# REQUEST FOR QUALIFICATIONS/PROPOSALS LEASE-LEASEBACK SERVICES FOR PACIFIC HIGH SCHOOL-CAREER TECHNICAL EDUCATION TRANSPORTATION TECHNOLOGIES BUILDING AND CAMPUS MODERNIZATION PHASE I - RFQ/P NO. 231

The San Bernardino City Unified School District Facilities Management, Planning ("District"), Development Department, is requesting proposals from those interested in providing LEASE-LEASEBACK SERVICES FOR PACIFIC HIGH SCHOOL-CAREER TECHNICAL **EDUCATION** TRANSPORTATION **TECHNOLOGIES BUILDING** AND **CAMPUS** MODERNIZATION PHASE I - RFQ/P NO. 231. Copies of the Request for Qualifications/Proposals ("RFQ/P") can be obtained via e-mail, in person at the address below, or from the District's web sites: www.sbcusdfacilities.com

http://sbcusd.com/district\_offices/business\_services\_division/Purchasing/. The proposals must be received at the address indicated below by no later than **June 27, 2019 by 3:00 P.M. Pacific Time.** All qualified providers of services described in the RFQ/P, including the local firms/individuals are encouraged to participate in this process and submit their proposals.

#### PREQUALIFICATION OF CONTRACTORS:

As a condition of submitting Qualifications/Proposals for this Project, and in accordance with California Public Contract Code section 20111.6, prospective contractors are required to submit to the District a completed set of prequalification documents on forms provided by the These documents will be the basis for determining which contractors are qualified to submit a proposal on this Project. Proposals will not be accepted if a Contractor has not been pregualified where prequalification is required. Prequalification documents are available from the San Bernardino City Unified School District Website http://sbcusdfacilities.com/transforming-schools/bid-opportunities/, and at Facilities Planning & Development Department, located at 956 West 9th Street, San Bernardino, CA 92411. Prequalification documents must be submitted to the Facilities Planning & Development Department by Monday, June 17, 2019. Contractors will be notified by e-mail, telephone, fax or by mail of their prequalification rating within a reasonable period of time after submission of their prequalification documents, but not less than five business days prior to the RFQ/P proposals due date.

The District, at its sole discretion, reserves the right to reject any proposals received after the deadline stated herein. The proposals must be submitted in a sealed envelope, addressed as indicated below, with the name and address of the respondent clearly printed in the upper, left corner. The envelope should be clearly printed: QUALIFICATIONS/PROPOSALS FOR LEASE-LEASEBACK SERVICES FOR PACIFIC HIGH SCHOOL—CTE AND MODERNIZATION—PHASE I RFQ/P NO. 231.

SAN BERNARDINO CITY UNIFIED SCHOOL DISTRICT
Facilities Management and Maintenance & Operations
Department
956 West 9<sup>th</sup> Street
San Bernardino, California 92411
Contact: Sherri Lien
(909) 388-6100
Sherri Jien@sbcusd.k12.ca.us



# REQUEST FOR PROPOSALS NO. 231 FOR LEASE-LEASEBACK SERVICES

for the

Pacific High School

Career Technical Education

Transportation Technologies Building and

Campus Modernization Phase I

1020 Pacific Street
SAN BERNARDINO, CALIFORNIA 92404

Proposal Deadline Date
June 27, 2019 at 3:00 PM

**Submit Proposal to:** 

San Bernardino City Unified School District 956 W. 9<sup>th</sup> Street San Bernardino, CA 92411 Attn: Sherri Lien



#### **REQUEST FOR PROPOSALS NO. 231**

# **Table of Contents**

l.	PURPOSE OF THE RFP:	. 3
II.	BACKGROUND ON THE PROJECT:	. 3
III.	LEASE-LEASEBACK CONTRACTOR PREQUALIFICATION	. 5
IV.	PRECONSTRUCTION SERVICES	. 6
V.	CONTRACT DOCUMENTS	. 6
VI.	RFP TIMELINE	. 7
VII.	QUESTIONS AND CLARIFICATION OF THE RFP	. 7
VIII.	COMMUNICATIONS RESTRICTIONS	. 8
IX.	DIR REGISTRATION AND PREVAILING WAGES	. 8
Χ.	SUBCONTRACTOR DESIGNATIONS	. 9
XI.	CONTENTS OF THE PHASE I PROPOSAL	10
XII.	PHASE II SUBMITTALS & INTERVIEW (To be submitted by SHORTLISTED FIRMS ONLY.)	17
XIII.	PREPARATION AND SUBMITTAL OF THE PROPOSAL	18
XIV.	PHASE I PROPOSAL EVALUATION	22
XV.	PHASE II EVALUATION CRITERIA AND BEST VALUE SCORE	24
XVI.	GENERAL TERMS AND CONDITIONS	25
RFP	ATTACHMENT 1 – FIRM QUESTIONNAIRE	27
RFP	ATTACHMENT 2A – Proposed Lease Leaseback Fee (for Phase I)	32
RFP	ATTACHMENT 2B – PRECONSTRUCTION SERVICES FEE PROPOSAL (for Phase II)	33
RFP	ATTACHMENT 3 – SAMPLE AGREEMENT FORMS	34



#### I. PURPOSE OF THE RFP:

- A. By way of this Request for Proposals ("RFP"), the San Bernardino City Unified School District ("District") seeks proposals from lease-leaseback contractors ("Contractor" or "Firm") to provide lease-leaseback construction services for the Pacific High School Career Technical Education Facility (CTE) and Campus Modernization Phase I ("Project").
- B. The purpose of this RFP is to obtain information that will enable the District to select a lease-leaseback Contractor using the "best value" competitive procurement process under Education Code section 17400 et seq. that can assist the District with lease-leaseback construction services for the Projects. The "best value" competitive procurement process is an evaluation process whereby a Firm is selected by the District on the basis of objective criteria for evaluating the qualifications of Firms, with the selected Firm representing the best combination of price and qualifications. Each Contractor responding to this RFP should be prepared and qualified to provide the lease-leaseback construction services described in this RFP to the District in an expeditious and timely manner and on relatively short notice so as to enable the District to meet critical time deadlines and schedules.
- C. The procurement of the Contractor under this RFP will be through a two-step process. Those interested Contractors may submit their Phase I proposals for evaluation as per "Evaluation Categories" Section XI (C) 1 through 13 of this RFP. The District's Proposal Evaluation Committee will score those proposals in accordance with Section XV "Proposal Evaluation and Best Value Score". The highest scoring firms (up to three firms) will then be invited to participate in Phase II, providing Guaranteed Maximum Price (GMP) fee proposals and interview with the District's selection panel. The best value Contractor will be determined in accordance with Section XV.

#### II. BACKGROUND ON THE PROJECT:

- A. San Bernardino City Unified School District is located in the Inland Empire and serves approximately 50,000 students at 74 campuses from across the City of San Bernardino, unincorporated portions of San Bernardino, and the City of Highland. The District has several facility renovation and new construction projects in the planning and design stages, including this project. This Project will be constructed using the lease-leaseback project delivery method authorized by Education Code section 17400 et seq.
- B. The District has independently contracted with Ruhnau Clarke Architects to develop a Master Plan for the Pacific High School campus and to be the Architect of Record for the High School Career Technical Education Facility (CTE) and Campus Modernization Phase I Project. Future phases of the work will likely include a new Administration Building, a new Performing Arts Center, possible additional Technical Education facilities and other campus modernizations and site improvements.



- C. The CTE design has recently been approved by the Division of the State Architect (DSA) and Phase I Modernization project design documents are currently under review and have not yet received final approval by the DSA for construction. During the Preconstruction phase of the project, the selected lease-leaseback Contractor will work with the Architect as directed by the District to complete the constructability review, value engineering, construction sequencing, scheduling, phasing, and various other preconstruction services as described in the contract documents.
- D. If the District elects to proceed with the Construction Phase of the effort, the selected Lease-Leaseback Contractor is to act as a General Contractor providing lease-leaseback construction services for various modernization and new construction efforts in accordance with the contrat documents for work at the Pacific High School campus located at 1020 Pacific Street, San Bernardino, California 92404.
- E. The scope of work (SOW) for the Pacific HS CTE and Modernization project may include but is not limited to:
  - 1. Providing all Preconstruction Services as detailed in contract documents (see RFP Section IV. and Attachment 3, CSA Exhibit "A1" Preconstruction Services Scope of Work),
  - 2. Construction of the new CTE facility of approximately 21,000 sq. ft. and installation of all specified equipment,
  - 3. Modernization of various classroom buildings and supporting structures as detailed in contract documents. These modernizations involve approximately 78,492 sq.ft. of existing campus facilities,
  - 4. The addition of approximately 7,920 sq. ft. to the gymnasium,
  - 5. Construction of miscellaneous site improvements,
  - Construction of various temporary facilities to support the project, including temporary fencing, temporary parking facilities for school staff, school visitors, and construction staff, rerouting of impacted irrigation lines, and temporary relocation of supporting systems (power, phones, data cabling, etc.)
- F. The estimated construction budget for this Project is \$40 million. The estimated period of performance for the entire Pacific HS CTE and Modernization project is a 23 month construction duration.
- G. The District is exploring the possibility of instituting an Owner Controlled Insurance Program (OCIP) to cover the project. Currently, proposers should plan on providing the insurance coverages as detailed in this RFP. Should the District deem that an OCIP program would prove a better financial and risk-management option, an RFP Addendum will be issued



H. Contractor is to invoice District on a monthly basis commencing 30 days after both party's execution of contract documents and District issuance of Notice to Proceed. Monthly invoices are to be calculated by dividing total contract price by 23 equal monthly installments to cover the construction duration and lease back period, less a 5% retention. Invoicing for retention payment shall be allowable after receipt of the Certificate of Occupancy and acceptable completion of any/all "punchlist" items.

#### III. LEASE-LEASEBACK CONTRACTOR PREQUALIFICATION

- A. Lease-Leaseback Contractor Prequalification *All Firms submitting a proposal to this RFP must be prequalified with the District pursuant to Public Contract Code section 20111.6 (b)-(m) without exception prior to submitting a proposal.* Any Firm that submits a proposal and is not prequalified will be deemed non-responsive and that Firm's proposal will be rejected and returned to the Firm unopened.
- B. Prequalification documents are available from the San Bernardino City Unified School District website at www.sbcusdfacilities.com, and at 956 W. 9th Street, San Bernardino, CA 92411. Contractor Prequalification documents must be submitted by Contractor by February 4, 2019 at Noon. Contractors will be notified by telephone, email, or mail of their prequalification rating within a reasonable period of time after submission of their prequalification documents, but not less than five business days prior to the proposal submission deadline. Contractors holding a currently valid notice of successful prequalification with the District for lease-leaseback projects will not need to submit another prequalification package. Provide a copy of your current (2019) prequalification status letter from the District with your proposal.
- C. Proposal submissions require only <u>Prime Contractor</u> prequalification. Sometime <u>after award</u> of a Lease-leaseback contract, the selected Contractor will be required to prepare a Guaranteed Maximum Price (GMP) proposal for the construction of the project. At that time, any/all mechanical, electrical or plumbing ("MEP") subcontractors (defined as contractors that hold a C-4, C-7, C-10, C-16, C-20, C-34, C-36, C-38, C-42, C-43 or C-46 license) must also be prequalified with the District. This prequalification requirement applies even if the subcontractor will perform, or is designated and identified to perform, work that does not require one of the licenses listed above, but the subcontractor holds one of the licenses listed above.
- D. MEP subcontractors (as defined above) must be prequalified prior to the award of their respective subcontract. A list of prequalified MEP subcontractors will be made available by the District to the selected Contractor upon request, but not less than five business days prior to the GMP proposal submission deadline. However, it is the responsibility of the selected Contractor to ensure that all MEP subcontractors holding any of the licenses listed above are properly pregualified.



#### IV. PRECONSTRUCTION SERVICES

- A. The District anticipates that the successful Contractor will provide preconstruction services including, but not limited to, reviewing the Project's plans and specifications during the design of the Project to identify and note all deficiencies, incongruities and inconsistencies that may affect constructability of the Project including, but not limited to, design and specification omissions, incomplete and/or inconsistent plans, details and specifications, and any lack of coordination, together with all other appropriate, necessary and/or required services to facilitate and prepare for the successful development and construction of the Project.
- B. The preconstruction services will also include, but not be limited to, the following tasks: design meeting with the architects and engineers; review and validation of estimates prepared by the architect; preparation of a master critical path method schedule for the Project; preparation of cost estimates based on the final construction documents, including allowances, contingencies, general conditions, costs and fees; constructability reviews; value engineering; construction planning and phasing, and cost proposal strategies all with the goal that the DSA-approved plans and specifications for the Project will be complete such that the Project can be constructed by a competent licensed general building contractor in strict accordance with the DSA-approved plans and specifications without change orders, delays, or additional charges to District. (See also Attachment 3, CSA Exhibit "A1" Preconstruction Services Scope of Work for additional detail or required preconstruction services.)
- C. The successful Contractor shall not provide any preconstruction services or work that requires a contractor's license pursuant to Business and Professions Code section 7065 et seq.

#### V. CONTRACT DOCUMENTS

Contract documents may be viewed and ordered through the Crisp Imaging PlanWell Service online by clicking on 'PUBLIC PLANROOM' at www.crispimg.com after Thursday, May 16, 2019. There is a refundable deposit of one hundred dollars (\$100.00) for each set of proposal documents and specifications, upon payment by cashier's or company check made payable to San Bernardino City Unified School District. Prospective Proposers may secure up to two sets of documents. Eligible deposits will be refunded upon return of said documents to Crisp Imaging in good acceptable condition within five (5) business days after proposals are evaluated. Proposers in need of more than two sets of proposal documents may purchase at their own cost based on Crisp Imaging's current rates at that time. Crips Imaging contact information is as follows:

Crisp Imaging

3180 Pullman Street

Costa Mesa, CA 92626

Phone: (866) 632-8329

Public Plan Room: www.crispimg.com



#### VI. RFP TIMELINE

Request for Proposals Issued	May 30, 2019
Mandatory Project Site Walk-Through	June 11, 2019 at 9:00am
Deadline for Submittal of Question	June 18, 2019
Contractor Prequalification Due	June 17, 2019, by 3:00 pm
Responses to the Questions Submitted	June 21, 2019
Due Date for Submittal of Proposals	June 27, 2019 by 3:00 pm
Notice to Shortlisted Firms	July 11, 2019
Short List Interviews	July 22-26, 2019
Anticipated Notice of Intent to Award	July 30, 2019
Anticipated Board Approval of Award of Preconstruction	Contract August 6, 2019
Development of GMP Proposal	August 7 – Sept 16, 2019
Subcontractor Prequalification Due	September 9, 2019
Due Date for GMP Fee Proposals	September 17, 2019
Anticipated Board Approval of GMP	October 15, 2019
Anticipated Construction Start	October 28, 2019

(Note: Timeline is estimated, all dates are subject to revision at the District's discretion.)

#### VII. QUESTIONS AND CLARIFICATION OF THE RFP

- A. All questions, requests for explanation or clarifications of any kind in regard to this RFP shall be made in written form submitted via email to Juan Cantoran, at <a href="mailto:icantoran@ruhnauclarke.com">icantoran@ruhnauclarke.com</a> and a cc to <a href="mailto:sherri.lien@sbcusd.k12.ca.us">sherri.lien@sbcusd.k12.ca.us</a>; by no later than 3:00 p.m., June 7, 2019. Email header line must contain the text, "Question re: RFP No.231".
- B. A response will not be provided to any questions, requests for explanation or clarifications received after the specified date and time. All addenda and clarifications will be posted on the District's website, www.sbcusdfacilities.com, and provided to those Firms that have registered with the District.
- C. Any interpretation, clarification, or correction of this RFP will only be made by addendum as noted above. No person or Firm is authorized to make any oral interpretation of any provision in this RFP, nor shall any oral interpretation be binding on the District.



#### VIII. COMMUNICATIONS RESTRICTIONS

- A. To ensure a level playing field with an open and uniform competitive process, Respondents and their Representatives must adhere to restrictions on unallowable communications set below:
- B. Prohibited Communication Examples of prohibited communication by Respondents and their Representatives include, but are not limited to:
  - Contact of SBCUSD Officials, including members of the department initiating a contract, or members who will serve on an evaluation team for any contract information that is not uniformly available to all other bidders, proposers or contractors;
  - 2. Contact of SBCUSD Officials, including Board Members and their staff, to lobby on any aspect relating to a contract matter under consideration, negotiation, protest or dispute;
  - 3. Contact of SBCUSD Officials in the particular department requesting a competitive contract to discuss other business or partnership opportunities.
- C. Exceptions The following are exceptions to the Communication Restrictions:
  - 1. Open and uniform communications which are made as part of the procurement process such as requests for clarification or information made in writing, under the terms expressly allowed for in the RFP document; (ii) interviews or presentations to evaluation committee members which are part of the procurement process;
  - 2. Negotiations with SBCUSD's designated negotiation team members;
  - 3. Protests which follow the process outlined by SBCUSD's protest policies and procedures; and
  - 4. Requests for technical assistance approved by SBCUSD contract officials. (For example, questions relating to SBCUSD's LBOP Program.)

#### IX. DIR REGISTRATION AND PREVAILING WAGES

A. DIR Registration. Contractors and their subcontractors (of any tier) shall not be qualified to submit or be listed on a proposal, or engage in the performance of any contract for public work, as defined in the Labor Code, unless currently registered and qualified to perform public work pursuant to Section 1725.5 of the Labor Code. It is not a violation of this section for an unregistered contractor to submit a proposal that is authorized by Section 7029.1 of the Business and Professions Code or by Section 10164 or 20103.5 of the Public Contract Code, provided the contractor is registered to perform public work pursuant to Section 1725.5 of the Labor Code at the time the contract is awarded.



B. Prevailing Wages. The Contractor and all subcontractors shall comply with the requirements set forth in Division 2, Part 7, Chapter 1 of the Labor Code. Pursuant to Labor Code section 1770 et seq., the District has obtained from the Director of the Department of Industrial Relations the general prevailing rate of per diem wages and the general prevailing rate for holiday and overtime work in the locality in which this work is to be performed for each craft, classification or type of worker needed to execute the contract. Copies are available from the District to any interested party on request and are also available from the Director of the Department of Industrial Relations.

#### X. SUBCONTRACTOR DESIGNATIONS

- A. Pursuant to Education Code section 17406(a)(4)(A), the District is **NOT** requiring all Firms to identify and designate the subcontractors in their proposals. However, after award of the lease-leaseback contract for the Project, and in accordance with Education Code section 17406(a)(4)(B), any subcontractor whose subcontract value exceeds one-half of one percent of the price allocable to the total construction work on the project must be awarded a subcontract in accordance with the following process:
  - 1. Provide public notice of availability of work to be subcontracted in accordance with publication requirements applicable to the competitive bidding process of the District, including a fixed date and time on which qualifications statements, bids, or proposals will be due.
  - 2. Establish reasonable qualification criteria and standards.
  - 3. Award the subcontract either on a best value basis or to the lowest responsible bidder.
- B. The process above may include prequalification or short-listing. Subcontractors awarded subcontracts as set forth above shall be afforded all the protections of the Subletting and Subcontracting Fair Practices Act (Public Contract Code section 4100 et seq.)
- C. All subcontractors (of any tier) performing any portion of the Work must comply with Labor Code sections 1725.5 and 1771.1 and must be properly and currently registered with the California Department of Industrial Relations and qualified to perform public works pursuant to Labor Code section 1725.5 throughout the duration of the Project.



#### XI. CONTENTS OF THE PHASE I PROPOSAL

- A. Firms must submit one (1) original, five (5) hard copies and one (1) digital copy (provided on a CD, DVD or USB drive) of the proposal. All proposals should address the requested information for each of the evaluation categories below. All proposal responses should be provided in 10 point font or greater, tables and captions excluded. Please note page-count limits for responses as identified in each section. Double sided printing is preferred, with each printed side counted as a page. Pages shall be 8 ½" x 11", although the use of fold-out pages (11" x 17") is allowable, but fold-outs will be counted as two pages for each printed side). Responses should be in arranged in the order as sections/topics are presented in this RFP.
- B. The proposal shall demonstrate the qualifications, competence, and capacity of the Firm. The proposal should contain the following:
  - 1. Cover Letter/Letter of Interest Include a cover letter, addressed to Tom Pace, stating the eligibility of the Firm to respond to this RFP, a brief description and history of the Firm, and a statement of interest. (1-2 pages)
  - 2. Table of Contents The table of contents shall reflect the order stated herein and shall include section titles and page numbers. (1-2 pages)
  - 3. Mandatory Requirements The proposal requirements as presented in Section XII. C.2-5 below are mandatory and must be satisfied. The mandatory requirements will be scored on a pass/fail basis. Failure to meet any one of the mandatory requirements specified in this Section XI.(C)(1 through 4) will disqualify your Firm from any further consideration for this RFP.
  - 4. Scored Evaluation Categories RFP items found in Section XI.(C)(5 through 13) and Section XI.(C) constitute the scored elements of the phase I proposal. Responses will be scored using point valuations as presented in RFP Section XV.B.2

#### C. Evaluation Categories

- 1. Contractor Responsibility (Not to exceed 2 pages) Identify if your Firm has ever had the following (items 1-6) occur in the past seven (7) years. For the purposes of this paragraph, "Firm" shall include any present or past (over the last seven years), officers, owners, principals, partners, or any qualifying individuals including any Responsible Managing Employee (RME) or Responsible Managing Officer (RMO). Any occurrence of the following in the past seven (7) years shall render the Firm not qualified to submit a proposal:
  - a) Found to be a non-responsible contractor by any public agency;
  - b) Convicted for false claims;
  - c) Firm's license revoked or suspended;
  - d) Debarred or otherwise ineligible to bid on or be awarded a public works contract;



- e) Terminated for cause or defaulted on a construction contract; or
- f) Convicted of a crime involving the awarding of a construction contract, or the bidding or performance of a construction contract.
- 2. License Requirements (1 page) Pursuant to Business and Professions Code section 7028.15 and Public Contract Code section 3300, the Contractor must possess a California Contractor's Class "B" license at the time of submittal of its proposal, and for the duration of the contract, if awarded. Subcontractors must possess the appropriate license for the work to be performed on the Project.
- 3. Performance and Payment Bonds (Not to exceed 2 pages) All Firms submitting a proposal to this RFP must be able to provide separate faithful payment and performance bonds, each in an amount equal to 100% of the total contract amount. All bonds must be issued by a California admitted surety as defined in California Code of Civil Procedure section 995.120. Firms must provide a letter from their surety indicating the Firm's current and overall bonding capacity, and the ability to meet the bond requirements in Section 35.13 of the Construction Services Agreement.
- 4. Insurance Requirements (Not to exceed 4 pages) All Firms submitting a proposal to this RFP must have the ability to meet all of the insurance requirements set forth in Section 35.4 of the Construction Services Agreement. Firms must include a copy of their current certificate of insurance in their proposals evidencing the following minimum insurance requirements:
  - a) Comprehensive general liability insurance with a combined single limit per occurrence of not less than \$2,000,000.00 or commercial general liability Insurance (including automobile insurance) which provides limits of not less than:
  - b) Per occurrence (combined single limit): \$1,000,000.00
  - c) Project Specific Aggregate (for this Project only): \$2,000,000.00
  - d) Products and Completed Operations: \$1,000,000.00
  - e) Personal and Advertising Injury Limit: \$1,000,000.00
  - f) The following special hazards shall be covered by riders or riders to above mentioned public liability insurance or property damage insurance policy or policies of insurance, in amounts as follows:
  - g) Automotive and truck where operated: \$1,000,000.00
  - h) Material Hoist where used: \$1,000,000.00
  - i) Explosion, Collapse and Underground (XCU coverage): \$1,000,000.00



- j) Excess Liability Insurance coverage in the amount of Five Million Dollars (\$5,000,000.00).
- k) Workers' Compensation Insurance shall be provided in accordance with the provisions of the California Labor Code adequate to protect any person, firm, or corporation employed directly or indirectly in connection with the work of the Contractor from claims under Workers' Compensation Acts which may arise for operations, whether such operations be by any person, firm, or corporation, employed directly or indirectly by the Contractor upon or in connection with the work.
- I) Sexual Abuse/Molestation coverage shall be carried by Contractor and maintained during the term of this AGREEMENT a policy with a limit of not less than \$1,000,000 Sexual Abuse Injury Limit of Insurance. The CONTRACTOR shall hold the DISTRICT harmless from any and all claims for injury, damage, and loss.
- m) Provide a letter from your insurance broker stating that you will be able to provide and meet the insurance requirements in Section 35.4 of the Construction Services Agreement.
- 5. Firm Overview, Personnel, Capacity, and Methodology Attachment 1. Each Firm must completely answer all questions in Attachment 1 of the RFP. Provide a narrative stating the following:
  - a) Description of Firm (Not to exceed 2 pages) Include a description of the Firm's qualifications for providing preconstruction and lease-leaseback services on California school construction projects. Include information regarding the size of the Firm, location of the office from which the required services will be performed, nature of all work performed, and the number of years in this particular business. The Firm shall provide an affirmative statement that it is independent of the District as defined by generally accepted standards.
  - b) Firm's Personnel and Staffing Resources (Not to exceed 20 pages for section) Submit resume(s) or profiles for each key staff who will be proposed to provide the requested services, including their qualifications and recent relevant experience providing similar services. Each resume shall include, without limitation, the following information; (a) education; (b) years of relevant experience; (c) professional registrations, certifications and affiliations (d) project-specific experience with focus on public works projects and emphasis on K-12 projects providing preconstruction and lease-leaseback services, including dates and durations of each project listed and the name of the firm where employed. Include a discussion on the Firm's philosophy and approach for providing outstanding customer service.



- c) Capacity & Methodology (Not to exceed 4 pages) Provide an overview as to how the Firm will provide services and fulfill the requirements and expectations of the District and this RFP. Use this section to address the ability of your Firm to undertake and accomplish the required scope of services while meeting deadlines, the Firm's record of meeting schedules and deadlines of other clients, advantages over other firms in the same industry, strength and stability as a business, and supportive client references. Describe the Firm's ability to provide preconstruction and lease-leaseback services exclusively and in a timely manner for the District and the Firm's commitment to providing experienced personnel assigned to District's Project.
- d) Litigation (Not included in page count) Furnish and provide specific information on any termination for convenience, litigation settled or judgments entered within the last five (5) years, as well as any civil judgments within the last five (5) years. Identify if the Firm or any employee of the Firm is a party to an existing dispute with an owner, or owner's consultants, related to any project for which the Firm provided construction services. If so, please describe the nature of the dispute and its anticipated outcome.
- e) Identify if the Firm has ever filed a petition for bankruptcy. If so, please provide the date the petition was filed and identify the jurisdiction in which it was filed.
- 6. Relevant Experience and Past Performance (Not to exceed 6 pages)
  Description of past performance and related experience. Each Firm is required to submit a list of its most relevant preconstruction and lease-leaseback services provided in the past five (5) years that are of the approximate size of the Project described in the RFP. The list shall include:
  - a) A description and size of the project,
  - b) Scope of the work,
  - c) Dates services were performed for preconstruction services and for lease-leaseback services,
  - d) Total price for the project (please state amounts separately for preconstruction services and for lease-leaseback services and include the final guaranteed maximum price and all contingencies and allowances),
  - e) Owner's name, address, and phone number.
- 7. Preconstruction Phase Services (not to exceed 8 pages). Describe your methodology in providing Preconstruction phase services for the Project, specifically addressing the following:
  - a) Preconstruction Services Staffing
  - b) Planning of construction sequencing, phasing and logistics.



- c) Constructability review process, providing examples of constructability review that you preformed that resulted in the identification of significant design conflicts or omissions. Describe experience in using Bluebeam software in the plan review and markup process.
- d) Value engineering process, providing examples which resulted in savings of significant savings of time or money.
- e) Construction schedule development and adherence. Provide insight on firm's record of meeting schedules and deadlines of other clients.
- f) Construction estimating process used.
- g) Building Information Modeling (BIM): State whether you firm has BIM capability and describe your use of BIM on prior projects.
- h) Describe your program(s) for trade contractor outreach and soliciting subcontractor bids.
- 8. Construction Phase Services (not to exceed 8 pages). Describe your methodology in providing Construction phase services for the Project. Address the ability and planned approach of the respondent to accomplish the required construction phase scope of services, including but not limited to:
  - a) Construction schedule development and schedule adherence. Provide insight on firm's record of meeting schedules and deadlines of other clients;
  - b) Construction phase project staffing.
  - c) Construction quality.
  - d) Construction Field Office space.
  - e) Construction staging and logistics.
  - f) Managing K-12 construction efforts and DSA interface, including DSA Close-out process. Demonstrate familiarization with DSA, PR 13-01, "BOX" Duties
  - g) Documenting as-builts,
  - h) Building and systems commissioning,
  - i) Firm's approach to the challenges of the LLB delivery method.
  - j) Provide evidence that the firm is a licensed general contractor in the State of California in good standing (B license).
  - k) Describe advantages or differentiators over other entities providing similar services.



- I) Note: Information regarding the items above should be furnished for both the respondent and any sub consultants/sub-contractors included in the Phase I proposed team.
- 9. Proposed Timeline (not to exceed 2 11X17 pages pages). Provide a proposed project schedule in Gantt chart format, indicating major subtasks and proposed durations addressing but not limited to the elements as listed below. (Schedule may be provided in 11" x 17" fold-out format.)
  - a) Provide an outline of general project phasing plan
  - b) Construction Planning, Phasing and Logistics
  - c) Construction Procurement
  - d) Construction Mobilization
  - e) Temporary Facilities Construction
  - f) Construction (Modernization and New Construction)
  - g) Commissioning
  - h) Project Closeout
- 10. Labor Compliance/Skilled and Trained Workforce (Not to exceed 4 pages)
  - a) Describe your ability to comply with statutory requirements for the payment of prevailing wages, including the monitoring and enforcement of your subcontractor's payment of prevailing wages. Provide copies of any DIR Civil Wage and Penalty Assessment issued against your Firm, explain the circumstances for the Civil Wage and Penalty Assessment, and the final resolution.
  - b) Further describe your plan and methodology to comply with the requirements for the use of a "skilled and trained workforce" as defined in Education Code section 17407.5 and Public Contract Code section 2600 et seq., for each apprenticeable occupation that will be used on the Project, including all subcontractors of any tier. Include in your discussion your plan and methodology to comply with the percentage requirements for the use of "skilled journeypersons" for each apprenticeable occupation and the required monthly report demonstrating compliance. Please include a copy of a sample monthly report prepared by your firm for another owner if available. (Sample is not included in page count.)
  - c) Finally, identify and discuss which apprenticeable occupation(s) will be the most difficult to meet the percentage requirements for skilled journeypersons on the Project and state why.
- 11. Safety (Not to exceed 4 pages)



Discuss your plan to maintain a safe worksite, especially regarding work on an active school campus. In your discussion, include whether your Firm has an Injury and Illness Prevention Program that complies with 8 CCR § 1509, whether your Firm has a safety program that meets Cal/OSHA requirements, and whether your Firm will provide a full-time person dedicated to safety on the Project.

- a) Please state whether you have had any accidents in the past five years that resulted in a construction fatality on any of your projects and provide any details for each incident.
- b) Please state whether you have had any recordable injuries in the past five years and provide the average total recordable injuries for the past five years.
- c) Please provide an EMR verification from the State of California or an insurance company for each of the past five years (2018, 2017, 2016, 2015, and 2014).
- 12. Local Business Outreach and Participation (Not to exceed 3 pages)

The District is vitally interested in promoting the growth of small and local businesses within the boundaries of the District and the City of San Bernardino by means of increasing the participation of these businesses in the District's purchase of goods and services. The District has a goal of ten (10) percent of all contracts for these services be awarded to local businesses. A locally-owned business, for purposes of satisfying the locality requirements of this provision, is one which holds a valid business license issued by the city where the District is located or the city or cities where the District has schools. Describe the Firm's plan for inclusion of local businesses in the services to be provided for the District.

13. Exceptions to the Lease-Leaseback Agreement – (Not in page count)

The form of the Lease-Leaseback Agreement (Site Lease, Sublease, and Construction Services Agreement) is attached to this RFP as Attachment 3. Please review each agreement and provide any proposed exceptions to those agreements on Attachment 1, Firm Questionnaire, Section E.

D. Phase I Subcontractor Mark-up Fee Proposal

Proposers shall submit Subcontractor Mark-up (Lease-Lease Back Fee) proposals in format specified in RFP "Attachment 2A". Subcontractor Mark-up Fee proposals shall include the following:

1. Proposed Subcontractor Mark-up (Lease-Lease Back Fee) – State proposed percentage (%) mark-up to be applied on all subcontracts. Said mark-up is to include Contractor overhead and profit on all construction trade labor and materials. Submit proposed fee as per RFP Attachment 2A.



#### XII. PHASE II SUBMITTALS & INTERVIEW (To be submitted by SHORTLISTED FIRMS ONLY.)

- A. Firms selected to participate in Phase II of the selection process (shortlisted firms) may be asked to interview with the District's Selection Panel and to submit additional fee proposals (see item C. below).
- B. Interview of Shortlisted Firms

Following review of the Phase I proposals received, the District may request any or all proposing respondents to participate in a presentation and/or interview in regards to their proposal. If interviews are conducted, the interview will consist of a presentation (15 minutes) followed by a question and answer period (up to 45 minutes). The District reserves the right to conduct negotiations with any number of respondents, as determined by the District, for entering into contract agreements.

- C. GMP Fee Proposals (To be submitted by **SHORTLISTED FIRMS ONLY**.)
  - 1. Proposed Preconstruction Services Fee State Guaranteed Maximum Price per month to provide preconstruction services as detailed in this RFP and Statement of Work. Preconstruction services fee shall include any/all "general conditions" items or services provided during the preconstruction phase of the project. Submit proposed fee as per RFP Attachment 2B.
  - 2. Proposed General Conditions Fee State Guaranteed Maximum Price (GMP) to provide construction phase General Conditions staffing, services, materials, furnishings and equipment. Submit proposed fee as per RFP Attachment 2B. The proposed fees should include all direct labor costs, fringe benefits, insurance, overhead, profit, and all other expenses the Contractor will incur in providing the lease-leaseback construction services. General Conditions GMP shall include all of the following:
    - a) Project Staff to include: Site Superintendent (full time), Foreman, Project Manager, Project Engineer, Administrative/Accounting, support staff (Scheduler, Estimator, etc.) and Safety Representative.
    - b) Safety Planning, Training, Oversight & Monitoring
    - c) Site Signage, Public Relations & Ceremonies
    - d) Temporary Fencing & Barricades
    - e) Office and Site Equipment Rental
    - f) Engineering & Layout
    - g) CPM Schedule development and updates
    - h) Safety & First Aid Supplies, Fire Extinguishers, misc. equipment
    - i) Ceremonies/PR/Site Signage
    - j) Temporary Toilets
    - k) Cell Phones/Internet Service/Copier(s)/Printer(s)/Office Supplies
    - 1) Temporary Water



- m) Temporary Power
- n) As Builts & Plans
- o) Temporary Office field office and furnishings
- p) Temporary Office and furnishings for Inspector of Record
- q) Materials Storage
- r) Site Vehicles and Fuel
- s) Equipment Maintenance
- t) Regular Site & Office Clean Up
- u) Street Cleaning/Sweeping/Rumble Plates
- v) Waste receptacles (dumpster(s) and regular pick up
- w) Storm Water Pollution Prevention Plan (SWPPP) development, implementation, upkeep and monitoring
- x) Construction site security

#### D. Interview of Shortlisted Firms

Following review of the Phase I proposals received, the District may request any or all proposing respondents to participate in a presentation and/or interview in regards to their proposal. The District reserves the right to conduct negotiations with any number of respondents, as determined by the District, for entering into contract agreements.

#### XIII. PREPARATION AND SUBMITTAL OF THE PROPOSAL

- A. Phase I Proposal Submittal and Deadline One (1) original, five (5) hard copies and one (1) digital copy (provided on a CD, DVD or USB drive) of the proposal must be submitted under sealed cover by no later than 3:00 p.m. on June 27, 2019. Label the outside of the sealed proposal envelope or box with your company name, proposal title and RFP deadline.
- B. Phase I Proposals shall be delivered to the attention of:

Sherri Lien San Bernardino City Unified School District 956 W. 9th Street San Bernardino, CA 92411

C. It is the sole responsibility of the Firm submitting the proposal to ensure that its proposal is actually received in the office prior to the deadline time and due date. Unless this RFP is extended by a written amendment, proposals received after the time on the due date will not be considered. Faxed or emailed proposals will not be accepted.



- D. Phase II GMP Fee Proposal Submittal and Deadline (*for shortlisted firms only*) One (1) original and five (5) hard copies of the GMP Fee Proposal must be submitted under sealed cover by no later than 3:00 p.m. on July 22, 2019. Label the outside of the sealed proposal envelope or box with your company name, proposal title and RFP deadline.
- E. Phase II GMP Fee Proposals shall be delivered to the attention of:

Sherri Lien San Bernardino City Unified School District 956 W. 9th Street San Bernardino, CA 92411

- F. It is the sole responsibility of the Firm submitting the proposal to ensure that its proposal is actually received in the office prior to the deadline time and due date. Unless this RFP is extended by a written amendment, proposals received after the time on the due date will not be considered. Faxed or emailed proposals will not be accepted.
- G. Proposal Completeness Proposals (whether competing in Phase I or both Phase I & II) shall be completed in all respects as required by the instructions herein. A proposal may be rejected if it is conditional or incomplete, or if it contains alterations of form or major irregularities of any kind as determined by the District. A proposal will be rejected if, in the opinion of the District, the information contained therein was intended to mislead the District in the evaluation of the proposal.
- H. District Not Responsible For Preparation Costs All costs incurred in the preparation, submission and/or presentation of Firms responding to the RFP including, but not limited to, the Firm's travel expenses or long distance charges to attend any preconferences, presentations, interviews, and negotiation sessions, shall be the sole responsibility of the Firm and will not be reimbursed by the District. The District shall not pay for any costs incurred for proposal or contract preparation as a result of termination of this RFP or termination of the contract resulting from this RFP.
- I. Right to Use Ideas All proposals and other materials submitted become the property of the District. District reserves the right to use any ideas presented in any response to the RFP. Selection or rejection of the proposal shall not affect this right.
- J. Modification or Withdrawal Of Proposal A Firm may modify or withdraw a proposal after submission by written request of withdrawal and re-submission, provided that the proposal withdrawal is prior to the proposal deadline specified.
- K. Amendments Firms are advised that the District reserves the right to amend this RFP at any time. Amendments will be done formally by providing written amendments to all potential Firms known to have received a copy of the RFP and/or by publishing the amendment on the District's website.



- L. Equal Opportunity The Firm shall certify that it is an Equal Opportunity Employer and has made a good faith effort to improve minority employment and agrees to meet federal and state guidelines. Legal residents of the United States of America shall be used in providing all services under this RFP.
- M. Firm shall not discriminate nor permit discrimination against any person because of race, color, religion, age, national origin, ancestry, creed, handicap, sexual orientation, or union membership in the performance of the work, including but not limited to preparation, manufacturing, fabrication, installation, erection and delivery of all supplies and equipment. In the event of receipt of such evidence of such discrimination by the Firm or its agents, employees or representatives, District shall have the right to rescind and terminate the contract.
- N. The successful Firm agrees to include the paragraph (M.) above with appropriate adjustments in all subcontracts, which are entered into for work to be performed pursuant to the contract.
- O. Waiver or Breach Thereof No term or provision of this RFP shall be deemed waived and no breach excused, unless such waiver or consent shall be in writing and signed by an individual authorized to so waive or consent. Any consent by either party to, or waiver of, a breach by the other, whether express or implied, shall not constitute a consent to, waiver of, or excuse for, any other breach or subsequent breach, except as may be expressly provided in the waiver or consent.
- P. Covenant Against Gratuities The Firm warrants that no gratuities (in the form of entertainment, gifts, or otherwise) were offered or given by the Firm or any agent or representative of the Firm, to any officer or employee or consultant of the District with a view toward securing the resultant contract or securing favorable treatment with respect to any determinations concerning the award of the contract. For breach or violation of this provision, the District shall have the right to terminate any negotiation or the resultant contract, either in whole or in part, and any loss or damage sustained by the District in procuring on the open market any items which the Firm agreed to supply shall be borne and paid for by the Firm. The rights and remedies of the District provided in this clause shall not be exclusive and are in addition to any other rights and remedies provided by law.
- Q. Indemnification and Insurance The Firm, at its own expense and without exception, shall indemnify, defend and pay all damages, costs, expenses, including attorney fees, and otherwise hold harmless the District, its employees and representatives, from any liability of any nature or kind in regard to the delivery of these services. (See, Construction Services Agreement Section 35.4 for insurance requirements and Section 36 for hold harmless and indemnify requirements.)



- R. Conflict of Interest The Firm is in agreement that it presently has no interest and will not acquire any interest, directly or indirectly, which would conflict in any manner or degree with the performance of the services hereunder. The Firm further agrees that no person having any such known interest or conveyed an interest shall be employed, directly or indirectly, in the delivery of services under this RFP.
- S. Independent Contractor The Firm represents itself as an independent contractor offering such services to the general public and shall not represent him/herself or his/her employees to be an employee of the District. Therefore, the Firm shall assume all legal and financial responsibility for taxes, FICA, employee fringe benefits, workers compensation, employee insurance, minimum wage requirements, overtime, and other expenses.
- T. Precedence of Documents
  - 1. The contract between the District and the successful Firm(s) shall consist of (1) this Request for Proposals (RFP) and any amendments thereto, (2) the Agreements included herein to be executed with the successful Firm(s); and (3) the proposal submitted by the Firm to the District in response to the RFP. In the event of a conflict in language between the documents referenced above, the provisions and requirements set forth and/or referenced in the Agreement shall govern. However, the District reserves the right to clarify any contractual relationship in writing with the concurrence of the Firm, and such written clarification shall govern in case of conflict with the applicable requirements stated in the RFP or the Firm's proposal. In all other matters not affected by the written clarification, if any, the RFP shall govern.
  - 2. In case of a difference between RFP documents, contract documents, figures, specifications, and cited environmental surveys and reports, the order of precedence shall be as follows
    - a) Contract Agreement,
    - b) Technical Specifications,
    - c) RFP Instructions to Bidders,
    - d) Project drawings (figures)
    - e) Project technical reports (Phase I Environmental Report and Hazardous Materials Survey Report(s), Underground Utility Survey Report(s), Topographical Survey Report(s), Geotechnical and soil boring reports, etc.)
    - f) The proposal submitted by the firm to the District in response to the RFP



- 3. In case of discrepancy in the figures, the RFP text, in the specifications, project related reports, or in the contract text, the matter shall be promptly submitted to the District's Project Representative, who shall promptly make a determination in writing. Any adjustment by the Contractor without such a determination shall be at his own risk and expense.
- U. Compliance with Laws In connection with the furnishing of services or performance of work under this RFP, the Firm agrees to comply with the Fair Labor Standards Act, Equal Opportunity Employment Act, and all other applicable federal and state laws, regulations and executive orders to the extent that the same may be applicable.

#### XIV. PHASE I PROPOSAL EVALUATION

A. Proposal Evaluation Committee –

The District's Proposal Evaluation Committee will consist of at least three (3) members and will score each proposal based on the evaluation categories and points set forth in the RFP. (See Section XVI.B. below). Each Firm's proposal will be evaluated and scored only on the information that is included in the Firm's proposal. If any information is missing or incomplete in your proposal, you will not be provided the opportunity to supply the missing or incomplete information, nor will the District seek clarification of any information included in the proposals. Each proposal must be capable of being evaluated independently based solely on the information contained in the proposal.

- B. Evaluation Categories, Points, and Scoring
  - 1. Each member on the Proposal Evaluation Committee will independently score each proposal and each Firm's score will be equal to the average score from the Evaluation Committee (i.e., the total number of points from the entire Proposal Evaluation Committee divided by the number of Proposal Evaluation Committee members: initial score = sum total of points/number of committee members). The initial score will be calculated to two decimal places. Although the Proposal Evaluation Committee will independently score each proposal, the members reserve the right to discuss the RFP process and information in any proposal with other members.
  - 2. The RFP contains eleven (11) Evaluation Categories, as discussed in Section VIII, and the maximum number of points for each category is shown in the table below. There are 1000 possible points in each phase.

(See table next page)



Phase I EVALUATION CATEGORIES	POINTS
Mandatory Requirements	Pass/Fail
Firm Personnel, Capacity, and Methodology	200
Relevant Experience and Past Performance	200
Proposed Subcontractor % Mark-up	200
Labor Compliance/Skilled and Trained Workforce	50
Safety	200
Local Business Outreach and Participation	50
Acceptance or Exceptions to Preconstruction/LLB Agreements	100
MAXIMUM TOTAL PHASE I SCORE	1000
Phase II EVALUATION CATEGORIES	scored after shortlist
GMP Fee Proposals	400
Interview - Project Team and Assignments	300
Interview - Responses to Questions	300
MAXIMUM TOTAL PHASE II SCORE	1000



#### XV. PHASE II EVALUATION CRITERIA AND BEST VALUE SCORE

- A. The selection panel's scoring of the shortlisted Firms will be based on the panel's view of which proposing Contractor provides the best value to the District and Stakeholders. Key points for the Selection panel's evaluation of Phase II submittals and interview will be:
  - 1. Project Team and Assignments Strength and coverage of proposed project team. (300 points)
  - 2. Responses to Interview Questions Evaluation of responses to interview panel's questions. (300 points)
  - 3. Best Value / Project Budget Provides overall best value to District (400 points) as evidenced by Lease-Leaseback Fee, Preconstruction Services Fee, and General Conditions Fee Proposals
- B. The total final scores will be used to determine the Best Value Scores (as defined in Education Code section 17400). Best Value Score = sum total of points/number of committee members. A Best Value Score will not be calculated for those Firms not invited to interview with the Proposal Evaluation Committee.
- C. Following completion of Phase II Shortlisted Firms' presentations to the selection panel, the panel will forward their award recommendation to the SBCUSD Board of Education. The SBCUSD Board of Education reserves the right to make the final decision as to which Contractor will be ultimately selected. The District also reserves the right to reject as unsuccessful any or all entries.
- D. The District reserves the right to conduct negotiations with any number of respondents, as determined by the District, for entering into contract agreements.
- E. Phase II Short List Interviews After each Firm's proposal score is calculated, the Proposal Evaluation Committee may, at its sole discretion, determine the short list of Firms to interview with the Proposal Evaluation Committee. If interviews are conducted, the interview will consist of a presentation (15 minutes) followed by a question and answer period (up to 45 minutes). After the interview, the Proposal Evaluation Committee will add the points for the "Interview" Evaluation Category based on the information presented at the interviews and the Firm's responses to questions along with scores of the Firm's GMP Fee Proposal to determine the Firm's total final score.
- F. If the Firm provides information during the Interview that differs from, or otherwise clarifies any information in the proposal submitted, the Proposal Evaluation Committee reserves the right to adjust any previous points given to the Firm in the appropriate Evaluation Category.



- G. The total proposal score plus the GMP Fee Proposal and Interview scores will be used to determine the Best Value Scores (as defined in Education Code section 17400). Best Value Score = sum total of points/number of committee members. A Best Value Score will not be calculated for those Firms not invited to interview with the Proposal Evaluation Committee (those Firms not shortlisted to compete in Phase II).
- H. The District reserves the right to conduct negotiations with any number of respondents, as determined by the District, for entering into contract agreements.

#### XVI. GENERAL TERMS AND CONDITIONS

- A. District Obligation Receipt of proposals and responses to this RFP does not obligate the District in any way. The District reserves the right to accept or reject any or all proposals, and to waive any irregularities or informalities in any proposal or in the RFP process.
- B. Award of Contract This RFP implies no obligation to award contracts to any Firm. If it is in the best interest of the District, the District retains the sole and absolute right to select the Firm that best meets the District requirements. The award is subject to acceptance by the Governing Board of the District. The District also reserves the right to reject any or all proposals.
- C. Approval to Start Work The successful Firm may perform work once a Lease-Leaseback Agreement (Site Lease, Sublease, and Construction Services Agreement) has been fully executed and approved by both parties and all appropriate documentation has been received and approved by the District, and a purchase order has been issued. The District shall not be responsible for work done, even in good faith, prior to approval of the agreement and purchase order issuance by the District.
- D. Ownership of Documents All proposals and materials submitted in response to this RFP shall become the property of the District and shall be considered a part of public records and subject to disclosure under the California Public Records Act, unless exempted by law. In addition, all designs, drawings, specifications, notes and other work developed in the performance of any services resulting from this RFP shall be the sole property of District and may be used by District for any purposes without additional compensation to the selected Firm. The selected Firm agrees not to assert any rights or to establish any claim under the design patent or copyright laws.
- E. Joint Ventures Where two or more Firms desire to submit a single response to this RFP, they should do so on a prime-subcontractor basis rather than as a joint venture or informal team. The District intends to contract with a single Firm and not with multiple Firms doing business as a joint venture.



F. Fingerprinting - Per the provisions of Education Code section 45125.1, the District has a zero tolerance policy for all Firms having any contacts with students without clearance from the State Department of Justice. All assigned personnel shall comply with the fingerprinting clearance law prior to providing services at the school sites. Firms affidavit of compliance to include DOJ print outs of every person proposed to be on school site(s). Affidavit and records shall maintained by Firm's Record Custodian.



#### RFP ATTACHMENT 1 – FIRM QUESTIONNAIRE

### (To be submitted with Phase I Proposal)

The Respondent shall furnish all the following information accurately and completely for the Respondent and each of the proposed staff and submit this with the proposal. Failure to comply with this requirement may cause rejection of the Respondent's qualifications. Additional sheets may be attached if necessary. "You" or "your" or "Respondent" as used herein refers to the Respondent and/or any of its owners, officers, directors, shareholders, parties, principals, or any qualifying individuals including any RME or RMO.

If the same information is provided elsewhere in your qualification and qualification materials, then please clearly identify such in the following questions.

Please be advised that the District may request verbal or written clarifications, additional information, an interview or presentation at any time regarding this questionnaire.

#### **SECTION A - GENERAL INFORMATION**

Telephone:	Facsimile:			
Email and Interr	net Addresses:			
Type of Respondent: (check one)				
Individual	Partnership Corpor	ration		
Names and titles of all principals/officers of the Respondent:				
Name	Title	Phone Number		



(5)	e list any applicable certifications and licenses and their associated numbers:			
(6)	Have you or any of your principals ever conducted similar services under a different name or certification or different license number?			
	(a)	If yes, give other name, address and certification or license number.		
		Name		
		Address		
		License No. (if any)		
(7)	How name	many years has Respondent been in business under its present business?		
(8)	How many years of experience does Respondent have providing similar services			
(9)	For h	now many public agencies has Respondent provided similar services?		
(10)	Pleas	se list the public agencies, including any school districts that Respondent has ded similar services for:		
(11)	nation	se attach a short history of the Respondent including whether it is local, nal, or international as well as approximate number of employees. Also de the number of offices and locations.		
(12)		ify preconstruction and lease-leaseback construction services performed for school districts in accordance with parameters described above.		

Describe how Respondent has successfully provided preconstruction and lease-

(13)



leaseback construction services such as those described herein.

(14) Describe the unique or innovative preconstruction and lease-leaseback construction services utilized on previous projects.

# SECTION B - LEGAL

(15)	Have you or any of your principals been in any claim, litigation or arbitration of any kind on a question or questions relating to similar services involving a school or community college district during the prior five (5) years?		
	(a)	If yes, provide the name of the public agency and briefly detail the dispute:	
(16)	Have you ever had a service agreement terminated for convenience in the prior five (5) years?		
	(a)	If yes, provide details including the name of the other party:	
	ondent	spondent, owners, and/or any principal or manager involved in or is aware of any pending litigation regarding professional misconduct, bad faith, n, or sexual harassment?	
	(a)	If yes, provide details:	



(18)	any pe	spondent, owners, and/or any principals or manager involved in or aware of ending disciplinary action and/or investigation conducted by any local, state eral agency?		
	(a)	If yes, provide details:		
(20)		Respondent comply with all District, local, state and federal legal ements, regulations and laws?		
SEC1	TION C	- ADDITIONAL INFORMATION		
(21)	your	Please provide any other information that may assist the District in ascertaining our qualifications, capability and customer service under any resultangreement.		
SEC1	TION D	- CONFLICT OF INTEREST		
(22)	with a	you ever had any direct or indirect business, financial or other connection ny official, employee or consultant of the District? Identify any conflict of st in (a):		
	(a)	Please elaborate and discuss any potential, apparent or actual conflict of interest (use additional pages as required):		



# **SECTION E.** Exceptions to Agreement Forms

	The Firm is required to list any exceptions to terms in the Agreement			
	Forms below. (Use add	ditional pages as required):		
		f perjury under the laws of the State oing Firm Questionnaire is true and		
Executed this	s day of	, 20, at	, State of	
Company Na	ame	Signature		
Title		Print Name		
( <b>NOTE:</b> Form n	nust be signed by individual	authorized to commit the firm to the terms	of the proposal.)	



#### RFP ATTACHMENT 2A – Proposed Lease Leaseback Fee (for Phase I)

## A completed "Attachment 2A" must be submitted with the Phase I Proposal.

The Firm proposes the following fees:

1. Lease-Leaseback Fee - The lease-leaseback fee shall include the Firm's overhead, profit, and all other costs (excluding preconstruction services and general conditions), and should be expressed as a percentage and shall be the same as the "Contractor's Fee" as set forth in Article 3 of the Construction Services Agreement in Attachment 3 to this RFP and as proposed by Contractor in response to the Phase I RFP Evaluation Category.

**NOTE**: Should the Firm try to revise the Lease-Leaseback Fee (a.k.a. "Contractor's Fee") so that it exceeds the percentage below, the Firm agrees and acknowledges that the District has the right to deem the Firm's proposal non-responsive, cancel the lease-leaseback contract without owing any fees or costs to the Firm, and award a contract to another contractor/Firm.

Lease-Leaseback Fee (Contractor's F	<b>ee):</b> % of trade construction cost.
Executed this day of	, 20
Company Name	Signature
Title	Print Name

(**NOTE:** Must be signed by individual authorized to commit the firm to the terms of the proposal.)



#### RFP ATTACHMENT 2B - PRECONSTRUCTION SERVICES FEE PROPOSAL (for Phase II)

# A completed "Attachment 2B" and requested back up documentation must be submitted with the <u>Phase II</u> GMP Fee proposal.

## (To be submitted by Shortlisted firms only.)

The Firm proposes the following fees:

Lease-Leaseback Fee - The lease-leaseback fee shall include the Firm's overhead, profit, and all other costs (excluding general conditions), and should be expressed as a percentage and shall be the same as the "Contractor's Fee" as set forth in Article 3 of the Construction Services Agreement in Attachment 3 to this RFP and as proposed by Contractor in response to the Phase I RFP Evaluation Category.

<u>NOTE</u>: Should the Firm try to revise the Lease-Leaseback Fee (a.k.a. "Contractor's Fee") so that it exceeds the percentage below, the Firm agrees and acknowledges that the District has the right to deem the Firm's proposal non-responsive, cancel the lease-leaseback contract without owing any fees or costs to the Firm, and award a contract to another contractor/Firm.

and award a contract to another contractor/	Firm.
Lease-Leaseback Fee:% of submitted on Attachment 2A in the Phase I p	trade construction cost. (Should match proposed % Firm proposal)
provide preconstruction services as detailed	tion XII C.1, state Guaranteed Maximum Price per month to d in this RFP and CSA Exhibit "A1" Preconstruction Services fee shall include any/all labor, expenses, items or services of the project.
Preconstruction Services: \$	/month*
	2, The general conditions should be expressed as a monthly dule, and description in Section II of the RFP.
General Conditions: \$	/month*
(above) a line item listing of the proposed m	the preconstruction services and general conditions totals nonthly fee elements and the corresponding monthly cost on the Project. Failure to include this backup line item listing will be a second to the project.
Executed this day of	, 20
Company Name	Signature
 Title	Print Name

(NOTE: Must be signed by individual authorized to commit the firm to the terms of the RFP.)



# RFP ATTACHMENT 3 - SAMPLE AGREEMENT FORMS

RFP ATTACHMENT 3 Construction Services Agreement

CSA Exhibit "A" Scope of Work

CSA Exhibit "A1" Preconstruction Services Scope of Work

RFP ATTACHMENT 3A Site Lease Agreement

RFP ATTACHMENT 3B Sublease Agreement



# RFP Attachment 3 SAN BERNARDINO CITY UNIFIED SCHOOL DISTRICT LEASE-LEASE BACK PROJECTS MASTER CONSTRUCTION SERVICES AGREEMENT

Between

# SAN BERNARDINO CITY UNIFIED SCHOOL DISTRICT

and

Dated as of \_\_\_\_\_\_, 2019



1.	GENERAL INTENT	3
2.	TITLE 24 RESPONSIBILITIES – GENERAL INTENT OF THE CSA	5
3.	CONTRACT INFORMATION	7
4.	DEFINITIONS	7
5.	ESTABLISHMENT OF GUARANTEED MAXIMUM PRICE "GMP"	13
6.	OPEN BOOK ACCOUNTING AND SELECTION OF SUBCONTRACTORS	17
7.	SAVINGS AND VALUE ENGINEERING	19
8.	DISTRICT CONTINGENCY	20
9.	SCHEDULE	20
10.	INSPECTION OF WORK/ INSPECTOR AND ARCHITECT	22
11.	ARCHITECT	25
12.	DISTRICT RESPONSIBILITIES	25
13.	CONTRACTOR RESPONSIBILITIES	27
14.	CONTRACT DOCUMENTS AND INTERPRETATIONS	35
15.	SUBMITTALS	35
16.	REQUEST FOR SUBSTITUTIONS	42
17.	EXTRA WORK/MODIFICATIONS (INCLUSION OF CCD COSTS, DSA COSTS, AND AN ICD PROCESS)	43
18.	TIME OF COMPLETION	47
19.	TERMINATION OF AGREEMENT	49
20.	RESOLUTION OF AGREEMENT CLAIMS	51
21.	MAINTENANCE OF RECORDS; AUDIT/OWNERSHIP OF DOCUMENTS	59
22.	PREVAILING RATES OF WAGES; RECORDS, APPRENTICES	60
23.	RECORDS OF WAGES PAID	61
24.	APPRENTICES	63
25.	REGISTRATION WITH DEPARTMENT OF INDUSTRIAL RELATIONS	64
26.	HOURS OF WORK	65
27.	SKILLED AND TRAINED WORKFORCE	65
28.	PROTECTION OF PERSONS AND PROPERTY	66
29.	PAYMENTS AND RETENTION	70
30.	NONCONFORMING WORK	73
31.	SUBCONTRACTOR PAYMENTS	73
32.	SEPARATE CONTRACTS	73
33.	USE OF PREMISES/SAFETY	74
34.	CLEANING UP	74
35.	INSURANCE	75
36.	HOLD HARMLESS AND INDEMNITY	79
37.	SUBSTITUTION OF SECURITY	80
38.	TITLE TO WORK	80
39.	COMPLIANCE WITH STATE STORM WATER PERMIT FOR CONSTRUCTION	81
40.	EQUAL OPPORTUNITY CLAUSE	81
41.	SPECIAL NOTICE OF AMERICAN'S WITH DISABILITIES ACT	82



42.	PATENTS, ROYALTIES, AND INDEMNITIES	82			
43.	EXCISE TAX	82			
44.	PROHIBITED INTERESTS	82			
45.	COMPLIANCE WITH DTSC GUIDELINES – IMPORTED SOIL/SOILS INSPECTION	83			
46.	HAZARDOUS WASTE AND UNKNOWN PHYSICAL CONDITIONS83				
47.	NO ASBESTOS CERTIFICATION84				
48.	LAWS AND REGULATIONS85				
49.	AGREEMENT MODIFICATIONS				
50.	NOTICES				
51.	THIRD-PARTY CLAIMS	85			
52.	ASSIGNMENT				
53.	HEADINGS	85			
54.	INTEGRATION/MODIFICATION				
55.	APPLICABLE LAW/ PROVISIONS REQUIRED BY LAW DEEMED INSERTED				
56.	SUCCESSION OF RIGHTS AND OBLIGATIONS				
EXHIBI' Attachme	ent 2 — Insurance Documents & Endorsements ent 3 — Drug-Free Workplace Certification ent 4 - Contractor's Certificate regarding Alcoholic Beverage and Tobacco Free Campus Policy ent 5 - Statement of Intent of Meeting DVBE Participation Goal ent 6 — Payment Bond ent 7 — Faithful Contract Performance Bond ent 8 — Prime/General Contractor Information ent 9 — Vendor Tax Information ent 10 — Vendor Application ent 11 — Certification of Non-Utilization of Asbestos Materials ent 12 — Contractor Prevailing Wage Compliance Certification ent 13 - Guarantee ent 14 — Contractor's Certification regarding Background Checks chment 14A - Contractor's Certification regarding Background Checks — List of Names chment 14B — Fingerprint and Criminal Background Check Certification Form/ Request for Waiver ent 15 - Escrow Agreement for Security Deposits in Lieu of Retention				
	ent 16 – Disabled Veteran Business Enterprise (DVBE) Contractor Close Out Statement ent 17 – Conduct Rules for Contractors				
	ent 18 - (a.k.a. Exhibit K) – Contract Information				
	Division 1 Forms				

Immediate Construction Change Directive Certificate of Substantial Completion



# SAN BERNARDINO CITY UNIFIED SCHOOL DISTRICT PROJECTS

#### MASTER CONSTRUCTION SERVICES AGREEMENT

This Master Construction Services Agreement is made as of	F, 2019, by and	l between the San				
Bernardino City Unified School District, a California School District of	organized and existing und	ler the laws of the				
State of California (hereinafter called the "District"), and	, a (	California licensed				
contractor operating under the laws of the State of California ("Contractor").						

General intent of agreement:

WHEREAS, the District entered into an agreement with \_\_\_\_\_\_\_ (the "Architect") to provide architectural services for the District for the purpose of developing Construction Documents for the construction of improvements at [INSERT SCHOOL SITES] (the "Projects" or "\_\_\_\_\_\_\_ Projects".)

#### 1. GENERAL INTENT

- 1.1 The Board of Education has reviewed the different methodologies available to deliver a public works project and has carefully considered the options of competitive bid to a general contractor who would be responsible for the entire project, a construction management managed multi-prime trade contract project, an at-risk construction management contract, turn-key delivery by another public entity or delivered by another public entity through a joint use project, but have through Board action and independent staff and Board review determined that there are benefits and detriments to each delivery method.
- 1.2 The Board of Education has also reviewed the lease-leaseback methodology under California Education Code section 17406 which permits the governing board of a school district to lease to any person, firm, or corporation any real property owned by the District if the instrument by which such property is leased requires the lessee to construct on the leased premises, or provide for the construction thereon, of a building for the use of the school district, during the term of the lease, and provides that title to that building shall vest in the school district prior to or at the expiration of the lease.
- As part of the Board of Education's consideration of the possible methods of delivery, the Board has also reviewed available information from the Coalition of Adequate School Housing materials on delivery methods, California School Board Association, California Association of School Business Officials, Office of Public School Construction Meeting Minutes and SAB Implementation Committee meeting minutes and considered the benefits and detriments of the lease-leaseback delivery method.
- 1.4 Further, the Board of Education understands that unique to the lease-leaseback delivery method, the lease-leaseback Contractor will not only be undertaking the traditional due diligence of investigating existing Project related information, documents and the Project site, but now included as part of the Contractor's "Due Diligence" (as defined herein) as part of this lease-leaseback delivery method, the Contractor will be performing a review of the Construction Documents to visualize conflicts that may have not been located by the Architect as part of the Architect's constructability review when the Construction Documents were being prepared.
- 1.5 The Board of Education in its consideration of the substantial evidence that is available to the District staff and through the Board's own research has determined that this ability to work between the Contractor and the Architect to resolve a greater percentage of construction claims that would ordinarily arise through any of the other delivery methods addressed in Article 1.1 above also provides the ability of the Contractor to determine the likely level of errors and omissions, and provides a Guaranteed Maximum Price for the Project based on the Contractor's Due Diligence.



The unique ability to determine with certainty the budget numbers for the Project provides this Board of Education the ability to not only ensure that the District is best serving the community and its school children, but also provides the ability to focus resources towards future and simultaneous projects that could not be undertaken during any of the other delivery methods since a sizable contingency needs to be set aside for potential claims, litigation, arbitration, mediation, and delays that could jeopardize the ability to plan for occupancy of the building or the possibility of having to spend significant resources to procure alternative facilities.

- As part of this lease-leaseback Master Construction Services Agreement, a site lease with Contractor (the "Master Site Lease"), for the Projects has been entered into and attached as Exhibits and Attachments to the Master Site Lease is a description of the sites (the "Sites") in order for Contractor to construct improvements to this existing school Sites under the possessory interest of a lease with a greater degree of control over the overall Project, including ability to coordinate Site related items such as utilities, ability to insure both the Projects and the Sites against a broader range of risks, and greater primary control and oversight over Subcontractors and suppliers for the Projects as the lessee of the Sites.
- 1.7 In addition, the Contractor subleases the constructed portions of the Sites and the Projects back to the District pursuant to a Master Sublease Agreement (the "Master Sublease") under which the District will be required to make Sublease Payments as described therein; and
- 1.8 It is agreed that upon the expiration of the Master Site Lease and Master Sublease, title to the Project shall vest in the District; and
- 1.9 Contractor represents that Contractor is uniquely experienced in Construction of public schools and community colleges including, but not limited to, the specific requirements and regulations of the Field Act as administered by the Division of State Architect, working with the Division of State Architect, Office of Public School Construction, California Department of Education and work with the various applicable other State and local agencies that have jurisdiction over the Project, is duly licensed as a contractor in the State of California, and is prepared to analyze, synthesize and efficiently perform construction work for the District as more fully set forth in this Agreement
- 1.10 Contractor has thoroughly Due Diligence as defined in Articles 4 and 5 to establish a Guaranteed Maximum Price for the Projects (which may include an Errors and Omissions Contingency and a Construction Contingency for Contractor's own errors and omissions) that will not be exceeded. Contractor has investigated the site conditions and reviewed the Construction Documents to establish that there are no known problems with respect to the site conditions or the Construction Documents and that Contractor can and will construct the Project for the Guaranteed Maximum Price as set forth in Article 3 and defined in Article 5 of this Master Construction Services Agreement, and Contractor will not seek any additional compensation whatsoever, including, without limitation, any requests based upon known site conditions, extensions on the Lease beyond the Lease period or any requests, except for such additional compensation provided for herein based upon unforeseen conditions and/or errors or omissions contained within the plans and specification or Construction Documents.
- 1.11 Since the Contractor has entered into a negotiated Lease and is performing this Master Construction Services Agreement as the Lessee of the Premises, Contractor understands and agrees that:
  - 1.11.1 Public Contract Code section 4100 et seq. addressing subcontractor listing shall not apply except to the extent applicable under Education Code section 17406(a)(4). However, the District is requiring an open book accounting and the public selection of Subcontractors pursuant to Article 6.3 of this Agreement.
  - 1.11.2 Public Contract Code section 20111 addressing competitive bidding does not apply to the Project pursuant to the specific language of Education Code section 17406 which



provides for a competitive procurement process through request for sealed proposals from qualified proposers.

- 1.11.3 Public Contract Code section 3400 addressing proprietary specifications does not apply since the Contractor has entered into a negotiated Lease pursuant to which is obligated to build the Project. The Contractor agrees and acknowledges that it has had great opportunity throughout the Due Diligence process and negotiation of the Lease and related agreements to propose any changes or substitutions, and warranties that it shall propose no further changes or substitutions pursuant to Public Contract Code section 3400. Substitutions and Value Engineering are allowed to address cost savings and to more efficiently build the Project at Articles 5.3 and 16.
- 1.11.4 The requirements in Public Contract Code section 22300 shall not apply.
- 1.12 Prequalification of Contractor and MEP Subcontractors. In accordance with California Public Contract Code section 20111.6, the Contractor is required to submit to the District a completed set of prequalification documents on forms provided by the District and be deemed prequalified by the District prior to entering into the Contract for the Project. In addition, all mechanical, electrical or plumbing ("MEP") Subcontractors of any tier (contractors that hold C-4, C-7, C-10, C-16, C-20, C-34, C-36, C-38, C-42, C-43 or C-46 licenses), must also be prequalified. It is the responsibility of the bidder to ensure that all MEP Subcontractors holding any of the licenses listed above are properly prequalified. This prequalification requirement for MEP Subcontractors applies even if the subcontractor will perform, or is designated to perform, work that does not require one of the licenses listed above, but the subcontractor holds one of the licenses listed above.

#### 2. <u>TITLE 24 RESPONSIBILITIES – GENERAL INTENT OF THE CSA</u>

Contractor accepts the contractual relationship established between it and District by this Master Construction Services Agreement, and Contractor covenants with District to furnish reasonable skill and judgment in constructing the Project as set forth in the Construction Documents, as defined in Article 4 for the Project which are described and/or set forth herein as forth in the Construction Documents, exhibits and attachments. Contractor agrees to furnish efficient business administration, coordination review of the Construction Documents, coordination of the work of the Subcontractors and vendors and superintendence to furnish at all times an adequate supply of professionals, workers, and materials and to perform the work appropriately, expeditiously, economically, and consistent with the Master Construction Services Agreement and Construction Documents as defined in Article 14, below.

- 2.1 <u>Title 24 Responsibilities</u>. The Contractor shall continually supervise and direct the Work using the Contractor's best skill and attention. The Contractor shall be solely responsible for and have control over construction means, methods, techniques, sequences, procedures; and shall coordinate all portions of the Work in conformance with the Contract Documents. Specific duties of the Contractor shall include those set out in Section 43 of Title 21 of the California Code of Regulations and Section 4-343 of Title 24 of the California Code of Regulations. These duties include, but are not limited to the following:
  - 2.1.1 *Responsibilities.* It is the duty of the Contractor to complete the Work covered by his or her Contract in accordance with the approved Construction Documents. The Contractor in no way is relieved of any responsibility by the activities of the Architect, Engineer, Inspector or DSA in the performance of their duties.
  - 2.1.2 Performance of the Work. The Contractor shall carefully study the approved Construction Documents and shall plan its schedule of operations well ahead of time. If at any time it is discovered that work is being done which is not in accordance with the approved Construction Documents, the Contractor shall correct the Work immediately.



- 2.1.3 *Inconsistencies*. All inconsistencies or timing or sequences which appear to be in error in the Construction Documents shall promptly be called to the attention of the Architect or, Engineer, for interpretation or correction. Local conditions which may affect the structure shall be brought to the Architect's attention at once. In no case, shall the instruction of the Architect be construed to cause work to be done which is not in conformity with the approved plans, specifications, change orders, construction change documents, and as required by law. (See Title 24 Section 4-343)
- 2.1.4 *Verified Reports.* The Contractor shall make and submit to the office from time to time, verified reports as required in Title 24 Section 4-366. As part of the Close-Out of the Project (see Article 13.16), Contractor shall be required to execute a Form 6-C as required under Title 24 Sections 4-343.
- 2.1.5 *Reporting Requirements*. Contractor shall fully comply with any and all reporting requirements of Education Code sections 17315, et seq., in the manner prescribed by Title 24, as applicable.
- 2.1.6 Contractor Responsibility. The Contractor shall be responsible to the District for acts and omissions of the Contractor's employees, Subcontractors, material and equipment suppliers, and their agents, employees, invitees, and other persons performing portions of the Work under direct or indirect contract with the Contractor or any of its Subcontractors.
- 2.1.7 All Work is performed Under the Direction of Inspector. Pursuant to Title 24 requirements, the Contractor shall not carry on Work except with the knowledge of the Inspector. (See Title 24 generally)
- 2.1.8 Contractor to Establish Timing and Protocol with Inspector. Contractor shall establish a protocol for requesting inspection with Inspector so as to not delay the Work and provide adequate time for the Inspector to perform inspection. If such a protocol is not established ahead of time, Inspector may utilize the time criteria set by Title 24 of 48 hours in advance of submitting form DSA 156 for each new area. DSA requirements under PR 13-01 specifically gives the Special Inspector fourteen (14) days to post to the DSA website. Contractor is responsible for delays and for failure to plan.
- 2.1.9 *Conformance with Approved Submittals.* This conformance includes performing all Work only in conformance with approved Submittals, Shop Drawings, and Samples or the Inspector may be required to issue a DSA Form 154 Notice of Deviation from approved DSA Contract Documents.
- 2.1.10 *Incremental Assemblies*. For some Projects, there may be a need to incrementally install certain assemblies. It is up to Contractor to identify areas and assemblies that may be constructed incrementally. Contractor must identify and establish incremental areas of construction and establish protocols with Inspector for DSA 152 approvals so they may be presented to DSA. See PR-13 item 2.1.10 for further discussion.
- 2.1.11 Coordination with Outside Contractors. If any of the Work for the Project is known to include Work performed by contractors retained directly by the District, Contractor shall be responsible for the coordination and sequencing of the Work of those other contractors so as to avoid any impact on the Project Schedule.



# 3. <u>CONTRACT INFORMATION</u>

3.1	<u>District</u> :	San Bernardino City Unified School District 956 W. 9 <sup>th</sup> Street San Bernardino, CA 92411 (909) 388-6100			
3.2	Notices:	Tom Pace			
		tom.pace@sbcusd.k12.ca.us			
3.3	Contractor:				
3.4	Notices:				
	of Work/Co 6), and Pe Construction the parties of through Co	proposes projects at the following District sites:  School. During preconstruction services, it is possible that the Sites will be combined into one Project to save time and costs. Each individual Project will have separate contract information for Sections 3.5-3.9, Scope onstruction Documents (Exhibit A), Master Budget (TBD), Payment Bond (Attachment rformance Bond (Attachment 7), which will be set forth in Exhibit K to Master on Services Agreement which Exhibit K shall be amended by the mutual agreement of from time to time to reflect the Projects. The following are established entractor's review of the Program, Contract Documents and through Contractor's Due prior to entering into this Agreement:			
3.5	Contract Ti	me is TBD Days. See Exhibit K for each Project.			
3.6	-	Damages for overstaying Lease (Art. 18) is \$TBD per calendar day. See Exhibit K for Project.			
3.7 Guaranteed Maximum Price (Art. 5) is \$TBD. See Exhibit K for each		Maximum Price (Art. 5) is \$TBD. See Exhibit K for each Project.			
	3.7.1	Construction Contingency (within GMP) is \$TBD. See Exhibit K for each Project.			
	3.7.2	Errors and Omissions Contingency (within GMP) is \$TBD. See Exhibit K for each Project.			
3.8	The only exception to the GMP is Unforeseen Underground Conditions, and District Contingency for Owner requested extras as follows:				
	3.8.1	District's Contingency (Art. 8) is \$TBD. District Contingency is carried outside of the GMP. See Exhibit K for each Project.			
	3.8.2	Unforeseen Allowance is \$TBD. Unforeseen Allowance is carried outside of the GMP. See Exhibit K for each Project.			
3.9		ctor's fee for this Project is percent (%) and is included in the GMP. See Exhibit Project.			
<u>DEFI</u>	<u>NITIONS</u>				
4.1	Action of th	Action of the Governing Board is a vote of a majority of the District's Governing Board.			

4.



- Allowances are separate from the Unforeseen Allowance and mean budgets established for specific scopes of the Work which cannot be fully defined in the Construction Documents at the time that the GMP is established. Allowances may only be drawn upon pursuant to a Change Order issued pursuant to Article 17. In the event that an Allowance is included, the Contractor shall provide all services, work, labor and materials reasonably implicit in the description of the Allowance for the amount stated for the Allowance, all in accordance with the Construction Documents. Contractor acknowledges and agrees that it has had ample time and consideration to fully assess any Allowance(s) and to negotiate the description and amount of the Allowance(s), such that Contractor fully accepts and shall bear the entire risk and responsibility of providing all services, work, labor and materials required for the Allowance(s) under this Agreement. Expenditures from the GMP will either arise from Construction Contingency or Errors and Omissions Contingency and shall be submitted pursuant to Article 17 addressing Change Orders. The amount of the Change Order shall reflect the difference between actual costs approved by the District and the allowance amounts established in the GMP.
- 4.3 <u>As-Builts</u> are a set of Construction Documents maintained by the Contractor clearly showing all changes, revisions, substitutions, field changes, final locations, and other significant features of the Project. The As-Builts shall be maintained continuously throughout the Work for the Project and is both a prerequisite to the issuance of Pay Application and a requirement for Contract Close-Out. See Article 13.14.
- 4.4 <u>Architect</u> means the architect, engineer, or other design professional engaged by the District to design and perform general observation of the work of construction and interpret the drawings and specifications for the Project.
- 4.5 <u>Beneficial Occupancy</u> is the point in time when a building or buildings are fit for occupancy is fit for occupancy and its intended use Basic requirements are the building is safe, at or near Substantial Completion, and all life safety is operational. The fact that a building is occupied does not mean that the building is ready for Beneficial Occupancy if there are elements that are unsafe or if life safety items are not operational. Taking occupancy on a structure that is under a fire watch is not considered Beneficial Occupancy. Beneficial Occupancy is not be used by the Contractor as a basis to request Retention Payment unless the entire Project is Substantially Complete in accordance with Article 4.45.
- 4.6 <u>Claims.</u> A Claim is a request for payment, supported by back-up documentation which includes, invoices time sheets, or other documents substantiating legitimacy or entitlement that is submitted during the Project or immediately following the Project made prior to the Application for Retention Payment and prior to Final Completion of the Project. A "Claim" means a separate demand by the Contractor for (1) time extension, (2) payment of money or damages arising from Work done by or on behalf of the Contractor pursuant to the Contract and payment of which is not otherwise expressly provided for or the claimant is not otherwise entitled to, or (3) and amount the payment of which is disputed by the District. See Article 20.
- 4.7 <u>Close-Out</u> means the process for Final Completion of the Project, but also includes the requirements for the DSA Certification that the Project is Complete (See DSA Certification Guide). See Article 13.16.
- 4.8 <u>Commencement Date</u> shall mean the Project commencement date found in the Notice to Proceed for the Project in accordance with Article 4.28 of this Construction Services Agreement.
- 4.9 <u>Complete/ Final Completion</u> means that all Work in the Contract Documents is finished, the requirements of the Contract Documents have been met, successful testing, startup and satisfactory operation of the Project as a total unit has been accomplished in substantial conformance with the Contract Documents, the Project is completed, all Work has ceased on the Project and the Project has been accepted by the District's Board. This may also be referred to as Final Completion. In



- most cases, the recording of a Notice of Completion shall represent Completion of the Project. Beneficial Occupancy or Substantial Completion does not mean the Work is Complete.
- 4.10 <u>Completion Date</u> is the date when all Work for the Project shall be Substantially Complete and is the date assigned at the end of the Contract Time for the Project.
- 4.11 <u>Construction Change Document (CCD).</u> A Construction Change Document is a DSA term that is utilized to address changes to the DSA approved Construction Documents. There are two types of Construction Change Documents. (1) DSA approved CCD Category A (DSA Form 140) for work affecting Structural, Access or Fire-Life Safety of the Project which will require a DSA approval; and, (2) CCD Category B (DSA Form 140) for work NOT affecting Structural Safety, Access Compliance or Fire and Life Safety that will not require a DSA approval (except to confirm that no Approval is required). See Article 17.4.
- 4.12 <u>Master Construction Services Agreement (CSA)</u> means this Master Construction Services Agreement, together with any duly authorized and executed amendments hereto.
- 4.13 <u>Construction or Construction Services</u> means all labor and services necessary for the construction of the Project, and all materials, equipment, tools, supplies and incidentals incorporated or to be incorporated in such construction as fully described in the Contract Documents.
- 4.14 Construction Costs means any and all costs incurred by the Contractor with respect to the construction and equipping, as the case may be, of the improvements performed, whether paid or incurred prior to or after the date hereof, including, without limitation, costs for Site preparation, the removal or demolition of existing structures, the construction of the Project and related facilities and improvements, and all other work in connection therewith, security of the Site and Project, Contractors' overhead and supervision at the Project Site, all costs and expenses including any taxes or insurance premiums paid by the Contractor with respect to the Property, and administrative and other expenses necessary or incident to the Project, excluding Contractors' and Developers' home office overhead and profit. The term "Construction Costs" includes all Contractor's costs associated with preparing or generating additional copies of any Construction Documents, as defined below, related to or required for the Project, including preparation or generation of additional Plans and/or Specifications for Contractor's Subcontractors. In no event shall Construction Costs exceed the Guaranteed Maximum Price.
- 4.15 Construction Documents comprise the Plans and Specifications approved by DSA for each Project Allowances stipulated in the Contract Documents, and all Addenda, if any, issued prior to the entry into this Agreement. The Construction Documents shall include all Modifications generated after the Effective Date in accordance with the Contract Documents, including, without limitation, a written amendment to the Contract signed by the Contractor and duly executed and approved by the District, a Change Order, a Construction Change Document, or a written order for a minor change in the Work issued by the Architect.
- 4.16 <u>Contract Documents</u> means those documents which form the entire Contract by and between District and Contractor. The Contract Documents consist of the Master Site Lease, Master Sublease, General, Supplementary and other Conditions, this Master Construction Services Agreement, including all exhibits and attachments hereto, and the Construction Documents. The Contract Documents collectively form the Contract. The Contract represents the entire and integrated Agreement between the parties hereto and supersedes prior negotiations, representations, or agreements, either written or oral. The Contract may be amended or modified only by a written Modification. The Contract Documents shall be binding solely upon the District and Contractor, do not create a contractual relationship of any kind between the Architect and Contractor, between the District and any Subcontractor or Sub-subcontractor, or between any persons or entities other than the District and the Contractor, and are not intended to and do not create any third party beneficiary.



- 4.17 <u>Contract Time</u> is the time period specified in the Contract Documents in which the Project shall be completed. This is sometimes referred to a Contract Duration, or "time in which the Contractor has to Complete the Project". See Article 9.
- 4.18 <u>Day</u> means a calendar day unless specifically designated as a business day.
- 4.19 <u>Drawings or Plans</u> are graphic and pictorial portions of the Contract Documents prepared for the Project and approved changes thereto, wherever located and whenever issued, showing the design, location, and scope of the Work, generally including plans, elevations, sections, details, schedules, and diagrams as drawn or approved by the Architect. Sometimes Drawings will also be included in Addenda, Change Orders, and Specifications.
- 4.20 <u>Due Diligence</u> is the review and analysis of as-built documents, title documents, any prior design documents for the Project or Site, geotechnical reports, surveys, site investigations and other documents and information provided by the District, and synthesizing of information utilized to determine the components of the GMP. Requirements for Due Diligence are further addressed at Article 5.
- 4.21 <u>DSA</u> is the Division of State Architect. DSA is the agency that provides design and construction oversight for K-12 Schools, Community Colleges, and State Funded Charter School Projects. DSA is the responsible agency for this Project and Contractor has submitted a bid for the Project since Contractor is familiar with Contractor's responsibilities under the DSA requirements more thoroughly set forth at Title 24 of the California Code of Regulations. Contractor agrees to abide by the jurisdiction of DSA and shall construct the Project to conform with the approved plans, specifications, Addenda, and Change Orders (inclusive of approved CCD's and ICD's issued by the District pending CCD approval). The DSA website is at http://www.dgs.ca.gov/dsa.
- 4.22 <u>Effective Date</u> is the latter of the date upon which the District Board approves the Master Site Lease and the Master Sublease and Contractor has executed the Master Site Lease and Master Sublease.
- 4.23 <u>Float</u> the total number of days an activity may be extended or delayed without delaying the Completion Date shown in the schedule. Float will fall into three categories: (1) Rain Days; (2) Governmental Delays; and, (3) Project Float. See Article 9.2.
- 4.24 <u>Immediate Change Directive (ICD) is a</u> written order prepared by the Architect and signed by the District and the Architect, directing a change in the Work where the Work must proceed immediately and stating a proposed basis for adjustment, if any, in the Contract Sum or Contract Time, or both. See Article 17.4.1.2
- 4.25 <u>Inspector of Record (IOR)</u> or Project Inspector (PI) is the individual retained by the District in accordance with Title 24 of the California Code of Regulations who will be assigned to the Project
- 4.26 <u>Guaranteed Maximum Price or GMP</u> means the Guaranteed Maximum Price established pursuant to Article 5 to be paid to Contractor for Contractor's construction of the Project hereunder, subject to any adjustments for Extra Work/Modifications as provided in Article 17.
- 4.27 <u>Notice of Non-Compliance (DSA Form 154)</u> is a document issued by the Inspector if there is a deviation from the DSA approved Plans, Specifications, and Change Orders. See Article 17.2.
- 4.28 Notice to Proceed. After execution of this Master Construction Services Agreement and the Master Site Lease(s) and Master Sublease(s) between the parties, the District shall issue a notice to the Contractor to proceed with the Project ("Notice to Proceed") for each Project, which Notice to Proceed shall include the date upon which commencement for the Project shall commence. If a legal challenge is made as to the validity of the Contract Documents, within 90 days from the date of the board award, the Parties may mutually elect to rescind any Notice to Proceed that has been issued by the District and the Master Construction Services Agreement and



- the Master Site Lease(s) and Master Sublease(s) between the Parties shall be terminated for the convenience of the Parties. Upon such termination, the District shall be obligated to pay Contractor in accordance with Section 19.2.
- 4.29 <u>Plans</u> are that portion of the Construction Documents consisting of the drawings and other pictorial or other graphic expression of requirements for the work of improvement to be completed by Contractor, including, without limitation, services, work, material, equipment, construction systems, instructions, quality assurance standards, workmanship, and performance of related services.
- 4.30 <u>Project</u> means the specific improvements to be constructed and installed by the Contractor at a particular Site, as more particularly described and/or referenced in Exhibit "A" attached hereto and as set forth in Exhibit "K" as may be applicable.
- 4.31 <u>Provide</u> shall include "provide complete in place," that is "furnish and install complete."
- 4.32 <u>Punch List</u> is a list of minor repair items, prepared after the issuance of a Certificate of Substantial Completion, by the Inspector and Architect of Work required in order to complete the Contract Documents and ensure compliance with the DSA Approved Plans so the Project may be Closed Out. Issuance of the Retention Payment is dependent upon the proper completion of the Punch List. See Article 13.16 and Article 29.
- 4.33 Request for Information (RFI) is a written request prepared by the Contractor requesting the Architect to provide additional information necessary to clarify or amplify an item which the Contractor believes is not clearly shown or called for in the drawings or specifications, or to address problems which have arisen under field conditions.
- 4.34 <u>Schedule</u> is the Contractor's view of the practical way in which the Work will be accomplished. In this Agreement there is a requirement for a Baseline Schedule and regular Schedule Updates that show all Work to be completed during the Contract Time and shall include all items listed under Article 9.3.
- 4.35 <u>Schedule of Values</u> is a detailed breakdown of the Contract Price for each Project, building, Phase of Work or Site as determined by the District. This Schedule of Values shall adequately detail the price for the Work so that the status of the construction of any improvements can be meaningfully reviewed by the Inspector, Architect of Record, Engineer of Record, and District. (See Article 13.12)
- 4.36 <u>Separate Contracts</u> are Contracts that the District may have with other Contractors, vendors, suppliers, or entities to perform Work on the Project. This may include, but is not limited to Multi-Prime Trade Contractors, furniture installers, testing agencies, clean-up contractors, or network or low voltage contractors. Contractor shall plan for certain other contractors that may also be working on the Project site and address these other contractors in Contractor's Schedule. See Article 32.
- 4.37 Site refers to the grounds of the Project or in some cases may refer to multiple sites as defined in the Contract Documents and such adjacent lands as may be directly affected by the performance of the Work.
- 4.38 <u>Master Site Lease and/or Lease</u> means the Master Site Lease(s) of even date herewith, by and between the District and the Contractor together with any duly authorized and executed amendment thereto under which the District leases the Site to the Contractor.
- 4.39 <u>Specifications</u> are that portion of the Construction Documents consisting of the written requirements for the work of improvement to be completed by Contractor, including, without limitation, services, work, material, equipment, construction systems, instructions, quality assurance standards, workmanship, and performance of related services.



- 4.40 <u>Standards</u>, <u>Rules</u>, <u>and Regulations</u> referred to are recognized printed standards and shall be considered as one and a part of these specifications within limits specified. Federal, state and local regulations are incorporated into the Contract Documents by reference. The Contractor shall not be responsible for any violations of federal, state or local codes, laws or regulations applicable to the Construction Documents that are part of the Architect's standard of care. Nothing in this paragraph, however, shall be construed as a limitation or waiver of the Contractor's applicable standard of care for the Project.
- 4.41 Stop Work Order, or an Order to Comply is issued when either (1) the Work proceeds without DSA approval; (2) the Work proceeds without a DSA Project Inspector, or (3) where DSA determines that the Work is not being performed in accordance with applicable rules and regulations, and would compromise the structural integrity of the Project or would endanger lives. If a Stop Work Order is issued, the Work in the affected area shall cease until DSA withdraws the Stop Work Order. Pursuant to Education Code section 17307.5(b) and Education Code section 81133.5, the District shall not be held liable in any action filed against the District for any delays caused by compliance with the Stop Work Order.
- 4.42 <u>Subcontractor</u> means any person or entity, including trade contractors, who have a contract with Contractor to perform any work or supply materials for the Project.
- 4.43 <u>Master Sublease(s)</u> means the Master Sublease(s) of even date herewith by and between the District and Contractor together with any duly authorized and executed amendment hereto under which the District subleases the Site from the Contractor.
- 4.44 <u>Sublease Payment</u> means any payment required to be made by the District pursuant to Section 7 of the Master Sublease.
- 4.45 <u>Substantial Completion</u> is not reached unless and until each of the following four (4) conditions have been met: (1) all contractually required items have been installed with the exception of only minor and Incomplete Punch Items (See Article 13.16); (2) All Fire/Life Safety Systems have been installed, and are working and signed off on the DSA Form 152 Inspection Card, all building systems including mechanical, electrical and plumbing are all functioning; (3) all other items on the DSA Form 152 Inspection Card for the Project have been approved and signed off; and (4) the Project is fit for occupancy and its intended use, as certified by the Architect pursuant to the Certificate of Substantial Completion set forth in the Division 1 Forms attached hereto.
- 4.46 <u>Substitution</u> is a change in product, material, equipment, or method of construction from those required by the Construction Documents proposed by the Contractor. Specific requirements for substitutions are set forth at Article 16.
- 4.47 <u>Unforeseen Allowance</u> means the budget established for hazardous substances and underground conditions that differ from representations in the Contract Documents or Due Diligence Documents and meet the requirements under Article 13.15.5 and 18.4. The Unforeseen Allowance may also include other costs as allocated in the District's sole and absolute discretion related to the Project. The District, in its sole and absolute discretion, may use the District Contingency to fund any costs allowed under the Unforeseen Allowance. Any funds remaining in the Unforeseen Allowance at the completion of the Project shall remain unspent and allocated to the District as the District sees fit to use.
- 4.48 Work shall include all labor, materials, services and equipment necessary for the Contractor to fulfill all of its obligations pursuant to the Contract Documents as it pertains to each individual \_\_\_\_\_\_\_ Project. It shall include extension of Contractor's obligations to Subcontractor to perform Subcontractor Due Diligence including, but not limited to, visiting the Site of the proposed Work (a continuing obligation after the commencement of the Work), fully acquainting and familiarizing itself with the conditions as they exist and the character of the operations to be carried out under the Contract Documents, and make such investigation as it may



see fit so that it shall fully understand the facilities, physical conditions, and restrictions attending the Work under the Contract Documents. Each such Contractor or Subcontractor shall also thoroughly examine and become familiar with the Drawings, Specifications, and associated Contract Documents.

4.49 Workers include laborers, workers, and mechanics.

#### 5. ESTABLISHMENT OF GUARANTEED MAXIMUM PRICE "GMP"

- 5.1 <u>Guaranteed Maximum Price (GMP)</u> is the amount agreed upon between the District and Contractor that shall not be exceeded for the Construction of each Project within the Contract Time based on Contractor's thorough review of the Contract Documents, Due Diligence in investigation of all aspects of the Project, as set forth in Exhibit K for each Project. The GMP includes the costs for the Sublease Payments being paid by the District as Progress Payments and Retention Payment during construction in accordance with the terms of this Master Construction Services Agreement. Any references to Progress Payments shall also mean Sublease Payments. A Construction Contingency (Article 5.2.1) and an Errors and Omissions Contingency (Article 5.2.2) is contained within the GMP. Costs that are outside of the GMP shall be as follows:
  - 5.1.1 Owner requested additional work (See Article 8) to be paid under the District Contingency.
  - 5.1.2 Unforeseen underground soil conditions or unforeseen hazardous materials that meet the requirements of Article 13.15.5 and 18.4 to be paid under the Unforeseen Allowance.
- 5.2 GMP. As a result of the Due Diligence of Contractor, the GMP for each Project is set forth under Article 3 and in Exhibit K for each Project. The GMP is based upon all Due Diligence performed, the approved Construction Documents, and all other Contract Documents existing and reviewed by the Contractor at the time this Master Construction Services Agreement is entered into as more fully described and referenced in the Scope of Work set forth in Exhibit "A." Contractor's detailed line item costing of the Project, or Master Budget, totaling the GMP is attached hereto as Exhibit "B." Furthermore, the District and Contractor represent and warrant that the GMP is separate and distinct from the Sublease Payments to be paid by the District under the Sublease. District represents and warrants and Contractor acknowledges that: 1) the total amount of Sublease Payments and any optional Prepayment under the Master Sublease include the total rental for the Project, which total does not exceed the fair market value for the Project, 2) said rental amount is separate and distinct from Progress Payments and Retention, and 3) said rental amount shall be paid by the District with District non-local match contribution local funds.

The GMP is an "all inclusive" price for the construction of the Project that is calculated after Due Diligence and shall not be exceeded except as set forth in this Agreement. Contractor has taken on all contingencies and calculated those contingencies out in the form of the Construction Contingency. Contractor specifically agrees that once the Construction Contingency is fully exhausted, that Contractor can and shall Complete the Project pursuant to the terms of this Agreement within the Contract Time. No disputes concerning compensation, extras, or application of Contingencies shall be utilized as grounds to slow down or to stop work. The following two contingencies have been calculated through the Due Diligence of the Contractor and shall be calculated against the contingency amounts based on application of the Change Order language of Article 17.

5.2.1 Construction Contingency. The Construction Contingency set forth at Article 3.8.1 is for the use of the Contractor, as approved by the District, to pay for miscellaneous work items which are required to complete the Project including to cover trade scope gaps, missed work, areas of damage that may occur between trades during construction,



Subcontractor coordination problems, and Contractor coordination errors. The Contractor shall not use the Construction Contingency to pay for costs related to the following: (a) errors or omissions in the Construction Documents; (b) discrepancies with the Construction Documents pertaining to applicable building code requirements; and/or (c) enhancements or additions to the Scope of Work desired by the District. The Contractor shall obtain written approval from the District prior to using the Construction Contingency. The following may be considered, at the District's sole discretion, valid Construction Contingency items: 1) overtime and premium time, 2) costs to address safety items, 3) Contractor coordination issues and errors, 4) scope gaps, 5) trade damage, and 6) for other items requested by the Contractor if approved by the District and in the District's sole discretion. If on Final Completion of the Project, funds are remaining in the Construction Contingency, such funds shall remain unspent and allocated to the District as the District sees fit to use.

5.2.2 Errors and Omissions Contingency. Within the GMP shall be a line item amount to cover errors and omissions in the Construction Documents ("Errors and Omissions Contingency"). The Errors and Omissions Contingency at Article 3.7.2 is calculated based on coordination review of the Construction Documents and coordination meetings that have been held with the Subcontractors and Architect. Specifically, it is the coordination items that could not be addressed through coordination meetings and a factor determined based on the coordination review that has been performed by Contractor. The Errors and Omissions Contingency is created from Contractor's Due Diligence and based on Contractor's experience on similar projects. As a result, Contractor agrees that Contractor shall not seek to charge District for Errors and Omissions in excess of the Errors and Omissions Contingency, where such Errors and Omissions should have reasonably been discovered by Contractor during the performance of the preconstruction services and/or Due Diligence.

Contractor shall notify the District under the Change Order Provisions of the need for such work and specifically identify the Work as Errors and Omissions by submitting to the District for its consideration and approval or disapproval, a written request for the work before such work is performed. If District approves such request in writing, the costs of the work, shall be added to or deducted from the Errors and Omissions Contingency within the GMP. Any funds remaining in the Errors and Omissions Contingency at the completion of the Project shall remain unspent and allocated to the District as the District sees fit to use, except for any portions of Savings added to the Errors and Omissions Contingency, which Savings shall be allocated between the parties as provided in Article 7 below.

# 5.3 Due Diligence

- 5.3.1 Documents Reviewed. Contractor has visited the site, entered and evaluated the structures on the site, reviewed all as-built information, environmental reports, Asbestos Hazard Emergency Response Act of 1986 reports applicable to the Project, lead reports, reports on any other hazardous substances, reviewed environmental impact reports, reviewed applicable mitigation measures for the Project, reviewed and observed the current site conditions, reviewed available records from City and/or County Records on the Project. All documents provided or reviewed by the Contractor shall be referred to collectively as the Due Diligence Documents.
- 5.3.2 Review of Existing Conditions. Contractor must have performed basic confirmation of the As-Built information that exists as part of the Due Diligence process. This basic confirmation shall include:
- 5.3.3 <u>Confirmation of overall dimensions</u> of major column lines, location of elements where coordination of new construction to existing construction is to occur, confirmation that



the rooms noted are located on the drawings, review and confirmation that rooms have not been reconfigured.

- 5.3.3.1 Confirmation of location for utilities and supporting infrastructure.

  Contractor shall review the utilities and confirm that the infrastructure from the As-Builts and Contract Documents are consistent with the actual As-Built Conditions of the Project site.
- 5.3.3.2 <u>Confirmation that fire/life safety elements</u> are consistent with expectations of the Contract Documents. Specifically, confirmation of the integrity of one-hour corridors, fire separations, working fire sprinklers, working fire alarms, communications systems, EMS systems, and other systems that are to remain in use and relied upon as part of the anticipated Project.
- 5.3.3.3 Review of the Environmental Documents (Asbestos, Lead, PCB's, etc.) and general confirmation that the scope of hazardous substances is consistent with that which is shown on the environmental reports that are provided.
- 5.3.3.4 Confirmation of Working hours and specific conditions which will affect the ability to work. Contractor shall check requirements for the local city and county and confirm working hours and days, testing schedules at the District for days when work shall not occur, other critical days when work cannot occur, mitigation measures in the EIR or Negative Declaration that may affect the ability to Work on the Project. This review shall help Contractor build a working schedule for the Project.
- 5.3.4 Review of Construction Documents. Contractor has performed a complete and diligent review of all plans, specifications, addenda, bulletins or other documents provided as the Construction Documents or otherwise mentioned in the Construction Documents. The Contractor has written and submitted RFIs to address potential design issues prior to the GMP development to obtain a comprehensive GMP that addresses design and constructability issues.
- 5.3.5 *Inconsistencies*. All inconsistencies, timing or sequences which appear to be in error in the Construction Documents shall promptly be called to the attention of the Architect or, Engineer, for interpretation or correction. Local conditions which may affect the structure shall be brought to the Architect's attention at once. In no case, shall the instruction of the Architect be construed to cause work to be done which is not in conformity with the approved plans, specifications, change orders, construction change documents, and as required by law. (See Title 24 Section 4-343)
- 5.3.6 *Coordination Review.* Contractor shall perform a constructability review of the Construction Documents as part of its Due Diligence to determine the level of Errors and Omissions that should be included in the Errors and Omissions Contingency.
- 5.3.7 Price Fluctuations. As part of Contractor's Due Diligence responsibilities, Contractor is required schedule and plan to order, obtain, and store materials and equipment sufficiently in advance of its Work at no additional cost to assure that there will be no delays. Contractor understands that this may be a multi-year contract and that materials fluctuate in value and shall have adequately addressed market fluctuations through agreements with Contractor vendors or by other means. Contractor further understands and incorporates into Contractor's bid or proposal cost any wage rate increases during the Project for the Contractor's labor force as well as all other subcontractor and vendor labor forces. Contractor also understands the length of the Project schedule and has



incorporated an appropriate budget to include labor, material, and equipment escalation costs into the GMP. At no time will the District accept any costs associated with these increases. District shall not be responsible for market fluctuations in costs or labor rate increases during the Project. Contractor further has incorporated any and all cost increases in areas of Work where there may be schedule variations so that cost increases are not passed through to the District.

- 5.3.8 *Coordination Review*. Contractor has thoroughly reviewed the plans, specifications, and other Due Diligence Documents and satisfied itself that the Construction Contingency is adequate to complete the Project for the GMP.
- 5.3.9 *Due Diligence Determinations*. Contractor has utilized all the available Due Diligence information to verify that the contingencies are adequate and that the Project can be constructed without exceeding the GMP:
  - 5.3.9.1 Construction Contingency. Based on review of the scope of work submitted from each Subcontractor, Contractor's Due Diligence and review shall be utilized to determine the size of the Construction Contingency to cover unforeseen conditions (other than noted in Article 5.1), cover trade scope gaps, missed work, areas of damage that may occur between trades during construction, Subcontractor coordination problems, Contractor coordination errors, and miscellaneous work items.
  - 5.3.9.2 <u>Errors and Omission Contingency.</u> Based on a thorough review of the available Construction Documents and information located pursuant to the Due Diligence performed, a set-aside has been made for an Errors and Omissions Contingency that may be utilized to compensate for construction work to correct Errors and Omissions in the Construction Documents.
  - 5.3.9.3 <u>District Contingency (sometimes called Owner Contingency).</u> District Contingency is a sum that is set aside by the District to address any additional services. In the District's sole discretion, design errors or omissions as determined by the District (to the extent the Errors and Omissions Contingency is exhausted) and unforeseen conditions as approved by the District, may be allocated to the District Contingency. Specifics on application of the Owner Contingency are set forth at Article 8.
  - 5.3.9.4 <u>Unforeseen Allowance</u>. Unforeseen Allowance is a sum set aside for unforeseen conditions that differ from representations in the Contract Documents or Due Diligence Documents or meet the requirements under Article 13.15.5 and 18.4. The Unforeseen Allowance may also include other costs as allocated in the District's sole and absolute discretion related to the Project.
- 5.3.10 Schedule. Contractor's Due Diligence will also be critical to the Contractor's determination of the number of days required to complete the Project. Contractor will determine if the suggested number of days from the District and Architect can be performed and shall also consider whether the Project requires Governmental or Rain day float that exceeds that set forth in Article 9. If Contractor does not note any concerns with the suggested Contract Time, then it is presumed that Contractor is in agreement with the proposed completion date the Contractor, by entering into this Agreement, has determined for itself that the Project Contract Time is realistic, reasonable and includes all required Float under Article 9.



#### 6. OPEN BOOK ACCOUNTING AND SELECTION OF SUBCONTRACTORS

- 6.1 Open Book Accounting. The Contractor's GMP shall be based on actual procured quotes and bids from Subcontractors, vendors, and suppliers or based on estimated costs. In addition, Contractor shall include an estimated overhead and profit line item along with the cost for Contractor supplied labor. This total construction cost, or Base Cost, shall be added to Subcontractor, vendor and supplier contingencies and the Construction Contingency (which includes an Errors and Omissions Contingency) to form the entire GMP. As costs are incurred during the course of the Project, the Job Cost Accounting shall be updated to include actual costs incurred. A report on costs shall be prepared as part of the GMP process and shall be provided on a regular basis to the District.
  - 6.1.1 Purpose. While competitive bidding is often viewed as the lowest price, utilizing the lowest bid neither results in the best contractor, efficient construction, or a properly completed product. In some cases, the Project becomes significantly more expensive because competitive bid contractors either don't understand the drawings, aren't qualified to build the Project, or are seeking to utilize the legal process to make money by bringing claims against the District. The lease leaseback methodology provides the ability to negotiate for the most qualified competent contractor and allow coordination and interaction between the Contractor, Architect and District to alleviate unnecessary problems or areas that would result in claims. However, in exchange for this flexibility and reduction in claims, it is in the District's best interests, as a public entity, to ensure that the Project accounting information is available for review and the financial aspects of the Project can be fully reviewed. Thus, Contractor agrees that all job cost information shall be kept in an "open book" manner, shall show the actual transactions that occurred for the Project and shall be disclosable to the State if State funds are being utilized.
  - 6.1.2 State Allocation Board Issues. The Office of Public School Construction, the administering agency for the State Allocation Board, audits the costs for construction under the general authority of Education Code section 17076.10 and under the specific authority of Regulation Section 1859.100 et seq. governing program accountability audit, material inaccuracy, and expenditure audits. Given the fact the State has approved the lease-leaseback delivery method, and the likelihood that the records of the Project will be audited if there are State Funds involved, a permanent record of all the financial transactions for the construction of the Project shall be available through an Open Book Accounting of the Project expenditures of both hard and soft costs including, but not limited to labor, material and services costs, including the subcontract and material costs that were utilized to build the Project.
  - 6.1.3 *Value Engineering During the Project.* In addition to Value Engineering addressed at Article 7 below, Contractor may have occasion where better pricing can be obtained from Subcontractors or suppliers. This better pricing shall be treated as part of Savings under Article 7.



- 6.2 <u>Scope Reduction Not Savings</u>. The District at all times shall have the right to reduce the scope of the Project. If the District reduces the scope of the Project, the GMP shall be reduced to reflect the reduced Scope of Work, pursuant to the provisions of Article 17. To the extent possible, it is the mutual goal of the District and Contractor to maximize the Scope of Work as allowed by the GMP. Reductions in scope are not considered Savings.
- 6.3 Selection of Subcontractors.
  - 6.3.1 If identified or requested in the District's Request for Proposal/ Qualifications ("RFP/RFQ"), the Contractor must use any Subcontractors identified and included in the Contractor's response to the District's RFP/RFQ pursuant to Education Code section 17406(a)(4). All Subcontractors identified and included in the Contractor's response to the District's RFP/RFQ shall be afforded the protections of the Subletting and Subcontracting Fair Practices Act (Chapter 4 (commencing with Section 4100) of Part 1 of Division 2 of the Public Contract Code).
  - 6.3.2 Following the award of the Contract to the Contractor by the District's Board of Education, and for all Subcontractors not identified in the Contractor's response to the District's RFP/RFQ, the Contractor shall proceed as follows in awarding construction Subcontracts with a value exceeding one-half of one percent of the price allocable to construction work:
    - 6.3.2.1 Provide public notice of availability of work to be subcontracted in accordance with the publication requirements applicable to the competitive bidding process of the District, including a fixed date and time on which qualifications statements, bids, or proposals will be due.
    - 6.3.2.2 Establish reasonable qualification criteria and standards.
    - 6.3.2.3 Award the subcontract either on a best value basis or to the lowest responsible bidder. The process may include prequalification or short-listing. The process shall not apply to Subcontractors identified and included in the Contractor's response to the District's RFP/RFQ. Subcontractors awarded construction subcontracts under this Article 6.3.2 shall be afforded all the protections of the Subletting and Subcontracting Fair Practices Act (Chapter 4 (commencing with Section 4100) of Part 1 of Division 2 of the Public Contract Code).
    - 6.3.2.4 All MEP Subcontractors must be prequalified as set forth in Article 1.12 above.
  - 6.3.3 In no case will the Contractor award any subcontracts until the District has concurred to the scope and price of the subcontracted services.
  - 6.3.4 All subcontractors (of any tier) performing any portion of the Work must comply with the Labor Code sections 1725.5 and 1771.1 and must be properly and currently registered with the California Department of Industrial Relations and qualified to perform public works pursuant to Labor Code section 1725.5 throughout the duration of the Project.
  - 6.3.5 Contractor shall provide the District with full documentation regarding the bids or competitive quotes received by Contractor. In no event shall such documentation be redacted or obliterated. In the event the Contractor does not comply with this provision, the District may terminate this Construction Services Agreement in accordance with the provisions of Article 19 below.



6.3.6 Compliance with Disabled Veteran Business Enterprise (DVBE) contracting goals is required under this Construction Services Agreement. In accordance with Education Code section 17076.11 the District has a DVBE participation goal of 3% per year of the overall dollar amount of state funds allocated to the District pursuant to the Leroy F. Greene School Facilities Act of 1998, and expended each year by the District. The District is seeking DVBE participation under this Construction Services Agreement. The Contractor must require Subcontractors to make a good faith effort to contact and utilize DVBE contractors and suppliers in securing bids for performance of the Project. Their efforts shall be documented on the DVBE Good Faith Effort Form attached as Attachment No.5.

#### 7. SAVINGS AND VALUE ENGINEERING

- 7.1 General Intent. The purpose of Savings is to minimize the expenditure of funds for the construction of the Project on items that exceed the minimum criteria required without a corresponding benefit to the District. The District also wishes to eliminate any excess quality levels or performance criteria provided in the Construction Documents so long as such elimination does not alter the design, aesthetics, safety standards or configuration or space, and does not increase future maintenance and operation costs. The District and the Contractor shall work cooperatively with each other, in good faith, to identify appropriate opportunities to reduce the Project costs and promote Savings. There are two stages when Savings may be generated. They are (1) Value Engineering when establishing the GMP and (2) Savings generated through changes, reductions, or Subcontractor negotiations that may occur after the GMP is established.
  - 7.1.1 Value Engineering is a review of systems so excess quality, unnecessary design elements, reconfiguration for efficiency, or other changes may be made to reduce the cost of a project. Sometimes, timing and sequences or re-use of materials that are unique to a project or area may generate savings. For example, if export soil is generated on a site which may have a substantial cost for transportation and removal could be sold to offset the costs incurred then a savings may be generated for the Project. Similarly, if concrete is ground, it may be sold for aggregate rather than as demolished construction materials.
  - 7.1.2 Other Savings generated over the course of the Project through Subcontractor negotiations, replacement of Subcontractors, or through other means shall be calculated as part of the overall costs for the Project as part of the "Open Accounting" of the Project and shall be counted towards Project Savings.
- Sharing and Calculation for Return of Savings. If Contractor realizes a Savings on an aspect of the Project, including but not limited to, Value Engineering or other Savings after the GMP is established and after execution of this Construction Services Agreement, such Savings shall be divided in the following proportion: Seventy Five Percent (75%) of any Savings shall be returned to the District and Twenty Five Percent (25%) of any Savings shall be returned to the Contractor. Calculation of Savings shall be determined by adding all expenses for the Project (excluding Change Orders and Owner and Construction Contingency Expenses), separating out overhead costs and either using the actual overhead costs, or the percentage set for overhead in the Article 5.3, whichever is higher an applying the percentage for profit against the GMP (less Change orders, Owner and Construction Contingency). Any remaining money shall be considered Savings. If the Project expenses exceed the GMP, then there are no Savings for the Project and the GMP shall apply. A separate calculation of whether there are savings associated with Change Orders under the Owner and Construction Contingency may be performed to determine if there are any savings that remain on these areas and applied to the overall savings calculation
- 7.3 <u>Savings Determined Through Audit.</u> District may, at its own costs, have an audit conducted of the Project related job costs to determine Savings as further outlined in Article 21.



# 8. <u>DISTRICT CONTINGENCY</u>

- 8.1 The District Contingency is an allowance for use by the District that can be used to pay the Contractor to perform additional services ("Additional Services") not described in this Construction Services Agreement. This District Contingency is outside of the GMP, is not part of the original bond, except to the extent that District contingency is utilized as a Change to the Contract under Article 17, and may be used for Owner requested additions, revisions to the Project, moving furniture or equipment, and other District unforeseen items. Contractor shall provide a cost estimate and a written description of the Additional Services required to perform such work. The District shall set aside a contingency amount outside the GMP, defined at Article 5 ("District Contingency") in the amount set forth at Article 3, which District Contingency shall be used for such Additional Services. Compensation for such Additional Services shall be negotiated and agreed upon in writing, in advance of Contractor's performing or contracting for such Additional Services. Nothing in this Construction Services Agreement shall be construed as limiting the valuation and amount to be paid to Contractor for such Additional Services or its implementation should a written agreement for such services be executed. Contractor shall not be entitled to compensation for Additional Services required as a result of Contractor's acts, errors or omissions. Further any Architectural Errors and Omissions shall not come out of District Contingency unless agreed upon in writing by the District in its sole discretion.
- 8.2 Additionally, while District is in no way limited by the manner in which it decides to utilize the District Contingency, said District Contingency shall not be used for any costs associated with errors or omissions in the Construction Documents until such time, if ever, the Errors and Omissions Contingency has been fully exhausted. Any funds remaining in the District Contingency at the completion of the Project shall remain unspent and remain allocated to the District.

# 9. SCHEDULE

- 9.1 <u>Contract Time:</u> Contractor shall perform and reach Substantial Completion (See Article 4.45) within the time specified in the Agreement. Moreover, Contractor shall proceed on a properly developed and approved CPM Master Baseline Schedule, which represents the Contractor's view of the practical way in which the Work will be accomplished. Note that Contract Time includes and incorporates all Float and other Baseline inclusions as noted in Article 9.3 and as otherwise specifically noted in Article 9
- 9.2 <u>Float</u> is the total number of days an activity may be extended or delayed without delaying the Completion Date shown in the schedule. Float will fall into three categories: (1) Rain Days; (2) Governmental Delays; and (3) Project Float. Project Float and Rain Days are owned by the Project and may be utilized as necessary for critical path delays once the days become available for consumption (i.e. the rain day arrives and is not utilized since rain did not occur or Work was performed on the interior of a building). However, Governmental Delay float shall not be utilized for purposes other than to address critical path delays that arise due to approvals, Inspector approvals or verifications on governmental forms.
  - 9.2.1 Governmental Delay Float. Given DSA requirements for submission and approval of CCD's prior to a DSA Form 152 sign off on areas of Work that deviate from approved Construction Documents, and the anticipated delays that may arise from this CCD procedure, no less than twelve (12) days per calendar year shall be set aside as Governmental Float to be utilized on critical path delays. A pro-rated number of days shall be calculated based on length of Contract Time. (For example, a two (2) year Contract Time shall require twenty-four (24) days of Governmental Float. If the Contract Time is 182 days, then the Contract Time shall require six (6) days of Governmental Float) This Governmental Delay float must be incorporated into the schedule and should be incorporated in each critical activity as Contractor deems fit. Specifically, major categories of Work under the DSA 152 (Project Inspection Card) should be allocated Governmental Delay Float at the Contractor's discretion.



Governmental Delay Float on the Project may exceed 12 days per one (1) year period, but Contractor is required to include not be less than 12 days of Governmental Delay Float during each one (1) year period.

- 9.2.2 Inclement Weather (Rain Days). The Contractor will only be allowed a time extension for unusually severe weather if it results in precipitation or other conditions which in the amount, frequency, or duration is in excess of the norm at the location and time of year in question as established by the National Oceanic and Atmospheric Administration (NOAA) weather data. No less than 22 calendar days for each Calendar year for Southern California. The NOAA weather related days (22 days in Southern California) shall be set aside as float within the Baseline Schedule. Additional days beyond the NOAA shall be considered under the same criteria that weather days are granted below.
- 9.2.3 Granting of Days beyond those Anticipated. A Rain Day shall be granted by Architect or CM if the weather prevents the Contractor from beginning Work at the usual daily starting time, or prevents the Contractor from proceeding with seventy-five (75%) of the normal labor and equipment force towards completion of the day's current controlling item on the accepted schedule for a period of at least five hours, and the crew is dismissed as a result thereof, the Architect will designate such time as unavoidable delay and grant one (1) critical path activity calendar-day extension if there is no available float for the calendar year.
- 9.2.4 *Project Float* is all remaining float, including extra days included in a particular activity.
- 9.3 <u>Inclusions in Baseline.</u> In addition to Scheduling requirements set forth at Article 9, Contractor is specifically directed to include in Contractor's Baseline Schedule and all Schedule updates that provide for the following items required pursuant to this CSA, including but not limited to:
  - 9.3.1 Rain Day Float (excluding inclement weather) as required under Article 9.2.2. For example, if the NOAA provides 22 days of rain days, all 22 days must be incorporated and noted in the schedule. Further, any days required to clean-up or dry out shall be included for operations that are likely to require a clean-up or dry out period. Days that are not utilized shall be considered float owned by the Project.
  - 9.3.2 Governmental Delay Float under Article 9.2.1. This Governmental Delay Float shall only be utilized for Governmental Delays and shall not be considered available float owned by the Project. This float shall be distributed to the Project as granted and approved by the District, and shall be used to offset liquidated damages for overstaying the Lease, and shall not generate compensable delays.
  - 9.3.3 Submittal and Shop drawing schedule under Article 9.6 and 15.6.
  - 9.3.4 Deferred Approvals under Article 15.3 and 15.6
  - 9.3.5 Time for separate contractors, including furniture installation and start up activities, under Article 32.
  - 9.3.6 Coordination and timing of any drawings, approvals, notifications, permitting, connection, and testing for all utilities for the Project. Article 13.15.2.
  - 9.3.7 Testing, special events, or District activities.
- 9.4 <u>Schedule Updates.</u> Contractor shall update the schedule each month to address actual start dates and durations, the percent complete on activities, actual completion dates, estimated remaining duration for the Work in progress, estimated start dates for Work scheduled to start at future times and changes in duration of Work items



- 9.4.1 Listing of Items Causing Delays. Schedule Updates shall provide a listing of activities which are causing delay in the progress of Work and a narrative shall be provided showing a description of problem areas, anticipated delays, and impacts on the Construction Schedule. Simply stating "District Delay" or "Architect Delay" shall be an inadequate listing.
- 9.4.2 Recovery Schedule. In addition to providing a schedule update every thirty (30) days, the Contractor, shall take the steps necessary to improve Contractor's progress and demonstrate to the District and Architect that the Contractor has seriously considered how the lost time, the Completion Date, or the milestones that are required to be met within the terms of the Contract. Contractor shall provide a Recovery Schedule showing how Milestones and the Completion Date will be met.
  - 9.4.2.1 <u>Failure to Provide a Recovery Schedule.</u> Shall subject Contractor to the assessment of Liquidated Damages for failure to meet the Contract Time.
- 9.5 <u>Time of the Essence.</u> Time limits stated in the Contract Documents are of the essence to the Contract. By executing the Agreement, the Contractor confirms that the Contract Time is a reasonable period for performing the Work
- 9.6 <u>Time for Preparing Submittals Must Be Incorporated in Schedule</u>: Contractor shall include Submittals as line items in the Baseline Schedule. Time for preparing and coordinating Submittals shall not delay the Work, Milestones, or the Completion Date, and shall be in conformance with Article 15.6.

# 10. <u>INSPECTION OF WORK/ INSPECTOR AND ARCHITECT</u>

- 10.1 <u>Inspection of Work/Inspector</u>. The District shall hire its own Division of State Architect Inspector as required by law. District, District's Representatives, and the Division of the State Architect shall at all times have access to the work whether it is in preparation or progress, and Contractor shall provide proper facilities for such access and for inspection.
  - 10.1.1 *General*. One or more Project Inspectors employed by the District and approved by the Division of the State Architect will be assigned to the Work in accordance with the requirements of Title 24 of the California Code of Regulations. The Inspector(s) duties are as specifically defined in Title 24 Section 4-333 and 4-342 and in DSA IR A-8.
  - 10.1.2 Inspector's Duties and DSA Noted Timelines for Inspection. All Work shall be under the observation of the Inspector. Contractor shall establish a protocol for requesting inspection with Inspector so as to not delay the Work and provide adequate time for the Inspector to perform inspection. If such a protocol is not established ahead of time, Inspector may utilize the time criteria set by Title 24 of 48 hours in advance of submitting form DSA 156 for each new area. The Inspector shall have free access to any or all parts of the Work at any time. The Contractor shall furnish the Inspector such information as may be necessary to keep the Inspector fully informed regarding progress and manner of Work and character of materials. Such observations shall not, in any way, relieve the Contractor from responsibility for full compliance with all terms and conditions of the Contract, or be construed to lessen to any degree the Contractor's responsibility for providing efficient and capable superintendence. The Inspector is not authorized to make changes in the drawings or specifications nor shall the Inspector's approval of the Work and methods relieve the Contractor of responsibility for the correction of subsequently discovered defects, or from its obligation to comply with the Contract Documents.
  - 10.1.3 *Electronic Posting.* Inspector shall electronically post DSA required documents on the DSA electronic posting website. It is the Contractor's responsibility to determine the



status of posting and determine if all the criteria for sign off of a category of Work on the Project Inspection Card (Form DSA 152) as defined more thoroughly in the most current version of the DSA 152 manual posted on the DSA website.

- 10.1.4 *Incremental Approvals under PR-13.* Inspector may collaborate with Contractor about approval of areas that may be constructed and approved incrementally under the DSA 152 card pursuant to the guidelines of PR-13. Inspector shall work with Contractor to present incremental approval proposals to DSA.
- 10.1.5 Inspector's Authority to Reject or Stop Work. The Inspector shall have the authority to reject Work whenever provisions of the Contract Documents are not being complied with, and Contractor shall instruct its Subcontractors and employees accordingly. In addition, the Inspector may stop any Work that poses a probable risk of harm to persons or property. The Contractor shall instruct its employees, Subcontractors, material and equipment suppliers, etc., accordingly. The absence of any Stop Work Order or rejection of any portion of the Work shall not relieve the Contractor from any of its obligations pursuant to the Contract Documents.
- 10.1.6 *Inspector's Facilities.* Within seven (7) days after notice to proceed, the Contractor shall provide the Inspector with the temporary facilities as required. More specific requirements for the Inspector facilities may be further described under Division 1 of the Specifications.
- 10.1.7 Testing Times. The District will provide inspection and testing at its cost during the normal eight (8) hour day Monday through Friday (except holidays). Work by the Contractor outside of the normal eight (8) hour day shall constitute an authorization from the Contractor to the District to provide inspection and testing as required outside of the normal eight (8) hour day. Contractor shall provide adequate time for inspections so as to not delay the Work. An advanced timing protocol may be established pursuant to Article 10. If the Contractor is behind Schedule then it is incumbent on the Contractor to provide advance forecast through look ahead of the anticipated date for inspection so the Inspector may plan their activities so as to not delay the Project. Contractor shall reimburse District for any additional costs associated with inspection and testing (including re-inspection and re-testing) outside the normal eight-hour day and for any retests caused by the Contractor pursuant to Article 10.4.
- 10.1.8 Contractor Is Required to Coordinate Testing and Inspections. It is the Contractor's responsibility to request special inspections with sufficient time so all testing may be timely completed and posted so work may proceed and the Inspector's signature is attached to the Project Inspection Card (Form 152). Specifically, timely request for special inspection under the DSA Verified Report Forms 291 (laboratory), DSA Verified Report Form 292 (Special Inspection), and DSA Verified Report 293 (geotechnical) since DSA requirements under PR 13-01 specifically gives the Special Inspections 14 days to post to the DSA website. It is the Contractor's responsibility to timely schedule and pay (if applicable) for Special Inspections as to not delay the Project, and any failure or resulting delay is not considered Governmental Delay Float under Article 9.2.1.
- 10.1.9 Special Inspection Out of State, Out of Country or Remote from Project. If Contractor has a Subcontractor or supplier that requires in plant or special inspections or tests that are out of the country, out of state or a distance of more than 200 miles from the Project site, the District shall provide the Special Inspector or individual performing tests time for inspection and testing during normal work hours. Contractor, however, is responsible for the cost of travel, housing, food, out of area premiums that may be in the Inspector/Testing Agreement with District, or other expenses necessary to ensure proper inspection or testing is provided by a DSA Certified Inspector, Special Inspector,



or individual performing tests. In some cases all three (DSA Inspector, Special Inspector, and Testing) may be required. In addition, if the DSA Certified Inspector, Special Inspector, or individual performing test has contractual travel clauses or special rates for out of town inspection, Contractor is responsible for all costs associated with the contractual travel costs in addition to all other costs. Arrangements for inspection and/or testing shall be made far enough in advance so as to not delay the Work.

- STOP WORK ORDER. DSA may issue a Stop Work Order, or an Order to Comply, when either (1) the Work proceeds without DSA approval; (2) the Work proceeds without a DSA Project Inspector, or (3) where DSA determines that the Work is not being performed in accordance with applicable rules and regulations, and would compromise the structural integrity of the Project or would endanger lives. If a Stop Work Order is issued, the Work in the affected area shall cease until DSA withdraws the Stop Work Order. Pursuant to Education Code section 17307.5(b) and Education Code section 81133.5, the District shall not be held liable in any action filed against the District for any delays caused by compliance with the Stop Work Order, except to the extent that an error or omission by the District is the basis for the issuance of the Stop Work Order.
- 10.3 <u>Inspector's Field Office.</u> Contractor shall provide for the use of inspector a separate trailer or temporary private office of not less than seventy five square feet of floor area to be located as directed by District and to be maintained until removal is authorized by District. The Office shall be of substantial waterproof construction with adequate natural light and ventilation. Door shall have a key type lock or padlock hasp. The Inspector's field office shall have heating and air-conditioning and shall be equipped with a telephone, internet connection, working computer, a fax machine and use of an on-site copier at Contractor's expense. A table satisfactory for the study of plans and two chairs shall be provided by Contractor. Contractor shall provide and pay for adequate electric lights, and adequate heat and air conditioning for the field office until authorized removal.

# 10.4 RESPONSIBILITY FOR ADDITIONAL CHARGES INCURRED BY THE DISTRICT FOR PROFESSIONAL SERVICES

- 10.4.1 If at any time prior to the completion of the requirements under the Contract Documents, the District is required to provide or secure additional professional services (including CM, Inspection, Architect, Engineering and Special Consultant Services) for any reason by any act of the Contractor, the District may seek a Deductive Change Order for any costs incurred for any such additional services, which costs shall be deducted from the next scheduled Progress Payment. A Deductive Change Order shall be independent from any other District remedies and shall not be considered a waiver of any District rights or remedies. If payments then or thereafter due to the Contractor are not sufficient to cover such amounts, the Contractor shall pay the difference to the District. Additional services shall include, but shall not be limited to, the following:
  - Services made necessary by the default of the Contractor (Article 19 or Article 12.2).
  - b) Services made necessary due to the defects or deficiencies in the Work of the Contractor.
  - c) Preparation of a CCD or ICD to correct a Contractor Deficiency, or Contractor Caused Notices of Non-Compliance (Article 17.2)
  - Services required by failure of the Contractor to perform according to any provision of the Contract Documents.
  - e) Services in connection with evaluating substitutions of products, materials, equipment, Subcontractors' proposed by the Contractor, and making subsequent revisions to drawings, specifications, obtaining DSA approvals, DSA costs for



- review of CCD's, other governmental agency review costs, and providing other documentation required (except for the situation where the specified item is no longer manufactured or available). (Article 16
- f) Services for evaluating and processing Claims or Disputes submitted by the Contractor in connection with the Work outside the established Change Order or Claims or Disputes process.
- g) Services required by the failure of the Contractor to prosecute the Work in a timely manner in compliance within the specified time of completion.
- Services in conjunction with the testing, adjusting, balancing and start-up of equipment other than the normal amount customarily associated for the type of Work involved.
- i) Services in conjunction with more than one (1) re-review of Submittals of Shop Drawings, product data, samples, RFI's etc.

# 11. ARCHITECT

- 11.1 Architect's Status. In general and where appropriate and applicable, the Architect shall observe the progress and quality of the work on behalf of the District. The Architect shall have the authority to act on behalf of District only to the extent expressly provided in this Construction Services Agreement. After consultation with the Inspector and after using his/her best efforts to consult with the District, the Architect shall have authority to stop work whenever such stoppage may be necessary in his reasonable opinion to insure the proper execution of the Construction Services Agreement. Contractor further acknowledges that the Architect shall be, in the first instance, the judge of the performance of this Construction Services Agreement
- 11.2 <u>Architect's Decisions.</u> Contractor shall promptly notify District in writing if the Architect fails within a reasonable time, make decisions on all claims of the District or Contractor and on all other matters relating to the execution and progress of the Project.

#### 12. DISTRICT RESPONSIBILITIES

- District Site Representations. District warrants and represents that, District has, and will continue to retain at all times during the course of construction, legal title to the Site and that said land is properly subdivided and zoned so as to permit the construction and use of said Site. District further warrants and represents that title to said land is free of any easements, conditions, limitation, special permits, variances, agreements or restrictions which would prevent, limit, or otherwise restrict the construction or use of said facility. However, in the event easements for permanent structures or permanent changes in existing facilities are necessary, they shall be secured and paid for by District, unless otherwise specified. Reference is made to the fact that District has provided information on the Site to Contractor. Such information shall not relieve the Contractor of its responsibility; and the interpretation of such data regarding the Site, as disclosed by any borings or other preliminary investigations, is not warranted or guaranteed, either expressly or implicitly, by the District. The Contractor shall be responsible for having ascertained pertinent local conditions such as location, accessibility and general character of the Site and for having satisfied itself as to the observable, known or documented conditions under which the work is to be performed.
- 12.2 Partial Default: District Right to Take Over Work (Two (2) day notice to Cure and Correct). If the Contractor Defaults or neglects to carry out the Work in accordance with the Contract Documents, the District may provide a two (2) business day written notice to cure (a shorter period of time in the case of Emergency or a critical path delay) Contractor's Partial Default in a specific segregated area of work. The District's right to issue a Partial Default of the Contractor's Work and take over that segregated area of Work includes, but is not limited to:



- a) Failure to supply adequate workers on the entire Project or any part thereof;
- b) Failure to supply a sufficient quantity of materials;
- c) Failure to perform any provision of this Contract;
- d) Failure to comply with safety requirements, or due to Contractor is creation of an unsafe condition;
- e) Cases of bona fide emergency;
- f) Failure to order materials in a timely manner;
- g) Failure to prepare deferred-approval items or Shop Drawings in a timely manner;
- h) Failure to comply with Contractor's Baseline or Update Schedule, meet critical Milestones which would result in a Delay to the Critical Path, or Delay the Contract Time;
- i) Failure to comply with the Subcontractor selection and award requirements under Education Code section 17406(a)(4);
- j) Failure to meet the requirements of the American's with Disabilities Act;
- k) Failure to complete Punch List work; or
- 1) Failure to proceed on an Immediate Change Directive.
- 12.2.1 Failure to correct a Notice of Deviation. If during the two (2) business day period, the Contractor fails to Cure and correct the deficiency noted in the notice of Partial Default with diligence and promptness, the District may correct such deficiencies without prejudice to other remedies the District may have, including a Termination for Cause as set forth in Article 19.
- 12.2.2 Service of Notice of Partial Default with Right to Cure. A written notice of Partial Default and right to Cure under Article 12.2("Article 12.2 Notice" or "Notice of Partial Default") shall be served by facsimile (with a copy provided by e-mail to the e-mail address provided and copied to the Project Superintendent).
- 12.2.3 Shortened Time for Partial Default in the Case of Emergencies. In an Emergency situation, the District may correct any of the deficiencies described in Article 12.2 without prejudice to other remedies by providing service of written notice of Emergency requiring a shortened time for Partial Default specifying the time given to Cure, if any.
- 12.2.4 Shortened Time for Partial Default in the Case of Critical Path Delay. In the case of critical path delay, the District may correct any of the deficiencies described in Article 12.2 without prejudice to other remedies providing service of written notice of Critical Path Delay to the Contractor with a specific description of the critical path delay items noting the line item or area of Work that is on the Critical Path and prescribe the length of shortened time to Cure, if any.
- 12.2.5 Written Notice of Partial Default to be Deducted by Deductive Change Order. The District shall have the right to determine the reasonable value of the Article 12.2 Partial Default Work, or if there is an actual value for the Work, shall use that value and issue a Deductive Change Orders under Article 17.6.



# 13. CONTRACTOR RESPONSIBILITIES.

- 13.1 Full Time Supervision. Contractor shall keep on the Work at all times during its progress a competent, English speaking construction Superintendent satisfactory to the District. The Superintendent shall be present on a full-time basis, shall be dedicated exclusively to the Project and shall not share superintendency duties with another project or job. The Superintendent shall not be replaced except with written consent of the District. The Superintendent shall represent the Contractor in its absence and shall be fully authorized to receive and fulfill any instruction from the Architect, the Inspector, the District or any other District representative (including CM in the cases where the District has a CM representative). All Requests for Information shall be originated by the Superintendent and responses thereto shall be given to the Superintendent. No Work shall begin on any day by any Subcontractor or other person on the Project site until the Superintendent has arrived, or shall any Work continue during the day after the Superintendent has departed from the The Superintendent shall have authority to bind Contractor through the Superintendent's acts. The Superintendent shall represent the Contractor, and communications given to the Superintendent shall be binding on the Contractor. Before commencing the Work, Contractor shall give written notice to District (and CM representative) and Architect of the name and a Statement of Qualifications of such superintendent. Superintendent shall not be changed except with written consent of District, unless a superintendent proves to be unsatisfactory to Contractor and ceases to be in its employ, in which case, Contractor shall notify District and Architect in writing. Contractor shall provide a replacement superintendent approved by the District prior to performing additional work.
- Staff. Notwithstanding other requirements of the Contract Documents, the Contractor and each Subcontractor shall: (1) furnish a competent and adequate staff as necessary for the proper administration, coordination, supervision, and superintendence of its portion of the Work; (2) organize the procurement of all materials and equipment so that the materials and equipment will be available at the time they are needed for the Work; and (3) keep an adequate force of skilled and fit workers on the job to complete the Work in accordance with all requirements of the Contract Documents.
- 13.3 Contractor shall notify District and Architect, in writing, when Contractor desires to change the Project Manager for the Project, and shall provide the information specified above. The new Project Manager cannot serve on the Project until approved by District. District shall have the right, at any time, to direct a change in Contractor's Project Manager if performance is unsatisfactory, as determined by District, in its sole discretion.
- 13.4 Contractor shall give efficient supervision to the work, using its skill and attention and shall cause working drawings and specifications to be prepared and submitted to the District. Following agreement by Contractor and District with respect to said working drawings and specifications, it shall be Contractor's responsibility to perform the work described in said working drawings and specifications in substantial compliance with the Construction Documents.
- 13.5 <u>Right to Remove</u>. District shall have the right, but not the obligation, to require the removal from the Project of any superintendent, staff member, agent, or employee of any Contractor, Subcontractor, material or equipment supplier.
- 13.6 <u>Discipline</u>. The Contractor shall enforce strict discipline and good order among the Contractor's and Subcontractor's employees, and other persons carrying out the Contract. The Contractor shall not permit employment of unfit persons or persons not skilled in tasks assigned to them. As used in this subsection, "unfit" includes any person who the District concludes is improperly skilled for the task assigned to that person, who fails to comply with the requirements of this Article, or who creates safety hazards which jeopardize other persons and/or property.
- 13.7 <u>Labor and Materials</u>



- 13.7.1 Contractor to Provide. Unless otherwise provided in the Contract Documents, the Contractor shall provide and pay for labor, material, equipment, tools, construction equipment and machinery, water, heat, air conditioning, utilities, transportation, and other facilities, services and permits necessary for proper execution and completion of the Work whether temporary or permanent and whether or not incorporated or to be incorporated in the Work.
- 13.7.2 Quality. Unless otherwise specified, all materials and equipment to be permanently installed in the Project shall be new and shall be of the highest quality or as specifically stated in the Contract Documents. The Contractor shall, if requested, furnish satisfactory evidence as to kind and quality of all materials and equipment within ten (10) days of a written request by the District, including furnishing the District with bona fide copies of invoices for materials or services provided on the Project. All labor shall be performed by workers skilled in their respective trades, and shall be of the same or higher quality as with the standards of other public school construction.
- 13.7.3 *Replacement.* Any work, materials, or equipment, which do not conform to these requirements or the standards set forth in the Contract Documents, may be disapproved by the District, in which case, they shall be removed and replaced by the Contractor at no additional cost or extension of time to the District.
- 13.8 Pre-Construction Orientation/Construction Meetings. The Contractor, in conjunction with the District and the Architect, shall conduct pre construction orientation conferences for the benefit of Subcontractors to orient the Subcontractors to the various reporting procedures and site rules prior to the commencement of actual construction. These Pre-Construction meetings shall include coordination of the Subcontractor Work to help reduce Errors and Omissions and Construction Contingency requests and shall incorporate the Constructability Due Diligence review done by Contractor.
- 13.9 Owner Meetings. The Contractor shall conduct construction and progress meetings with District Representatives, and Construction Managers that occur at least weekly and as otherwise requested by the District, to discuss such matters as procedures, progress problems and scheduling. The Contractor shall prepare and promptly distribute official minutes of such meetings to all parties in attendance including Architect, District and Inspector.
- 13.10 <u>Budget/Cash Flow Reports.</u> The Contractor shall incorporate approved changes as they occur, and develop cash flow reports and forecasts for submittal to the District on a monthly basis. The Contractor shall provide regular monitoring of the approved estimates of Construction Costs, showing actual costs for activities in progress, and estimates for uncompleted tasks. The Contractor shall identify variances between actual and budgeted or estimated costs, and advise the District and the Architect whenever Project costs exceed budgets or estimates. The Contractor shall maintain cost accounting records on authorized additional services or work performed under unit costs, additional work performed on the basis of actual costs of labor and materials, or other work requiring accounting records.
- 13.11 Progress Reports. The Contractor shall record the progress of the Project, and shall submit monthly written progress reports to the District and the Architect including information on the entire Project, showing percentages of completion and the number and amounts of proposed Extra Work/Modifications and their effect on the Construction Costs as of the date of the report. The Contractor shall also keep a daily log containing a record of weather, Contractors, work on the site, number of workers, work accomplished, problems encountered, and other similar relevant data as the District may require. The Contractor shall make the log available to the District and the Architect. The District shall be promptly informed of all anticipated delays. In the event that the Contractor determines that a schedule modification is necessary, the Contractor shall promptly submit a revised Schedule for approval by the District

#### 13.12 <u>Schedule of Values.</u>

- 13.12.1 *Break Down of Schedule of Values.* Schedule of Values shall be broken down by Project, site, building, milestone, or other meaningful method to measure the level of Project Completion as determined by the District. The schedule of values shall include, but not be limited, to Subcontractor costs, the costs for the Submittals, Punch Lists, Commissioning and Start-Up, Close Out Submittals, and As-Builts.
- 13.12.2 Based on Contractor Costs. The Schedule of Values shall be based on the costs from Contractor to the District. However, the submission of the Schedule of Values shall not be front loaded so the Contractor is paid a greater value than the value of the Work actually performed and shall not shift funds from parts of the Project that are later to Work that is performed earlier.
- 13.12.3 Largest Dollar Value for Each Line Item. Identify Subcontractors and materials suppliers proposed to provide portions of Work equal to or greater than ten thousand dollars (\$10,000) or one-half (1/2) of one percent (1%) of their Contract Price, whichever is less, or as otherwise approved in writing by the District.
- 13.12.4 *Allowances*. Any Allowances provided for in the Contract shall be a line item in the Schedule of Values.
- 13.12.5 *Labor and Materials Shall Be Separate*. Labor and Materials shall be broken into two separate line items unless specifically agreed in writing by the District.
- 13.12.6 *District Approval Required.* The District shall review all submissions of Schedule of Values received pursuant to this Article in a timely manner. All submissions must be approved by the District before becoming the basis of any payment.
- 13.13 <u>Scheduling.</u> Contractor shall complete the construction pursuant to the CPM Schedule as required under Article 9.
- As-Builts. Throughout the duration of the Project, Contractor shall maintain on a current basis an accurate and complete set of As-Built Drawings (and Annotated Specifications) clearly showing all changes, revisions to specifications and substitutions during construction, including, without limitation, field changes and the final location of all electrical and mechanical equipment, utility lines, ducts, outlets, structural members, walls, partitions, and other significant features. In case a specification allows Contractor to elect one of several brands, makes, or types of material or equipment, the annotations shall show which of the allowable items the Contractor has furnished. The Contractor will update the As-Built Drawings and Annotated Specifications as often as necessary to keep them current, but no less often than weekly.
  - 13.14.1 *Updates*. Contractor shall update As-Built Drawings with complete information on an area of Work at or near the time when the Work is being performed and prior to any DSA 152 sign off and prior to any Work being covered.
  - 13.14.2 *Storage*. The As-Built Drawings and Annotated Specifications shall be kept at the Site and available for review and inspection by the District and the Architect. Failure to maintain and update the As-Built Drawings is a basis to withhold scheduled Progress Payments pursuant to Article 29.4.
  - 13.14.3 *Upon Beneficial Occupancy*. Contractor shall obtain and pay for reproducible plans upon Beneficial Occupancy. Contractor shall deliver Plans to District Representative (Construction Manager if one is hired for the Project).
  - 13.14.4 As-Builts at Completion of Work. On completion of the Work and prior to and as a condition precedent to the Application for Retention Payment, the Contractor will



provide one neatly prepared and complete set of As-Built Drawings and Annotated Specifications to the District. Contractor shall certify the As-Builts as a complete and accurate reflection of the actual construction conditions of the Work by affixing a Stamp indicating the Drawings are As-Builts and Certifying Accuracy on the final set of As-Builts.

- 13.14.5 Log of Control and Survey Documentation. Contractor shall complete and maintain an accurate log or all control and survey documentation for the Project as the Work progresses. All reference and control points shall be recorded on the As-Built drawings. The basis of elevations shall be one of the established benchmarks that must be maintained on the As-Builts.
- 13.14.6 Record Coordinates for Key Items. Contractor shall record, by coordinates, all utilities on-site with top of pipe elevations, major grade and alignment changes, rim, grate or top of curb and flow line elevations of all drainage structures and sewer manholes. Contractor shall update record information at or near the time when work is occurring in an area and prior to DSA 152 sign off on any category of Work and prior to covering the Work.

# 13.15 <u>Miscellaneous Obligations of Contractor</u>

- 13.15.1 District Permit and Other Obligations. It is expressly understood that the District shall pay the DSA for the DSA inspector, soils testing, DSA fees, special testing, etc. If additional review or permits become necessary for reasons not due to Contractor's fault or because of DSA requirements or regulations implemented after the date the GMP is established and not reasonably anticipated at the time the GMP is established, Contractor may seek compensation only for the direct cost (without mark up or added fees) of that review, as an additional cost. In the alternative, District may pay such costs directly to DSA. (Offsite costs and additional inspection costs)
- 13.15.2 Contractor Permit Obligations. Contractor shall pay for all remaining general building permits and ancillary permits and licenses not paid by District prior to the commencement of this Construction Services Agreement. Contractor shall also be responsible for arranging and overseeing all necessary inspections and tests, including inspections by the DSA, permits and occupancy permits, and ensure compliance with any Federal and State laws. All municipal charges for permanent utilities including, but not limited to, sewer, electrical, phone, gas, water, and irrigation shall be paid for by District. Contractor shall be responsible for arranging the payment of such fees by District at least one (1) week in advance of when the payment is due. Contractor may either request reimbursement from District for such fees (at direct cost only), or obtain the funds from District prior to paying such fees.
- 13.15.3 *Protection.* The Contractor shall establish procedures for the protection of all existing structures, equipment, utilities, and other existing improvements, both on site and off site.
- 13.15.4 *Nuisance Abatement.* The Contractor shall develop a mutually agreed upon documented program with the District to abate and minimize noise, dust, and disruption to normal activities at the existing facilities on the Site, including procedures to control on site noise, dust, and pollution during construction.
- 13.15.5 Site Mitigation and Remediation. Contractor shall be required to undertake Site mitigation or remediation at its sole cost for items identified in the Due Diligence Documents provided to Contractor. For hazardous substances and underground conditions that differ from representations in Contract Documents or Due Diligence Documents, Contractor shall provide notice within five (5) days after the discovery of



the occurrence of the unforeseen conditions. If Due Diligence Documents and information provided to Contractor does not provide notice of the unforeseen condition, then the costs for such work shall be added as an extra pursuant to Article 17. Costs shall be allocated to the Unforeseen Allowance. However, to the extent Unforeseen Allowance is exceeded, District may, in its sole and absolute discretion, allocate any costs that exceed the Unforeseen Allowance arising from unforeseen underground conditions and hazardous substances that are not documented in the Construction Documents or in the Due Diligence Documents reviewed to the District Contingency.

- 13.15.6 *Utilities*. The Contractor shall perform and pay for all temporary utility hook ups and connections; the District shall pay for use of utilities during construction, as well as any fees owed to utility suppliers for connection to existing mainline facilities. Buildings shall be connected to water, gas, sewer, and electric services, complete and ready for use. Service connections shall be made and existing services reconnected.
- 13.15.7 *Sanitary Facilities.* The Contractor shall provide a sanitary temporary toilet building as directed by the inspector for the use of all workers. The building shall be maintained in a sanitary condition at all times and shall be left at the site until the inspector directs removal. Use of toilet facilities in the work under construction shall not be permitted except by approval of the Inspector.
- 13.15.8 Layout and Field Engineering. All field engineering required for laying out this work and establishing grades for earthwork operations shall be furnished by the Contractor at its expense. Such work shall be done by a qualified civil engineer or land surveyor licensed in California and approved by the Architect. Any required "as built" drawings of site development shall be prepared by a qualified civil engineer or land surveyor licensed in California and approved by the Architect.
- 13.15.9 *Cutting and Patching.* Contractor shall do all cutting, fitting, or patching of work as required to make its several parts come together properly and fit it to receive or be received by work of other contractors showing upon, or reasonably implied by, the drawings and specifications for the completed structure. Contractor shall make good after them as Architect may direct. All cost caused by defective or ill-timed work shall be borne by party responsible therefore. Contractor shall not endanger any work by cutting, excavating, or otherwise altering work and shall not cut or alter work of any other contractor without consent or at the direction of Architect.
- 13.15.10 Documents on the Project Site. Contractor shall keep one copy of all Contract Documents, including addenda, change orders, Division I, Title 21 of the California Code of Regulations, Parts 1-5 and 12 of Title 24, and Title 22 of the California Code of Regulations, and the prevailing wage rates applicable to the Project, which are a part of Contract Documents, on job at all times. Said documents shall be kept in good order and shall be available to District representative, Architect and his representatives. Contractor shall be acquainted with and comply with the provisions of said Titles 21, 22 and 24 as they relate to this Project. (See particularly Duties of the Contractor, Title 24 California Code of Regulations, Section 4-343.) Contractor shall also be acquainted with and comply with all California Code of Regulations provisions relating to this Project, particularly Titles 17, 19, 21, 22 and 24.) Contractor shall also make available all books, records, accounts, contracts, bids, etc. upon request of District.
- 13.15.11 *Contractor to Bind Subcontractors to the Provisions of this Contract.* Contractor shall ensure that Subcontractors are bound to the same extent as Contractor is bound to District.
- 13.15.12 *Contractor Responsible for Means and Methods.* Contractor shall be solely responsible for the construction means, methods, techniques, sequences, procedures, and



coordinating all portions of the work under the Contract Documents, unless the Contract Documents give other specific instructions concerning these matters. Contractor shall be responsible to see that the finished work complies accurately with the Contract Documents. Contractor shall not perform the work without utilizing the Contract Documents or, where required, approved shop drawings, product data, or samples for any such portion of the work.

- 13.15.13 Contractor Responsible for Acts and Omissions of Employees. Contractor shall be responsible to District for acts and omissions of Contractor's employees, Subcontractors, material and equipment suppliers, and their agents, employees, invitees, and other persons performing portions of the work under direct or indirect contract with Contractor or any of its Subcontractors.
- 13.15.14 General DSA Compliance. During the entire term of this Agreement, Contractor shall coordinate its services with the District, Architect, Project Inspector, and other parties to ensure that all requirements set forth in the DSA's Inspection Card (Form 152) and any subsequent revisions or updates thereto issued or required by DSA, or any other/alternate processes are being met in compliance with DSA requirements. Contractor shall take all action necessary as to not delay progress in meeting any DSA requirements. Contractor shall meet any applicable requirements set forth in DSA's Construction Oversight Process Procedure (PR 13-01) and any subsequent revisions or updates thereto issued or required by DSA. Any references to DSA requirements for the Project shall be deemed to include and incorporate any revisions or updates thereto.

#### 13.16 Close Out

- 13.16.1 All DSA Close-Out requirements (See DSA Certification Guide). Contractor is also specifically directed to the DSA Certification Guide and the applicable certificates for the DSA-311 form.
- 13.16.2 Punch List Is Prepared Only After the Project Is Substantially Complete. The Inspector and Architect shall prepare a Punch List of items which is an inspection report of the Work, if any, required in order to complete the Contract Documents and ensure compliance with the DSA Approved Plans so the Project may be Completed by the Contractor and a final DSA Close-Out is approved. When all Work for the Project is Complete, including Punch Lists and all Work complies with the approved Contract Documents and Change Orders, the Project has reached Final Completion.
- 13.16.3 Time for Completion of Punch List. Contractor shall only be given a period of no more than thirty (30) days to complete the Punch List on Project. During the Punch List period Contractor Superintendent and Project Manager shall remain engaged in the Project and shall not be removed or replaced. If the Punch List is not completed at the end of the Punch List time then Contractor shall issue a valued Punch List within 5 days after the date the Punch List time ends. If Contractor does not issue such a list, the Owner or Architect may issue a valued Punch List to the Contractor and withhold up to 150% of the value of the Punch List Work.
- 13.16.4 As-Builts Up to Date and Complete. The intent of this procedure is to obtain an exact "As-Built" record of the Work upon completion of the Project. The following information shall be carefully and correctly drawn on the prints and all items shall be accurately located and dimensioned from finished surfaces of building walls on all As-Built drawings:
  - 13.16.4.1 The exact location and elevations of all covered utilities, including valves, cleanouts, etc. must be shown on As-Builts



- 13.16.4.2 Contractor is liable and responsible for inaccuracies in As-Built drawings, even though they become evident at some future date.
- 13.16.4.3 Upon completion of the Work and as a condition precedent to approval of release of the Retention Payment, Contractor shall obtain the Inspector's approval of the "As-Built" information. When completed, Contractor shall deliver corrected sepias and/or a Diskette with an electronic file in a format acceptable to the District.
- 13.16.4.4 District may withhold the cost to hire a draftsman and potholing and testing service to complete Record As-Built Drawings at substantial cost if the Contractor does not deliver a complete set of Record As-Built Drawings. This shall result in withholding of between \$10,000 to \$20,000 per building that does not have a corresponding Record As-Built Drawing.
- 13.16.5 Any Work not installed as originally indicated on drawings
- 13.16.6 All DSA Close-Out requirements (See DSA Certification Guide). Contractor is also specifically directed to the DSA Certification Guide and the applicable certificates for the DSA-311 form.
- 13.16.7 Submission of Form 6-C. Contractor shall be required to execute a Form 6-C as required under Title 24 Sections 4-343. The Contractor understands that the filing with DSA of a Form 6-C is a requirement to obtain final DSA Approval of the Construction by Contractor and utilized to verify under penalty of perjury that the Work performed by Contractor complies with the DSA approved Contract Documents.
- 13.16.8 Contractor shall be Responsible for All Costs to Certify the Project. The District may Certify the Project complies with Approved Construction Documents by utilizing the procedures under the Project Certification Guide (Located at the DSA website at http://www.documents.dgs.ca.gov/dsa/plan\_review\_process/project\_certification\_guid e\_updated\_03-15-13.pdf). All costs for professionals, inspection, and testing required for an alternate Project Certification shall be the Contractor's responsibility and the District reserves its right to institute legal action against the Contractor and Contractor's Surety for all costs to certify the Project and all costs to correct Non-Compliant Work that is discovered during the Alternate Certification Process.
- 13.16.9 ADA Work that must be corrected to receive DSA certification. See Article 41.
- 13.16.10 *Maintenance Manuals.* At least thirty (30) days prior to final inspection, three (3) copies of complete operations and maintenance manuals, repair parts lists, service instructions for all electrical and mechanical equipment, and equipment warranties shall be submitted. All installation, operating, and maintenance information and drawings shall be bound in 8½" x 11" binders. Provide a table of contents in front and all items shall be indexed with tabs. Each manual shall also contain a list of Subcontractors, with their addresses and the names of persons to contact in cases of emergency. Identifying labels shall provide names of manufactures, their addresses, ratings, and capacities of equipment and machinery.
- 13.16.11 Maintenance manuals shall also be delivered in electronic media for the Project. Any demonstration videos shall also be provided on electronic media.
- 13.17 <u>Correction of Work: Warranty.</u> Neither a Progress Payment, Sublease Payment nor any provision in the Contract Documents shall relieve Contractor of responsibility for faulty materials or workmanship incorporated in the Project. Contractor warrants that all work under this Construction



Services Agreement will be free of faulty materials or workmanship and hereby agrees, within ten (10) days upon receiving notification from District, to remedy, repair or replace, without cost to District, all defects which may appear as a result of faulty materials or workmanship in the Project, at any time, or from time to time, during a period beginning with commencement of the Project and ending one (1) year after the date of completion of the specific the warranty is being provided, as defined in Article 18 hereof. The foregoing warranty of Contractor also applies to the remedy, repair or replacement of defects which may in the documents prepared by Contractor and/or any party retained by, through or under Contractor in connection with the Project, but the foregoing warranty of Contractor does not guarantee against damage to the Project sustained by use, wear, intentional acts, accidents, or lack of normal maintenance or as a result of changes or additions to the Project made or done by parties not directly responsible to Contractor, except where such changes or additions to the Project are made in accordance with Contractor's directions. No guarantee furnished by a party other than Contractor with respect to equipment manufactured or supplied by such party shall relieve Contractor from the foregoing warranty obligation of Contractor. The warranty period set forth herein above shall not apply to latent defects appearing in the Project, and with respect to such defects, the applicable statute of limitations shall apply. Contractor agrees to provide the District with all equipment and materials warranties provided by manufacturers to District but has no obligation to assist in processing such warranty claims after said one (1) year warranty period.

- 13.17.1 Assignment of Subcontracts. Upon the Completion of the Warranty period, Contractor shall assign to the District all subcontracts with Subcontractors, material suppliers or other vendors that provided Work for the Project. This assignment shall include all purchase orders and any change orders or addenda that were executed with the assigned Subcontractor.
  - 13.17.1.1 <u>Documents to be Provided to District.</u> Contractor shall provide the following documents to the District as part of Close Out of the Project:
    - a. *Subcontractor Warranty*. Contractor shall provide any warranty documents, including warranties consistent with the requirements of this Contract and the Contract Documents.
    - b. *Contracts*. Contractor shall provide copies of all subcontracts, amendments, change orders and other documents associated with the Subcontractor's scope of work and price for work on the Project.
    - c. Subcontractors Bound to the Same Extent as Contractor. The Subcontractors shall be bound to the same extent as the Contractor is bound by this CSA and Subcontractors shall be required to include assignment of their contracts to the District.
    - d. *Bonds Assignable*. Contractor shall ensure that Subcontractor performance and payment bonds are assignable and can be assigned to the District.
    - e. *Unconditional Releases*. Contractor shall provide as part of the Close Out of the Project, Unconditional Releases for each Subcontractor and Material supplier that provided Work for the Project.
    - f. *Project Files*. Contractor shall provide the District a copy of the entire Subcontractor file, including any submittals or shop drawings that were provided by Subcontractor.



- g. District Reserves the Right to Assume Subcontractor Contracts
  Prior to the End of the Warranty Period. District reserves the
  right to take assignment of Subcontractor contracts prior to the
  end of the warranty period.
- 13.18 Assignment of Anti-Trust Claims. The Contractor offers and agrees to assign to the District all rights, title and interest in and to all causes of action it may have under Section 4 of the Clayton Act (15 USC Sec. 15) or under the Cartwright Act (Chapter 2 (commencing with section 16700) of Part 2 of Division 7 of the Business and Professions Code), arising from purchase of goods, services, or materials pursuant to the Construction Services Agreement. This assignment shall become effective at the time the District tenders the final Sublease Payment to Contractor, without further acknowledgment by the parties.

# 14. CONTRACT DOCUMENTS AND INTERPRETATIONS

- 14.1 The Contract Documents shall be executed, and/or initialed as appropriate, in duplicate by District and Contractor. The Contract Documents are complementary, and what is required by any one shall be as binding as if required by all. The intention of the Contract Documents is to include all labor, services and materials reasonably necessary for the proper execution of the work.
- 14.2 It is not intended that work and/or services not covered under any heading, section, branch, class or trade of the specifications shall be supplied, unless it is required elsewhere in the Contract Documents or is reasonably inferable therefrom as being necessary to produce the intended results, in which case such work and/or services shall be supplied by Contractor. Words which have well known technical or trade meanings are used herein in accordance with such recognized meanings. Mutual agreement shall be reached with respect to words which do not have a well-known technical or trade meaning and the definition of which come into question.
- 14.3 Plans and Specifications are intended to be fully cooperative and to agree. All Plan and Specification changes shall be dated and sequentially recorded. All modifications to Plans and Specifications shall be interpreted in conformity with the Contract Documents, which shall govern, unless otherwise specified.

#### 15. SUBMITTALS

#### 15.1 Definitions

- 15.1.1 Deferred Approvals. Approval of certain aspects of the construction may be deferred until the construction Contract has been awarded. To facilitate the design process, DSA grants deferred approval to the design and detailing of certain elements of the Project at the request of the Architect or Engineer of Record. Design elements that may be deferred may include, but are not limited to Access floors, Bleachers, Elevator guide rails and related elevator systems, Exterior wall systems precast concrete, glass fiber reinforced concrete, etc., Skylights, Window wall systems, storefronts, Stage rigging, and other systems as noted in the Contract Documents. (Also see Article 15.3 and 15.6).
- Shop Drawings. The term "Shop Drawings" as used herein means drawings, diagrams, equipment or product schedules, and other data, which are prepared by Contractor, Subcontractors, manufacturers, suppliers, or distributors illustrating some portion of the Work, and includes: illustrations; fabrication, erection, layout and setting drawings; manufacturer's standard drawings; schedules; descriptive literature, instructions, catalogs, and brochures; performance and test data including charts; wiring and control diagrams; and all other drawings and descriptive data pertaining to materials, equipment, piping, duct and conduit systems, and methods of construction as may be required to show that the materials, equipment, or systems and their position conform to the requirements of the Contract Documents.



- 15.1.3 Manufactured applies to standard units usually mass-produced, and "Fabricated" means items specifically assembled or made out of selected materials to meet individual design requirements. Shop drawings shall: establish the actual detail of all manufactured or fabricated items, indicate proper relation to adjoining work, amplify design details of mechanical and electrical systems and equipment in proper relation to physical spaces in the structure, and incorporate minor changes of design or construction to suit actual conditions.
- 15.1.4 Submittals is a term used interchangeably and sometimes refers to Shop Drawings, Product Data, and Samples since all Subcontractor submissions are tracked in a Submittal Log and may include any of the noted items. However, generally, a Submittal is a manufacturer's product information and product data including description, characteristics, size, physical characteristics, and requirements to prepare the jobsite for receiving of the particular manufactured item.
- 15.1.5 Samples. The term "samples" as used herein are physical examples furnished by Contractor to illustrate materials, equipment, or quality and includes natural materials, fabricated items, equipment, devices, appliances, or parts thereof as called for in the Specifications, and any other samples as may be required by the Architect to determine whether the kind, quality, construction, finish, color, and other characteristics of the materials, etc., proposed by the Contractor conform to the required characteristics of the various parts of the Work. All Work shall be in accordance with the approved samples.

## 15.2 Shop Drawings.

- When Shop Drawings Are Required. Shop drawings are required for prefabricated components and for installation and coordination of these prefabricated components into the Project. In addition, Shop Drawings, are prepared to address the actual size and installation of components from various Subcontractors and provides an opportunity for the Contractor to coordinate and address conflicts between the subcontracting trades. In some cases, each Subcontractor or trade will provide Shop Drawings in a format agreed upon by District.
- 15.2.2 Purpose for Shop Drawings. Shop drawings are the Contractor's manufacturer, Subcontractor, supplier, vendor or the Contractor's detailed drawings showing particularized method for assembly, specifics to a manufacturer, manufacturer component installation requirements, specifics as to a manufactured item, alterations to a manufactured, a custom created item, or drawn version of more detailed information expanding on the Architect's design shown in the Contact Documents. Drawings address the appearance, performance, size, weight, characteristics and prescriptive descriptions associated with the Contractor or Contractor's Subcontractor's plan for installation or assembly based on the design in the specifications and Contract Documents. The shop drawing often is more detailed than the information shown in the Contract Documents to give the Architect and Engineer the opportunity to review the fabricator's version of the product (along with particulars specific to that particular product), prior to fabrication. References to the Contract Documents, Construction Documents, Drawings, Plans, and Specifications assist the Architect and Engineer in their review of the Shop Drawings. Attachment of manufacturer's material specifications, "catalog cut sheets," and other manufacturer's information may be provided to accompany Shop Drawings. Because Shop Drawings facilitate the Architect's and Engineer's approval of the system, they should be as clear and complete as possible so they may be reviewed by Architect or Engineer for the Project.
- 15.2.3 Shop Drawing Requirements. The Contractor shall obtain and submit with Shop Drawings all seismic and other calculations and all product data from equipment



manufacturers. "Product data" as used herein are illustrations, standard schedules, performance charts, instructions, brochures, diagrams, and other information furnished by the Contractor to illustrate a material, product, or system for some portion of the Work.

- Not a Reproduction of Architectural or Engineering Drawings. The shop drawing are not a reproduction of the architectural or engineering drawings. Instead, they must show more detail than the Construction Documents and details the fabrication and/or installation of the items to the manufacturer's production crew or Contractor's installation crews.
- 15.2.5 Shop Drawings Engineering Requirements: Some shop drawings require an engineer stamp to be affixed on the drawings and calculations. In such cases, a current and valid engineering stamp shall be affixed by a California registered engineer. No out of State engineers shall stamp Shop Drawings. (See DSA IR A-18). In most cases, an engineer means California registered mechanical, structural, electrical or plumbing engineer. California Registered Civil Engineers will not be accepted for structural details unless specifically approved by DSA.
- DSA approvals Required Prior to Work. No work on a Shop Drawing that requires DSA approval may proceed until DSA approval is received. Contractor has provided DSA approval time and allowed adequate time for corrections in Contractor's Schedule as required pursuant to Article 9.
- 15.2.7 Shop Drawing Identification. All Shop Drawings must be properly identified with the name of the Project and dated, and accompanied by a letter of transmittal referring to the name of the Project and to the Specification section number for identification of each item clearly stating in narrative form, as well as "clouding" all qualifications, departures, or deviations from the Contract Documents. Shop drawings, for each section of the Work shall be numbered consecutively and the numbering system shall be retained throughout all revisions. All Subcontractor submissions shall be made through the Contractor. Each drawing shall have a clear space for the stamps of Architect and Contractor.
- 15.3 Deferred Approvals. Deferred approvals shall be submitted and processed to ensure all DSA and other governmental approvals are secured so as to not delay the Project. There may be additional requirements for deferred approvals in Division 1 of the Specifications. All deferred approvals shall be prepared by Contractor or Contractor's agent early enough so as to not delay the Project. Contractor is aware that Title 24 California Code of Regulations Section 4-317 has specific requirements for deferred approvals as to governing agencies and as to the Architect and Engineer for the Project. As a result, any delay associated with the time for approval by applicable agencies or by the Architect or Architect's consultants shall be Contractor's. Contractor is required to comply with inclusion of Deferred Approvals in the Schedule as required under Article 9
  - DSA Approvals Required Prior to Work. No work on a deferred approval item may proceed on the components until DSA approval is received. Contractor has provided DSA approval time and allowed adequate time for any DSA revisions in Contractor's Schedule as required pursuant to Article 9.

#### 15.4 Submittals and Samples

15.4.1 *Information Required With Submittals*: Manufacturer, trade name, model or type number and quantities: Information provided must be of sufficient detail to allow Architect and Engineer to compare the submitted item with the specified products and acceptable products listed, in the specification and addenda.



- 15.4.2 Description of Use and Performance Characteristics: Information should be furnished describing the normal use and expected performance of the product. The Architect and Contractor review this information to confirm that the product is appropriate for the intended use.
- 15.4.3 Size and Physical Characteristics: The size and physical characteristics, such as adjustment capabilities, which is reviewed by both the Contractor and Architect. The Contractor has the most available information for comparing adjoining materials and equipment. The Contractor also needs to know the size and weight of the equipment for lifting and handling considerations.
- 15.4.4 *Finish Characteristics:* The Architect reviews the available finishes and selects the appropriate finish, if the finish was not previously specified in the documents. The Contractor should confirm that finish requirements in the specification are being met by the product.
- 15.4.5 *Contractor Responsible for Jobsite Dimensions*: Some material is custom-fabricated to job conditions, requiring dimensions from the jobsite. These jobsite dimensions are provided by the Contractor as part of the Contractor's responsibilities for the Project and shall be provided prior to release of the product for manufacture. Contractor shall not rely on Architect or Engineers to provide jobsite dimensions.
- 15.4.6 Full Range of Samples Required (When Specific Items Not Specified). Except in cases where the exact color and type of item is specified since the District is utilizing items Standardized or pre-selected by District, the full range of color, graining, texture, or other characteristics are anticipated for review in finished products, a sufficient number of samples of the specified materials shall be furnished by the Contractor to indicate the full range of characteristics which will be present in the finished products. Products delivered or erected without Submittal and approval without providing a full range of samples shall be subject to rejection. Except for range samples, and unless otherwise called for in the various sections of the Specifications, samples shall be submitted in duplicate.
- 15.4.7 *Labeling of Samples*. All samples shall be marked, tagged, or otherwise properly identified with the name of the submitting party, the name of the Project, the purpose for which the samples are submitted and the date.
- 15.4.8 *Transmittal letter.* All samples shall be accompanied by a letter of transmittal containing similar information, together with the Specification section number.
- 15.4.9 *Labels and Instructions*. All samples of materials shall be supplied with the manufacturer's descriptive labels and application instructions. Each tag or sticker shall have clear space for the review stamps of Contractor and Architect.
- 15.4.10 Architect's Review. The Architect will review and, if appropriate, approve submissions and will return them to the Contractor with the Architect's stamp and signature applied thereto, indicating the timing for review and appropriate action in compliance with the Architect's (or District's) standard procedures. In the cases where a CM is hired by the District, CM may be the party that receives and performance logging and initial processing of the Samples. CM may, in some cases, reject samples that are not in conformance with Contract requirements.

#### 15.5 Submittal Submission Procedure

15.5.1 *Transmittal Letter and Other Requirements*. All Submittals must be properly identified with the name of the Project and dated, and each lot submitted must be accompanied by



a letter of transmittal referring to the name of the Project and to the Specification section number for identification of each item clearly stating in narrative form, as well as "clouding" on the submissions, all qualifications, departures, or deviations from the Contract Documents. Shop drawings, for each section of the Work shall be numbered consecutively and the numbering system shall be retained throughout all revisions. All Subcontractor submissions shall be made through the Contractor. Each drawing shall have a clear space for the stamps of Architect and Contractor. In the case where a CM is hired on the Project, the CM may be designated to receive the Submittals for the Project, log the Submittals, and in some cases reject Submittals that do not conform to Contract requirements.

- Copies Required. Each Submittal shall include one (1) legible, reproducible (if electronic is available, electronic copies shall also be provided) and five (5) legible prints of each drawing or schedule, table, cut sheet, etc., including fabrication, erection, layout and setting drawings, and such other drawings as required under the various sections of the Specifications, until final acceptance thereof is obtained. Subcontractor shall submit copies, in an amount as requested by the Contractor, of: (1) manufacturers' descriptive data for materials, equipment, and fixtures, including catalog sheets showing dimensions, performance, characteristics, and capacities; (2) wiring diagrams and controls; (3) schedules; (4) all seismic calculations and other calculations; and (5) other pertinent information as required by the District or Architect.
- 15.5.3 Corrections. The Contractor shall make all corrections required by Architect, District or CM and shall resubmit, as required by Architect or CM, corrected copies of Shop Drawings or new samples until approved. Contractor shall direct specific attention in writing or on resubmitted Shop Drawings to revisions other than the corrections required by the Architect on previous submissions. Professional services required for more than one (1) re-review of required Submittals of Shop Drawings, product data, or samples are subject to charge to the Contractor pursuant to Article 10.4.
- 15.5.4 Approval Prior to Commencement of Work. No portion of the Work requiring a shop drawing or sample submission or other Submittal shall be commenced until the submission has been reviewed by Contractor and Architect (and CM, if applicable) and approved by Architect (and CM where applicable) unless specifically directed in writing by the Architect. All such portions of the Work shall be in accordance with approved Shop Drawings and samples.
- 15.5.5 *District's Property*. All Submittals, Shop Drawings, computer disks, constructability reviews, schedules, annotated specifications, samples and other Submittals shall become the District's property upon receipt by the District or Architect.
- Schedule Requirements for Submittals. Contractor shall obtain and shall submit all required Submittals (i.e. Shop Drawings, Deferred Approvals, Samples, etc.), in accordance with Contractor's "Schedule for Submission of Shop Drawings and Samples" as required in the scheduling portion of the CSA at Article 9 and the Specifications (as long as the Specifications do not conflict with CSA. In the case of conflict, the conflicting provision shall be controlled by the CSA and the remaining specification sections shall be interpreted as if the CSA language is inserted) with such promptness as to cause no delay in its own Work or in that of any other contractor or Subcontractor but in no event later than thirty five (35) days after the Notice to Proceed is issued except in the specific cases noted as an exception as set forth below. No extensions of time will be granted to Contractor or any Subcontractor because of its failure to have Shop Drawings and samples submitted in accordance with this Article 15 and the Schedule. Each Subcontractor shall submit all Shop Drawings, samples, and manufacturer's descriptive data for the review of the District, the Contractor, and the Architect through the Contractor.



- 15.6.1 *Consideration of Schedule*. Contractor has considered lead times, DSA or other agency governmental review times, Architect or Engineer review times, manufacturing seasons, and specific long lead procurement concerns for all submittals for the Project.
  - 15.6.1.1 <u>All Submittals for the Project</u> except those specifically agreed upon by District and Architect, in writing, shall be specifically incorporated into the Submittal section of the Schedule so as to not delay the Work. The agreement to allow a later Submittal does not mean that Article 15.6 is waived. Contractor shall order materials and ensure prices are honored and secured for the Project.
    - a. Structural Steel may be included as a Submittal later than 35 days if Structural Steel is a significant portion of the Work, at least one or some of the Project is a structural steel structural system, or as specifically agreed upon by the Architect or District.
    - b. It is specifically agreed that submissions of structural steel Submittals shall not be piecemeal (unless some portion is requested separately by the Owner or Architect), shall provide complete designs, shall be stamped by the Structural Steel Subcontractor, Contractor, and Structural Steel Subcontractor's structural engineer at time of submission and as further addressed in this Article.
    - c. In no case shall the submission of Structural Steel Drawings delay the critical path for the schedule. If a Milestone is provided for submission of complete structural steel Shop Drawings then the date shall be no later than as set forth in the Milestone
  - Exceptions to Submittal Within Thirty-Five (35) Days by Written Agreement. A written request detailing the specific reasons for a submission later than 35 days due to complexity of design, or non-critical path status of the Submittal shall be submitted at the time the Baseline Schedule is submitted. The Baseline Schedule shall not include a delayed Submittal until written agreement is provided. In addition to the request for providing a Submittal after the thirty-five (35) day period, a copy of the Contract with the Subcontractor who shall be performing the Submittal, a written statement from the Subcontractor verifying that work has commenced on the Submittal and providing Subcontractor's own schedule of milestones and completion dates, and a corresponding Submittal designation in the Schedule as required under Article 9
    - Approval of a delayed Submittal shall not result in any increase in the Contract Price or result in an extension of time for the completion of the Project.
  - 15.6.1.3 Piecemeal Submissions of Submittals. Piecemeal Submittals mean providing portions of Shop Drawings or Submittals as they are being completed. The submission of piecemeal Submittals results in the appearance of a submission when there is inadequate information for the Architect or Engineer to adequately review a submission. Piecemeal differs from submission of complete buildings or phases of buildings or complete assemblies. The Architect may agree to allow submission of single buildings or areas as long as the Submittals are complete.



## 15.7 General Submittal Requirements

- 15.7.1 Contractor Submittal Representations. By submitting Shop Drawings, product data, samples, etc., the Contractor represents that it has determined and verified all materials, field measurements, catalog numbers, related field construction criteria, and other relevant data in connection with each such submission, and that it has checked, verified, and coordinated the information contained within such Submittals with the requirements of the Work and of the Contract Documents, including the construction schedule.
- 15.7.2 Contractor Coordination. By submitting Shop Drawings, Submittals, product data, samples, etc., the Contractor represents that it has determined and verified all materials, field measurements, catalog numbers, related field construction criteria, and other relevant data in connection with each such submission, and that it has checked, verified, and coordinated the information contained within such Submittals with the requirements of the Work and of the Contract Documents, including the construction schedule. Contractor shall stamp, sign, and date each Submittal indicating its representation that the Submittal meets all of the requirements of the Contract Documents and evidence Contractor's review through execution of the following stamp to be placed on each Shop Drawings:

"The [contractor] has reviewed and approved the field dimensions and the construction criteria, and has also made written notation regarding any information in the Shop Drawings and Submittals that does not conform to the Contract Documents. This Shop Drawing or Submittal has been coordinated with all other Shop Drawings and Submittals received to date by me as Contractor and this duty of coordination has not been delegated to Subcontractors, material suppliers, the Architect, or the Engineers on this Project.

Signature of Contractor and date

- 15.7.3 No Deviation from Contract Documents. The submission of the Shop Drawings, product data, samples, etc., shall not deviate from the requirements of the Contract Documents including detailing and design intent which is specifically outlined in Contract Documents except as specifically authorized by the Architect or through an accepted substitution pursuant to Article 16. All deviations from the Contract Documents shall be narratively described in a transmittal accompanying the Shop Drawings. However, Shop Drawings shall not be used as a means of requesting a substitution, the procedure for which is defined in Article 16, "Substitutions."
- 15.7.4 Contractor Responsibility for Shop Drawings Conformance to Contract Documents.

  Review by District and Architect shall not relieve the Contractor or any Subcontractor from its responsibility in preparing and submitting proper Shop Drawings in accordance with the Contract Documents.
- 15.7.5 *Incomplete Submittals*. Any submission, which in Architect's opinion is incomplete, contains errors, or has been checked superficially will be returned unreviewed by the Architect for resubmission by the Contractor.
- 15.7.6 Shop Drawings and Submittals Shall Not Be Used as a Method to Make a Substitution. Shop drawings and Submittals shall not be used as a means of requesting a substitution or to make changes in the Contract Documents. If changes are made to the Contract Documents through the Shop Drawings, the Architect shall have the right to reject the Submittal. If the Architect does not note the deviation from the approved Construction



Documents, the Contractor is still responsible for the change and the Architect or the District may require the Shop Drawings be revised to properly reflect the approved Contract Documents. The Architect or District may also require that the Contractor bear all costs under Article 10.4 and consequential damages associated with a CCD to revise Construction Documents to accommodate the deviation from approved Construction Documents.

15.7.7 Extent of Review. In reviewing Shop Drawings, the Architect will not verify dimensions and field conditions. The Architect will review and approve Shop Drawings, product data, samples, etc., for aesthetics and for conformance with the design concept of the Work and the information in the Contract Documents. The Architect's review shall neither be construed as a complete check which relieves the Contractor, Subcontractor, manufacturer, fabricator, or supplier from responsibility for any deficiency that may exist or from any departures or deviations from the requirements of the Contract Documents unless the Contractor has, in writing, called the Architect's attention to the deviations at the time of submission. The Architect's review shall not relieve the Contractor or Subcontractors from responsibility for errors of any sort in Shop Drawings or schedules, for proper fitting of the Work, coordination of the differing Subcontractor trades and Shop Drawings and Work which is not indicated on the Shop Drawings at the time of submission of Shop Drawings. Contractor and Subcontractors shall be solely responsible for any quantities which may be shown on the Submittals or Contract Documents.

# 16. REQUEST FOR SUBSTITUTIONS

- 16.1 For purposes of this provision the term "substitution" shall mean a change in product, material, equipment, or method of construction from those required by the Construction Documents proposed by the Contractor.
- Public Contract Code section 3400 does not apply to this agreement since the materials, services, and equipment used has been investigated as part of the Due Diligence investigation by Contractor and incorporated in the overall GMP.
- 16.3 Contractor may submit requests together with substantiating data for substitution of any "or equal" material, process or article. Any savings generated from the substitution shall be considered Project Savings under Article 7. The District shall not be responsible for any costs of Contractor associated with "or equal" substitution requests. The District has the complete and sole discretion to determine if a material, process or article is an "or equal" material, process or article that may be substituted. The data required to substantiate requests for substitutions of an "or equal" material, process or article data shall include a signed affidavit from the Contractor stating that the substituted "or equal" material, process or article is equivalent to that specified in the specification in every way except as listed on the affidavit. Substantiating data shall also include:
  - 1. Is equal in quality/service/ability to the Specified Item;
  - 2. Will entail no changes in detail, construction, and scheduling of related work;
  - 3. Will be acceptable in consideration of the required design and artistic effect;
  - 4. Will provide no cost disadvantage to the District;
  - 5. Will require no excessive or more expensive maintenance, including adequacy and availability of replacement parts; and
  - 6. Will required no change of the construction schedule



- 16.4 Failure to submit all the needed substantiating data, including the signed affidavit, to the Architect in a timely fashion so that the substitution can be adequately reviewed may result in the rejection of the proposed substitution. The District is not obligated to review multiple substitution submittals for the same product or item due to the Contractor's failure to submit a complete package initially.
- 16.5 Contractor shall bear the costs of all architectural and engineering work, DSA CCD review fees, and other costs associated with the review of submittals for substitution. See Article 10.4.
- 16.6 Contractor agrees to include the provisions of this Article in all Subcontractor contracts.

# 17. <u>EXTRA WORK/MODIFICATIONS (INCLUSION OF CCD COSTS, DSA COSTS, AND AN ICD PROCESS)</u>

No Changes Without Authorization. There shall be no change whatsoever in the drawings, specifications, or in the Work without an executed Change Order, Change Order Request, Immediate Change Directive, or order by the Architect for a minor change in the Work as herein provided. District shall not be liable for the cost of any extra work or any substitutions, changes, additions, omissions, or deviations from the Drawings and Specifications unless authorized District representative has approved the cost in writing by Change Order or executed Construction Change Document. No extension of time for performance of the Work shall be allowed hereunder unless claim for such extension is made at the time changes in the Work are ordered, and such time duly adjusted in writing in the Change Order. The provisions of the Contract Documents shall apply to all such changes, additions, and omissions with the same effect as if originally embodied in the Drawings and Specifications. Notwithstanding anything to the contrary in this Article 17, all Change Orders shall be prepared and issued by the Architect and shall become effective when executed by the authorized District representative (utilizing either a Construction Contingency Amount or a District Contingency Amount), the Architect, and the Contractor.

CONTRACTOR UNDERSTANDS, ACKNOWLEDGES, AND AGREES THAT THE REASON FOR THIS NOTICE REQUIREMENT IS SO THAT DISTRICT MAY HAVE AN OPPORTUNITY TO ANALYZE THE WORK AND DECIDE WHETHER THE DISTRICT SHALL PROCEED WITH THE CHANGE ORDER OR ALTER THE PROJECT SO THAT SUCH CHANGE IN WORK BECOMES UNNECESSARY AND TO AVOID THE POSSIBLE DELAYS ASSOCIATED WITH THE ISSUANCE OF A NOTICE OF NON-COMPLIANCE.

- Notices of Non-Compliance. Contractor deviation or changes from approved Construction Documents may result in the issuance of a Notice of Non-Compliance (See DSA Form 154). Contractor is specifically notified that deviations from the Construction Documents, whether major or minor, may result in the requirement to obtain a DSA Construction Change Document to correct the Notice of Non-Compliance. (See Article 17.4.1.1 for Definition of CCD). In some cases, the lack of a DSA approved CCD AND verification from the Inspector that a Notice of Non-Compliance has been corrected may result in a critical path delay to the next stage of Work on the Project. Specifically, a deviation from approved Construction Documents may prevent approval of the category of Work listed in the DSA 152 Project Inspection Card. Any delays that are caused by the Contractor's deviation from approved Construction Documents shall be the Contractor's responsibility.
- 17.3 <u>Architect Authority.</u> The Architect will have authority to order minor changes in the Work that do not involve DSA Approval not involving any adjustment in the Contract Sum, or an extension of the Contract Time.
- 17.4 CONSTRUCTION CHANGE DOCUMENT (CCD Category A, and CCD Category B) and IMMEDIATE CHANGE DIRECTIVE (ICD)



## 17.4.1 Definitions

17.4.1.1

Construction Change Document (CCD). A Construction Change Document is a DSA term that is utilized to address changes to the DSA approved Construction Documents. There are two types of Construction Change Documents. (1) DSA approved CCD Category A (DSA Form 140) for Work affecting Structural, Access or Fire-Life Safety of the Project which will require a DSA approval; and, (2) CCD Category B (DSA Form 140) for work NOT affecting Structural Safety, Access Compliance or Fire and Life Safety that will not require a DSA approval (except to confirm that no Approval is required);

17.4.1.2 <u>Immediate Change Directive (ICD)</u>. An Immediate Change Directive is a written order to the Contractor prepared by the Architect and signed by the District (and CM if there is a CM on the Project) and the Architect, directing a change in the Work and stating a proposed basis for adjustment, if any, in the Contract Sum or Contract Time, or both. The District may by ICD, without invalidating the Contract, direct immediate changes in the Work within the general scope of the Contract consisting of additions, deletions, or other revisions within. If applicable, the Contract Sum and Contract Time will be adjusted accordingly.

In the case of an Immediate Change Directive being issued, Contractor must commence Work immediately or delays from failure to perform the ICD shall be the responsibility of Contractor and the failure to move forward with Work immediately shall also be grounds for Termination under Article 19 or determination of partial default under Article 12.2.

An ICD does not automatically trigger an Article 20 Dispute or Claim. Contractor must timely follow the procedures outlined at Article 20 and this Article where applicable.

Refer to Forms for a copy of the proposed Immediate Change Directive form.

17.4.1.3 <u>Use to Direct Change.</u> An ICD shall be used to move work forward immediately and to avoid delay. In some cases, an ICD shall be issued in the absence of agreement on the terms of an Extra, or RFP. A copy of an ICD form is provided in the Forms included with this CSA. The anticipated not to exceed price for the Work will be inserted into the ICD. In the case of an ICD issued to correct Contractor Deficiencies or to correct a Contractor caused Notice of Non-Compliance, the ICD may be issued with \$0 and 0 time. Contract may prepare an Extra associated with the ICD pursuant to Article 17. However, Contractor shall proceed with all Work required under an Approved ICD immediately upon issuance. Failure to proceed with the Work under an ICD shall be grounds for Termination for Cause under Article 19 or take over the Work under Article 12.2.

If adequate time exists, an ICD may be subject of an RFP for pricing and determination if any time that may be required. However, if an RFP is not completed, Contractor shall immediately commence Work when an ICD is issued. If the RFP is incomplete, it may still be completed to be submitted for Pricing Purposes as long as the PR is submitted within the timeline provided by the PR, or within 10 days following issuance of the ICD.



- 17.4.1.4 ICD Issued Over a Notice of Non-Compliance or to Cover Work Subject to a DSA 152 Sign Off. In some cases, an ICD shall be for the purpose of proceeding with Work to keep the Project on Schedule and as an acknowledgement by the District that Contractor is proceeding with Work contrary to a Notice of Non-Compliance, prior to issuance of a DSA approved CCD Category A, or to direct the covering of Work which has not yet received a DSA 152 Inspection Approval to move forward.
  - a. Contractor Compliance with all Aspects of an ICD. Contractor is to undertake the ICD and comply with all aspects of the Work outlined in the ICD. Inspector is to inspect the Work pursuant to the ICD. Failure to follow the ICD may result in deduction of the ICD Work under Article 12.2 or Termination of the Contractor pursuant to Article 19.
  - b. Exception in the Case of DSA Issued Stop Work Order.

    Contractor must proceed with an ICD even if a CCD has not been approved by DSA except in the case of a DSA issued Stop Work Order. If a DSA Stop Work Order is issued, Contractor must stop work and wait further direction from the District.
  - c. ICD Due to Contractor Deficiency or Contractor Caused Notice of Non-Compliance. If an ICD is issued to correct a Contractor Deficiency or a Contractor caused notice of Non-Compliance, Contractor specifically acknowledges responsibility for all consequential damages associated with the Contractor Deficiency or Contractor Caused Notice of Non-Compliance and all consequential damages and costs incurred to correct the deficiency under Article 10.4.
- 17.5 Extras Request. Extra work or a modification or reduction of requirements or of methods of performing the Construction which differ from the work or requirements set forth in the Construction Documents ("Extra Work/Modifications"); and for such purposes, the District may at any time during the life of this Construction Services Agreement by written order, make such changes as it shall find necessary from Construction Contingency if District approves such request in writing. The costs of the Extra Work/Modifications, as established pursuant to this Article, shall be deducted from the Construction Contingency as mutually agreed in writing or the Errors and Omissions Contingency or the Unforeseen Allowance as determined by the District, and shall not affect the GMP.
  - 17.5.1 *Format*. The following format shall be used, as applicable by the District and the Contractor to communicate proposed additions and deductions to the Contract. A copy of a proposed Construction Change Document form is provided in Division 1 of the Specifications. The most stringent guidelines will apply to all forms.

		EXTRA	CREDIT
(a)	Material (attach itemized quantity and unit cost plus sales tax)		
(b)	Equipment (attach invoices)		



(c)	Labor Not to Exceed Applicable Prevailing Wage Rates (attach itemized hours and rates)	<u>EXTRA</u>	CREDIT
(d)	Subtotal (a-d)	<del></del>	
(e)	If Subcontractor performed work, add Subcontractor's overhead and profit to portions performed by Subcontractor, not to exceed 10% of item (d).		
(f)	Subtotal		
(g)	Contractor's Overhead and Profit: Not to exceed 10% of Item (d) if Contractor performed the work. No more than 5% of Item (d) if Subcontractor performed the work. If work was performed by Contractor and Subcontractors, portions performed by Contractor shall not exceed 10% if Item (d), and portions performed by Subcontractor shall not exceed 10% of Item (d)		
(h)	Subtotal		
(i)	Bond not to exceed one percent (1%) of Item (h)		
(j)	TOTAL		
(k)	Time/ Days	<del></del>	

The undersigned Contractor approves the foregoing Extra Work as to the changes, if any, and the contract price specified for each item and as to the extension of time allowed, if any, for completion of the entire work on account of said Extra Work, and agrees to furnish all labor, materials and service and perform all work necessary to complete any additional work specified therein, for the consideration stated herein. It is understood that said Extra Work shall be effective upon approval from the District's Designee if such amounts are against the GMP and if Owner Contingency is used when approved by the Governing Board of the District.

It is expressly understood that the value of such extra Work or changes, as determined by any of the aforementioned methods, expressly includes any and all of the Contractor's costs and expenses, both direct and indirect, resulting from additional time required on the Project or resulting from delay to the Project. Any costs, expenses, damages or time extensions not included are deemed waived.

The Contractor expressly acknowledges and agrees that any change in the Work performed shall not be deemed to constitute a delay or other basis for claiming additional compensation based on theories including, but not limited to, acceleration, suspension or disruption to the Project.

17.5.2 Should Contractor claim that any instruction, request, drawing, specification, action, condition, omission, default, or other situation (i) obligates the District to pay additional compensation to the Contractor; or (ii) obligates the District to grant an extension of time for the completion of the Construction Services Agreement; or (iii) constitutes a waiver of any provision in this Construction Services Agreement, CONTRACTOR SHALL NOTIFY THE DISTRICT, IN WRITING, OF SUCH CLAIM AS SOON AS POSSIBLE, BUT IN NO EVENT WITHIN MORE THAN TEN (10) BUSINESS DAYS FROM THE DATE CONTRACTOR HAS ACTUAL OR CONSTRUCTIVE



NOTICE OF THE CLAIM. CONTRACTOR SHALL ALSO PROVIDE DISTRICT WITH SUFFICIENT WRITTEN DOCUMENTATION SUPPORTING THE FACTUAL BASIS OF THE CLAIM under Article 20. Contractor shall be required to certify under penalty of perjury the validity and accuracy of any claims submitted. The Contractor's failure to notify the District within the ten (10) business day period shall be deemed a waiver and relinquishment of the claim against the District. If such notice be given within the specified time, the procedure for its consideration shall be as stated above in this Section.

17.5.3 All costs associated with the Extra Work/Modification may be in terms of time, money or both.

## 17.6 Deductive Change Orders

- 17.6.1 All Deductive Change Order(s) must be prepared utilizing the form under Paragraph 17.5 (a)-(d) only setting forth the actual costs incurred. Except in the case of an Article 12.2 or 29.4 Deductive Change Order where no mark-up shall be allowed, Contractor will be allowed a maximum of 5% total profit and overhead.
- 17.6.2 For Unilateral Deductive Change Orders, or where credits are due from Contractor for Allowances, Deductive Items, Inspection, Damage, DSA CCD review costs, Architect or Inspector costs for after hours or corrective services, Work removed from the Agreement under Article 12.2 or Article 29.4, there shall be no mark-up.
- 17.6.3 District may, at any time, after a Deductive Change Order is presented to Contractor by District for items under Article 12.2 or Article 29.4 of if there is disagreement as to the Deductive Change Order, issue a unilateral Deductive Change Order on the Project and deduct the Deductive Change Order from a Progress Payment or the Retention Payment.

## 18. <u>TIME OF COMPLETION</u>

- ONCE THE DISTRICT HAS ISSUED A NOTICE TO PROCEED, CONTRACTOR SHALL 18.1 PROCEED WITH THE CONSTRUCTION OF THE PROJECT WITH REASONABLE DILIGENCE. CONTRACTOR AGREES THAT THE PROJECT WILL BE SUBSTANTIALLY COMPLETE WITHIN THE CALENDAR DAYS DESIGNATED IN ARTICLE 3 FROM THE NOTICE TO PROCEED. SAID CONTRACT TIME MAY BE EXTENDED FOR SUCH PERIODS OF TIME AS ALLOWED UNDER THE CONTRACT DOCUMENTS. IF THE PROJECT IS NOT SUBSTANTIALLY COMPLETED IN ACCORDANCE WITH THE FOREGOING, IT IS UNDERSTOOD THAT THE DISTRICT WILL SUFFER DAMAGE SINCE CONTRACTOR HAS OVERSTAYED ITS LEASE TERM. IT BEING IMPRACTICAL AND INFEASIBLE TO DETERMINE THE AMOUNT OF ACTUAL DAMAGE, IT IS AGREED THAT CONTRACTOR'S EXTENSION OF THE LEASE SHALL RESULT IN LIQUIDATED DAMAGES, AND NOT AS A PENALTY, THE SUM SET FORTH IN ARTICLE 3 FOR EACH CALENDAR DAY OF DELAY UNTIL WORK IS SUBSTANTIALLY COMPLETED. CONTRACTOR AND ITS SURETY SHALL BE LIABLE FOR THE AMOUNT THEREOF. ANY MONEY DUE OR TO BECOME DUE THE CONTRACTOR MAY BE RETAINED BY THE DISTRICT TO COVER SAID LIQUIDATED DAMAGES FOR OVERSTAYING THE LEASE. SHOULD SUCH MONEY NOT BE SUFFICIENT TO COVER SAID LIQUIDATED DAMAGES, THE DISTRICT SHALL HAVE THE RIGHT TO RECOVER THE BALANCE FROM THE CONTRACTOR OR ITS SURETIES, WHO WILL PAY SAID BALANCE FORTHWITH.
- 18.2 Within five (5) business days after the Project commencement date in the District's Notice to Proceed, Contractor shall furnish District with a Baseline CPM (Critical Path) Schedule pursuant to Article 9. The Contractor shall include the District's occupancy requirements showing portions of the Projects having occupancy priority.



- 18.3 Contractor shall not be charged for liquidated damages, as set forth in the Agreement, for materially differing underground soil conditions than those outlined in the soils report and from hazardous substances that are encountered that are not documented in the Contract Documents or in the Due Diligence Documents provided to Contractor.
  - In case of encountering such unforeseen conditions noted above, Contractor shall notify the District in writing immediately and no later than seven (7) days following encountering the unforeseen condition. After providing written notice, Contractor shall test and provide District with Test results (unless District choses to test) and shall proceed with Work based on the Test results. A Change Order pursuant to Article 17 shall be submitted. All time and expenses shall be verified with the Inspector or District Designee either on the day the extra work occurs, but no later than 10 am the following business day.
  - 18.3.2 Change Orders associated with approved unforeseen conditions shall be billed as Change Order Work and allocated to the Unforeseen Allowance, and if the Unforeseen Allowance is exceeded, the District, in its sole and absolute discretion, may allocate such costs to the District Contingency to the extent unforeseen conditions as defined in this Article are encountered.
- 18.4 Contractor shall within ten (10) calendar days of beginning of any such delay notify District in writing of causes of delay. Thereupon District shall ascertain the facts and extent of delay and grant extension of time for completing work when, in its judgment, the findings of fact justify such an extension. Extension of time shall apply only to that portion of work affected by the delay, and shall not apply to other portions of work not so affected. Contractor agrees that the extension of time granted under this Article shall be its sole and exclusive remedy for the consequences of any delay described above. For any such delay resulting from the actions or inactions of Architect, District, or their officers, agents, and employees, or changes to the scope of the Work which impact the schedule, Contractor shall be entitled to reimbursement for its reasonable additional costs resulting from such delay, but not any additional profit or fee.
- 18.5 Contractor acknowledges the extreme importance of promptly notifying and thoroughly documenting any request for time extension and further specifically acknowledges that District will suffer extreme prejudice should Contractor fail in any way to comply with this requirement. Failure to comply with the procedures and time limits established in this Article shall constitute a waiver of such request. Evidence presented by Contractor that District had actual notice of the time extension request, that District was not prejudiced by Contractor's failure to comply with this requirement, and/or that District considered Contractor's request despite Contractor's failure to strictly comply with this provision shall not render this requirement unenforceable.
- 18.6 Contractor is required to order, obtain, and store materials and equipment sufficiently in advance of its work at no additional cost or advance payment from District to assure that there will be no delays. An extension of time will not be granted for a delay caused by a shortage of materials.
- 18.7 Contractor shall not be entitled to additional compensation for delays within its control. Contractor is aware that governmental agencies, such as the Department of General Services, gas companies, electrical utility companies, water districts and other agencies may have to approve Contractor-prepared drawings or approve a proposed installation. In the event of delays to the Project from such agencies for which Contractor has no control, provided such delays are not caused by Contractor's or any Subcontractor's acts or omissions, Contractor may be entitled to a time extension for such delays, but shall not be allowed additional compensation for the costs of such delays not impacting the Project's critical path.
- 18.8 District reserves the right to occupy any building or portion thereof or use any improvement contemplated by the Contract Documents prior to the completion of the entire Project. A list of work to be completed and corrected by Contractor, if any, shall be prepared and agreed to between



District and Contractor before any such occupancy or use. Such occupancy or use shall not operate as an acceptance of any part of the Project but shall start the guaranty-warranty period on the structure or portion thereof so occupied or improvement or equipment so used; provided, however, that such occupancy or use shall not start the guaranty-warranty period as to items appearing on the list of work yet to be completed and corrected or as to structures or improvements (or portions thereof) that are not occupied or used. No such occupancy or use shall be deemed to have occurred unless and until District has given Contractor written notice of its intention to so occupy or use any particular structure or improvement specifying the portion or portions of the structure, improvement or equipment which will be deemed so occupied or used. District and Contractor shall take reasonable steps to obtain the consent of Contractor's insurance company or companies and shall, without mutual written consent, take no action with respect to partial occupancy or use that would cause cancellation, lapse of or reduction of such insurance. Such occupancy or use by District shall relieve Contractor of (and District shall assume) the responsibility for injury or damage to said occupied or used portions of the Project resulting from use by District or the public or from the action of the elements or from any other cause, except injury or damage resulting from the operations, negligence or intentional acts of Contractor, any Subcontractors or materialmen of any tier, or their officers, employees or agents.

#### 19. TERMINATION OF AGREEMENT

#### 19.1 Termination for Breach.

- 19.1.1 If the Contractor refuses or fails to proceed with the construction of the Project or any separable part thereof with such diligence as will insure its completion within the time specified by this Construction Services Agreement or any extension thereof, or fails to Complete the Project within the Contract Time, or if the Contractor should be adjudged bankrupt, or if it should make a general assignment for the benefit of its creditors, or if a receiver should be appointed on account of its insolvency, or the Contractor or any of its Subcontractors should violate any of the provisions of this Construction Services Agreement, the District may serve written notice upon the Contractor and its Surety of the District's intention to terminate this Construction Services Agreement. This notice of intent to terminate shall contain the reasons for such intention to terminate this Construction Services Agreement and a statement to that effect that the Contractor's right to perform work on the Project shall cease and terminate upon the expiration of ten (10) days unless such violations have ceased and arrangements satisfactory to the District have been made for correction of said violations.
- In the event that the District serves such written notice of termination upon the Contractor and the Surety, the Surety shall have the right to take over and perform this Construction Services Agreement. If the Surety does not: (1) give the District written notice of Surety's intention to take over and commence performance of this Construction Services Agreement within fifteen (15) days of the District's service of said notice of intent to terminate upon Surety; and (2) actually commence performance of this Construction Services Agreement within thirty (30) days of the District's service of said notice upon Surety; then the District may take over the Project and prosecute the same to completion by separate contract(s) or by any other method it may deem advisable for the account and at the expense of the Contractor.
- 19.1.3 In the event that the District elects to obtain an alternative performance of the Construction Services Agreement as specified above: (1) the District may, without liability for so doing, take possession of and utilize in completion of the Project such materials, appliances, plants and other property belonging to the Contractor that are on the site and reasonably necessary for such completion; and (2) Surety shall be liable to the District for any cost or other damage to the District necessitated by the District securing an alternate performance pursuant to this Article.



## 19.2 <u>Termination for Convenience.</u>

- 19.2.1 The District may terminate performance of the Project called for by the Contract Documents in whole or, from time to time, in part, if the District determines that a termination is in the District's interest.
- 19.2.2 The District shall terminate all or any part of the Project upon delivery to the Contractor of a "Notice of Termination" specifying that the termination is for the convenience of the District, the extent of termination, and the effective date of such termination.
- 19.2.3 After receipt of Notice of Termination, and except as directed by the District's Representative, the Contractor shall, regardless of any delay in determining or adjusting any amounts due under this Termination for Convenience clause, immediately proceed with the following obligations:
  - 1. Stop Work as specified in the Notice of Termination.
- 2. Complete any work specified in the Notice of Termination in a least cost/shortest time manner while still maintaining the quality called for under the Contract Documents.
- 3. Leave the Property upon which the Contractor was working and upon which the facility (or facilities) forming the basis of the Contract Documents is situated in a safe and sanitary manner such that it does not pose any threat to the public health or safety.
- 4. Terminate all subcontracts to the extent that they relate to the portions of the work terminated.
- 5. Place no further subcontracts or orders, except as necessary to complete the continued portion of the Construction Services Agreement.
- 6. Submit to the District's Representative, within ten (10) days from the Project termination date found in the Notice of Termination, all of the usual documentation called for by the Contract Documents to substantiate all costs incurred by the Contractor for labor, materials and equipment through the Project termination date, including termination costs related to demobilizing and closing out the Project, found in the Notice of Termination. Any documentation substantiating costs incurred by the Contractor solely as a result of the District's exercise of its right to terminate this Construction Services Agreement pursuant to this clause, which costs the Contractor is authorized under the Construction Services Agreement to incur, shall: (i) be submitted to and received by the District no later than thirty (30) days after the Project termination date found in the Notice of Termination; (ii) describe the costs incurred with particularity; and (iii) be conspicuously identified as "Termination Costs occasioned by the District's Termination for Convenience."
  - 19.2.4 Termination of the Construction Services Agreement shall not relieve the Surety of its obligation for any just claims arising out of or relating to the work performed on the Project.
  - 19.2.5 In the event that the District exercises its right to terminate this Construction Services Agreement pursuant to this clause, the District shall pay the Contractor, upon the Contractor's submission of the documentation required by this provision, and other applicable provisions of the Construction Services Agreement the following amounts not already paid to Contractor:
    - 1. All actual costs incurred according to the provisions of this Construction Services Agreement including but not limited to insurance costs incurred in connection with the Project.
    - 2. A reasonable allowance for profit on the cost of the work on the Project performed and not otherwise paid for the District, provided Contractor



establishes to the satisfaction of the District, that it is reasonably probable that the Contractor would have made a profit had the Construction Services Agreement been completed and provided further, that the profit allowed shall in no event exceed five percent (5%) of costs. In no event shall the total amount exceed GMP.

- A reasonable allowance for Contractor's administrative costs in determining the amount payable due to termination of the Construction Services Agreement under this Article.
- 19.3 <u>Termination of Agreement by Contractor.</u> The Contractor may terminate the Construction Services Agreement upon ten (10) days written notice to the District, whenever: (1) there is a substantial failure of performance on the part of the District; or (2) the District shall elect not to appropriate funds and/or not to make two (2) successive Sublease Prepayments (if exercised by the District in its sole discretion) following the receipt by District of a request from the Contractor in its capacity as Lessor for each such Sublease Prepayment pursuant to Article 21 of the Sublease. In the event of such termination, the Contractor shall have no claims against the District except for payment for the value of the work performed on the Project as of the date of termination.
- 19.4 <u>Assignment of Subcontractors and Suppliers.</u> If the Contract is Terminated, Contractor shall provide District copies of all subcontracts, purchase orders, addenda, invoices, payment records, and Project files associated with each Subcontractor and Material Supplier. The District shall have the option to assume any Subcontracts, contracts or purchase orders the District choses. To the extent that vendors are not paid in full for the labor, materials, or services provided, Contractor shall provide an accounting statement showing the amounts paid and the amounts due to the Subcontractor and a statement on the anticipated payment status associated with the Termination.
- 19.5 <u>Continuation of Work During Disputes.</u> In the event of a dispute between the parties as to performance of the work or the interpretation of this contract, or payment dispute, the parties shall attempt to resolve the dispute. Pending resolution of this dispute, Contractor agrees to continue the work diligently to completion and shall neither rescind nor terminate the agreement.

#### 20. RESOLUTION OF AGREEMENT CLAIMS

20.1 Decision of Architect. "Disputes" or "Claims" as defined in Article 20.9.1.1 between District and Contractor involving money or time, including those alleging an error or omission by the Architect shall be referred initially to the Architect for action as provided in Article 20.2 within ten (10) days after Contractor's Article 17 request for extra work/ modification is denied. If there is a CM, the CM shall receive the Dispute and may review and also assemble opinions and documents to assist the Architect. A decision by the Architect, as provided in Article 20.5, shall be required as a condition precedent to proceeding with remedies set forth in Article 20.9 as to all such matters arising prior to the date Retention Payment Application is due, regardless of whether such matters relate to execution and progress of the Work, or the extent to which the Work has reached Final Completion.

The condition precedent of an Architect decision shall be waived if: (1) the position of Architect is vacant; (2) the Architect has failed to take action required under Article 20.5 within the time periods required therein; or (3) the Dispute or Claim relates to a stop notice claim not arising from any extra Change Order or Immediate Change Directive for which approval has not been provided.

Architect's Review. The Architect (and CM) will review the Dispute and take one or more of the following preliminary actions upon receipt of a Dispute: (1) request additional supporting data from the claimant; (2) submit a schedule to the parties indicating when the Architect expects to take action; (3) reject the Dispute in whole or in part, stating reasons for rejection; (4) recommend approval of the Dispute; or (5) suggest a compromise. The Architect may also, but is not obligated to, notify the Surety, if any, of the nature and amount of the Dispute..



- 20.2.1 Architectural Immunity. Architect review of Disputes and Claims shall be impartial and meant to resolve Disputes and Claims. Pursuant to the case, <u>Huber, Hunt & Nichols, Inc. v. Moore</u> (1977) 67 Cal.App.3d 278, the Architect is provided a quasi-judicial immunity for interpreting and deciding Disputes and Claims between the District and Contractor.
- 20.3 <u>Documentation if Resolved.</u> If a Dispute has been resolved, the Architect (and/or CM) will prepare a Change Order or obtain appropriate documentation to document the terms for Board approval.
- Actions if Not Resolved. If a Dispute has not been resolved and all documentation requested pursuant to Article 20.2 has been provided, the Contractor shall, within ten (10) days after the Architect's initial response, assemble all the documents involved in the Dispute including copies of all back-up documentation of costs and the basis for the Dispute and take one or more of the following actions: (1) modify the initial Dispute; (2) notify the Architect that the initial Dispute stands; or (3) supplement with additional supporting data and re-submit to the Architect under Article 20.2.
- Architect's Written Decision. If a Dispute has not been resolved after consideration of the foregoing and of other evidence presented by the parties or requested by the Architect, the Architect (or Architect through CM) shall provide a written decision twenty (20) days after compliance with Article 20.4. Upon expiration of such time period, the Architect (or Architect through CM) will render to the parties its written decision relative to the Dispute, including any change in the Contract Sum or Contract Time or both. The Architect may also request reasonable additional time to complete Architect's written decision.

If the resolution of the Dispute by the Architect is not satisfactory to the Contractor and copies of all back-up documentation of costs and the basis for the Dispute is fully articulated in a package of material that is complete, the Contractor may then submit a Claim to the District under Article 20.9.

- 20.6 <u>Continuing Contract Performance.</u> Pending final resolution of a Dispute or Claim, including, negotiation, mediation, arbitration, or litigation, the Contractor shall proceed diligently with performance of the Contract, and the District shall continue to make any undisputed payments in accordance with the Contract (less any withholdings or offsets). If the Claim is not resolved, Contractor agrees it will neither rescind the Contract nor stop the progress of the work, but Contractor's sole remedy shall be to submit such controversy to determination by a court of competent jurisdiction in the county where the Project is located, after the Project has been completed, and not before.
  - 20.6.1 District's Option to Submit Individual Disputes to Arbitration during Claims and Disputes Process. At the District's sole option, in order to more efficiently resolve Claims during the Project and prior to the completion of the Claims Process, pursuant to Government Code section 9201, the District may submit individual Disputes or Claims for binding arbitration and Contractor agrees to the resolution of for each individual Dispute or Claim by an Arbitrator, including resolution of time and delays. If binding arbitration is utilized for individual Disputes or Claims, such resolution is full and final as to that particular Dispute or Claim. THIS INDIVIDUAL DISPUTE ARBITRATION PROCESS IS NOT AN ARBITRATION CLAUSE AND SHALL NOT BE CONSTRUED AS AN AGREEMENT TO ARBITRATE. INDIVIDUAL DISPUTES ARBITRATION PROCESS IS FOR THE SOLE PURPOSE OF STREAMLINING AND RESOLVING DISPUTES OR CLAIMS DURING CONSTRUCTION AND SHALL BE REQUESTED ON SPECIFIC INDIVIDUAL ITEMS BY THE DISTRICT PRIOR TO RETENTION PAYMENT (EVEN IF THERE ARE DEDUCTIONS MADE FROM RETENTION PAYMENT) WHICH REPRESENTS THE FINAL COMPLETION OF THE PROJECT.



- 20.6.1.1 If there is no Retention remaining on the Project, individual Disputes initiated prior to Project Final Completion shall continue until a final disposition of the Arbitration or resolution of the individual Claim or Dispute.
- 20.6.1.2 The Arbitration process shall not toll the Disputes or Claims process under Article 20 or the requirement to submit Claims to Court under Article 20.13.
- 20.7 <u>Claims for Concealed Trenches or Excavations Greater Than Four Feet Below the Surface.</u> When any excavation or trenching extends greater than four feet below the surface:
  - 20.7.1 *Immediately upon discovery*, The Contractor shall promptly, and before the following conditions are disturbed, notify the District, by telephone and in writing of any:
    - 20.7.1.1 Subsurface or latent physical conditions at the Site differing from those indicated in the Drawings, Specifications, or pursuant to the documents and information from Contractor's Due Diligence or Due Diligence Documents.
    - 20.7.1.2 Unknown physical conditions at the Site of any unusual nature, different materially from those ordinarily encountered and generally recognized as inherent in Work of the character provided for in the Contract.
    - 20.7.1.3 Hazardous waste condition, except, if Contractor's bid includes removal or disposal of hazardous substances, or is part of Contractor's Due Diligence or Due Diligence Documents. Material that the Contractor believes may be a material that is hazardous waste, as defined in Section 25117 of the Health and Safety Code, is required to be removed to a Class I, Class II, or Class III disposal site in accordance with the provisions of existing law. In such case, the notice procedures and requirements of Article 17.5.2 shall apply.
  - 20.7.2 The District shall investigate the conditions, and if District finds that the conditions do materially so differ, and cause a decrease or increase in the Contractor's cost of, or the time required for, performance of any part of the Work shall issue a change order or Construction Change Document under the procedures described in the Contract.
  - 20.7.3 *In the event that a dispute* arises between a public entity or District and the Contractor whether the conditions materially differ, or cause a decrease or increase in the Contractor's cost of, or time required for, performance of any part of the Work, the Contractor shall not be excused from any scheduled Completion Date provided for by the Contract, but shall proceed with all Work to be performed under the Contract. The Contractor shall retain any and all rights provided either by Contract or by law which pertain to the resolution of disputes and protests between the contracting parties.
- 20.8 <u>Dispute Concerning Extension of Time.</u> If Contractor and District cannot agree upon an extension of time, whether compensable or not, then Contractor must have first completed the procedures set forth in Article 18. Upon completion of the procedures set forth under Article 18, Contractor must then comply with the requirements in this Article including those set forth under Article 20.9.
- 20.9 <u>Claims Procedures.</u> Pursuant to the remedies under Public Contract Code section 9201 and Government Code section 930.2, Contractor, through execution of this Agreement, also agrees to comply with the Claims requirements under Article 20 to quickly and efficiently resolve disputes. Further, to provide a level of accuracy to the records submitted, the District shall have the right to



audit books and records pursuant to Article 21 based on the actual costs incurred and to reduce the uncertainty in resolving disputes with limited information.

#### 20.9.1 Procedure Applicable to all Claims

- 20.9.1.1 Definition of Claim: A "Claim" is where a Dispute between the parties rises to the level where backup documentation is assembled and provided to the District as a separate demand by the Contractor for: (1) a time extension, including, without limitation, for relief from damages or penalties for delay assessed by the District under the Contract; (2) payment by the District of money or damages arising from Work done by, or on behalf of, the Contractor pursuant to the Contract and payment for which is not otherwise expressly provided for or to which the Contractor is not otherwise entitled to; or (3) an amount of payment disputed by the District. If the Claim is for damages associated with a DSA Stop Work Order, the Contractor shall not be entitled to a request for Compensation, but shall be entitled to utilize Governmental Delay Float (See Article 9.2.1.)
- 20.9.1.2 Filing Claim Is Not Basis to Discontinue Work: The Contractor shall promptly comply with Work under the Contract or Work requested by the District even though a written Claim has been filed. The Contractor and the District shall make good faith efforts to resolve any and all Claims that may arise during the performance of the Work covered by this Contract.
- 20.9.1.3 Claim Notification: The Contractor shall within seven (7) calendar days after the written decision of the Architect, or if the time period for Architect's decision has passed under Article 20.5, submit a notification in writing sent by registered mail or certified mail with return receipt requested, with the District (and the District's CM) stating clearly the basis for the Claim and including all relevant and required documents. If the notification is not submitted within seven (7) days after the written decision of the Architect or the passage of time under Article 20.5, the Contractor shall be deemed to have waived all right to assert the Claim, and the Claim shall be denied. Claims submitted after the Retention Payment date shall also be considered null and void by the District. All Claims shall be reviewed pursuant to Articles 20.1 through 20.5.
- 20.9.1.4 The Formal Notification of Claim must be presented as follows:
  - a. The term "Claim" must be at the top of the page in no smaller than 20 point writing.
  - b. All documentation submitted pursuant to Article 20 to the Architect shall be submitted with the title "claim."
  - c. A stack of documents, copy of all Project documents, or the submission of random documents shall not constitute an adequate reference to supporting documentation
  - d. Any additional or supporting documentation that Contractor believes is relevant should be submitted at this time.
- 20.9.1.5 Reasonable Documents to Support Claim: The Contractor shall furnish reasonable documentation to support the Claim. The Contractor shall provide all written detailed documentation which supports the Claim, including but not limited to: arguments, justifications, cost, estimates, Schedule analysis and detailed documentation. The format of the



required reasonable documentation to support the Claim shall include, without limitation:

- a. Cover letter.
- b. Summary of factual basis of Claim and amount of Claim.
- c. Summary of the basis of the Claim, including the specific clause and section under the Contract under which the Claim is made.
- d. Documents relating to the Claim, including:
  - 1. Specifications sections in question.
  - 2. Relevant portions of the Drawings
  - 3. Applicable Clarifications (RFI's)
  - Other relevant information, including responses that were received.
  - 5. Contractor Analysis of Claim merit.
    - (a) Contractor's analysis of any Subcontractor vendor claims that are being passed through.
    - (b) Any analysis performed by outside consultants.
    - (c) Any legal analysis that Contractor deems relevant.
- e. Breakdown of all costs associated with the Claim.
- f. For Claims relating to time extensions, an analysis and supporting documentation evidencing any effect upon the critical path in conformance with the requirements of Article 9 and a chronology of events and related correspondence.
- g. Chronology of events and related correspondence.
- h. Applicable daily reports and logs.
  - 1. If the daily reports or logs are not available, lost or destroyed, there shall be a presumption that the lost documentation was unfavorable to the Contractor. See California Civil Jury Instruction 204.
- i. For Claims involving overhead, cost escalation, acceleration, disruption or increased costs, a full version of job costs reports organized by category of work or Schedule of Values with budget information tracked against actual costs. Any and all supporting back-up data, including the original bid or cost documents (and associated original unaltered metadata).
  - 1. The metadata and bid or cost information shall be provided confidentially and subject to a protective



- order to prevent dissemination to other contractors or to the public. However, the bid or cost documentation should remain intact and available for review and inspection in case of this type of increased cost Claim.
- 2. This data on the bid or cost information shall be made available to any District attorneys or experts and shall also be utilized as evidence for any legal proceedings.
- 3. If the bid or cost documentation is not available, lost or destroyed, there shall be a presumption that the lost bid or cost documentation was unfavorable to the Contractor. See California Civil Jury Instruction 204.
- j. Certification: The Contractor (and Subcontractors, if applicable) shall submit with the claim a certification under penalty of perjury:
  - 1. That the Contractor has reviewed the Claim and that such Claim is made in good faith;
  - 2. Supporting data are accurate and complete to the best of the Contractor's knowledge and belief;
  - The amount requested accurately reflects the amount of compensation for which the Contractor believes the District is liable.
  - 4. That the Contractor is familiar with Government Code section 12650 et seq. and Penal Code section 72 and that false claims can lead to substantial fines and/or imprisonment.
- k. Signature of Certification: If the Contractor is not an individual, the certification shall be executed by an officer or general partner of the Contractor having overall responsibility for the conduct of the Contractor's affairs.
- 1. Upon receipt of a Claim and all supporting documents as required above, the District shall conduct a reasonable review of the Claim and, within a period not to exceed 45 days, shall provide the Contractor a written statement identifying what portion of the Claim is disputed and what portion is undisputed. Upon receipt of a Claim, the District and Contractor may, by mutual agreement, extend the time period provided in this paragraph.
- m. If the District needs approval from its governing Board to provide the Contractor a written statement identifying the disputed portion and the undisputed portion of the Claim, and the governing Board does not meet within the 45 days or within the mutually agreed to extension of time following receipt of a Claim sent by registered mail or certified mail, return receipt requested, the District shall have up to three days following the next duly publicly noticed meeting of the governing Board after the 45-day period, or extension, expires to provide the



Contractor a written statement identifying the disputed portion and the undisputed portion.

- n. Any payment due on an undisputed portion of the Claim shall be processed and made within 60 days after the District issues its written statement. If the District fails to issue a written statement, paragraph t below shall apply.
- o. If the Contractor disputes the District's written response, or if the District fails to respond to a Claim issued pursuant to Article 20.9 within the time prescribed, the Contractor may demand in writing an informal conference to meet and confer for settlement of the issues in dispute. Upon receipt of a demand in writing sent by registered mail or certified mail, return receipt requested, the District shall schedule a meet and confer conference within 30 days for settlement of the Claim.
- Within 10 business days following the conclusion of the meet p. and confer conference, if the Claim or any portion of the Claim remains in dispute, the District shall provide the Contractor a written statement identifying the portion of the Claim that remains in dispute and the portion that is undisputed. Any payment due on an undisputed portion of the Claim shall be processed and made within 60 days after the District issues its written statement. Any disputed portion of the Claim, as identified by the Contractor in writing, shall be submitted to nonbinding mediation, with the District and the Contractor sharing the associated costs equally. The District and Contractor shall mutually agree to a mediator within 10 business days after the disputed portion of the Claim has been identified in writing. If the parties cannot agree upon a mediator, each party shall select a mediator and those mediators shall select a qualified neutral third party to mediate with regard to the disputed portion of the Claim. Each party shall bear the fees and costs charged by its respective mediator in connection with the selection of the neutral mediator. If mediation is unsuccessful, the parts of the Claim remaining in dispute shall be subject to applicable procedures in Article 20.13.
- q. For purposes of Article 20.9, mediation includes any nonbinding process, including, but not limited to, neutral evaluation or a dispute review board, in which an independent third party or board assists the parties in dispute resolution through negotiation or by issuance of an evaluation. Any mediation utilized shall conform to the timeframes in this section.
- r. Unless otherwise agreed to by the District and the Contractor in writing, the mediation conducted pursuant to Article 20.9 shall excuse any further obligation under Public Contract Code section 20104.4 to mediate after litigation has been commenced.
- s. This Claims process does not preclude the District from requiring arbitration of disputes under private arbitration or the Public Works Contract Arbitration Program, if mediation under



Article 20.9 does not resolve the parties' Claim. This Claims process does not preclude the District from submitting individual Disputes or Claims to binding arbitration pursuant to Article 20.12 below.

- t. Failure by the District to respond to a Claim from the Contractor within the time periods described in this subdivision or to otherwise meet the time requirements of Article 20.9 shall result in the Claim being deemed rejected in its entirety. A Claim that is denied by reason of the District's failure to have responded to a Claim, or its failure to otherwise meet the time requirements of Article 20.9, shall not constitute an adverse finding with regard to the merits of the Claim or the responsibility or qualifications of the Contractor.
- If a subcontractor or a lower tier subcontractor lacks legal u. standing to assert a Claim against a District because privity of contract does not exist, the Contractor may present to the District a Claim on behalf of a subcontractor or lower tier subcontractor. A subcontractor may request in writing, either on his or her own behalf or on behalf of a lower tier subcontractor, that the Contractor present a Claim for work which was performed by the subcontractor or by a lower tier subcontractor on behalf of the subcontractor. The subcontractor requesting that the Claim be presented to the District shall furnish reasonable documentation to support the Claim. Within 45 days of receipt of this written request, the Contractor shall notify the subcontractor in writing as to whether the Contractor presented the Claim to the District and, if the Contractor did not present the Claim, provide the subcontractor with a statement of the reasons for not having done so.
- v. Upon receipt of a Claim, the parties may mutually agree to waive, in writing, mediation and proceed directly to the commencement of a civil action or binding arbitration, as applicable.
- w. The Contractor's Claim shall be denied if it fails to follow the requirements of this Article.
- 20.10 <u>District (through CM or District's Agent or Attorney) May Request Additional Information</u>. Within thirty (30) days of receipt of the Claim and the information under this Article, the District may request in writing any additional documentation supporting the Claim or documentation relating to defenses to the Claim which the District may assert. If additional documents are required, the time in which the Claim is evaluated may be extended by a reasonable time so the Claim and additional documents may be reviewed.
- 20.11 <u>Claims Procedures in Addition to Government Code Claim</u>. Nothing in the Claims procedures set forth in Article 20 of the CSA shall act to waive or relieve the Contractor from meeting the requirements set forth in Government Code section 900 et seq.
- 20.12 <u>Binding Arbitration of Individual Claim Issues</u>. To expedite resolution of Claims pursuant to Public Contract Code section 9201, at the District's sole option, the District may submit individual Claims to Arbitration prior to Retention Payment consistent with the requirements of Article 20.6.1.



- 20.13 <u>Resolution of Claims in Court of Competent Jurisdiction</u>. If Claims are not resolved under the procedure set forth and pursuant to Article 20.9, such Claim or controversy shall be submitted to a court in the County of the location of the Project after the Project has been completed, and not before
- Warranties, Guarantees and Obligations. The duties and obligations imposed by this CSA and the rights and remedies available hereunder to the parties hereto, and, in particular but without limitation, the warranties, guarantees and obligations imposed upon Contractor by the Contract Documents and amendments thereto; and all of the rights and remedies available to District and Architect thereunder, are in addition to, and are not to be construed in any way as a limitation of any rights and remedies available to any or all of them which are otherwise imposed or available by laws or regulations by special warranty or guarantee or by other provisions of the Contract Documents, and the provisions of this Article will be as effective as if repeated specifically in the Contract Documents in connection with each particular duty, obligation, right and remedy to which they apply.

#### 21. MAINTENANCE OF RECORDS; AUDIT/OWNERSHIP OF DOCUMENTS

- 21.1 State Audit. Pursuant to and in accordance with the provisions of Government Code § 10532, or any amendments thereto, all books, records, and files of the District, the Contractor, or any Subcontractor connected with the performance of this Contract involving the expenditure of state funds in excess of Ten Thousand Dollars (\$10,000.00), including, but not limited to, the administration thereof, shall be subject to the examination and audit of the Office of the Auditor General of the State of California for a period of five (5) years after Retention Payment is made or a Notice of Completion is Recorded, whichever occurs first. Contractor shall preserve and cause to be preserved such books, records, hard drives, electronic media, and files for the audit period.
- District Audit. Pursuant to the remedies under Public Contract Code section 9201 and Government Code section 930.2, Contractor, through execution of this Agreement, also agrees the District shall have the right to review and audit, upon reasonable notice, the books and records of the Contractor concerning any monies associated with the Project. The purpose of this Audit is to quickly and efficiently resolve disputes based on the actual costs incurred and to reduce the uncertainty in resolving disputes with limited information. The District shall perform any audits at its own cost and any such audit shall be performed by an independent auditor, having no direct or indirect relationship with the functions or activities being audited or with the business conducted by the Contractor or District. In the event the independent auditor determines that Change Orders, Response to Request for Proposals, Claims, Appeal of Claims, or other requests for payment the Auditor shall report the results of the Audit findings to the District and provide a copy to the Contractor after giving the District Board the opportunity for at least 10 days review. If the Contractor disputes the findings of the independent auditor, such dispute shall be handled in the manner set forth under Article 20 entitled Disputes.
- Failure to Produce Books or Records. If Contractor having agreed to the terms of this Contract fails to produce books or records requested by Auditor, such failure to produce books or records that were required to be preserved for audit, it shall be presumed that the information contained in the withheld books or records were unfavorable to the Contractor and the Auditor shall note this refusal in the results of the Audit findings for further evaluation by the District and the District's Board. The refusal to release records that are concerning monies associated with the Project may be used as a grounds to Debar the Contractor from future Projects for failure to preserve records under this Article and the failure to produce required audit records may also be used as a grounds for a negative finding against the Contractor depending on the significance of the records that are withheld by Contractor. Failure to produce Job Cost Data tied to Job cost categories and budgets shall be presumed an intentional failure to produce key audit records. Similarly, failure to produce daily time records (prepared at or near the time of the Work actually took place shall be presumed an intentional failure to produce key audited records.



- 21.4 Inefficiency, Acceleration or Delay Claims. If Contractor is seeking costs for inefficiency, home office overhead, or unanticipated increased costs due to delays or acceleration, Contractor shall also produce copies of the original bid or cost tabulation utilized in submitting Contractor's cost for the Project. This document shall be considered confidential and shall not be subject to disclosure through a Public Records Act and shall not be distributed to anyone other than the District and the District's counsel. This bid or cost tabulation shall only be used in litigation, arbitration, evaluation of Claims or Disputes, Audit, and trial. If the records for the bid or cost tabulation are kept on a computer, the Contractor shall also produce all metadata (in native format) that accompanies the bid or cost tabulation for inspection to prove the authenticity of the underlying bid or cost tabulation. Failure to produce the bid or cost tabulation for review of inefficiency, home office overhead, or unanticipated increased costs due to delays or accelerations shall be considered material evidence that the bid or cost tabulation was not favorable to the Contractor. This evidence shall be entered as a jury instruction for trial that the bid or cost tabulation was not produced and the bid or cost tabulation information was unfavorable to the Contractor. The evidence may also be used in Debarment Proceedings, and noted as an exception to an Audit Findings.
- 21.5 Upon notification of Contractor concerning the results of the audit and a reasonable time has passed for Contractor to respond to Audit Findings and if either there is no Dispute of the Audit findings under this Article or if the result after utilizing the Disputes Clause confirms the Audit findings, the District may seek any Savings that have not been accounted for with District and may also seek reimbursement for overstated Claims, Change Orders, or Appeal of Claims.
- 21.6 Ownership of Drawings. Notwithstanding any provision of this Agreement, all drawings, specifications, and copies thereof furnished by District are its property. They are not to be used on other work and with exception of signed contract sets, are to be returned to District on request at completion of work.

# 22. PREVAILING RATES OF WAGES; RECORDS, APPRENTICES

Wage Rates. Pursuant to the provisions of Article 2 (commencing at § 1720), Chapter 1, Part 7, Division 2, of the Labor Code, the District has obtained the general prevailing rate of per diem wages and the general prevailing rate for holiday and overtime work in the locality in which this public works project is to be performed for each craft, classification, or type of worker needed for this Project from the Director of the Department of Industrial Relations ("Director"). These rates are on file at the administrative office of the District and are also available from the Director of the Department of Industrial Relations. Copies will be made available to any interested party on request. The Contractor shall post a copy of such wage rates at appropriate, conspicuous, weatherproof points at the Site.

Any worker employed to perform Work on the Project, but such Work is not covered by any classification listed in the published general prevailing wage rate determinations or per diem wages determined by the Director of the Department of Industrial Relations, shall be paid not less than the minimum rate of wages specified therein for the classification which most nearly corresponds to the employment of such person in such classification.

- 22.2 <u>Holiday and Overtime Pay</u>. Holiday and overtime work, when permitted by law, shall be paid for at the rate set forth in the prevailing wage rate determinations issued by the Director of the Department of Industrial Relations or at least one and one-half (1½) times the specified basic rate of per diem wages, plus employer payments, unless otherwise specified in the Contract Documents or authorized by law
- 22.3 <u>Wage Rates Not Affected by Subcontracts.</u> The Contractor shall pay and shall cause to be paid each worker engaged in the execution of the Work on the Project not less than the general prevailing rate of per diem wages determined by the Director, regardless of any contractual relationship which may be alleged to exist between the Contractor or any Subcontractor and such workers.



- 22.4 <u>Per Diem Wages.</u> The Contractor shall pay and shall cause to be paid to each worker needed to execute the Work on the Project per diem wages including, but not limited to, employer payments for health and welfare, pensions, vacation, travel time and subsistence pay as provided for in Labor Code §1773.1.
- 22.5 Forfeiture and Payments. Pursuant to Labor Code §1775, the Contractor shall forfeit to the District, not more than Two Hundred Dollars (\$200.00) for each calendar day, or portion thereof, for each worker paid less than the prevailing wages rates as determined by the Director of the Department of Industrial Relations, for the work or craft in which the worker is employed for any Work done under the Agreement by the Contractor or by any Subcontractor under it. The amount of the penalty shall be determined by the Labor Commissioner and shall be based on consideration of: (1) whether the Contractor or Subcontractor's failure to pay the correct rate of per diem wages was a good faith mistake and, if so, the error was promptly and voluntarily correct upon being brought to the attention of the Contractor or Subcontractor; and (2) whether the Contractor or Subcontractor has a prior record of failing to meet its prevailing wage obligations.

#### 23. RECORDS OF WAGES PAID

## 23.1 Payroll Records

- Pursuant to §1776 of the Labor Code, each Contractor and Subcontractor shall keep an accurate payroll record showing the name, address, social security number, work classification and straight time and overtime hours worked each day and week, and the actual per diem wages paid to each journeyman, apprentice, worker or other employee employed by him or her in connection with the Project.
- 23.1.2 All payroll records shall be certified and submitted to the District with each application for payment, but not less than once per month or as otherwise requested by the District. All payroll records shall be available for inspection at all reasonable hours at the principal office of the Contractor on the following basis:
- 23.1.3 A certified copy of an employee's payroll record shall be made available for inspection or furnished to the employee or his or her authorized representative on request.
- 23.1.4 A certified copy of all payroll records shall be made available for inspection or furnished upon request to a representative of District, the Division of Labor Standards Enforcement or the Division of Apprenticeship Standards of the Department of Industrial Relations.
- A certified copy of all payroll records shall be made available upon request by the public for inspection or for copies thereof. However, a request by the public shall be made through the District, the Division of Apprenticeship Standards or the Division of Labor Standards Enforcement. If the requested payroll records have not been provided pursuant to Paragraph (2) above, the requesting party shall, prior to being provided the records, reimburse the costs, according to law for the preparation by the Contractor, Subcontractor(s), and the entity through which the request was made. The public shall not be given access to such records at the principal office of the Contractor.
- 23.1.6 Unless required to be furnished directly to the Labor Commissioner in accordance with Labor Code section 1771.4, the certified payroll records shall be on forms provided by the Division of Labor Standards Enforcement or shall contain the same information as the forms provided by the Division of Labor Standards Enforcement.
- 23.1.7 The Contractor or Subcontractor(s) shall file a certified copy of all payroll records with the entity that requested such records within 10 calendar days after receipt of a written request.



- 23.1.8 Any copy of records made available for inspection as copies and furnished upon request to the public or any public agency by the District, the Division of Apprenticeship Standards or the Division of Labor Standards Enforcement shall be marked or obliterated to prevent disclosure of an individual's name, address and social security The name and address of the Contractor awarded the Contract or the Subcontractor(s) performing the Contract shall not be marked or obliterated. Any copy of records made available for inspection by, or furnished to, a joint labor-management committee established pursuant to the federal Labor Management Cooperation Act of 1978 (Section 175a of Title 29 of the United States Code) shall be marked or obliterated only to prevent disclosure of an individual's name and social security number. Notwithstanding any other provision of law, agencies that are included in the Joint Enforcement Strike Force on the Underground Economy established pursuant to Section 329 of the Unemployment Insurance Code and other law enforcement agencies investigating violations of law shall, upon request, be provided non-redacted copies of certified payroll records.
- 23.1.9 The Contractor shall inform the District of the location of all payroll records, including the street address, city and county, and shall, within five working days, provide a notice of a change of location and address.
- 23.1.10 The Contractor or Subcontractor(s) shall have ten (10) calendar days in which to comply subsequent to receipt of a written notice requesting payroll records. In the event that the Contractor or Subcontractor(s) fails to comply within the 10-day period, the Contractor or Subcontractor(s) shall, as a penalty to the District, forfeit One Hundred Dollars (\$100.00) for each calendar day, or portion thereof, for each worker, until strict compliance is effectuated. Upon the request of the Division of Apprenticeship Standards or the Division of Labor Standards Enforcement, these penalties shall be withheld from Progress Payments or Retention Payment then due.
- 23.1.11 Responsibility for compliance with this Article shall rest upon the Contractor.

# 23.2 <u>Withholding of Payments & Penalties</u>

- 23.2.1 The District may withhold or delay Progress Payments to the Contractor or a Sublease Payment or Retention if:
  - 23.2.1.1 The required prevailing rate of per diem wages determined by the Director of the Department of Industrial Relations is not paid to all workers employed on the Project; or
  - 23.2.1.2 The Contractor or Subcontractor(s) fail to submit all required certified payroll records with each application for payment, but not less than once per month; or
  - 23.2.1.3 The Contractor or Subcontractor(s) submit incomplete or inadequate payroll records; or
  - 23.2.1.4 The Contractor or Subcontractor(s) fail to comply with the Labor Code requirements concerning apprentices; or
  - 23.2.1.5 The Contractor or Subcontractor(s) fail to comply with any applicable state laws governing workers on public works projects.



## 24. APPRENTICES

- Apprentice Wages and Definitions. All apprentices employed by the Contractor to perform services under the Contract shall be paid the standard wage paid to apprentices under the regulations of the craft or trade for which he or she is employed, and as determined by the Director of the Department of Industrial Relations, and shall be employed only at the craft or trade to which he or she is registered. Only apprentices, as defined in §3077 of the Labor Code, who are in training under apprenticeship standards that have been approved by the Chief of the Division of Apprenticeship Standards and who are parties to written apprenticeship agreements under Chapter 4 (commencing with §3070) of Division 3, are eligible to be employed under this Contract. The employment and training of each apprentice shall be in accordance with the apprenticeship standards and apprentice agreements under which he or she is training, or in accordance with the rules and regulations of the California Apprenticeship Council.
- 24.2 Employment of Apprentices. Contractor agrees to comply with the requirements of Labor Code §1777.5. The Contractor awarded the Project, or any Subcontractor under him or her, when performing any of the Work under the Contract or subcontract, employs workers in any apprenticeable craft or trade, the Contractor and Subcontractor shall employ apprentices in the ratio set forth in Labor Code §1777.5. The Contractor or any Subcontractor must apply to any apprenticeship program in the craft or trade that can provide apprentices to the Project site for a certificate approving the contractor or subcontractor under the apprenticeship standards for the employment and training of apprentices in the area or industry affected. However, the decision of the apprenticeship program to approve or deny a certificate shall be subject to review by the Administrator of Apprenticeship. The apprenticeship program or programs, upon approving the Contractor or Subcontractor, shall arrange for the dispatch of apprentices to the Contractor or Subcontractor upon the Contractor's or Subcontractor's request. "Apprenticeable craft or trade" as used in this Article means a craft or trade determined as an apprenticeable occupation in accordance with the rules and regulations prescribed by the California Apprenticeship Council. The ratio of work performed by apprentices to journeyman employed in a particular craft or trade on the Project shall be in accordance with Labor Code §1777.5.
- Submission of Contract Information. Prior to commencing Work on the Project, the Contractor and Subcontractors shall submit contract award information to the applicable apprenticeship program(s) that can supply apprentices to the Project and make the request for the dispatch of apprentices in accordance with the Labor Code. The information submitted shall include an estimate of journeyman hours to be performed under the Contact, the number of apprentices proposed to be employed, and the approximate dates the apprentices would be employed. A copy of this information shall also be submitted to the District if requested. Within sixty (60) days after concluding Work on the Project, the Contractor and Subcontractors shall submit to the District, if requested, and to the apprenticeship program a verified statement of the journeyman and apprentice hours performed on the Project.
- Apprentice Fund. The Contractor or any Subcontractor under him or her, who, in performing any of the Work under the Contract, employs journeymen or apprentices in any apprenticeable craft or trade shall contribute to the California Apprenticeship Council the same amount that the Director determines is the prevailing amount of apprenticeship training contributions in the area of the Project. The Contractor and Subcontractors may take as a credit for payments to the California Apprenticeship Council any amounts paid by the Contractor or Subcontractor to an approved apprenticeship program that can supply apprentices to the Project. The Contractor and Subcontractors may add the amount of the contributions in computing its bid or costs for the Contract.
- 24.5 <u>Prime Contractor Compliance.</u> The responsibility of compliance with this Article 13 §1777.5 of the Labor Code for all apprenticeable occupations is with the Prime Contractor. Any Contractor or Subcontractor that knowingly violates the provisions of this Article or Labor Code §1777.5 shall be subject to the penalties set forth in Labor Code §1777.7.



24.6 WHEN DETERMINING GMP, CONTRACTOR SHALL INCLUDE TO THE EXTENT POSSIBLE ANTICIPATED GENERAL PREVAILING WAGE RATES FOR THE TIME WHEN WORK ON THE PROJECT WILL ACTUALLY BE PERFORMED.

#### 25. REGISTRATION WITH DEPARTMENT OF INDUSTRIAL RELATIONS

- Strict compliance with all DIR registration requirements in accordance with Labor Code sections 1725.5 and 1771.1 is a material obligation of the Contractor and all of its subcontractors (of any tier) under the Contract Documents. The foregoing includes, without limitation, compliance with DIR registration requirements at all times during performance of the Work by the Contractor and all of its subcontractors of any tier. The failure of the Contractor and all subcontractors of any tier to be properly registered with DIR at all times during performance of the Work is a material breach of the Contract Documents and subject to termination for cause.
- An affirmative and ongoing obligation of the Contractor under the Contract Documents is the verification that all subcontractors of any tier are at all times during performance of the Work are in full and strict compliance with the DIR registration requirements. The Contractor shall not permit or allow any subcontractor of any tier to perform any Work without the Contractor's verification that all subcontractors are in full and strict compliance with the DIR registration requirements. Any subcontractors of any tier not properly registered with DIR shall be substituted in accordance with Labor Code section 1771.1. Contractor or its subcontractors of any tier shall not be entitled to any additional costs or time arising from or in any way related to compliance with the DIR registration requirements.
- 25.3 The Contractor and all subcontractors shall furnish certified payroll records as required pursuant Labor Code section 1776 directly to the Labor Commissioner in accordance with Labor Code section 1771.4 on at least on a monthly basis (or more frequently if required by the District or the Labor Commissioner) and in a format prescribed by the Labor Commissioner. The District reserves the right to withhold Progress Payments or Retention Payment if the District is notified, or determines as the result of its own investigation, that Contractor is in violation of any of the requirements set forth in Labor Code section 1720 et seq. at no penalty or cost to the District. Monitoring and enforcement of the prevailing wage laws and related requirements will be performed by the Labor Commissioner/ Department of Labor Standards Enforcement (DLSE).
- 25.4 The Labor Commissioner and the Division of Labor Standards Enforcement (DLSE) may conduct various compliance monitoring and enforcement activities including, but not limited to, confirming the accuracy of payroll records, conducting worker interviews, conducting audits, requiring submission of itemized statements prepared in accordance with Labor Code section 226, and conducting random in-person inspections of the Project site ("On-Site Visits"). On-Site Visits may include inspections of records, inspections of the Work site and observation of work activities, interviews of workers and others involved with the Project, and any other activities deemed necessary by the Labor Commissioner/DLSE to ensure compliance with prevailing wage requirements. The Labor Commissioner/DLSE shall have free access to any construction site or other place of labor and may obtain any information or statistics pertaining to the lawful duties of the Labor Commissioner/DLSE.
- 25.5 Any lawful activities conducted or any requests made by the Labor Commissioner/DLSE shall not be the basis for any delays, claims, costs, damages or liability of any kind against the District by the Contractor. Contractor and all Subcontractors shall cooperate and comply with any lawful requests by the Labor Commissioner's office. The failure of the Labor Commissioner, DLSE, or any other part of the Department of Industrial Relations to comply with any requirement imposed by the California Code of Regulations, Title 8, Chapter 8 shall not of itself constitute a defense to the failure to pay prevailing wages or to comply with any other obligation imposed by Division 2, Part 7, Chapter 1 of the Labor Code.



25.6 Prior to commencing any Work on the Project, the Contractor shall post the notice/poster required under the California Code of Regulations and Labor Code section 1771.4 in both English and Spanish at a conspicuous, weatherproof area at the Project site. The required notice/poster is available on the Labor Commissioner's website.

## 26. HOURS OF WORK

- 26.1 Eight (8) hours of work shall constitute a legal day's work. The Contractor and each subcontractor shall forfeit, as penalty to the District, twenty five dollars (\$25) for each worker employed in the execution of work on the Project by the Contractor or any subcontractor under him for each calendar day during which such worker is required or permitted to work more than eight (8) hours in any one calendar day and forty (40) hours in any calendar week in violation of the provisions of the Labor Code, and in particular, Section 1810 to Section 1815, thereof, inclusive, except that work performed by employees of the Contractor and his subcontractors in excess of eight hours per day at not less than one and one half times the basic rate of pay, as provided in Labor Code section 1815.
- Generally, construction work on the Project shall be accomplished on a regularly scheduled eight (8) hour per day work shift basis, Monday through Friday, between the hours of 7:00 a.m. and 5:00 p.m., however nothing herein shall prevent Contractor from working weekends and after school hours in order to complete the Project so long as not otherwise prohibited by law or local ordinances or regulations.
- Any work necessary to be performed after regular working hours, or on Sundays or other holidays shall be performed and included within the GMP, unless otherwise agreed to in writing before the work in question is commenced pursuant to Article 9, Extra Work/Modifications.

## 27. SKILLED AND TRAINED WORKFORCE

- 27.1 Contractor and all Subcontractors of any tier must comply with the requirements set forth in Education Code section 17407.5, including providing an enforceable commitment that the Contractor and all Subcontractors of any tier will use a "Skilled and Trained Workforce" as defined in Education Code section 17407.5 (b)(3). Contractor and all Subcontractors are to carefully review all requirements set forth in Education Code section 17407.5 before entering into the Contract for the Project.
- 27.2 The Contractor's commitment that a Skilled and Trained Workforce will be used to perform Work on the Project and the Contract shall be established by the following:
  - 27.2.1 Contractor shall include in all of its subcontracts, and Subcontractors shall require in its subcontracts of any tier, mandatory compliance with Education Code section 17407.5.
  - 27.2.2 Contractor shall provide to the District, on a monthly basis while the Project or Contract is being performed, a written report demonstrating that the Contractor and all Subcontractors of any tier are complying with the requirements set forth in Education Code section 17407.5. If Contractor fails to provide the monthly report, the District shall withhold payment for the portion of the monthly pay application related to the non-compliance of the Contractor or portion of the monthly pay application related to the non-compliance of any subcontractor. Upon notice to the Contractor of withholding for non-compliance, the non-compliant Contractor and/or subcontractor may cure the non-compliance. If Contractor and/or any subcontractor cures the non-compliance or substantially complies with required percentages of Public Contract Code section 2601 any monies withheld by the District will be released no later than the next pay application. If Contractor and/or any subcontractor fails to cure the non-compliance or substantially comply with the requirements within 60 days of notice of the noncompliance, the Contractor and/or non-compliant subcontractor shall meet and confer with the District to demonstrate the Contractor's and/or subcontractor's efforts and plan



to achieve substantial compliance with the requirements, on a cumulative basis, by completion of the Project. A meet and confer may be held earlier upon Contractor's written request to the District. The District, after reasonable demonstration by the Contractor and/or subcontractor that good faith and best efforts have been and are being made to substantially comply with the requirements of Section 2601, the District shall release any monies withheld. The Section 2601 percentages shall not apply to punchlist or warranty for a Contractor and/or subcontractor if that Contractor and/or subcontractor has otherwise substantially complied with the requirements of Section 2601 during the project. If Contractor and/or subcontractor is unable to substantially comply with the required percentages, on a cumulative basis, by completion of the project, the District, in its discretion may assess a charge to the non-compliant party of \$1,000.00 per percentage that the Contractor and/or subcontractor fails to meet the graduation percentage requirements, not to exceed \$5,000.00 or 10 percent of the total contract or subcontract value, whichever is less. This shall be the sole and exclusive remedy for Contractor's and/or any subcontractor's non-compliance with this section.

- 27.2.3 The monthly report provided to the District's Governing Board as required above shall be a public record under the California Public Records Act (Chapter 3.5 (commencing with Section 6250) of Division 7 of Title 1 of the Government Code), and shall be open to public inspection.
- 27.2.4 Contractor's commitment that a Skilled and Trained Workforce will be used to perform Work on the Project and the Contract may also be established by the Contractor providing evidence and any other information or documents reasonably requested by the District showing that the Contractor has entered into a project labor agreement that includes the requirements of Education Code section 17407.5(c) that will bind the Contractor and all its Subcontractors of any tier performing Work on the Project or Contract.
- 27.3 If the District's Governing Board has entered into a project labor agreement that will bind all contractors and subcontractors performing Work on this Project or Contract that includes the requirements of Education Code section 17407.5(c), the Contractor's agreement that it will become a party to that project labor agreement shall satisfy the requirements under Education Code section 17407.5(c).
- 27.4 If the Contractor or Subcontractor of any tier is not in compliance with all of the requirements set forth in Education Code section 17407.5, the District shall exercise any rights or remedies allowed under Education Code section 17407.5 or other applicable law.

#### 28. PROTECTION OF PERSONS AND PROPERTY

28.1 Fingerprinting. If any portion of the work for the Project is to be performed at an operating school, Contractor shall comply with the applicable requirements of Education Code sections 45125.1 and 45125.2 with respect to fingerprinting of employees who may have contact with District's pupils. Contractor shall also ensure that its Subcontractors on the Project comply with the applicable requirements of Sections 45125.1 and 45125.2. To this end, Contractor and its Subcontractors must provide for the completion of the Fingerprint Certification form attached as Attachment 14, 14A & 14B and incorporated herein by this reference prior to commencing work on the Project. In no event shall any employees of Contractor or its Subcontractors come into contact with District's pupils before the certification is completed. Contractor's failure to comply with this law shall be considered a material breach of the Agreement upon where the Agreement may be terminated, at District's sole discretion, without any further compensation to Contractor. Contractor and Subcontractor personnel on Site shall not have been convicted of any criminal offense which may have a discernible adverse impact on District or its students. Contractor shall advise its employees of these requirements before they enter on the Site and shall immediately remove from the Site any



- employee in violation of these requirements as determined by Contractor or by District. Contractor shall impose these requirements on its Subcontractors.
- 28.2 Contractor has been advised and is aware that District has adopted a Board Policy which prohibits the use of tobacco products, including smokeless tobacco, anywhere on District property. Contractor shall be responsible for the enforcement of District's tobacco-free policy among all Contractor's employees and Subcontractors while on District property. Contractor understands and agrees that should any employee or Subcontractor of Contractor violate the Board Policy, after having already been warned once for violating District's tobacco-free policy, Contractor shall remove the individual for the duration of the Project. Contractor shall not be entitled to any additional compensation and/or time in completing the Project as a result of such removal.
- 28.3 Contractor shall take all steps necessary to insure that employees of Contractor or any of its Subcontractors' employees do not use, consume, or work under the influence of alcohol or illegal drugs while on the Project. Contractor shall prevent any of its employees or its Subcontractors' employees from playing any recorded music devices or radios or wearing any radio headphone devices for entertainment while working on the Project. Contractor shall also prevent its employees or Subcontractors' employees from bringing any animal onto the Project.
- 28.4 Contractor shall be responsible for all damages to persons or property that occur as a result of its fault or negligence in connection with the prosecution of this Contract and shall take all necessary measures and be responsible for the proper care and protection of all materials delivered and work performed until completion and final acceptance by District.
- 28.5 Contractor shall take, and require Subcontractors to take, all necessary precautions for safety of workers on the work and shall comply with all applicable federal, state, local and other safety laws, standards, orders, rules, regulations, and building codes to prevent accidents or injury to persons on, about, or adjacent to premises where work is being performed and to provide a safe and healthful place of employment. In addition to meeting all requirements of OSHA, Cal-OSHA, state, and local codes, Contractor shall furnish, erect and properly maintain at all times, as directed by District or required by conditions and progress of work, all necessary safety devices, safeguards, construction canopies, signs, audible devices for protection of the blind, safety rails, belts and nets, barriers, lights, and watchmen for protection of workers and the public and shall post danger signs warning against hazards created by such features in the course of construction. Contractor shall designate a responsible member of its organization on the work, whose duty shall be to post information regarding protection and obligations of workers and other notices required under occupational safety and health laws, to comply with reporting and other occupational safety requirements, and to protect the life, safety and health of workers. Name and position of person so designated shall be reported to District by Contractor. Contractor shall correct any violations of safety laws, rules, orders, standards or regulations. Upon the issuance of a citation or notice of violation by the Division of Occupational Safety and Health, such violation shall be corrected promptly.
- In an emergency affecting safety of life or of work or of adjoining property, Contractor, without special instruction or authorization from District, is hereby permitted to act, at its discretion, to prevent such threatened loss or injury; and Contractor shall so act if so authorized or instructed by District. Any compensation claimed by Contractor on account of emergency work shall be determined by agreement.
- 28.7 Contractor shall provide such heat, covering, and enclosures as are necessary to protect all work, materials, equipment, appliances, and tools against damage by weather conditions.
- 28.8 Contractor shall take adequate precautions to protect existing roads, sidewalks, curbs, pavements, utilities, adjoining property and structures (including, without limitation, protection from settlement or loss of lateral support), and to avoid damage thereto, and repair any damage thereto caused by construction operations. All permits, licenses, or inspection fees required for such repair work shall be obtained and paid for by Contractor.



- 28.9 Trenches Five Feet or More in Depth. The Contractor shall submit to the District, in advance of excavation, a detailed plan showing the design of shoring, bracing, sloping or other provisions to be made for worker protection from the hazard of caving ground during the excavation of any trench or trenches five feet or more in depth. The Contractor shall also submit a copy of its annual trench/excavation permit approved by CAL-OSHA. The plan shall be prepared by a registered civil or structural engineer. As part of the plan, a note shall be included stating that the registered civil or structural engineer certifies that the plan complies with CAL OSHA Construction Safety Orders, or stating that the registered civil or structural engineer certifies that the plan is not less effective than the shoring, bracing, sloping, or other provisions of the Safety Orders.
  - 28.9.1 All shoring submittal shall include surcharge loads from adjacent embankments, construction loads and spoil bank. Submittal shall indicate minimum horizontal distance from top of trench to edge of all surcharge loads for all cases of shoring and side slopes.
  - 28.9.2 Nothing in this Section shall relieve Contractor of the full responsibility for providing shoring, bracing sloping, or other provisions adequate for worker protection. If such plan varies from the shoring system standards established by the Construction Safety Orders, the plan shall be prepared by a registered civil or structural engineer and shall be approved by CAL-OSHA. No excavation of such trench or trenches shall be commenced until said plan has been accepted by the District or the person to whom authority to accept has been delegated by the District.
- 28.10 Contractor shall (unless waived by District in writing):
  - 28.10.1 When performing construction on existing sites, become informed and take into specific account the maturity of the students on the site; and when performing work which may interfere with the school routine before, during or after school hours, enclose working area with a substantial barricade, and arrange work to cause minimum amount of inconvenience and danger to students and faculty in their regular school activities.
  - 28.10.2 Not allow any person, other than workers on the Project, or individuals authorized by District to come upon any portion of the premises where work is being performed. Contractor shall require all workers on the Project to be conspicuously identified either by a firm logo on their clothing, or by means of a prominent identification badge.
  - 28.10.3 Provide substantial barricades around any shrubs or trees indicated to be preserved.
  - 28.10.4 Deliver materials to building area over route designated by District.
  - 28.10.5 Take preventive measures to eliminate dust.
  - 28.10.6 Confine apparatus, the storage of materials, and the operations of workers to limits indicated by law, ordinances, permits, or directions of District; and shall not interfere with the work or unreasonably encumber premises or overload any structure with materials; and enforce all instructions of District regarding signs, advertising, fires, smoking, the presence of liquor, and the presence of firearms and require that all workers comply with all regulations while on construction site.
  - 28.10.7 Take care to prevent disturbing or covering any survey markers, monuments, or other devices marking property boundaries or corners. If such markers are disturbed by accident, they shall be replaced by an approved land surveyor or civil engineer at no cost to District.
  - 28.10.8 Not allow personal radios on the work site



- 28.10.9 Where the Project involves work at an operating school, inform and take such preventive measures necessary to insure that all employees, Subcontractors and other individuals authorized on the Project site refrain from any personal contact or conversations with the students on site.
- 28.10.10 Contractor shall not impose structural loading upon any part of the work under construction or upon existing construction on or adjacent to the Site in excess of safe limits, or loading such as to result in damage to the structural, architectural, mechanical, electrical, or other components of the work. The design of all temporary construction equipment and appliances used in construction of the work and not a permanent part thereof, including, without limitation, hoisting equipment, cribbing, shoring, and temporary bracing of structural steel, is the sole responsibility of Contractor. All such items shall conform with the requirements of governing codes and all laws, ordinances, rules, regulations, and orders of all authorities having jurisdiction. Contractor shall take reasonable and customary precautions, such as shoring of masonry walls and temporary tie bracing of structural steel work, to prevent possible wind damage during construction of the work. The installation of such bracing or shoring shall not damage the work in place or the work installed by others. Any damage which does occur shall be promptly repaired by Contractor at no cost to District.
- 28.10.11 Contractor shall require that Subcontractors participate in, and enforce, the safety and loss prevention programs established by Contractor for the Project, which will cover all work performed by Contractor and its Subcontractors. All Subcontractors and material or equipment suppliers shall cooperate fully with Contractor, District, and all insurance carriers. Subcontractors shall immediately, within twenty four (24) hours, report in writing to Contractor all accidents whatsoever arising out of, or in connection with, the performance of the work, whether on or off the Site, which caused death, personal injury, or property damage, giving full details and statements of witnesses. Contractor shall thereafter immediately, within two (2) days, report the facts in writing to District giving full details of the accident.
- 28.10.12 Contractor and Subcontractors shall use only those ingress and egress routes designated by District, observe the boundaries of the Site designated by District, park only in those areas designated by District, which areas may be on or off the Site, and comply with any parking control program established by District, such as furnishing license plate information and placing identifying stickers on vehicles.
- 28.10.13 Contractor shall be responsible for providing security services for the Site as needed for the protection of the Site and as determined in District's reasonable discretion.
- 28.10.14 Contractor shall, for all contracts involving state funds, submit a "Drug-Free Workplace Certification." Contractor shall take all reasonable steps necessary to ensure that any employees of Contractor or any of its Subcontractors' employees report for work in a manner fit to do their job. Such employees shall not be under the influence of or in possession of any alcoholic beverage or of any controlled substance (except a controlled substance as prescribed by a physician so long as the performance or safety at the Project Site is not affected thereby). Contractor shall advise its employees of these requirements before they enter on the Site and shall immediately remove from the site any employee in violation of these requirements as determined by Contractor or by the District. Contractor shall impose these requirements on its Subcontractors.
- 28.10.15 Contractor and Subcontractors shall at all times enforce strict discipline and good order among their employees and other persons carrying out the Contract and shall not employ on work any unfit person or anyone not skilled in work assigned to such person. It shall be the responsibility of Contractor to ensure compliance with this Article. Any person in the employ of Contractor or Subcontractors whom District may deem



incompetent, unfit, intemperate, troublesome or otherwise undesirable shall be excluded from the work Site and shall not again be employed on it except with written consent of District. Contractor must sign and cause all Subcontractors to sign the Conduct Rules for Contractors form attached as "Attachment 17" and incorporated herein by this reference prior to commencing work on the Project.

28.11 Contractor shall be at all times during the performance of work hereunder in full compliance with the provisions of the Immigration Reform and Control Act of 1986 ("IRCA") in the hiring of its employees, and Contractor shall indemnify, hold harmless and defend District against any and all actions, proceedings, penalties or claims arising out of Contractor's failure to comply strictly with the IRCA.

## 29. PAYMENTS AND RETENTION

The Construction Cost of the each Project shall not exceed the GMP identified in Exhibit K as may be amended from time to time, except as otherwise provided in this Construction Services Agreement and Sublease. During the progress of construction, Contractor will provide monthly progress payment applications for the total scheduled value of the work completed under the GMP set forth in Article 3. District shall pay to Contractor a monthly progress payment comprising a sum equal to ninety-five percent (95%) of the scheduled value of the work approved and completed up to the last day of the previous month, less aggregate of previous payments("Progress Payment"). If all of the necessary information is submitted and accurate (including the schedule of values), District shall approve the Progress Payments within fifteen (15) days after District's receipt of the periodic estimate for partial payment and District shall pay such payments within fifteen (15) days after the District's approval of the periodic estimate for partial payment. Progress Payments shall be made on the basis of monthly estimates which shall be prepared by Contractor on a form approved by District and certified by Architect and Project Inspector, or any other approved representative of the District, and filed before the fifth day of the month during which payment is to be made. Work completed as estimated shall be an estimate only and no inaccuracy or error in said estimate shall release Contractor or any bondsman from such work or from enforcing each and every provision of this document and District shall have the right subsequently to correct any error made in any estimate for payment. Contractor shall not be entitled to have any payment estimates processed or be entitled to have any payment made for work performed so long as any lawful or proper direction concerning non-complying work or any portion thereof given by the District lacks correction by Contractor. District shall withhold from the Progress Payments 150% of the estimated value of non-complying work unless satisfactorily corrected or remedied. Contractor shall, at a minimum, provide the following documents as part of its request for a Progress Payment: (1) Schedule of Values, (2) Project Contingency Trackers, (3) Project Allowance Trackers, (4) Project Savings Reports (Refer to the Project Savings Section for the Project Savings Items) including the budget versus actual costs of Project Management and General Condition Expenses, (5) Project Daily Reports (Contractor and Subcontractor), (6) Project Safety Reports, (7) Monthly Lien Releases Unconditional and Conditional Waivers (all contractors), and (8) Monthly Schedule Update and Narratives (with Recovery Schedules as needed).

- 29.1 The District shall retain five percent (5%) "Retention" from Progress Payments and release Retention as required in this CSA and specifically, not until after Close-Out under Article 13.16.
- In no event shall the cumulative total of the Progress Payments/ Sublease Payments and Retention ever exceed the GMP as defined herein, unless specifically allowed under Article 5.
  - 29.2.1 Title to new materials and/or equipment for the work of this contract, on a continuous basis while the Project is being completed, shall vest in the District. However, responsibility for such new material and/or work of this contract shall remain with the Contractor until incorporated into the work and accepted by District; no part of said materials and/or equipment shall be removed from its place of storage except for immediate installation in the work of this contract; and Contractor shall keep an accurate



inventory of all said materials and/or equipment in a manner satisfactory to the owner or his authorized representative.

Notwithstanding anything to the contrary stated above, the Contractor may include in its request for payment the value of any structural steel, glue laminated beams, trusses, bleachers and other such custom-made materials prepared specifically for the Project and unique to the Project so long as all of the following requirements are satisfied:

- 29.2.1.1 The aggregate cost of materials stored off-site shall not exceed Twenty Five Thousand Dollars (\$25,000) at any time or as otherwise agreed to be District in writing;
- 29.2.1.2 Title to such materials shall be vested in the District as evidenced by documentation satisfactory in form and substance to the District, including, without limitation, recorded financing statements, UCC filings and UCC searches;
- 29.2.1.3 With each request for payment, the Contractor shall submit to the District a written list identifying each location where materials are stored off-site (which must be a bonded warehouse) and the value of the materials at each location. The Contractor shall procure insurance satisfactory to the District (in its reasonable discretion) for materials stored off-site in an amount not less than the total value thereof;
- 29.2.1.4 The consent of any Surety shall be obtained to the extent required prior to payment for any materials stored off-site;
- 29.2.1.5 Representatives of the District shall have the right to make inspections of the storage areas at any time; and
- 29.2.1.6 Such materials shall be (1) protected from diversion, destruction, theft and damage to the reasonable satisfaction of the District; (2) specifically marked for use on the Project; and (3) segregated from other materials at the storage facility.
- 29.3 <u>Reasons to Withhold Payment.</u> The District may withhold any payment, in whole, or in part, to such extent as may be necessary to protect the District from loss because of, but not limited to:
  - 1. Defective Work not remedied;
  - 2. Stop Notices served upon the District;
  - 3. Liquidated damages assessed against the Contractor;
  - 4. The cost of completion of the Contract if there exists reasonable doubt that the Work can be Completed for the unpaid balance of any Contract Price or by the completion date;
  - 5. Damage to the District or other contractor;
  - 6. Unsatisfactory prosecution of the Work by the Contractor;
  - 7. Failure to store and properly secure materials;
  - 8. Failure of the Contractor to submit on a timely basis, proper and sufficient documentation required by the Contract Documents, including, without limitation, acceptable monthly progress schedules, Shop Drawings, Submittal schedules, schedule of values, product data



- and samples, proposed product lists, executed Change Order, Construction Change Documents, and verified reports;
- 9. Failure of the Contractor to maintain As-Built drawings;
- 10. If, in the District's opinion, the representations to the District required pursuant to Article 9.4 cannot be made;
- 11. Erroneous estimates by the Contractor of the value of the Work performed, or other false statements in an application for payment;
- 12. Unauthorized deviations from the Contract Documents (including but not limited to Unresolved Notices of Deviations (DSA Form 154);
- 13. Failure of the Contractor to prosecute the Work in a timely manner in compliance with established progress schedules and completion dates;
- 14. Failure to properly pay prevailing wages as defined in Labor Code section 1720, et seq.;
- 15. Failure to properly maintain or clean up the Site;
- 16. Payments to indemnify, defend, or hold harmless the District;
- 17. Any payments due to the District including, but not limited to, payments for failed tests, or utilities changes or permits;
- 18. Failure to submit an acceptable schedule in accordance with Article 9;
- 19. Failure to pay Subcontractor or suppliers;
- 20. Failure to secure warranties, including the cost to pay for warranties
- 21. Failure to provide release from material suppliers or Subcontractors when requested to do so
- 22. Items deducted pursuant to Article 17.6
- 23. Incomplete Punch List items under Article 13.6 which have gone through the Article 12.2 process
- 24. Allowances that have not been used
- 29.4 Reallocation of Withheld Amounts. District may, in its discretion, apply any withheld amount to payment of outstanding claims or obligations. In so doing, District shall make such payments on behalf of Contractor. If any payment is so made by District, then such amount shall be considered as a payment made under this CSA to Contractor and District shall not be liable to Contractor for such payments made in good faith. Such payments may be made without prior judicial determination of claim or obligation. District will render Contractor an accounting of such funds disbursed on behalf of Contractor.

If Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents or fails to perform any provision thereof, District may, after ten (10) calendar days written notice to the Contractor and without prejudice to any other remedy make good such deficiencies. The District shall adjust the total Contract price by reducing the amount thereof by the cost of making good such deficiencies. If District deems it inexpedient to correct Work which is damaged, defective, or not done in accordance with Contract provisions, an equitable reduction in the Contract price (of at least 150% of the estimated reasonable value of the nonconforming Work) shall be made therefor.



29.5 <u>Payment After Cure.</u> When the grounds for declining approval are removed, payment shall be made for amounts withheld because of them. No interest shall be paid on any retention or amounts withheld due to the failure of the Contractor to perform in accordance with the terms and conditions of the Contract Documents.

# 30. NONCONFORMING WORK

Contractor shall promptly remove from premises all Work identified by District as failing to conform to the Contract whether incorporated or not. Contractor shall promptly replace and re-execute its own Work to comply with the Contract without additional expense to District and shall bear the expense of making good all Work of other contractors destroyed or damaged by such removal or replacement.

If Contractor does not remove such Work which has been identified by District as failing to conform to the Contract Documents within a reasonable time, fixed by written notice, District may remove it and may store the material at Contractor's expense. If Contractor does not pay expenses of such removal within ten (10) calendar days' time thereafter, District may, upon ten (10) calendar days' written notice, sell such materials at auction or at private sale and shall account for net proceeds thereof, after deducting all costs and expenses that should have been borne by Contractor.

## 31. SUBCONTRACTOR PAYMENTS

- 31.1 Payments to Subcontractors. No later than ten (10) days after receipt, or pursuant to Business and Professions Code section 7108.5, the Contractor shall pay to each Subcontractor, out of the amount paid to the Contractor on account of such Subcontractor's portion of the Work, the amount to which said Subcontractor is entitled. The Contractor shall, by appropriate agreement with each Subcontractor, require each Subcontractor to make payments to Sub-subcontractors in a similar manner.
- 31.2 <u>No Obligation of District for Subcontractor Payment.</u> The District shall have no obligation to pay, or to see to the payment of, money to a Subcontractor except as may otherwise be required by law.
- 31.3 <u>Payment Not Constituting Approval or Acceptance.</u> An approved request for a Progress Payment, a Certificate of Substantial Completion, or partial or entire use or occupancy of the Project by the District shall not constitute acceptance of Work that is not in accordance with the Contract Documents.
- Joint Checks. District shall have the right, if necessary for the protection of the District, to issue joint checks made payable to the Contractor and Subcontractors and material or equipment suppliers. The joint check payees shall be responsible for the allocation and disbursement of funds included as part of any such joint payment. In no event shall any joint check payment be construed to create any contract between the District and a Subcontractor of any tier, any obligation from the District to such Subcontractor, or rights in such Subcontractor against the District. The District may choose to issue joint checks at District's sole discretion and only after all the requirements of that particular school district and county are specifically met. Some school districts cannot issue joint checks, so the ability to issue joint checks will depend on the District and the specific circumstances.

# 32. <u>SEPARATE CONTRACTS</u>

32.1 Reservation of Rights to have other Contractors on Site. District reserves the right to let other contractors enter the Site to perform work as part of its use of the Site. Contractor shall afford other contractors reasonable opportunity for the introduction and storage of their materials and the execution of their work and shall properly connect and coordinate the Project with the work of such Contractors. Such contractors shall comply with all applicable State safety laws and regulations and shall provide a certificate of insurance naming Contractor as additional insured



Notice of Coordination of Work. If the proper execution of any part of the Contractor's work on the Project depends upon the work of any such contractors, Contractor shall inspect and promptly report to District any patent defects or other problems it identifies in such work that render it unsuitable for such proper execution and results. Contractor is only required to inspect the work of such other contractors prior to commencing its own further work in connection with or in relation to that other work. Further, Contractor is only expected to identify patent defects or other problems, and is not required to do any destructive testing or to monitor the progress of such work by other contractors prior to its completion. In no event shall the work of such other contractors be covered by the warranty given by Contractor to the District, nor shall Contractor be required to provide insurance for such work.

# 33. USE OF PREMISES/SAFETY

Contractor shall confine operations at the Site to areas permitted by law, ordinances, permits and the Construction Documents and shall not unreasonably encumber the Site or existing facilities on the Site with any materials or equipment. Contractor shall not load or permit any part of the work to be loaded with a weight so as to endanger the safety of persons or property at the Site. The Contractor shall maintain emergency first aid treatment for his employees which complies with the Federal Occupational Safety and Health Act of 1970 (29 USC, section 651 et seq.).

## 34. CLEANING UP

- 34.1 <u>Contractor's Responsibility to Clean Up.</u> Contractor at all times shall keep premises free from debris such as waste, dust, excess water, storm water runoffs, rubbish, and excess materials and equipment. Contractor shall not leave debris under, in, or about the premises, but shall promptly remove same from the premises and dispose of it in a lawful manner. Disposal receipts or dump tickets shall be furnished to the Architect within five (5) days of request.
  - Contractor shall remove rubbish and debris resulting from the Work on a daily basis. Contractor shall maintain the structures and Site in a clean and orderly condition at all times until acceptance of the Project by the District. Contractor shall keep its access driveways and adjacent streets, sidewalks, gutters and drains free of rubbish, debris and excess water by cleaning and removal each day. All concrete, sidewalks, and paths of travel shall be broom cleaned daily.
- 34.2 <u>General Final Clean-Up.</u> Upon completion of Work, Contractor shall employ experience workers or professional cleaners for final cleaning. Clean each surface to the condition expected in a normal, commercial, building cleaning and maintenance program.
  - 1. Clean interior and exterior of buildings, including fixtures, equipment, walls, floors, ceilings, roofs, window sills and ledges, horizontal projections, and any areas where debris has collected, so surfaces are free from foreign material or discoloration;
  - 2. Clean the Project site. The grounds should be cleared of any Contractor equipment, raked clean of debris and trash removed. Sweep paved areas broom clean.
  - 3. Repair or replace any damaged materials. Replace any chipped or broken glass.
  - 4. Remove any and all stains.
  - 5. Remove labels that aren't permanent labels.
  - 6. Clean and polish all glass, plumbing fixtures, equipment, finish hardware and similar finish surfaces. Remove any glazing compounds
  - 7. Remove temporary utilities, fencing, barricades, planking, sanitary facilities and similar temporary facilities from Site.



- 8. Remove temporary film that remains on any hardware, doors or other surfaces.
- 9. Seal the bottom and tops of all doors
- 10. Special Clean-Up.
- 11. In addition to the general cleaning, the following special cleaning shall be done at the completion of the Work in accordance with the specifications including, but not limited to:
  - a. Remove putty stains from glazing, then wash and polish glazing.
  - Remove marks, stains, fingerprints and other soil or dirt from painted, stained or decorated work.
  - c. Remove temporary protection and clean and polish floors and waxed surfaces.
  - d. Clean and polish hardware and plumbing trim; remove stains, dust, dirt, plaster and paint
  - e. Wipe surfaces of mechanical and electrical equipment.
  - f. Remove spots, soil, plaster and paint from tile work, and wash tile.
  - g. Clean all fixtures and equipment, remove excess lubrication, clean light fixtures and lamps, polish metal surfaces.
  - h. Vacuum-clean carpeted surfaces.
  - i. Remove debris from roofs, down spout and drainage system.
- 34.3 <u>Failure to Cleanup.</u> If the Contractor fails to clean up as provided in the Contract Documents, the District may do so, and the cost thereof shall be the responsibility of the Contractor pursuant to Article 12.2 and seek a Deductive Change Order.

# 35. INSURANCE

- Insurance Requirements. Before the commencement of the Work, the Contractor shall purchase from and maintain in a company or companies lawfully authorized to do business in California as admitted carriers with a financial rating of at least an A status as rated in the most recent edition of Best's Insurance Reports or as otherwise amended in these Contract Documents, such insurance as will protect the District from claims set forth below, which may arise out of or result from the Contractor's operations under the Contract and for which the Contractor may be legally liable, whether such operations are by the Contractor, by a Subcontractor, by anyone directly or indirectly employed by any of them, or by anyone for whose acts any of them may be liable:
  - 12. Claims for damages because of bodily injury, sickness, disease, or death of any person District would require indemnification and coverage for employee claim;
  - 13. Claims for damages insured by usual personal injury liability coverage, which are sustained by a person as a result of an offense directly or indirectly related to employment of such person by the Contractor or by another person;
  - 14. Claims for damages because of injury or destruction of tangible property, including loss of use resulting therefrom, arising from operations under the Contract Documents;



- 15. Claims for damages because of bodily injury, death of a person, or property damage arising out of the ownership, maintenance, or use of a motor vehicle, all mobile equipment, and vehicles moving under their own power and engaged in the Work;
- 16. Claims involving contractual liability applicable to the Contractor's obligations under the Contract Documents, including liability assumed by and the indemnity and defense obligations of the Contractor and the Subcontractors; and
- 17. Claims involving Completed Operations, Independent Contractors' coverage, and Broad Form property damage, without any exclusions for collapse, explosion, demolition, underground coverage, and excavating. (XCU)
- 18. Claims involving sudden or accidental discharge of contaminants or pollutants.
- 35.2 <u>Subcontractor Insurance Requirements.</u> The Contractor shall require its Subcontractors to take out and maintain similar public liability insurance and property damage insurance required under this Article in like amounts. A "claims made" or modified "occurrence" policy shall not satisfy the requirements of this Article without prior written approval of the District.
- Additional Insured Endorsement Requirements. The Contractor shall name, on any policy of insurance required under this Article, the District, Architect, Inspector, the State of California, their officers, employees, agents and independent contractors as additional insureds. Subcontractors shall name the Contractor, the District, Architect, Inspector, the State of California, their officers, employees, agents and independent contractors as additional insureds. The Additional Insured Endorsement included on all such insurance policies shall be an ISO CG 20 33 (04/813), or an ISO CG 20 38 (04/13) and ISO CG 20 37 (04/13) or their equivalent as determined by the District in its sole discretion, and must state that coverage is afforded the additional insured with respect to claims arising out of operations performed by or on behalf of the insured. If the additional insureds have other insurance which is applicable to the loss, such other insurance shall be on an excess or contingent basis. The insurance provided by the Contractor pursuant to this Article must be designated in the policy as primary to any insurance obtained by the District. The amount of the insurer's liability shall not be reduced by the existence of such other insurance.
- 35.4 Specific Insurance Requirements
  - Contractor shall take out and maintain and shall require all Subcontractors, if any, whether primary or secondary, to take out and maintain:
  - 35.4.2 Comprehensive General Liability Insurance with a combined single limit per occurrence of not less than \$2,000,000.00 or Commercial General Liability Insurance (including automobile insurance) which provides limits of not less than:

a a a y man a a (a a mhin a d ain a la limit)

19.	Per occurrence (combined single limit)	\$1,000,000.00
20.	Project Specific Aggregate (for this Project only)	\$2,000,000.00
21.	Products and Completed Operations	\$1,000,000.00
22.	Personal and Advertising Injury Limit	\$1,000,000.00

Professional Liability Insurance, including contractual liability, with limits of \$1,000,000 per claim and \$2,000,000 in the aggregate. Such insurance shall be maintained during the term of this Preconstruction Services Scope of Work and renewed at rates consistent with the time of execution of the Construction Services Agreement adjusted for inflation. If the Professional Liability Insurance policy is written on a claims made basis, it shall be maintained continuously for a period of no less than three (3) years after a Notice of Completion is recorded for the Project to which it applies.

\$1,000,000,00



The "retro date" must be shown and must be before the date of this Construction Services Agreement.

- 35.4.4 Sexual Abuse/Molestation Coverage The CONTRACTOR shall carry and maintain during the term of this AGREEMENT a policy with a limit of not less than \$1,000,000 Sexual Abuse Injury Limit of Insurance. The CONTRACTOR shall hold the DISTRICT harmless from any and all claims for injury, damage, and loss.
- Employer's Liability Insurance in an amount not less than \$1,000,000 per accident for bodily injury or disease.
- 35.4.6 Insurance Covering Special Hazards. The following Special hazards shall be covered by riders or riders to above mentioned public liability insurance or property damage insurance policy or policies of insurance, in amounts as follows:

23	Automotive and truck where operated in amounts	\$1,000,000.00
4.7.	Automotive and truck where operated in amounts	D1.000.000.00

24. Material Hoist where used in amounts \$1,000,000.00

25. Explosion, Collapse and Underground (XCU coverage) \$1,000,000.00

- 26. In addition, provide Excess Liability Insurance coverage in the amount of Five Million Dollars (\$5,000,000.00).
- Workers' Compensation Insurance. During the term of this Contract, the Contractor shall provide workers' compensation insurance (not less than \$1M) for all of the Contractor's employees engaged in Work under this Contract on or at the Site of the Project and, in case any of the Contractor's Work is subcontracted, the Contractor shall require the Subcontractor to provide workers' compensation insurance (not less than \$1M) for all the Subcontractor's employees engaged in Work under the subcontract. Any class of employee or employees not covered by a Subcontractor's insurance shall be covered by the Contractor's insurance. In case any class of employees engaged in Work under this Contract on or at the Site of the Project is not protected under the Workers' Compensation laws, the Contractor shall provide or cause a Subcontractor to provide adequate insurance coverage for the protection of those employees not otherwise protected. The Contractor shall file with the District certificates of insurance and in comply with Labor Code § 3700.

# 35.6 Builder's Risk/All Risk

35.6.1 Course-of-Construction Insurance Requirements. The Contractor, during the progress of the Work and until final acceptance of the Work by District upon completion of the entire Contract, shall maintain Builder's Risk, Course of Construction or similar first party property coverage issued on a replacement value basis consistent with the total replacement cost of the structures where work is being performed inclusive of all Work for the Project included within the Contract Documents. Coverage is to insure against all risks of accidental direct physical loss, and must include, by the basic grant of coverage or by endorsement, the perils of vandalism, malicious mischief (both without any limitation regarding vacancy or occupancy), fire, sprinkler leakage, civil authority, sonic boom, earthquake, flood, collapse, wind, lightning, smoke and riot. The coverage must include debris removal, demolition, increased costs due to enforcement of building ordinance and law in the repair and replacement of damage and undamaged portions of the property, and reasonable costs for the Architect's and engineering services and expenses required as a result of any insured loss upon the Work and Project which is the subject of the Contract Documents, including the underlying structure where Work is being performed, completed Work and Work in progress, to the full insurable value thereof. Such insurance shall include the District and the Architect as additional named insureds, and any other person with an insurable interest as designated by the District.



The Contractor shall submit to the District for its approval all items deemed to be uninsurable. The risk of the damage to the Work due to the perils covered by the "Builder's Risk/All Risk" Insurance, as well as any other hazard which might result in damage to the Work, is that of the Contractor and the surety, and no claims for such loss or damage shall be recognized by the District nor will such loss or damage excuse the complete and satisfactory performance of the Contract by the Contractor.

- 35.7 <u>Fire Insurance.</u> Before the commencement of the Work, the Contractor shall procure, maintain, and cause to be maintained at the Contractor's expense, fire insurance on all Work subject to loss or damage by fire. The amount of fire insurance shall be sufficient to protect the Project against loss or damage in full until the Work is accepted by the District.
- 35.8 <u>Other Insurance.</u> The Contractor shall provide all other insurance required to be maintained under applicable laws, ordinances, rules, and regulations.
- 35.9 <u>Proof of Insurance.</u> The Contractor shall not commence Work nor shall it allow any Subcontractor to commence Work under this Contract until all required insurance and certificates have been obtained and delivered in duplicate to the District for approval subject to the following requirements:
  - 35.9.1 Certificates and insurance policies shall include the following clause:
    - 27. "This policy shall not be non-renewed, canceled, or reduced in required limits of liability or amounts of insurance until notice has been mailed to the District. Date of cancellation or reduction may not be less than thirty (30) days after the date of mailing notice."
    - 28. Certificates of insurance shall state in particular those insured, the extent of insurance, location and operation to which the insurance applies, the expiration date, and cancellation and reduction notices.
    - 29. Certificates of insurance shall clearly state that the District and the Architect are named as additional insureds under the policy described and that such insurance policy shall be primary to any insurance or self-insurance maintained by District.
    - 30. The Contractor and its Subcontractors shall produce a certified copy of any insurance policy required under this Article upon written request of the District.
- 35.10 <u>Compliance</u>. In the event of the failure of Contractor to furnish and maintain any insurance required by this Article 34, the Contractor shall be in default under the Contract. Compliance by Contractor with the requirement to carry insurance and furnish certificates or policies evidencing the same shall not relieve the Contractor from liability assumed under any provision of the Contract Documents, including, without limitation, the obligation to defend and indemnify the District and the Architect.
- 35.11 No Waiver Created through Payments. The making of any payments under this CSA or the Sublease shall not be construed as creating an insurable risk interest by or for the District or be construed as relieving the Contractor or his Subcontractors of responsibility for loss from any direct physical loss, damage, or destruction occurring prior to Completion of the Project.
- 35.12 <u>Waiver of Subrogation.</u> Contractor waives (to the extent permitted by law) any right to recover against the District for damages to the Work, any part thereof, or any and all claims arising by reason of any of the foregoing, but only to the extent that such damages and/or claims are covered by property insurance and only to the extent of such coverage (which shall exclude deductible amounts) by insurance actually carried by the District.



The provisions of this section are intended to restrict each party to recovery against insurance carriers only to the extent of such coverage and waive fully and for the benefit of each, any rights and/or claims which might give rise to a right of subrogation in any insurance carrier. The District and the Contractor shall each obtain in all policies of insurance carried by either of them, a waiver by the insurance companies thereunder of all rights of recovery by way of subrogation for any damages or claims covered by the insurance.

## 35.13 Performance and Payment Bonds

35.13.1 Bond Requirements. Prior to commencing any portion of the Work, the Contractor shall furnish separate payment and performance bonds for its portion of the Work which shall cover 100% faithful performance of and payment of all obligations arising under the Contract Documents and/or guaranteeing the payment in full of all claims for labor performed and materials supplied for the Work. All bonds shall be provided by a corporate surety authorized and admitted to transact business in California as sureties.

To the extent, if any, that the Contract Price is increased in accordance with the Contract Documents, the Contractor shall, upon request of the District, cause the amount of the bonds to be increased accordingly and shall promptly deliver satisfactory evidence of such increase to the District. To the extent available, the bonds shall further provide that no change or alteration of the Contract Documents (including, without limitation, an increase in the Contract Price, as referred to above), extensions of time, or modifications of the time, terms, or conditions of payment to the Contractor will release the surety. If the Contractor fails to furnish the required bonds, the District may terminate the Contract for cause.

- 35.13.2 Surety Qualification. Only bonds executed by admitted Surety insurers as defined in Code of Civil Procedure § 995.120 shall be accepted. Surety must be a California-admitted surety and listed by the U.S. Treasury with a bonding capacity in excess of the Project cost.
- 35.13.3 Alternate Surety Qualifications. If a California-admitted surety insurer issuing bonds does not meet these requirements, the insurer will be considered qualified if it is in conformance with § 995.660 of the California Code of Civil Procedure and proof of such is provided to the District.
- 35.13.4 Contractor is hereby authorized to obtain a performance and payment bond from any Subcontractors selected by Contractor at its discretion and cost. Any bonds required by this subsection shall comply with the requirements set forth above.

## 36. HOLD HARMLESS AND INDEMNITY

Contractor shall defend, indemnify and hold harmless District, Construction Manager, and their officers, employees, and agents (excluding the Inspector, Architect, and any other design or engineering professionals retained by District or Architect) from all liabilities, claims, actions, liens, judgments, demands, damages, losses, costs or expenses of any kind arising from death, personal injury, property damage or other cause based or asserted upon any act, omission, or breach connected with or arising from the progress of Work or performance of service under this Agreement or the Contract Documents. As part of this indemnity, Contractor shall protect and defend, at its own expense, District, Construction Manager, and their officers, employees, and agents (excluding the Inspector, Architect, and any other design or engineering professionals retained by District or Architect) from any legal action including attorney's fees or other proceeding based upon such act, omission, breach or as otherwise required by this Article.

Furthermore, Contractor agrees to and does hereby defend, indemnify and hold harmless District, Construction Manager, and their officers, employees, and agents (excluding the Inspector, Architect, and any other design or engineering professionals retained by District or Architect) from every claim or demand made, and every liability, loss, damage, expense or attorney's fees of any nature whatsoever, which may be incurred by reason of:



- Liability for (1) death or bodily injury to persons; (2) damage or injury to, loss (including theft), or loss of use of, any property; (3) any failure or alleged failure to comply with any provision of law or the Contract Documents; or (4) any other loss, damage or expense, sustained by any person, firm or corporation or in connection with the Work called for in this Agreement or the Contract Documents, except for liability resulting from the sole or active negligence, or the willful misconduct of the District.
- Any bodily injury to or death of persons or damage to property caused by any act, omission or breach of Contractor or any person, firm or corporation employed by Contractor, either directly or by independent contract, including all damages or injury to, loss (including theft), or loss of use of, any property, sustained by any person, firm or corporation, including District, arising out of or in any way connected with Work covered by this Agreement or the Contract Documents, whether said injury or damage occurs either on or off District property, but not for any loss, injury, death or damages caused by the sole or active negligence or willful misconduct of the District.
- Any dispute between Contractor and Contractor's Subcontractors/supplies/sureties, including, but not limited to, any failure or alleged failure of the Contractor (or any person hired or employed directly or indirectly by the Contractor) to pay any Subcontractor or Materialman of any tier or any other person employed in connection with the Work and/or filing of any stop notice or mechanic's lien claims.

Contractor, at its own expense, cost, and risk, shall defend any and all claims, actions, suits, or other proceedings that may be brought or instituted against the District, its officers, agents, CM, or employees (excluding the Inspector, Architect, and any other design or engineering professionals retained by District or Architect) on account of or founded upon any cause, damage, or injury identified herein and shall pay or satisfy any judgment that may be rendered against the District, its officers, agents or employees (excluding the Inspector, Architect, and any other design or engineering professionals retained by District or Architect) in any action, suit or other proceedings as a result thereof.

Contractor shall ensure that its contract with each of its Subcontractors contains provisions requiring the Subcontractors to defend, indemnify and hold harmless the District, Architect, Inspector, the State of California to a minimum level as set forth in this Article and consistent with the language of this Article.

The Contractor's and Subcontractors' obligation to defend, indemnify and hold harmless the District, Architect, and their officers, employees, and agents (excluding the Inspector, Architect, and any other design or engineering professionals retained by District or Architect)hereunder shall include, without limitation, any and all claims, damages, and costs for the following: (1) any damages or injury to or death of any person, and damage or injury to, loss (including theft), or loss of use of, any property; (2) breach of any warranty, express or implied; (3) failure of the Contractor or Subcontractors to comply with any applicable governmental law, rule, regulation, or other requirement; (4) products installed in or used in connection with the Work; and (5) any claims of violation of the Americans with Disabilities Act ("ADA") claims arising from failure to comply with the Construction Documents.

## 37. SUBSTITUTION OF SECURITY

In accordance with Public Contract Code section 22300, the District will permit the substitution of securities for any moneys withheld by the District to ensure performance under the Construction Services Agreement. At the request and expense of the Contractors, securities equivalent to the amount withheld shall be deposited with the District, or with a state or federally chartered bank as the escrow agent, who shall then pay such moneys to the Contractor. Upon satisfactory completion of the Construction Services Agreement the securities shall be returned to the Contractor.

#### 38. TITLE TO WORK

Title to all work completed and in the course of construction paid for by District and title to all materials on account of which payment has been made by District to Contractor shall vest in District pursuant to the applicable provisions of the Sublease.



# 39. COMPLIANCE WITH STATE STORM WATER PERMIT FOR CONSTRUCTION

The Contractor shall be required to comply with all conditions of the State Water Resources Control Board (State Water Board) National Pollutant Discharge Elimination System General Permit for Waste Discharge Requirements for Discharges of Storm Water Runoff Associated with Construction Activity (Permit) for all construction activity which results in the disturbance of in excess of one acre of total land area or which is part of a larger common area of development or sale. The Contractor shall be responsible for filing the Notice of Intent and for obtaining the Permit. The Contractor shall be solely responsible for preparing and implementing a Storm Water Pollution Prevention Plan (SWPPP) prior to initiating Work. The Contractor's Qualified SWPPP Developer (QSD) shall work with the Architect and its engineers in preparing an approved SWPPP and revising it as necessary or required. It shall be the Contractor's responsibility to evaluate the cost of procuring the Permit and preparing the SWPPP as well as complying with the SWPPP and any necessary revision to the SWPPP. The Contractor shall employ a Qualified SWPPP Practitioner (QSP) to implement the approved SWPPP during construction. The Contractor shall comply with all requirements of the State Water Resources Control Board. The Contractor shall include all costs of compliance with specified requirements in the GMP.

Contractor shall be responsible for procuring, implementing and complying with the provisions of the Permit and the SWPPP, including the standard provisions, monitoring and reporting requirements as required by the Permit. Contractor shall provide copies of all reports and monitoring information to the District, Architect and the District's third party SWPPP consultant.

The Contractor shall comply with the lawful requirements of any applicable municipality, the County, drainage district, and other local agencies regarding discharges of storm water to separate storm drain system or other watercourses under their jurisdiction, including applicable requirements in municipal storm water management programs.

Failure to comply with the Permit is in violation of federal and state law. The Contractor hereby agrees to indemnify and hold harmless the District, its Board members, officers, agents, employees and authorized volunteers from and against any and all claims, demands, losses or liabilities of any kind or nature which the District, its Board members, officers, agents, employees and authorized volunteers may sustain or incur for noncompliance with the Permit arising out of or in connection with the Project, except for liability resulting from the sole negligence, willful misconduct or active negligence of the District, its Board members, officers, agents, employees or authorized volunteers. District may seek damages from the Contractor for delay in completing the Project caused by the Contractor's failure to comply with the Permit.

## 40. EQUAL OPPORTUNITY CLAUSE

The Contractor herein agrees not to discriminate in its recruiting, hiring, promotion, demotion or termination practices on the basis of race, religious creed, color, national origin, ancestry, sex, age or physical handicap in the performance of this Construction Services Agreement and to comply with the provisions of the following laws:

- 40.1 California Fair Employment and Housing Act (Gov. Code 12900 et seq., prohibiting discrimination in employment on account of race, religious creed, color, national origin, ancestry, physical disability, mental disability, medical condition, marital status, or sex, and prohibiting harassment of an employee or applicant because of race, religious creed, color, national origin, ancestry, physical disability, mental disability, medical condition, marital status, sex, or age);
- 40.2 Federal Civil Rights Act of 1964 (42 USC '2000e et seq., prohibiting discrimination in employment on the basis of race, color, national origin, religion, or sex); Title I of the Americans With Disabilities Act of 1990 (42 USC 12101 et seq., prohibiting discrimination against qualified individuals with a disability in hiring and employment practices);
- 40.3 The Age Discrimination in Employment Act (29 USC 621 et seq., prohibiting age discrimination in employment against individuals who are at least forty years of age);



- 40.4 California Labor Code section 1102.1 (prohibiting discrimination in any aspect of employment or opportunity for employment based on actual or perceived sexual orientation);
- 40.5 Sexual orientation;
- 40.6 American with Disabilities Act (ADA) (See Article 41); and
- 40.7 Any other laws or regulations prohibiting discrimination as may be applicable to Contractor.

# 41. SPECIAL NOTICE OF AMERICAN'S WITH DISABILITIES ACT

Some of the requirements in the Construction Documents are meant to comply with the American's with Disabilities Act ("ADA"). The requirements of the ADA are technical in nature and may appear to be minor in nature (i.e. whether a walkway or ramp has a 2% cross-slope). Contractor is warned that even the slightest deviation from the specific requirements from the ADA is considered a Civil Rights Violation and subjects the District to fines of three times actual damages sustained by a handicap individual or up to \$4,000 per violation and attorney's fees required to enforce the ADA violation. As a result of the significant liability and exposure associated with ADA aspects of the Contract, Contractor shall take special care to meet all ADA requirements detailed in the Construction Documents. Failure to comply with ADA rules that results in a Notice of Non-Compliance shall be repaired to meet ADA requirements promptly. In addition, any ADA violations that are not identified by Inspector or Architect that are later identified shall be repaired and charged back to the Contractor through a Deductive Change Order.

41.1 <u>Indemnification of ADA Claims.</u> ADA claims arising from failure to comply with Construction Documents shall be indemnified, held harmless and defended by Contractor. Further, any withholdings for ADA violations in Article 29.4 shall include potential redesign costs and an accelerated repair costs due to the potential for ADA claims arising from DSA posting of ADA violations on the Project.

#### 42. PATENTS, ROYALTIES, AND INDEMNITIES

The Contractor shall hold and save the District and its officers, agents, and employees harmless from liability of any nature or kind, including cost and expense, for or on account of any patented or unpatented invention, process, article, or appliance manufactured or used in the performance of this Construction Services Agreement, including its use by the District, except to the extent a method or means was specifically required by the Contract Documents.

# 43. EXCISE TAX

If under federal excise tax law any transaction hereunder constitutes a sale on which a federal excise tax is imposed and the sale is exempt from such excise tax because it is a sale to a state or local government for its exclusive use, the District, upon request, will execute a certificate of exemption which will certify (1) that the District is a political subdivision of the state for the purposes of such exemption and (2) that the sale is for the exclusive use of the District. No excise tax for such materials shall be included in the GMP.

## 44. PROHIBITED INTERESTS

No official of District and no District representative who is authorized in such capacity and on behalf of District to negotiate, make, accept, or approve, or to take part in negotiating, making, accepting or approving any architectural, engineering, inspection, construction or material supply contract or any subcontract in connection with construction of Project, shall be or become directly or indirectly interested financially in this Construction Services Agreement or any part thereof. No officer, employee, architect, attorney, engineer or inspector of or for District who is authorized in such capacity and on behalf of District to exercise any executive, supervisory or other similar functions in connection with construction of Project, shall become directly or indirectly interested financially in this Construction Services Agreement or in any part thereof.



# 45. COMPLIANCE WITH DTSC GUIDELINES – IMPORTED SOIL/SOILS INSPECTION

- 45.1 If the Project requires the use of imported soils, the Contractor shall be responsible to use and shall certify that the imported material it uses is free of any hazardous and/or toxic substance or material of any nature or type as defined in accordance with California Law and the California Health and Safety Code. The District reserves the right to reject any imported material that has come from agricultural or commercial land uses. Contractor must notify the District of the source of material and comply with the applicable Regional Water Quality Control Board Resolution and when applicable, with the guidelines of the Department of Toxic Substances Control (DTSC).
- 45.2 Unless otherwise provided, when a soils investigation report obtained from test holes at the site is available, such report shall not be a part of this contract. Nevertheless, with respect to any such soils investigation and/or geotechnical report regarding the site, it shall be the responsibility of the Contractor to review and be familiar with such report. Any information obtained from such report or any information given on drawings as to subsurface soil condition or to elevations of existing grades or elevations of underlying rock is approximate only, is not guaranteed, and does not form a part of the contract, unless otherwise specifically provided. Contractor is required to make a visual examination of site and must make whatever tests it deems appropriate to determine the underground condition of the soil. Limited soil tests and subsurface investigations, if any, are available for review and consideration by Contractor and were conducted for the purpose of design only. Subsurface investigation information is made available by District solely as a matter of convenience and general information for Contractor and Contractor is expected to review and be familiar with such information. No representation is made by District or Architect that information provided is completely representative of all conditions and materials which may be encountered. If such a report is referenced in the Contract Documents for performance of the Project, such reference shall be to establish minimum requirements only. Further, no representation is made by District or Architect that information provided is solely adequate for purposes of construction. District disclaims responsibility for interpretations by Contractor of soil and subsurface investigation information, such as in protecting soil-bearing values, rock profiles, presence and scope of boulders and cobbles, soil stability and the presence, level and extent of underground water. Contractor shall determine means, methods, techniques and sequences necessary to achieve required characteristics of completed Work. Conditions found after execution of the Construction Services Agreement to be materially different from those reported and which are not customarily encountered in the geographic area of the Project hall be governed by provisions of this Construction Services Agreement for unforeseen conditions.

# 46. HAZARDOUS WASTE AND UNKNOWN PHYSICAL CONDITIONS

Contractor shall promptly, and before the following conditions are disturbed, notify the District, in writing, of any:

- 1. Material that Contractor believes may be material that is hazardous waste, as defined in Section 25117 of the Health and Safety Code that is required to be removed to a Class I, Class II, or Class III disposal site in accordance with provisions of existing law.
- 2. Subsurface or latent physical conditions at the Site differing from those indicated, including geological, soils, and or water table issues which impede construction or increase Construction Costs.
- Unknown physical conditions at the Site (not including structures or improvements) of any
  unusual nature, different materially from those ordinarily encountered and generally
  recognized as inherent in work of the character provided for in the Construction Services
  Agreement.
- 46.1 District shall promptly investigate the conditions, and if it finds that the conditions materially so differ, and the materials that are not on reports or documents supplied or reviewed as part of



- Contractor's Due Diligence shall be submitted as a Change Order under Article 17 and, upon approval, shall be allocated to the Unforeseen Allowance.
- 46.2 In the event that a dispute arises between District and Contractor whether the conditions materially differ from Due Diligence Documents reviewed for hazardous substances, or cause a decrease or increase in Contractor's cost of, or time required for, performance of any part of the work, Contractor shall not be excused from any scheduled completion date provided for by this Construction Services Agreement but shall proceed with all work to be performed under the Construction Services Agreement.

# 47. NO ASBESTOS CERTIFICATION

- 47.1 <u>Asbestos Free Installation Certification:</u> Contractor shall execute and submit an "Asbestos Free Materials Certification," and further, is aware of the following
  - 47.1.1 Should asbestos containing materials be installed by the Contractor in violation of this certification, or if removal of asbestos containing materials is part of the Project, decontaminations and removals will be performed in accordance with the requirements of all applicable laws and will meet the following criteria:
    - 47.1.1.1 Decontamination and removal of work found to contain asbestos or work installed with asbestos containing equipment shall be done only under the supervision of a qualified consultant, knowledgeable in the field of asbestos abatement and accredited by the Environmental Protection Agency (EPA).
    - 47.1.1.2 The asbestos removal contractor shall be an EPA accredited contractor qualified in the removal of asbestos and shall be chosen and approved by the asbestos consultant who shall have sole discretion and final determination in this matter.
    - 47.1.1.3 The asbestos consultant shall be chosen and approved by the District which shall have sole discretion and final determination in this matter.
    - 47.1.1.4 The work will not be accepted until asbestos contamination is reduced to levels deemed acceptable by the asbestos consultant.
  - 47.1.2 If removal of asbestos containing materials is part of the Project, the cost of all asbestos removal, including, but not necessarily limited to the cost of the asbestos removal contractor, the cost of the asbestos consultant, analytical and laboratory fees, time delays and additional costs that may be incurred by the District shall be borne entirely by the Contractor.
  - 47.1.3 Hold Harmless: Interface of work for the Project with work containing asbestos shall be executed by the Contractor at his/her risk and at his/her discretion with full knowledge of the currently accepted standards, hazards, risks and liabilities associated with asbestos work and asbestos containing products. By execution of the Construction Services Agreement the Contractor acknowledges the above and agrees to the fullest extent permitted by law to hold harmless the District, its Board and each member of the Board, its officers, employees, agents, representatives, including its Architect and assigns, for all asbestos liability which may be associated with this work. The Contractor further agrees to instruct his/her employees with respect to the above mentioned standards, hazards, risk and liabilities.



# 48. <u>LAWS AND REGULATIONS</u>

Contractor shall give all notices and comply with all laws, ordinances, rules, and regulations bearing on conduct of work as indicated and specified. If Contractor observes that drawings and specifications are at variance therewith, it shall promptly notify Architect in writing and any necessary changes shall be adjusted as provided in this Construction Services Agreement for changes in work. If Contractor performs any work knowing it to be contrary to such laws, ordinances, rules and regulations, and without such notice to the District's Architect, it shall bear all costs arising therefrom.

# 49. <u>AGREEMENT MODIFICATIONS</u>

No waiver, alteration or modification of any of the provisions of this Construction Services Agreement shall be binding upon either District or Contractor unless the same shall be in writing and signed by both District and Contractor.

# 50. NOTICES

All communications in writing between District and Contractor, including without limitation, applications for payment, shall be deemed to have been received by the addressee if delivered to the person for whom they are intended or if sent by registered mail, return receipt requested, or by telex, telegram, or fax followed by regular mail, addressed pursuant to the Notice Section of Article 3.

## 51. THIRD-PARTY CLAIMS

Pursuant to Public Contract Code section 9201(b) and (c), District shall provide Contractor with timely notification of the receipt of any third-party claim, relating to the Contract. District is entitled to recover its reasonable costs incurred in providing such notification.

## 52. ASSIGNMENT

Except Contractor's responsibility to assign Subcontractors and material suppliers to District upon Project Completion and the running of the Warranty Period, Contractor shall not assign or sublet the Lease, Sublease or this Construction Services Agreement, nor shall Contractor assign any monies due or to become due to it hereunder. Contractor has unique abilities and understanding of the Project from negotiations and the Due Diligence that has been undertaken and, thus, any assignment will not transfer to the assignee the specific understanding associated with Contractor on this Project.

## 53. HEADINGS

The headings herein contained are inserted only as a matter of convenience and reference and are not meant to define, limit or describe the scope or intent of the Contract Documents or in any way to affect the terms and provisions set forth herein.

# 54. <u>INTEGRATION/MODIFICATION</u>

This Construction Services Agreement represents the entire understanding of District and Contractor as to those matters contained herein, and supersedes and cancels any prior oral or written understanding, promises or representations with respect to those matters covered herein, and it shall not be amended, altered or changed except by a written agreement signed by the parties hereto.

## 55. APPLICABLE LAW/ PROVISIONS REQUIRED BY LAW DEEMED INSERTED

The terms and provisions of this Construction Services Agreement shall be construed in accordance with the laws of the State of California. If any action is brought in a court of law to enforce any term of this Construction Services Agreement the action shall be brought in a state court situated in the County where the District is located, State of California, unless a court finds jurisdiction or venue is only proper in a federal court, or a court outside this county.



In the event of any such litigation between the parties, the parties shall pay for their respective costs incurred, including attorneys' fees.

Each and every provision of law and clause required by law to be inserted in this Construction Services Agreement shall be deemed to be inserted herein and the Construction Services Agreement shall be read and enforced as though it were included herein, and if through mistake or otherwise any such provision is not inserted, or is not correctly inserted, then upon application of either party, the Construction Services Agreement shall forthwith be physically amended to make such insertion or correction.

# 56. <u>SUCCESSION OF RIGHTS AND OBLIGATIONS</u>

All rights and obligations under this Construction Services Agreement shall inure to and be binding upon the successors and assigns of the parties hereto.

IN WITNESS WHEREOF, the parties hereto have, by their duly authorized representatives, executed this Master Construction Services Agreement, in duplicate, as of the day and year first above written.

CONTRACTOR	DISTRICT:		
	SAN BERNARDINO CITY UNIFIED SCHOOL DISTRICT		
By:Name Title	By:		
DATE	DATE:		



#### **EXHIBIT "A"**

# SCOPE OF WORK / CONSTRUCTION DOCUMENTS

#### Exhibit A-1

# PRECONSTRUCTION SERVICES SCOPE OF WORK

Contractor will provide consulting services to the District with respect to reviewing the Plans and Specifications to identify and call out all deficiencies, incongruities and inconsistencies that may affect constructability of the Project, including but not limited to design and specification omissions, incomplete and/or inconsistent plans, details and specifications, and any lack of coordination, and scheduling, pricing, and phasing, together with all other appropriate, necessary and/or required services in accordance with the applicable standard of care, excluding only responsibility for the professional negligence of any licensed engineer or architect in the preparation of the Plans and Specifications ("Services" or "Preconstruction Services") to facilitate, and in preparation for, the successful development and construction of the Project.

In performing these Preconstruction Services, Contractor acknowledges and agrees that no work for which a contractor is required to be licensed in accordance with Business & Professions Code section 7065, et seq., will be performed on this Project until such time that the Division of State Architect has approved the Project's plans and specifications.

The parties further acknowledge that, after the completion of preconstruction services and prior to start of construction, Contractor and District will negotiate a guaranteed maximum price for the Project along with the contingencies amounts, performance period, and liquidated damages rate in Section 3 of the Construction Services Agreement.

## **ARTICLE 1 DEFINITIONS**

- 1.1 <u>DEFINITIONS</u>. As used in this Preconstruction Services Scope of Work, the following terms shall have the meanings specified herein unless the context requires otherwise.
- 1.1.1 "Architect" shall mean the Architect of Record for the design of the Project, currently \_\_\_\_\_\_, or any successor architect of record approved and appointed by the Board for the design of the Project.
- 1.1.2 "Board" shall mean the Board of Education of the San Bernardino City Unified School School District.
- 1.1.3 "Construction Budget" shall mean the amount of money that the District has allocated for all construction.
- 1.1.4 "Construction Cost" shall mean the cost to perform all Work pursuant to the Construction Documents.
- 1.1.5 "Construction Documents" shall mean those documents which are required for the actual construction of the Project as accepted and approved by DSA and the District's Governing Board, including not limited to the complete final working drawings and specifications setting forth in detail the work to



be done and the materials, workmanship, finishes and equipment required, as well as all related correspondence providing additional direction as to the design intent, including RFIs, reviewed submittals, CCDs, change orders, etc.

- 1.1.6 "Consultant" includes an architect, engineer, planner, landscape architect, inspector or other professional/advisor with whom the District contracts with directly or indirectly to perform Project-related services.
- 1.1.7 "Contractor" shall mean the licensed Contractor performing the professional services under this Preconstruction Services Scope of Work, as authorized by Government Code sections Government Code 4525, 4526, 4529.5.
  - 1.1.8 "Day" shall mean a calendar day unless otherwise specifically designated.
- 1.1.9 "District Representative" shall mean \_\_\_\_\_\_, and any successor appointed by District.
- 1.1.10 "DSA Laws and Regulations" shall mean, in connection with each construction phase, the laws and regulations relating to the jurisdiction and authority of the Division of the State Architect in effect at the time construction is approved and the applicable permits, if any, are obtained, including, without limitation, the Field Act, Education Code sections 17280 et seq., and the California Disabled Access Law, Government Code sections 4450, et seq., along with all related laws, regulations rules and policies.
- 1.1.11 "Educational Specifications" shall mean the District's approved educational specifications for school facility construction and incorporated herein by this reference and approved by the Board.
- 1.1.12 "General Conditions" shall mean the agreed upon overhead, temporary utilities, trailers, equipment and other on site and off site costs borne by the Contractor during Construction Phase of the Project.
- 1.1.13 "GMP" shall mean the Guaranteed Maximum Price" as that term is defined by State law for purposes of the Lease Leaseback delivery method of public school construction.
  - 1.1.14 "IOR" shall mean Inspector of Record for the Project.
- 1.1.15 "Master Project Schedule" shall mean the Project schedule and any Master Project Schedule presented to, and approved by, the Board at a later date.
- 1.1.16 "Program Manager" shall mean \_\_\_\_\_\_, or any successor approved and appointed by the Board to provide program management services.
- 1.1.17 "Project" shall mean the preconstruction and construction of the facilities that will comprise the project named on page 1 of this Preconstruction Services Scope of Work.
- 1.1.18 "Project Budget" shall mean the budget for the Project, prepared and revised by the Program Manager and the Contractor and approved by District during the preconstruction phase and approved by the Board.
  - 1.1.19 "Reimbursable Expenses" shall mean, any item of expense approved by the District as a



reimbursable expense in connection with this Preconstruction Services Scope of Work and as detailed in Project Budget, Exhibit B.

1.1.20 "Work" shall mean all the construction, work, labor, materials, machinery, equipment, tools, supplies, services and other items that the Contractor is to perform or provide in connection with the Project pursuant to the Construction Documents.

# ARTICLE 2 PRECONSTRUCTION CONTRACTOR BASIC SERVICES AND RESPONSIBILITIES

Contractor represents to the District that: (i) it has previously acted as a Contractor; (ii) it has the necessary license(s) required by law for the Services set forth in this Preconstruction Services Scope of Work, (License No. \_\_\_\_\_\_); and (iii) it has expertise and experience in constructability reviews, cost estimating, value engineering, construction supervision, bid preparation, evaluation of construction projects, project scheduling, cost benefit analysis, claims review and negotiation, and general management and administration of construction projects.

Contractor covenants to provide its best skill and judgment in furthering the interests of the District in the performance of its obligations under this Preconstruction Services Scope of Work. Contractor agrees to furnish efficient business administration and management services and to perform in an expeditious and economical manner consistent with the best interests of the District. Contractor shall provide all services with respect to the Project as set forth in this Preconstruction Services Scope of Work and the attached exhibits (the "Services").

It is understood and agreed that time is of the essence in connection with the funding plan and the design and construction of the Project and Contractor agrees to use its best efforts to ensure that the Project is submitted to DSA for approval as soon as reasonably possible in conjunction with the Architect.

Unless directed otherwise by the District, the District's Representative, and/or the Program Manager, the Contractor shall direct all communication, correspondence, and other interactions with the District through the Program Manager, including communication with the District's personnel, the Architect, the District's consultants, and any other agencies, organizations, or outside entities.

Contractor further agrees that the personnel it named for the specified positions in its proposal and assigned to the District constituted a material basis upon which it received its best value determination, and that in the event Contractor replaces any of the named personnel with a substitute individual, Contractor shall pay to District, as liquidated damages and not as a penalty, the sum of \$20,000.00 per substitution. Contractor and District agree that if Contractor replaces any named personnel with a substitute individual, District's damages would be extremely difficult or impracticable to determine and that the aforesaid amount is a reasonable estimate of and reasonable sum for such damages for District's staff time to train and bring on board Contractor's new individual. District may deduct any liquidated damages due from Contractor from any amounts otherwise due to Contractor under this Preconstruction Services Scope of Work. This provision shall not limit any right or remedy of District in the event of any other default of Contractor other than personnel substitutions.

2.1 <u>BASIC SERVICES</u>. The Basic Services shall include project design review and evaluation, planning for construction mobilization and supervision, construction cost estimating and analysis, project scheduling, and cost-benefit analysis, including, but not limited to, the tasks identified below.



- 2.1.1 Contractor shall communicate and coordinate with the District and the Architect to ascertain the requirements of the Project and shall arrive at a mutual understanding of such requirements with the District.
- 2.1.2 Contractor shall provide a preliminary evaluation of the District's schedule and Construction Budget, each in terms of the other.
- 2.1.3 The Architect's agreement with the District may include numerous phases of services described in such agreement. During the Architect's services, Contractor shall coordinate with the Architect as necessary to deliver the Services and support the schematic design, design development, construction documents, DSA submittal development and approval, and bid preparation, administration, review of bids, and development of proposed guaranteed maximum price ("GMP").
- 2.1.4 Contractor shall perform the Preconstruction Services as defined in the Recitals and further detailed in this Article 2 in accordance with the applicable standard of care for a licensed contractor, excluding only responsibility for the professional negligence of any licensed engineer or architect in the preparation of the Plans and Specifications:
- (1) Perform an ongoing review of the Architect's programming plan including the size of space, proposed finishes, ceiling heights, building height, exterior finishes, circulation spaces, any necessary ancillary spaces, and any anticipated site work;
- (i) Contractor shall submit to the Program Manager, at each document review phase, an analysis of the Architect's program in comparison to the District's approved Educational Specifications, including quantified cost and time impacts associated with each variance.
- (2) Perform an ongoing analyses and review of the Construction Documents during their development and advise and make recommendations on proposed site use and improvements, facility improvements, selection of materials, building systems and equipment, constructability reviews, value engineering and related quality assurance/quality control consulting, scheduling, and methods of project delivery.
- (3) Contractor shall advise and provide recommendations on relative feasibility of construction methods, availability of materials and labor, time requirements for procurement, installation and construction, and factors related to construction cost and scheduling including, but not limited to, costs of alternative designs or materials, preliminary budgets, and possible economies;
- (4) Regularly revise and update a Project Scope of Work document in coordination with the Architect to:
- (i) Identify, quantify, and delineate the trade-specific scopes of work, how they are separate from each other, and where coordination is required to deliver a complete system for all components of the Project Scope of Work,
- (ii) Identify potential scope gaps, or scope overlaps between trades and present such findings to the Architect and the Program Manager in a timely manner for review and consideration,



- (iii) Identify long lead procurement items and approval activities required for each trade's scope of work, and
- (iv) Identify submittal requirements, agency approvals, permit requirements, licensing requirements, and any other necessary items that are required for timely completion of each trade's scope of work;
- (v) Ensure that all Construction Documents submitted to DSA shall be constructible by a competent general building contractor duly licensed by the State of California, without need for any Requests for Information, Supplemental Instructions, Change Orders or similar inquiries or changes in order to complete construction of the full Scope of Work within a Construction Cost, including all contingencies and allowances, not to exceed 90% of the Construction Budget and to form the basis of the Guaranteed Maximum Price for the Project.
- (5) Coordinate actively with the Architect to provide trade coordination input into the design process to ensure that all Construction Documents are fully coordinated and that all clashes and inconsistencies are identified and remedied through, or to the equivalent extent of Building Information Management clash detection analysis;
- (6) Perform ongoing and accurate Construction Cost estimating to confirm that cost to perform the Work does not exceed the Construction Budget, including regular reconciliation reports between Architect's and Contractor's cost estimates, including square foot pricing at schematics, detailed line item quantities and costs at conceptual design, and regular cost estimate updates at design development, construction documents, DSA submittal, bid set and further phases as needed;
- Prepare an ongoing and accurate, and periodically update, Master (7) Project Schedule for the Architect's review and the District's acceptance showing major construction milestones including but not limited to: start of construction, mobilization, demolition, abatement, site work, foundations, structure, mechanical/electrical/plumbing/fire sprinkler (MEPF) systems, building envelope, exterior finishes, interior finishes, landscaping/hardscaping, final inspection, and acceptance by the District. The Master Project Schedule must include the following information: detailed work activities properly sequenced for trade coordination planning as needed to ensure that the Project can be completed within the allotted construction schedule, long lead items are identified, curing times are identified, procurement schedule requirements are defined, submittal schedule requirements are defined, and other timeline and schedule planning as necessary to ensure that the Project can be constructed within the allotted timeframe. Contractor coordinate and collaborate with the Architect as necessary to prepare, and shall prepare accordingly the portion of the preliminary Project schedule relating to the performance of the Architect's services in accordance with the Architect's agreement(s) with the District. In the Master Project Schedule, Contractor shall coordinate and integrate Contractor's Services, the Architect's services, the construction of the Project, the District's responsibilities, inspection requirements, document review periods, and all other activities required for Project completion, highlighting critical and long-lead-time items;
- (8) Develop a list of recommended contingencies, allowances, and estimated escalation;
- (9) Develop proposed General Conditions and all proposed markups including but not limited to: fee, insurance, and bonding. Develop site logistics and safety plan showing laydown areas, construction traffic flow and construction personnel parking;



- (10) Develop a complete list of bid alternates, and proposed bid list of trade contractors as well as criteria for trade contractors prequalification, exercising all due diligence to obtain at least five (5) trade contractors per trade required for major trades, and three (3) trade contractors per trade required for minor trades, and in the event that Contractor is unable, despite the exercise of due diligence, to obtain the minimum number of trade contractor bids required, present to District the record of due diligence which District shall not unreasonably reject as adequate in lieu of obtaining the minimum number of trade contractor bids required;
- (11) Develop proposed GMP with full detail, bid results, and notes, including bid alternates and associated pricing.
- 2.1.5 Further, Contractor shall provide ongoing advice to the District and the Architect in a team effort to assure that the Project is delivered on time and on budget. To provide such ongoing support and consulting, the Contractor shall:
- (1) Participate in Project progress meetings, as scheduled by the Program Manager, with Architect and Program Manager to provide ongoing updates of status of items set forth in 2.1.4 above, and to discuss any and all issues that arise that may affect the Project;
- (2) Prepare a monthly progress report and provide weekly updates as needed to include, but not be limited to, the following information:
- (i) Status of all required deliverables in progress, and required within 4 weeks of date of report,
  - (ii) Design intent and scope questions,
  - (iii) Programming status,
  - (iv) Coordination reviews,
  - (v) Regulatory and agency review updates,
  - (vi) Progress on any required studies and deliverables,
  - (vii) Contract administration,
  - (viii) Budget and value engineering, and
  - (ix) Schedule status;
- (3) Provide support to the Program Manager as requested and or required to provide accurate and complete monthly updates to the Board and the Citizen's Bond Oversight Committee, including but not limited to (i) attending meetings with Program Manager, (ii) preparing reports and presentations to demonstrate project progress, (iii) coordinating with Architect and Consultant to ensure complete and accurate information is provided at all times to the Board and Citizens' Bond Oversight Committee.



- 2.1.6 Following the District's approval of each phase of the development of Construction Documents, Contractor shall update and submit the latest estimate of the Construction Cost and the Master Project Schedule, and all other Phase Deliverables.
- 2.2 <u>ADDITIONAL PRECONSTRUCTION SERVICES</u>. Services in addition to those set forth in this Preconstruction Services Scope of Work will require written request or pre-authorization in writing by the District following specific approval of such services by the Board if the fee for the additional services causes an increase to the Preconstruction Services Fee in Section 5.1 of this Preconstruction Services Scope of Work. It is understood and agreed that Contractor shall not perform any services in addition to those set forth in this Preconstruction Services Scope of Work unless and until Contractor receives specific written approval for such additional services from the Board. It is understood and agreed that if Contractor performs services in addition to those set forth in this Preconstruction Services Scope of Work without receiving prior written approval from the Board, Contractor shall not be paid for such services.

## 2.3 TIME

- 2.3.1 Contractor shall perform the Services set forth in this Preconstruction Services Scope of Work as expeditiously as is consistent with reasonable skill and care and the orderly progress of the Project. Time is of the essence in connection with the Project and with all of Contractor's Services.
- 2.3.2 Contractor shall be entitled to an extension of time for the time of completion for delay which may arise due to an act of God, such as an earthquake, flood or fire, or an act of a public enemy or act of war, if such act results in delays on any approvals necessary for completion of the Project, but Contractor shall have no claim for any other compensation for such delay.
- 2.3.3 Should the schedule for the construction of the Project be extended due to an added scope of work as directed by the District and approved by the Board or an extension of the schedule related to governmental agency approvals necessary for completion of the Project, the time for performance under this Preconstruction Services Scope of Work shall be extended and Contractor shall be compensated for this extension as mutually agreed by the parties.

# **ARTICLE 3** THE DISTRICT'S RESPONSIBILITIES

- 3.1 The District shall provide all information actually known to District, without obligation or duty to undertake any investigation, research, inspection, inquiry, regarding the requirements of the Project including the District's objectives, constraints and criteria.
- 3.2 The District shall designate a District Representative to act on the District's behalf with respect to the Project. The District, or the District Representative, if authorized, shall render decisions promptly to avoid unreasonable delay in the progress of Contractor's Services.
- 3.3 The District shall furnish tests, inspections and reports as required by law or the Construction Documents.
- 3.4 If the District observes or otherwise becomes aware of any fault or defect in the Project, or nonconformance with the Construction Documents, prompt notice thereof shall be given by the District to Contractor. District has no obligation or duty to undertake any investigation, research, inspection, inquiry or other steps to discover any fault or defect in the Project, or nonconformance with the



Construction Documents, but only the obligation to inform Contractor of any specific fault, defect or non-conformance of which the District actually becomes aware.

- 3.5 The District reserves all rights regarding the Project and any development, progress or work thereon, including the right to cease any or all work on or related to the Project, the right to perform work related to the Project with the District's own forces and/or whether to award any contracts to any person or entity in connection with the Project. Contractor understands and acknowledges that this Preconstruction Services Scope of Work contains no promise to enter into or negotiate any further agreement, work or engagement with or for District by and between the District and Contractor.
- 3.6 The District shall retain the Architect whose services, duties and responsibilities are described in the agreement between the District and the Architect. The District-Architect agreement shall be furnished to Contractor.

# **ARTICLE 4 CONSTRUCTION COST**

- 4.1 Construction Cost shall not include the compensation of Contractor for the Services performed under this Preconstruction Services Scope of Work, nor all services of the Architect and Consultant, the cost of land, rights-of-way and other costs that are the responsibility of the District.
- 4.2 Contractor shall consult with the Architect and the District to suggest reasonable adjustments in the scope of the Project, and to suggest alternate bids in the Construction Documents to adjust the Construction Costs so that it does not exceed the allowable Construction Budget indicated in the attached Exhibit "B".
- 4.3 Contractor shall provide for the District's review and acceptance, a monthly report showing the status of the Project. With the District's assistance, and in accordance with District procedures, Contractor shall provide all construction related Board agenda items. Examples: change orders, notices to proceed, notice of completion, authorization to bid, award of Prime Contracts, etc.

## ARTICLE 5 BASIS OF COMPENSATION AND PAYMENT

Reimbursable expenses, other than Approved Charges, as designated in Exhibit "B," are included in the Preconstruction Services Fee. Approved Charges, as designated in Exhibit "B," shall be reimbursed by the District as described in this Article 5.

5.2 <u>METHOD OF PAYMENT</u>. Contractor shall submit for the District's approval a proposed Schedule of Values ("SOV") within 14 days of receipt of the executed Lease-Leaseback Agreement, indicating the Contractor's distribution of the Preconstruction Services Fee among the various Services for use in determining the billable amounts to be invoiced by the Contractor to the District in accordance with



Exhibit B to this Preconstruction Services Scope of Work. The District approval of the SOV shall not be unreasonably withheld.

5.3 <u>INVOICING FOR SERVICES</u>. Following completion of the Services applicable to each phase set forth in the SOV, or agreement by the District to consider an interim invoice, Contractor shall submit an invoice in form and substance satisfactory to the District in an amount not to exceed the amount specified as the portion of the Preconstruction Services Fee to be paid for that phase set forth in the SOV for the Services identified in the invoice.

Contractor shall identify all reimbursable expenses or charges included in the invoice or request for payment as separate from the Preconstruction Services Fee line items, and provide a cumulative total of reimbursable expenses billed to date, current reimbursable amount billed, and remaining amount for reimbursable expenses as provided for in this Preconstruction Services Scope of Work. All reimbursable expenses shall be identified using the categories agreed upon by the parties. Requests for reimbursable expenses shall be limited to the categories of charges listed in contract documents, and any other categories of charges agreed to at a later date by the Board, and must be within the total amount allowable per this Preconstruction Services Scope of Work. A request for reimbursement of an Approved Charge is limited to the categories listed in contract documents and any other categories of charges agreed to at a later date by the Board.

Each invoice or request for payment shall also be accompanied by a certificate from Contractor to the effect that invoice or request for payment is a true and accurate reflection of the Services performed by Contractor and that the items for which compensation is requested have not been previously paid for or denied compensation by the District. Contractor shall use the District's Invoice Approval Form

5.4 <u>TIMING OF PAYMENT</u>. District shall pay Contractor for all undisputed amounts, which are approved by the District pursuant to this Preconstruction Services Scope of Work no later than thirty (30) calendar days from the date of receipt by the District of an approved invoice from Contractor. No invoicing or payments will be allowed for any work performed by the Contractor prior to Board approval of this Preconstruction Services Scope of Work. Any work performed prior to Board approval will be done at the Contractor's own risk and cost.

## ARTICLE 6 TERMINATION, ABANDONMENT OR SUSPENSION OF WORK

## 6.1 TERMINATION OF PRECONSTRUCTION CONTRACTOR SERVICES

6.1.1 The District may terminate all or any portion of this Preconstruction Services Scope of Work for cause in the event Contractor fails to promptly and efficiently perform the Services or otherwise fails to comply with the terms of this Preconstruction Services Scope of Work. The termination shall be effective if Contractor fails to cure such default within thirty (30) Days following issuance of written notice thereof by the District, or if the cure by its nature takes longer, fails to commence such cure within thirty (30) Days of issuance of the notice and diligently prosecute such cure to the satisfaction of the District. For termination of this Preconstruction Services Scope of Work by the District, cause shall mean the occurrence of any of the following events at any time:

(1) Contractor becomes insolvent or files for relief under the bankruptcy laws of the United States;



- (2) Contractor makes a general assignment for the benefit of its creditors or fails to pay its debts as the same become due;
  - (3) A receiver is appointed to take charge of Contractor's property;
- (4) Contractor has replaced an individual named in its proposal to provide services to the District without the District's prior written approval; or
- (5) Contractor is in default of any other material obligation under the Preconstruction Services Scope of Work.
- 6.1.2 Contractor may give thirty (30) Days written notice to the District of Contractor's intent to terminate this Preconstruction Services Scope of Work for the District's failure to perform its duties and responsibilities under this Preconstruction Services Scope of Work. If, after the expiration of such thirty (30) Days, the District fails to cure the performance as set forth in Contractor's notice of intent to terminate the Preconstruction Services Scope of Work, Contractor may issue a notice of termination. If the actions to be taken to cure the default would reasonably exceed thirty (30) Days and the District commences curing the default within said period of time, and thereafter continuously continues to cure the default, Contractor agrees not to suspend or terminate the Preconstruction Services Scope of Work until the District has had a reasonable opportunity to fully cure the default.
- 6.1.3 The District shall also have the right in its absolute discretion to terminate this Preconstruction Services Scope of Work without cause following forty-five (45) Days written notice from the District to Contractor.

# 6.2 CONTINUANCE OF WORK

In the event of a dispute between the parties as to performance of the Services by Contractor or the interpretation of this Preconstruction Services Scope of Work, or payment or nonpayment for Services performed or not performed, the parties shall attempt to resolve the dispute. The District and Contractor agree to seek, in good faith, a timely and equitable resolution of a dispute. All efforts will be made by both the District and Contractor to avoid any legal proceedings arising from a dispute.

However, pending resolution of a dispute, Contractor agrees to continue the Services diligently to completion and the District agrees to continue paying Contractor all undisputed compensation in accordance with Article 5. If the dispute is not resolved, Contractor agrees it shall neither terminate the Preconstruction Services Scope of Work nor stop the progress of its Services, but Contractor's sole remedy shall be to submit such controversy to determination by a court having competent jurisdiction of the dispute.

# 6.3 ABANDONMENT OF THE PROJECT

The District has the absolute discretion to suspend or abandon all or any portion of the work on the Project and may do so upon fourteen (14) Days' written notice to Contractor. Upon notice of suspension or abandonment, Contractor shall immediately discontinue any further action on the Project or the abandoned portion of the Project, as applicable. If the entire work to be performed on the Project is abandoned, the parties shall each be relieved of the remaining executory obligation of the Preconstruction Services Scope of Work, as it relates to the Project, but shall not be relieved of any obligations arising prior to said abandonment.

# 6.4 <u>COMPENSATION IN THE EVENT OF TERMINATION, ABANDONMENT OR SUSPENSION</u>



In the event the District terminates this Preconstruction Services Scope of Work for cause, abandons or suspends the work on the Project, there shall be due and payable within thirty (30) Days following such termination, abandonment or suspension a sum of money sufficient to increase the total amount paid to Contractor to an amount which bears the same proportion to the Preconstruction Services Fee as the amount of Services performed or provided by Contractor prior to the time of such termination, suspension or abandonment of this Preconstruction Services Scope of Work bears to the entire Services Contractor is required to perform pursuant to this Preconstruction Services Scope of Work.

In the event of termination due to a breach of this Preconstruction Services Scope of Work by Contractor, the compensation due Contractor upon termination shall be reduced by the amount of damages sustained by the District due to such breach.

# 6.5 DELIVERY/OWNERSHIP OF DOCUMENTS

Upon termination, abandonment or suspension, Contractor shall deliver to the District all documents and materials related to the Project. It is agreed that the District is the sole owner of all documents, schedules, reports, and any other materials concerning the Project.

# **ARTICLE 7 INDEMNIFICATION**

7.1 Contractor shall defend, indemnify, and hold harmless as required under Section 36 of the Construction Services Agreement.

# ARTICLE 8 SUCCESSORS, SUCCESSORS AND ASSIGNS

**RESERVED.** 

# ARTICLE 9 APPLICABLE LAW

RESERVED.

# ARTICLE 10 PRECONSTRUCTION CONTRACTOR NOT AN OFFICER OR EMPLOYEE OF DISTRICT

While engaged in carrying out and complying with the terms and conditions of this Preconstruction Services Scope of Work, Contractor is an independent construction management consultant and not an officer or employee of the District.

## ARTICLE 11 INSURANCE

- 11.1 Without in any way affecting the indemnity provided in or by Article 7 or Sections 35 and 36 of the Construction Services Agreement, before commencement of any Services, Contractor shall review Section 35 of the Master Construction Services Agreement for the onsite coverage through the District's OCIP program and the required offsite insurance requirements.
- 11.2 Additional Insurance. Contractor shall procure and maintain the types and amounts of coverage as follows:



- 11.2.1 Professional Liability Insurance, including contractual liability, with limits of \$1,000,000 per claim and \$2,000,000 in the aggregate. Such insurance shall be maintained during the term of this Preconstruction Services Scope of Work and renewed per 11.3.1 below and/or at rates consistent with the time of execution of the Construction Services Agreement adjusted for inflation.
- 11.2.2 Employer's Liability Insurance in an amount not less than \$1,000,000 per accident for bodily injury or disease.
- 11.2.1 The Contractor shall carry and maintain during the term of this AGREEMENT a policy with a limit of not less than \$1,000,000 Sexual Abuse Injury Limit of Insurance. The Contractor shall hold the DISTRICT harmless from any and all claims for injury, damage, and loss.

# 11.3 MINIMUM SCOPE OF INSURANCE

11.3.1 If the Professional Liability Insurance policy is written on a claims made basis, it shall be maintained continuously for a period of no less than three (3) years after a Notice of Completion is recorded for the Project to which it applies. The "retro date" must be shown and must be before the date of this Construction Services Agreement.



# Exhibit A-2 CONSTRUCTION SERVICES SCOPE OF WORK

(See RFP/Q, Construction Documents, Drawings, & Specifications)



# **Exhibit B - Project Budget**

(To be completed at a later date)



# **ATTACHMENT NO. 1 TO AGREEMENT**

# CERTIFICATE REGARDING WORKERS' COMPENSATION

Every employer, except the State, shall secure the payment of compensation in one or more of the following ways:

- 1. By being insured against liability to pay compensation in one or more insurers duly authorized to write compensation insurance in the State of California.
- 2. By securing from the Director of Industrial Relations, a certificate of consent to self-insure either as an individual employer, or as one employer in a group of employers, which may be given upon furnishing proof satisfactory to the Director of Industrial Relations of ability to self-insure and to pay any compensation that may become due to its employees.
- 3. For any county, city, city and county, municipal corporation, public district, public agency, or any political subdivision of the state, including each member of a pooling arrangement under a joint exercise of powers agreement (but not the state itself), by securing from the Director of Industrial Relations a certificate of consent to self-insure against workers' compensation claims, which certificate may be given upon furnishing proof satisfactory to the director of ability to administer workers' compensation claims properly, and to pay workers' compensation claims that may become due to its employees. On or before March 31, 1979, a political subdivision of the state which, on December 31, 1978, was uninsured for its liability to pay compensation, shall file a properly completed and executed application for a certificate of consent to self-insure against workers' compensation claims. The certificate shall be issued and be subject to the provisions of Section 3702.

I am aware of the provisions of Section 3700 of the Labor Code, which require every employer, including subcontractors, to be insured against liability for workers' compensation or to undertake self-insurance in accordance with the provisions of the code, and I will comply with such provisions before commencing the performance of the work of this Contract.

Signature	Date	
Printed Name	_	
Official Title	_	

In accordance with Article 5 (commencing at section 1860), Chapter 1, Part 7, Division 2 of the Labor Code, the above certificate must be signed and submitted with the Contractor's bid.

**CONTRACTOR** 



# **ATTACHMENT NO. 2 TO AGREEMENT**

# **INSURANCE DOCUMENTS & ENDORSEMENTS (Not Required for OCIP or Wrap Up Insurance)**

The following insurance endorsements and documents must be provided to the SAN BERNARDINO CITY UNIFIED SCHOOL DISTRICT within five (5) calendar days after receipt of notification of award. If the apparent low bidder fails to provide the documents required below, the District may award the contract to the next lowest responsible and responsive bidder or release all bidders, and the bidder's bid security will be forfeited. All insurance provided by the bidder shall fully comply with the requirements set forth in Article 11 of the General Conditions.

1. <u>General Liability Insurance</u>: Certificate of Insurance with all specific insurance coverages set forth in Article 11 of the General Conditions, proper Project description, designation of the District as the Certificate Holder, a statement that the insurance provided is primary to any insurance obtained by the District and minimum of 30 days' cancellation notice. Bidder shall also provide required additional insured endorsement(s) designating all parties required in Article 11 of the General Conditions. The additional insured endorsement shall be an ISO CG 20 10 (11/85), or an ISO CG 20 10 (10/93 or 07/04) and ISO CG 20 37 (10/93 or 07/04) or their equivalent as determined by the District in its sole discretion.

Incidents and claims are to be reported to the insurer at:

(Title)		(Department)
(Company)		
(Street Address)		
(City)	(State)	(Zip Code)
() (Telephone Number)		

- Workers' Compensation/ Employer's Liability Insurance: Certificate of Workers' Compensation Insurance meeting the coverages and requirements set forth in Article 11 of the General Conditions, minimum of 30 days' cancellation notice, proper Project description, waiver of subrogation and any applicable endorsements.
- Automobile Liability Insurance: Certificate of Automobile Insurance meeting the coverages and requirements set forth in Article 11 of the General Conditions, minimum 30 days' cancellation notice, any applicable endorsements and a statement that the insurance provided is primary to any insurance obtained by the District.
- 4. Sexual Abuse/Molestation The CONTRACTOR shall carry and maintain during the term of this AGREEMENT a policy with a limit of not less than \$1,000,000 Sexual Abuse Injury Limit of Insurance. The CONTRACTOR shall hold the DISTRICT harmless from any and all claims for injury, damage, and loss.



# Incidents and claims are to be reported to the insurer at:

	Attn:			
		(Title)		(Department)
		(Company)		
		(Street Address)		
		(City)	(State)	(Zip Code)
		() (Telephone Number)		
DATE:				
DATE			CONTRACTOR	
			Ву:	
			Cianatura	
			Signature	

Page **103** 



# ATTACHMENT NO. 3 TO AGREEMENT

# CONTRACTOR'S CERTIFICATE REGARDING DRUG-FREE WORKPLACE

This Drug-Free Workplace Certification form is required from all successful bidders pursuant to the requirements mandated by Government Code Sections 8350 et seq., the Drug-Free Workplace Act of 1990. The Drug-Free Workplace Act of 1990 requires that every person or organization awarded a contract or grant for the procurement of any property or service from any State agency must certify that it will provide a drug-free workplace by performing certain specified acts. In addition, the Act provides that each contract or grant awarded by a State agency may be subject to suspension of payments or termination of the contract or grant, and the CONTRACTOR or grantee may be subject to debarment from future contracting, if the contracting agency determines that specified acts have occurred.

Pursuant to Government Code Section 8355, every person or organization awarded a contract or grant from a State agency shall certify that it will provide a drug-free workplace by doing all of the following:

- Publishing a statement, notifying employees that the unlawful manufacture, distribution, dispensation, possession, or use of a controlled substance is prohibited in the person's or organization's workplace, and specifying actions which will be taken against employees for violations of the prohibition;
- 2. Establishing a drug-free awareness program to inform employees about all of the following:
  - a. The dangers of drug abuse in the workplace;
  - b. The person's or organization's policy of maintaining a drug-free workplace;
  - c. The availability of drug counseling, rehabilitation and employee-assistance programs; and
  - d. The penalties that may be imposed upon employees for drug abuse violations;
- 3. Requiring that each employee engaged in the performance of the contract or grant be given a copy of the statement required by subdivision (a) and that, as a condition of employment on the contract or grant, the employee agrees to abide by the terms of the statement.

I, the undersigned, agree to fulfill the terms and requirements of Government Code Section 8355 listed above and will (a) publish a statement notifying employees concerning the prohibition of controlled substance at the workplace, (b) establish a drug-free awareness program, and (c) require each employee engaged in the performance of the contact be given a copy of the statement required by section 8355(a) and require such employee agree to abide by the terms of that statement.

I also understand that if the SAN BERNARDINO CITY UNIFIED SCHOOL DISTRICT determines that I have either (a) made a false certification herein, or (b) violated this certification by failing to carry out the requirements of Section 8355, that the contract awarded herein is subject to termination, suspension of payments, or both. I further understand that, should I violate the terms of the Drug-Free Workplace Act of 1990, I may be subject to debarment in accordance with the requirements of Sections 8350 et. seq.

I acknowledge that I am aware of the provisions of Government Code Sections 8350 et. seq. and hereby certify that I will adhere to the requirements of the Drug-Free Workplace Act of 1990.

DATE:	Contractor:	Signature:



# **ATTACHMENT NO. 4 TO AGREEMENT**

# CONTRACTOR'S CERTIFICATE REGARDING ALCOHOLIC BEVERAGE and TOBACCO-FREE CAMPUS POLICY

The CONTRACTOR agrees that it will abide by and implement the DISTRICT's Alcoholic Beverage and Tobacco-Free Campus Policy (SAN BERNARDINO CITY UNIFIED SCHOOL DISTRICT Policy No. 4119.14), which prohibits the use of alcoholic beverages and tobacco products, of any kind and at any time, on DISTRICT-owned or leased buildings, on DISTRICT property and in DISTRICT vehicles. The CONTRACTOR shall procure signs stating "ALCOHOLIC BEVERAGE AND TOBACCO USE IS PROHIBITED" and shall ensure that these signs are prominently displayed in all entrances to school property at all times.

DATE:	Contractor:	Signature:



# **ATTACHMENT NO. 5 TO AGREEMENT**

# DISABLED VETERAN BUSINESS ENTERPRISE (DVBE) PARTICIPATION STATEMENT

Each bidder must complete this form in order to comply with the San Bernardino Unified School District ("District") policy for participation of disabled veteran business enterprises (School Din projects funded in whole or in part by the State of California pursuant to the Leroy F. Greene School Facilities Act of 1998. (Education Code §17070.10, et seq.)

Project Name:
Bid No.:
OSA No.:
The undersigned, on behalf of the Contractor named below, certifies that the Contractor has made easonable efforts to secure participation by DVBE in the contract to be awarded for the above-referenced Bid No., including participation by DVBE subcontractors and/or material suppliers. <b>Check only one of the ollowing</b> :
The Contractor was unable after reasonable efforts to secure DVBE participation in the contract for the above-referenced Project/Bid No. However, the Contractor will use DVBE services if the opportunity arises at any time during construction of the Project. Upon completion of the Project, the Contractor will report to the District the total dollar amount of DVBE participation in any contract awarded to Contractor, and in any change orders, for the above-referenced Project/Bid No.  The Contractor has secured DVBE participation in the contract for the above referenced Project/Bid No., and anticipates that such DVBE participation will equal approximately
the Project, Contractor will report to the District the actual total dollar amount of DVBE participation in the contract awarded to Contractor, and in any change orders, for such Bid No.
Company:
Name:
Title:
Signature:
Date:



# ATTACHMENT NO. 6 TO AGREEMENT

# **PAYMENT BOND**

(CALIFORNIA PUBLIC WORK)

KNOW ALL MEN BY THESE PRESENTS, that whereas, the SAN BERNARDINO CITY UNIFIED SCHOOL DISTRICT OF SAN BERNARDINO COUNTY (sometimes referred to hereinafter as "DISTRICT") has awarded to CONTRACTOR NAME, hereinafter designated as the "CONTRACTOR", an agreement for the public work described as follows:

PROJECT NAME:
BID NO.:
WHEREAS, said CONTRACTOR is required to furnish a bond in connection with said Contract, and pursuant to California Civil Code section 9550;
NOW, THEREFORE, We, CONTRACTOR NAME, the undersigned Contractor, as Principal; and , a corporation organized and existing under the laws of the State
of, and duly authorized to transact business under the laws of the State of California,
as Surety, are held and firmly bound unto the SAN BERNARDINO CITY UNIFIED SCHOOL DISTRICT and
to any and all persons, companies, or corporations entitled by law to file stop notices under California Civil
Code section 9100, or any person, company, or corporation entitled to make a claim on this bond, in the
sum of AMOUNT IN DOLLARS AND 00/100 (\$XX,XXX.00), such sum being not less than one hundred
percent (100%) of the total amount payable by said Obligee under the terms of said Contract, for which
payment will and truly to be made, we
bind ourselves, our heirs, executors and administrators, successors and assigns, jointly and severally, firmly by these presents.

THE CONDITION OF THIS OBLIGATION IS SUCH that if said Principal, its heirs, executors, administrators, successors, or assigns, or subcontractor, shall fail to pay any person or persons named in Civil Code section 9100; or fail to pay for any materials, provisions, or other supplies, used in, upon, for, or about the performance of the work contracted to be done, or for any work or labor thereon of any kind, or for amounts due under the Unemployment Insurance Code, with respect to work or labor thereon of any kind; or shall fail to deduct, withhold, and pay over to the Employment Development Department, any amounts required to be deducted, withheld, and paid over by Unemployment Insurance Code section 13020 with respect to work and labor thereon of any kind, then said Surety will pay for the same, in an amount not exceeding the amount herein above set forth, and in the event suit is brought upon this bond, also will pay such reasonable attorneys' fees as shall be fixed by the court, awarded and taxed as provided in California Civil Code section 9550 et seg.

This bond shall inure to the benefit of any and all persons, companies, and corporations entitled to file claims under Section 9100 of the California Civil Code, so as to give a right of action to them or their assigns in any suit brought upon this bond.

It is further stipulated and agreed that the Surety of this bond shall not be exonerated or released from the obligation of the bond by any change, extension of time for performance, addition, alteration or modification in, to, or of any contract, plans, or specifications, or agreement pertaining or relating to any scheme or work of improvement herein above described; or pertaining or relating to the furnishing of labor, materials, or equipment therefor; nor by any change or modification of any terms of payment or extension of time for payment pertaining or relating to any scheme or work of improvement herein above described; nor by any rescission or attempted rescission of the contract, agreement or bond; nor by any conditions precedent or



subsequent in the bond attempting to limit the right of recovery of claimants otherwise entitled to recover under any such contract or agreement or under the bond; nor by any fraud practiced by any person other than the claimant seeking to recover on the bond; and that this bond be construed most strongly against the Surety and in favor of all persons for whose benefit such bond is given; and under no circumstances shall the Surety be released from liability to those for whose benefit such bond has been given, by reason of any breach of contract between the DISTRICT and the CONTRACTOR or on the part of any obligee named in such bond; that the sole condition of recovery shall be that the claimant is a person described in California Civil Code Sections 9100, and who has not been paid the full amount of his or her claim; and that the Surety does hereby waive notice of any such change, extension of time, addition, alteration or modification herein mentioned.

IN WITNESS WHEREOF this instrument has been duly executed b on the, 20	y the Principal and Surety above named,
PRINCIPAL: Corporate Seal of Principal if Corporation	Principal
Ву:	
Print Name:	
Title:	
SURETY:	
By:	
Title:	
Telephone:	

#### IMPORTANT: THIS IS A REQUIRED FORM.

Surety companies executing bonds must possess a certificate of authority from the California Insurance Commissioner authorizing them to write surety insurance defined in California Insurance Code section 105, and if the work or project is financed, in whole or in part, with federal, grant or loan funds, Surety's name must also appear on the Treasury Department's most current list (Circular 570 as amended).

Any claims under this bond may be addressed to: (Name and Address of Surety)

(Name and Address of agent or representative for service for service of process in California)



Telephone:	Telephone:
	leting this certificate verifies only the identity of the individual who ertificate is attached, and not the truthfulness, accuracy, or validity
STATE OF CALIFORNIA	) ) ss.
COUNTY OF	) 55. )
On	, before me,, who proved on the basis of satisfactory ame(s) is/are subscribed to the within instrument and acknowledged
to me that he/she/they executed the sofon the instrument the person(s), or the	ame(s) is/are subscribed to the within instrument and acknowledged ame in his/her/their authorized capacity(ies) as the Attorney-in-Fac (Surety) and acknowledged to me that by his/her/their signature(se entity upon behalf of which the person(s) executed the instrument.  JRY under the laws of the State of California that the foregoing
WITNESS my hand and official seal.	
Notary Public in and for said State	(SEAL)
Commission expires:	
NOTE: A copy of the power-of-attorno	ey to local representatives of the bonding company must be attached



# ATTACHMENT NO. 7 TO AGREEMENT

### PERFORMANCE BOND

(CALIFORNIA PUBLIC WORK)

WHEREAS, the **SAN BERNARDINO CITY UNIFIED SCHOOL DISTRICT** OF SAN BERNARDINO COUNTY, (sometimes referred to hereinafter referred to as "District or as "Obligee") has awarded CONTRACTOR'S NAME, (hereinafter designated as the "CONTRACTOR"), an agreement for the work described as follows: PROJECT NAME (hereinafter referred to as the "Public Work"); and

described as follows: PROJECT NAME (hereinafter referred to as the "Public Work"); and
PROJECT NAME:
BID NO.:
WHEREAS, the work to be performed by the CONTRACTOR is more particularly set forth in that certain contract for said Public Work dated February 6, 2019, (hereinafter referred to as the "Contract"), which Contract is incorporated herein by this reference; and
WHEREAS, the CONTRACTOR is required by said Contract to perform the terms thereof and to provide a bond both for the performance and guaranty thereof.
NOW THEREFORE, we, CONTRACTOR'S NAME, the undersigned CONTRACTOR, as Principal, and, a corporation organized and existing under the laws of the State of, and duly authorized to transact business under the laws of the State of California as Surety, are held and firmly bound unto the SAN BERNARDINO CITY UNIFIED SCHOOL DISTRICT in the sum of NINE HUNDRED TWO THOUSAND THREE HUNDRED EIGHTY NINE DOLLARS AND 00/100 (\$902,389.00), said sum being not less than one hundred percent (100%) of the total amount payable by said Obligee under the terms of said Contract, for which amount well and truly to be made, we bind ourselves, our heirs, executors, administrators, successors, and assigns, jointly and severally, firmly by these presents.
THE CONDITION OF THIS OBLIGATION IS SUCH THAT, if the bounded CONTRACTOR, his or her heirs executors, administrators, successors or assigns, shall in all things stand to and abide by, and well and truly keep and perform the covenants, conditions, and agreements in the contracts and any alteration thereof made as therein provided, on his or her part, to be kept and performed at the time and in the manner.

THE CONDITION OF THIS OBLIGATION IS SUCH THAT, if the bounded CONTRACTOR, his or her heirs, executors, administrators, successors or assigns, shall in all things stand to and abide by, and well and truly keep and perform the covenants, conditions, and agreements in the contracts and any alteration thereof made as therein provided, on his or her part, to be kept and performed at the time and in the manner therein specified, and in all respects according to their intent and meaning; and shall faithfully fulfill guarantees of all materials and workmanship; and indemnify, defend and save harmless the Obligee, its officers and agents, as stipulated in said Contract, then this obligation shall become null and void; otherwise it shall be and remain in full force and effect.

The Surety, for value received, hereby stipulates and agrees that it shall not be exonerated or released from the obligation of this bond (either by total exoneration or pro tanto) by any change, extension of time, alteration in or addition to the terms of the contract or to the work to be performed there under or the specifications accompanying the same, nor by any change or modification to any terms of payment or extension of time for any payment pertaining or relating to any scheme of work of improvement under the contract. Surety also stipulates and agrees that it shall not be exonerated or released from the obligation of this bond (either by total exoneration or pro tanto) by any overpayment or underpayment by the Obligee that is based upon estimates approved by the Architect. The Surety stipulates and agrees that none of the aforementioned changes, modifications, alterations, additions, extension of time or actions shall in any way affect its obligation on this bond, and it does hereby waive notice of any such changes, modifications, alterations, additions or extension of time to the terms of the contract, or to the work, or the specifications as well notice of any other actions that result in the foregoing.



Whenever Principal shall be, and is declared by the Obligee to be, in default under the Contract, the Surety shall promptly either remedy the default, or shall promptly take over and complete the Contract through its agents or independent contractors, subject to acceptance and approval of such agents or independent contractors by Obligee as hereinafter set forth, in accordance with its terms and conditions and to pay and perform all obligations of Principal under the Contract, including, without limitation, all obligations with respect to warranties, guarantees and the payment of Liquidated Damages; or, at Obligee's sole discretion and election, Surety shall obtain a bid or bids for completing the Contract in accordance with its terms and conditions, and upon determination by Obligee of the lowest responsible bidder, arrange for a contract between such bidder and the Obligee and make available as Work progresses (even though there should be a default or succession of defaults under the contract or contracts of completion arranged under this paragraph) sufficient funds to pay the cost of completion less the "balance of the Contract Price" (as hereinafter defined), and to pay and perform all obligations of Principal under the Contract, including, without limitation, all obligations with respect to warranties, guarantees and the payment of Liquidated Damages. The term "balance of the Contract Price," as used in this paragraph, shall mean the total amount payable to Principal by the Obligee under the Contract and any modifications thereto, less the amount previously paid by the Obligee to the Principal, less any withholdings by the Obligee allowed under the Contract. Obligee shall not be required or obligated to accept a tender of a completion contractor from the Surety.

Surety expressly agrees that the Obligee may reject any agent or contractor which may be proposed by Surety in fulfillment of its obligations in the event of default by the Principal. Unless otherwise agreed by Obligee, in its sole discretion, Surety shall not utilize Principal in completing the Contract nor shall Surety accept a bid from Principal for completion of the work in the event of default by the Principal.

No final settlement between the Obligee and the Contractor shall abridge the right of any beneficiary hereunder, whose claim may be unsatisfied.

The Surety shall remain responsible and liable for all patent and latent defects that arise out of or relate to the Contractor's failure and/or inability to properly complete the Public Work as required by the Contract and the Contract Documents. The obligation of the Surety hereunder shall continue so long as any obligation of the Contractor remains.

Contractor and Surety agree that if the Obligee is required to engage the services of an attorney in connection with enforcement of the bond, Contractor and Surety shall pay Obligee's reasonable attorneys' fees incurred, with or without suit, in addition to the above sum.

In the event suit is brought upon this bond by the Obligee and judgment is recovered, the Surety shall pay all costs incurred by the Obligee in such suit, including reasonable attorneys' fees to be fixed by the Court.

IN WITNESS WHER	EOF this instrument has b	een duly executed by the Pri	ncipal and Surety above named,
on the	day of	, 20	
PRINCIPAL:			
Corporate Seal of Pr	rincipal if Corporation		Principal
Ву:			
Print Name:			
Title:			



SURETY:	
Ву:	
Title:	
Telephone:	
The rate of premium on this bond is	per thousand.
The total amount of premium charged: \$corporate surety).	(This must be filled in by a



### IMPORTANT: THIS IS A REQUIRED FORM.

Surety companies executing bonds must possess a certificate of authority from the California Insurance Commissioner authorizing them to write surety insurance defined in California Insurance Code section 105, and if the work or project is financed, in whole or in part, with federal, grant or loan funds, Surety's name must also appear on the Treasury Department's most current list (Circular 570 as amended).

Any claims under this bond may be addressed to: (Name and Address of Surety)	(Name and Address of agent or representative for service for service of process in California)
Telephone:	Telephone:
	rtificate verifies only the identity of the individual who tached, and not the truthfulness, accuracy, or validity
STATE OF CALIFORNIA ) ss. COUNTY OF )	
evidence to be the person(s) whose name(s) is/are to me that he/she/they executed the same in his/he of (Surety) and on the instrument the person(s), or the entity upon	, who proved on the basis of satisfactory subscribed to the within instrument and acknowledged er/their authorized capacity(ies) as the Attorney-in-Fact acknowledged to me that by his/her/their signature(s) behalf of which the person(s) executed the instrument. he laws of the State of California that the foregoing
WITNESS my hand and official seal.  Notary Public in and for said State	(SEAL)
Commission expires:	



attached hereto.

A copy of the power-of-attorney to local representatives of the bonding company must be

NOTE:



# **ATTACHMENT NO. 8 TO AGREEMENT**

# PRIME/GENERAL CONTRACTOR INFORMATION

Company Name:			
Mailing Address:			
Delivery Address, if different:			
	COMPANY CONT	ACTS	
General Correspondence:		Phone:	
·	Name & Title	Гом	
Billing & Pay Requests:		Phone:	
	Name & Title	Fax:	
Field Issues/Coordination:		Phone:	
	Name & Title		
Proposal Requests &		Phone:	
Change Order Pricing:	Name & Title	Гом	
Submittals:		Phone:	
	Name & Title	Fax:	
RFI's:		Phone:	
	Name & Title	Fax:	
	24-HOUR EMERGENCY	CONTACT	
Name:			
Title:			
Telephone:			
Cell Phone/Pager:			



#### ATTACHMENT NO. 9 TO AGREEMENT

Form W-9

# Request for Taxpayer

Give Form to the

Identification Number and Certification  Internal Florence Service					send t	to the	IRS.
	•	your income tax return)					
page 2.	Business name/disr	agarded entity name, if different from above					
	Check appropriate	box for federal tax classification:		Examption	s (see ins	truction	5):
hint or type Instructions on	Individual/sole	proprietor C Corporation S Corporation Partnership 1	Trust/estate	Exempt per	una cada	W	
불음					•		
Print or type Instruction	Limned liability	y company. Enter the tax classification (C=C corporation, S=S corporation, P=partnersh		examption code (if an		CA rep	orting
	Other (see ins	tructions) >					
Specific	Address (number, s	treet, and apt. or suite no.)	Requester's name ar	nd address	(optional)		
See	City, state, and ZIP	code					
	List account number	r(s) here (optional)					
Par	Taxpay	ver Identification Number (TIN)					
Enter	your TIN In the app	propriate box. The TIN provided must match the name given on the "Name" i	ine Social secu	urtty numb	er		
		ding. For individuals, this is your social security number (SSN). However, for a	а 🗆			$\neg \neg$	$\Box$
entitie	s, it is your employ	tetor, or disregarded entity, see the Part I Instructions on page 3. For other ver identification number (EIN). If you do not have a number, see How to get a	a <u>                                    </u>	] -			
TIN or	page 3.						
Note.	If the account is in	more than one name, see the chart on page 4 for guidelines on whose	Employer i	dentificati	on numbe	ar	
numbe	er to enter.		<u> </u>	$\cdot \square$			
Part	Certific	cation	· ' '				

Under penalties of perjury, I certify that:

- 1. The number shown on this form is my correct taxpayer identification number (or I am waiting for a number to be issued to me), and
- 2. I am not subject to backup withholding because: (a) I am exempt from backup withholding, or (b) I have not been notified by the internal Revenue Service (IRS) that I am subject to backup withholding as a result of a failure to report all interest or dividends, or (c) the IRS has notified me that I am no longer subject to backup withholding, and
- I am a U.S. citizen or other U.S. person (defined below), and
- 4. The FATCA code(s) entered on this form (if any) indicating that I am exempt from FATCA reporting is correct.

Certification instructions. You must cross out item 2 above if you have been notified by the IRS that you are currently subject to backup withholding because you have falled to report all interest and dividends on your tax return. For real estate transactions, item 2 does not apply. For mortgage interest paid, acquisition or abandonment of secured property, cancellation of debt, contributions to an individual retirement arrangement (IRA), and generally, payments other than interest and dividends, you are not required to sign the certification, but you must provide your correct TIN. See the instructions on page 3.

Sign Signature of U.S. person ▶ Date ▶

#### General Instructions

Section references are to the Internal Revenue Code unless otherwise noted.

Future developments. The IRS has created a page on IRS.gov for information about Form W-9, at www.irs.gov/w9. Information about any future developments affecting Form W-9 (such as legislation enacted after we release it) will be posted

#### Purpose of Form

A person who is required to file an information return with the IRS must obtain your correct taxpayer identification number (TIN) to report, for example, income paid to you, payments made to you in settlement of payment card and third party network transactions, real estate transactions, mortgage interest you paid, acquisition or abandonment of secured property, cancellation of debt, or contributions you made

Use Form W-9 only if you are a U.S. person (including a resident alleri), to provide your correct TIN to the person requesting it (the requester) and, when applicable, to:

- 1. Certify that the TIN you are giving is correct (or you are waiting for a number
- 2. Certify that you are not subject to backup withholding, or
- Claim exemption from backup withholding if you are a U.S. exempt payee. If applicable, you are also certifying that as a U.S. person, your allocable share of any partnership income from a U.S. trade or business is not subject to the

withholding tax on foreign partners' share of effectively connected income, and

Cortify that FATCA code(s) entered on this form (if any) indicating that you are exempt from the FATCA reporting, is correct.

Note. If you are a U.S. person and a requester gives you a form other than Form W-9 to request your TIN, you must use the requester's form if it is substantially similar to this Form W-9.

Definition of a U.S. person. For federal tax purposes, you are considered a U.S. person if you are:

- An individual who is a U.S. citizen or U.S. resident alien,
- A partnership, corporation, company, or association created or organized in the United States or under the laws of the United States,
- An estate (other than a foreign estate), or
- A domestic trust (as defined in Regulations section 301.7701-7).

A domestic trust (as doffined in Regulations section 301.7701-7). Special rules for partnerships. Partnerships that conduct a trade or business in the United States are generally required to pay a withholding tax under section 1446 on any foreign partners' share of effectively connected taxable income from such business. Further, in certain cases where a Form W-9 has not been received, the rules under section 1446 require a partnership to presume that a partner is a foreign person, and pay the section 1446 withholding tax. Therefore, if you are a U.S. person that is a partner in a partnership conducting a trade or business in the United States, provide Form W-9 to the partnership to establish your U.S. status and avoid section 1446 withholding on your share of partnership income.

Cat. No. 10231X

Form W-9 (Rev. 8-2013)

For further instructions on how to fill this page out, please visit <a href="www.irs.gov">www.irs.gov</a>.

Prime/General Contractor Information



# **ATTACHMENT NO. 10 TO AGREEMENT**

# **VENDOR APPLICATION**

When completed mail to:

San Bernardino City Unified School District Purchasing Department 793 N. 'E' Street, San Bernardino, CA 92410 (909) 381-1126

Business Name:				
Business License Number: Expiration Date:				
Representative's Name:				
Business Address:				
Number of years in business				
Business Telephone Numbe				
Products or Services Providence				
Comments:				
List of references where you				r school districts)
Other Schools or Business Name/Address	Contact	Phone	Dates of Service	Products/Service
1.				
2.				
3.				
"By signing below, I certify understand it is that as a vendor of the District that are a vendor's total responsibility to opprovided, totally conforms to a guarantee that this company will understand the District reservatives and the Sun newspaper."	he vendor's responsinis company will conformation of some as required by the determine specific desuch legal requirement be used as a vendoes the right to use, as	ibility to update the form to all Federal, ervices provided, e San Bernardino tails of such requirents. I understand for the District or	e above information as State, County and City including but not limite City Clerk's Office. I rements and warrant that d the submission of the requested to quote on	needed. I further agree laws, ordinances, codes do to, obtaining a *San understand that it is the at all work performed, or all sapplication does not any or all requirements.
Authorized Vendor Represe	ntative Signature: _		Dat	e:



Name:	Title:	



# **ATTACHMENT NO. 11 TO AGREEMENT**

# **CERTIFICATION OF NON-UTILIZATION OF ASBESTOS MATERIAL**

PROJECT:	
WE HEREBY CERTIFY THAT NO ASBESTOS OR INCORPORATED OR USED IN THIS CONSTRUCT EQUIPMENT USED TO EFFECT THIS CONSTRUCT	ON OR IN ANY TOOLS, DEVICES, CLOTHING OR
Asbestos and/or asbestos-containing products are to chrysotile, crocidolite, amosite, anthophyllite, to	
Any or all material containing greater than one-tenth of asbestos-containing material.	of one percent (>0.1%) asbestos shall be defined as
Any disputes involving the question of whether or not n shall be settled by electron microscopy; the cost of an	
All work or materials installed by the contractor which installed with asbestos-containing equipment, will be it and replaced by the Contractor at no additional cost to	mmediately rejected and this work shall be removed
Decontamination and removal of work found to contain equipment shall be done only under supervision of a field of asbestos abatement and accredited by the Environment	qualified Asbestos Consultant, knowledgeable in the
The Asbestos Removal Contractor shall be an EPA asbestos and shall be chosen and approved by the A and final determination in this matter.	
The Asbestos Consultant shall be chosen and approving final determination in this matter.	ed by the District who shall have sole discretion and
The work will be not accepted until asbestos contamin Asbestos Consultant.	ation is reduced to levels deemed acceptable by the
Subcontractor (if applicable):	General/Prime Contractor:
Name:	Name:
By:	Ву:
Signature	Signature:



# **ATTACHMENT NO. 12 TO AGREEMENT**

# CONTRACTOR PREVAILING WAGE COMPLIANCE CERTIFICATION

To: San Bernardino City Unified School District Facilities Management Department 956 W. 9<sup>th</sup> St San Bernardino, California 92411

I hereby certify that I will comply with the State of California Public Works Contract Requirements and Department of Industrial Relations Wage Orders regarding wages, benefits, on site audits with 48-hour notice, payroll records and apprentice and trainee employment requirements.

CONTRACTOR
CONTRACTOR'S PRINCIPAL'S SIGNATURE
DATED



# **ATTACHMENT NO. 13 TO AGREEMENT**

# **GUARANTEE**

Guarantee for, which	We hereby guarantee that the we have installed in
/	we have installed in done in accordance with the Contract Documents,
including without limitation, the drawings and spec	cifications, and that the work as installed will fulfill the undersigned and its surety agrees to repair or replace
any or all such work, together with any other adjac	cent work, which may be displaced in connection with tive in workmanship or material within a period of
Three (3) years from the dat	te of the Notice of Completion of the above-mentioned ED SCHOOL DISTRICT, ordinary wear and tear and
unusual abuse or neglect excepted.	ED School District, ordinary wear and tear and
a reasonable period of time, as determined by the notified in writing by the DISTRICT or within 48 ho undersigned and its surety authorizes the DISTRIC good at the expense of the undersigned and its su	Is to comply with the above-mentioned conditions within DISTRICT, but not later than ten (10) days after being turns in the case of an emergency or urgent matter, the CT to proceed to have said defects repaired and made rety, who will pay the costs and charges therefor upon bintly and severally liable for any costs arising from the
	Countersigned:
(Proper Name)	(Proper Name)
By:	By:
(Signature of Subcontract or Contractor)	(Signature of General Contractor if for Subcontractor)
Representatives to be contacted for service:	
Name:	
Address:	
Phone Number:	



# **ATTACHMENT NO. 14 TO AGREEMENT**

# CONTRACTOR CERTIFICATION REGARDING BACKGROUND CHECKS (Modernization Projects)

<u>CONTRACTOR'S NAME</u> certifies that it has performed one of the following:
[Name of contractor/consultant]

Pursuant to Education Code Section 45125.1, Contractor has conducted criminal background checks, through the California Department of Justice, of all employees providing services to the SAN BERNARDINO CITY UNIFIED SCHOOL DISTRICT, pursuant to the contract/purchase order dated February 6, 2019, and that none have been convicted of serious or violent felonies, as specified in Penal Code Sections 1192.7(c) and 667.5(c), respectively.

As further required by Education Code Section 45125.1, attached hereto as Attachment "A" is a list of the names of the employees of the undersigned who may come in contact with pupils.

OR

Pursuant to Education Code Section 45125.2, Contractor will ensure the safety of pupils by one or more of the following methods:

- 1. The installation of a physical barrier at the worksite to limit contact with pupils.
- Continual supervision and monitoring of all employees of the entity by an employee of the entity whom the Department of Justice has ascertained has not been convicted of a violent or serious felony.

I declare under	penalty of perjury under	the laws of the United States that the foregoing is true and correct.
Date	, 20	[Name of Contractor/Consultant]
		By its:



# ATTACHMENT 14A: CONTRACTOR CERTIFICATION REGARDING BACKGROUND CHECKS

(INSERT NAMES OF EMPLOYEES WHO MAY COME IN CONTACT WITH PUPILS)



# 14B - Fingerprint and Criminal Background Check Certification

In accordance with Department of Justice (DOJ) fingerprint and criminal background investigation requirements of Education Code section 45125.1 et seq.

With respect to	the Agreement dated	, between t	he San Bernardino City Unified School
	TRICT") and the individual, comp		("VENDOR"), for
-	ll appropriate boxes and sign belo		
REQUIREME			
☐ <b>A</b> ) The VI requirements o	ENDOR hereby certifies to the DI f Education Code (EC) section 45 dents have been convicted of a fe	5125.1 and that none of its employees t	completed the criminal background check hat may come into contact with (c) or a serious felony listed in Penal Code
	, or attach, all employees' named check clearance in accordance	nes that have successfully complete ce with the law.	d the fingerprinting and criminal
	Services may l	pegin after a purchase order (P.O.) is issued	l to the vendor.
-OR- REQUEST FO	OR WAIVER:		
		, BE ADVISED THAT NO SERVIC DISTRICT AND A P. O. IS ISSUE	ES ARE TO BE PROVIDED UNTIL D.
		artment of Justice (DOJ) fingerprint and crim	ninal background investigation for the reason(s)
	ucation Code section 45125.1 et seq.	have NO CONTACT with pupils. (No scho	ol site services will be provided )
_		1 1 ,	ttach information about the length of time on
schoo and a	ol grounds, proximity of work area to provide a substantiate limit	pupil areas, whether VENDOR/its employe ed contact.) [EC 45125.1(c)]	es will be working by themselves or with others,
MOR	E of the following methods are utilized	have OTHER THAN LIMITED CONTACTED to ensure pupil safety. [EC 45125.2(a)]	with pupils but will assure that ONE (1) OR
C	heck all methods to be used:		
		barrier at the worksite to limit contact with	pupils. OOR by an employee of the VENDOR who has
	not been convicted o	of a serious or violent felony as ascertained	
		es of the VENDOR by school personnel.	0)
<u> </u>	he services provided by the VENDOI	R are for an "EMERGENCY OR EXCEPTI	ONAL SITUATION, such as when pupil health
is accurate. I	understand that it is the VENDOR	's sole responsibility to maintain, update	e certification form and attached employee list(s and provide the DISTRICT with current t, throughout the duration of VENDOR provide
or saf	ety is endangered or when repairs are	needed to make school facilities safe and h	abitable". [EC 45125.1(b)]
Authorizes VENI	· ·	lame Title	Date
BOTH DISTRICT API	PROVALS SHOWN BELOW ARE REQUIRED		
For	WAIVER REQUEST: By:	AI <b>P</b> ROVED	DENIED □ Date:
District Use		ss Services Division OR Other Authorized District Agent	
Only	WAIVER REQUEST: By: Perry Wiseman, Assistant Superintendent, H	APPROVED   Tuman Resources Division OR Other Authorized District 4	DENIED Date:
	- 211 , 11 deman, 12 distant Superintendent, 11	One manorige District	0



# **ATTACHMENT NO. 15 TO AGREEMENT**

# ESCROW AGREEMENT FOR SECURITY DEPOSITS IN LIEU OF RETENTION (Optional)

This Escrow Agreement is made and entered into by and between the SAN BERNARDINO CITY

UNIFIED SCHOOL DISTRICT,	, San E	Bernardino, Cali	fornia	,
UNIFIED SCHOOL DISTRICT,hereinafter called "OWNER", and		whose	e address	is
, hereinafter ca	alled "CONTRACTOR",	and		
whose address is	_, hereinafter called "Esc	row Agent".		
For the consideration hereinafter set tas follows:	orth, the OWNER, CONT	RACTOR and E	scrow Agent a	ıgree
(1) Pursuant to section 22300 CONTRACTOR has the option to deposit securequired to be withheld by OWNER pursual OWNER and CONTRACTOR for (hereinafter referred to as contractor, the OWNER shall make payments CONTRACTOR deposits the securities as a sthe OWNER within ten (10) days of deposit. To shall be at least equal to the cash amount the Contract between the OWNER and CONTRACTOR as before the contract of the contract o	rities with Escrow Agent and to the Construction Contract in the "Contract"). Altern of the retention earnings ubstitute for Contract earning market value of the secon required to be withheld CTOR. Securities shall be	as a substitute for Contract entered he amount of atively, on writt directly to the es nings, the Escro- curities at the time as retention und	r retention earn I into between en request of crow agent. W w Agent shall r e of the substit der the terms of	nings the dated f the Vhen notify ution of the
(2) The OWNER shall make progress pa otherwise would be withheld from progress pa Escrow Agent holds securities in the form and	yments pursuant to the C	ontract provision		
(3) When the OWNER makes pa Escrow Agent shall hold them for the benefit of this contract is terminated. The Contractor materials and conditions of this agreement and applicable and binding when the OWNER pay	of the Contractor until suc ay direct the investment the rights and responsibi	h time as the eso of the payments lities of the partion	crow created u into securities	ınder s. All

(5) The interest earned on the securities or the money market accounts held in escrow and all interest earned on that interest shall be for the sole account of CONTRACTOR and shall be subject to withdrawal by CONTRACTOR at any time and from time to time without notice to the OWNER.

Escrow Agent in administering the Escrow Account and all expenses of the OWNER. These expenses and

payment terms shall be determined by the OWNER, CONTRACTOR, and Escrow Agent.

CONTRACTOR shall be responsible for paying all fees for the expenses incurred by

- (6) CONTRACTOR shall have the right to withdraw all or any part of the principal in the Escrow Account only by written notice to Escrow Agent accompanied by written authorization from the OWNER to the Escrow Agent that OWNER consents to the withdrawal of the amount sought to be withdrawn by CONTRACTOR.
- (7) The OWNER shall have a right to draw upon the securities in the event of default by the CONTRACTOR. Upon seven (7) days' written notice to the Escrow Agent from the OWNER of the notice of default under Article 2.2, Article 9.6 or Article 14, the Escrow Agent shall immediately convert the securities to cash and shall distribute the cash as instructed by the OWNER.



- (8) Upon receipt of written notification from the OWNER certifying that the Contract is final and complete, and that the CONTRACTOR has complied with all requirements and procedures applicable to the Contract, Escrow Agent shall release to CONTRACTOR all securities and interest on deposit less escrow fees and charges of the Escrow Account. The escrow shall be closed immediately upon disbursement of all moneys and securities on deposit and payment of fees and charges.
- (9) Escrow Agent shall rely on the written notifications from the OWNER and the CONTRACTOR pursuant to Sections (5) to (8), inclusive, of this agreement and the OWNER and CONTRACTOR shall hold Escrow Agent harmless from Escrow Agent's release and disbursement of the securities and interest as set forth above.
- (10) The names of the persons who are authorized to give written notice or to receive written notice on behalf of the OWNER and on behalf of CONTRACTOR in connection with the foregoing, and exemplars of their respective signatures are as follows:

On behalf of Owner:		
Title		
Name		
Signature		
Address		
On behalf of Contractor:		
Title		
Name		
Signature		
Address		



On behalf of Agent:	
Title	
Name	
Signature	
Address	
Escrow Agent a fully executed counter	t is opened, the OWNER and CONTRACTOR shall deliver to the art of this Agreement.  arties have executed this Agreement by their proper officers on the
OWNER	CONTRACTOR
Title	Title
Name	Name
Signature	Signature



# **ATTACHMENT NO. 16 TO AGREEMENT**

# DISABLED VETERAN BUSINESS ENTERPRISE (DVBE) CONTRACTOR CLOSE-OUT STATEMENT

The Contractor shall complete this form, as a condition to final payment, for purposes of reporting participation by Disabled Veteran Business Enterprises (DVBE) in the contract for the Project/Bid No. specified below. Project Name: Bid No.: DSA No.: Category of \$ Amount of Name Address/Phone Work\* Contract \*Categories of work include: (1) construction services (specify services that DVBE will provide); (2) architecture and engineering services; (3) procurement of materials, supplies and equipment; and (4) information technology. The undersigned, on behalf of the Contractor, certifies that DVBE participation on the contract for Bid No. dollars (\$ equaled represents approximately \_\_\_\_% of the total contract price including change orders for the Project. Company: \_\_ Name: Title:

Signature: \_\_\_\_\_ Date: \_\_\_\_



### **ATTACHMENT NO. 17 TO AGREEMENT**

#### **CONDUCT RULES FOR CONTRACTORS**

Each contractor/subcontractor, when performing work on San Bernardino City Unified School District property, in addition to complying with the provisions of the Master Construction Services Agreement, shall adhere to the following rules of conduct:

- 1. Professional and courteous conduct is expected and will be displayed at all times.
- 2. Interaction with students, staff, and/or other visitors is prohibited with the exception of designated administrators.
- 3. The use of profanity and/or disparaging language will not be tolerated.
- 4. All contractors/subcontractors shall wear a means of identification on site when school is in session which must be approved by the District prior to commencement of work.
- 5. All contractors/subcontractors shall remain in the vicinity of his/her work and will not stray to other areas of the property not involved in the Project, including student and staff toilet facilities.
- 6. Pursuant to Government Code section 8350 et seq., the San Bernardino City Unified School District is a drug-free workplace. This policy shall be strictly enforced.
- 7. Alcoholic beverages are prohibited from being consumed or brought on any District property.
- 8. The use of any tobacco products on District property is strictly prohibited.
- 9. Any lewd, obscene or otherwise indecent acts, words, or behavior by any contractor/subcontractor shall not be tolerated.
- 10. All contractors/subcontractors shall conform to a dress code whereby:
  - A. No clothing that contains violent, suggestive, derogatory, obscene, or racially-biased material may be worn.
  - B. Garments, accessories or personal grooming artifacts with slogans, graphics, or pictures promoting drugs, alcohol, tobacco, or any other controlled substances which are prohibited to minors will not be allowed.
- 11. No firearms are allowed on campuses/District property.
- 12. All contractors/subcontractors shall comply with Education Code section 45125 et seq. with respect to all fingerprinting requirements.

Non-compliance with any of the above-stated rules of conduct by any contractor/subcontractor may be sufficient grounds for immediate removal from the job site and termination of the contract.

I acknowledge that I am aware of the above-stated rules of conduct and hereby certify that all of my Company's employees, consultants, suppliers, and/or any subcontractors will adhere to these provisions.

Date	Authorized Signature
	Print Name
	Company



# **ATTACHMENT NO. 18 TO AGREEMENT**

# (a.k.a. EXHIBIT K)

# PROJECT CONTRACT INFORMATION

Project Name:	
Description of th	e Project:
Description of th	e Site:
Scope of Work/C	Construction Documents:
DSA Application	n Number:
DSA File No.:	
Master Budget:	See Attached
Section 3.5	Contract Time is Days.
Section 3.6	Liquidated Damages for overstaying lease (Art. 18) is \$ per calendar day.
Section 3.7	Guaranteed Maximum Price (Art. 5) is \$
3.7.1	Construction Contingency (within GMP) is \$
3.7.2	Errors and Omissions Contingency (within GMP) is \$
Section 3.8	The only exception to the GMP is Unforeseen Underground Conditions, and District Contingency for Owner requested extras as follows:
3.8.1	District's Contingency (Art. 8) is \$ District Contingency is carried outside of the GMP.
3.8.2	Unforeseen Allowance is \$ Unforeseen Allowance is carried outside of the GMP.
Section 3.9 the GMP.	The Contractor's fee for this Project is% percent and is included in



# **Division 1 Forms**

# IMMEDIATE CONSTRUCTION CHANGE DIRECTIVE NO.

PROJECT:	
TO:	
You are hereby directed to provide the extra work necessary to comply with this ICD.	
DESCRIPTION OF CHANGE:	
COST (This cost shall not be exceeded):	
TIME FOR COMPLETION:	
NOTE:	
Pursuant to Article 17.4.1.2 An Immediate Change Directive is a written order to the Contract Architect and signed by the District (and CM if there is a CM on the Project) and the Architect, dithe Work and stating a proposed basis for adjustment, if any, in the Contract Sum or Contract District may by ICD, without invalidating the Contract, direct immediate changes in the Work scope of the Contract consisting of additions, deletions, or other revisions within. If applicable, the Contract Time will be adjusted accordingly. CONTRACTOR SHALL PROCEED WITH WOR THIS ICD IMMEDIATELY UPON RECEIPT OR THE DISTRICT MAY EITHER HOLD THE IN EITHER PARTIAL DEFAULT PURSUANT TO ARTICLE 12.2 OR TOTAL DEFAULT ARTICLE 19.	irecting a change in Time, or both. The within the general e Contract Sum and K SET FORTH IN E CONTRACTOR
Architect	
 District	



# **CERTIFICATE OF SUBSTANTIAL COMPLETION**

PROJECT:
TO:
As the Architect for the Project described above, the Project has reached Substantial Completion. Substantial Completion is not reached unless and until each of the following three (3) conditions have been met: (1) all contractually required items have been installed with the exception of only minor and Incomplete Punch Items (See Article 13.16 of the Construction Services Agreement); (2) All Fire/Life Safety Systems have been installed, and are working and signed off on the DSA Form 152 Inspection Card, all building systems including mechanical, electrical and plumbing are all functioning; and (3) the Project is fit for occupancy and its intended use
I certify that the Project has reached Substantial Completion as defined above on the following date:
Architect



# **RFP Attachment 3A**

# SAN BERNARDINO CITY UNIFIED SCHOOL DISTRICT PROJECT

# **MASTER SITE LEASE**

Between

SAN BERNARDINO CITY UNIFIED SCHOOL DISTRICT

and

Dated as of \_\_\_\_\_\_, 2019



# SAN BERNARDINO CITY UNIFIED SCHOOL DISTRICT PROJECT

# **MASTER SITE LEASE**

Unified School Distric	R SITE LEASE is dated as of, 2019, and is by and between the San Bernardino City t, a school district duly organized and existing under the laws of the State of California (the nd, a California corporation operating under the laws of the e "Lessee").
	ne District desires to provide for the construction of certain public improvements at <mark>[insert cts]</mark> (the "Projects"); and
for the common bene buildings at the vario specifically described	ne District's governing board has determined that it is in the best interests of the District and effit of the citizens it serves to construct the Projects by leasing to the Lessee land and existing ous District school sites at which the public improvements are to be constructed, as more in Exhibit "A," (the "Sites"), and subleasing from the Lessee the Sites and the Projects under a reement (the "Master Sublease") attached hereto as Exhibit "B" and by this reference and
of the sites, site condi	ne Lessee has conducted Due Diligence of the Sites and the Projects to determine the suitability tions, utilities, hazardous substances, and other conditions for the construction of the Projects t Article 5 of the Master Construction Services Agreement); and
	ne District is authorized under Section 17406 of the California Education Code to lease the Site y has duly authorized the execution of this Master Site Lease; and
Sublease, which is incorretain beneficial use at the constructed improved Master Site Lease, the improvements on the Construction Services	ursuant to this Master Site Lease, the District and Lessee have agreed to the terms of the orporated and attached hereto as Exhibit "B," by which the District will sublease the Sites and and occupancy of the Sites during which term, improvements will be constructed by Lessee. As ovements are completed, the District shall receive full beneficial use and occupancy of the ments upon payment for such improvements by the District to the Lessee. As part of this e District and the Lessee have agreed to terms by which the Lessee will perform construction e Sites during the term of the Master Sublease according to the terms of the Master Agreement ("Master CSA"), which is incorporated and attached to the Master Site Lease as that the improvements will meet the District's expectations and comply with applicable law.
NOW THERE follows:	FORE, in consideration of the covenants hereinafter set forth, District and Lessee agree as
1.	<u><b>DEFINITIONS.</b></u> Unless the context otherwise requires, the terms defined in this Article shall, for all purposes of this lease, have the meanings as herein specified.
A.	"Commencement Date" shall mean the Project commencement date found in the Notice to Proceed for each Project in accordance with the Master Construction Services Agreement.
В.	<u>"Master Construction Services Agreement" (CSA)</u> means the Master Construction Services Agreement, together with any duly authorized and executed amendments hereto.
C.	"Construction Documents" consist of the Plans and Specifications approved by DSA for each project, Allowances stipulated in the Contract Documents, and all



Addenda(s), if any, issued prior to the entry into this Agreement. The Construction Documents shall include all Modifications generated after the Effective Date in accordance with the Contract Documents, including, without limitation, a written amendment to the Contract signed by the Contractor and duly executed and approved by the District, a Change Order, a Construction Change Document, or a written order for a minor change in the Work issued by the Architect.

- D. "Contract Documents" means those documents which form the entire Contract by and between District and Contractor. As of the effective date of the Master Site Lease and Master Sublease, the Contract Documents consist of the Lease, the Sublease, any General, Supplementary and other Conditions, the Master Construction Services Agreement, including all exhibits and attachments hereto, and the Construction Documents. The Contract Documents collectively form the Contract. The Contract represents the entire and integrated Agreement between the parties hereto and supersedes prior negotiations, representations, or agreements, either written or oral. The Contract may be amended or modified only by a written Modification. The Contract Documents shall be binding solely upon the District and Contractor, do not create a contractual relationship of any kind between the Architect and Contractor, between the District and any Subcontractor or Sub-subcontractor, or between any persons or entities other than the District and the Contractor, and are not intended to and do not create any third party beneficiary. The Architect shall, however, be entitled to performance and enforcement of obligations under the Contract intended to facilitate performance of the Architect's duties. (See Article 14 of the Master CSA).
- E. <u>"Day"</u> means a calendar day unless specifically designated as a business day.
- F. <u>"District"</u> means the San Bernardino City Unified School District, a school district duly organized and existing under the laws of the State of California.
- G. <u>"Effective Date"</u> is the latter of the date upon which the District Board approves the Master Site Lease and the Master Sublease and Contractor has executed the Master Site Lease and Master Sublease.
- H. <u>"Lessee"</u> shall mean , and its successors and assigns.
- I. <u>"Project"</u> means the improvements and related work to be constructed and installed by the Contractor, as part of this Master Site Lease and in accordance with the Master Construction Services Agreement attached hereto as Exhibit "C".
- J. <u>"Site"</u> refers to the grounds of the Projects or in some cases may refer to multiple sites as defined in the Contract Documents and such adjacent lands as may be directly affected by the performance of the Work, more particularly described in Exhibit "A" attached hereto.
- K. <u>"Master Site Lease" or "Lease"</u> means this Master Site Lease together with any duly authorized and executed amendment hereto under which the District leases the Sites to the Lessee.
- L. <u>"Master Sublease"</u> means the Sublease attached hereto and incorporated as Exhibit "B", together with any duly authorized and executed amendment thereto.
- M. <u>"Sublease Payment"</u> means any payment required to be made by the District pursuant to Article 7 of the Sublease.



N. <u>"Term of this Lease" or "Term"</u> means the time during which this Lease is in effect, as provided for in Article 3 of this Master Site Lease.

#### 2. **SITE LEASE**.

The District leases to the Lessee, and the Lessee leases from the District, on the terms and conditions set forth herein, the various Sites situated in the City of San Bernardino City Unified, County of San Bernardino, State of California, more specifically described in Exhibit "A" attached hereto, including any improvements now or hereafter affixed thereto. The term of each Site Lease shall commence upon the completion of Lessee's Due Diligence with regard to the sites and the issuance of the Notice to Proceed for each Project.

#### 3. <u>TERM</u>.

The Term of this Master Site Lease shall become effective upon the authorized execution of this Master Site Lease and shall terminate as of the last day of the Master Sublease, unless sooner terminated as provided thereby. If on the scheduled date of termination of this Master Site Lease, any Sublease Payments for any of the various Projects shall have therefore been abated at any time and for any reason, then the term of this Master Site Lease shall be subject to a Liquidated Damages cost as set forth in Article 3.6 of the Master Construction Services Agreement and the Master Site Lease shall be extended until the date upon which all such Sublease Payments shall be fully paid. Without limiting any other term or provision of the Master Sublease Agreement or Master Construction Services Agreement between the parties, at the termination of this Master Site Lease, natural or otherwise, title to the Site, and any improvements constructed thereon by the Lessee, shall vest in the District in accordance with Education Code section 17406.

- 4. **REPRESENTATIONS, COVENANTS, AND WARRANTIES OF THE DISTRICT**. The District represents, covenants and warrants to the Lessee that:
  - A. The District has good and merchantable fee title to the Sites and has authority to enter into and perform its obligations under this Master Site Lease;
  - B. There are no liens on the Sites other than Permitted Encumbrances;
  - C. All taxes, assessments or impositions of any kind with respect to the Sites, if applicable, except current taxes, have been paid in full;
  - D. The Sites are properly zoned (or subject to an exception from zoning) for the intended purpose and utilization of the Site;
  - E. The District is in compliance with all laws, regulations, ordinances and orders of public authorities applicable to the Sites;
  - F. Except for Validation Actions concerning the Projects, there is no litigation of any kind currently pending or threatened regarding the Sites or the District's use of the Sites for the purposes contemplated by this Master Site Lease;
  - G. To the best of the District's knowledge, except for that which shall be disclosed by the District prior to the Project commencement date in the Notice to Proceed:



- (1) no dangerous, toxic or hazardous pollutants, contaminants, chemicals, waste, materials or substances, as defined in or governed by the provisions of any State or Federal Law relating thereto (hereinafter collectively called "Environmental Regulations", and also including, but not limited to, urea-formaldehyde, polychlorinated biphenyls, asbestos, asbestos containing materials, nuclear fuel or waste, radioactive materials, explosives, carcinogens and petroleum products, or any other waste, material, substance, pollutant or contaminant which would subject the owner of the Site or the Lessee or the Lessee's subcontractors to any damages, penalties or liabilities under any applicable Environmental Regulation (hereinafter collectively called "Hazardous Substances", are now or have been stored, located, generated, produced, processed, treated, transported, incorporated, discharged, emitted, released, deposited or disposed of in, upon, under, over or from the Sites;
- (2) no threat exists of a discharge, release or emission of a Hazardous Substance upon or from the Sites into the environment;
- the Sites have not been used as or for a mine, a landfill, a dump or other disposal facility, industrial or manufacturing facility, or a gasoline service station;
- (4) no underground storage tank is now located in the Sites or has previously been located therein;
- (5) no violation of any Environmental Regulation now exists relating to the Sites, no notice of any such violation or any alleged violation thereof has been issued or given by any governmental entity or agency, and there is not now any investigation or report involving the Sites by any governmental entity or agency which in any way relates to Hazardous Substances;
- (6) no person, party or private or governmental agency or entity has given any notice of or asserted any claim, cause of action, penalty, cost or demand for payment or compensation, whether or not involving any injury or threatened injury to human health, the environment or natural resources, resulting or allegedly resulting from any activity or event described in (1) above;
- there are not now any actions, suits, proceedings or damage settlements relating in any way to Hazardous Substances, in, upon, under over or from the Sites;
- (8) the Sites are not listed in the United States Environmental Protection Agency's National Priorities List of Hazardous Waste Sites or any other list of Hazardous Substance sites maintained by any federal, state or local governmental agency; and
- (9) the Sites are not subject to any lien or claim for lien or threat of a lien in favor of any governmental entity or agency as a result of any release or threatened release-of any Hazardous Substance.
- H. To the extent permitted by law, the District shall not abandon the Sites for the use for which it is currently required by the District and further, shall not seek to substitute or acquire property to be used as a substitute for the uses for which the Sites and Projects are to be maintained under the Master Site Lease.



I.	The term "Permitted Encumbrances" as used herein shall mean, as of any particular time:

- (1) liens for general ad valorem taxes and assessments, if any, not then delinquent;
- this Master Site Lease; the Master Sublease; any right or claim of any mechanic, laborer, materialman, supplier, or vendor, if applicable, not filed or perfected in the manner prescribed by law; easements, rights of way, mineral rights, drilling rights, and other rights, reservations, covenants, conditions, or restrictions which exist of record as of the date of this Master Site Lease and which will not materially impair the use of the Sites;
- (3) easements, rights of way, mineral rights, drilling rights and other rights, reservations, covenants, conditions, or restrictions established following the date of recordation of this Master Site Lease and to which the Lessee and the District consent in writing which will not impair or impede the operation of the Sites.

5. **REPRESENTATIONS AND WARRANTIES OF THE LESSEE.** The Lessee represents and warrants to the District that:

- A. The Lessee is duly organized in the State of California, and in good standing under the laws of the State of California, with full corporate power and authority to lease and own real and personal property;
- B. The Lessee has full power, authority and legal right to enter into and perform its obligations under this Master Site Lease, and the execution, delivery and performance of this Master Site Lease has been duly authorized by all necessary corporate actions on the part of the Lessee and does not require any further approvals or consents;
- C. Execution, delivery and performance of this Master Site Lease does not and will not result in any breach of or constitute a default under any indenture, mortgage, contract, agreement or instrument to which the Lessee is a party or by which it or its property is bound;
- D. There is no pending or, to the best knowledge of the Lessee, threatened action or proceeding before any court or administrative agency which will materially adversely affect the ability of the Lessee to perform its obligations under this Master Site Lease; and

6.	RENTAL.

The Lessee shall pay to the District as and for advance rental hereunder \$1.00 per year or part thereof, on or before the date of commencement of the Term of this Master Site Lease. The Lessee shall have no obligation to make rental payments hereunder in the event the Commencement Date of this Master Site Lease does not occur as a result of the District's inability to issue a Notice to Proceed for the each \_\_\_\_\_\_\_ Project pursuant to the provisions of the Master Construction Services Agreement.

#### 7. PURPOSE.

The Lessee shall use the Sites solely for the purpose of constructing the Projects thereon and for subleasing the Sites and the Projects to the District; provided, that upon the occurrence of an Event of Default by the District under the Master Sublease, the Lessee may exercise the remedies provided for in the Master Construction Services Agreement or the Master Sublease.



- 8. <u>TERMINATION</u>. The Lessee agrees, upon termination of this Master Site Lease or the end of the Term of this Master Site Lease:
  - A. To quit and surrender the Sites in the same good order and condition as it was in at the time of commencement of the Term hereunder, reasonable wear and tear excepted;
  - B. To release and re-convey to the District any liens and encumbrances created or caused by the Lessee; and
  - C. That any permanent improvements and structures existing upon the Sites at the time of the termination of this Master Site Lease shall remain thereon and title thereto shall vest in the District.

Notwithstanding the District's foregoing rights in the event of termination, the Lessee shall retain the right to full compensation for all services rendered prior to the termination, including all rights they have under the Master Construction Services Agreement and the Master Sublease as well as all recourse provided by California law including common law, for the value of the work performed on the Site and/or the Project.

In the event the Master Construction Services Agreement is terminated pursuant to the provisions therein, this Master Site Lease shall immediately terminate.

#### 9. **QUIET ENJOYMENT**.

Subject to the terms of the Master Sublease attached hereto as Exhibit "B", the District covenants and agrees that it will not take any action to prevent the Lessee's quiet enjoyment of the Sites during the Term hereof; and, that in the event District's fee title to the Sites is ever challenged so as to interfere with the Lessee's right to occupy, use and enjoy the Sites, the District will use all governmental powers at its disposal, including the power of eminent domain, to obtain unencumbered fee title to the Sites and to defend the Lessee's right to occupy, use, and enjoy the Sites. The District, however, retains the right, throughout the Master Site Lease Term, to use the Sites for District purposes, pursuant to the terms of the Master Sublease.

#### 10. NO LIENS.

The District shall not mortgage, sell, assign, transfer or convey the Sites or any part thereof to any person during the Term of this Master Site Lease, without the written consent of the Lessee. Nothing herein shall preclude the District from granting utility easements across the Site to facilitate the use and operation of the Project for which it is intended.

#### 11. RIGHT OF ENTRY.

The District reserves the right for any of its duly authorized representatives to use the Project during the Term of this Master Site Lease or Master Sublease and enter upon the Site at any reasonable time to inspect the same or to make any repairs, improvements or changes necessary for the preservation thereof, but in doing so shall not interfere with the Lessee's operations on the Project.

#### 12. **ASSIGNMENT AND SUBLEASING.**

The Lessee will not assign or otherwise dispose of or encumber any of the Sites or this Master Site Lease without the written consent of the District.



#### 13. NO WASTE.

The Lessee agrees that at all times that it is in possession of the Site it will not commit suffer or permit any waste on the Site, and it will not willfully or knowingly use or permit the use of the Site for any illegal act or purpose.

#### 14. **DEFAULT**.

In the event the Lessee shall be in default in the performance of any obligation on its part to be performed under the terms of the Master Construction Services Agreement and this Master Site Lease, which default continues for thirty (30) days following notice and demand for correction thereof to the Lessee, the District may exercise any and all remedies granted by law, except that no merger of this Master Site Lease and of the Master Sublease shall be deemed to occur as a result thereof.

#### 15. **TITLE**.

During the Term of this Master Site Lease, the District shall hold title to the Sites and obtain title to the Project from the Lessee, including any and all additions which comprise improvements, fixtures, repairs, replacements or modifications, as such improvements are built and paid for pursuant to the Master Construction Services Agreement with full title vesting in the District to all improvements upon the end of the Term of this Master Site Lease.

#### 16. **TAXES**.

The terms of this Master Site Lease may result in the creation of a possessory interest. If such a possessory interest is vested in a private party to this document, the private party may be subjected to the payment of personal property taxes levied on such interest. Pursuant to Section 107.6 of the California Revenue and Taxation Code, District hereby notifies Lessee that: (i) the property interest obtained by Lessee pursuant to the Master Site Lease may be subject to property taxation; and (ii) Lessee may be subject to the payment of property taxes levied on the property interest obtained by Lessee.

#### 17. **EMINENT DOMAIN**.

In the event the whole or any part of the Sites or the improvements thereon, including but not limited to the Project, is taken by eminent domain, the financial interest of the Lessee shall be recognized and is hereby determined to be the amount of all Sublease Payments and Retention Payment, as applicable, then due or past due, less any allowed withholdings or offsets, and unearned interest as of the date the Lessee receives payment in full. The balance of the award in such eminent domain action, if any, shall be paid to the District.

### 18. <u>LIQUIDATED DAMAGES</u>.

Pursuant to Lessee's Due Diligence, as further described in Article 5 of the Master Construction Services Agreement, Lessee has determined the Term of this Master Site Lease which shall extend at least until the Punch List is completed under Article 13 of the Master Construction Services Agreement. Pursuant to the Master Construction Services Agreement, Liquidated Damages shall apply if the Contract Time is exceeded.

#### 19. **PARTIAL INVALIDITY**.



If any one or more of the terms, covenants or conditions or this Master Site Lease shall to any extent be declared invalid, unenforceable, void or voidable for any reason whatsoever by a court of competent jurisdiction, the finding or order or decree of which becomes final, none of the remaining terms, provisions, covenants and conditions of this Master Site Lease shall be affected thereby, and each provision of this Master Site Lease shall be valid and enforceable to the fullest extent permitted by law.

#### 20. **NOTICES**.

Any notices or filings required to be given or made under this Master Site Lease shall be served, given or made in writing upon the District or the Lessee, as the case may be, by personal delivery or registered mail to the respective addresses given below. Any change in the addresses noted shall not be binding upon the other party unless preceded by no less than thirty (30) days prior written notice. Any such notices shall be deemed to have been received by the addressee if delivered to the person for whom they are intended or if sent by registered mail, return receipt requested, or by email, or fax followed by regular mail, addressed as follows:

If to Lessee:		
	Attn:	
	Email:	

If to District: San Bernardino City Unified School District

956 W. 9th Street

San Bernardino, CA 92411

Attn: Tom Pace

Email: tom.pace@sbcusd.k12.ca.us

#### 21. BINDING EFFECT.

This Master Site Lease shall inure to the benefit of and shall be binding upon the District, the Lessee and its respective successors in interest and assigns.

### 22. <u>AMENDMENTS AND MODIFICATIONS</u>.

This Master Site Lease shall not be effectively amended, changed, modified, altered or terminated without the written agreement of the District and the Lessee.

# 23. **EXECUTION IN COUNTERPARTS**.

This Master Site Lease may be simultaneously executed in several counterparts, each of which shall be an original and all of which shall constitute but one and the same instrument.

### 24. LAWS, VENUE AND ATTORNEYS' FEES.

The terms and provisions of this Master Site Lease shall be construed in accordance with the laws of the State of California. If any action is brought in a court of law to enforce any term of this Master Site Lease, the action shall be brought in a state court situated in the County of San Bernardino, State of California, unless a court finds jurisdiction or venue is only proper in



a federal court, or a court outside this county. In the event of any such litigation between the parties, the parties shall pay for their respective costs incurred, including attorneys' fees.

## 25. <u>INTEGRATION/MODIFICATION</u>.

This Master Site Lease represents the entire understanding of the District and Lessee as to those matters contained herein, and supersedes and cancels any prior oral or written understanding, promises or representations with respect to those matters covered herein and shall not be amended, altered, or changed except by a written agreement signed by the parties hereto.

### 26. **HEADINGS**.

The captions or headings in this Master Site Lease are for convenience only and in no way define, limit or describe the scope or intent of any provisions or sections of this Master Site Lease.

#### 27. **TIME.**

Time is of the essence in this Master Site Lease and each and all of its provisions.

IN WITNESS WHEREOF, the parties hereto have executed this Master Site Lease by their authorized officers as of the day and year first written above.

"DISTRICT"	"LESSEE"
SAN BERNARDINO CITY UNIFIED SCHOOL DISTRICT	[CONTRACTOR]
BY:	BY:
Name:	Name:
Title:	Title:
Date:	Date:



# **EXHIBIT "A"**

# **DESCRIPTION OF SITES**



## **EXHIBIT "B"**

## **SUBLEASE**

(SEE RFP NO.231 ATTACHMENT 3B)

## **EXHIBIT "C"**

## MASTER CONSTRUCTION SERVICES AGREEMENT

[TO BE INSERTED]



# **RFP Attachment 3B**

SAN BERNA	ARDINO CITY UNIFIED SCHOOL DISTRICT
	PROJECT

**MASTER SUBLEASE AGREEMENT** 

# Between

## SAN BERNARDINO CITY UNIFIED SCHOOL DISTRICT

and

Dated as of \_\_\_\_\_, 2019



#### SAN BERNARDINO CITY UNIFIED SCHOOL DISTRICT

	PROJECT
N	MASTER SUBLEASE AGREEMENT

This MASTER SUBLEASE AGREEMENT ("Master Sublease") is dated as of \_\_\_\_\_\_, 2019, and is by and between the San Bernardino City Unified School District, a school district duly organized and existing under the laws of the State of California ("District"), and \_\_\_\_\_\_\_, a California corporation and operating under the laws of the State of California ("Lessor" or "Contractor").

## **RECITALS:**

WHEREAS, the District deems it essential for its own governmental purpose, to finance the construction of certain improvements as described \_\_\_\_\_\_\_ in Exhibit "A" attached hereto (the "Projects") and situated on the [insert school sites] sites described in Exhibit "B" attached hereto (the "Site"); and

WHEREAS, pursuant to Section 17400 *et seq.* of the Education Code, the District may enter into leases and agreements relating to real property and buildings used by the District; and

WHEREAS, pursuant to Section 17406 of the Education Code, the District is leasing the Sites to Lessor under a lease agreement dated the date hereof (the "Master Site Lease") for the purpose of Lessor constructing improvements on the Sites during the Term of the Master Site Lease on the terms and conditions the District finds to be in its best interest and set forth in this Master Sublease and the Master Construction Services Agreement attached as Exhibit "C" to the Master Site Lease; and

WHEREAS, the District owns the Sites, and pursuant to the Master Construction Services Agreement, will prepare and adopt plans and specifications for the completion of improvements, which will be approved pursuant to law as required by Section 17402 of the Education Code prior to the start of any construction for any project; and

WHEREAS, the District and Lessor agree to mutually cooperate now or hereafter, to the extent possible, in order to sustain the intent of this Master Sublease and the bargain of both parties hereto, and to provide Sublease Payments to be made on the dates and in the amount set forth herein.

#### WITNESSETH:

In consideration of the mutual covenants hereinafter set forth, the District and Lessor parties hereto agree as follows:

- 1. **<u>DEFINITIONS.</u>** Unless the context otherwise requires, the terms defined in this Article shall, for all purposes of this Master Sublease, have the meanings as herein specified.
  - A. <u>"Commencement Date"</u> shall mean the Project commencement date found in the Notice to Proceed for each Project in accordance with the Master Construction Services Agreement.
  - B. <u>"Construction Costs"</u> means any and all costs incurred by the Lessor with respect to the construction and equipping, as the case may be, of the Project, whether paid or incurred prior to or after the date hereof, including, without limitation, costs for Site preparation, the removal or demolition of existing structures, the construction of the Project and related facilities and improvements, and all other work in connection therewith, security of the Site



and Project, Lessor's overhead and supervision at the Project Site, all costs and expenses including any taxes or insurance premiums paid by the Lessor with respect to the Property, and administrative and other expenses necessary or incident to the Project, excluding Lessor's and Developer's home office overhead and profit. The term "Construction Costs" includes all Lessor's costs associated with preparing or generating additional copies of any Construction Documents, as defined below, related to or required for the Project, including preparation or generation of additional plans and specifications for Lessor's subcontractors. In no event shall Construction Costs exceed the Guaranteed Maximum Price.

- C. <u>"Master Construction Services Agreement"</u> (CSA) means the Master Construction Services Agreement attached hereto, together with any duly authorized and executed amendments hereto.
- D. "Construction Documents" consist of the Plans and Specifications approved by DSA for each project, Allowances stipulated in the Contract Documents, and all Addenda, if any, issued prior to the entry into this Agreement. The Construction Documents shall include all Modifications generated after the Effective Date in accordance with the Contract Documents, including, without limitation, a written amendment to the Contract signed by the Contractor and duly executed and approved by the District, a Change Order, a Construction Change Document, or a written order for a minor change in the Work issued by the Architect.
- E. "Contract Documents" means those documents which form the entire Contract by and between District and Contractor. As of the effective date of the Master Site Lease and Master Sublease, the Contract Documents consist of the Master Site Lease, the Master Sublease, any General, Supplementary and other Conditions, the Master Construction Services Agreement, including all exhibits and attachments hereto, and the Construction Documents. The Contract Documents collectively form the Contract. The Contract represents the entire and integrated Agreement between the parties hereto and supersedes prior negotiations, representations, or agreements, either written or oral. The Contract may be amended or modified only by a written Modification. The Contract Documents shall be binding solely upon the District and Contractor, do not create a contractual relationship of any kind between the Architect and Contractor, between the District and any Subcontractor or Sub-subcontractor, or between any persons or entities other than the District and the Contractor, and are not intended to and do not create any third party beneficiary. The Architect shall, however, be entitled to performance and enforcement of obligations under the Contract intended to facilitate performance of the Architect's duties. (See Article 14 of the CSA).
- F. "Day" means a calendar day unless specifically designated as a business day.
- G. <u>"District"</u> means the San Bernardino City Unified School District, a school district duly organized and existing under the laws of the State of California.
- H. <u>"Effective Date"</u> is the latter of the date upon which the District Board approves the Master Site Lease and the Master Sublease and Contractor has executed the Master Site Lease and Master Sublease.
- "Event of Default" means one or more events of default as defined in Article 16 of this Master Sublease.



- J. <u>"Guaranteed Maximum Price" or "GMP"</u> means the Guaranteed Maximum Price established pursuant to Article 5 of the CSA to be paid to Lessor for Lessor's construction of the Project hereunder, subject to any adjustments for Extra Work/Modifications as provided in Article 17 of the CSA.
- K. <u>"Lessor"</u> shall mean \_\_\_\_\_, and its successors and assigns.
- L. <u>"Project"</u> means the improvements and related work to be constructed and installed by the Lessor, as more particularly described and/or referenced in Exhibit "A" attached hereto.
- M. <u>"Site"</u> refers to the grounds of the Projects or in some cases may refer to multiple sites as defined in the Contract Documents and such adjacent lands as may be directly affected by the performance of the Work, particularly described in Exhibit "B" attached hereto.
- N. <u>"Master Site Lease" or "Lease"</u> means the Master Site Lease of even date herewith, by and between the District and the Lessor together with any duly authorized and executed amendment thereto under which the District leases the Sites to the Lessor.
- O. <u>"Master Sublease"</u> means this Master Sublease together with any duly authorized and executed amendment hereto.
- P. <u>"Sublease Payment"</u> means any payment required to be made by the District pursuant to Article 7 of this Sublease.
- Q. <u>"Term of this Master Sublease" or "Term"</u> means the time during which this Master Sublease is in effect, as provided for in Article 3 of this Master Sublease.

#### 2. MASTER SUBLEASE.

Lessor hereby leases and subleases to District, and District hereby leases and subleases from Lessor the Projects and the Sites, including any real property improvements now or hereafter affixed thereto in accordance with the provisions herein for the full Term of this Master Sublease. The leasing by the Lessor to the District of the Sites shall not effect or result in a merger of the District's leasehold estate pursuant to this Master Sublease and its fee estate as lessor under the Master Site Lease, and the Lessor shall continue to have and hold a leasehold estate in said Sites pursuant to the Master Site Lease throughout the Term thereof and the Term of this Master Sublease.

## 3. TERM OF THE MASTER SUBLEASE.

A. The Term of this Master Sublease shall become effective upon the authorized execution of this Master Sublease and issuance of a Notice to Proceed for any \_\_\_\_\_\_ Project under the terms of the CSA and payment of the last Sublease Payment, unless otherwise terminated pursuant to this Master Sublease, the Master Site Lease, or the CSA.



- B. Termination of Term. Except as otherwise provided, the Term of this Master Sublease shall terminate upon the earliest of any of the following events:
  - (1) An Event of Default and the Lessor's election to terminate this Master Sublease pursuant to the provisions of Articles 16 and 17, hereof;
  - (2) The arrival of the last day of the Term of this Master Sublease and payment of all Sublease Payments hereunder; or
  - (3) The exercise of the District's option under Article 21 hereof.
- 4. REPRESENTATIONS, WARRANTIES AND COVENANTS OF DISTRICT. The District represents and warrants to Lessor that:
  - A. District is a public school district, duly organized and existing under the Constitution and laws of the State of California with authority to enter into this Master Sublease and to perform all of its obligations hereunder;
  - B. District's governing body has duly authorized the execution and delivery of this Master Sublease and further represents and warrants that all requirements have been met and procedures followed to ensure its enforceability;
  - C. The execution, delivery and performance of this Master Sublease does not and will not result in any breach of or constitute a default under any indenture, mortgage, contract, agreement or instrument to which District is a party by which it or its property is bound;
  - D. There is no pending or, to the knowledge of District, threatened action or proceeding before any court or administrative agency which will materially adversely affect the ability of District to perform its obligations under this Master Sublease;
  - E. The Projects and the Sites are essential to District in the performance of its governmental functions and their estimated useful life to the District exceeds the Term of this Master Sublease;
  - F. District shall take such action as may be necessary to include all Sublease Payments in its annual budget and annually to appropriate an amount necessary to make such Sublease Payments;
  - G. District shall not abandon the Sites for the use for which it is currently required by District and, to the extent permitted by law, District shall not seek to substitute or acquire property to be used as a substitute for the uses for which the site is maintained under the Master Sublease; and
  - H. District shall not allow any Hazardous Substances (as such term is defined in the Site Lease and limited by that which shall be disclosed by the District prior to the Project commencement date in the Notice to Proceed) to be used or stored on, under or about the Sites.
- 5. **REPRESENTATIONS AND WARRANTIES OF LESSOR.** Lessor represents and warrants to District that:
  - A. Lessor is duly organized in the State of California, and in good standing as a corporation under the laws of the State of California, with full corporate power and authority to lease and own real and personal property;



- B. Lessor has full power, authority and legal right to enter into and perform its obligations under this Master Sublease, and the execution, delivery and performance of this Master Sublease has been duly authorized by all necessary corporate actions on the part of Lessor and does not require any further approvals or consents;
- C. The execution, delivery and performance of this Master Sublease does not and will not result in any breach of or constitute a default under any indenture, mortgage, contract, agreement or instrument to which Lessor is a party by which they or their property is bound;
- D. There is no pending or, to the knowledge of Lessor, threatened action or proceeding before any court or administrative agency which will materially adversely affect the ability of Lessor to perform their obligations under this Master Sublease; and
- E. Lessor will not mortgage or encumber the Sites or the Master Sublease or assign this Master Sublease or their rights to receive Sublease Payments hereunder, except as permitted herein.

#### 6. **APPROPRIATION OF FUNDS.**

A. In order to ensure that moneys sufficient to pay all costs will be available for this purpose when required, District shall maintain on deposit, and shall annually appropriate funds sufficient to make all Sublease Payments which become due to Lessor under this Master Sublease Agreement.

## 7. **SUBLEASE PAYMENTS**.

- A. District shall pay Lessor sublease payments (the "Sublease Payments") for the improvements, use and occupancy of the various Projects and Sites. The obligation of the District to pay Sublease Payments hereunder shall constitute a current expense of the District and shall not in any way be construed to be a debt of the District in contravention of any applicable constitutional or statutory limitations or requirements concerning the creation of indebtedness by the District, nor shall anything contained herein constitute a pledge of the general tax revenues, funds or moneys of the District. The Sublease Payments, which the parties acknowledge and agree, are good and sufficient consideration for the improvements and the District's use and occupancy of the Project and the Site.
- B. The District shall pay Lessor the portion of the GMP in accordance with the CSA. No Sublease Payment shall be made by the District in an amount that exceeds the aggregate cost approved in accordance with the CSA to the Lessor of the work on the Project completed to the date the Lessor submits an application for payment, less the aggregate amount of all Sublease Payments previously made by the District to the Lessor.
- C. In the event the District elects to exercise its option under Article 21.B below, the District's obligations under this Sublease including, but not limited to, the District's obligations to make Sublease Payments under this Article, shall thereupon cease and terminate.
- D. Except as specifically provided in this Article and in Article 9 hereof or as otherwise provided by law, the obligation of the District to make Sublease Payments when due and payable hereunder will be absolute and unconditional in all events and will not be subject to any set-off, defense, counterclaim, abatement or recoupment for any reason whatsoever.



#### 8. **FAIR RENTAL VALUE**.

Sublease Payments shall be paid by District in consideration of the right of possession of, and the continued quiet use and enjoyment of, the Projects and the Sites during the Term, as well as payment for any tenant improvements made by the Lessor which title to the tenant improvements shall vest progressively in the District as such tenant improvements are built and paid for pursuant to the Master Construction Services Agreement. Full ownership of each Project shall occur at the end of the Term of this Master Sublease and payment of any amounts owed under this Master Sublease, unless this Master Sublease, the Master Site Lease or Master Construction Services Agreement is terminated in accordance with their respective terms and conditions. The parties hereto have agreed and determined that such total rental is not in excess of the fair rental value of the Projects and the Sites. In making such determination, consideration has been given to the fair market value of the Projects and the Sites, that title to the improvements completed and paid for by District as to which the District shall have the right to possess, occupy and use, the uses and purposes which may be served by the Projects and the Sites and the benefits therefrom which will accrue to the District and the general public, the ability of the District to make additions, and modifications and improvements to the Projects and the Sites which are not inconsistent with the Master Construction Services Agreement (Exhibit "C" to Master Site Lease) and which do not interfere with the Lessor's work on the Projects and the Sites.

#### 9. **SUBLEASE ABATEMENT**.

In addition to delay of Sublease Payments provided in Article 7, above, Sublease Payments due hereunder with respect to the Projects and the Sites shall be subject to abatement prior to the commencement of the use of the Projects and the Sites by the District or during any period in which, by reason of material damage to or destruction of the Projects or the Sites, there is substantial interference with the use and right of possession by the District of the Projects and the Sites or any substantial portion thereof. For each potential incident of substantial interference, decisions to be made on i) whether or not abatement shall apply; ii) the date upon which abatement shall commence; iii) the applicable portion of Sublease Payments to be abated and; iv) the concluding date of the particular abatement shall all be subject to determinations by the District. The amount of Sublease abatement shall be such that the Sublease Payments paid by the District during the period of Projects and Sites restoration do not exceed the fair rental value of the usable portions of the Projects and Sites. In the event of any damage or destruction to the Projects or the Sites, this Master Sublease shall continue in full force and effect.

#### 10. USE OF SITES AND PROJECTS.

Subject to reasonable interference from construction operations by the Lessor under the terms of the Master Construction Services Agreement during the Term of this Master Sublease, Lessor shall provide the District with quiet use and enjoyment of the Sites without suit, or hindrance from Lessor or their assigns, provided District is in compliance with its duties under this Master Sublease. District will not use, operate or maintain the Sites or Projects improperly, carelessly, in violation of any applicable law or in a manner contrary to that contemplated by this Master Sublease. District shall provide all permits and licenses, if any, necessary for the operation of the Projects and Sites. In addition, the District agrees to comply in all respects (including, without limitation, with respect to the time, maintenance and operation of the Projects and Sites) with laws of all jurisdictions in which its operations involving the Projects and Sites may extend and any legislative, executive, administrative or judicial body exercising any power or jurisdiction over the Sites or the Projects; provided, however, that District may contest in good faith the validity or application of any such law or



rule in any reasonable manner which does not, in the opinion of Lessor, adversely affect the estate of Lessor in and to the Sites or the Projects or its interest or rights under this Master Sublease. Lessor acknowledges that at any time during the Term of this Master Sublease, District may access the Site to conduct District business. Lessor acknowledges and agrees to the District's use or occupation of the Sites, so long as such use or occupation does not unreasonably interfere with construction of the Projects. Upon substantial completion of the Projects or severable portions hereof, the Lessor shall provide the District with quiet use and enjoyment of the Sites without suit or hindrance from the Lessor or its assigns, subject to reasonable interference from ongoing construction operations on any remaining portion of the Sites under construction by the Lessor. Notwithstanding any provision to the contrary in this Master Sublease or the Master Construction Services Agreement, the District shall, concurrent with any occupancy, use or possession of any portion of the Projects, furnish property and loss liability insurance to cover any such portion of the Projects or Sites it occupies, uses or possesses. District shall provide certificates of insurance and additional insured endorsement naming Lessor.

## 11. <u>LESSOR'S INSPECTION/ACCESS TO THE SITE</u>.

District agrees that Lessor and any of Lessor's representatives shall have the right at all reasonable times to enter upon the Sites or any portion thereof to construct and improve the Project, to examine and inspect the Sites or Projects, to make repairs or service warranty obligations, and to exercise its remedies pursuant to the section in this Master Sublease entitled "Remedies on Default." District further agrees that Lessor and any of Lessor's representatives shall have such rights of access to the Sites as may be reasonably necessary to cause the proper maintenance of the Sites and the Projects in the event of failure by District to perform its obligations hereunder.

## 12. **PROJECT ACCEPTANCE**.

District shall acknowledge final inspection and completion of the Projects by executing and recording a Notice of Completion. The validity of this Master Sublease will not be affected by any delay in or failure of completion of the Projects.

ALTERATIONS AND ATTACHMENTS. All permanent additions and improvements that are made to the Projects shall belong to and become the property of Lessor, subject to the provisions of this Master Sublease. Separately identifiable attachments added to the Projects by the District shall remain the property of the District. At Lessor's request, the District agrees to remove the attachments and restore the Projects to substantially as good a condition as when acquired and constructed, normal wear and tear excepted, in the event of failure by the District to perform its obligations hereunder.

#### 14. **MAINTENANCE AND UTILITIES**.

Until the date of beneficial occupancy by the District of each Project and Substantial Completion of each Project as defined in the CSA, Lessor shall, in its own name, contract for and pay the expenses of all utility services required for each Project and Substantial Completion of each Project, the District shall, in its own name, contract for and pay the expenses of all utility services including, but not limited to, all air conditioning, heating, electrical, gas, refuse collection, water, and sewer units. The District shall be responsible for all utilities and maintenance of only the portion of the Sites occupied solely or beneficially by the District during construction of the Projects by Lessor. Once the Projects are accepted by the District as finally complete, the District shall have

13.



responsibility for maintenance and repair of each \_\_\_\_\_\_ Project and the Sites, except for warranty or other obligations of Lessor relating to the improvements as set forth in the Master Construction Services Agreement.

## 15. **TAXES**.

16.

District shall keep the Projects and the Sites free and clear of all levies, liens, and encumbrances and shall pay all license fees, registration fees, assessments, charges, and taxes (municipal, state, and federal) if applicable, which may now or hereafter be imposed upon the ownership, leasing, renting, sale, possession, or use of the Projects and the Sites, excluding, however, all taxes on or measured by Lessor's income.

- **EVENTS OF DEFAULT.** The term "Event of Default," as used in this Master Sublease means the occurrence of any one or more of the following events:
- A. The District fails to make any unexcused Sublease Payment (or any other payment) within fifteen (15) days after the due date thereof or the District fails to perform or observe any other covenant, condition or agreement to be performed or observed by it hereunder and such failure to either make the payment or perform the covenant, condition or agreement is not cured within ten (10) days after written notice thereof by Lessor;
- B. The Lessor discovers that any statement, representation or warranty made by the District in this Master Sublease, or in any document ever delivered by the District pursuant hereto or in connection herewith is misleading or erroneous in any material respect;
- C. The District becomes insolvent, is unable to pay its debts as they become due, makes an assignment for the benefit of creditors, applies or consents to the appointment of a receiver, trustee, conservator or liquidator of the District or of all or a substantial part of its assets, or a petition for relief is filed by the District under federal bankruptcy, insolvency or similar laws.
- 17. REMEDIES ON DEFAULT. Upon the happening of any Event of Default, Lessor may exercise remedies set forth below; provided, however, that notwithstanding anything herein to the contrary, there shall be no right under any circumstances to accelerate the Sublease Payments or otherwise declare any Sublease Payments not then in default to be immediately due and payable. The District shall continue to remain liable for the payment of Sublease Payments and damages for breach of this Master Sublease and the performance of all conditions herein such Sublease Payments and damages shall be payable to Lessor at the time and in the manner set forth in subsections (A) and (B) of this Article:
  - A. In the event that Lessor does not elect to terminate this Sublease pursuant to subsection (B) below, the District agrees to and shall remain liable for the payment of Sublease Payments and the performance of all conditions herein and shall reimburse Lessor for the full amount of the Sublease Payments to the end of the Sublease Term.
  - B. In the event of termination of this Sublease by Lessor at its option and in the manner hereinafter provided on account of default by the District, the District shall pay Lessor Sublease Payments then owing for past Sublease Payments due and not paid, not to exceed the approved costs for all labor, materials and services provided up to the date of Lessor's termination of the Master Sublease. Neither notice to pay Sublease Payments, nor to deliver up possession of the Project and the Site given pursuant to law, nor any proceeding in unlawful detainer taken by Lessor shall of itself operate to terminate this Master Sublease. In the event of any litigation between the parties, the parties shall pay for their respective costs incurred, including attorneys' fees.



C. No right or remedy herein conferred upon or reserved to Lessor is exclusive of any other right or remedy herein, but each shall be cumulative of every other right or remedy given hereunder or now or hereafter existing at law or in equity or by statute or otherwise, and may be enforced concurrently therewith or from time to time; provided, however, that notwithstanding any provisions to the contrary herein, Lessor shall not under any circumstances have the right to accelerate the Sublease Payments that fall due in future Sublease periods or otherwise declare any Sublease Payments not then in default to be immediately due and payable.

#### 18. **NON-WAIVER**.

No covenant or condition to be performed by District or Lessor under this Master Sublease can be waived except by the written consent of the other party. Forbearance or indulgence by District or Lessor in any regard whatsoever shall not constitute a waiver of the covenant or condition in question. Until complete performance by the District or Lessor of said covenant or condition, the other party shall be entitled to invoke any remedy available to it under this Master Sublease or by law or in equity despite said forbearance or indulgence.

#### 19. **ASSIGNMENT**.

Without the prior written consent of the Lessor, which consent shall not be unreasonably withheld, the District shall not (a) assign, transfer, pledge, or hypothecate this Master Sublease, the Project and the Site, or any part thereof, or any interest therein, or (b) sublet or lend the use of the Project or any part thereof, except as authorized by the provisions of the California Civic Center Act, Education Code section 38130 et seq. However, District may lease, license or otherwise allow use or occupation of the Sites for third party use so long as such use or occupation does not unreasonably interfere with construction of the Project. Consent to any of the foregoing prohibited acts applies only in the given instance and is not a consent to any subsequent like act by the District or any other person. The Lessor shall not assign its obligations under this Master Sublease with the exception of their obligation to issue default notices and to convey or re-convey their interest in the Project and Site to the District upon full satisfaction of the District's obligations hereunder; however, the Lessor may assign their right, title and interest in this Master Sublease, the Sublease Payments and other amounts due hereunder and the Project in whole or in part to one or more assignees or sub-assignees at any time upon written notice to the District. No assignment shall be effective as against the District unless and until the District is so notified in writing. The District shall pay all Sublease Payments due hereunder pursuant to the direction of Lessor or the assignee named in the most recent assignment or notice of assignment. During the Sublease Term, the District shall keep a complete and accurate record of all such assignments. Subject always to the foregoing, this Master Sublease inures to the benefit of, and is binding upon, the heirs, legatees, personal representatives, successors, and assigns of the parties hereto.

## 20. **OWNERSHIP**.

During the Term of this Master Sublease, the District shall hold title to the Sites and progressively obtain title to the Projects from the Lessor, and any and all additions which comprise fixtures, repairs, replacements or modifications thereof, as Sublease Payments are made to Lessor. During the Term of this Master Sublease, the Lessor shall have a leasehold interest in the Sites pursuant to the Master Site Lease. If the District prepays the Sublease Payments in full pursuant to Article 21 hereof or otherwise pays all required Sublease Payments, all remaining rights, title and interests of the Lessor, if any, in and to the Project and Sites, shall be fully transferred to and vested in the District. Title shall be transferred to and vested in the District hereunder without the necessity for any further instrument of



transfer. At the termination of this Master Sublease, title to the Site, and any improvements constructed thereon shall vest in the District.

## 21. SUBLEASE PREPAYMENT/ PURCHASE OPTION.

- A. <u>Sublease Prepayments</u>. At any time during the term of this Master Sublease, the District may in its sole discretion, upon the request of the Lessor or on upon its own initiative, make Sublease Prepayments to the Lessor. No Sublease Prepayments requested by the Lessor may be made by the District in an amount exceeding the aggregate true cost to the Lessor of the work on the Project completed to the date the Lessor submits the request for a Sublease Prepayment less the aggregate amount of: (1) all Sublease Payments previously made by the District to the Lessor; (2) all Sublease Prepayments previously made by the District to the Lessor; (3) all amounts previously retained pursuant to Article 21(A)(3), below, from Sublease Prepayments previously made by the District to the Lessor; and (4) the retention for such Sublease Prepayment. Lessor must submit evidence that the conditions precedent set forth in Article 21(A)(1) below, have been met. In the event District elects to make Sublease Prepayments, the Prepayment Price, contemplated in Article 21(B), below, shall be adjusted accordingly.
  - (1) The following are conditions precedent to any Sublease Prepayments made to the Lessor pursuant to a request of the Lessor and exercised by the District in its sole discretion:
    - a. Satisfactory progress of the work and construction pursuant to the approved schedule and "Contract Time" pursuant to Article 9 of the Master Construction Services Agreement shall have been made as determined in Article 21(A)(2), below.
    - b. Lessor shall also submit to the District (i) duly executed conditional lien releases and waivers (in the form provided in California Civil Code section 8132) from the Lessor and all subcontractors, consultants and other persons retained by the Lessor in connection with the Project, whereby such persons conditionally waive all lien and stop notice rights against the District, the Project and the Project site with respect to the pending Sublease Prepayment to be made by the District, (ii) duly executed unconditional lien releases and waivers (in the form provided in California Civil Code section 8134) from the Lessor and all subcontractors, consultants and other persons retained by the Lessor in connection with the Project, whereby such persons unconditionally and irrevocably waive all lien and stop notice rights against the District, the Project and the Project site with respect to all previous Sublease Prepayments made by the District, and (iii) any other items that the Lessor may be required to collect and distribute to the District pursuant to the terms and provisions of the CSA. Lessor shall promptly pay all amounts due to each subcontractor, consultant and other person retained by Lessor in connection with the Project no later than ten (10) days after Lessor's receipt of a Sublease Prepayment from the District.
  - (2) The determination of whether satisfactory progress of the Construction pursuant to the approved schedule and "Contract Time" has occurred shall be made by the Project Inspector hired by the District pursuant to Article 10 of the CSA. If the Project Inspector determines that pursuant to the approved schedule and "Contract Time", the work required to be performed, as stated in the Lessor's Sublease Prepayment



request has not been completed and approved, the Lessor shall not be eligible to receive the requested Sublease Prepayment.

B. <u>Purchase Option</u>. If the District is not in default hereunder, the District shall be granted options to purchase not less than all of the Project in as-is condition. The Prepayment Price at any given time shall be an amount equal to the GMP, as it may be revised from time to time, less the sum of any Sublease Payments and/or Sublease Prepayments made by the District prior to the date on which the District elects to exercise its option under this Article.

#### 22. RELEASE OF LIENS.

- A. Notwithstanding Article 21 above, upon Substantial Completion of the various

  Projects as defined in the CSA and the recording of a Notice of Completion for the various

  Projects, Lessor or its assignee and the District shall release Lessor's leasehold interest in Projects and the Sites. However, District shall retain any and all claims and or warranties it may have under the CSA.
- B. Lessor shall authorize, execute and deliver to the District all documents reasonably requested by the District to evidence (i) the release of any and all liens created pursuant to the provisions of this Master Sublease and the Master Site Lease, and (ii) any other documents required to terminate the Master Site Lease and this Master Sublease.

#### 23. TERMINATION OF CONSTRUCTION SERVICES AGREEMENT.

In the event the Master Construction Services Agreement is terminated pursuant to the provisions contained therein, this Master Sublease shall immediately terminate.

### 24. **SEVERABILITY**.

If any provision of this Master Sublease shall be held invalid or unenforceable by a court of competent jurisdiction, such holdings shall not invalidate or render unenforceable any other provision of this Master Sublease, unless elimination of such provision materially alters the rights and obligations embodied in this Master Sublease.

## 25. <u>INTEGRATION/MODIFICATION</u>.

This Master Sublease constitutes the entire agreement between Lessor and the District as to those matters contained herein, and supersedes and cancels any prior oral or written understanding, promises or representations with respect to those matters covered herein, and it shall not be amended, altered, or changed except by a written agreement signed by the parties hereto.

#### 26. **NOTICES**.

Services of all notices under this Master Sublease shall be sufficient if given personally or mailed to the party involved at its respective address hereinafter set forth or at such address as such party may provide in writing from time to time. Any change in the addresses noted shall not be binding upon the other party unless preceded by no less than thirty (30) days prior written notice. Any such notices shall be deemed to have been received by the addressee if delivered to the person for whom they are intended or if sent by registered mail, return receipt requested, or by email, or fax followed by regular mail, addressed as follows:



	If to Lessor:	_(Name		
		_(Address)		
		_(City, State, Z	Zip)	
		Attn:		
		Email:		
	If to District:	San Bernardin 956 W. 9 <sup>th</sup> Stre San Bernardin		
		Attn: Tom Pac		
27.	TITLES.			
			s of this Master Sublease are solely for the c e interpretation thereof.	onvenience of
28.	TIME.			
	Time is of the essent	ce in this Master	er Sublease and each and all of its provisions	
29.	LAWS, VENUE AND	ATTORNEYS' FEE	<u>:ES</u> .	
	laws of the State of C this Master Sublease Bernardino, State of	California. If any e, the action shal f California, unle ourt outside this	laster Sublease shall be construed in accord y action is brought in a court of law to enforall be brought in a state court situated in the less a court finds jurisdiction or venue is on is county. In the event of any such litigation or attorney's fees.	ce any term of County of San aly proper in a
IN WITNESS as of the day and yea	•	es hereto have ex	executed this Master Sublease by their auth	orized officers
DISTRICT			"LESSOR"	
SAN BERNARDINO CI	TY UNIFIED SCHOOL [	DISTRICT	[CONTRACTOR]	
BY:				
			BY:	



## **SUBLEASE EXHIBIT A**

## **DESCRIPTION OF PROJECT**

[TO BE INSERTED]



## **SUBLEASE EXHIBIT B**

## **DESCRIPTION OF SITE**

[TO BE INSERTED



# **Division I Specifications**

# **TABLE OF CONTENTS**

SECTION 01 11 00 SUMMARY OF WORK	1
SECTION 01 11 14 - WORK SEQUENCE AND PHASING	10
SECTION 01 20 00 - PRICE AND PAYMENT PROCEDURES	16
SECTION 01 26 00 - CONTRACT MODIFICATION PROCEDURES	20
SECTION 01 31 00 - PROJECT COORDINATION AND MEETINGS	31
SECTION 01 31 14 LABOR COMPLIANCE PROGRAM (LCP) COORDINATION.	35
SECTION 01 32 16 - CONSTRUCTION PROGRESS SCHEDULE	39
SECTION 01 33 00 - SUBMITTAL PROCEDURES	45
SECTION 01 35 16 - ALTERATION PROJECT PROCEDURES	58
SECTION 01 45 00 - QUALITY CONTROL	61
SECTION 01 45 23 - TESTING AND INSPECTION	74
SECTION 01 52 00 - CONSTRUCTION FACILITIES	81
SECTION 01 57 00 - TEMPORARY CONTROLS	91
SECTION 01 57 13 - TEMPORARY EROSION AND SEDIMENT CONTROL	94
SECTION 01 60 00 - MATERIALS AND EQUIPMENT	96
SECTION 01 71 23 - FIELD ENGINEERING	103
SECTION 01 73 29 - CUTTING AND PATCHING	105
SECTION 01 77 00 - CLOSEOUT PROCEDURES	119
SECTION 01 78 36 - WARRANTIES AND BONDS	125
SECTION 01 78 39 - PROJECT RECORD DOCUMENTS	129

## **SECTION 01 11 00 SUMMARY OF WORK**

#### 1. PART 1 - GENERAL

#### 1.1. SECTION INCLUDES

- 1.1.1.Summary of the Work of these Contract Documents for the construction of: RFP No. 231 Pacific High School Career Technical Education Facility and Campus Modernization Phase 1.
- 1.1.2. Architect's Name: Ruhnau Clarke Architects

#### 1.2. GENERAL

- 1.2.1.Work under this Contract includes furnishing all labor, materials, services and transportation including both preconstruction and construction services as required for completion of the Project in accordance with the Contract Documents.
- 1.2.2. The Work includes both new construction and modernization efforts at the existing Pacific High School campus at 1020 Pacific Street, San Bernardino, CA.
  - 1.2.2.1. Construction of the new Career Technical Education Transportation Technologies (CTE) facility of approximately 21,000 sq. ft. and installation of all specified equipment,
  - 1.2.2.2. Modernization of various classroom buildings and supporting structures as detailed in contract documents. These modernizations involve approximately 78,492 sq. ft. of existing classrooms and other campus facilities,
  - 1.2.2.3. The addition of approximately 7,920 sq. ft. to the gymnasium,
  - 1.2.2.4. Construction of miscellaneous site improvements,
  - 1.2.2.5. Construction of various temporary facilities to support the project, including temporary fencing, temporary parking facilities for school staff, school visitors, and construction staff, rerouting of impacted irrigation lines, and temporary relocation of supporting systems (power, phones, data cabling, etc.)
- 1.2.3.The scope shall also include providing preconstruction services, including but not limited to planning of project phasing, logistics and construction sequence, review of design documents for constructability, value engineering, schedule development, sourcing of long lead items, prequalification of trade subcontractors and development of subcontractor bid packages, and development of Guaranteed Maximum Pricing.
- 1.2.4. The Contract Time for completion shall be that shown in the Construction Progress Schedule. Once the CONTRACTOR has received a notice to proceed, the CONTRACTOR shall complete the work within ## calendar months from receipt of the notice to proceed. It is expressly understood that time is of the essence.
- 1.2.5. The work is to be performed within a portion of an active school campus. All work shall be conducted in a manner that does not impact the health and safety of school staff, students, site workers and project personnel, adjacent property owners, and/or the

- general public. Contractor shall at all times employ safety practices and environmental controls which take into consideration the fact that work is being performed on an active school campus. All work shall be performed in a manner which maximizes safety.
- 1.2.6. Contract Drawings: The Drawings provided with and identified in the Project Manual are the Drawings referenced in the Agreement.
  - 1.2.6.1. The location, extent and configuration of the required construction and improvements are shown and noted on Drawings.
    - 1.2.6.1.1. The Drawings are referenced in the Agreement.
    - 1.2.6.1.2. An index of Drawings is included in the set of Drawings.
  - 1.2.6.2. Drawings are arranged into series according to design discipline. Such organization and all references to trades, subcontractor, specialty contractor or supplier shall not control the Contractor in dividing the Work among subcontractors or in establishing the extent of the Work to be performed by any trade.
  - Where the terms "as shown", "as indicated", "as noted", "as detailed", "as 1.2.6.3. scheduled", or terms of like meaning, are used in the Drawings or Specifications, it shall be understood that reference is being made to the Drawings referenced in the Agreement.
  - 1.2.6.4. Where reference to the word "plans" is made anywhere in Drawings, Specifications and related Contract Documents, it shall be understood to mean the Drawings referenced in the Agreement.
- 1.2.7.Contractor's Safety Performance: SBCUSD places safety and safe work practices at a premium, especially in regard to operations on active District campuses.
- 1.2.8. All work shall be performed in a manner that minimizes impact to the environment, minimizes waste and maximizes the amount of salvageable material recovered throughout the project(s).
- 1.2.9. All work shall be performed in a manner that minimizes noise and vibration impacts to the adjacent classroom buildings and school operations. In some cases, loud or high vibration activities may have to be rescheduled to accommodate school instructional and/or testing activities. Such activities may require work on weekends or during holiday breaks. Holiday break periods for the 2019-20 school year are as follows:

Spring Break: 3/25/19-4/5/19

Memorial Day: 5/27/19

Summer Break: Begins 6/10/19, Ends approx. 8/1/19 (TBD)

**Labor Day** 9/2/19 October 7th 10/7/19 Veteran's Day 11/11/19

Thanksgiving 11/25/19 - 11/29/19 Winter Break: 12/23/19-1/10/20

MLK Day: 1/20/20 Lincoln's Birthday: 2/10/20President's Day: 2/17/20

• Spring Break 3/23/20 – 4/3/20

• Memorial Day: 5/25/20

- 1.2.10. All work shall be performed in a manner that protects existing infrastructure, landscaping, and other structures designated to remain.
- 1.2.11. All work shall be performed in a manner that meets the District's expectation for safe work execution, as well as adherence to schedule and project budget.

#### 1.3. BACKGROUND

- 1.3.1. Career Technical Education Facility:
  - 1.3.1.1. The CTE site is located on the west side of the PHS campus, with the campus located in what is predominantly a residential community. The new CTE facility will be approximately 21,000 sq. ft.
- 1.3.2.PHS Campus Modernization
  - 1.3.2.1. The Modernization portion of the work will renovate 11 existing campus buildings, totaling approximately 78,492 sq. ft. A new addition to the gymnasium will add approximately 7,920 sq. ft. to that facility.
- 1.3.3.Temporary Facilities:
  - 1.3.3.1. The scope of work includes the necessary temporary construction required to relocate the classrooms, training room(s), offices and support spaces vacated by the modernization effort, as well as temporary and/or permanent installation of supporting systems (fire alarm, clock and bell systems, IT/data cabling and switches, security systems and public address system controls) to designated temporary spaces on campus.
  - 1.3.3.2. The scope shall also include temporary fencing and signage as required to designate staff and separate visitor parking areas, student drop off, as well as contractor's laydown areas. Areas shall be fenced and demarcated to safely route and protect student, staff, visitors, construction site personnel and the general public.
  - 1.3.3.3. During construction, Contractor shall provide temporary field office, IOR office, safety and restroom facilities to support the construction effort, in accordance with the Contract Documents.
  - 1.3.3.4. Temporary work includes fencing and signage as required to enclose worksite and to establish and designate staff and separate visitor parking areas, student drop off, as well as contractor's laydown areas. Areas shall be fenced and demarcated to safely route and protect student, staff, visitors, construction site personnel and the general public.

1.3.3.5. Temporary signage shall include providing and installing a 4' by 8' all-weather sign depicting project information, including project name, District information, contractor name and contact information, architect name, etc., in accordance with the Contract Documents.

#### 1.4. EXISTING CONDITIONS

Certain information relating to existing surface and subsurface conditions and structures is available to bidders but will not be part of the Contract Documents, as follows:

- 1.4.1.Geotechnical Report: (Converse)
  - 1.4.1.1. Original copy is available for inspection at District's offices during normal business hours.
  - 1.4.1.2. This report identifies properties of below grade conditions and offers recommendations for the design of foundations, prepared primarily for the use of Architect.
  - 1.4.1.3. Report may also include information on soil and subsurface investigations conducted at site by an independent testing laboratory and report with log of borings prepared.
  - 1.4.1.4. The District, Architect and Engineers disclaim all responsibility for the accuracy of information prepared by others.
  - 1.4.1.5. The District, Architect and Engineers disclaim all responsibility for the information to be completely representative of conditions and materials which may be encountered and as being adequate for the purposes of construction.
  - 1.4.1.6. Variations in kind, depth, quantity, and condition of soils may occur.
  - 1.4.1.7. The District, Architect and Engineers further disclaim responsibility for interpretation by Bidding Contractors and others of soil and subsurface investigation information, such as in projecting soil-bearing values, rock profiles, presence and scope of boulders and cobbles, soil stability and the presence, level and extent of underground water.
  - 1.4.1.8. The recommendations described shall not be construed as a requirement of this Contract, unless specifically referenced in the Contract Documents.
  - 1.4.1.9. This report, by its nature, cannot reveal all conditions that exist on the site. Should subsurface conditions be found to vary substantially from this report, changes in the design and construction of foundations will be made, with resulting credits or expenditures to the Contract Sum accruing to District.
  - 1.4.1.10. If variances from Geotechnical Report are found, make written report and timely submit to the District Representative. Note that District requires claims for additional compensation must be submitted in writing no more than 10 business days from the date contractor has notice of the claim. (Contract section 17.5.2.)
- 1.4.2. Hazardous Materials Survey Report: (Converse)

- 1.4.2.1. The recommendations described in the referenced report shall not be construed as a requirement of this Contract, unless specifically referenced in the Contract Documents.
- 1.4.2.2. This report, by its nature, cannot reveal all conditions that exist on the site.

  Should hidden conditions be found to vary substantially from this report, changes in work may be required, with resulting credits or expenditures to the Contract Sum accruing to District.
- 1.4.2.3. If variances from Hazardous Materials Survey Report are found, make written report and timely submit to the District Representative. Note that District requires claims for additional compensation must be submitted in writing no more than 10 business days from the date contractor has notice of the claim. (Contract section 17.5.2.)
- 1.4.2.4. Contractor shall be responsible proper removal and disposal of any/all hazardous materials designated in Contract documents to be removed or that are encountered in the course of the project(s).
- 1.4.2.5. Contractor shall be responsible for any/all required certifications, employee training, regulatory agency notifications, and jobsite signage to support subject hazardous materials removal.

#### 1.5. BIDDER'S INVESTIGATIONS

- 1.5.1.Bidder's Investigation: Bidder shall visit site and become familiar with site conditions at the project location.
- 1.5.2.Immediately upon completion of Bidder's investigation, bidder shall return site areas to condition existing prior to start of Bidder investigations as directed by District Representative.

#### 1.6. WORK COMPONENTS

The following work components are required by the Contract, Technical Specifications and Bid Proposal Exhibits and text of this RFP:

- 1.6.1.Preconstruction Services
  - 1.6.1.1. Preconstruction services shall be provided as per "Attachment 3 Exhibit A1 Preconstruction Services Scope of Work" and as otherwise described in project documents.

1.6.1.1.1.	Preconstruction services shall include but not be limited to:
1.6.1.1.2.	Project Design Review
1.6.1.1.3.	Preliminary Review of District's Schedule and Construction Budget
1.6.1.1.4.	Construction Cost Estimating
1.6.1.1.5.	Constructability Review(s) for build-ability, suitability of materials,
1.6.1.1.6.	Planning Construction Mobilization, Phasing and Logistics
1.6.1.1.7.	Project Scheduling
1.6.1.1.8.	Value Engineering and Cost-Benefit Analysis
1.6.1.1.9.	Ongoing Review of Architect's Programming Plan

- 1.6.1.1.10. Development of a proposed Guaranteed Maximum Price (GMP) with full detail, bid results, and notes including bid alternates and associated pricing.
- 1.6.2. Activities Prior to Start of On-site Construction Work
  - 1.6.2.1. Obtain ALL permits necessary to perform the scope of work. Prepare and file all required notifications, including but not limited to notifications in compliance with South Coast Air Quality Management District (SCAQMD) Rule 1403 requirements.
    - 1.6.2.1.1. SCAQMD must be notified, using their online web app, **prior to** any work activities taking place for (a) renovations that impact ACM (except for renovations involving less than 100 square feet total of ACM surface area); (b) all renovations involving the clean-up of damaged or disturbed ACM; and (c) all demolitions. Pre-registration with SCAQMD is required to use the web app.
    - 1.6.2.1.2. Please refer to SCAQMD Rule 1403
      (http://www.aqmd.gov/docs/default-source/rule-book/reg-xiv/rule-1403.pdf)
      for all current requirements, and the Rule 1403 Frequently Asked Questions
      page (http://www.aqmd.gov/docs/default-source/compliance/AsbestosDemolition-/1403-frequently-asked-questions.pdf) for more information.
  - 1.6.2.2. Submit and fully adhere to Contractor's health and safety plan in full compliance with CalOSHA, OSHA and project specifications. Site work may not proceed until this plan is delivered to and accepted by District.
  - 1.6.2.3. Submit and fully adhere to a site specific Technical Execution Plan (TEP or Work Plan) including a Hazardous Materials Abatement Plan. On site work activities may not proceed until the Technical Execution Plan (TEP) is reviewed and accepted by District's representative, engineer and/or environmental consultant, and proof of required regulatory agency notifications is provided.
  - 1.6.2.4. Identify and procure the services of licensed waste haulers and properly permitted Waste Disposal/Management Facilities for the transportation and disposal of all material generated during hazardous materials abatement and demolition activities.
  - 1.6.2.5. Submit a detailed work schedule for the project(s) for review and acceptance by District.
  - 1.6.2.6. Contractor shall erect perimeter site fencing with fabric screen prior to initiation of any other site work as per specifications and drawings.
- 1.6.3. Hazardous Material Abatement and/or Mitigation activities are to include abatement of but are not limited to the following materials as listed in sections 1.06 B.1, 2, and 3 below:
  - 1.6.3.1. Asbestos Containing Materials (ACMs)
    - 1.6.3.1.1. District has contracted survey(s) of asbestos, lead-based paint and other hazardous wastes to confirm presence of these materials. (See survey reports provided as attachments to project specifications).
    - 1.6.3.1.2. All ACM must be abated prior to commencing demolition or construction activities. The abatement must be performed by a Cal/OSHA

- licensed asbestos abatement contractor using methods in accordance with Title 8 of California Code of Regulations (CCR) 1529 and South Coast Air Quality Management District Rule 1403.
- 1.6.3.1.3. In the event that suspect asbestos containing asbestos-cement (Transite<sup>™</sup>) piping is encountered during subsurface demolition or construction activities, Contractor shall remove and dispose of these materials on a unit price basis as directed. (Note unit pricing required on Bid Form.)
- 1.6.3.2. Lead-Based Paint (LBP) and Lead containing items Lead was detected above concentrations greater than 1.0 mg/CM in the following:
  - 1.6.3.2.1. District has contracted survey(s) of asbestos, lead-based paint and other hazardous wastes to confirm presence of these materials. (See survey reports provided as attachments to project specifications).
- 1.6.3.3. Any/all hazardous materials abatement work shall be completed between the hours of 4:00 p.m. and 7:00 a.m., or during holidays or weekends when school is not in session.
- 1.6.3.4. Other Hazardous Materials.
  - 1.6.3.4.1. Fluorescent light fixtures were observed in the subject PHS buildings.
  - 1.6.3.4.2. All lighting ballasts which are not clearly marked "No PCBs" or "PCB Free" shall be assumed to contain PCBs, and shall be removed intact, packaged, and disposed of appropriately as hazardous waste. All other ballasts may be incinerated or recycled at an appropriate disposal site.
  - 1.6.3.4.3. All fluorescent tubes will be disposed as universal waste. Remaining bulbs will be disposed according to applicable regulations.
  - 1.6.3.4.4. Smoke detectors were observed in the subject buildings. Smoke detectors designated to be removed shall be disassembled by Contractor and categorized as either ionization detector (radioactive) or photoelectric detectors, which can be completed by checking for the required radioactive stickers on the baseplates. Photoelectric detectors may be discarded as construction debris. Ionization detectors will require appropriate off-site disposal per appropriate regulations.
  - 1.6.3.4.5. Exit signs Contractor shall disassemble exit signs to confirm whether they are paper, electrically lit or tritium containing. Paper and electric may be disposed as construction debris. Tritium units shall be disposed off-site according to regulations.

## 1.6.4.Demolition

1.6.4.1. All employees engaged in demolition activities shall be instructed regarding the contents of the Contractor's Health & Safety Plan(s).

- 1.6.4.2. Any/all demolition shall be performed in a manner that emphasizes and maximizes the safety of students, staff, area residents as well as project personnel and support staff.
- 1.6.4.3. Any/all demolition shall be performed in full compliance with local, state and federal regulations, including but not limited to SCAQMD, Cal/OSHA, Federal OSHA, EPA and AHERA requirements.
- 1.6.4.4. Demolition shall be performed in a manner that does not encroach upon or cause damage to adjacent properties and structures.
- 1.6.4.5. Demolition shall be performed in a manner that facilitates safe and efficient handling and load out of materials for disposal.
- 1.6.4.6. The sequence of Demolition, material stockpiling, loadout, transport, and disposal shall be performed in a manner that promotes a safe and smooth workflow to meet schedule milestones.
- 1.6.4.7. Contactor shall take measures to protect in place adjacent trees and landscaping designated to remain.

#### 1.7. SEQUENCING OF WORK

- 1.7.1.Proper regulatory notifications must be filed and Health & Safety plans be submitted, and permits be secured prior to commencing site work. Proof of filing of regulatory agency notifications will be required prior to start of work.
- 1.7.2.Installation of perimeter fencing and screening must be completed prior to initiation of other site activities.
- 1.7.3. Hazardous Materials Abatement work must be completed, inspected and approved by the District representative and/or District consultant prior to the start of any demolition.
- 1.7.4. Controls and wiring for low voltage systems (IT, Fire Alarm, Public Address, Clocks and Bells, etc.) serving adjacent facilities and classrooms shall be re-routed to the temporary classrooms and offices as necessary to provided uninterrupted operations prior to commencing demolition or construction activities.

#### 1.8. PERMITS, LICENSES AND FEES

## 1.8.1.Permits:

- 1.8.1.1. For Work included in the Contract, Contractor shall obtain all permits from authorities having jurisdiction including but not limited to City of San Bernardino, serving utility companies and other state and local regulatory agencies.
- 1.8.1.2. District will reimburse Contractor for amount charged for such permits, without mark-up.

## 1.8.2.Licenses and certifications:

1.8.2.1. Contractor shall obtain and pay all licenses and certifications associated with project demolition, abatement and construction activities, such as business licenses, contractors' licenses and vehicle and equipment licenses.

1.8.2.2. All costs for licenses and or certifications shall be included in the Contract Sum.

## 1.8.3. Assessments:

1.8.3.1. District will pay all assessments and utility service connection fees. Costs of assessments shall not be included in the Contract Sum.

## 1.8.4.Test and Inspection Fees:

- 1.8.4.1. Contractor shall pay all fees charged by authorities having jurisdiction and from serving utility companies and agencies, for tests and inspections conducted by those authorities, companies and agencies.
- 1.8.4.2. District will reimburse Contractor for actual amount of such fees, without mark-up.
- 1.8.4.3. Refer to Section 01 45 00 Quality Requirements for additional information on tests and inspections and responsibility for payment of fees.

## **END OF SECTION**

#### SECTION 01 11 14 - WORK SEQUENCE and PHASING

## PART 1 - GENERAL

#### 1.01 SECTION INCLUDES

A. Requirements for phasing of the Work include logistics, phasing, and completion of designated phases prior to commencement of subsequent phases.

## 1.02 RELATED SECTIONS

- A. Summary of the Work
- B. Project Coordination and Meetings
- C. Submittals
- D. Construction Progress Schedule
- E. Construction Facilities
- F. Temporary Controls
- G. Contract Closeout

#### 1.03 SUBMITTALS

- A. Prior to commencement of the Work, CONTRACTOR shall prepare and submit to the DISTRICT a Project Logistics Plan, including a Logistics Site Plan, showing in detail the Contractor's Work Sequence/Phasing plan, in the same size and scale as the architectural site plan, including, but not limited to, the following, items:
  - 1. Truck access route to and from the Project site, in accordance with local ordinances.
  - 2. Location of any overhead wire restrictions for power, street lighting, signal, and/or cable.
  - 3. Local sidewalk access and street closure requirements.
  - 4. Protection of sidewalk pedestrians and vehicular traffic.
  - 5. Project site fencing and access gate locations.
  - 6. Construction parking.
  - 7. Material staging and/or delivery areas.
  - 8. Material storage areas.
  - 9. Temporary trailer locations.
  - 10. Temporary service location and proposed routing of all temporary utilities.
  - 11. Location of temporary and/or accessible fire protection
  - 12. Trash removal and location of dumpsters.
  - 13. Concrete pumping locations.
  - 14. Crane locations.
  - 15. Location of portable sanitary facilities.
  - 16. Mixer truck wash out locations.
  - 17. Traffic control signage.
  - 18. Perimeter and site lighting.
  - 19. Provisions for Storm Water Pollution Prevention Plan SWPPP
  - 20. Stockpile and/or lay down areas.

- 21. Areas for separately identified phases of the work.
- 22. Barriers to separate construction activities from on-going school operations and circulation.

#### 1.04 PHASING OF THE WORK --GENERAL

- A. Project will be constructed in separate phases, or milestone increments, as identified or described in this Section and other parts of the Contract Documents. The Logistics Plan must define and delineate Work to be completed in each designated phase.
- B. Each phase will be required to be completed according to the Milestones included in the approved Construction Progress Schedule, prior to the commencement of the next subsequent phase, unless exception is granted by the District. CONTRACTOR shall incorporate in the plan and coordinate the Work of separate work contracts or of DISTRICT relative to each separate phase of this Project.
- C. CONTRACTOR shall prepare the Construction Progress Schedule in order to complete the Work and related activities in accordance with the phasing requirements, and to meet both the Milestone and Contract Time requirements.
- D. CONTRACTOR shall install all necessary Work for utilities and services, including, but not limited to, power, lighting, signal, HVAC, drainage, and plumbing systems in phased Work before completion of the designated phase. All valves, pull boxes, stub outs, temporary valves or capping, and other Work necessary for phased completion and operation of all necessary systems shall be provided whether or not such Work is specifically identified in the Contract Documents.

## 1.05 PHASING OF THE WORK - SPECIFIC

A. CONTRACTOR shall prepare the Construction Progress Schedule including phased Milestones, under the following general headings:

Phase 1 Mobilization – (# of days) calendar days:

Milestone Tasks

Phase 2 Utility Re-routing and Hazardous Materials Abatement – (# of days) calendar days:

Milestone Tasks

Phase 3 Demolition – (# of days) calendar days:

Milestone Tasks

B. The Contract Time shall be that shown in the Construction Progress Schedule.

#### **END OF SECTION**

#### **SECTION 01 11 40 - WORK RESTRICTIONS**

#### PART 1 - GENERAL

#### **SECTION INCLUDES:**

Contractor's Use of Premises

Access Roads

Parking

Work Hours

Restrictions on Noise, Dust, and Odor Emissions

Restrictions on Air Emissions of Toxic Chemicals

Protection of Existing Utilities

#### **CONTRACTOR'S USE OF PREMISES:**

- A. Contractor shall confine all operations, including the storage of materials, to the designated areas of the Project Site as shown in the Drawings, or as otherwise approved in writing by the Owner's Representative. Contractor shall be responsible for arranging for, and paying the costs of, any necessary off-site storage. No Impacted Materials shall be stored or stockpiled outside of the Project Site.
- **B.** Contractor's use of the premises shall be limited to the Work being performed under the Specifications and Drawings.
- Contractor shall be responsible for the security and safety of Contractor's equipment and facilities. Owner and the Owner's Representative shall not be liable for loss or damage of Contractor's tools, vehicles, equipment, or materials, whatever the cause. Such loss or damage shall not be sufficient reason for changes in the Project Schedule.
- **D.** Contractor shall be responsible for any damage to roadways, facilities, (unless otherwise marked for removal), or structures on, or adjacent to, the site due to negligence, carelessness, actions, errors, or omissions on the part of the Contractor.

#### **ACCESS ROADS:**

- **A.** Contractor vehicles shall enter and exit the site only at the location designated or as otherwise approved in writing by the Owner's Representative.
- **B.** Contractor shall be responsible for obtaining any permits and paying any fees necessary for Contractor's use of public streets or roads.
- **C.** Contractor shall abide by local, state, and federal regulations, including, but not limited to, any flaggers and signage for impeded traffic flow on public streets.
- **D.** Contractor shall, at all times, provide for unimpeded access for emergency vehicles to the Project Site and nearby properties.

#### **PARKING:**

- **A.** Contractor shall park construction vehicles and construction equipment only in areas designated for such purpose in accordance with Specifications.
- **B.** Contractor employees shall park personal vehicles only in an employee parking area as designated by the Owner's Representative.
- **C.** Vehicles shall not be parked in any locations where they impede traffic or access to areas where Work is being conducted.

#### **WORK HOURS:**

- A. Normal Work Hours (for activities other than hazardous materials abatement) will be 7:00 a.m. to 5:00 p.m. Monday through Friday, or as determined in advance of Work between the Owner's Representative and Owner. Work hours established by any ordinance, law, or regulation shall supersede the requirements of this Specification.
- **B.** Hazardous materials abatement activities shall not be conducted during normal school hours. Hazardous materials abatement activities shall occur on school holidays or weekends or, on school days, between the hours of 4:00 p.m. and 7:00 a.m.
- C. Should alternate or extended work hours be approved, Contractor shall conduct all Work during daylight hours so that the Work can be conducted safely and the Owner's Representative can effectively observe the Work, or Contractor may furnish adequate lighting for activities conducted by prior written approval of the Owner's Representative. Contractor shall provide adequate lighting at all times, as deemed necessary by the Owner's Representative for safety reasons, provided that the Contractor can demonstrate that light levels in the Work area meet or exceed OSHA Regulations.
- **D.** Contractor may conduct regular equipment maintenance during hours outside of the Normal Work Hours defined in this Section. The Contractor shall notify the Owner's Representative of such activities.
- **E.** Contractor personnel shall not work on site alone.
- F. Any variation from Normal Work Hours, or work on weekends or holidays shall be subject to approval by the Owner's Representative and Owner. Contractor shall submit notice to the Owner's Representative no less than 24 hours prior to requesting any necessary variation from Normal Work Hours, to allow for adequate review and coordination of staff. Contractor's notice to the Owner's Representative and Owner shall include Work activities to be conducted outside of Normal Work Hours, the hours and days that those activities shall be conducted, and the requested duration of the change in Normal Work Hours.
- G. Emergency repairs of equipment outside of Normal Work Hours may be performed without 24-hour notice, but Contractor shall verbally notify the Owner's Representative prior to such emergency maintenance.

#### RESTRICTIONS ON NOISE, DUST, AND ODOR EMISSIONS:

**A.** Contractor is responsible for conducting all Work in accordance with all applicable Laws and Regulations concerning work hours, noise or sound levels including but not limited to the requirements of the City of San Bernardino. Work involving high noise or high vibration

- levels may be restricted so as not to conflict with school testing and/or instructional activities.
- **B.** Contractor is responsible for conducting all Work in accordance with all applicable Laws and Regulations concerning airborne dust emissions including but not limited to the requirements of SCAQMD and the City of San Bernardino.
- **C.** Contractor is responsible for conducting all Work in accordance with all applicable Laws and Regulations concerning odor emissions including but not limited to the provisions of the City of San Bernardino.
- D. Contractor shall control the Work at all times such that noise, dust, and odor measurements do not exceed the Action Levels in the Specifications, Contractor's Health and Safety Plan and or regulatory limits.
- E. The Owner's Representative and Owner shall have authority to direct Contractor to stop Work or modify Work methods or activities as necessary to comply with the Health and Safety Plan, to prevent interruption to school testing or instructional activities, or should the Owner's Representative deem odor emissions, noise levels, or dust emissions be excessive.

#### 1.07. RESTRICTIONS ON AIR EMISSIONS OF HAZARDOUS OF TOXIC MATERIALS:

- A. Contractor shall be responsible for conducting all Work in accordance with Laws and Regulations concerning airborne emissions of hazardous dusts or toxic chemicals including but not limited to the requirements of SCAQMD, California DTSC and the City of San Bernardino.
- **B.** Contractor shall control the Work at all times such that concentrations of airborne constituents measured at the perimeter of the work area are below the Action Levels set forth in the Health and Safety Plan and/or regulations.
- C. The District and/or Owner's Representative shall have authority to direct the Contractor to stop Work or modify Work methods or activities as necessary to enforce compliance with the Action Levels for airborne emissions of toxic chemicals.

#### 1.08. PROTECTION OF EXISTING UTILITIES:

- **A.** Contractor shall contact and cooperate with utility companies to locate and mark all utilities (including pipelines, cables, power poles, and other structures) on the site prior to beginning the Work. Utility location shall be in compliance with Specifications, Drawings and Contract documents.
- **B.** Contractor shall comply with the requirements of specific utility protection Laws or Regulations.
- **C.** All utilities shall be protected from damage during construction, unless otherwise indicated to be removed or abandoned. If damaged, the utilities shall be repaired as required by the utility's Owner at the Contractor's expense.
- D. If a utility is encountered or otherwise made known to the Contractor prior to beginning the Work, the Contractor shall promptly take necessary steps to assure that the utility is not damaged, and give written notice to the Owner's Representative. The Owner's

Representative shall then review the conditions and determine the extent, if any, to which a change is required in the Contract Documents to reflect and document the consequences of the existence of the utility.

**E.** Contractor will be aware of and plan to prevent damage to underground utilities that might be caused by walking heavy equipment across the site. The Contractor will prevent mitigation measures in their costs and work plan to prevent damage to underground utilities.

**END OF SECTION** 

## SECTION 01 20 00 - PRICE AND PAYMENT PROCEDURES

## PART 1 - GENERAL

- 1.01 SECTION INCLUDES
  - A. Payment Procedures
  - B. Schedule of Values
- 1.02 RELATED SECTIONS
  - 1. Construction Progress Schedule
  - 2. Contract Closeout
  - Allowances

## 1.03 SCHEDULE OF VALUES

- A. Submit a Schedule of Values to the ARCHITECT for review and approval within 10 calendar days after the date of DISTRICT-CONTRACTOR Agreement. Submit in electronic Excel spreadsheet format.
- B. In the Schedule of Values, the Contract Sum shall be broken down into specific elements of the Work, as follows, coded in accordance with the DISTRICT'S coding structure.
  - 1. General Contractor's Overhead and Profit
  - 2. Site Mobilization
  - 3. Bonds and Insurance
  - 4. Field Supervision
  - 5. Project Close-Out (Section of General Requirements)
  - 6. Other General Conditions and General Requirements
  - 7. Demolition
  - 8. Site Clearing and Preparation
  - 9. Site Earthwork
  - 10. Site Improvements (Paving, etc.)
  - Site Utilities
  - 12. Landscape Irrigation
  - 13. Landscape Planting
  - 14. Hazardous Materials Abatement
  - 15. HVAC Work

- 16. Plumbing
- 17. Fire Protection Sprinklers
- 18. Electrical Power Rerouting
- 19. Electrical Site Lighting
- 20. Fire Alarm and Smoke Detection Systems
- 21. Electrical Communications and Security Systems
- C. On projects of more than one building, provide separate schedules for each building.
- D. The percent-complete values from the approved cost-loaded Construction Progress Schedule shall provide the basis for each Application for Payment. Before each Application, update the Progress Schedule with all approved Change Orders.

#### 1.04 APPLICATION FOR PAYMENT

- A. Submit Application for Payment to the ARCHITECT and IOR for review, in electronic format. Upon approval submit three (3) signed and original copies of each certified application. All copies shall be complete, including the updated Schedule of Values or Construction Progress Schedule, releases and similar attachments. Transmit each copy with a transmittal form listing attachments and recording appropriate information related to the application, in a manner acceptable to ARCHITECT
- B. Each certified Application for Payment shall be consistent with previous applications and payments as reviewed by ARCHITECT and IOR; paid for by OWNER.
- C. Payment Application Times: The period of Work covered by each Application for Payment is based on the payment date for each progress payment as specified in the General Conditions. The period covered by each Application for Payment is the previous month.
- D. Initial Application for Payment: Administrative actions and submittals that must precede or coincide with the first certified Application for Payment include, but are not limited to, the following:
  - Certified Schedule of Values or Cost-Loaded Schedule
  - 2. Performance and payment bonds
  - 3. List of principal suppliers and fabricators
  - 4. Worker Compensation certificates
  - Auto Insurance

- 6. Hazardous Material Insurance Certificates
- 7. Construction Progress Schedule
- 8. Submittal Schedule
- 9. Emergency Contact List
- Copies of authorizations and licenses from governing authorities for performance of the Work
- E. Application for Payment at Substantial Completion: Following OWNER issuance of the certificate of Substantial Completion, submit an Application for Payment together with the following:
  - Occupancy permits and similar approvals by authorities having legal jurisdiction over the Work
  - 2. Removal of temporary facilities and services
  - 3. Testing, adjusting and balance records
  - 4. Removal of surplus materials, rubbish, and similar elements
  - 5. Meter readings
  - 6. Start-up performance reports
  - 7. OWNER training and orientations
  - 8. Change-over information related to OWNER occupancy, use, operation, and maintenance
  - 9. Final cleaning
  - 10. Ensure that incomplete Work is not accepted and will be completed without undue delay
  - 11. Advice on shifting insurance coverage
  - 12. List of defective Work, recognized as exceptions to certificate of Substantial Completion
  - 13. Change of door locks to OWNER system
- F. Final Payment Application: Administrative actions and submittals that must precede or coincide with submittal of the final Application for Payment include, but are not limited to, the following:
  - 1. Completion of Contract Closeout requirements
  - 2. Project record and other closeout documents
  - 3. Completion of final punch list items
  - 4. Delivery of extra materials, products and or stock
  - 5. Identification of unsettled claims

- 6. Proof that taxes, fees, and similar obligations are paid
- 7. Evidence of payment and release of liens
- 8. Operating and maintenance instruction manuals
- 9. Consent of surety to final payment
- 10. Waivers and releases
- 11. Warranties, guarantees and maintenance agreements

### G. Retention

1. Retention will be released no sooner than 35 days and not later than 60 days after Notice of Completion has been recorded with the County Recorder's Office.

**END OF SECTION** 

### SECTION 01 26 00 - CONTRACT MODIFICATION PROCEDURES

### **PART 1 - GENERAL**

#### 1.01 SUMMARY

- A. This Section specifies administrative and procedural requirements for making modifications to the contract including:
  - 1. Change Orders
  - 2. Construction Change Documents (see General Conditions)
  - Contract Credits
  - 4. Contract Additions
  - 5. Construction Change Directives
  - 6. Immediate Change Directives (see General Conditions)
  - Instructions

### B. Modifications:

- Provide full written data required to evaluate contract modifications, including breakdown of labor, material, equipment and description of work with unit costs for each category.
- 2. Maintain detailed records of work done on a time-and-material basis
- 3. Provide full documentation for all proposed Change Orders to the Architect for his review.
- C. Designate in writing the member of Contractor's organization:
  - 1. Who is authorized to accept changes in the Work.
  - 2. Who is responsible for informing others in the Contractor's employ of the authorization of changes in the Work.

### 1.02 RELATED SECTIONS

- A. Addenda: All issued Addendums
- B. Agreement: The amounts of unit prices if any as established in the Contract.
- C. General Conditions Article 7, Changes in the Work.
- D. Section 01 30 00 Administrative Requirements for Submittal Procedures.
- E. Section 01 60 00 Product Requirements

### 1.03 REFERENCES

- A. Change Order Requirements per Title 24 Part 1 CCR.
  - 1. Change Orders: Changes or alterations of the approved plans or specifications after a contract for the work has been awarded are to be made by means of Change Orders. State the reason for the change and provide supplementary drawings where necessary. Change Orders must be manually signed by the Architect or Engineer in general responsible charge of observation of the work

- or by the Architect or Engineer delegated responsibility for observation of the portion of the work affected by the Change Order.
- 2. Change Orders are required to bear the approval of the School Board or their authorized representative upon delegated authority.

#### PRELIMINARY PROCEDURES 1.04

- A. The Architect or School District may initiate changes by submitting a Request for Proposal. The request will include:
  - Detailed description of the Change, Products, and location of the change in the Project. Changes may include additions and deletions from the Contract.
  - 2. Supplementary or revised Drawings and Specifications.
  - The projected time span for making the change and a specific 3. statement as to whether overtime work is, or is not, authorized.
  - A specific period of time during which the requested price will be 4. considered valid.
  - 5. Such request is for information only, and is not an instruction to execute the changes, nor to stop Work in progress.
- B. Contractor may initiate changes by submitting a written Change Order Request to the Architect or School District containing:
  - Description of the proposed change. 1.
  - 2. Statement of the reason for making the changes.
  - Statement of the effect on the Contract Sum/ Contract Price and the 3. Contract Time.
  - 4. Statement of the effect on the Work of separate contractors with breakdown of costs for labor, materials and equipment.
  - 5. Documentation supporting any change in Contract Sum/ Contract Price or Contract Time, as appropriate.

#### 1.05 **CONSTRUCTION CHANGE DIRECTIVES**

- In lieu of Proposal Request, the School District through the Construction Α. Manager may issue, a Construction Change Directive (also referred to as an Immediate Change Directive in the General Conditions) for Contractor to proceed with a change which shall state a basis for adjustment, if any, in the Contract Sum/ Contract Price or Contract Time, or both.
- B. Authorization will describe changes in the Work, both additions and deletions, with attachments of revised Contract Documents to define details of the change, and will designate the method of determining any change in the Contract Sum/ Contract Price and any change in Contract Time.
- C. The School District and Architect will sign and date the Construction Change Directive as authorization for the Contractor to proceed with the changes.
- D. Contractor may sign and date the Construction Change Directive to indicate agreement with the terms therein.

#### 1.06 **DOCUMENTATION OF PROPOSALS AND CLAIMS**

- A. Support each quotation for a lump-sum proposal, and for each unit price which has not previously been established, with sufficient substantiating data to allow the Engineer and School District to evaluate the quotation.
- B. On request provide additional data to support time and cost computations:
  - Labor required in hours with unit costs. 1.
  - 2. Equipment required.
  - 3. Products required in units
    - Recommended source of purchase and unit cost. a.
    - Quantities required h.
  - Taxes, insurance and bonds. 4.
  - 5. Credit for Work deleted from Contract, similarly documented.
  - 6. Overhead and profit.
  - Justification for any change in Contract Time.
- C. Support each claim for additional costs, and for work done on a time and material basis, with documentation as required for a lump-sum proposal, plus additional information:
  - 1. Name of the School District's authorized agent who ordered the work, and date of the order.
  - 2. Dates and times work was performed, and by whom.
  - Time record, summary of hours worked, and hourly rates paid. 3.
  - Receipts and invoices for: 4.
    - Equipment used, listing dates and times of use. a.
    - Products used, listing of quantities. b.
    - Subcontracts C.
- D. Document requests for Substitution of Products as specified in Section 01 60 00.

#### 1.07 **CONSTRUCTION CREDITS**

- Α. Work deleted and no work has been completed by the Contractor: Work deleted from the contract is to be credited back to the District and subtracted from the contract amount. Credits are to be included in Change Orders.
  - Contractor shall credit back to the District total value for the work 1. deleted from the contract. Cost of credits shall be determined by the amount stated in the Contractor's Schedule of Values.
  - 2. Where the value of credits cannot be determined from the Contractor's Schedule of values, total value of the credit is to be determined by the cost of materials, labor, overhead and profit, insurance, bonds, etc. All General Contractor, Subcontractor and Material Supplier levels of the Contract are to be included in the total value of credits back.
  - 3. No amount at any level of the contract shall be withheld from credits for overhead and profit, insurance, bonds, time delays, construction schedule changes and administrative expenses.
- B. Work deleted and a portion of the work has been completed by the Contractor: Work deleted from the contract is to be credited back to the

District and subtracted from the contract amount. Credits are to be included in Change Orders.

- Contractor shall credit back to the District the total value of the work 1. deleted from the contract less any work already completed on the credit item. Cost of credits shall be determined by the amount stated in the Contractor's Schedule of Values less any work already completed. Completed work may include cost of shop drawings. submittals, site preparation, partially completed work on the credit item or other expenses related to the item.
- 2. Where the value of credits cannot be determined from the Contractor's Schedule of values, total value of the credit is to be determined by the cost of materials, labor, overhead and profit, insurance, bonds, etc. All General Contractor, Subcontractor and Material Supplier levels of the Contract are to be included in the total value of credits back.
- 3. An amount equal to the percentage of work already completed on the deleted item may be withheld from credits back for overhead and profit, insurance, bonds, construction schedule adjustments and administrative expenses, as indicated in the General Conditions.

#### 1.08 PREPARATION OF CHANGE ORDERS

- The Architect will prepare each Change Order. Α.
- Change Order will describe changes in the Work, both additions and B. deletions, with attachments of revised Contract Documents to define details of the change.
- C. Change Order will provide an accounting of the adjustment in the Contract Sum/ Contract Price and in the Contract Time.

#### 1.09 **LUMP-SUM/FIXED PRICE CHANGE ORDER**

- Α. Content of Change Orders will be based on either:
  - The School District's Proposal Request and Contractor's 1. responsive Proposal as mutually agreed with the School District.
  - 2. Contractor's Proposal for a change, as recommended by the School District or their authorized agent.
- B. The School District, Division of the State Architect and Architect or Engineer in responsible charge will sign and date the Change Order as an authorization for the Contractor to proceed with the changes.
- The Contractor will sign and date the Change Order to indicate agreement C. with the terms therein.

#### **UNIT PRICE CHANGE ORDER** 1.10

- Content of Change Orders will be based on either: Α.
  - The School District's definition of the scope of the required 1. changes.

- 2. Contractor's Proposal for a change, as recommended by the School District or Authorized Agent.
- Survey of completed work. 3.
- B. The amounts of the unit prices to be:
  - Those stated in the Agreement.
  - Those mutually agreed upon between School District and 2. Contractor.
- C. When quantities of each of the items affected by the Change Order can be determined prior to start of the work:
  - The School District and Architect or Engineer in responsible charge 1. will sign and date the Change Order as authorization for Contractor to proceed with the changes.
  - 2. Contractor is to sign and date the Change Order to indicate agreement with the terms therein.
- D. When quantities of the items cannot be determined prior to start of the work:
  - The School District through the Architect will issue a Construction Change Directive directing the Contractor to proceed with the change on the basis of unit prices, and will cite the applicable unit prices.
  - 2. At completion of the change, the School District or its authorized agent will determine the cost of such work based on the unit prices and quantities used.
  - 3. The Contractor shall submit documentation to establish the number of units of each item and any claims for a change in Contract Time.
  - The School District, Division of the State Architect and Architect or 4. Engineer in responsible charge will sign and date the Change Order as authorization for the Contractor to proceed with the Changes.
  - 5. The Contractor will sign and date the Change Order to indicate agreement with the terms therein.

#### 1.11 TIME AND MATERIALS CHANGE ORDER/CONSTRUCTION CHANGE **DIRECTIVE:**

- Α. The School District through the Architect will issue a Construction Change Directive directing Contractor to proceed with the changes
- B. At completion of the change, Contractor shall submit itemized accounting and supporting data as provided in the Article 1.6, "Documentation of Proposals and Claims," of this Section.
- C. The School District or its authorized representative will determine the allowable cost of such work, as provided in General Conditions and Supplementary Conditions.
- D. The School District, Division of the State Architect and Architect or Engineer in general responsible charge will sign and date the Change Order to authorize the change in Contract Sum/ Contract Price and in Contract Time.
- E. The Contractor will sign and date the Change Order to indicate agreement with the terms therewith.

#### 1.12 **INSTRUCTIONS**

- Α. Architect's Supplemental Instructions:
  - Minor changes in the work shall be carried out in accordance with supplemental instructions issued in accordance with the Contract Documents without change in Contract Sum/ Contract Price or Contract Time.
  - 2. The Architect will issue, sign, and date Supplemental Instructions.
  - The Contractor will sign and date Supplemental Instructions to 3. indicate acceptance of minor changes consistent with the Contract Documents and return signed copy to Architect.

#### 1.13 CORRELATION WITH CONTRACTOR'S SUBMITTALS

- A. Periodically revise Schedule of Values and Request for Payment forms to record each change as a separate item of Work and to record the adjusted contract amounts.
- B. Periodically revise the Construction Schedule to reflect each change in Contract Time.
- C. Revise sub-schedules to show changes for other items of work affected by the changes.
- D. Upon completion of work under a Change Order, enter pertinent changes in Record Documents.

#### 1.14 **FORMS**

- Α. Submit Proposal Request typed on AIA Document G709. A Copy of this form may be obtained from the local American Institute of Architects, **Chapter Office**
- B. Submit Change Orders typed on the Change Order Form included in this Project Manual. Form is included in General Conditions and at the end of this Section.
- C. Submit Supplemental Instructions typed on the form provided by Architect, Requests for Interpretation (RFI's).
- D. Immediate Change Directive Form is included in the Supplementary General Conditions.

PART 2 – PRODUCTS - (Not Applicable)

PART 3 – EXECUTION - (Not Applicable) **END OF SECTION (Form Attached)** 

### **SECTION 01 30 00 - ADMINISTRATIVE REQUIREMENTS**

### **PART 1 GENERAL**

### 1.01 SECTION INCLUDES

- Preconstruction meeting.
- B. Site mobilization meeting.
- C. Progress meetings.
- D. Construction progress schedule.
- E. Progress photographs.
- F. Coordination drawings.
- G. Requests for Interpretation.
- H. Submittals for review, information, and project closeout.
- I. Number of copies of submittals.
- J. Submittal procedures.
- K. Labor Compliance Program

### 1.02 RELATED REQUIREMENTS

- A. Section 01 20 00 Price and Payment Procedures:
  - 1. Proposal Requests, Construction Change Directives and Change Orders.
  - 2. Applications for Payment and the Schedule of Values.
- B. Section 01 32 16 Construction Progress Schedule: Form, content, and administration of schedules.
- C. Section 01 40 00 Quality Requirements: Test and inspection reports.
- D. Section 01 60 00 Product Requirements: Requests for substitutions of materials, products, equipment and systems.
- E. Section 01 70 00 Execution and Closeout Requirements: Additional coordination requirements.
  - 1. Requirements for preparation and submission of operation and maintenance data.
  - 2. Lien and bonding company releases, keys, inspection records from authorities having jurisdiction and insurance documents.
  - 3. Submittals for occupancy, Acceptance and Final Payment.
- F. Section 01 78 00 Closeout Submittals: Project record documents.
  - 1. Procedures for submitting warranty and guarantee documentation.
- I. Divisions 2 through 33 Sections (as applicable): Procedures for specific submittals specified in those Sections to be made at Contract closeout.

### 1.03 REFERENCE STANDARDS

A. AIA G810 - Transmittal Letter; 2001.

### 1.04 DEFINITIONS

- A. Action Submittals: Written and graphic information that requires responsive action by Construction Manager and Architect or other responsible design professional.
- B. Informational Submittals: Written information that does not require responsive action by Construction Manager and Architect or other responsible design professional.
- C. Unsolicited Submittals: Action or informational submittals not required by the Contract Documents or not requested by the reviewer. Unsolicited submittals may be returned with notation "not reviewed."
- D. Product Data: Standard published information ("catalog cuts") and specially prepared data for the Work of the Contract, including standard illustrations, schedules, brochures, diagrams, performance charts, instructions and other information to illustrate a portion of the Work.
- E. Request for Interpretation (RFI): A document submitted by the Contractor requesting clarification of a portion of the Contract Documents, hereinafter referred to as an RFI.
- F. Samples: Physical examples that demonstrate the materials, finishes, features, workmanship and other characteristics of a portion of the Work. Accepted samples shall serve as quality basis for evaluating the Work.
- G. Shop Drawings, Product Data and Samples: Instruments prepared and submitted by Contractor, for Contractor's benefit, to communicate to Architect the Contractor's understanding of the design intent, for review and comment by Architect on the conformance of the submitted information to the general intent of the design. Shop drawings, product data and samples are not Contract Documents.
- H. Shop Drawings: Drawings, diagrams, schedules and illustrations, with related notes, specially prepared for the Work of the Contract, to illustrate a portion of the Work.
- I. Other Submittals: Technical data, test reports, calculations, surveys, certifications, special warranties and guarantees, operation and maintenance data, extra stock and other submitted information and products shall not be considered as Contract Documents but shall be information from Contractor to Architect to illustrate a portion of the Work for confirmation of understanding of design intent.

### **PART 2 PRODUCTS - NOT USED**

### PART 3 EXECUTION

#### 3.01 PROGRESS PHOTOGRAPHS

- A. Submit photographs with each application for payment, taken not more than 3 days prior to submission of application for payment.
- B. Maintain one set of all photographs at project site for reference; same copies as submitted, identified as such.
- C. Photography Type: Digital; electronic files.
- D. Provide photographs of site and construction throughout progress of Work produced by an experienced photographer, acceptable to Architect.
- E. In addition to periodic, recurring views, take photographs of each of the following events:
  - Completion of site clearing.

- 2. Excavations in progress.
- 3. Foundation removals in progress and upon completion.
- 4. Final completion, minimum of ten (10) photos.

### F. Views:

- 1. Provide non-aerial photographs from four cardinal views at each specified time, until Date of Substantial Completion.
- 2. Consult with Architect for instructions on views required.
- 3. Provide factual presentation.
- 4. Provide correct exposure and focus, high resolution and sharpness, maximum depth of field, and minimum distortion.
- 5. Point of View Sketch: Provide sketch identifying point of view of each photograph.
- G. Digital Photographs: 24 bit color, minimum resolution of 2584 x 1936 ("5 megapixel"), in JPG format; provide files unaltered by photo editing software.
  - 1. Delivery Medium: Via email (for individual photos) and with project record photos on DVD.
  - 2. File Naming: Include project identification, date and time of view, and view identification.
  - 3. Point of View Sketch: Include digital copy of point of view sketch with each electronic submittal; include point of view identification in each photo file name.
  - 4. PDF File: Assemble all photos into printable pages in PDF format, with 2 to 3 photos per page, each photo labeled with file name; one PDF file per submittal.
  - 5. Photo CD(s): Provide 1 copy including all photos cumulative to date and PDF file(s), with files organized in separate folders by submittal date.
  - 6. Hard Copy: Printed hardcopy (grayscale) of PDF file and point of view sketch.

### 3.02 CONTRACTOR'S REQUESTS FOR INTERPRETATION (RFI)

- A. Contractor shall comply with procedures specified herein to make a Request for Interpretation (RFI).
  - 1. Prior to submitting a written RFI, when possible, the Contractor shall have a verbal conversation with the Architect to discuss the item in question.
- B. The Contractor shall request that the Architect make an interpretation of the requirements of the Contract Documents for resolution of the following:
  - Inability to determine from the Contract Documents the exact material, process, or system to be installed;
  - 2. Or when the elements of construction are required to occupy the same space (interference);
  - 3. Or when an item of Work is described differently at more than one place in the Contract Documents; .
- C. Submission of RFIs: RFIs shall be prepared and submitted on form acceptable to Architect.
  - 1. Forms shall be completely filled in, and if prepared by hand, shall be fully legible after copying by xerographic process.
  - 2. Each RFI shall be limited to a single discrete subject.

- 3. Each RFI shall be given a discrete, consecutive number.
- 4. Each page of the RFI and each attachment to the RFI shall bear the Project name, Architect's Project number, date, RFI number and a descriptive title.
- 5. Contractor shall sign all RFIs attesting to good faith effort to determine from the Contract Documents the information requested for interpretation.
- 6. Make submission of RFIs to Architect.
- 7. RFI may be submitted by email as a PDF attachment or through an electronic service if agreed to in advance by all parties.
- D. Subcontractor-Initiated and Supplier-Initiated RFIs:
  - RFIs from subcontractors and material suppliers shall be submitted through, be reviewed by and be attached to an RFI prepared, signed and submitted by Contractor.
    - a. RFIs submitted directly by subcontractors or material suppliers will be returned unanswered to the Contractor.
  - 2. Contractor shall review all subcontractor- and supplier-initiated RFIs and take actions to resolve issues of coordination, sequencing and layout of the Work.
    - a. RFIs submitted to request clarification of issues related to means, methods, techniques and sequences of construction or for establishing trade jurisdictions and scopes of subcontracts will be returned without interpretation.
      - 1) Such issues are solely the Contractor's responsibility.
    - b. Contractor shall be responsible for delays resulting from the necessity to resubmit an RFI due to insufficient or incorrect information presented in the RFI.

### E. Requested Information:

- Contractor shall carefully study the Contract Documents to ensure that information sufficient for interpretation of requirements of the Contract Documents is not already included. RFIs that request interpretation of requirements clearly indicated in the Contract Documents will be returned without interpretation.
- 2. In all cases in which RFIs are issued to request clarification of issues related to
- 3. In all cases, the Contractor shall furnish all information required for the Architect to analyze and/or understand the circumstances causing the RFI and prepare a clarification or direction as to how the Contractor shall proceed for RFIs issued to request clarification of issues related to:
  - a. Means, methods, techniques and sequences of construction, for example
  - b. Pipe and duct routing, clearances;
  - c. Specific locations of Work shown diagrammatically;
  - d. Apparent interferences and similar items.
- 4. If information included with this type RFI by the Contractor is insufficient, the RFI will be returned unanswered.
- F. Unacceptable Uses for RFIs: RFIs shall not be used for the following purposes:
  - 1. To request approval of submittals.
  - 2. To request approval of substitutions. See Section 01 60 00 Product Requirements.

- 3. To request changes that entail change in Contract Time and Contract Sum. (Comply with provisions of the Conditions of the Contract), as discussed in detail during pre-construction conference).
- To request different methods of performing Work than those indicated in the Contract Drawings and Specifications (comply with provisions of the Conditions of the Contract).
- G. Disputed Requirements: In the event the Contractor believes that a clarification by the Architect results in additional cost or time, Contractor shall not proceed with the Work indicated by the RFI until authorized to proceed by the District and claims, if any, are resolved in accordance with provisions in the Conditions of the Contract.
- H. RFI Log: Contractor shall prepare and maintain a log of RFIs, and at any time requested by the Architect or District, the Contractor shall furnish copies of the log showing all outstanding RFIs.
- I. Review Time: Architect will return RFIs to Contractor and District within ten calendar days of receipt.
  - 1. RFIs received after 12:00 noon shall be considered received on the next regular working day for the purpose of establishing the start of the response period.
  - 2. Architect will endeavor to respond in a timely fashion to RFIs, in less than the allotted time. Some RFI's could take longer depending on third party responses.

### **SECTION 01 31 00 - PROJECT COORDINATION AND MEETINGS**

#### PART 1 GENERAL

### 1.01 SECTION INCLUDES

- A. Coordination and Coordination Drawings.
- B Pre-construction meeting.
- C. Progress meetings.
- D. Pre-installation meetings.

### 1.02 RELATED SECTIONS

- A. Construction Progress Schedule.
- B Work Sequence and Phasing.
- C. Labor Compliance Program (LCP Coordination).
- D. Submittal Procedures.
- E. Closeout Procedures.
- F. Field Engineering.

### 1.02 COORDINATION

- A. Coordinate scheduling, submittals and work of the various portions of the Contract Documents to assure efficient and orderly sequence of installation of interdependent construction elements, with provisions for accommodating items installed later.
- B. When separate or multiple-prime contracts exist, work cooperatively with the DISTRICT and all other CONTRACTORs, subcontractors, suppliers and other entities working on any portion of the Project.
- C. Work closely with the DISTRICT to coordinate work and to maintain the Construction Progress Schedule.
- D. Verify that utility-requirements of equipment to be installed are compatible with building utilities. Coordinate work of various Sections having interdependent responsibilities for installing, connecting to, and placing in service, such equipment, as well as work of utility companies.
- E. Coordinate space requirements and installation of mechanical and electrical work which are indicated diagrammatically on drawings. Follow routing shown for pipes, ducts and conduit, as closely as practicable; place runs parallel with line of building. Utilize spaces efficiently to maximize accessibility for other installations, for maintenance and for repairs.
- F. In finished areas, except as otherwise indicated, conceal pipes, ducts and wiring within the construction. Coordinate locations of fixtures and outlets with finish elements.

- G. In locations where several elements of mechanical and electrical work must be sequenced and positioned with precision in order to fit into available space, prepare Coordination Drawings showing the actual conditions required for the installation prior to purchasing, fabricating or installing the elements required to be coordinated. Submit Coordination Drawings to DISTRICT.
- H. Closing up of walls, partitions or furred spaces, backfilling and other covering up operations shall not proceed until all enclosed or covered work and inspections have been completed. Verify before proceeding.
- I. Coordinate completion and cleanup of work of separate sections in preparation for substantial completion, including portions of work designated for DISTRICT's full and/or partial occupancy).
- J. After DISTRICT occupancy of premises, coordinate access to site for correction of defective work and work not in accordance with Contract Documents, in a manner to minimize disruption of DISTRICT's activities.

### 1.03 PRE-CONSTRUCTION MEETING

- A. The DISTRICT will schedule a pre-construction meeting immediately after receipt of the Notice of Award.
- B. Mandatory attendance includes DISTRICT Representative, INSPECTOR OF RECORD (IOR), Testing Laboratory Representative, ARCHITECT, CONTRACTOR and CONTRACTORS Job Superintendents, and major subcontractors.
- C. Optional Attendance includes Architect's consultants, subcontractors and utility company representatives.
- D. The DISTRICT Representative will preside at the conference, and will record meeting minutes and distribute copies in a timely manner.

# E. Agenda:

- 1. Execution of Agreement between DISTRICT and CONTRACTOR.
- 2. Issue Notice to Proceed.
- Submission of executed bonds and insurance certificates.
- 4. Distribution of Contract Documents, Notice of Award, Forms, sample Schedule of Values.
- 5. Submission of list of subcontractors, list of products.
- 6. Designation of responsible personnel representing the parties.
- 7. Procedures for processing of field decisions, submittals, substitutions, applications for payments, proposal requests, change orders, construction directives, and contract closeout procedures.
- 8. Preliminary Construction Progress Schedule.
- 9. Critical work sequencing.
- 10. Submittals, substitutions.

- 11. Procedures and forms for preparation and maintenance of project record/as-built documents
- 12. Use of the Project site and parking availability
- 13. Temporary facilities
- 14. Equipment deliveries and priorities
- 15. Safety procedures
- 16 Security
- 17. Housekeeping
- 18. Working hours
- 19. Labor Compliance Officer presentation
- 20. Insurance Services including OCIP
- 21. Environmental Health & Safety
- 22. Review of Logistics Plan
- 23. Progress payments
- 24. Communications procedures
- 25. Fingerprinting requirements
- 26. Construction permits
- 27. Inspections and tests
- 28. SWPPP
- 29. Project meetings

### 1.04 PROGRESS MEETINGS

- A. CONTRACTOR shall schedule and administer progress meetings throughout progress of the work at regular intervals, typically weekly, or more frequently if needed.
- B. CONTRACTOR shall make arrangements for meetings, prepare agenda and preside at meetings, record minutes and distribute copies.
- C. Attendance required: DISTRICT, Architect, IOR, CONTRACTOR.
- D. Agenda:
  - 1. Review minutes of previous meetings.
  - 2. Review work progress.
  - 3. Field observations, problems and decisions.
  - 4. Identification of problems which impede planned progress.
  - 5. Review of submittals' status and schedule of submittals.
  - 6. Review of off-site fabrication progress and delivery schedules.
  - 7. Maintenance of Progress Schedule.
  - 8. Corrective measures to regain projected schedules.
  - 9. Planned progress during succeeding work period.
  - 10. Coordination of projected progress.
  - 11. Maintenance of quality and workmanship standards.
  - 12. Proposed changes and effect on progress schedule and coordination.
  - 13. Other business appropriate to the status of the Project.

### 1.05 PRE-INSTALLATION MEETING

- A. When required in individual Specification Sections, convene a pre-installation meeting prior to commencing work of the Section.
- B. Require attendance of parties directly affecting, or affected by, work of the specific section, including but not limited to the ARCHITECT, IOR, CONTRACTOR, and affected sub-contractors, manufacturers and fabricators.
- C. Notify DISTRICT and ARCHITECT at least five (5) days in advance of meeting date.
- D. Prepare agenda, preside at conference, record minutes and distribute copies within three (3) days after conference to participants
- E. Review and discuss pre-installation conditions, preparation, installation procedures, coordination with related work, and orientation of Maintenance and Operations personnel.

**END OF SECTION** 

# SECTION 01 31 14 LABOR COMPLIANCE PROGRAM (LCP) COORDINATION

### PART 1 GENERAL

### 1.01 SECTION INCLUDES

- A. LCP Coordination Activities
- B. LCP Meeting

# 1.02 RELATED SECTIONS

- A. Project Coordination and Meetings
- B. Price and Payment Procedures

### 1.03 LABOR COMPLIANCE PROGRAM MEETING

After the DISTRICT awards the Contract, and prior to the commencement of the work, a mandatory pre-job LCP Meeting will be conducted by the LCP representative with the CONTRACTOR(s) and those subcontractors listed in the Bid Documents – as part of the Pre-Construction Meeting or as a separate meeting.

At that meeting, the LCP representative will discuss the federal and state labor law requirements applicable to the contract including prevailing wage requirements, respective record keeping responsibilities, the requirement for the submittal of certified payroll records to the DISTRICT, and the prohibition against discrimination in employment.

The LCP representative will provide the CONTRACTOR and each subcontractor with a checklist of labor law requirements and will discuss, in detail, the following checklist items:

### 1. Payment of Prevailing Wage Rates

The CONTRACTOR to whom the contract is awarded and its subcontractors hired for the public works project are required to pay no less than the specified general prevailing wage rates to all workers employed in the execution of the contract, including each subcontract.

The CONTRACTOR is responsible for ascertaining and complying with all current general prevailing wage rates for crafts and any rate changes that occur during the life of the contract. Information on all prevailing wage rates and all rate changes are to be posted at the job site for all workers to view.

### 2. Apprentice

It is the duty of the CONTRACTOR and subcontractors to employ registered apprentices on the public works project under Labor Code Section 1777.5;

### 3. Penalties

There are penalties required for CONTRACTOR'S/ subcontractor's failure to pay prevailing wages and for failure to employ apprentices, including forfeitures and debarment under Labor Code Sections 1775; 1776; 1771.1; 1777.7 and 1813;

## 4. Certified Payroll Reports

Under Labor Code Section 1776, CONTRACTORS and subcontractors are required to keep accurate payroll records showing the name, address, social security number and work classification for each employee and owner performing work; also the straight time and overtime hours worked each day and each week, the fringe benefits and the actual per diem wage paid to each owner, journey person, apprentice worker or other employee hired in connection with the public works project.

Employee payroll records shall be certified and shall be made available for inspection at all reasonable hours at the principal office of the contractor/subcontractor, or shall be furnished to any employee, or his/her authorized representative on request, pursuant to Labor Code Section 1776;

Each CONTRACTOR and every lower-tier subcontractor and supplier is required to submit certified payrolls and labor compliance documentation electronically as specified by the DISTRICT.

Electronic submittal will be a web-based system, accessed on the World Wide Web by a web browser. Each contractor and subcontractor will be given a Log On identification and a password to access the DISTRICT's reporting system.

Use of the system may entail additional data entry of weekly payroll information including; employee identification, labor classification, total hours worked and hours worked on this project, wage and benefit rates paid, etc. The CONTRACTOR's payroll and accounting software may be capable of generating a 'comma delimited file' that will interface with the software.

This electronic reporting requirement will be 'flowed down' to every lower-tier subcontractor and vendor required to provide labor compliance documentation.

Under Labor Code Section 1776(g) there are penalties required for contractor's/ subcontractor's failure to maintain and submit copies of certified payroll records on request.

# 5. Nondiscrimination in Employment

There exists prohibition against employment discrimination under Labor Code Sections 1735 and 1776.6, the Government Code, the Public Contracts Code and Title VII of the Civil Rights Act of 1964;

### 6. Kickbacks Prohibited

CONTRACTORS and subcontractors are prohibited from recapturing wages illegally or extracting "kickbacks" from employee wages under Labor Code Section 1778;

# 7. Itemized Wage Deduction Statement

Under Labor Code Section 226, every employer shall at the time of each payment of wages, furnish each of his or her employees, an accurate itemized statement in writing showing the gross wages, total hours worked, all deductions, net wages earned, the inclusive dates of the period for which the employee is

paid, name of the employee and his/her social security number, the name and address of the employer and all applicable hourly rates in effect during the pay period.

# 8. Acceptance of fees prohibited

There exists a prohibition against CONTRACTOR/subcontractor acceptance of fees for registering any person for public work under Labor Code Section 1779; or for filling work orders on public works contracts pursuant to Labor Code Section 1780:

### 9. Listing of Subcontractors

All prime CONTRACTORS are required to list properly all subcontractors hired to perform work on the public works projects covering more than one half of one percent, pursuant to Government Code Section 4100 et seq;

# 10. Proper Licensing

CONTRACTORS are required to be licensed properly and to require that all subcontractors be properly licensed. Penalties are required for employing workers while unlicensed under Labor Code Section 1021 and under the California Contractors License Law found at Business and Professions Code Section 7000 et seq.

# 11. Unfair Competition Prohibited

CONTRACTORS/ subcontractors are prohibited from engaging in unfair competition as specified under Business and Professions Code Sections 17200 to 17208:

# 12. Workers Compensation Insurance

Labor Code Section 1861 requires that CONTRACTORS and subcontractors be insured properly for Workers Compensation.

### 13. OSHA

Contractors and subcontractors are required to abide by the Occupational Safety and Health laws and regulations that apply to the particular construction project.

### 14. Employment Diversity

The requirement to demonstrate employment diversity in the hiring of women and ethnic groups as outlined in the Public Contracts Code Section 10115 and in the DISTRICT contract.

The CONTRACTOR's and subcontractors present at the meeting will be given the opportunity to ask questions of the LCP representative relative to the items contained in the Labor Law Requirements Checklist. The checklist will then be signed by the CONTRACTOR's representative and the DISTRICT's LCP representative, a representative of each subcontractor, and the DISTRICT's LCP representative.

At the meeting, the DISTRICT's LCP representative will provide the CONTRACTOR with a copy of the DISTRICT's LCP package.

It will be the CONTRACTOR's responsibility to provide copies of the LCP package to all of the CONTRACTOR's listed subcontractors and to any substituted subcontractors.

# **END OF SECTION**

### **SECTION 01 32 16 - CONSTRUCTION PROGRESS SCHEDULE**

### PART 1 - GENERAL

### 1.01 SUMMARY

- A. Provide a Construction Progress Schedule ("schedule") for the entire Work, including all necessary and related sub-schedules.
- B. Provide updates of the Schedule and periodic reports as required.

### 1.02 RELATED SECTIONS

- A. Summary of the Work
- B. Work Sequence and Phasing
- C. Price and Payment Procedures
- D. Project Coordination and Meetings
- E. Submittals
- F. Testing and Inspection
- G. Contract Closeout

### 1.03 FORM AND CONTENT OF SCHEDULES

- A. Schedule shall be in the form of a computer-generated Critical Path Method (CPM) network in Precedence Diagram Mode (PDM) showing all construction activities required to complete the Work of the Project within the Contract Time and any DISTRICT-defined Milestones.
- B. Schedule shall include but not be limited to the following:
  - 1. Complete sequence, with start and completion dates, of each and every activity of construction or element of the construction process.
  - 2. Phases of construction, with start and completion Milestones, as well as any other Milestones defined by the DISTRICT.
  - 3. Critical submittals, including DISTRICT and ARCHITECT review and approval periods, including 15 workdays for the first submittal (10 days for resubmittal), 21 days when the ARCHITECT's Consultants must review, and 30 days for review of submittals of Structural Steel, Door Hardware, and Steel Doors and Frames.
  - 4. Procurement, manufacture and/or fabrication; testing and delivery to the Project site of special long-lead-time material and equipment.
  - 5. Operational start-up, test and balance, performance testing, and training of operators for systems and equipment; for Substantial Completion and for Final Completion.
  - 6. Temporary facilities; construction of mock-ups, prototypes and/or samples; punch list; interfaces with Separate Work Contracts; and

- regulatory agency approvals and permits required for performance of the Work.
- 7. Deferred Approvals by the Department of the State Architect (DSA), allowing a minimum of ninety (90) days for all Deferred Approval items.
- 8. DISTRICT interfaces and owner-furnished equipment, either installed by CONTRACTOR (OFCI) or by OWNER (OFOI).
- 9. Decision dates for products specified by allowances, selection of finishes, and other ARCHITECT- or DISTRICT-furnished schedules or decisions.
- C. Schedule shall be updated periodically as specified to show progress of each activity and all changes since the previous submission, including:
  - 1. Major changes in scope.
  - 2. Activities modified since previous updating.
  - Revised projections due to changes.
  - 4. Other identifiable changes.

### 1.04 QUALITY ASSURANCE

- A. Scheduler: Contractor's personnel or consultant specializing in CPM scheduling shall have five (5) years minimum experience in scheduling construction work of the size and complexity comparable to this Project, including use of Primavera P6 or other as approved by DISTRICT, and shall have use of computer facilities with high-speed Internet access. DISTRICT must approve the Scheduler's resume, experience, and demonstrated skills.
- B. Contractor's Administrative Personnel: Two years minimum experience in using and monitoring CPM schedules on comparable projects.

### 1.05 SCHEDULE SUBMITTALS

- A. CONTRACTOR shall submit Construction Progress Schedules as follows:
  - 1. Preliminary Schedule: Submit a preliminary Baseline Schedule within fourteen days after Notice of Award. DISTRICT will review the Preliminary Schedule and return comments within ten workdays.
  - 2. Initial Schedule: Revise the preliminary Schedule and resubmit within ten days, to provide the Project's Baseline Schedule.
  - 3. Monthly Schedule Update: While retaining the Baseline Schedule, revise copies to show actual construction progress to date, and submit at scheduled monthly dates, or as otherwise required by the DISTRICT.
  - 4. In the event that the progress of the Work or the sequencing of the activities of the Work differs significantly from that indicated in the Baseline Schedule, the Contractor shall submit a Recovery Schedule to the DISTRICT, demonstrating the Contractor's plan to recover lost time, achieve all contractual milestones, and complete the work within the

Contract Time. Appropriate recovery actions include, but are not limited to, assignments of additional labor or equipment, shift or overtime work, expediting of submittals or deliveries, overlapping of activities, or sequencing changes to increase activity concurrence. An accompanying narrative shall describe the cause of the problems and the actions planned by the Contractor to recover the schedule. The DISTRICT will review the Recovery Schedule and provide comments, leading to approval of the schedule.

B. CONTRACTOR shall include with schedule submittals a written narrative report sufficiently comprehensive to explain the basis and determination of CONTRACTOR's approach to the Work, including but not limited to: activity durations; manpower flow; average crew sizes; equipment requirements; production rates; potential problem areas; permits; all necessary coordination with authorities, utilities suppliers, Separate Work Contracts, and other parties; and long lead delivery items requiring more than thirty (30) days from the date of order to delivery on the Project site. Report shall define problem areas, anticipated delays, or other factors having an impact on the Schedule.

### 1.06 SCHEDULE REQUIREMENTS

- A. Schedule shall represent CONTRACTOR's plan to complete the Work within the Milestones and/or Contract Time. However:
  - 1. A schedule extending beyond the Milestones and/or Contract Time will not be acceptable.
  - A schedule indicating Work completed in less than the Milestones and/or Contract Time will not be acceptable. CONTRACTOR shall indicate any available float.
  - 3. A schedule found unacceptable by the DISTRICT shall be revised by CONTRACTOR and resubmitted within five (5) days.
- B. Schedule shall be in sufficient detail to assure adequate planning and execution of Work, including but not limited to:
  - 1. Start and completion of all items of Work and their major components, and all designated dates identified as Milestones by DISTRICT.
  - Construction activity durations shall be limited to no more than two
    reporting periods, with exception of fabrication and procurement activities,
    unless approved otherwise by DISTRICT. Activity durations shall be total
    of actual workdays to perform and complete that activity and shall not
    include consideration of weather impact on the activity.
  - 3. Activities for procurement, delivery, and installation of equipment, materials and other supplies, including time for submittals, reviews and re-submittals. Include decision dates for selection of finishes.

- 4. Time for fabrication and delivery of manufactured products for the Work, showing interdependence of procurement and construction activities.
- 5. Identify each activity with applicable CSI Specification Division number, and coordinate with the CONTRACTOR's approved "Schedule of Values." Include adequate breakdown of activities for the Mechanical and Electrical elements of the work, to enable accurate monitoring and to assure full coordination with DISTRICT operating personnel.
- 6. Each activity shall be capable of being cost and resource-loaded with the resulting cost total equal to the Contract Amount
- Activities shall include all associated interface activities contained within the Contract Documents including, but not limited to, DISTRICT maintenance-and-operations activities
- 8. Each activity shall be defined to permit reasonable monitoring and evaluation of progress in performance of the Work.
- 9. Activities shall include:
  - a. A description of what is to be accomplished and where.
  - b. Workday duration.
  - c. Responsibility code identifying the performing party for each individual activity.
  - d. Area of Work coded on each Work activity.
  - e. Phase of Work coded on each Work activity.
- 10. Network shall show continuous flow from left to right.
- 11. Network shall be capable of multiple sorts as required for DISTRICT review.
- 12. Program shall be capable of compiling monetary value of completed and partially completed activities, of accepting revised completion dates and re-computation of all dates and float.
- 13. Contractor shall not sequester float through strategies such as extending activity durations estimates to consume available float, using preferential logic, using extensive or insufficient crew or resource loading, or by using float-suppression techniques, special lead or lag logic restraints, or imposed dates.
- 14. Identify days per week and shifts per day worked; also, non-work days and holidays.

- 15. Identify activities that constitute controlling operations, i.e., Milestones or Critical Path.
- 16. DISTRICT may require additional coding of activities.
- C. Notwithstanding acceptance of the Schedule, failure to identify and/or include any element of the Contract into the Schedule shall not release CONTRACTOR from obligation of completing all required Work in accordance with the Contract Completion Date or any Milestones.
- D. Submittal of the Schedule shall constitute CONTRACTOR's confirmation that the Schedule meets the requirements of the Contract Documents, and the Work will be executed in the sequence indicated in the Schedule.

### 1.07 COST LOADING OF SCHEDULE

- A. The Contract Schedule shall contain sufficient detail and information so that the CONTRACTOR can cost load the schedule in accordance with the District's coding structure.
- B. Power, security, telephone, PA/intercom, data, clock, video, fire alarm, and HVAC controls cabling shall not be scheduled together in an activity.
- C. The CONTRACTOR shall assign a cost value to each activity consisting of the sum of labor, material, equipment, overhead, profit, and general conditions costs allocable to that activity. The sum of all such values assigned shall equal the Contract total.
- D. Unless authorized by the DISTRICT, no site-related activity shall have a value exceeding \$100,000. The CONTRACTOR shall not cost load procurement and submittal related activities, unless authorized by the DISTRICT.
- E. For site-stored materials that are eligible for payment as provided by the Contract Documents, the Contractor may load the value of the materials on a one-day delivery activity. Payment for uninstalled materials is limited to major pieces of equipment with a cost value in excess of \$10,000. If the Work includes items covered by allowances, the Contractor shall include one activity in the schedule for each allowance that is loaded with the cost of that allowance. The scheduling of the allowance activities shall reflect the Contractor's best estimate of the duration and sequence of the Work.
- F. Upon District approval of a Change Order, the Contractor shall add separate cost-loaded activities to the Contract Schedule for each Change Order individually. If the DISTRICT so determines, the Contractor must further divide each Change Order as necessary to comply with the District's cost coding system.

### 1.08 REQUIREMENTS FOR UPDATING AND REVIEW

A. Prepare updated Schedule by one of following methods:

- 1. When updating current Schedule with actual Work progress only (non logic changes), status current Schedule with actual start and finish dates, remaining durations, and percent completion of cost and resource loaded activities and submit to DISTRICT and ARCHITECT for review.
- When updating current Schedule with logic changes, Construction
   Directives, Change Orders, delay / disruption activities, or recovery plans,
   prepare an explanatory description or computer-generated fragnet for
   DISTRICT and ARCHITECT review.
- When Work is associated with a Change Order, the adjustments shall be resource-loaded with material unit quantities, corresponding cost account, resources account codes, activity description, accepted costs and time adjustments. The activity ID number shall identify the number of the Change Order.
- B. CONTRACTOR shall attend weekly and monthly Schedule review meetings in order to accomplish the following:
  - 1. Discuss actual activity start and/or completion dates and any applicable variances, forecast activity start and/or completion dates and any applicable variances, progress of all activities underway at the time of the review, and to plan remedial action to mitigate schedule variances.
  - 2. Identify activities modified by CONTRACTOR since last update.
  - 3. Indicate changes that may be required to maintain the Milestones and Date of Completion.
- C. Submit updated schedules:
  - 1. With each Application for Payment.
  - 2. After Milestones, changes in scope, major delays, or other significant points in the construction process.

### 1.09 FAILURE TO COMPLY WITH REQUIREMENTS

- A. If CONTRACTOR fails to comply with the specified requirements, DISTRICT reserves the right to engage an independent scheduling consultant and/or provide its own expertise to fulfill these requirements, and shall be entitled to recover by assessment all incurred costs for the services from the CONTRACTOR.
- B. Submittal of any Schedule is subject to review and acceptance by ARCHITECT and/or DISTRICT. DISTRICT retains the right to withhold progress payments in whole or part until CONTRACTOR submits a Schedule acceptable to DISTRICT.

### 1.10 RECORD DOCUMENTS

A. Prior to Final Completion of the Work, CONTRACTOR shall submit as-built report and time-scaled network diagram reflecting as-built Project critical paths.

### **END OF SECTION**

### **SECTION 01 33 00 - SUBMITTAL PROCEDURES**

### PART 1 GENERAL

### 1.01 SECTION INCLUDES

- A. Submittal Procedures
- B. Shop Drawing Submittals
- C. Product Data Submittals
- D. Samples Submittals
- E. Manufacturers' Instruction Submittals
- F. Manufacturers' Certificate Submittals
- G. Coordination Drawing Submittals
- H. DSA Deferred Approvals

### 1.02 SUBMITTAL PROCEDURES

- A. Provide submittals wherever required by other sections of this Specification. Transmit ONE (1) electronic copy, and hard copy as requested, of each submittal directly to the ARCHITECT/IOR/DISTRICT/DISTRICT REPRESENTATIVE on forms prescribed by the ARCHITECT, with a copy of the transmittal form to the DISTRICT. Clearly identify each item submitted. Sequentially number the transmittal forms. For re-submittals use original submittal number with an alphabetic suffix.
  - 1. Include ARCHITECT's job number as it appears on Contract Documents, and all information required by the prescribed form.
  - 2. Include state agency application or approval number.
  - 3. Bind drawing and data submittals sturdily, clearly label covers.
- B. Identify Project, CONTRACTOR, subcontractor or supplier; pertinent Drawing sheet and detail number(s) and specification Section number, as well as name and telephone number of individual who may be contacted for further information.
- C. Determine and verify all field dimensions and conditions, materials, catalog numbers and similar data.
- D. Provide space for CONTRACTOR and ARCHITECT review stamps.
- E. Apply CONTRACTOR's dated stamp with CONTRACTOR's original signature or initials, certifying that review, verification of Products, field dimensions, adjacent construction Work and coordination of information is in accordance with the requirements of the Work and Contract Documents. Stamped signatures or initials are not acceptable.
  - F. Identify clearly, on the submittal and the transmittal form, any changes or variations from the Contract Documents. State effect of changes on Construction Progress Schedule and changes required in other Work or products (including "no effect"). Any change not so noted, even though stamped reviewed, will not

- be considered approved. Specific written approval by the ARCHITECT must be provided for any deviation from the Contract Documents.
- G. Revise and resubmit submittals as required; identify all changes made since previous submittal.
- H. Coordinate as required with all trades and all public agencies involved.
- I. Unless otherwise specifically authorized by ARCHITECT, make all submittals in groups containing all associated items. ARCHITECT may reject partial submittals as not complying with the provisions of this section.

### 1.03 SCHEDULES FOR SUBMITTALS

- A. Schedule submittals in accordance with the Construction Progress Schedule, far enough in advance of scheduled dates of installation to provide required time for the review and approval process, including possible revision and resubmittal and for placing orders and securing delivery.
- B. Within thirty (30) days from the Notice of Award, or in accordance with the Project Schedule, whichever is sooner, submit to the ARCHITECT and the DISTRICT for review and acceptance a "Schedule for Submission of Shop Drawings, Product Data, and Samples" ("Submittal Schedule") listing all submittals with planned dates of submission and return approved.
- C. Submittal Schedule will be incorporated into the Construction Progress Schedule. Update and submit revised schedule not less often than monthly.
- D. Allow in the Submittal Schedule sixteen (16) days after receipt for the ARCHITECT's review, both for initial submittals and for resubmittals; more for complex changes.

### 1.04 SHOP DRAWINGS

- A. Shop Drawings shall include fabrications and installation drawings, setting diagrams, schedules, patterns, templates and similar drawings. Include the following information:
  - 1. Dimensions.
  - Identification of products and materials included.
  - Compliance with specified standards.
  - 4. Notation of coordination requirements.
  - 5. Notation of dimensions established by field measurement.
- B. Sheet Size: Except for templates, patterns and similar full-size drawings, submit Shop Drawings on sheets at least 8-1/2-inch x 11 inch, but not larger than 30-inch x 42 inch.

- C. Stamp: Each page of shop drawings shall bear the CONTRACTOR's stamp, which shall signify the CONTRACTOR's representation that he has determined and verified materials, field measurements and field construction criteria related thereto, or will do so, and has checked and coordinated the information contained in the shop drawings. Each stamp shall be accompanied by a wet signature or initial of an employee of the CONTRACTOR who may be contacted for information. Stamped signatures or initials are not acceptable.
- D. Review Process: Make initial submittal of one (1) electronic copy and hard copy as required by AOR of each shop drawing. Comments or corrections will be noted on the reproducible and returned to the CONTRACTOR. If resubmittal is required, CONTRACTOR shall identify all changes made since the previous submittal and resubmit in the same manner. ARCHITECT will stamp or note drawings as follows:
  - 1. "NO EXCEPTION TAKEN" indicating that construction or fabrication may proceed.
  - 2. "MAKE CORRECTIONS NOTED" indicating that no resubmittal is required contingent upon corrections being made.
  - 3. "REJECTED" or "REVISE & RESUBMIT" indicating that corrections shall be made and drawings resubmitted for review.

After the final review, the CONTRACTOR shall copy and distribute the stamped drawings to the ARCHITECT.

- E. The ARCHITECT will review shop drawings for conformance with the requirements of the Contract Documents. The ARCHITECT's favorable review of a separate item shall not indicate acceptance of an assembly in which the item functions.
- F. The ARCHITECT's review of shop drawings shall not relieve the CONTRACTOR of responsibility for any deviation from the requirements of the Contract Documents unless the CONTRACTOR has informed the ARCHITECT in writing of such deviation at the time of submission and the ARCHITECT has given written acceptance to the specific deviation. The ARCHITECT's favorable review shall not relieve the CONTRACTOR from responsibility for errors or omissions in the shop drawings.
- G. No portion of work requiring shop drawings shall be commenced until the shop drawings have been returned with a favorable review by the ARCHITECT.
- H. ARCHITECT's CAD Drawings: The CONTRACTOR may request the use of the ARCHITECT's computer-generated drawings for use in preparing shop drawings. If the ARCHITECT approves this request, any costs incurred must be paid by the CONTRACTOR to the ARCHITECT. The CONTRACTOR must assume all liability for the accuracy and completeness of the shop drawings so prepared, and must hold the ARCHITECT harmless. The request must be in writing to the ARCHITECT, specifying the format and media requested.

#### 1.05 PRODUCT DATA

- A. Product Data includes manufacturers' standard drawings, catalogs, certificates of conformance, substantiating calculations, and similar relevant data as specified in individual Specification sections.
- B. Submit six (6) copies loose-leaf in binders, to facilitate copying of individual sheets. Provide the CONTRACTOR's stamp on the cover sheet of each submittal.
- C. Mark each copy to identify applicable products, models, options and other data. Supplement manufacturers' standard data to provide information unique to this Project.
- D. Review process, corrections, final distribution, and other conditions shall be similar to that for Shop Drawings.

### 1.06 MANUFACTURER'S CERTIFICATES

- A. When specified in individual specification sections, submit manufacturers' certificates to ARCHITECT for review in quantities specified for Product Data.
- B. Indicate material or product conforms to or exceeds specified requirements. Submit supporting reference data, affidavits and certifications as appropriate.
- C. Certificates may be recent or previous test results on material or product, but must be acceptable to ARCHITECT.
- D. Review process, corrections, final distribution, and other conditions shall be similar to that for Shop Drawings.

### 1.07 COORDINATION DRAWINGS

- A. The CONTRACTOR shall prepare and submit for review Coordination Drawings of all major spaces. Coordination Drawings indicate routing, locations, sizes, types and numbers of components for each class of work in concealed spaces where potential conflict may occur between structures, mechanical, electrical, fire sprinklers, communications and ceiling suspension systems. They include both plans and section drawings. (See also the General Requirements Section "PROJECT COORDINATION & MEETINGS.")
- B. Show all systems components, including plan locations of all ceiling penetrations and surface-mounted items. Provide cross sections wherever necessary to indicate proper support of ceilings and non-interference with work of other sections of the specifications. Cross sections shall indicate coordination required and proposed solutions for routing of elements where potential conflict exists.
- C. Drawings shall be based on field measurements, shop drawings and product data. They shall be prepared early enough to allow time to identify and resolve conflicts without delaying the progress of the Work. Conflicts shall be brought to

- ARCHITECT's attention immediately, together with CONTRACTOR's recommendations for resolution.
- D. Submit the Coordination Drawings in a scale of not less than 1/8" = 1' 0," with necessary sections and profiles at an appropriate, clearly readable enlarged scale. Submit the coordinated drawings as one reproducible and two prints.
- E. The ARCHITECT will review the submittals, make appropriate notations and comments to ensure the solutions meet the intent of the Contract Documents, and then return to CONTRACTOR for implementation.
- F. It shall be the responsibility of the CONTRACTOR to assure that all fabricators and installers of work involved in the Coordination Drawings be informed, consulted and advised in sufficient advance time to arrive at solutions where no extension of contract time or extra cost to the DISTRICT will be involved.

### 1.08 DSA DEFERRED APPROVALS

- A. Installation of deferred approval items shall not be started until detailed plans, specifications and engineering calculations have been accepted and signed by the ARCHITECT or Engineer in responsible charge of design and signed by a California registered ARCHITECT or Professional Engineer who has been delegated responsibility covering the work shown on a particular plan or specification, and the design has been approved by the Division of the State Architect (DSA). Deferred approval items for this project are the following:
  - 1. n/a
- B. Deferred approval drawings and specifications become part of the approved documents for the project when they have been approved by the Division of the State ARCHITECT.
- C. Submit electronic files and hard copies of each drawing as required by AOR and DISTRICT.
- D. Submit color electronic copies of calculations, product data and test reports and hardcopies as required by AOR.
- E. Identify and specify all supports, fasteners, spacing, penetrations, etc. for each of the deferred approval items, including calculations for each and all fasteners.
- F. Submit documents to the ARCHITECT for review prior to submitting to the Division of the State Architect. Submission shall be made within 30 days of the award of contract.
- G. Documents shall bear the stamp and signature of the Structural, Mechanical, or Electrical Engineer licensed in the State of California who is responsible for the Work shown on the documents.

- H. ARCHITECT will submit the documents as appropriate to the Project Structural, Mechanical and Electrical Engineers for review. Their review shall only be for conformance with the design intent shown in the Contract Documents.
- J. After review by ARCHITECT, forward submittal to the Division of the State Architect for approval, with copy of the transmittal to the DISTRICT.
- K. Respond to review comments made by the Division of the State Architect and revise and resubmit submittal to DSA for final approval. Provide copies of the DSA-approved documents to the ARCHITECT and the DISTRICT.

**END OF SECTION** 

# SECTION 01 33 00 – 1 -SUPPLEMENTAL SUBMITTAL PROCEDURES (For Projects involving DEMOLITION and/or hazardous materials abatement)

### PART 1 - GENERAL

### 1.01 SECTION INCLUDES:

- A. Supplemental Submittal Procedures
- **B.** Requests for Information
- C. Startup Submittals
- **D.** Outline of Contractor's Technical Execution Plan

#### 1.02 SUPPLEMENTAL SUBMITTAL PROCEDURES:

- A. Contractor shall prepare and transmit two copies of each of the following Submittals to the DISTRICT REPRESENTATIVE:
  - 1. Contractor shall submit the initial Project Schedule as discussed in this Section.
  - Contractor shall submit the Contractor's Health & Safety Plan (HASP) as discussed in this Section.
  - 3. Contractor shall submit a Technical Execution Plan (Work Plan) as discussed in this Section.
  - 4. Contractor shall submit all required written notifications to regulatory agencies at least 10 days prior to commencing any demolition and/or hazardous materials abatement activities. Copies evidencing proper and timely filing of these notifications shall be submitted to the District representative. Subject regulatory agencies include but are not limited to:
    - **a.** South Coast Air Quality Management District (SCAQMD)
    - **b.** Cal/OSHA
    - **c.** California Department of Public Health (CDPH)
  - **5.** Contractor shall submit Contractor's Daily Construction Report electronically by 10:00 A.M. the next Working day. Daily report shall include:
    - a. Summary of day's activities.
    - **b.** Summary of corrective actions that were taken to improve site safety, security, and erosion and sediment control BMPs.
    - **c.** Summary of materials imported and exported
    - d. Listing of equipment that was mobilized or demobilized
    - **e.** Summary of any safety issues.
    - **f.** Any "Near Miss" observations for the day.

- g. Summary of Daily Safety Meeting
- **h.** Summary of day's weather conditions
- i. Total number of personnel onsite for the day
- j. Listing of personnel onsite
- **k.** Listing of subcontractors onsite
- I. Listing of visitors onsite
- **6.** Contractor shall submit weight tickets, Bill of Lading documents, generators initial waste manifest copies and any other form of shipping documents on a daily basis as specified in other Sections of the Specifications.
- **7.** Contractor shall submit a two week look-ahead schedule at each weekly construction meeting.
- **8.** Contractor shall submit biweekly revisions and updates of Progress Schedule and Technical Execution Plan as required by the DISTRICT REPRESENTATIVE.
- **9.** Contractor shall submit monthly Health and Safety reports, as specified in Specifications Section 01-45-15 Health and Safety Requirements.
- **10.** Contractor shall submit weekly safety reports.
- 11. Contractor shall submit equipment inspection logs on a daily basis. To be included in the Daily Construction Report.
- **12.** Additional submittals as described in the Specifications.
- **B.** Contractor shall transmit each Submittal to the DISTRICT REPRESENTATIVE at the Project Site. Each submittal will be reviewed and returned with one of the following Classifications:
  - 1. *No exceptions taken*; Contractor may proceed with the work.
  - **2.** Conformed as Noted: Contractor may proceed with the work subject to the comments and/or notes on the Submittal. Re-submittal is not required.
  - **3.** Revise and Resubmit: Contractor may not proceed with the work. Re-submittal is required for certain items.
  - 4. Rejected
- Contractor shall develop a submittal register for review at weekly progress meetings. Register shall be based on submittals listed in the Submittal Summary following this section, requirements throughout the Specifications and additional items as deemed necessary by the Construction Manager or Engineer. In the event a submittal is not listed in the Submittal Summary it does not relieve the Contractor from the responsibility to provide such submittal.
- **D.** Contractor shall submit copies (with all signatures affixed) of all waste manifests, weigh tickets, Certificates of Destruction, and other shipping documentation.

- E. Contractor shall transmit each Submittal with a cover letter signed by Contractor's Project Superintendent. Contractor shall, by signing each Submittal, certify that Contractor has reviewed the Submittal, and that the submitted information conforms to the requirements of the Work and these Specifications.
- F. Contractor shall sequentially number the transmittals (e.g., Submittal No. 001). Contractor shall number revised Submittals with original number and a sequential alphabetic suffix (e.g., Submittal No. 001a).
- **G.** Each Submittal shall include Project title, Contractor, Subcontractor or Supplier, title of Submittal, Specifications Section number and, if applicable, Drawing number.
- **H.** Submittals that do not conform to the requirements of the Specifications shall be returned with a notation of deficiencies. Contractor shall revise to correct noted deficiencies and resubmit. When revised for resubmission, Contractor shall identify all changes made since previous submission.
- I. Submittals must be submitted to the Engineer prior to the execution of work that requires approval of submittal(s) associated with that work.
- J. The Engineer shall be allowed an ample amount of time to review, supply comment, and provide additional review and approval of all submittals.
- **K.** Submittals not required by the Specifications shall not be recognized or processed.

### 1.03 REQUESTS FOR INFORMATION:

- A. Contractor shall submit all Requests for Information (RFI) to the Engineer in writing. Requests for information shall be numbered sequentially and shall include the related Specifications Section number or Drawing number.
- **B.** RFIs shall be used for the purposes of providing clarification, proposing an alternative procedure or method, and providing a platform for discussion with regards to any changes in the work or proposed changes in the work.
- C. The Engineer will provide any revisions to the Specifications or Drawings in writing.
- **D.** Contractor shall request written confirmation of any interpretations or clarifications provided verbally by the Engineer.

### 1.04 STARTUP SUBMITTALS:

- A. This paragraph specifies Submittals that Contractor shall prepare and transmit prior to commencing the Work at the Project Site. Additional Submittals are specified in other Sections of these Specifications.
  - Contractor shall submit the initial Project Schedule. The Project Schedule shall identify
    milestones and shall be consist with the Contractors TEP with regards to Abatement
    and Demolition sequence. The Project Schedule shall be prepared using Microsoft
    Project software.
    - 2. Contractor shall submit the Contractor's HASP as specified in Specifications Section 01415 – Health and Safety Requirements, including documentation of worker's OSHA training and medical monitoring and the name and qualifications of the full-time Site Safety and Health Officer.

- Technical Execution Plan: Certain elements of the Work require the Contractor to provide detailed written information for review, comment, and approval by the Engineer prior to the execution of the work. Prior to commencing any work on the site, Contractor shall submit a draft Technical Execution Plan (TEP), conforming to the outline specified in Paragraph 1.05, for the Engineer's review and comment. Contractor shall revise the draft TEP as requested by the Engineer and submit a final TEP, subject to the Engineer's review, approval, and acceptance, prior to commencing Work. Any material changes in the Work, process, staffing, major equipment or materials will require a TEP amendment and review and approval by the Engineer.
- 4. Contractor shall file required notifications of abatement and/or demolition activities with appropriate regulatory agencies prior to initiation of regulated site activities. Copies of all such notifications shall be submitted to Engineer concurrent with Contractor's submittal to said regulatory agencies.
- 5. Contractor shall provide for Engineer's approval the name and qualifications of Subcontractors providing any sampling, laboratory analyses, geotechnical, material testing or surveying services as required in the Specifications and/or contract documents.

#### 1.05 OUTLINE OF CONTRACTOR'S TECHNICAL EXECUTION PLAN

- A. Contractor shall prepare and submit a Proposed Technical Execution Plan to the Engineer at least 10 days prior to the planned start of site work. The Technical Execution Plan shall, at a minimum, include the following sections:
  - 1. Project Coordination.
    - **a.** Detailed Project staffing plan showing staffing levels for each task and phase of Work. Note: No undocumented workers are allowed to work on site.
    - **b.** Resume of key project staff including proposed Project Superintendent(s).
    - **c.** List of all proposed subcontractors, including hazardous material abatement companies, transportation companies, and disposal facilities.
    - **d.** List of major equipment and materials.
    - **e.** List of Permits and Approvals to be obtained by Contractor, including contact names, titles, and phone numbers.
    - **2.** Progress Schedule.
      - **a.** Include Contractor's initial Baseline Project Schedule, including line items for all major project work elements.
  - 3. Construction Facilities and Temporary Controls.
    - **a.** Locations, sizes, and requirements for utility services.
    - **b.** Layout of Work Zones
    - **c.** Proposed design of Abatement Enclosures and Decontamination Stations.

**d.** Decontamination Methods and Equipment.

Procedures to prevent contamination of clean areas.

Vehicle decontamination and inspection procedures.

Procedures for collection, treatment, and disposal or discharge of decontamination residuals and used PPE.

#### 4. Erosion and Sediment Control

- **a.** Proposed detail description and design of Erosion and Sediment controls to be used during demolition activities.
- b. The Contractor shall implement storm water best management practices in general conformance with the requirements of the California Storm Water Quality Association (CASQA) Storm water Best Management Practice Handbook, Revised August 2011.
- **5.** Site Security Requirements.
  - **a.** Detailed plan describing site security measures to be used during working and non-working hours to prevent unauthorized access to the property.
- Notifications and Permits.
  - **a.** List of all required Permits and Notifications.
  - **b.** Proof of submittal of required notifications to appropriate regulatory agencies (Demolition and abatement notifications, etc.)
  - **c.** Description of information and assistance required for Contractor to obtain above-referenced notifications and permits.
- **7.** Asbestos and Lead Paint Abatement and Regulated Waste Work Plan. (If applicable)
  - a. Identify proposed asbestos and lead paint abatement subcontractor to be used, if work will not be performed by Contractor. Include detailed work procedures to be used in the removal and demolition of the asbestos containing material, lead paint material, and universal waste. Contractor shall review the Asbestos Specifications and Hazardous Materials Survey Report provided in the Appendices. The work plan will identify proposed asbestos, lead paint, and regulated waste disposal and recycling facilities. Included within the plan written certification as specified herein that employees have received appropriate training regarding hazards of asbestos and lead paint exposure, respirator use, personnel decontamination, procedures and OSHA regulations. The Contractor shall provide proof of appropriate licenses to perform the Work. Additional information to be submitted from Asbestos abatement contractor are:
  - **b.** Equipment List;
  - **c.** Proof of Worker Training and required Medical Examinations;

- d. Proof of employee respirator fit testing, and
- **e.** Contractor and subcontractor's Respiratory Protection Plans.
- **8.** Building Demolition Plan.
  - **a.** Proposed sequence of demolition work.
  - **b.** Planned methods and equipment to be used for demolition activities.
  - **c.** Procedures and measures to ensure that workers as well as unauthorized personnel, both during and outside of working hours are safely away from the buildings especially during active demolition.
  - **d.** Description of utility termination process.
  - **e.** Proposed methods of protecting in place specified utilities.
  - f. Proposed debris and salvage stockpile locations.
  - **g.** Equipment and procedures to be employed to control dust and noise.
  - **h.** Dust and noise monitoring procedures.
- 9. Stockpile Management and Loading.
  - **a.** Show on-site truck routes and loading areas for off-site transportation.
  - **b.** Truck loading areas, staging areas for incoming empty trucks.
  - **c.** Coordination of demolition, stockpiling, and loading.
- **10.** Off-site Transportation and Traffic Control Plan.
  - **a.** Provide an estimate, by day, of the expected quantities of material to be shipped from the site. Describe the number of trucks to be used, the expected turn-around-times, and the expected number of trips per day.
  - **b.** If onsite truck scales are employed, describe locations and equipment to be used to weigh haul trucks. Include frequency for obtaining true weight of trucks.
  - c. Provide a Traffic Control Plan showing how trucks will enter and exit the site, the location of flaggers and signs, designated haul route to and from the off-site disposal facilities with posted speed limits, warnings, etc., and incident reporting procedures for trucking related incidents. The Traffic Control Plan shall establish truck staging area located in area approved by District representative and/or as detailed in contract documents.
  - **d.** Provide a plan for verifying the accuracy of weight scales if employed.
  - **e.** Describe procedures to inspect trucks and loads before leaving the site to ensure nothing falls off the truck on roads and highways.
- 11. Site Cleanup

- **a.** Describe the method for site clean-up activities and disposal of materials (fine debris, trash, etc.). This includes, but will not be limited to, preventing cross contamination of waste streams and quality control procedures to ensure hazardous materials are not mixed with nonhazardous materials; Describe dust mitigation during site clean-up activities.
- **12.** Site Specific Health and Safety Plan (HASP).
  - **a.** Provide a HASP, including measures to be taken for operational and worker safety, protection of the general public, and measures to control exposure to airborne dusts, asbestos, lead-based paint, as well as hazards related to demolition activities.
  - **b.** Contractor's HASP shall designate a qualified, dedicated Site Safety and Health Officer (SSHO) to be present on the Project site during the Work.
- 13. Disposal and Recycling Facilities.
  - **a.** List of proposed disposal and recycling facilities to be used and their daily capacities for this project.

#### **END OF SECTION**

## **SECTION 01 35 16 - ALTERATION PROJECT PROCEDURES**

#### PART 1 GENERAL

#### 1.01 **SECTION INCLUDES**

- A. Products and installation for patching and extending Work.
- B. Transitions and adjustments.
- C. Repair of damaged surfaces, finishes and cleaning.

## PART 2 PRODUCTS

#### 2.01 PRODUCTS FOR PATCHING AND EXTENDING WORK

- Α. New Materials: As specified in product Sections; match existing products and work for patching and extending work.
- B. Type and Quality of Existing Products: Determine by inspection, and testing products where necessary, referring to existing work as a standard.

#### PART 3 EXECUTION

#### 3.01 **EXAMINATION**

- Verify that demolition is complete and areas are ready for installation of new Work. Α.
- B. Beginning of renovation work means acceptance of existing conditions.

#### 3.02 **PREPARATION**

- Move, cut, or remove items as necessary for access to alterations and renovation Α. work. Replace and restore at completion.
- B. Remove unsuitable material not marked for salvage, such as rotted wood, corroded metals and deteriorated masonry and concrete. Replace materials as specified for finished work.
- C. Remove debris and abandoned items from area and from concealed spaces.
- D. Prepare surface and remove surface finishes to provide for proper installation of new work and finishes.
- E. Close openings in exterior surfaces to protect existing work and salvage items from weather and extremes of temperature and humidity. Insulate ductwork and piping to prevent condensation in exposed areas.

#### 3.03 **INSTALLATION**

A. Coordinate work of alterations and renovations to expedite completion sequentially and to accommodate District occupancy.

- B. Remove, cut and patch work in a manner to minimize damage and to provide a means of restoring products and finishes to original or specified condition.
- C. Refinish visible existing surfaces to remain in renovated rooms and spaces, to specified condition for each material with a neat transition to adjacent finishes.
- D. Restore existing and remaining plumbing, heating, ventilating and air conditioning, electrical and fire alarm systems to full operating condition and advise ARCHITECT of any deficiencies discovered during the course of the work.
- E. Install products as specified in individual Sections.

#### 3.04 **TRANSITIONS**

- Α. Where new work abuts or aligns with existing, perform a smooth and even transition. Patched work shall match existing adjacent work in texture and appearance.
- B. When finished surfaces are cut so that a smooth transition with new work is not possible, terminate existing surface along a straight line at a natural line of division and make recommendation to ARCHITECT for a satisfactory resolution.

#### 3.05 **ADJUSTMENTS**

- Where removal of partitions or walls results in adjacent spaces becoming one, Α. rework floors, walls and ceilings to a smooth plane without breaks, steps or bulkheads.
- B. Where a change of plane of 1/4 inch or more occurs, request instructions from ARCHITECT.
- C. Trim existing doors as necessary to clear new floor finish. Refinish trim as required.
- D. Fit work at penetrations in fire-rated assemblies as specified in "Cutting and Patching" section.

#### REPAIR OF DAMAGED SURFACES 3.06

- A. Patch or replace portions of existing surfaces which are damaged, lifted, discolored or showing other imperfections.
- B. Repair substrate prior to patching finish.

#### 3.07 **FINISHES**

- Α. Finish surfaces as specified in individual product Sections.
- B. Finish patches to produce uniform finish and texture over entire area. When finish cannot be matched, refinish entire surface to nearest intersections.

A. Conform to "Contract Closeout" requirements.

**END OF SECTION** 

#### **SECTION 01 45 00 - QUALITY CONTROL**

#### PART 1 GENERAL

#### 1.01 SECTION INCLUDES

- A. Quality assurance and control of installation
- B. Reference Standards
- C. Field Samples
- D. Mock-up
- E. Project Inspector
- F. Permits and Fees
- G. Verified Reports
- H. Manufacturers' field services and reports

#### 1.02 QUALITY ASSURANCE/CONTROL OF INSTALLATION

- A. Monitor quality control over suppliers, manufacturers, products, services, site conditions and workmanship to produce work of specified quality.
- B. Comply fully with manufacturers' instructions including each step in sequence.
- C. Should manufacturers' instructions conflict with Contract Documents, request clarification from Architect before proceeding.
- D. Comply with specified standards as a minimum quality for the Work except when more stringent tolerances, codes or specified requirements indicate higher standards or more precise workmanship.
- E. Perform work by persons qualified to produce workmanship of specified quality.
- F. Where experience minimums for workmen, applicators, companies or manufacturers are required in individual sections, written certification and documentation substantiating such minimums shall be submitted and approved by the Architect, when requested.
- G. Secure products in place with positive anchorage devices designed and sized to withstand stresses, vibration, physical distortion or disfigurement.

#### 1.03 REFERENCE STANDARDS

- A. Conform to reference standard by date of issue current on date of Contract Documents.
- B. Obtain copies of standards when required by Contract Documents.
- C. Should specified reference standards conflict with Contract Documents, request clarification from the Architect before proceeding.

D. The contractual relationship of the parties to the Contract shall not be altered from the Contract Documents by mention or inference otherwise in any reference document.

## 1.04 FIELD SAMPLES

- A. Install field samples at the site as required by individual specifications Sections for review by Architect.
- B. Accepted samples represent a quality level for the Work.
- C. Where field sample is specified in individual Sections to be removed, clear area after field sample has been accepted by Architect and is no longer required for reference.

#### 1.05 MOCK-UP

- A. Tests will be performed under provisions identified in this section.
- B. Assemble and erect specified items with specified attachment and anchorage devices, flashings, seals and finishes.
- C. Where mock-up is specified in individual Sections to be removed, clear area after mock-up has been accepted by Architect and is no longer required for reference.

#### 1.06 PROJECT INSPECTOR

- A. An Inspector, herein referred to as the "Project Inspector" or "Inspector of Record", will be employed by the District and approved by Office of Regulation Services, Division of State Architect (ORS/DSA) in accordance with Part 1, Title 24, Section 4-333, California Code of Regulations. His duties are described in Part 1, Title 24, Section 4-342, CCR. His duties are also required and defined in Sections 17309, 17311, 81141 and 81143 of the California Education Code as they relate to schools.
- B. The work of construction in all stages of progress shall be subject to the personal continuous observation of the Inspector of Record (IOR). He shall have free access to any or all part of the work at any time. The Contractor shall furnish the Inspector reasonable facilities for obtaining such information as may be necessary to keep him fully informed respecting the progress and manner of the work and the character of the materials. Inspection of the work shall not relieve the Contractor from any obligation to fulfill the requirements of this Contract.

## 1.07 VERIFIED REPORTS

A. Contractor shall comply with Part 1, Title 24, Sections 4-336 and 4-343, California Code of Regulations and issue verified reports through the Architect as required.

## 1.08 MANUFACTURERS' FIELD SERVICES AND REPORTS

- A. When specified in individual specification Sections, require material or Product suppliers or manufacturers to provide qualified staff personnel to observe site conditions, conditions of surfaces and installation, quality of workmanship, start-up of equipment, test, adjust and balance of equipment and as applicable and to initiate instructions when necessary.
- B. Manufacturers' Representatives shall report observations and site decisions or instructions given to applicators or installers that are supplemental or contrary to manufacturers' written instructions.
- C. Submit report of observation to Architect for review.

**END OF SECTION** 

#### **SECTION 01 45 15 - HEALTH AND SAFETY REQUIREMENTS**

#### PART 1 - GENERAL

#### 1.01 SECTION INCLUDES:

- **A.** Summary
- B. References
- C. Contractor's Responsibility for Health and Safety
- **D.** Submittals
- E. Notifications
- F. Equipment and Facilities
- **G.** Personal Protective Equipment
- H. Other Health and Safety Equipment
- **I.** Training
- J. Work Planning and Meetings
- K. Engineering Controls
- L. Monitoring
- M. Evaluation of Performance
- N. Site Security Other Safety Considerations
- O. Work by Others
- P. EHS Incident Report Form
- Q. Hot Work Permit Form
- R. Job Safety and Hazard Analysis Form

#### 1.02 SUMMARY:

**A.** This Section includes requirements for Health and Safety during performance of Work, including identification of applicable Laws and Regulations, Submittals, notification requirements, and Health and Safety Specifications.

#### 1.03 REFERENCES:

- **A.** Applicable regulations and publications include, but are not limited to, the following:
  - 1. ANSI, Emergency Eyewash and Shower Equipment, Z358.1, 1981.
  - ANSI, Practice for Occupational and Educational Eye and Face Protection, Z87.1, 1979.
  - 3. ANSI, Protective Footwear, Z41.1, 1983.
  - 4. DOT Standards and Regulations, 49 CFR 171, 49 CFR 172 and 49 CFR 214.
  - 5. NFPA, Flammable and Combustible Liquids Code, NFPA 30, most recent revision.
  - 6. OSHA, Title 29 CFR Part 1910, Occupational Safety and Health Standards, and Title 29 CFR Part 1926, Safety and Health Regulations for Construction Sites.
  - 7. USEPA, Health and Safety Requirements for Personnel Engaged in Field Activities, USEPA Order No. 14402.
  - 8. USEPA, Standard Operating Safety Guidelines, November 1984.
  - 9. Title 8, California Code of Regulations, Section 1532.1, Lead. Cal/OSHA
  - 10. Title 8, California Code of Regulations, Section 1529. Asbestos. Cal/OSHA
  - 11. Title 8, California Code of Regulations, Section 1529(I)(2), Asbestos Waste Disposal, Cal/OSHA
  - 12. South Coast Air Quality Management District (SCAQMD) regulations including but not limited to Rule 402, 403, and 1403)

Where two or more regulations/documents conflict, the one(s) offering the greatest degree of protection shall apply.

## 1.04 CONTRACTOR'S RESPONSIBILITY FOR HEALTH AND SAFETY:

- **A.** Contractor shall comply with any and all state, federal, and local Ordinances, Laws and Regulations.
  - Contractor shall be responsible for the Health and Safety of Contractor's employees, its Subcontractors, Suppliers, agents, inspectors, visitors, the general public, and any others associated with or interacting with Contractor who provides labor, goods, or other services on the Project site.
  - 2. Contractor shall be responsible for emergency response planning and notification, and for actual response to any and all emergencies that may occur during the course of the Work, including emergencies that may occur when Contractor is not present at the Project site.
  - 3. Contractor is responsible for communicating daily with the District Representative regarding Health and Safety issues for the District Representative's safe conduct of the District Representative's duties, but such communication shall not imply any duty or responsibility on the part of the District Representative with regard to Health and Safety of Contractor's employees, its Subcontractors, Suppliers, the general public, or others. The District Representative's responsibility and duty with regard to Health and Safety shall be limited to the District Representative's employees. Contractor shall have responsibility and duty to the District Representative to communicate Health and Safety issues accurately and in a timely manner to allow the District Representative to take appropriate actions to protect the District Representative's employees and the Owner's employees.
  - 4. Contractor shall designate a dedicated Site Safety and Health Officer (SSHO) on the Project site during the Work.
    - A. The SSHO duties and responsibilities shall be the prevention of accidents and the maintaining and supervising of safety precautions and programs associated with the Contractor's activities at the Project site. The designated SSHO shall be certified in applicable OSHA Construction Safety training. At a minimum, the designated SSHO shall have at least 1 year of experience as a SSHO on demolition and construction sites. Contractor's SSHO shall be solely dedicated to Health and Safety issues from the start of the site activities through completion.
    - B. The SSHO shall enforce the requirements of safety for all Contractor personnel onsite at all times. The SSHO shall ensure that all Contractor personnel, Subcontractor personnel, and Contractor visitors follow the Contractor's site Health and Safety Plan (HASP), including wearing the designated level of PPE. If the SSHO elects to require a higher level of protection than that specified in the District Representatives HASP, the extra costs associated with such higher level shall be borne by Contractor, unless such extra costs are approved in advance in writing by the District Representative.
    - **C.** Prior to mobilization and continually through the duration of the Work, the SSHO shall inspect the Project site and document area-specific and worker-specific protection requirements.
    - **D.** After mobilization, the SSHO shall monitor activities and shall document the need for additional worker protection as required, based on activities performed and Action Levels specified in the HASP.
    - **E.** The SSHO shall verify that all activities are performed in accordance with the HASP and all federal, state, local, and Health and Safety standards, Laws and Regulations, and guidelines.

- F. In the event of a health or safety risk, as determined by the SSHO or by other Contractor personnel or by the District Representative, Contractor shall not proceed with the Work until a method for handling the risk has been determined in consultation with the District Representative and implemented. Any health or safety risk resulting in a stoppage of Work shall be reported immediately to the District Representative.
- Contractor shall be responsible for implementing a behavior-based safety process and providing site training, observation, and feedback for Contractor personnel employed at the Site.
- District Representative shall provide the Contractor with a copy of the District Representative's HASP as a reference. Contractor shall be responsible for preparing their own HASP under which their employees shall work.

#### SUBMITTALS: 1.05

- Contractor shall prepare and submit a HASP to the District Representative as a part of the A. TEP. The Contractor shall follow all applicable local, state, and federal Health and Safety standards, Laws and Regulations, and guidelines implemented through, but not limited to, CalOSHA, SCAQMD, OSHA and USEPA. Where these are in conflict, the most stringent requirement shall be followed. The following points shall be addressed in the Contractor's HASP:
  - i. Names of key personnel and alternates responsible for Health and Safety, including a Contractor Health and Safety Representative and SSHO. The District Representative must approve the SSHO.
  - 2. A Health and Safety risk or Task Hazard Analysis (THA) associated with each portion of the Work (i.e., list potential hazards), including THAs for abatement, demolition, construction of retaining structures, loading and transportation of demolition debris and materials, decontamination, truck traffic, and restoration.
  - 3. A requirement that Contractor locate Underground Facilities by using Southern California "Dig Alert" procedures prior to the start of the Work.
  - 4. PPE to be used for each of the site tasks and operations being conducted, as required by Cal/OSHA and 29 CFR Subpart I, and 29 CFR 1926.
  - 5. Frequency and types of dust monitoring and instrumentation to be used by the Contractor, including methods of maintenance and calibration of monitoring and sampling equipment. Dust monitoring requirements will be determined by the South Coast Air Quality Management District (SCAQMD Rule 403).
  - 6. Corrective actions and upgrading of PPE based on monitoring of dust, with specific Action Levels identified.
  - 7. Site control measures in accordance with the control program required Cal/OSHA and OSHA.
  - Decontamination procedures in accordance with Specifications.
  - 9. An emergency response plan meeting federal, state, and local requirements for safe and effective responses to emergencies, including the necessary PPE and other equipment. Explanation of potential emergencies and contingency plan of action, including description of the route to the nearest appropriate hospital, hospital route map, and posting of emergency telephone numbers at the Project
  - 10. If confined space entry is required, include confined space entry procedures in accordance with Cal/OSHA Title 8, Subchapter 7 § 5157 Permit Required Confined Spaces and a list of all anticipated confined space entries required by Contractor in the course of the Work.
  - 11. A spill containment program meeting the requirements of all applicable local, state, and federal Health and Safety standards.

- 12. A list of Health and Safety and emergency equipment available on the Site.
- 13. A description of engineering controls used to reduce the hazards of equipment operation.
- 14. Training for emergency response procedures as outlined in the District Representative's HASP.
- 15. Heat stress program consistent with the references provided in the District Representative's HASP.
- 16. Cold stress program consistent with the references provided in the District Representative's HASP.
- 17. Lockout/Tagout where the operation of machinery and/or equipment in which the unexpected energization on start up or the release of stored energy could cause injury to personnel.
- 18. Measures in place to ensure accountability of the location of all workers onsite at all times specifically to prevent workers from entering into unsafe areas such as buildings being demolished;
- 19. Measures in place to verify that unauthorized personnel such as passersby and homeless personnel have not entered the site prior to beginning of demolition in the morning, after lunch or other times when the site is momentarily left unattended:
- 20. Securing the site each night to ensure there are no large pieces of metal, brick, concrete, etc. that might fall on workers during the next shift or unauthorized people that might enter the site after hours;
- 21. Ensuring that all holes and pits where personnel might fall or trip are covered, backfilled fenced or barricaded as needed to prevent injuries;
- 1.06 Contractor's Daily Construction Report, submitted in accordance with Specifications Section 01330, shall include a summary of daily safety issues and a summary of Contractor's Daily Safety Meeting.
  - A. Contractor shall submit weekly safety reports that include:
    - The names of all Contractor and Subcontractor personnel employed at the Site at any time during the week, and the names and duties of key personnel including Contractor's Project Manager, Project Superintendent, SSHO, and all competent personnel.
    - 2. A summary of all Health and Safety incidents describing any medical treatment that was provided during the week, the current Work status of any individuals affected the names of individuals who may have observed the incident, and actions taken by Contractor to address the unsafe act or unsafe condition.
    - 3. A summary of all Health and Safety near-misses or observations providing an opportunity for shared learning and future hazard avoidance. For any Health or Safety incident or near-miss, list the date, the nature of the incident or near-miss, and the names of individuals involved.
    - 4. The total number of labor hours worked at the Site during that week.
    - 5. Internal Health and Safety audits performed by the Contractor as part of the Contractor's HASP.
    - 6. Results of Contractor behavioral observation and feedback evaluations as described in the District Representative's HASP.
  - **B.** Contractor shall submit documentation of training and experience for the designated competent persons.
  - **C.** Contractor shall maintain all required and applicable training records on-site including, but not limited to those specified in Part 3.01 (A) of this Section.

- **D.** Contractor shall submit a Hot Work Permit for any torch cutting, or activities that generate sparks. If the Contractor does not have a permit format readily available, they may request a permit from the District Representative.
- **E.** Contractor shall conduct a THA for significant activities and submit the documentation to the District Representative for review prior to the start of the activities. Contractor's THA shall be submitted on the THA forms attached to this Section, or other form acceptable to the District Representative.
- F. Contractor shall submit copies of all daily equipment inspections completed.

#### 1.07 NOTIFICATIONS:

- A. Contractor shall immediately (within 30 minutes) verbally report to the District Representative the occurrence of any and all Health and Safety incidents. A Supervisor's Accident/ Incident Report (SAIR), which may be requested from the District Representative, shall be submitted within 24 hours of occurrence of the incident or issue.
- B. Contractor shall immediately and fully investigate any such incident or near-miss and conduct a root cause analysis, and shall submit to the District Representative, the Contractor's written corrective action plan for such incident within one day after the incident occurs in accordance with Specifications Section 01330 Submittal Procedures.
- C. Contractor shall notify the District Representative in writing at least 5 days prior to bringing any hazardous material, equipment, or process to the site, or using the same on the Site. Contractor shall provide the District Representative with a MSDS for all chemicals brought on to the Site.
- D. Contractor shall immediately notify the District Representative in writing of any hazard that Contractor discovers or observes on the site and corrective measures planned or taken to eliminate or minimize such hazard. Hazard reporting will be completed as a Near Miss Report as described in 1.05(C)(3) of this Section.

#### **PART 2 - PRODUCTS**

#### 2.01 EQUIPMENT AND FACILITIES:

**A.** Contractor shall provide all equipment, temporary facilities, and personnel required to perform activities onsite safely in accordance with all Laws and Regulations and standards, and with the Contractor's HASP.

## 2.02 PERSONAL PROTECTIVE EQUIPMENT:

- A. The appropriate level of PPE shall be determined by the Contractor for specific tasks as described in the Contractor's HASP. If hazards are identified that require a level of protection greater than Level D (defined in paragraph C below), Work shall be suspended and the District Representative notified. The Contractor's SSHO, in consultation with the District Representative, shall determine what actions are required prior to restarting Work. Contractor shall determine and document the appropriateness of suggested minimum PPE requirements for Contractor's employees and others at the Project site.
- B. Contractor shall furnish and maintain materials and equipment for the Health and Safety of Contractor employees, its Subcontractors, Suppliers, and visitor personnel. Contractor shall provide all required Health and Safety equipment, first aid equipment, tools, monitoring equipment, PPE, and ancillary equipment and methods required to ensure workers' Health and Safety and to comply with the Contractor's HASP. District Representative will furnish PPE and monitoring for District Representative's employees.
- C. Level D protection will be required at all times while onsite by all personnel and visitors. Level D PPE consists of:
  - 1. Hard hat
  - 2. Steel-toed boots

- 3. Safety glasses with permanent side shields
- 4. Work clothes (long pants, shirts with sleeves)
- 5. High visibility reflective safety vests
- 6. Hearing protection (as needed to prevent exposure exceeding 85 dB level) if noise level warrants.
- D. In most cases, Level D will be the maximum allowed level of PPE. Level C may be required as certain hazards are faced provided that personnel are properly trained and certified. Contractor shall notify District Representative immediately when upgrades to Level C are employed by the Contractor.
- E. In cases where the Owner requires additional PPE, the District Representative will notify the Contractor of these additional requirements in advance of mobilization so that Contractor may obtain the necessary equipment.

#### 2.03 OTHER HEALTH AND SAFETY EQUIPMENT:

- **A.** Contractor is required to have the following equipment available on the Site for the Health and Safety of Contractor, Subcontractors, Suppliers, and visitors:
  - 1. First aid kits
  - 2. Fire suppression equipment (appropriate to location and type of flammable materials present). Equipment will be certified ready for use within the previous twelve months and will also have been inspected each month; documentation supporting certification and inspections will be available for review.
  - 3. Emergency eyewash facilities meeting OSHA specifications
  - **4.** Other equipment or supplies as determined to be necessary or prudent by Contractor or the District Representative
  - **5.** Flammable liquids storage cabinet(s), if necessary
  - **6.** Fall protection equipment appropriate for the hazards on the project
  - 7. Heavy blankets

## PART 3 - EXECUTION

#### 3.01 WORKER QUALIFICATION:

- **A.** Contractor shall provide the following training to each worker, unless otherwise specified:
  - Cal OSHA, OSHA, AHERA, and or California Department of Public Health (CDPH)
    compliant worker training as required by regulations including but not limited to
    CCR Title 8, 29 CFR, and 40 CFR.
  - **2.** Current cardiopulmonary resuscitation (CPR) and first aid certification for at least two workers assigned to Work on the site.
  - **3.** Confined Space Entry Training for workers entering confined spaces.
  - **4.** Contractor shall designate one "competent person" for Demolition as defined by 29 CFR Part 1926.850.
    - A. For one who is assigned the role of a "competent person," documentation of sufficient and relevant training and experience to perform the assigned duties and responsibilities of that role. As defined in 29 CFR 1926.32(f), the competent person shall be "one who is capable of identifying existing and predictable hazards, and who has authority to take prompt corrective measures to eliminate them."
    - B. Relevant training and experience shall be in the same type of Project activities included in the Work under this contract.
    - C. Training as required for Asbestos and Lead abatement workers

#### 3.02 WORK PLANNING AND MEETINGS

- a. Contractor and the District Representative shall conduct a daily Health and Safety meeting, prior to beginning Work for that day, to address Health and Safety issues, changing site conditions, activities and personnel. All Contractor and Subcontractor employees working on the Site on that day shall attend the meeting. All meetings shall be documented and attendees shall sign acknowledgement of their presence at the meeting. Daily meetings shall include an evaluation of the Work to be conducted, the hazards associated with the work, and control measures being used to reduce exposure.
- b. Contractor personnel who are not in attendance for the daily Health and Safety meeting shall be briefed on the meeting notes upon arrival at the Site and prior to commencing their Work activities. Employees shall sign acknowledgement of briefings prior to commencing Work.
- c. Contractor shall hold and document additional safety meetings at the start of each major task and whenever site conditions affecting personnel safety change. Any major task undertaken shall require the completion, or modification, of a THA as described in this Section.

#### 3.03 ENGINEERING CONTROLS

- **A.** Contractor shall, at a minimum, provide the following Engineering controls to reduce the hazards of equipment operation and exposure during demolition and lead and asbestos abatement activities:
  - 1. Roll-over cages for bulldozers, back hoes, loaders, and tractors
  - 2. Back-up alarms for all trucks and moving equipment
  - 3. Wetting of media or other means to control dust during the Work
  - 4. Decontamination of equipment in accordance with Specifications.
  - 5. Enclosures for abatement activities.
  - 6. Barricades around restricted areas.
  - 7. Others as determined to be necessary or prudent by Contractor or as directed by the District Representative
  - 8. Contractor shall post ground-level warning signs every 50 feet below all overhead utilities onsite.

## 3.04 MONITORING:

- **A.** Contractor shall perform heat exposure and cold exposure monitoring activities as required by weather conditions.
  - 1. Contractor shall perform all atmospheric monitoring of tanks, pits, sumps, vaults, and enclosures to ensure that toxic or explosive gases are not present prior to performing demolition activities or personnel entry. At a minimum, atmospheric monitoring shall include the Lower Explosive Limit (LEL), % Oxygen, and Hydrogen Sulfide gas.
  - Contractor should monitor workers for dust exposure using a personal dust monitor. Workers with the greatest likelihood of being exposed to dust, as evaluated by the SSHO, should don a personal dust monitor.

#### 3.05 EVALUATION OF PERFORMANCE:

a. Contractor shall routinely conduct internal safety audits on Subcontract and Subsubcontract Work sites in accordance with the Contractor's HASP. The focus of these routine audits will be on compliance with OSHA and local occupational safety regulations. b. Contractor shall conduct routine behavioral observations and provide immediate feedback during Work activities to promote safe behavior of Contractor employees and Subcontractor employees.

#### 3.06 SITE SECURITY - OTHER SAFETY CONSIDERATIONS

- **A.** The Site is located in a heavily developed community, with frequent traffic on adjacent streets, and truck traffic delivering to the neighboring retail locations. Site workers, visitors, and truck drivers need to become familiar with the local traffic pattern to prevent traffic accidents and impeding traffic when entering and exiting the site.
- **B.** There is evidence of some transients and/or trespassers attempting to enter the portions of the campus.
- **C.** Site Security measures in the form of fencing, barricades and signage are critically important to maintain a safe work environment and to protect the public. The Contractor shall be responsible for maintaining adequate security measures for the duration of the project.

#### 3.07 WORK BY OTHERS

**A.** District representative or consultant may perform site perimeter dust monitoring activities and monitor for emissions of nuisance dust and/or hazardous materials to areas outside the Work limits.

**END OF SECTION** 

**HEALTH AND SAFETY FORMS FOLLOW** 

# **Hot Work Permit**

**Permit Valid** 

		For 1 W	Vork Day
Site Name:	Project Number:		
EHS Officer:			
Hot Work Description:			
Workers/Welders Conducting Hot W	Vork:		
Permits MUST be co	ompleted in its Entirety Before Hot Work B	egins	
		Vaa	Nia

	Yes	No
Has Project supervisor been notified of intended Hot Work?		
Does client representative need to be notified of the intended Hot Work?		
Will Hot Work impact the general public, clients, or operation employees?		
Will the intended Hot Work need to be coordinated with other contractors who may be working on the site to make them aware of any hazards and the scope of work to be performed?		
Have hazardous energy sources been identified, isolated, and locked out/tagged out before the start of the Project?		
Will Hot Work be conducted within a confined space?		
All testing equipment (i.e., CGI, oxygen meter, etc.) and firefighting equipment (i.e., extinguisher, etc.) have been checked to ensure proper operation and calibration before the start of this Project?		
Has a fire watch been designated and on station?		
Have coatings on metal surfaces been tested for ignitability and flame spread?		
Has the area been cleared of all flammable materials?		
Have all fuel sources been identified and protected?		
Has the area been restricted with proper barriers and signs?		
Has the area been tested to be certain that atmosphere is 0% LEL before starting Hot Work?		
Have flame sensitive areas and equipment (including cylinders and gas delivery lines) exposed to slag and sparks been protected by flame resistant blankets or removed from the area?		
Have all equipment and hoses been protected from falling metal structures and debris?		
Have escape routes been identified before starting work?		
Is ventilation equipment needed? Type needed:		

## The Following Protective Equipment Will be Required:

	Yes	No		Yes	No
Welding Goggles/Shield Tint			Supplied Air Respirator		
Safety Boots			Head Protection		
Leather gloves			Safety Harness		
Hearing Protection			Welding Leathers – Top		
APR Cartridge			Welding Leathers - Bottom		

Permit Valid for 1 Work Day			
The following procedures will be applicable prior to Hot Work on tanks or other types of enclosed structures. (Check all that apply and fill in appropriate information.)			
☐ Ventilate to 0% LEL			
☐ Confined Space Entry Per	mit		
☐ Mechanical Ventilation Re	quired		
☐ Cold Cut Only	Method Allowed:		
☐ Hot Cutting Permitted	Method Allowed:		
Inert to <% Oxygen			
Approvals:			
Date			
District Representative			
Contractor's Site Safety Officer			
Fire Watch			
Performed Hot Work Employee			
File Permit in Project Work File and He	ealth and Safety Department		

#### **SECTION 01 45 23 - TESTING AND INSPECTION**

#### PART 1 GENERAL

## 1.01 SECTION INCLUDES

This Section includes CONTRACTOR's responsibilities with regard to mandatory testing and inspection services:

- A. Testing and inspection services to meet requirements of the California Code of Regulations (CCR), Title 24, California Building Code (CBC).
- B. Tests of materials required by the DISTRICT's DSA certified testing agency as set forth in Section 4-335 of the California Building Standards Administrative Code.
- C. Inspection by DSA certified inspectors, employed by the DISTRICT in accordance with the requirements of California Building Standards Administrative Code, assigned to the Work with duties specifically defined in Section 4-333(b).

## 1.02 TESTING AGENCY

- A. DISTRICT will select an independent testing agency approved by the California Division of the State Architect to conduct tests, sampling, and testing of materials.
- B. Selection of material to be tested shall be by the agency or the INSPECTOR OF RECORD (IOR) and not by CONTRACTOR.
- C. Any material shipped from the source of supply prior to having satisfactorily passed such testing and inspection or prior to the receipt of notice from IOR that such testing and inspection is not required shall not be incorporated into the Work.
- D. DISTRICT will select and directly reimburse testing agency the costs for all DSA and/or DSA required tests and inspections, but may be reimbursed by CONTRACTOR for such costs as noted in related portions of the Contract Documents.
- E. The independent testing agency is not authorized to release, revoke, alter, or enlarge requirements of the Contract Documents or approve or accept any portion of the Work. The testing agency shall not perform any duties of CONTRACTOR. The agency does not have authority to stop the Work.

## 1.02 TEST REPORTING

- A. Test reports shall include all tests performed, regardless of whether such tests indicate the material is satisfactory or unsatisfactory. Samples taken but not tested shall also be reported. Records of special sampling operations as required shall also be reported. Reports shall indicate the material or materials were sampled and tested in accordance with requirements of CBC, Title 24, Parts 1 and 2. Test reports shall indicate specified design strength. They shall also definitely state whether or not material or materials tested comply with the specified requirements. When requested by DISTRICT or Architect, provide interpretation of test results.
- B. After each inspection and test, testing agency will promptly submit one (1) copy of laboratory report to the following distribution list:

- 1. Division of State Architect
- 2. District. (or District Representative if applicable)
- 3. Project Inspector.
- 4. Architect.
- 5. Structural Engineer.
- 6. Mechanical and Electrical Engineers (Related Tests and Inspections).
- 7. Contractor

## C. Each test report will include:

- Date issued.
- 2. Project title, Architect's number, DSA application and file number.
- 3. Name of agency's inspector.
- 4. Date and time of sampling or inspection.
- 5. Identification of product and Specifications Section.
- 6. Location in the Project.
- 7. Type of inspection or test.
- 8. Date of test and ambient conditions at time of test.
- 9. Results of tests.
- Statement of Conformance with Contract Documents.
- 11. Signature by Registered Professional Engineer licensed in California.
- 12. Statement that tests were conducted in accordance with Parts 1 and 2, Title 24, California Code of Regulations.
- D. Immediately upon testing agency determination of a test failure, the agency will telephone the results of the test to the ARCHITECT. On the same day, the agency will send written test results to those on the distribution list.

#### 1.03 TEST AND INSPECTION VERIFICATION REPORT

A. Testing agency shall submit to the Division of the State Architect a verified report in duplicate, with copy to the DISTRICT, covering each test which is required to be performed by that agency during progress of the Work. Such report shall be furnished each time construction on the Work is suspended, covering tests up to that time, and also prior to Final Completion of the Work, covering all tests.

#### 1.04 INSPECTION BY DISTRICT

- A. DISTRICT and its representatives shall at all times have access, for purpose of inspection, to all parts of the Work and to shops wherein the Work is in preparation, and CONTRACTOR shall at all times maintain proper facilities and provide safe access for such inspection.
- B. DISTRICT shall have the right to reject materials and/or workmanship deemed defective Work, and to require correction. Defective workmanship shall be corrected in a satisfactory manner and defective materials shall be removed from the premises and legally disposed of, all without charge to DISTRICT. If CONTRACTOR does not correct such defective Work within a reasonable time, fixed by written notice and in accordance with the terms and conditions of the Contract Documents, DISTRICT may correct such defective Work and proceed to recover costs in accordance with related Articles of the Contract Documents.

#### 1.05 INSPECTOR OF RECORD

- A. INSPECTOR OF RECORD (IOR) is employed by DISTRICT in accordance with requirements of Title 24 of the California Code of Regulations with their duties specifically defined therein.
- B. Inspection of Work shall not relieve CONTRACTOR from any obligation to fulfill all of the terms and conditions of the Contract Documents.
- C. CONTRACTOR shall be responsible for scheduling times of inspection, tests, sample taking, and similar activities of the Work.

## 1.06 CONTRACTOR RESPONSIBILITIES

- A. Cooperate with testing agency personnel, DISTRICT's Representative, INSPECTOR OF RECORD (IOR), CONSTRUCTION MANAGER and the ARCHITECT, to provide access to the Work including weekends and after work hours and to manufacturer's facilities.
- B. Provide incidental labor, materials and facilities to provide, at all times, safe access to Work to be tested, to obtain and handle samples at the site or at source of products to be tested, to facilitate tests and inspections, storage and curing of test samples.
- C. Notify CONSTRUCTION MANAGER, IOR and testing agency 24 hours in advance of required inspections or sampling, and 48 hours in advance of special testing or inspections. Notify DISTRICT in advance of the manufacturer or fabrication of materials in time to plan and schedule required testing at the source of supply. Extra expenses resulting from a failure to notify the agency shall be borne by the CONTRACTOR. Whenever extra expenses are indicated to be borne by the CONTRACTOR, they will be charged to the CONTRACTOR by Change Order.
- D. The DISTRICT, IOR, CONSTRUCTION MANAGER or the ARCHITECT shall have the right to reject materials and workmanship which are defective or to require their correction. Rejected workmanship shall be satisfactorily corrected and rejected materials shall be removed from the premises without cost to the DISTRICT. Extra expenses for retesting and re-inspection shall be borne by the CONTRACTOR. If the CONTRACTOR fails to correct such rejected work within a reasonable time, fixed by written notice, the DISTRICT will correct same and charge the expense to the CONTRACTOR by Change Order.
- E. Should it be considered necessary or advisable by the DISTRICT at any time before date of substantial completion of the entire work to make an examination of work already completed by removing or tearing out the same, the CONTRACTOR shall on request promptly furnish all necessary facilities, labor and materials. If such work is found to be defective in any respect due to fault of the CONTRACTOR or his subcontractor, all extra expenses shall be borne by the CONTRACTOR. If, however, such work is found to meet the requirements of the Contract, the additional cost of labor and material necessarily involved in the examination and replacement costs shall be allowed the CONTRACTOR by Change Order.
- F. When changes of construction progress schedule are necessary during construction, coordinate such changes with the testing agency as required.

- G. When the testing agency is ready to test according to the established schedule, but is prevented from testing or taking specimens due to incompleteness of the Work, extra charges for testing attributable to the delay shall be borne by the CONTRACTOR.
- H. CONTRACTOR is responsible for compliance to all applicable local, state, and federal regulations regarding codes, regulations, ordinances, restrictions, and requirements, regardless of the provisions of this Section.
- I. Inspecting and testing performed exclusively for the CONTRACTOR's convenience shall be the sole responsibility and expense of the CONTRACTOR.

## 1.07. TESTS AND INSPECTIONS

- A. The following tests and inspections do not limit inspection of the Work but are required by DSA, other agencies, or are required in related Sections of the Contract Documents. The list may not be all inclusive.
- B. Excavations, Foundations and Retaining Walls CBC, Chapter 18A and 33A
  - 1. Inspection:

a.	Earth Fill Compaction	1801A.2
b.	Inspection of Driven Pile Installation	1809A.6
C.	Inspection of Caissons	1809A.7

- C. Concrete CBC, Chapter 19A:
  - Materials:

a.	Test of Materials	1903A.1
b.	Portland Cement Tests	1903A.2
C.	Concrete Aggregate	1903A.3
d.	Shotcrete Aggregate	1903A.3; 1924A.3
e.	Reinforcing Bars	1903A.5.1; 1903A.5.2;
	-	1903A.5.3; 1903A.5.4;
f.	Prestressing Steel & Anchorage	1903A.5.5;
g.	Structural Steel, Steel Pipe or tubing	1903A.5.6
h.	Admixtures	1903A.6
d.	Batch Plant Inspection	1929A.4
e.	Waiver of Batch Plant	1929A.5, 1929A.6
	Inspection & Tests	

## 2. Quality:

a.	Proportions of Concrete	1905A.1; 1905A.2; 1905A.3;
	•	1905A.4; 1905A.5; 1905A.6,
b.	Mixing and Placing	1905A.1.1; 1905A.1.2;
		1905A.1.3
C.	Concrete Testing	1905A.6;
d.	Test Of Shotcrete	1905A.6; 1924A.10
e.	Composite Construction Cores	1929A.8
f.	Gypsum Concrete Strength Tests	1925A.1; 1929A.13
a.	Insulating Concrete Tests	DSA IR 27-1

#### 3. Inspection:

Project Site Inspection 1905A.7.1 a.

Batch Plant or Weighb.

master Inspection: 1929A.4, 1929A.5; 1929A.6

Pre-stressed Concrete Inspection 1929A.9 C. d. Shotcrete Inspection 1929A.10

Reinforcing Bar Welding Inspection 1929A.12, 1903A.10 e.

#### D. Lightweight Metal - CBC, Chapter 20A:

1. Materials:

> 2001A.2 a. Alloys b. Identification 2001A.4

2. Inspection:

> Welding 2004A.8

#### E. Masonry - CBC, Chapter 21A:

1. Materials:

> Masonry Units 2102A.2,.4,.5,.6 a. Portland Cement, Lime 2102A.2.2, .3; 2103A.2 b. Mortar & Grout Aggregates 2102A.2.1; 2103A.4.3 C. Reinforcing Bars 2102A.2.10; 1903A5, d.

> > 2102A.2.10

#### 2. Quality:

Portland Cement Tests a. 1903A.2, 1929(A.1) Mortar & Grout Tests 2105A.3.4.2 b.

Masonry Prism Tests 2105A.3.2, 2105A3.5 C.

Masonry Core Tests 2105A 3.1 d. Reinforcing Bars 2102A.2.10 e.

3. Inspection:

> a. Reinforced Masonry 2105A

Reinforcing Bar Welding 1903A.10, 1929A.12 b. Inspection

#### F. Steel - CBC, Chapters 17A & 22A:

1. Materials:

> Structural Steel. 2202A.1, 2231A.1 a.

> > Cold Formed Steel

Material Identification 2203.A4 b.

#### 2. Inspection and Tests:

a. Test of Structural Steel 2231.A

Tests of High Strength Bolts, 2231.A.2 Nuts, and Washers b.

Tests of End Welded Studs 2231.A.3 C. Shop Fabrication Inspection d. 2231.A.4

Welding Inspection 2231.A.5 e.

f.	Non-destructive Weld Testing	1703	3A
g.	High Strength Bolt Inspection	2231	1A.6
ĥ.	Steel Joist Load Tests	2231	1A.7
i	Spray applied fire resistance mate	rials	1701

- G. Wood CBC, Chapter 23A:
  - 1. Materials:

a.	Lumber and Plywood Grading	2303A.1, 2304A
b.	Glue-Laminated Members	2303A.2, 2304A

2. Inspection:

a.	Glue-Laminated Fabrication	2337A.1
b.	Timber Connectors	2337A.2
C.	Manufactured Trusses	2337A.3

- H. Exterior Wall Coverings CBC, Chapter 14A, 25A:
  - 1. Materials:

a. Portland Cement Plaster 2508A, 2509A, 2510A

2. Inspection:

a. Veneer Inspection 1405A

- I. Clay or Concrete Roof Tile CBC Chapter 15A:
  - Materials:

a. Clay or concrete tile 1507.7

2. Inspection: District Requirement

## 1.08 EARTHWORK

- A. The DISTRICT's testing agency, under the direction of the Geotechnical Engineer of Record, will provide continuous inspection of fill and will field test fill and earth backfill as placed and compacted, and inspect excavations and subgrade before concrete is placed and provide periodic inspection of open excavations, embankments, and other cuts or vertical surfaces of earth. Geotechnical Engineer will sign all reports of observation and testing.
- B. Unsatisfactory materials shall be removed from the site. Materials installed improperly shall be removed, replaced, moisture adjusted, re-compacted and otherwise re-worked to achieve a satisfactory installation.
- C. Imported fill materials from offsite or onsite shall be inspected and tested at the source before importing and placing, and a report issue attesting to the satisfactory nature of the material.
- D. The agency will perform all sampling and testing of materials and testing of work in place as required by the DSA Testing and Inspection Listing, or otherwise required. Testing will be performed in accordance with ASTM or Californiarequired test methods.

#### 1.09 CONCRETE

- A. The DISTRICT's testing agency will conduct one-time sampling of aggregate and preparation and testing of concrete mix design for each strength and/or aggregate size specified. Testing costs for additional mix designs shall be borne by the CONTRACTOR.
- B. Continuous plant inspection and other concrete installation tests will be conducted by the DISTRICT's testing agency. However, costs for retesting of materials that do not meet specification requirements shall be borne by the CONTRACTOR.

## 1.10 ROOFING

- A. Testing agency will conduct inspection and testing of built-up bituminous roofing in accordance with manufacturer's instructions, including:
  - 1. Attend pre-roofing conference.
  - 2. Check deck surfaces prior to roofing application to verify that substrate is in satisfactory condition to receive roofing.
  - 3. Check kettle temperature control system and monitor kettle control temperatures.
  - 4. Inspect and test materials including softening point of asphalt to ensure conformance with specifications.
  - Check for excessive moisture.
  - Observe roofing application to ensure conformance with specifications.
  - Supervise cutting and repair of cut-out tests and test and inspect cut-out samples for conformance with specifications.

## **END OF SECTION**

#### **SECTION 01 52 00 - CONSTRUCTION FACILITIES**

#### PART 1 GENERAL

## 1.01 SECTION INCLUDES

A. Furnishing and installing temporary facilities as indicated, specified or required for proper performance of the Work.

#### 1.02 RELATED SECTIONS

- A. Temporary. Storm Water Pollution Control
- B. Temporary Controls
- C. Construction Waste Management and Disposal

#### 1.03 GENERAL

- A. CONTRACTOR shall provide, maintain, relocate, and remove temporary facilities, including buildings, field office, toilets, utilities, storage units, fencing, barricades, chutes, elevators, hoists, scaffolds, railings and other facilities or services as required. CONTRACTOR shall be responsible for all use charges for the items provided as specified herein.
- B. CONTRACTOR shall furnish, install, maintain and pay for all necessary permits, inspections, temporary lines and connections and metering devices, use charges, move-ins/outs, connection fees, service, extension and distribution, deliveries/pickups, rentals, storage, transportation, taxes, labor, insurance, bonds, materials, equipment and all other required miscellaneous items for the temporary utilities systems required for completion of the work, and, upon substantial completion of the Work, remove all such temporary utilities systems and appurtenances.

#### 1.04 REGULATORY REQUIREMENTS

- A. Comply with governing ordinances, regulations and utility company requirements and recommendations.
- B. Comply with pollution and environmental protection codes and regulations for use of water and energy, for discharge of waste and storm drainage from the project site, and for control of dust, air pollution and noise.
- C. Temporary construction shall conform to requirements of State, County and local authorities and insurance requirements which pertain to operation, health, safety and fire hazard. Provide items necessary to comply with such requirements, whether or not specifically indicated or specified in the Contract Documents.

#### 1.05 TEMPORARY WATER

A. CONTRACTOR shall provide and maintain temporary potable water service, including water distribution piping and outlet devices of the size and required flow rates in order to provide service to all areas of the Project site at all times.

#### 1.06 TEMPORARY SANITARY FACILITIES

- A. CONTRACTOR shall provide portable chemical toilet facilities, in quantities based on total number of workers and shall be in accordance with CAL/OSHA standards.
- B. Portable chemical toilet facilities shall be maintained with adequate supplies and in a clean and sanitary condition and shall be removed from the Project site upon Substantial Completion of the Work.
- C. CONTRACTOR employees shall not use school toilet facilities.
- D. CONTRACTOR will define appropriate areas for break and lunch periods and will provide suitable containers for placement of trash in those areas. Areas shall be maintained clean and orderly.

#### 1.07 TEMPORARY TELEPHONE SERVICE

A. CONTRACTOR shall provide temporary telephone and data service for temporary facilities.

## 1.08 TEMPORARY ELECTRICAL POWER

- A. CONTRACTOR shall provide temporary electrical service for construction, temporary facilities, and connections for construction equipment requiring power or lighting, at all points required for the Work, for inspection and safety.
- B. CONTRACTOR shall ensure that welding equipment is supplied by electrical generators, not by the utility-furnished electrical power.

## 1.09 TEMPORARY LIGHTING

- A. CONTRACTOR shall provide and maintain all temporary lighting as necessary to provide safe access, performance and inspection of the work.
- B. Light levels provided shall be a minimum of 20 foot candles inside buildings and 5 foot candles outside for inspection, safety and security.

## 1.10 TEMPORARY HEATING, VENTILATION AND AIR CONDITIONING (HVAC)

- A. CONTRACTOR shall provide temporary heating, ventilating, cooling and filtration required for satisfactory completion of the Work.
- B. CONTRACTOR shall ventilate enclosed areas to assist cure of materials, dissipate humidity, and to prevent accumulation of dust, fumes, vapors, gases, or other irritants.
- C. CONTRACTOR shall maintain manufacturer-required levels of room and/or space temperature, humidity and ventilation necessary to install products, materials and/or systems of the Work.

D. Utilization of the HVAC system for temporary construction use does not constitute DISTRICT acceptance of the system.

#### 1.11 TEMPORARY GAS

A. CONTRACTOR shall provide temporary gas service for construction and temporary facilities, at all points required for the Work.

## 1.12 CONSTRUCTION EQUIPMENT AND FACILITIES

- A. CONTRACTOR shall erect, equip, and maintain construction equipment in strict accordance with applicable statues, laws, ordinances and regulations of authority having jurisdiction.
- B. CONTRACTOR shall provide, maintain and remove upon completion of the Work all temporary rigging, scaffolding, hoisting equipment, rubbish chutes, ramps, stairs, runways, platforms, ladders, railings and other temporary construction as required for all work hereunder.

## 1.13 FIELD OFFICES

- A. CONTRACTOR shall provide a temporary field office for his own use. It shall be weather-tight with lighting, electrical outlets, electronic communications capabilities, HVAC, and otherwise equipped to adequately conduct construction operations. Provide a conference room adequate for project meetings.
- B. In addition to the CONTRACTOR's field office, CONTRACTOR shall provide and maintain a minimum of one similarly equipped 800 sf temporary field office building on the Project site for use by the DISTRICT for the duration of the Work. The office shall be accessible by the DISTRICT, ARCHITECT and/or the IOR on a 7 day a week 24-hour basis. Office shall be provided with code-required ADA accessibility:
  - 1. Office building shall include a conference room with a conference table and adequate seating for twelve.
  - 2. Office building shall have two separate private offices; together with an open office space.
  - 3. Office shall be furnished with two (2) exterior entrance doors with one located in a separate office. Each door shall be furnished with both a dead bolt and cylinder lock with 6 keys.
  - Exterior doors and windows shall be provided with exterior mounted burglar bars. Windows shall be provided with operable window shades. Security of office and contents is a continuous obligation of CONTRACTOR.
  - 5. Office shall have ample headroom and shall be properly lighted, heated, ventilated, and air-conditioned, and shall have an electric drinking fountain or potable refrigerated bottled water service.
  - 6. The conference room shall be approximately 300 sq. ft. in size and shall

be furnished with a minimum of four single phase convenience outlets. It shall be furnished with a conference table capable of seating twelve, and twelve comfortable conference chairs, and shall have a 4' x 8' whiteboard on one of the long walls.

- 7. Provide phone, data transmission lines, related appurtenances, services, and equipment for use by DISTRICT as specified below:
  - a. Provide, install, & maintain any related equipment necessary to provide continuous internet access from each location.
- C. CONTRACTOR shall be responsible for maintaining all electrical distribution lines, equipment and related devices. If equipment and/or transmission equipment becomes inoperable and downtime exceeds two (2) days, CONTRACTOR shall replace and/or provide equivalent interim equipment.
- D. Office, furniture, equipment, and related ancillary devices shall remain property of CONTRACTOR. CONTRACTOR shall remove such property upon Final Completion of Work or as otherwise determined by the DISTRICT.
- E. At CONTRACTOR expense and without limitation, remove and/or relocate temporary office(s) and related facilities as rapidly as required in order to provide for progress of the Work.

## F. FIELD OFFICE SUPPLIES

1. CONTRACTOR shall provide the initial supply of field office supplies as need for IOR.

#### 1.14 STORAGE AND STAGING

- A. Operations of the CONTRACTOR, including storage of materials, shall be confined to areas approved by DISTRICT. CONTRACTOR shall be liable for damage caused by him during such use of property of the DISTRICT or other parties.
- B. Storage facilities shall provide protection of products from excessive cold, heat, moisture, humidity or physical abuse as specified in the respective sections for the products stored.
- C. CONTRACTOR shall save the DISTRICT, along with its respective officers, employees and agents, and the ARCHITECT and his employees, free and harmless from liability of any nature or kind arising from any use, trespass or damage occasioned by his operations on assigned premises of third parties.

#### 1.15 FENCES AND BARRICADES

A. CONTRACTOR shall install temporary Project site security fence(s) and/or barricade(s), as specified herein or indicated on Drawings, or as required for safety and security. New or used material may be furnished. Security of Project site and contents is a continuous obligation of CONTRACTOR.

- B. Unless otherwise indicated or specified, a site security fence shall be constructed of 8'-0" high chain link fencing with an 8'-0" high windscreen. Space posts not to exceed 10'-0" on centers. Posts shall be of following nominal pipe dimensions: terminal, corner, and gatepost 2-1/2", line posts 2". Chain link fence shall be not less than #13 gage, 2" mesh, and in one width. Posts, fence and accessories shall be galvanized and as follows:
  - 1. Posts shall be set in the earth a depth of 30" with soil firmly compacted around post, unless required otherwise in writing by DISTRICT.
  - 2. Fence fabric shall be attached to posts with #14 gage tie wire at 16" on centers. A #6 gage steel tension wire with turnbuckles shall be installed at top and bottom of barricade fencing. Wire tie fabric to tension wires at 18" centers.
  - 3. Windscreen shall be attached to fence fabric and steel tension wires at 18" centers with a minimum of #14 gage tie wire. Windscreen shall be maintained and all rips, tears, missing sections shall be corrected as soon as detected.
  - 4. Chain link fencing shall be free from barbs, icicles or other projections resulting from galvanizing process. Fence having such defects will be replaced even if it has been installed.
  - 5. Gates shall be fabricated of steel pipe with welded corners, and bracing as required. Fence and fabric to be attached to frame at 12" centers. Provide all gate hardware of a strength and quality to perform satisfactorily until barricade is removed upon Substantial Completion of the Work. Each gate shall have a chain and padlock. Provide two (2) gate keys to DISTRICT. At Substantial Completion of the Work, remove barricade from Project site, backfill and compact fence footing holes. Existing surface paving that is cut into or removed shall be patched and sealed to match surrounding areas.
  - 6. When directed by DISTRICT, CONTRACTOR shall at CONTRACTOR expense and without limitation, remove and/or relocate fencing, fabric and barricades or other security and protection facilities as rapidly as required in order to provide for progress of the Work. (Note: DISTRICT retains option to leave fencing, barricades and SWPPP protections in place for an unspecified period of time after completion of demolition and final grading operations, to be compensated on a monthly rate basis.)

## C. Other Temporary Enclosures & Barricades

- 1. Provide fences and barricades to prevent unauthorized entry to construction areas and to protect existing facilities and adjacent properties from damage from construction operations.
- 2. Provide lockable, temporary weather-tight enclosures at openings in exterior walls to create acceptable working conditions, to allow for temporary heating and for security.
- Provide protective barriers around trees, plants and other improvements designated to remain. Replace any damaged materials as directed by the ARCHITECT

- 4. Temporary partitions shall be installed at all openings where additions connect to existing buildings, and where necessary to protect areas, spaces, property, personnel, students and faculty and to separate and control dust, debris, noise, access, sight, fire areas, safety and security. Temporary partitions shall be as designated on the Drawings or as specified by ARCHITECT. At CONTRACTOR expense and without limitation, remove and/or relocate enclosures, barriers and temporary partitions as rapidly as required in order to provide for progress of the Work.
- 5. Since the Work of this Project may be immediately adjacent to existing occupied structures and vehicular and pedestrian right of ways, CONTRACTOR shall, in his sole judgment and in accordance with applicable safety standards, provide all temporary facilities, additional barricades, protection and care to protect existing structures, occupants, property, pedestrians and vehicular traffic. CONTRACTOR is responsible for any damage, which may occur to the property and occupants of the property of DISTRICT or adjacent private or public properties which in any way results from the acts or neglect of CONTRACTOR.
- Fences and barricades must completely separate construction activities and personnel from school operations, staff, students and the public.
   Construction workers shall not interact or communicate with students or staff except in emergency or safety related situations.
- 7. Provide barricades and covered walkways required by governing authorities for public rights-of-way and for public access to existing buildings.
- 8. Protect vehicles, stored materials, site and structures from damage.

#### 1.16 TEMPORARY DE-WATERING FACILITIES & DRAINAGE:

- A. For temporary drainage and de-watering facilities and operations not directly associated with construction activities included under individual sections, comply with de-watering requirements of applicable Division 01 sections or of sound practice. CONTRACTOR shall maintain the Work, Project site and related areas free of water.
- B. For temporary drainage and de-watering facilities and operations directly associated with new buildings, additions or other construction activities, comply with Division 01 & 02 Sections. CONTRACTOR shall be responsible for dewatering of excavations, trenches & below grade areas of buildings, structures, the Project site and related areas.

#### 1.17 TEMPORARY PROTECTION FACILITIES:

A. CONTRACTOR shall not change over from using temporary facilities and controls to permanent facilities until Substantial Completion, except as permitted by DISTRICT.

- B. CONTRACTOR shall provide fire protection during construction in accordance with CFC, Article 87
- C. Until permanent fire protection needs are supplied and approved by authorities having jurisdiction, CONTRACTOR shall provide, install and maintain temporary fire protection facilities of the types needed in order to adequately protect against fire loss. CONTRACTOR shall adequately supervise welding operations, combustion type temporary heating and similar sources of fire ignition.
- D. CONTRACTOR shall provide, install and maintain substantial temporary enclosures of partially completed areas of construction. Provide locking entrances to prevent unauthorized entrance, vandalism, theft and similar violations of security. Where materials, tools and equipment are stored within the Work area, CONTRACTOR shall provide secure lock up to protect against vandalism, theft and similar violations of security. DISTRICT accepts no financial responsibility for loss, damage, vandalism or theft.
- E. CONTRACTOR operations shall not block, hinder, impede or otherwise inhibit the use of required exits and/or emergency exits to the public way, except as approved by the DISTRICT. CONTRACTOR shall maintain unobstructed access to fire extinguishers, fire hydrants, temporary fire protection facilities, stairways and other access routes for firefighting equipment and/or personnel.
- F. With approval of DISTRICT and at the earliest feasible date in each area of the Work, complete installation of the permanent fire protection facilities including connected services and place into operation and use. Instruct DISTRICT personnel in use of permanent fire protection facilities.
- G. In the event of an emergency drill or an actual emergency, designated by the sounding of the fire alarm and/or other sounding device, all construction activities must cease. CONTRACTOR shall evacuate the Work area and remain outside the Work area until permitted to return. No Work shall be conducted during the evacuation of a building or during an emergency.

## 1.18 TEMPORARY SECURITY AND SAFETY MEASURES:

- A. During performance of the Work in existing facilities CONTRACTOR shall provide, install and maintain substantial temporary barriers and/or partitions separating all Work areas from areas occupied by students, faculty and/or administrative staff.
- B. During performance of the Work in existing facilities and/or on a Project site occupied by students and where temporary barriers and/or partitions are not physically feasible, CONTRACTOR shall provide an employee meeting the requirements of Education Code Section 45125.2. (2) to continually supervise and monitor all employees of the CONTRACTOR and Subcontractor. For the purposes of this Section, CONTRACTOR employee shall be someone whom the Department of Justice has ascertained has not been convicted of a violent or serious felony as listed in Penal Code Section 667.5(c) and/or Penal Code Section 1192.7(c). To comply with this Section, CONTRACTOR shall have his employee submit his or her fingerprints to the Department of Justice pursuant to Education Code Section 45125.1(a).

C. Penal Code Sections 290 and 290.4, commonly known as "Megan's Law," require, among other things, individuals convicted of sexually oriented crimes, to register with the chief of police where the convicted individual resides or with a county sheriff or other law enforcement officials. The CONTRACTOR shall check its own employees and require each Subcontractor to check its employees and report to the CONTRACTOR if any such employees are registered sex offenders. The CONTRACTOR shall check monthly during the life of the Contract to ascertain this information and report same to DISTRICT. Before starting the Work, and monthly thereafter during the life of Contract, CONTRACTOR shall notify the DISTRICT in writing if any of its employees and/or if any Subcontractor's employees is a registered sex offender. If so, CONTRACTOR shall proceed in accordance with the previous paragraph.

#### 1.19 TEMPORARY ACCESS ROADS AND PARKING:

- A. Due to the limited amount of on and off Project site space for the parking of staff, students and school visitors' vehicles, there will be no parking of CONTRACTOR vehicles in areas designated for school use only. CONTRACTOR shall provide legal access to and maintain CONTRACTOR designated areas for the legal parking, loading, off-loading & delivery of all vehicles associated with the Work. CONTRACTOR shall be solely responsible for providing and maintaining these requirements whether on or off the Project site.
- B. Contractor's onsite parking shall be in areas shown on the Logistics Site Plan or as otherwise designated by the DISTRICT.
- C. Temporary access roads are to be installed and maintained by CONTRACTOR to all areas of the Project site.
- D. CONTRACTOR will be permitted to utilize existing on-site roads as designated by DISTRICT. CONTRACTOR shall only utilize those entrances and exits as designated by DISTRICT, and CONTRACTOR shall observe all traffic regulations of DISTRICT.
- E. Provide and maintain access to fire hydrants, free of obstructions.
- F. Do not park or drive on concrete walks or in the new buildings at any time.
- G. CONTRACTOR shall maintain roads and walkways in a clean condition including removal of debris and/or other deleterious material on a daily basis.

## 1.20 TRENCHES

- A. CONTRACTOR shall comply with all applicable statutes, codes & regulations regarding trenching and trenching operations. Open trenches for installation of utility lines (water, gas, electrical and similar utilities) and open pits outside barricaded working areas shall be barricaded at all times in a legal manner determined by CONTRACTOR.
- B. Open trenches deeper than 3'-6", and not located within a public street access, shall be enclosed within an 8'-0" high chain-link fence.

- C. Trenches shall be backfilled and patch-paved within twenty-four (24) hours after approval of installation by authorities having jurisdiction or shall have "trench plates" installed.
- D. Required access to buildings shall be provided and maintained.

#### 1.21 PROJECT SIGNAGE

- A. CONTRACTOR shall furnish and install a Project sign on the Project site at a location established by ARCHITECT. A graphical layout of the proposed sign shall be submitted to ARCHITECT and DISTRICT for review before fabrication.
- B. Sign construction shall be 10'-0" wide by 6'-0" high with 6" x 6" posts and 1" exterior grade plywood, bolted to posts.
- C. Sign lettering shall be painted white with exhibit lettering by a professional sign painter, in accordance with details reviewed by ARCHITECT. The following shall be listed on sign:
  - 1. DISTRICT San Bernardino City Unified School District.
  - Name of School.
  - 3. Names of the Architect/Engineer and Consultants.
  - 4. Name of Prime Contractor.
  - 5. Other principal Contractors.
  - 6. Name of School Board member from District in which project is located.
- D. Except as otherwise specified herein, no other signs shall be displayed without approval of DISTRICT. At CONTRACTOR expense and without limitation remove and/or relocate Project signage and related facilities as rapidly as required in order to provide for progress of the Work.
- E. CONTRACTOR shall remove any Project signage at Substantial Completion of the Work.
- F. CONTRACTOR shall provide and install signage to provide directional, identification, and contact information to construction personnel and visitors as follows and as approved by DISTRICT.
  - 1. For construction traffic control/flow at entrances/exits, and as designated by DISTRICT.
  - 2. To direct visitors.
  - 3. For construction parking.
  - 4. To direct deliveries.
  - 5. For Warning Signs as required.
  - 6. Per CAL/OSHA standards as necessary.
  - 7. For office identification and Project site address.
  - 8. For "No Smoking" safe work site at designated locations.
  - 9. Emergency contact information and phone number of CONTRACTOR.
  - 10. Emergency contact information and phone number of local police, fire,

and emergency personnel.

# 1.22. CLOSE OUT

A. Remove all temporary facilities at the completion of construction, and restore the site and facilities to conditions acceptable to the DISTRICT, ARCHITECT and to local authorities.

#### **SECTION 01 57 00 - TEMPORARY CONTROLS**

#### PART 1 GENERAL

#### 1.01 SECTION INCLUDES

- Water Control.
- B. Dust Control.
- C. Erosion and Sediment Control.
- D. Noise Control.
- E. Pollution Control.

#### 1.02 RELATED SECTIONS

- A. Work Sequence and Phasing.
- B. Temporary Storm Water Pollution Control
- C. Construction Facilities

#### 1.04 GENERAL

- A. Include planned temporary control measures in the Project Logistics Plan of Section 01 32 16, Work Sequence and Phasing. Include hours of operation permitted by the Contract Documents or by local authorities.
- B. Update this Plan and provide status reports to the DISTRICT on temporary controls on a monthly basis

## 1.04 WATER CONTROL

- A. Do not permit surface or subsurface water or other liquids to accumulate on the site or in the immediate vicinity.
- B. Should such conditions be encountered or develop, control the accumulation of water or other liquid and suitably dispose of it by means of temporary pumps, piping, drainage lines, troughs, ditches, dams or other methods as approved by the ARCHITECT and/or the authority having jurisdiction.

#### 1.05 DUST CONTROL

- A. Conduct earthwork operations in a manner to prevent windblown dust and dirt from interfering with the progress of the Work, the District's activities, the existing occupied structures in the areas of the site immediately adjacent, and offsite adjacent properties.
- B. Water construction areas as necessary to minimize windblown dust and on-site accumulation of dust and dirt.
- C. Water spray or cover with tarpaulins truckloads of soil to minimize generation of dust and dirt from construction transportation operations.
- D. Prevent dust and dirt from accumulating on walks, roadways, parking areas and from washing into sewer and storm drain lines.

#### 1.06 EROSION AND SEDIMENT CONTROL

- A. Plan and execute construction by methods that will control surface drainage from cuts and fills and from borrow and waste disposal areas, and to prevent erosion and sedimentation.
- B. Minimize amount of bare soil exposed at one time.
- C. Provide temporary measures such as berms, dikes and drains to prevent water flow over adjacent properties or City rights-of-way.
- D. Construct fill and waste areas by selective placement to avoid erosive surface silts or clays. Avoid any eroded materials flowing off the property.
- E. Periodically inspect earthwork to detect evidence of erosion and sedimentation; and promptly apply corrective measures.

## 1.07 NOISE CONTROL

- A. Avoid excessive noise that would affect detrimentally adjacent activities and adjoining property.
- B. Confine operations to permissible hours of day, to eliminate neighborhood noise pollution.

#### 1.08 POLLUTION CONTROL

- A. Provide methods, means and facilities to prevent contamination of soil, water and atmosphere from discharge of noxious, toxic substances and pollutants produced by construction operations.
- B. Do not burn refuse, debris or other materials on the site.
- C. Comply with all State and local ordinances and regulatory requirements controlling environmental pollution during the course of construction and disposal operations.

## 1.09 PROGRESS CLEANING

- A. Maintain areas free of waste materials, debris and rubbish. Maintain site in a clean and orderly condition.
- B. CONTRACTOR shall assure the removal of debris and rubbish from pipe chases, plenums, attics, crawl spaces and other closed or remote spaces prior to the space being enclosed.
- C. CONTRACTOR shall assure the brooming and vacuum cleaning of interior areas prior to start of surface finishing, as well as continuing cleaning to eliminate dust.
- D. Until Substantial Completion of the Work, CONTRACTOR shall remove, as required, all graffiti from buildings, equipment, fences and other improvements on the Project site.

E. CONTRACTOR shall remove waste materials, debris and rubbish from site periodically and dispose off-site.

# 1.10 CLOSE OUT

A. Remove all temporary control measures at the completion of construction, and restore the site and facilities to conditions acceptable to the ARCHITECT and local authorities.

## SECTION 01 57 13 - TEMPORARY EROSION AND SEDIMENT CONTROL

#### **PART 1 - GENERAL**

## 1.01 SUMMARY

- A. The District will be filing with the State of California, State Water Resources Control Board a Notice of Intent (N.O.I.) to comply with the terms of the General Permit to Discharge Storm Water Associated with Construction Activity, prior to the beginning of construction on this site.
- B. A copy of the SWPPP will be made available to Contractors during the bidding period. The Contractor will need to implement and monitor the storm water pollution prevention plan prepared for this site. The Contractor will be required to review the storm water pollution prevention plan and to identify possible pollution sources and mitigation measures with all subcontractors at their starting of work on site.
- C. The Contractor will be obligated to comply with the requirements of the State's General Permit. Any fines or penalties due to failure to comply with the general permit shall be borne by the Contractor.
- D. Prior to construction and after commencement of construction activities, revisions to the SWPPP shall be submitted, by the Contractor, to the Architect for amendment to the general permit by the Civil Engineer.
- E. Storm water pollution prevention plan testing and reporting will be performed by the Contractor until such responsibility is reassigned by the District.

#### 1.02 REFERENCE STANDARDS

A. EPA (NPDES) - National Pollutant Discharge Elimination System (NPDES), Construction General Permit; Current Edition.

#### 1.03 QUALITY ASSURANCE

- A. Codes and Standards
  - 1. California Codes and Regulations; Title 24, California Building Code, Parts 1 & 2.
  - 2. State of California State Water Resources Control Board Regulations.
  - 3. US EPA's National Pollutant Discharge Elimination System EPA (NPDES).

## 1.04 SUBMITTAL

A. Comply with pertinent provisions of the general permit.

## PART 2 - PRODUCTS - (NOT USED)

## **PART 3 - EXECUTION**

## 3.01 SURFACE CONDITIONS

A. Examine the areas and conditions under which work of this section will be performed. Correct conditions detrimental to timely and proper completion of the work. Do not proceed until unsatisfactory conditions are corrected.

## 3.02 INSTALLATION

- A. Installation of the work shall be as indicated on the drawings as specified herein and regulatory requirements.
- B. Maintain the protection up to the project completion.

## 3.03 CLEANING

A. During and upon completion of the work comply with the general provisions of the general permit.

#### SECTION 01 60 00 - MATERIALS AND EQUIPMENT

#### PART 1 GENERAL

#### 1.01 SECTION INCLUDES

This Section includes administrative and procedural requirements governing products for incorporation into the Work.

#### 1.02 RELATED SECTIONS

A. Section 01300: Submittals

B. Section 01420: Testing and Inspection

C. Section 01640: Substitutions

D. Section 01740: Warranties

#### 1.03 DEFINITIONS

Definitions used in this Section are not intended to change the meaning of other terms used in the Contract Documents, such as "specialties," "systems," "structure," "finishes," "accessories," and other similar terms. Such terms are self-explanatory and have well-recognized meanings in the construction industry.

- A. "Products" are items purchased for incorporation into the Work, whether purchased for the Work or taken from previously purchased stock. The term "product" includes the terms "material" and "equipment" and terms of similar intent.
  - 1. "Named Products," are items identified by the manufacturer's product name, including make, model number or other designation, shown or listed in the manufacturer's published product literature, current as of the date of the Contract.
  - 2. "Foreign Products," as distinguished from "domestic products," are items substantially manufactured (50 percent or more of value) outside the United States and its possessions.
- B. "Materials" are products substantially shaped, cut, worked, mixed, finished, refined or otherwise fabricated, processed, or installed to form a part of the Work.
- C. "Equipment" is a product with operational parts, whether motorized or manually operated, that may require service connections, such as wiring or piping. Does not include machinery and equipment used for preparation, fabrication, conveying and erection of the Work.

## 1.04 QUALITY ASSURANCE

- A. Source Limitations: To the fullest extent possible, provide products of the same kind from a single source.
- B. Compatibility of Options: When the CONTRACTOR is given the option of selecting between two or more products for use in the Work, the product selected shall be compatible with products previously selected, even if previously selected products were also options.
- C. Nameplates: Except for required labels and operating data, do not attach or imprint manufacturer's or producer's nameplates or trademarks on exposed surfaces of products that will be exposed in view in occupied spaces or on the exterior.

- 1. Labels: Locate required product labels and stamps on concealed surfaces or, where required for observation after installation, on accessible surfaces that are not conspicuous.
- Equipment Nameplates: Provide a permanent nameplate on each item of service-connected or power-operated equipment. Locate on an easily accessible surface that is inconspicuous in occupied spaces. The nameplate shall contain the following information and other essential operating data:
  - a. Name of product and manufacturer
  - b. Model and serial number
  - c. Capacity
  - d. Speed
  - e. Ratings

## 1.05 PRODUCT DELIVERY, STORAGE, AND HANDLING

- A. Deliver, store, and handle products according to the manufacturer's recommendations, using means and methods that will prevent damage, deterioration, and loss, including theft.
  - 1. Schedule delivery to minimize long-term storage at the Project site and to prevent overcrowding of Work spaces.
  - 2. Coordinate delivery with installation time to assure minimum holding time for all items, but especially those that are flammable, hazardous, easily damaged, or sensitive to deterioration, theft, and other losses.
  - Deliver products to the Project site in an undamaged condition in the manufacturer's original sealed container or other packaging system, complete with labels and instructions for handling, storing, unpacking, protecting, and installing.
  - 4. Provide equipment and personnel to handle products by methods to prevent soiling, disfigurement or damage.
  - 5. Inspect products upon delivery to ensure compliance with the Contract Documents and to ensure that products are undamaged and properly protected.
  - 6. Store products at the Project site in a manner that will facilitate inspection and measurement of quantity or counting of units.
  - 7. Store heavy materials away from structures in a manner that will not endanger the structure's supporting construction.
  - 8. Store loose granular materials on solid flat surfaces in a well-drained area. Prevent mixing with foreign matter. Cover products subject to deterioration with impervious sheet covering.
  - 9. When approved by the District, provide off-site storage and protection in a bonded warehouse approved by District when site does not permit on-site storage or protection at no cost to the District.
  - 10. Store products subject to damage by the elements above ground, under cover in a weather-tight enclosure, with ventilation adequate to prevent

condensation. Maintain temperature and humidity within range required by manufacturer's instructions.

#### 1.06 MATERIAL SELECTION

- A. General Product Requirements: Provide products that comply with the Contract Documents, that are undamaged and, unless otherwise indicated, new at the time of installation.
  - 1. Provide products complete with accessories, trim, finish, safety guards, and other devices and details needed for a complete installation and the intended use and effect.
  - Standard Products: Where available, provide standard products of types that have been produced and used successfully in similar situations on other Projects.
- B. Product Selection Procedures: The Contract Documents and governing regulations govern product selection. Procedures governing product selection include the following:
  - 1. Proprietary Specification Requirements: Where Specifications name only a single material or manufacturer, provide the product indicated. No substitutions will be permitted.
  - 2. Semi-proprietary Specification Requirements: Where Specifications name two or more products or manufacturers, provide one of the products indicated throughout the Project. No substitutions will be permitted.
    - a. Where Specifications specify products or manufacturers by name, accompanied by the term "or equal," comply with General Conditions article on Substitutions to obtain approval for use of an unnamed product.
  - 3. Descriptive Specification Requirements: Where Specifications describe a product or assembly and list exact characteristics required, with or without use of a brand or trade name, provide a product or assembly that has the characteristics and otherwise complies with the Contract Documents.
  - 4. Performance Specification Requirements: Where Specifications require compliance with performance requirements, provide products that comply with these requirements and are recommended by the manufacturer for the application indicated.
    - a. Manufacturer's recommendations may be contained in published material literature or by the manufacturer's certification of performance.
  - 5. Compliance with Standards, Codes, and Regulations: Where Specifications only require compliance with an imposed code, standard or regulation, select a product that complies with the standards, codes, or regulations specified.
  - 6. Visual Matching: Where Specifications require matching an established Sample, decision of the ARCHITECT will be final on whether a proposed product matches satisfactorily.
  - 7. Visual Selection: Where specified product requirements include the phrase "... as selected from manufacturer's standard or premium colors, patterns, textures..." or a similar phrase, select a product and

manufacturer that complies with other specified requirements. The ARCHITECT will select the color, pattern, and texture from the product line selected.

## 1.07 INSTALLATION OF PRODUCTS

- A. Comply with manufacturer's instructions and recommendations for installation of products in the applications indicated. Anchor each product securely in place, accurately located, and aligned with other Work.
- B. Clean exposed surfaces and protect as necessary to ensure freedom from damage and deterioration until Substantial Completion.

# SECTION 01 60 00.01 - REQUEST FOR SUBSTITUTION

SU	Date:					
PR	PROJECT NAME: PACIFIC HIGH SCHOOL BUILDINGS G AND H DEMOLITION					
PR	OJECT NUMBER: 1-78-23					
то	: RUHNAU CLARKE ARCHITECTS . 3775 Tenth Street, Riverside, California 92501 From:					
	We hereby submit for your consideration the following product comparisons of the specified product and the proposed substitution. The undersigned fully understands that failure to answer any item below may be cause for rejection of request for substitution.					
	Request for substitution shall only be made during bidding (not later that to bid opening for inclusion by Addendum) except under conditions be Contractor.					
	Specified Product:					
	Project Manual Section Title Number Paragraph	age				
	Drawing No	Detail No.				
	Proposed Substitution:					
	Manufacturer:	Tel:				
A.	Reason request for substitution is being submitted:					
B.	Does proposed substitution affect in any way the Structural Safety, Accompliance, or Fire & Life Safety portions of the project? No Yes Explain	cess				

Does p	 proposed substitution affect dimensions, gages, weights, etc. on Drawing? No_		
Explaii	n		
	proposed substitution require changes in Drawings or design and installation es? No Yes		
cost of	f these changes is the responsibility of the Contractor.)		
	proposed substitution affect product cost, delivery time, or construction ule? NoYes Explain		
Does proposed substitution comply with specified ICC Number, UL Rating, ASTM Numbers? No Yes Explain			
	proposed substitution affect other trades and systems such as wiring, piping, prk, structure, etc.? No Yes (Explain which and how)		
	<del></del>		
If yes, Yes	has impact on their work been included in price of proposed substitution? No		
Does p	proposed substitution product guarantee differ from that of the specified product Yes Explain		
If the s \$	substitution request is accepted, it will result in: No cost impact credit of		
Attach substit	a listing of 3 projects (one in service for at least 3 years) using the proposed ution.		
specifi	antiating Data: Attach product data/brochures and Vendor qualifications for bo ed and substitute product. Provide samples for both specified and substitute cts, if applicable.		

specified product, understar	nas examined Construction Don't nds indicated application of pro t caused by the requested sub	oduct, and understands
Submitted by:		
(Type Name)	Signature	Date
Signature must be made by persterms.	son having legal authority to bi	ind his firm to the above
Architect's Comments: Accepted, accepted Reviewed by:	ed as noted, not accept	ted, received too late.
Architect	Date	
Construction Manager	Date	
District	 Date	

#### **SECTION 01 71 23 - FIELD ENGINEERING**

#### PART 1 - GENERAL

## 1.01 SECTION INCLUDES

A. Construction surveying requirements for the Work

## 1.02 RELATED SECTIONS

- A. Summary of the Work
- B. Project Coordination and Meetings
- C. Submittals
- D. Contract Closeout

## 1.03 SUBMITTALS

- A. CONTRACTOR shall submit the name and address of the State of California licensed surveyor to ARCHITECT and DISTRICT, including any changes as they may occur.
- B. At request of ARCHITECT and/or DISTRICT, CONTRACTOR shall submit copies of cut sheets, coordinate plots, data collector printouts, and other documentation as available to verify completeness and/or accuracy of field surveying Work

## PART 2 - PRODUCTS (Not applicable)

## PART 3 - EXECUTION

## 3.01 SURVEY REQUIREMENTS

- A. Establish a minimum of two permanent horizontal and vertical control points on the Project site, remote from the building area, referenced to data established by the survey control points.
- B. Indicate the reference points on the project record drawings with the basis of elevation being the established benchmarks.
- C. Establish lines, grades, locations and dimensions by instrumentation. Periodically, verify the layout of all Work by the same methods.
- D. Provide grade stakes and elevations for over-excavation and re-compaction, rough and final grades, paved areas, curbs, gutters, sidewalks, building pads, landscaped areas, and other areas as required.
- E. Calculate and layout proposed finished elevations and intermediate control as required to provide smooth transitions between the spot elevations indicated in the Contract Documents.
- F. Provide stakes and elevations for grading, fill, and topsoil placement.

- G. Provide adequate horizontal and vertical control to locate utility lines, including but not limited to, storm drains, sewers, water mains, gas, electric and signal, and provide vertical control in proportion to the slope of the line as required for accurate construction.
  - 1. Prior to trench closure, survey and record locations and invert and flow line elevations at manholes, POCs, and 50-foot intervals.
  - Survey and record top of curb and flow line elevations on finished concrete or AC surfaces at key locations such as BC's, EC's, grade breaks, corners or angle points in sufficient number to demonstrate the Work complies with the intent of the Contract Documents.
- H. Provide horizontal and vertical control for batter boards for drainage, utility, and other on-site structures as required.
- Furnish building corner offsets as required to adequately locate building pads.
   Provide cut and fill stakes within the building pad perimeter adequate to control
   both over excavation and re-compaction and the final sub-grade elevation of the
   building pad.
- J. Submit a certification signed by the surveyor confirming that the elevations and locations of improvements are in conformance with the Contract Documents. The statement shall include survey notes for the finish floor and building pad, showing the actual measured elevations on the completed sub-grade, recorded to the nearest 0.01'. Building pad tolerance will be +- 0.10'.

## 3.02 RECORD DRAWINGS

- A. The surveyor shall record all horizontal and vertical control information on "as-built" Record Drawings, as coordinates and elevations. Record drawings shall indicate locations of all utilities information, as described above.
- B. Upon Substantial Completion, CONTRACTOR shall deliver to the ARCHITECT Electronic CAD file as the final Record Drawings. CAD version to be determined by DISTRICT.
- C. Completed record drawing Electronic files shall be signed by the licensed surveyor, certifying that the information shown is correct and is in conformance with the Contract Documents within specified tolerances.
- D. Where other sections of the Contract Documents require verification or measurements of installed Work by survey, the surveyor shall perform and certify that all such surveys or verifications are completed in accordance with the Contract Documents.

#### **SECTION 01 73 29 - CUTTING AND PATCHING**

#### PART 1 GENERAL

#### 1.01 SECTION INCLUDES

Requirements and limitations for cutting and patching of work.

## 1.02 SCOPE

- A. Where the work requires that a particular existing building element, such as a partition, wall, paving, window or similar element of existing building construction, be removed, it is the intention of this specification that such work be a part of the demolition section and not a part of cutting and patching. Refer to individual category scope of work sheets to determine the limits of demolition work for each CONTRACTOR.
- B. New work required to replace such removals is considered as a part of the separate sections of the specifications covering similar new construction.
- C. Where incidental cutting and patching is required for the installation of a specific item or piece of equipment (including piping, ductwork, conduit, etc.), all such cutting and patching is considered to be specified as a part of the section requiring the cutting and patching, but shall also comply with the requirements of this Section.
- D. CONTRACTOR shall verify and check all areas to be cut and patched and shall coordinate the work of the various trades involved.
- G. Unless specifically designated otherwise, existing work cut, altered or revised to accommodate new work shall be patched to duplicate undisturbed adjacent finishes, colors, textures and profiles. New work in existing portions shall also be finished to match adjacent existing work unless noted otherwise.

#### 1.03 SUBMITTALS

- A. Submit written request in advance of cutting or alteration which affects:
  - 1. Structural integrity of any element of Project.
  - 2. Integrity of weather-exposed or moisture-resistant element.
  - 3. Efficiency, maintenance or safety of any operational element.
  - 4. Visual qualities of sight exposed elements.
  - 5. Work of DISTRICT or separate CONTRACTOR.

#### B. Include in request:

- 1. Identification of Project.
- 2. Location and description of affected work.
- 3. Necessity for cutting or alteration.
- 4. Alternatives to cutting and patching.
- 5. Description of proposed work and products to be used.

- Effect on work of District or separate CONTRACTOR.
- 7. Written permission of affected separate CONTRACTOR.
- 8. Date and time work will be executed.
- C. Obtain approval of ARCHITECT before proceeding with any cutting and patching:

# **PART 2 PRODUCTS**

#### 2.01 MATERIALS

A. Primary Products: Those required for original installation, unless specifically approved otherwise

### PART 3 EXECUTION

## 3.01 EXAMINATION

- A. Inspect existing conditions prior to commencing work, including elements subject to damage or movement during cutting and patching. Confirm status and current warranties and guarantees.
- B. After uncovering existing work, inspect conditions affecting performance of work.
  - Prior to cutting, boring or drilling through new or existing structural members or elements including reinforcing bars, CONTRACTOR shall prepare detailed drawings for review by the ARCHITECT and approval by the Division of the State Architect (DSA). Agency approvals shall be obtained by the ARCHITECT, not CONTRACTOR.
- C. Beginning of cutting or patching means acceptance of existing conditions.

#### 3.02 PREPARATION

- A. Provide temporary support to ensure structural integrity of the work. Provide devices and methods to protect other portions of Project from damage.
- B. Provide protection from elements for areas which may be exposed by uncovering work.
- C. Maintain excavations free of water.
- D. Where the Work requires sandblasting of existing surfaces in order to receive new materials secured by cementitious, adhesive or chemical bond, completely remove existing finishes, stains, oil, grease, bitumen, mastic and adhesives or other substances deleterious to the new bonding and/or fastening of new Work. Utilize wet sand blasting for interior surfaces and for exterior surfaces where necessary to prevent objectionable production of dust.

## 3.03 PERFORMANCE

- A. Employ skilled workmen to perform cutting and patching. Proceed with cutting and patching at the earliest feasible time and complete without delay. Carefully remove existing Work to be salvaged and/or reinstalled. Protect and store for reuse in the Work. Verify compatibility and suitability of existing substrates before starting the Work.
- B. The word "cutting" as used in the Contract Documents includes, but is not limited to, cutting, drilling, chopping, and other similar operations and the word "patching" includes, but is not limited to, patching, rebuilding, reinforcing, repairing, refurbishing, restoring, replacing, or other similar operations.
- C. **Cutting:** Cut existing construction using methods least likely to damage elements retained or adjoining Work. Where possible, review proposed procedures with the original installer; comply with the original installer's recommendations.
  - 1. In general, where cutting, use hand or small power tools designed for sawing or grinding, not hammering and chopping. Cut holes and slots as small as possible, neatly to size required, and with minimum disturbance of adjacent surfaces. Temporarily cover openings when not in use.
  - 2. To avoid marring existing finished surfaces, cut or drill from the exposed or finished side into concealed surfaces.
  - 3. Cut through concrete and masonry using a cutting machine, such as a carborundum saw or a diamond-core drill. Saw cut reinforcing bars and paint ends with bituminous paint except where bonded into new concrete or masonry.
  - 4. Comply with requirements of applicable Division 2 Sections where cutting and patching requires excavating, backfill, and/or recompaction.
  - 5. Woodwork: Cut and or remove to a panel or joint line.
  - 6. Sheet Metal: Remove back to joint, lap, or connection. Secure loose or unfastened ends or edges and seal watertight.
  - 7. Glass: Remove cracked, broken, or damaged glass and clean rebates and stops of setting materials.
  - 8. Plaster: Cut back to sound plaster on straight lines, and back bevel edges of remaining plaster. Trim existing lath and prepare for new lath.
  - 9. Gypsum Wallboard: Cut back on straight lines to undamaged surfaces with at least two opposite cut edges centered on supports.
  - 10. Acoustical ceilings: Remove hanger wires and related appurtenances where ceilings are not scheduled to be installed.
  - 11. Tile: Cut back to sound tile and backing on joint lines.

- 12. Flooring: Completely remove flooring and clean backing of prior adhesive. Carefully remove wood flooring for patching and repairing of existing wood flooring scheduled to remain.
- C. **Patching**: Patch with durable seams that are as invisible as possible. Comply with required tolerances.
  - Where feasible, inspect and test patched areas to demonstrate integrity of the installation. Verify conditions of existing substrates prior to executing Work.
  - 2. Restore exposed finishes of patched areas and extend finish restoration into retaining adjoining construction in a manner that will eliminate all evidence of patching and refinishing.
  - 3. Concrete: Maintain cut edges in a moist condition for twenty-four (24) hours prior to the placement of new concrete. In lieu of this an epoxy adhesive may be provided. Finish placed concrete to match existing unless noted otherwise. Concrete shall have a compressive strength of 3,000 psi where installed to repair and/or match existing improvements, unless noted otherwise.
  - 4. Metal Fabrications: Items to remain exposed shall have their edges cut and ground smooth and rounded.
  - 5. Sheet Metal: Replace removed or damaged sheet metal items as required for new Work.
  - 6. Glass: Install matching glass and re-seal exterior window assemblies.
  - 7. Lath and Plaster: Install new lath materials to match existing and fasten to supports at 6" centers. Provide a 6" lap where new lath to adjoins existing lath. Fasten new lath as required for new Work. Restore paper backings as required. Apply a bonding agent on cut edges of existing plaster. Apply three coat plaster of the type, thickness, finish, texture, and color to match existing.
  - 8. Gypsum Wallboard: Fasten cut edges of wallboard. Install patches with at least two opposite edges centered on supports and secure at 6" centers. Tape and finish joints and fastener heads. Patching shall be non-apparent when painted or finished.
  - 9. Acoustical Ceilings: Comply with the requirements for new Work specified in related sections of the Contract Documents.
  - 10. Resilient Flooring: Completely remove flooring and prepare substrate for new material.
  - 11. Paint: Prepare areas to be painted as specified for painting specific surfaces in the painting and coatings Sections of the Specifications.
- D. Fit work air tight to pipes, sleeves, ducts, conduit and other penetrations through surfaces.

- E. At penetrations of fire-rated walls, partitions, ceiling or floor construction, completely seal voids with fire-rated devices or material in accordance with Section 07270, to full thickness of the penetrated element.
- F. Refinish surfaces to match adjacent finish. For continuous surfaces, refinish to nearest intersection or natural break. For an assembly, refinish entire unit.

## 3.05 SLEEVES AND HANGERS

- A. Provide conduit, outlets, piping sleeves, boxes, inserts or other materials or equipment necessary to be built into work.
- B. In the event delays occur in delivery of sleeves or other materials, arrange to have boxes or other forms set at locations where piping or other material is to pass through or into slabs or other work.
- C. Upon subsequent installation of sleeves or other material, install fill materials to completely seal voids with fire-rated devices or moisture-resistant material, to full thickness of the penetrated element. Necessary expenditures incurred for boxing out or filling shall be without extra cost to the DISTRICT.

#### SECTION 01 74 19 - CONSTRUCTION WASTE MANAGEMENT AND DISPOSAL

## **PART 1 GENERAL**

#### 1.01 WASTE MANAGEMENT REQUIREMENTS

- A. Comply with the requirements Section 5.408 of the California Green Building Standards Code.
  - Recycle and/or salvage for reuse a minimum of 50 percent of the nonhazardous construction and demolition waste in accordance with Section 504.8.1.1, 5.408.1.2, or 5.408.1.3; or meet a local construction and demolition waste management ordinance, whichever is more stringent.
- B. District requires that this project generate the least amount of trash and waste possible.
- C. Employ processes that ensure the generation of as little waste as possible due to error, poor planning, breakage, mishandling, contamination, or other factors.
- D. Minimize trash/waste disposal in landfills; reuse, salvage, or recycle as much waste as economically feasible.
- E. Required Recycling, Salvage, and Reuse: The following may not be disposed of in landfills or by incineration:
  - 1. Aluminum and plastic beverage containers.
  - 2. Corrugated cardboard.
  - 3. Wood pallets.
  - 4. Clean dimensional wood: May be used as blocking or furring.
  - Land clearing debris, including brush, branches, logs, and stumps; see Section 31 10 00 - Site Clearing for use options.
    - a. Comply with California Green Code (CGC) 5.408.3; Excavated soil and land clearing debris: 100 percent of trees, stumps, rocks and associated vegetation and soils resulting primarily from land clearing shall be reused or recycled.
      - 1) Exception: Reuse, either on-or off-site, of vegetation or soil contaminated by disease or pest infestation.
  - 6. Concrete: May be crushed and used as riprap, aggregate, sub-base material, or fill.
  - 7. Bricks: May be used on project if whole, or crushed and used as landscape cover, sub-base material, or fill.
  - 8. Concrete masonry units: May be used on project if whole, or crushed and used as sub-base material or fill.
  - 9. Asphalt paving.
  - 10. Metals, including packaging banding, metal studs, sheet metal, structural steel, piping, reinforcing bars, door frames, and other items made of steel, iron, galvanized steel, stainless steel, aluminum, copper, zinc, lead, brass, and bronze.
  - 11. Glass.
  - 12. Gypsum drywall and plaster.
  - 13. Carpet, carpet cushion, carpet tile, and carpet remnants: DuPont (http://flooring.dupont.com) and Interface (www.interfaceinc.com) conduct reclamation programs.

- 14. Roofing.
- 15. Paint.
- 16. Plastic sheeting.
- 17. Rigid foam insulation.
- 18. Windows, doors, and door hardware.
- 19. Plumbing fixtures.
- 20. Mechanical and electrical equipment.
- 21. Fluorescent lamps (light bulbs).
- 22. Acoustical ceiling tile and panels.
- 23. Materials which could be hazardous and subject to special disposal regulations include but are not limited to the following:
  - a. Lead-Based Paint
  - b. Asbestos: Found in older pipe insulation, asphalt floor tiles, linoleum, insulation, etc.
  - c. Polychlorinated Biphenyls (PCBs):
    - 1) Found in electrical oil filled equipment manufactured prior to 1978 such as transformers, switches and fluorescent lamp ballasts.
    - 2) Also found in adhesive, sealant, caulk, glazing putty, roofing material, pesticide vehicle, ink, paper, fabric dye, gaskets, and hydraulic fluid.
  - d. HVAC Refrigerants: Containing Fluorinated and Chlorinated compounds.
  - e. Drinking Fountain Refrigerants: Containing Fluorinated and Chlorinated compounds.
  - f. Fluorescent Light Tubes: Contain mercury.
  - g. EXIT signs and Smoke Detectors: May contain unregulated, radioactive tritium. Required to be returned to manufacturer.
  - h. Contaminated Soils.
  - i. Pressure Treated Lumber.
- F. Contractor shall submit periodic Waste Disposal Reports; all landfill disposal, recycling, salvage, and reuse must be reported regardless of to whom the cost or savings accrues; use the same units of measure on all reports.
  - Contractor's quantitative reports for construction waste materials as a condition of approval of progress payments.
- G. Contractor shall develop and follow a Waste Management Plan designed to implement these requirements. CalGreen Section 5.408.1.1.
- H. The following sources may be useful in developing the Waste Management Plan:
  - 1. California Recycling Department, at www.bsc.ca.gov/Home/CALGreen.aspx.
  - 2. General information contacts regarding construction and demolition waste:
    - a. EPA Construction and demolition (C&D) debris website: www.epa.gov/epawaste/conserve/imr/cdm/.
    - b. Directory of Wood-Framed Building Deconstruction and Reused Building Materials Companies: www.fpl.fs.fed.us/documnts/fplgtr/fpl\_gtr150.pdf.
    - c. Additional resources to be developed by Contractor with assistance from District and Contractor, as requested.

- 3. Recycling Haulers and Markets: The source list below contains local haulers and markets for recyclable materials. This list is provided for information only and is not necessarily comprehensive; other haulers and markets are acceptable.
  - a. CAL-MAX: www.calrecycle.ca.gov/calmax/.
    - 1) A free service designed to help businesses find markets for nonhazardous materials they have traditionally discarded.
  - b. General Recycling/Reuse Centers: For information on qualified local solid waste haulers contact the California Department of Resources Recycling and Recovery - CalRecycle. The website lists wastes recycling facilities in counties throughout the State of California.
    - 1) http://www.calrecycle.ca.gov/default.asp
- I. Methods of trash/waste disposal that are not acceptable are:
  - 1. Burning on the project site.
  - 2. Burying on the project site.
  - 3. Dumping or burying on other property, public or private.
  - 4. Other illegal dumping or burying.
  - 5. Incineration, either on- or off-site.
- J. Regulatory Requirements: Contractor is responsible for knowing and complying with regulatory requirements, including but not limited to Federal, state and local requirements, pertaining to legal disposal of all construction and demolition waste materials.

#### 1.02 RELATED REQUIREMENTS

- A. Section 01 30 00 Administrative Requirements: Additional requirements for project meetings, reports, submittal procedures, and project documentation.
- B. Section 01 52 00 Construction Facilities: Additional requirements related to trash/waste collection and removal facilities and services.
- C. Section 01 60 00 Product Requirements: Waste prevention requirements related to delivery, storage, and handling.
- D. Section 01 70 00 Execution and Closeout Requirements: Trash/waste prevention procedures related to demolition, cutting and patching, installation, protection, and cleaning.
- E. Section 31 10 00 Site Clearing: Handling and disposal of land clearing debris.

## 1.03 DEFINITIONS

- A. Clean: Untreated and unpainted; not contaminated with oils, solvents, caulk, or the
- B. Construction and Demolition Waste: Solid wastes typically including building materials, packaging, trash, debris, and rubble resulting from construction, remodeling, repair and demolition operations.
  - 1. Debris that is not hazardous as defined in California Code of Regulations, Title 22, Section 66261.3 et seq.
  - 2. This term includes, but is not limited to, asphalt concrete, Portland cement concrete, brick, lumber, gypsum wallboard, cardboard and other associated packaging, roofing material, ceramic tile, carpeting, plastic pipe, and steel.

- 3. The debris may be commingled with rock, soil, tree stumps, and other vegetative matter resulting from land clearing and landscaping for construction or land development projects.
- C. Disposal: Removal off-site of demolition and construction waste and subsequent sale, recycling, reuse, or deposit in landfill or incinerator acceptable to authorities having jurisdiction.
- D. Diversion: Avoidance of demolition and construction waste sent to landfill or incineration. Diversion does not include using materials for landfill, alternate daily cover on landfills, or materials used as fuel in waste-to-energy processes.
- E. Enforcement Agency (EA). Enforcement agency as defined in CA Public Resources Code 40130.
- F. Hazardous: Exhibiting the characteristics of hazardous substances, i.e., ignitibility, corrosivity, toxicity or reactivity.
- G. Landfill, Inert waste or Inert Disposal Facility:
  - 1. A disposal facility that accepts only inert waste such as soil and rock, fully cured asphalt paving, uncontaminated concrete (including fiberglass or steel reinforcing rods embedded in the concrete), brick, glass, and ceramics, for land disposal.

## H. Landfill, Class III:

- A landfill that accepts non-hazardous resources such as household, commercial, and industrial waste, resulting from construction, remodeling, repair, and demolition operations.
- 2. A Class III landfill must have a solid waste facilities permit from the California Integrated Waste Management Board (CIWMB) and is regulated by the Enforcement Agency (EA).
- I. Mixed Debris: Loads that include commingled recyclable and non-recyclable materials generated at the construction site.
- J. Mixed Debris Recycling Facility: A processing facility that accepts loads of commingled construction and demolition debris for the purpose of recovering re-usable and recyclable materials and disposing the non-recyclable residual materials.
- K. Nonhazardous: Exhibiting none of the characteristics of hazardous substances, i.e., ignitibility, corrosivity, toxicity, or reactivity.
- L. Nontoxic: Neither immediately poisonous to humans nor poisonous after a long period of exposure.
- M. Recyclable: The ability of a product or material to be recovered at the end of its life cycle and remanufactured into a new product for reuse by others.
- N. Recycle: To remove a waste material from the project site to another site for remanufacture into a new product for reuse by others.
- O. Recycling: The process of sorting, cleansing, treating and reconstituting solid waste and other discarded materials for the purpose of using the altered form. Recycling does not include burning, incinerating, or thermally destroying waste.
- P. Recycling Center: A facility that receives only C&D material that has been separated for reuse prior to receipt, in which the residual (disposed) amount of waste in the material is less than 10% of the amount separated for reuse by weight.
- Q. Return: To give back reusable items or unused products to vendors for credit.
- R. Reuse: To reuse a construction waste material in some manner on the project site.
- S. Salvage: To remove a waste material from the project site to another site for resale or reuse by others.

- T. Sediment: Soil and other debris that has been eroded and transported by storm or well production run-off water.
- U. Separated for Reuse:
  - Materials, including commingled recyclables.
  - 2. Separated or kept separate from the solid waste stream for the purpose of:
    - a. Additional sorting or processing those materials for reuse or recycling.
      - In order to return them to the economic mainstream in the form of raw material for new, reused, or reconstituted products.
    - b. Products shall meet the quality standards necessary to be used in the marketplace.
    - c. Includes materials that have been "source separated".

## V. Solid Waste:

- 1. All putrescible and nonputrescible solid, semisolid, and liquid wastes, including:
  - a. Garbage, trash, refuse, paper, rubbish, ashes, industrial wastes, demolition and construction wastes.
  - b. Abandoned vehicles and parts thereof.
  - c. Discarded home and industrial appliances.
  - d. Dewatered, treated, or chemically fixed sewage sludge which is not hazardous waste.
  - e. Manure, vegetable or animal solid and semisolid wastes.
  - f. Other discarded solid and semisolid wastes.
- 2. "Solid waste" does not include hazardous waste, radioactive waste, or medical waste as defined or regulated by State law.
- W. Source Separation: The act of keeping different types of waste materials separate beginning from the first time they become waste.
  - Materials, including commingled recyclables, that have been separated or kept separate from the solid waste stream at the point of generation, for the purpose of additional sorting or processing of those materials for reuse or recycling in order to return them to the economic mainstream in the form of raw materials for new, reused, or reconstituted products which meet the quality standards necessary to be used in the marketplace.
- X. Toxic: Poisonous to humans either immediately or after a long period of exposure.
- Y. Trash: Any product or material unable to be reused, returned, recycled, or salvaged.
- Z. Waste: Extra material or material that has reached the end of its useful life in its intended use. Waste includes salvageable, returnable, recyclable, and reusable material.
- AA. Waste Hauler: A company that possesses a valid permit from the local waste management authority to collect and transport solid wastes from individuals or businesses for the purpose of recycling or disposal in the locality.

## 1.04 SUBMITTALS

- A. See Section 01 30 00 Administrative Requirements, for submittal procedures.
- B. Submit Waste Management Plan within 30 calendar days after receipt of Notice to Proceed, or prior to any trash or waste removal, whichever occurs sooner; submit projection of all trash and waste that will require disposal and alternatives to landfilling.
  - 1. Submit four copies of CWMP for review.

- a. Contractor's Construction Waste and Recycling Plan must be approved by the Architect and Construction Manager prior to the start of Work.
- 2. Approval of the Contractor's CWMP shall not relieve the Contractor of responsibility for adequate and continuing control of pollutants and other environmental protection measures.
- C. Waste Management Plan: Include the following information:
  - 1. Analysis of the trash and waste projected to be generated during the entire project construction cycle, including types and quantities.
  - 2. Landfill Options: The name, address, and telephone number of the landfill(s) where trash/waste will be disposed of, the applicable landfill tipping fee(s), and the projected cost of disposing of all project trash/waste in the landfill(s).
  - 3. Landfill Alternatives: List all waste materials that will be diverted from landfills by reuse, salvage, or recycling.
    - a. List each material proposed to be salvaged, reused, or recycled.
    - b. List the local market for each material.
  - 4. Meetings: Describe regular meetings to be held to address waste prevention, reduction, recycling, salvage, reuse, and disposal.
  - Materials Handling Procedures: Describe the means by which materials to be diverted from landfills will be protected from contamination and prepared for acceptance by designated facilities; include separation procedures for recyclables, storage, and packaging.
  - 6. Transportation: Identify the destination and means of transportation of materials to be recycled; i.e. whether materials will be site-separated and self-hauled to designated centers, or whether mixed materials will be collected by a waste hauler.
  - 7. Recycling Incentives: Describe procedures required to obtain credits, rebates, or similar incentives.
- D. Waste Disposal Reports: Submit at specified intervals, with details of quantities of trash and waste, means of disposal or reuse, and costs; show both totals to date and since last report.
  - 1. Submit updated Report with each Application for Progress Payment; failure to submit Report will delay payment.
    - a. Inert materials shall achieve a construction waste diversion rate of at least 95 percent.
      - 1) These materials include, but are not limited to, concrete, asphalt and rock.
      - Earthwork is not included.
      - 3) Excavated soil shall not be included in any of the calculations used to ensure compliance with this specification section.
    - The overall diversion rate must be based on weight.
    - c. The diversion rate of individual materials can be measured in either weight or volume, but the rate shall be converted into the units selected for calculating the overall diversion rate.
      - All individual material diversions must be converted to a consistent set of units when calculating the overall diversion rate for the all reports and submittals required for the Work.
    - d. Conversion rate numbers shall be based on standard conversion rate data for construction projects provided by the California Integrated Waste

Management Board (CIWMB). This data is available at the following internet location, http://www.calrecycle.ca.gov/LGCentral/Library/dsg/ICandD.htm.

- 2. Submit Report on a form acceptable to District.
- 3. Landfill Disposal: Include the following information:
  - a. Identification of material.
  - b. Amount, in tons or cubic yards, of trash/waste material from the project disposed of in landfills.
  - c. State the identity of landfills, total amount of tipping fees paid to landfill, and total disposal cost.
  - d. Include manifests, weight tickets, receipts, and invoices as evidence of quantity and cost.
- 4. Recycled and Salvaged Materials: Include the following information for each:
  - Identification of material, including those retrieved by installer for use on other projects.
  - b. Amount, in tons or cubic yards, date removed from the project site, and receiving party.
  - c. Transportation cost, amount paid or received for the material, and the net total cost or savings of salvage or recycling each material.
  - Include manifests, weight tickets, receipts, and invoices as evidence of quantity and cost.
  - e. Certification by receiving party that materials will not be disposed of in landfills or by incineration.
- 5. Material Reused on Project: Include the following information for each:
  - a. Identification of material and how it was used in the project.
  - b. Amount, in tons or cubic yards.
  - c. Include weight tickets as evidence of quantity.
- 6. Other Disposal Methods: Include information similar to that described above, as appropriate to disposal method.

## **PART 2 PRODUCTS**

## 2.01 PRODUCT SUBSTITUTIONS

- A. See Section 01 60 00 Product Requirements for substitution submission procedures.
- B. For each proposed product substitution, submit the following information in addition to requirements specified in Section 01 60 00:
  - 1. Relative amount of waste produced, compared to specified product.
  - 2. Cost savings on waste disposal, compared to specified product, to be deducted from the Contract Sum.
  - 3. Proposed disposal method for waste product.
  - 4. Markets for recycled waste product.

## PART 3 EXECUTION

#### 3.01 WASTE MANAGEMENT PROCEDURES

A. See Section 01 30 00 for additional requirements for project meetings, reports, submittal procedures, and project documentation.

- B. See Section 01 52 00 for additional requirements related to trash/waste collection and removal facilities and services.
- C. See Section 01 60 00 for waste prevention requirements related to delivery, storage, and handling.
- D. See Section 01 70 00 for trash/waste prevention procedures related to demolition, cutting and patching, installation, protection, and cleaning.

## 3.02 WASTE MANAGEMENT PLAN IMPLEMENTATION

- A. Manager: Designate an on-site person or persons responsible for instructing workers and overseeing and documenting results of the Waste Management Plan.
- B. Communication: Distribute copies of the Waste Management Plan to job site foreman, each subcontractor, District, and Architect.
- C. Instruction: Provide on-site instruction of appropriate separation, handling, and recycling, salvage, reuse, and return methods to be used by all parties at the appropriate stages of the project.
- D. Meetings: Discuss trash/waste management goals and issues at project meetings.
  - 1. Pre-bid meeting.
  - 2. Pre-construction meeting.
  - 3. Regular job-site meetings.
- E. Facilities: Provide specific facilities for separation and storage of materials for recycling, salvage, reuse, return, and trash disposal, for use by all contractors and installers.
  - 1. As a minimum, provide:
    - a. Separate area for storage of materials to be reused on-site, such as wood cut-offs for blocking.
    - b. Separate dumpsters for each category of recyclable.
    - c. Recycling bins at worker lunch area.
  - 2. Provide containers as required.
  - 3. Provide temporary enclosures around piles of separated materials to be recycled or salvaged.
  - 4. Provide materials for barriers and enclosures that are nonhazardous, recyclable, or reusable to the maximum extent possible; reuse project construction waste materials if possible.
  - 5. Locate enclosures out of the way of construction traffic.
  - 6. Provide adequate space for pick-up and delivery and convenience to subcontractors.
  - 7. If an enclosed area is not provided, clearly lay out and label a specific area onsite.
  - 8. Keep recycling and trash/waste bin areas neat and clean and clearly marked in order to avoid contamination of materials.
- F. Hazardous Wastes: Separate, store, and dispose of hazardous wastes according to applicable regulations.
- G. Recycling: Separate, store, protect, and handle at the site identified recyclable waste products in order to prevent contamination of materials and to maximize recyclability of identified materials. Arrange for timely pickups from the site or deliveries to recycling facility in order to prevent contamination of recyclable materials.

- H. Reuse of Materials On-Site: Set aside, sort, and protect separated products in preparation for reuse.
- I. Salvage: Set aside, sort, and protect products to be salvaged for reuse off-site.

## 3.03 DISPOSAL OPERATIONS AND WASTE HAULING

- A. Remove waste materials from Project Site and legally dispose of them in a landfill or incinerator acceptable to authorities having jurisdiction.
  - 1. Except for items or materials to be salvaged, recycled, or otherwise reused.
  - 2. Except as otherwise specified, do not allow waste materials that are to be disposed of to accumulate on site.
  - 3. Use a permitted waste hauler or Contractor's trucking services and personnel. To confirm valid permitted status of waste haulers, contact the local solid waste authority.
  - 4. Become familiar with the conditions for acceptance of new construction, excavation and demolition materials at recycling facilities, prior to delivering materials.
  - 5. Deliver to facilities that can legally accept new construction, excavation and demolition materials for purpose of re-use, recycling, composting, or disposal.
  - 6. Remove and transport debris in a manner that will prevent spillage on adjacent surfaces and areas.
  - 7. Do not burn or bury waste materials on or off site. Appropriate on-site topical application of ground gypsum or wood, or use of site paving as granulated fill is considered reuse, not waste.

## 3.04 PLAN AND REPORT FORMS

A. See suggested forms on the following pages.

#### **SECTION 01 77 00 - CLOSEOUT PROCEDURES**

#### PART 1 GENERAL

#### 1.01 SECTION INCLUDES

- A. This Section includes administrative and procedural requirements for Contract Closeout, including but not limited to, the following:
  - 1. Completion Procedures
  - 2. Project Record Documents
  - 3. Operation and Maintenance Manuals
  - 4. Orientation and Instruction of DISTRICT'S Personnel
  - 5. Warranties and Guarantees
  - 6. Spare Parts and Materials
  - 7. Final Cleaning
- B. Additional closeout requirements for specific Work activities are included in the appropriate Sections in Divisions 02 through 16.

### 1.02 RELATED SECTIONS

- A. Price and Payment Procedures
- B. Submittals
- C. Construction Progress Schedule
- D. Construction Facilities
- E. Temporary Controls
- F. Warranties
- G. Project Record Documents

## 1.03 COMPLETION PROCEDURES

- A. Substantial Completion and Partial Occupancy:
  - 1. Conform to Title 24, Part 1, Section 4-336 CCR, Requirements for Verified Reports and Closeout Procedures.
  - In conjunction with the IOR, prepare a list of items to be completed or corrected. List may be developed by areas, when approved by the ARCHITECT.
  - 3. Within a reasonable time after receipt of the list, the ARCHITECT will inspect to determine status of completion.
  - 4. Should the ARCHITECT determine that Work is not substantially complete:

- a. The ARCHITECT will promptly notify the CONTRACTOR in writing, giving the reasons for his determination.
- b. CONTRACTOR shall remedy the deficiencies and notify the ARCHITECT when Work is ready for re-inspection.
- c. The ARCHITECT will re-inspect the Work.
- 5. When the ARCHITECT concurs that work is substantially complete:
  - a. The ARCHITECT will prepare a "Certificate of Substantial Completion" on AIA Form G704, accompanied by the CONTRACTOR's list of items to be completed or corrected as verified by the ARCHITECT.
  - b. The ARCHITECT will submit the Certificate to the DISTRICT and to the CONTRACTOR for their written acceptance of the responsibilities assigned to them in the Certificate.

## B. Final Completion:

- 1. Verify the Work is complete.
- 2. Prepare and submit a notice that Work is ready for final inspection and acceptance.
- 3. Certify that:
  - a. Work has been inspected by all governing agencies and is in compliance with all governing regulations.
  - b. Work has been inspected for compliance with the Contract Documents.
  - c. Work has been completed in accordance with the Contract Documents.
  - d. Equipment and systems have been tested as required and are operational.
  - e. Work is completed and ready for final inspection.
- 4. The ARCHITECT will make an inspection to verify status of completion.
- 5. Should the ARCHITECT determine the Work is incomplete or defective:
  - a. The ARCHITECT will promptly notify the CONTRACTOR in writing, listing incomplete or defective work.
  - b. CONTRACTOR shall remedy the deficiencies promptly and notify the ARCHITECT when ready for re-inspection.

- 6. When the ARCHITECT determines the Work is acceptable under the Contract Documents, he will request the CONTRACTOR to make closeout submittals.
- C. Submit all closeout documents, including but are not limited to:
  - 1. Project Record Documents.
  - 2. Operation and Maintenance Manuals (for all items requiring special knowledge for operation or for maintenance, listed in pertinent Sections of these Specifications), and for other items when so approved by the ARCHITECT.
  - Warranties and Guarantees.
  - 4. Keys and Keying Schedule.
  - 5. Spare parts, materials, extra stock to be turned over to the DISTRICT.
  - 6. Evidence of payment and release of liens, when requested by DISTRICT.
  - 7. List of subcontractors, service organizations and principal vendors, including names, addresses and telephone numbers, where they may be contacted for emergency service at all times, including nights, weekends and holidays.
- D. Final Payment:

Submit a Final Payment Request, showing all adjustments to the Contract Sum.

#### 1.04 VERIFIED REPORTS

A. Construction progress of the Work shall be reported to DSA via a duly verified report in accordance with Sections 4-336 and 4-343 of the California Building Standards Administrative Code.

#### 1.05 OPERATION AND MAINTENANCE MANUALS

- A. Prior to Substantial Completion, submit three (3) sets of Operation and Maintenance (O&M) Manuals and one (1) electronic copy to the ARCHITECT for DISTRICT's records. Organize O&M data into sets of manageable size. Bind properly indexed data in individual, heavy-duty, 2"-3", 3-ring, durably covered binders, with pocket folders for folded sheet information. Mark appropriate identification on front and spine of each binder.
  - 1. Emergency instructions
  - 2. Manufacturer's operating and maintenance instructions, including any seasonal adjustments
  - 3. Spare parts list
  - 4. Copies of warranties
  - 5. Wiring diagrams
  - 6. Recommended "turn-around" cycles

- 7. Inspection procedures
- 8. Shop Drawings and Product Data
- 9. Fixture lamping schedule

## 1.06 ORIENTATION AND INSTRUCTION OF DISTRICT'S PERSONNEL:

- A. Instruct the DISTRICT's personnel in proper operation and maintenance of all systems, equipment and similar items, which were provided as part of the work. Provide maintenance and inspection schedules that conform to manufacturer's recommendations. Provide instruction by manufacturers' representatives if installers are not experienced in operation and maintenance procedures. Include a detailed review of the following items:
  - 1. Maintenance manuals
  - Record documents
  - 3. Spare parts and materials
  - 4. Tools
  - 5. Lubricants
  - Fuels
  - 7. Identification systems
  - 8. Control sequences
  - 9. Hazards
  - 10. Cleaning
  - 11. Warranties and bonds
  - 12. Maintenance agreements and similar continuing commitments
- B. CONTRACTOR shall provide a schedule to the DISTRICT for approval for each of the instruction periods required.
  - 1. Organize the instruction sessions into group sizes and schedule the elapsed time for instruction in a manner to provide complete coverage of the subject matter. Video tape each session and provide DISTRICT with two (2) copies.
- C. Instruction sessions will be held in a DISTRICT designated area on the project site and at DISTRICT's convenience. Amount of time required for each session shall be as specified in individual sections, but in no case less than the time needed to fully convey the information needed by DISTRICT personnel for operating and maintaining the products.
- D. Instructors shall be qualified by the product manufacturer in the subject matter presented at each session.
  - 1. Submit names of instructors and qualifications to the Architect and DISTRICT for approval, 30 days prior to each scheduled session.
  - 2. Substitution of instructors will not be permitted without prior approval of Architect or DISTRICT.

- E. As part of instruction for operating equipment, demonstrate the following procedures:
  - 1. Start-up
  - 2. Shutdown
  - 3. Emergency operations
  - 4. Noise and vibration adjustments
  - 5. Safety procedures
  - 6. Seasonal adjustments
  - 7. Economy and efficiency adjustments
  - 8. Effective energy utilization measures
- F. Schedule and provide seasonal or periodic training sessions when specified in technical sections of the Specifications.

### 1.07 WARRANTIES AND GUARANTEES

- A. Manufacturer's warranties and guarantees notwithstanding, warrant entire Work against defects in materials and workmanship for twelve (12) months from date of Substantial Completion. Warranties and guarantees between CONTRACTOR and manufacturers and CONTRACTOR and suppliers shall not affect warranties or guarantees between CONTRACTOR and DISTRICT.
- B. Execute and assemble documents from subcontractors, suppliers and manufacturers.
- C. Submit prior to final Application for Payment.
- D. For items of Work delayed beyond date of Substantial Completion, provide updated submittal within ten (10) days after acceptance, listing date of acceptance as start of warranty period.

## 1.08 SPARE PARTS AND MAINTENANCE MATERIALS

- A. Provide products, spare parts, maintenance and extra materials in quantities specified in individual specification Sections.
- B. Deliver to project site location as directed by DISTRICT.

### 1.09 FINAL CLEANING

- A. Final cleaning is provided by Contractor.
- B. Each CONTRACTOR shall leave his finished work in clean condition, including following as applicable:
  - 1. Remove labels that are not permanent labels.
  - 2. Clean transparent materials, including mirrors and glass in doors and windows. Remove glazing compounds and other substances that are

- noticeable vision-obscuring materials. Replace chipped or broken glass and other damaged transparent materials.
- Clean exposed exterior and interior hard-surfaced finishes to a dust-free condition, free of stains, films, and similar foreign substances. Restore reflective surfaces to their original condition. Leave concrete floors broom clean. Vacuum carpeted surfaces.
- 4. Wipe surfaces of mechanical and electrical equipment. Remove excess lubrication and other substances. Clean plumbing fixtures to a sanitary condition. Clean light fixtures and lamps.

#### **SECTION 01 78 36 - WARRANTIES AND BONDS**

#### PART 1 GENERAL

# 1.01 SECTION INCLUDES

- A. Preparation and submittal of warranties and bonds.
- B. Time and schedule of submittals.

#### 1.02 RELATED SECTIONS

- A. Contract Closeout Procedures.
- B. Product Requirements
- C. Materials and Equipment
- D. Technical Specifications Sections: Warranties required for specific products or Work.

#### 1.03 WARRANTY REQUIREMENTS

- A. Warranties or bonds shall provide for replacement or reconstruction of failed or defective Work to an acceptable condition complying with the requirements of the Contract Documents. Work shall be restored at no cost to the District regardless of whether the District has benefited from use of the Work for a portion of its anticipated useful service life.
- B. Provide warranties and bonds, executed in duplicate by responsible Subcontractors, suppliers, and manufacturers, within ten days after completion of the applicable item or work.
- C. When a designated portion of the Work is partially used and/or occupied by the DISTRICT, submit properly executed warranties within ten (10) days of the Partial Use or Occupancy of the designated portion of the Work
- D. Verify that documents are in proper form, contain full information and are notarized.
- E. DISTRICT Recourse: Expressed warranties made to DISTRICT are in addition to implied warranties and shall not limit the duties, obligations, rights, and remedies otherwise available under the law. Expressed warranty periods shall not be interpreted as limitations on the time in which DISTRICT can enforce such other duties, obligations, rights, or remedies.

## 1.04 FORM OF SUBMITTALS

- A. Prepare duplicate binders, commercial quality, 8-1/2 x 11 inch, three-ring side binders with hardback, cleanable, plastic covers.
- B. Label cover and spine of each binder with typed or printed title WARRANTIES AND BONDS, with title of Project. Number separate volumes in order.

- C. Table of Contents: Typed, in the sequence of the Table of Contents of the Project Manual, with each item identified with the number and title of the specification Section in which specified and the name of the product or work item.
- D. Separate each warranty or bond with index tab sheets keyed to the Table of Contents listing. Provide full information, using separate typed sheets as necessary. Use paper of durable, long-lasting quality. List Subcontractor, supplier, and manufacturer, with name, address and telephone number of responsible principal.

#### 1.05 TIME OF SUBMITTALS

- A. Except for specifically authorized exceptions, the date for beginning the period of warranty shall be the Date of Substantial Completion.
- B. For equipment or component parts of equipment put into service during construction with District's permission, submit documents within ten (10) days after acceptance.
- C. Make other submittals within ten (10) days after Date of Substantial Completion prior to final Application for Payment.
- D. For items of Work when acceptance is delayed beyond Date of Substantial Completion, submit within ten (10) days after acceptance, listing the date of acceptance as the beginning of the warranty period.

**END OF SECTION** 

# **GUARANTEE**

	the requirements	of the Contract Doc	, which we have installed for SAN ROJECT NAME has been performed uments and that the work as installed	
defective in workm displaced in conne individual trade spe the above-mention	anship or material ction with such rescribing to the control of th	I together with any of placement within a made re stringent requirem	such work that may prove to be ther adjacent work which may be ninimum period of ONE (1) YEAR (see nents) from the date of acceptance of TY UNIFIED SCHOOL DISTRICT, septed.	
reasonable period after being notified	of time, as determ in writing by the Depaired and made	nined by the District, District, the undersigr e good at the expens	ne above mentioned conditions within but not later than ten (10) working day ned authorizes the District to proceed se of the undersigned, who will pay the	ys I to
		DDIME CONTRACT		
		PRIME CONTRACT	OR	
		SIGNED:		
		NAME		
Representatives to	be contacted for	service subject to ter	rms of contract:	
NAME:				
ADDRESS:				
PHONE #:				

# CONTRACTOR'S CERTIFICATE REGARDING ASBESTOS MATERIAL

nis form is to be submitted at the time final billing is provided.	
"I certify that all the materials and supplies installed under this	
(Name of Contract)	
contract are free of asbestos-containing materials."	
ate	
fficial Name of Contractor	
<i>(</i>	
tle	
ue	
gnature	

**END OF SECTION** 

#### **SECTION 01 78 39 - PROJECT RECORD DOCUMENTS**

#### PART 1 GENERAL

#### 1.07 SECTION INCLUDES

A. This Section includes administrative and procedural requirements for preparing, maintaining, and submitting Project Record Documents.

#### 1.08 RELATED SECTIONS

- A. Price and Payment Procedures
- B. Submittals
- C. Closeout Procedures
- D. Field Engineering

#### 1.09 PROJECT RECORD DOCUMENTS

- A. CONTRACTOR shall prepare and maintain record documents throughout the course of construction, as specified herein.
- B. Provide access to record documents for ARCHITECT, IOR and DISTRICT REPRESENTATIVE reference during normal working hours.
- Do not use project record documents for construction purposes. Protect record documents from deterioration and loss.
- D. Record in concise and neat manner, concurrent with construction progress, and at least on a weekly basis, all actual revisions to the work:
  - 1. Changes made on the Drawings, including Clarification Drawings.
  - 2. Changes made to the Specifications.
  - 3. Changes made by Addenda.
  - 4. Changes made by Instruction Bulletins.
  - 5. Change Orders or other authorized Modifications to the Contract.
  - 6. Revisions made to shop drawings, product data and samples.
- E Record Drawings shall be a clean, clear electronic files of Drawings and Shop Drawings. File type shall be determined by DISTRICT. Mark the set with red erasable pencil to show the actual installation where the installation varies substantially from the Work as originally shown. Indicate which Drawing is most capable of showing conditions fully and accurately. Where Shop Drawings are used, record a cross-reference at the corresponding location on the Drawings. Provide detailed and accurate field dimensions for concealed elements that would be difficult to measure and record at a later date.
  - 1. Mark new information, including details, that is important to DISTRICT but was not shown on Drawings or Shop Drawings.

- 2. Show measured depths of foundations in relation to finish first floor datum.
- 3. Show measured horizontal and vertical locations of underground utilities and appurtenances, referenced to permanent surface improvements. Identify drains and sewers by invert elevation.
- 4. Verify surveyor's Record Drawings with CONTRACTOR'S utilities locations and depths markups.
- Show measured locations of internal utilities and appurtenances concealed in construction, referenced to visible and accessible features of the Work. Identify ducts, dampers, valves, access doors and control equipment wiring.
- 6. Show field changes of dimension and detail.
- 7. Note related Change Order or Construction Directive numbers on each affected sheet.
- 8. Organize Record Drawing sheets into manageable sets. Bind sets with durable-paper cover sheets; print suitable titles, dates, and other identification on the cover of each set.
- F. Record Specifications: Maintain a complete copy of the Specifications, including Addenda, Change Orders and Construction Directives issued during construction. Legibly mark at each Section description of actual products installed if different from that specified, including:
  - 1. Manufacturer's name, trade name, product model and number and supplier.
  - 2. Authorized product substitutions or alternates utilized.
  - 3. Changes made by Addenda and Modifications.
- G. <u>Record Product Data</u>: Maintain a copy of each Product Data submittal. Note related Change Orders and Construction Directives and mark-up of record drawings and Specifications.
  - Mark these documents to illustrate significant variations in actual Work performed in comparison with information submitted. Include variations in products delivered to the Project site and from the manufacturer's installation instructions and recommendations.
  - Provide detailed and accurate information regarding concealed products and portions of Work that cannot otherwise be readily discerned later by direct observation.
- H. <u>Record Samples</u>: Immediately prior to Substantial Completion, CONTRACTOR shall meet with ARCHITECT and DISTRICT at the Project site to determine

- which Samples are to be transmitted to DISTRICT for record purposes. Comply with DISTRICT instructions regarding delivery to DISTRICT storage area.
- I. <u>Miscellaneous Records</u>: Refer to other Specification sections for requirements of miscellaneous record keeping and submittals in connection with actual performance of the Work. Immediately prior to the date of Final Completion, complete and compile miscellaneous records and place in good order. Identify miscellaneous records properly and bind or file, ready for continued use and reference. Submit to ARCHITECT for DISTRICT records.

**END OF SECTION** 

# **General Conditions**

Pacific High School
Career Technical Education
Transportation Technologies Building and
Campus Modernization Phase I

ARTICLE 1	DEFINITIONS	4
1.1	BASIC DEFINITIONS	4
1.2	EXECUTION, CORRELATION AND INTENT	10
1.3	OWNERSHIP AND USE OF ARCHITECT'S DRAWINGS, SPECIFICATIONS	
	AND OTHER DOCUMENTS	14
ARTICLE 2	DISTRICT	15
2.1	INFORMATION AND SERVICES REQUIRED OF THE DISTRICT	15
2.1	DISTRICT'S RIGHT TO CARRY OUT THE WORK DUE TO PARTIAL	13
2.2	DEFAULT IN A SPECIFIC SEGREGATED AREA OF WORK (48 HOUR	
	NOTICE TO CURE AND CORRECT)	18
ARTICLE 3	THE CONTRACTOR	20
3.1	SUPERVISION AND CONSTRUCTION PROCEDURES	
3.2	SUPERVISIONLABOR AND MATERIALS	
3.3 3.4	WARRANTY	
3.4 3.5	TAXES	
3.6	PERMITS, FEES AND NOTICES	
3.7	SUBMITTALS REQUIRED AT THE COMMENCEMENT OF THE PROJECT	
3.8	DOCUMENTS, SAMPLES, AND COMPUTER AT THE SITE	
3.9	SUBMITTALS INCLUDING SHOP DRAWINGS, PRODUCT DATA, AND	
	SAMPLES	29
3.10	SUBSTITUTIONS	
3.11	INTEGRATION OF WORK	
3.12	CLEANING UP	
3.13	ACCESS TO WORK	
3.14	ROYALTIES AND PATENTS	
3.15	INDEMNIFICATION	
3.16	SUBMISSION OF DAILY REPORTS	
3.17 3.18	AS-BUILT DRAWINGS AND ANNOTATED SPECIFICATIONS EQUIPMENT MANUALS	
3.18	DIR REGISTRATION	
3.19	DIK REGISTRATION	42
ARTICLE 4	ADMINISTRATION OF THE CONTRACT AND CLAIMS	43
4.1	ARCHITECT	43
4.2	ARCHITECT'S ADMINISTRATION OF THE CONTRACT	43
4.3	PROJECT INSPECTOR	45
4.4	STOP WORK ORDER	47
4.5	RESPONSIBILITY FOR ADDITIONAL CHARGES INCURRED BY THE	
	DISTRICT FOR PROFESSIONAL SERVICES	
4.6	DISPUTES AND CLAIMS	48
ARTICLE 5	SUBCONTRACTORS	57

5.1	DEFINITIONS	57
ARTICLE 6	CONSTRUCTION BY DISTRICT OR BY SEPARATE CONTRACTORS	59
6.1	DISTRICT'S RIGHT TO PERFORM CONSTRUCTION AND TO AWARD	
0.1	SEPARATE CONTRACTS	59
6.2	CONSTRUCTIVE OWNERSHIP OF PROJECT SITE AND MATERIAL	61
6.3	DISTRICT'S RIGHT TO CLEAN UP	
ARTICLE 7	CHANGES IN THE WORK	62
7.1	CHANGES	62
7.2	CHANGE ORDERS ("CO")	63
7.3	CONSTRUCTION CHANGE DOCUMENT (CCD Category A, and CCD	
	Category B) and IMMEDIATE CHANGE DIRECTIVE (ICD)	63
7.4	REQUEST FOR INFORMATION ("RFI")	65
7.5	REQUEST FOR PROPOSAL ("RFP")	65
7.6	CHANGE ORDER REQUEST ("COR")	66
7.7	COST OF CHANGE ORDERS	66
ADTICLE 0	TIME AND COHEDINE	72
ARTICLE 8	TIME AND SCHEDULE	
8.1	DEFINITIONS	
8.2	HOURS OF WORK	
8.3	PROGRESS AND COMPLETION	
8.4	EXTENSIONS OF TIME - LIQUIDATED DAMAGES	79
ARTICLE 9	PAYMENTS AND COMPLETION	83
9.1	CONTRACT SUM	83
9.2	COST BREAKDOWN	
9.3	PROGRESS PAYMENTS	84
9.4	APPLICATIONS FOR PROGRESS PAYMENTS	
9.5	STOP NOTICE CLAIMS AND WARRANTY OF TITLE	88
9.6	DECISIONS TO WITHHOLD PAYMENT	89
9.7	NONCONFORMING WORK	90
9.8	SUBCONTRACTOR PAYMENTS	91
9.9	COMPLETION OF THE WORK	91
9.10	PARTIAL OCCUPANCY OR USE	
9.11	COMPLETION AND FINAL PAYMENT	
9.12	SUBSTITUTION OF SECURITIES	100
ARTICLE 10	PROTECTION OF PERSONS AND PROPERTY	101
10.1	SAFETY PRECAUTIONS AND PROGRAMS	
10.1	SAFETY OF PERSONS AND PROPERTY	
10.2	EMERGENCIES	
10.3	HAZARDOUS MATERIALS	
10.4	TIAZANDOUS WATENIALS	100
ARTICLE 11	INSURANCE AND BONDS	
11.1	CONTRACTOR'S LIABILITY INSURANCE	
11.2	WORKERS' COMPENSATION INSURANCE	110
11.3	SEXUAL ABUSE/MOLESTATION	110

11.4	BUILDER'S RISK/ "ALL RISK" INSURANCE	110
11.5	FIRE INSURANCE	111
11.6	AUTOMOBILE LIABILITY	
11.7	OTHER INSURANCE	
11.8	PROOF OF INSURANCE	
11.9	COMPLIANCE	
11.10	WAIVER OF SUBROGATION	
11.11	PERFORMANCE AND PAYMENT BONDS	113
ARTICLE 12	UNCOVERING AND CORRECTION OF WORK	114
12.1	COMPLIANCE WITH TITLE 24 INSTALLATION REQUIREMENTS	114
12.2	SPECIAL NOTICE OF AMERICAN'S WITH DISABILITIES ACT	114
12.3	UNCOVERING OF WORK	
12.4	CORRECTION OF WORK	115
ARTICLE 13	MISCELLANEOUS PROVISIONS	117
13.1	GOVERNING LAW	
13.2	SUCCESSORS AND ASSIGNS	
13.3	WRITTEN NOTICE	
13.4	RIGHTS AND REMEDIES	
13.5	TESTS AND INSPECTIONS	
13.6	TRENCH EXCAVATION	
13.7	WAGE RATES, TRAVEL, AND SUBSISTENCE	
13.8	RECORDS OF WAGES PAID	
13.9	APPRENTICESASSIGNMENT OF ANTITRUST CLAIMS	
13.10	STATE AND DISTRICT CONDUCTED AUDITS	
13.11 13.12	STORM WATER POLLUTION PREVENTION	
13.12	STORM WATER POLLUTION PREVENTION	120
ARTICLE 14	TERMINATION OR SUSPENSION OF THE CONTRACT	130
14.1	TERMINATION BY THE CONTRACTOR FOR CAUSE	
14.2	TERMINATION BY THE DISTRICT FOR CAUSE	130
14.3	TERMINATION OF CONTRACT BY DISTRICT (CONTRACTOR NOT AT	
	FAULT)	131
14.4	REMEDIES OTHER THAN TERMINATION	132
ARTICLE 15	DEBARMENT	134
15.1	DEBARMENT MEANS THERE HAS BEEN A FINDING THAT THE	
	CONTRACTOR IS NOT RESPONSIBLE.	
15.2	BOARD FINDING	
15.3	HEARING AND PRESENTATION OF EVIDENCE	134
SUPPLEMEN	TARY GENERAL CONDITIONS	138

# ARTICLE 1 DEFINITIONS

# 1.1 BASIC DEFINITIONS

<u>NOTE:</u> The following shall not be construed as a comprehensive list of all definitions in the Contract Documents and there may be other definitions set forth in the Contract Documents. Additionally, any references to any DSA forms, documents or requirements shall be construed to incorporate any updates, supplements, or additions. The Contractor shall be required to meet the latest DSA requirements applicable to the Project.

- 1.1.1 <u>Action of the Governing Board</u> is a vote of a majority of the District's Governing Board.
- 1.1.2 <u>Approval</u> means written authorization through action of the Governing Board. The Governing board has delegated to the Assistant Superintendent the authority to approve certain modifications, Change Orders or Immediate Change Directives (Subject to the limits of the Delegation of Authority provided by the Board). In no case shall the Assistant Superintendent have authority to approve total Change Orders or Modifications to the Project exceeding 10% of the Contract Sum.
- 1.1.3 <u>Architect</u> means the architect, engineer, or other design professional engaged by the District to design and perform general observation of the work of construction and interpret the Drawings and Specifications for the Project. (See ARTICLE 4)
- 1.1.4 <u>As-Builts</u> are a set of Plans and Specifications maintained by the Contractor clearly showing all changes, revisions, substitutions, field changes, final locations, and other significant features of the Project. The As-Builts shall be maintained continuously throughout the Work for the Project and is both a prerequisite to the issuance of Payment Application and a requirement for Contract Close-Out. (See Article 3.17)
- 1.1.5 <u>Beneficial Occupancy</u> is the point in time when a building or buildings are fit for occupancy is fit for occupancy and its intended use. Basic requirements are the building is safe, at or near Substantial Completion, and all fire/life safety items are approved and operational. The fact that a building is occupied does not mean that the building is ready for Beneficial Occupancy if there are elements that are unsafe or if fire/life safety items are not approved and operational. Taking occupancy on a structure that is under a fire watch is not considered beneficial occupancy. Further, taking of Beneficial Occupancy is not a point in time when retention is due unless the entire school has obtained a Certificate of Substantial Completion that meets the definition of 1.1.46.
- 1.1.6 <u>Claims.</u> A Claim is a request for payment, supported by back-up documentation which includes, invoices time sheets, or other documents substantiating legitimacy or entitlement that is submitted during the Project or immediately following the Project made prior to the Final Retention Payment Application and prior to Final Completion of the Project. A "Claim" means a separate demand by the Contractor for (1) time extension, (2) payment of money or damages arising from Work done by or on behalf of the Contractor pursuant to the CONTRACT and payment of which is not otherwise expressly provided for or the claimant is not otherwise entitled to, or (3) and amount the payment of which is disputed by the District. (See Article 4.6)

- 1.1.7 <u>Change Order (CO).</u> A CO is a written instrument prepared by the Architect and signed by the District (as authorized by the District's Governing Board), the Contractor, and the Architect, stating their agreement upon (1) A description of a change in the Work, (2) The amount of the adjustment in the Contract Sum, if any; and (3) The extent of the adjustment in the Contract Time, if any. (See Article 7.2)
- 1.1.8 <u>Change Order Request (COR).</u> A COR is a written request supported by backup documentation prepared by the Contractor requesting that the District and the Architect issue a CO based upon a proposed change, or a change that results in an adjustment in cost, time or both, or arising from an RFP, CCD or ICD. (See Article 7.6)
- 1.1.9 <u>Close-Out</u> means the process for Final Completion of the Project, but also includes the requirements for the DSA Certification that the Project is Complete (See DSA Certification Guide). (See Article 9.9)
- 1.1.10 Construction Change Document (CCD). A Construction Change Document is a DSA term that is utilized to address changes to the DSA approved Plans and Specifications. There are two types of Construction Change Documents. (1) DSA approved CCD Category A for work affecting structural, access or fire/ life safety of the Project which will require a DSA approval; and, (2) CCD Category B for work NOT affecting structural safety, access compliance or fire/ life safety that will not require a DSA approval (except to confirm that no approval is required). Both CCD Category A and Category B shall be set forth in DSA Form 140 and submitted to DSA as required. (See Article 7.3)
- 1.1.11 <u>Complete/ Completion / Final Completion</u> means that all Work in the Contract Documents is finished, the requirements of the Contract Documents have been met, the Project has been Closed Out, and all Work has ceased on the Project. This may also be referred to as Final Completion. In most cases, the recording of a Notice of Completion shall represent Completion of the Project. Beneficial Occupancy does not mean the Work is Complete.
- 1.1.12 <u>Completion Date</u> is the date when all Work for the Project shall be Substantially Complete and is the date assigned at the end of the Contract Time for the Project. (See Article 1.1.46)
- 1.1.13 <u>Construction Manager.</u> The Construction Manager is a consultant to the District contracted to assist in Project planning, management and construction of the Project. If there is a Construction Manager, they may assist in various aspects of the Project including, but not limited to Monitoring the progress of the construction, reviewing and monitoring the schedule, progress of work, monitoring pay requests, facilitating communications, advising the District and its Board of Education on various aspects of the construction process, monitoring the RFI, COR, CCD, ICD, RFP, Claims, Disputes and other Project related processes.
- 1.1.14 <u>Contract or Agreement</u> when the terms are used in these General Conditions shall be references to the Contract Documents as defined herein.
- 1.1.15 Contract Documents (sometimes referred to as Construction Documents) consist of the Agreement between District and Contractor (hereinafter the Agreement or Contract), Conditions of the Contract (General, Supplementary and other Conditions), Drawings, Specifications, Addenda issued prior to bid, instructions to bidders, notice to bidders, and the requirements contained in the Bid Documents, other documents listed in the Agreement, and Modifications issued after execution of the Contract. A Modification is a written amendment to the Contract signed by parties, a Change Order, a Construction Change Document, or a written order for a minor change in the Work issued by the Architect. The Contract

Documents collectively form the Contract. The Contract represents the entire and integrated Agreement between the parties hereto and supersedes prior negotiations, representations, or agreements, either written or oral. The Contract may be amended or modified only by a written Modification. The Contract Documents shall not be construed to create a contractual relationship of any kind between the Architect and Contractor, between the District and any Subcontractor or Sub-subcontractor, or between any persons or entities other than the District and the Contractor. The Architect shall, however, be entitled to performance and enforcement of obligations under the Contract intended to facilitate performance of the Architect's duties.

- 1.1.16 <u>Contract Time</u> is the time period specified in the Contract Documents in which the Project shall be completed. This is sometimes referred to a Contract Duration, or "time in which the Contractor has to complete the Project". (See Article 8.1.1)
- 1.1.17 <u>Contractor, District, and Architect</u> are those mentioned as such in the Agreement. They are treated throughout the Contract Documents as if they are of singular number and neuter gender. Any reference to "Owner" shall mean "District" or San Bernardino City Unified School District.
- 1.1.18 <u>Cure</u> is the act of remedying a material failure to perform under the terms of the Contract Documents during the time provided to correct Contractor's Default. Specific time periods are provided to Cure and Correct a Contractor Default under Article 14 and for a Partial Default under Article 2.2 as well as elsewhere in the Contract Documents.
  - 1.1.19 <u>Days</u> mean calendar days unless otherwise specifically stated.
- 1.1.20 <u>Default</u> is a material breach of Contract. A Termination for Cause under Article 14 is a declaration of Default of the Contract and shall act as a demand upon the Surety to perform under the terms of the Performance Bond. Partial Defaults may also be tendered to the Surety at District's discretion. (See Article 2.2)
- 1.1.21 <u>Dispute.</u> A dispute is a disagreement on terms or conditions of the Project where the Contractor's opinion of the Project, Payment, Change Order or Request for Proposal differs from that of the District or Architect. A dispute only rises to the level of a claim once the dispute is assembled with back-up documentation and presented for evaluation. (See Article 4.6)
- 1.1.22 <u>District Representative</u> is the person designated by the District to represent the District during the Construction for the Project. This District Representative shall have the delegated authority as further defined in Article 1.1.2. This District Representative may be an employee of the District who may have the delegated authority as set forth in Article 1.1.3, and may also include Construction Managers. In some cases, the District and its Board may be assisted by a Construction Manager. When a Construction Manager is assisting the District, the Contractor, Architect, and Inspector shall have a primary contact with the District's Construction Manager who will advise the District.
- 1.1.23 <u>Drawings/Plans</u> are graphic and pictorial portions of the Contract Documents prepared for the Project and approved changes thereto, wherever located and whenever issued, showing the design, location, and scope of the Work, generally including Plans, elevations, sections, details, schedules, and diagrams as drawn or approved by the Architect. Sometimes Drawings will also be included in Addenda, Change Orders, and Specifications.

- 1.1.24 <u>DSA</u> is the <u>Division of State Architect</u>. DSA is the agency that provides design and construction oversight for K-12 Schools, Community Colleges, and State Funded Charter School Projects. DSA is the responsible agency for this Project and Contractor has submitted a bid for the Project since Contractor is familiar with Contractor's responsibilities under the DSA requirements more thoroughly set forth at Title 24 of the California Code of Regulations. Contractor agrees to abide by the jurisdiction of DSA and shall construct the Project to conform with the approved Plans, Specifications, Addenda, and Change Orders (inclusive of approved CCD's and ICD's issued by the District pending CCD approval). See DSA website.
- 1.1.25 <u>Emergency</u> shall be defined as a sudden, unexpected occurrence, involving a clear and imminent threat to the continuation of school classes, a critical path delay that will result in not being able to occupy the school when students arrive to use the facility, danger from the facility or from outside the facility, Act of God, or other action which requires immediate action to prevent or mitigate loss of, or damage to, life, health, property, or essential public services.
- 1.1.26 <u>Float</u> the total number of days an activity may be extended or delayed without delaying the Completion Date shown in the schedule. Float will fall into three categories: (1) Rain Days; (2) Governmental Delays; and, (3) Project Float. (See Article 8.1.4)
- 1.1.27 <u>Immediate Change Directive. (ICD)</u> A written order prepared by the Architect and signed by the District and the Architect, directing a change in the Work where the Work must proceed immediately and stating a proposed basis for adjustment, if any, in the Contract Sum or Contract Time, or both. (See Article 7.3)
- 1.1.28 <u>Inspector of Record (IOR)/ Project Inspector (PI)</u> is the individual retained by the District in accordance with Title 24 of the California Code of Regulations and who will be assigned to the Project
- 1.1.29 <u>Notice of Non-Compliance (DSA Form 154)</u> is a document issued by the Inspector if there is a deviation from the DSA approved Plans, Specifications, and Change Orders. (See Article 7.1.2)
- 1.1.30 <u>Payment Application or Certificate of Payment</u> is the Contractor's certified representation of the actual level of Work performed on the Project. Payment Applications are sometimes also called "Certificate of Payment", "Request for Payment", "Payment Application", or similar terms, and shall follow the Schedule of Values that are approved by the Architect, Inspector and District. (See Article 9.3)
- 1.1.31 <u>Project</u> is the complete construction of the Work performed in accordance with the Contract Documents.
- 1.1.32 <u>Project Manual</u> is the volume assembled for the Work which may include, without limitation, the bidding requirements, sample forms, Conditions of the Contract, and Specifications.
  - 1.1.33 <u>Provide</u> shall include "provide complete in place," that is "furnish and install complete."
- 1.1.34 <u>Punch List/ Punch Item / Incomplete Punch Item</u> is a list of minor repair items, prepared after the issuance of a Certificate of Substantial Completion, by the Inspector and Architect of Work required in order to complete the Contract Documents and ensure compliance with the DSA Approved

Plans so the Project may be Closed Out. Issuance of the Retention Payment is dependent of the proper completion of the Punch List. (See Article 9.9)

- 1.1.34.1 *Contractor's List of Punch Items* is a list of minor repair items the Contractor submits when the Contractor considers the Work Substantially Complete. Submission of this List of Incomplete Punch Items is the Contractor's representation that the Project is Substantially Complete. (See Article 9.9.1.1)
- 1.1.35 Request for Information (RFI) is a written request prepared by the Contractor requesting the Architect to provide additional information necessary to clarify or amplify an item which the Contractor believes is not clearly shown or called for in the Drawings or Specifications, or to address problems which have arisen under field conditions. (See Article 7.4)
- 1.1.36 <u>Request for Proposal (RFP)</u> is a written request prepared by the Architect (and/or CM) requesting the Contractor to submit to an estimate of the effect of a proposed change on the Contract Price and (if applicable) the Contract Time. (See Article 7.5)
- 1.1.37 <u>Safety Orders</u> are those issued by any city, county, state or federal agency having jurisdiction over the Project.
- 1.1.38 <u>Schedule</u> is the Contractor's view of the practical way in which the Work will be accomplished. In this Agreement there is a requirement for a Baseline Schedule and regular Schedule Updates that show all Work to be completed during the Contract Time and shall include all items listed under Article 8.3.2.9. See Article 8 of the General Conditions.
- 1.1.39 <u>Schedule of Values</u> is a detailed breakdown of the Contract Price for each Project, building, Phase of Work or Site as determined by the District. This Schedule of Values shall adequately detail the price for the Work so Progress Payments Applications can be meaningfully reviewed by the Inspector, Architect of Record, Engineer of Record, and District. (See Article 9.2)
- 1.1.40 <u>Separate Contracts</u> are Contracts that the District may have with other Contractors, vendors, suppliers, or entities to perform Work on the Project. This may include, but is not limited to Multi-Prime Trade Contractors, furniture installers, testing agencies, clean-up contractors, or network or low voltage contractors. Contractor shall plan for certain other contractors that may also be working on the Project site and address these other contractors in Contractor's Schedule. (See Article 6)
- 1.1.41 <u>Site</u> refers to the grounds of the Project as defined in the Contract Documents and such adjacent lands as may be directly affected by the performance of the Work.
- 1.1.42 <u>Specifications</u> are that portion of the Contract Documents consisting of the written requirements for material, equipment, construction systems, instructions, quality assurance standards, workmanship, and performance of related services.
- 1.1.43 <u>Standards, Rules, and Regulations</u> referred to are recognized printed standards and shall be considered as one and a part of these Specifications within limits specified. Federal, state and local regulations are incorporated into the Contract Documents by reference.
- 1.1.44 <u>Stop Work Order, or an Order to Comply</u>, is issued when either (1) the Work proceeds without DSA approval; (2) the Work proceeds without a DSA Inspector of Record, or (3) where DSA

determines that the Work is not being performed in accordance with applicable rules and regulations, and would compromise the structural integrity of the Project or would endanger lives. If a Stop Work Order is issued, the Work in the affected area shall cease until DSA withdraws the Stop Work Order. Pursuant to Education Code section 17307.5(b), the District shall not be held liable in any action filed against the District for any delays caused by compliance with the Stop Work Order

- 1.1.45 <u>Subcontractor</u>, as used herein, includes those having direct or indirect contracts with Contractor and ones who furnished labor, material or services for a special design according to Plans, Drawings, and Specifications of this Work.
- 1.1.46 <u>Substantial Completion/ Substantially Complete(d)</u> is not reached unless and until each of the following four (4) conditions have been met: (1) all contractually required items have been installed with the exception of only minor and Incomplete Punch List Items (See Article 9.9.1.2); (2) All Fire/Life Safety Systems have been installed, and are working and signed off on the DSA Form 152 Inspection Card, and all building systems including mechanical, electrical and plumbing are all functioning; (3) all other items DSA Form 152 Inspection Card for the Project have been approved and signed off; and (4) the Project is fit for occupancy and its intended use. For the purposes of this Contract, any references to Completion Date means Substantial Completion Date.
- 1.1.47 <u>Substitution</u> is a change in product, material, equipment, or method of construction from those required by the Construction Documents proposed by the Contractor. For this Project, a Substitution is subject to the filing of a Construction Substitution Request Form at the time of bid and meeting the requirements of Article 3.10.
- 1.1.48 <u>Supplementary Conditions/ Supplementary General Conditions/ Special Conditions</u> are terms that are sometimes used interchangeably and refer to any additional requirements or changes to the General Conditions as noted.
- 1.1.49 <u>Surety</u> is the person, firm, or corporation that executes as a bid bond, Payment Bond or Performance Bond guarantor on the Contractor's Bid, Contractor's Performance on the Contract and Payment of the Contractor's Subcontractors, material suppliers, vendors and labor on the Project. The Surety is bound to the same extent as the Contractor is bound once a Default occurs. A default includes a Termination for Substantial Failure to Perform under Article 14, but also includes any breach of Contract and is subject to the requirements and responsibilities as set forth in the Performance Bond.
- 1.1.50 <u>Work</u> shall include all labor, materials, services and equipment necessary for the Contractor to fulfill all of its obligations pursuant to the Contract Documents. It shall include the initial obligation of any Contractor or Subcontractor who performs any portion of the Work, to visit the Site of the proposed Work (a continuing obligation after the commencement of the Work), to fully acquaint and familiarize itself with the conditions as they exist and the character of the operations to be carried out under the Contract Documents, and make such investigation as it may see fit so that it shall fully understand the facilities, physical conditions, and restrictions attending the Work under the Contract Documents. Each such Contractor and its Subcontractors shall also thoroughly examine and become familiar with the Drawings, Specifications, and associated Contract Documents and bid documents before preparing and submitting any bid.
  - 1.1.51 Workers include laborers, workers, and mechanics.

# 1.2 EXECUTION, CORRELATION AND INTENT

#### 1.2.1 Correlation and Intent

- 1.2.1.1 Documents Complementary and Inclusive. The Contract Documents are complementary and are intended to include all items required for the proper execution and completion of the Work. All Contract Documents form the Contractor's Contract with the District. Any item of Work mentioned in the Specifications and not shown on the Drawings, or shown on the Drawings and not mentioned in the Specifications, shall be provided by Contractor as if shown or mentioned in both. The Contractor is bound to provide the Work complete and is under a legal duty to carefully study Plans and schedule operations well ahead of time and identify inconsistencies with the Plans and Specifications and call such inconsistencies to the attention of the Architect or Registered Engineer through the Inspector under Section 4-343(b) of Title 24.
- 1.2.1.2 Work to be Complete. Contractor has thoroughly studied the Contract Documents and understands that the District contracted with Contractor to provide a complete Project which means complete systems and buildings. The entire set of Contract Documents shows a complete Project and Contractor agrees that there are multiple disciplines putting together a set of Contract Documents. Thus, if portions of a system are shown on some Drawings and not others, this does not mean the Contractor is to only provide part of a system. For example, if an air conditioning unit is shown on the mechanical Drawings, the plumbing for the air conditioning is shown on another Drawing, and the electrical shown on the electrical Drawings, the Contractor is to provide a complete and working air conditioning system. The only time when an item is supplied incomplete is if the system is shown specifically as incomplete since others will be completing the system. Work includes, but is not limited to materials, workmanship, and manufacture of fabrication of components for the Project.
- 1.2.1.3 Coverage of the Drawings and Specifications. The Drawings and Specifications generally describe the Work to be performed by Contractor. Generally, the Specifications describe Work which cannot be readily indicated on the Drawings and indicate types, qualities, and methods of installation of the various materials and equipment required for the Work. It is not intended to mention every item of Work in the Specifications, which can be adequately shown on the Drawings, or to show on the Drawings all items of Work described or required by the Specifications even if they are of such nature that they could have been shown. All materials or labor for Work, which is shown on either the Drawings or the Specifications (or is reasonably inferable therefrom as being necessary to complete the Work), shall be provided by the Contractor. The Contractor is responsible for the whole Project as contractually set forth as the Contract Documents. It is intended that the Work be of sound, quality construction, and the Contractor shall be responsible for the inclusion of adequate amounts to cover installation of all items indicated, described, or implied in the portion of the Work to be performed by them.
- 1.2.1.4 *Conflicts*. In the event there is a discrepancy between the various Contract Documents, it is intended that the more stringent, higher quality, and greater quantity of Work shall apply.
- 1.2.1.5 Conformance with Laws. Each and every provision of law required by law to be inserted in this Contract shall be deemed to be inserted herein, and the Contract shall be read and enforced as though it were included herein, even if through mistake or otherwise any such provision is not inserted, or is not correctly inserted.

Before commencing any portion of the Work, Contractor shall check and review the Drawings and Specifications for such portion for conformance and compliance with all laws, ordinances, codes, rules and regulations of all governmental authorities and public and municipal utilities affecting the construction and operation of the physical plant of the Project, all quasi-governmental and other regulations affecting the construction and operation of the physical plant of the Project, and other special requirements, if any, designated in the Contract Documents. Such checking shall include review of Title 24 of the California Code of Regulations, California Building Code, local utility, local water connection, local grading and all other applicable agencies. In the event Contractor observes any violation of any law, ordinance, code, rule or regulation, or inconsistency with the Contract Documents, Contractor shall, within five (5) days, notify the Inspector, Architect and District in writing of same and shall ensure that any such violation or inconsistency shall be corrected in the manner provided hereunder prior to the construction of that portion of the Project. (See Title 24 Section 4-343)

The Contractor shall bear all expenses of correcting Work done contrary to said laws, ordinances, rules, and regulations if the Contractor performed same (1) without first consulting the Architect for further instructions regarding said Work or (2) disregarded the Architect's instructions regarding said Work.

- 1.2.1.6 Ambiguity and Inconsistency. Before commencing any portion of the Work, Contractor shall carefully examine all Drawings and Specifications and other information given to Contractor as to materials and methods of construction and other Project requirements. Prior to commencing any portion of the Work, Contractor shall notify Architect and District in writing of any perceived or alleged error, inconsistency, conflict, ambiguity, or lack of detail or explanation in the Drawings and Specifications in the manner provided herein. If the Contractor or its Subcontractors, material or equipment suppliers, or any of their officers, agents, and employees performs, permits, or causes the performance of any Work under the Contract Documents, which it knows or should have known to be in error, inconsistent, or ambiguous, or not sufficiently detailed or explained, Contractor shall bear any and all costs arising therefrom including, without limitation, the cost of correction thereof without increase or adjustment to the Contract Price or the time for performance. Contractor shall maintain an adequate inspection system and perform personal observations and review work and pre-plan the project to ensure the Work performed under the Contract conforms to Contract requirements. Contractor shall maintain records of such review and observation to ensure strict compliance with the terms of the Contract.
- 1.2.1.7 *Typical Parts and Sections*. Whenever typical parts or sections of the Work are completely detailed on the Drawings, and other parts or sections which are of the same construction are shown in outline only, the complete or more detailed shall apply to the Work which is shown in outline.
- 1.2.1.8 *Dimensions*. Dimensions of Work shall not be determined by scale or rule. Figured dimensions shall be followed at all times. If figured dimensions are lacking on Drawings, Architect shall supply them on request. The Architect's decisions on matters relating to aesthetic effect will be final.

# 1.2.2 <u>Addenda and Deferred Approvals</u>

1.2.2.1 Addenda are the changes in Specifications, Drawings, Contract Documents, and Plans which have been authorized in writing by the District or Architect, and which alter, explain, or clarify the Contract Documents. Addenda shall govern over all other Contract Documents. Subsequent addenda issued shall govern over prior addenda unless otherwise specified in the addenda.

1.2.2.2 Deferred Approvals. Deferred Approvals are Submittals that are reviewed by the Architect (or Engineer of Record) and submitted to DSA for approval based on thorough detailing of manufacturer and Project specific design. See Article 3.9.1and 3.9.3. The Deferred Approval item cannot be fully detailed on the originally approved Drawings or Specifications because of variations in product design and manufacture. Contract Documents which require Deferred Approval items are meant to be for illustration purposes only. Approval of Plans for such a portion of the Work may be deferred until the material suppliers and Subcontractors are selected. All Deferred Approvals are noted in the Plans and Specifications. Contractor is responsible for all Deferred Approval requirements set forth in the Contract Documents. Contractor is responsible to comply with all laws, building codes, Title 24 and regulations necessary to obtain all necessary approvals, including those required from the Division of the State Architect ("DSA") and the State Fire Marshall. Contractor shall not be granted an extension of time for failure to plan, schedule for and obtain necessary approvals. Contractor shall Schedule all Deferred Approval items in the Baseline Schedule and Schedule Updates under Article 3.9.6

#### 1.2.3 <u>Specification Interpretation</u>

- 1.2.3.1 *Titles*. The Specifications are separated into titled sections for convenience only and not to dictate or determine the trade or craft involved.
- 1.2.3.2 As Shown, Etc. Where "as shown," "as indicated," "as detailed," or words of similar import are used, reference is made to the Drawings accompanying the Specifications unless otherwise stated. Where "as directed," "as required," "as permitted," "as authorized," "as accepted," "as selected," or words of similar import are used, the direction, requirement, permission, authorization, approval, acceptance, or selection by Architect is intended unless otherwise stated.
- 1.2.3.3 *General Conditions*. The General Conditions and Supplementary General Conditions are a part of the Contract Documents which further defines and refines the Contract entered between the Contractor and District.
- 1.2.3.4 Abbreviations. In the interest of brevity, the Specifications are written in an abbreviated form and may not include complete sentences. Omission of words or phrases such as "Contractor shall," "shall be," etc., are intentional. Nevertheless, the requirements of the Specifications are mandatory. Omitted words or phrases shall be supplied by inference in the same manner as they are when a "note" occurs on the Drawings. In the interest of brevity, the Contract Documents frequently omit modifying words such as "all" and "any" and articles such as "the" and "an," but the fact that a modifier or an article is absent from one statement and appears in another is not intended to affect the interpretation of either statement.
- 1.2.3.5 *Plural*. Words in the singular shall include the plural whenever applicable or the context so indicates.
- 1.2.3.6 *Metric*. The Specifications may indicate metric units of measurement as a supplement to U.S. customary units. When indicated thus: 1" (25 mm), the U.S. customary unit is specific, and the metric unit is nonspecific. When not shown with parentheses, the unit is specific. The metric units correspond to the "International System of Units" (SI) and generally follow ASTM E 380, "Standard for Metric Practice."
- 1.2.3.7 Standard Specifications. Any reference to standard specifications of any society, institute, association, or governmental authority is a reference to the organization's standard

specifications, which are in effect at the date of the Contractor's proposal unless directed otherwise. If applicable specifications are revised prior to completion of any part of the Work, the Contractor may, if acceptable to Architect, perform such Work in accordance with the revised specifications. The standard specifications, except as modified in the Specifications for the Project, shall have full force and effect as though printed in the Specifications. Architect will furnish, upon request, information as to how copies of the standard specifications referred to may be obtained.

#### 1.2.4 Rules of Document Interpretation

- 1.2.4.1 In the event of conflict within the Drawings, the following rules shall apply:
- a. General Notes, when identified as such, shall be incorporated into other portions of Drawings.
- b. Schedules, when identified as such, are complementary with other notes and other portions of Drawings including those identified as General Notes.
- c. Larger scale Drawings shall take precedence over smaller scale Drawings.
- d. At no time shall the Contractor base construction on scaled Drawings.
- 1.2.4.2 Specifications shall govern as to materials, workmanship, and installation procedures.
- 1.2.4.3 If Contractor observes that Drawings and Specifications are in conflict, Contractor shall, prior to commencing work, notify the Architect in writing for the purposes of obtaining an interpretation of the Contact Documents.
- 1.2.4.4 In the case of conflict or inconsistencies, the order of precedence shall be as follows:
  - a. General Conditions take precedence over Drawings and Specifications.
  - b. Supplemental Conditions take precedence over General Conditions.
  - c. The Agreement Form shall take precedence over the Supplemental Conditions.
  - d. In the case of disagreement or conflict between or within Specifications, and Drawings, the more stringent, higher quality, and greater quantity of Work shall apply.
  - e. Addenda shall take precedence over Drawings and Specifications.
  - f. General Conditions shall take precedence over Addenda.
  - g. Drawings and Specifications take precedence over the Soils Report.

# 1.3 <u>OWNERSHIP AND USE OF ARCHITECT'S DRAWINGS, SPECIFICATIONS AND OTHER DOCUMENTS</u>

The Drawings, Specifications, and other Contract Documents for the Project are the property of the District and/or Architect pursuant Contract requirements between the District and Architect. The Contractor may retain one Contract record set. Neither the Contractor nor any Subcontractor, or material or equipment supplier shall own or claim a Copyright in the Drawings, Specifications, and other documents prepared by the Architect. All copies except the Contractor's record set, shall be returned or properly accounted for upon completion of the Work. The Drawings, Specifications, and other documents prepared by the Architect, and copies thereof furnished to the Contractor are not to be used by the Contractor or any Subcontractor, Sub-subcontractor, or material or equipment supplier on other projects or for additions to this Project outside the scope of the Work. The District and/or Architect hereby grants the Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers a limited license to use applicable portions of the Drawings, Specifications, and other documents prepared for the Project in the execution of their Work under the Contract Documents. Submittal or distribution to meet official regulatory requirements or for other purposes in connection with this Project is not to be construed as publication in derogation of the District's property interest or other reserved right.

# ARTICLE 2 DISTRICT

# 2.1 <u>INFORMATION AND SERVICES REQUIRED OF THE DISTRICT</u>

#### 2.1.1 <u>Site Survey</u>

The District will furnish, at its expense, a legal description of the Site and a land survey showing the boundaries of the Site. Contractor shall be responsible for all surveys regarding location of construction, grading and site work.

# 2.1.2 Soils

When required by the scope of the Project, the District will furnish, at its expense, the services of geotechnical engineers or consultants when reasonably required and deemed necessary by the Architect or as required by local or state codes. Such services, with written reports and appropriate written professional recommendations, may include test boring, test pits, soil bearing values, percolation tests, air and water pollution tests, and ground corrosion and resistivity tests, including necessary operations for determining subsoil, air, and water conditions.

# 2.1.3 Soils Report Part of the Contract Documents: Contractor Reliance

A soils investigation report has been obtained from test holes at the Site, and such report is incorporated into this Contract and made available for the Contractor's use in preparing its bid and Work under this Contract. Where the Plans and Specifications are more specific and provide more significant structure, systems, reinforcing, thicknesses, or construction methods, the Drawings shall control over the soils report. The soils report is available at the Architect's office for review and it is Contractor's responsibility to ensure that Contractor has reviewed the soils investigation report. Any information obtained from such report or any other information given on Drawings as to subsurface soil condition or to elevations of existing grades or elevations of underlying rock is approximate only. If, during the course of Work under this Contract, Contractor encounters subsurface conditions which differ materially from those indicated in the soils report, then Contractor shall notify the District within five (5) calendar days of discovery of the condition, and changes to the Contract Price may be made in accordance with Article 7 entitled "Changes in the Work." Contractor agrees that no claim against District will be made by Contractor for damages and hereby waives any rights to damages in the event the Contractor fails to notify District within the five-day period mentioned above.

WARNING: DISTRICT DOES NOT WARRANT THE SOILS AT THE PROJECT SITE. CONTRACTOR HAS REVIEWED AND IS FAMILIAR WITH THE REQUIREMENTS OF THE SOILS INVESTIGATION REPORT. CONTRACTOR UNDERSTANDS THAT PLANS, DRAWINGS AND SPECIFICATIONS SUPERSEDE THE SOILS REPORT IF THERE ARE CONFLICTS. FURTHER, IN ADDITION TO THE INFORMATION IN THE SOILS REPORT, CONTRACTOR HAS CONDUCTED AN INDEPENDENT INVESTIGATION OF THE PROJECT SITE AND THE SOILS CONDITIONS OF THE SITE. DISTRICT DOES NOT WARRANT THE SOILS CONDITIONS OF THE SITE AND CONTRACTOR IS FULLY RESPONSIBLE TO ASCERTAIN SITE CONDITIONS FOR THE PURPOSES OF DETERMINING CONSTRUCTION MEANS AND METHODS PRIOR TO COMMENCING CONSTRUCTION.

#### 2.1.4 Utilities

- 2.1.4.1 Location of Point of Connection. The locations shown for the point of connection are approximate. It shall be the responsibility of the Contractor to determine the exact location of all service connections.
- 2.1.4.2 Regional Notification Center. Contractor, except in an emergency, shall contact the appropriate regional notification center at least two (2) business days prior to commencing any excavation if the excavation will be conducted in an area or in a private easement which is known, or reasonably should be known, to contain subsurface installations other than the underground facilities owned or operated by the District, and obtain an inquiry identification number from that notification center. See Government Code section 4216.3. No excavation shall be commenced and carried out by the Contractor unless such an inquiry identification number has been assigned to the Contractor or any Subcontractor of the Contractor and the District has been given the identification number by the Contractor. Any damages arising from failure to make appropriate regional notification shall be at the sole risk of Contractor. Contractor shall solely be responsible for any fines, penalties or damages for violation of this Article and Government Code section 4216.6 or 4216.7. Any delays caused by failure to make appropriate regional notification shall be at the sole risk of Contractor and shall not be considered for extension of time pursuant to Article 8.4.
- 2.1.4.3 *Utilities Removal and Restoration.* The District has endeavored to determine the existence of utilities at the Site of the Work from the records of the District of known utilities in the vicinity of the Work. The positions of these utilities as derived from such records are shown in the Contract Documents. Thus, the locations of the main or trunklines located on the Drawings are approximate locations and not exact.

No excavations were made to verify the locations shown for underground utilities. Other than the main or trunkline, which the District has endeavored to locate on the Plans, service connections or laterals to these utilities may not be shown on the Plans. It shall be the responsibility of the Contractor to determine the exact location of all service connections. The Contractor shall make its own investigations, including exploratory excavations, to determine the locations and type of service connections, prior to commencing work which could result in damage to such utilities. The Contractor shall immediately notify the District's representative as to any utility main or trunkline discovered by Contractor in a different position than provided by the Regional Notification Center. With respect to main or trunklines, Contractor is to immediately notify District if the location is substantially different than as shown in the Contract Documents.

Contractor shall coordinate its Work with all utilities, including, but not limited to electricity, water, gas and telephone and meet with said utilities prior to the start of any work. Contractor shall show timing of all utility coordination activities under the Scheduling requirements of Article 8.

2.1.4.4 *Other Utilities.* In case it should be necessary to remove, relocate, or temporarily maintain a utility because of interference with the Work, the work on the utility shall be performed and paid for as follows:

When it is necessary to remove, relocate or temporarily maintain a service connection, the cost of which is not required to be borne by the owner of the service connection, the Contractor shall bear all expenses incidental to the work on the service connection. The work on the service connection shall be done in a manner satisfactory to the owner thereof; it being understood that the owner

of the service connection has the option of doing such work with his own forces or permitting the work to be done by the Contractor.

When it is necessary to remove, relocate, or temporarily maintain a utility which is in the position shown on the Plans, the cost of which is not required to be borne by the owner thereof, the Contractor shall bear all expenses incidental to the work on the utility. The work on the utility shall be done in a manner satisfactory to the owner thereof; it being understood that the owner of the utility has the option of doing such work with his own forces or permitting the work to be done by the Contractor.

When it is necessary to remove, relocate, or temporarily maintain a utility which is not shown on the Plans or is in a position different from that shown on the Plans and were it in the position shown on the Plans would not need to be removed, relocated, or temporarily maintained, and the cost of which is not required to be borne by the owner thereof, the District will make arrangements with the owner of the utility for such work to be done at no cost to the Contractor, or will require the Contractor to do such work in accordance with Article 7 or will make changes in the alignment and grade of the Work to obviate the necessity to remove, relocate, or temporarily maintain the utility. Changes in alignment and grade will be ordered in accordance with Article 7 herein.

No representations are made that the obligations to move or temporarily maintain any utility and to pay the cost thereof is or is not required to be borne by the owner of such utility, and it shall be the responsibility of the Contractor to investigate to find out whether said cost is required to be borne by the owner of the utility.

The right is reserved to governmental agencies and to owners of utilities to enter at any time upon any street, alley, right-of-way, or easement for the purpose of making changes in their property made necessary by the Work and for the purpose of maintaining and making repairs to their property.

#### 2.1.5 Existing Utility Lines; Removal, Relocation

2.1.5.1 *Main or Trunkline Facilities*. If the Contractor while performing the Contract discovers utility facilities not identified in the Contract Documents, Contractor shall notify the District and utility in writing prior to commencing work.

The owner of the public utility shall have the sole discretion to perform repairs or relocation work or permit the Contractor to do such repairs or relocation work at a reasonable price.

The Contractor shall exercise reasonable care and shall be compensated by the District for the actual verified field costs of locating, and removing, relocating, protecting or temporarily maintaining such main or trunkline utility facilities located in a substantially different location than in the Plans and Specifications, and for equipment in use on the project necessarily idled during such work. This Work shall be performed in accordance with Article 7 of these General Conditions.

2.1.5.2 Assessment. Nothing in these subparagraphs shall be deemed to require the District to indicate the presence of existing service laterals or appurtenances whenever the presence of such utilities on the Site can be inferred from the presence of other visible facilities, such as buildings, or meter junction boxes on or adjacent to the Site and could be inferred from the Main or Trunkline shown on the Drawings.

2.1.5.3 *Notification*. If the Contractor, while performing Work under this Contract, discovers utility facilities not identified by the District in the Contract Documents. Contractor shall, within five (5) days, notify the District and the utility in writing. If Contractor fails to notify the District within forty-eight hours after discovery of any utility facilities not identified by District in the Contract Documents, Contractor waives all rights to be compensated for any extra Work or damages resulting from such discovered utilities.

#### 2.1.6 Easements

District shall secure and pay for easements for permanent structures or permanent changes in existing facilities, if any, unless otherwise specified in the Contract Documents.

# 2.2 <u>DISTRICT'S RIGHT TO CARRY OUT THE WORK DUE TO PARTIAL DEFAULT IN A SPECIFIC SEGREGATED AREA OF WORK (48 HOUR NOTICE TO CURE AND CORRECT)</u>

If the Contractor Defaults or neglects to carry out the Work in accordance with the Contract Documents, the District may provide forty-eight (48) hour written notice to cure (a shorter period of time in the case of Emergency or a critical path delay as defined in Article 2.2.1) Contractor's Partial Default in a specific segregated area of work. The District's right to issue a Partial Default of the Contractor's Work and take over that segregated area of Work includes, but is not limited to:

- 1. Failure to supply adequate workers on the entire Project or any part thereof;
- 2. Failure to supply a sufficient quantity of materials;
- 3. Failure to perform any provision of this Contract;
- 4. Failure to comply with safety requirements, or due to Contractor is creation of an unsafe condition:
- 5. Cases of bona fide emergency;
- 6. Failure to order materials in a timely manner;
- 7. Failure to prepare Deferred Approval items or Shop Drawings in a timely manner;
- 8. Failure to comply with Contractor's Baseline or Update Schedule, meet critical Milestones which would result in a delay to the critical path, or delay the Contract Time;
- 9. Failure to comply with the Subletting and Subcontracting Fair Practices, Public Contract Code section 4100, et seq.
- 10. Failure to meet the requirements of the Americans with Disabilities Act;
- 11. Failure to complete Punch List work;
- 12. Failure to proceed on an Immediate Change Directive
- 13. Failure to correct a Notice of Deviation

If during the forty-eight (48) hour period, the Contractor fails to Cure and correct the deficiency noted in the 48-hour notice of Partial Default with diligence and promptness, the District may correct such deficiencies without prejudice to other remedies the District may have, including a Termination for Cause as set forth in Article 14. If there are inadequate funds remaining the Project balance or in the Retention Escrow to address at least 150% of the costs set forth in the Article 2.2 notice, the District may copy the Surety on the written notice of Partial Default. If a notice to the Surety is provided, except in the cases of emergency or critical path delay, the Surety has the option to take over and complete the Work described in the written notice if Surety personally delivers notice to District that it intends to perform such work. In the case where written notice has been provided, the District shall allow Surety seven (7) days to perform the Work.

# 2.2.1 <u>Service of Notice of Partial Default with Right to Cure</u>

A written notice of Partial Default and right to cure under Article 2.2 ("Article 2.2 Notice" or "Notice of Partial Default") shall be served by e-mail (with a copy provided by regular mail) to the e-mail address provided on the Bid submitted and copied to the Project Superintendent.

#### 2.2.2 Shortened Time for Partial Default in the Case of Emergencies.

In an Emergency situation, the District may correct any of the deficiencies described in Article 2.2 without prejudice to other remedies by providing service of written notice of Emergency requiring a shortened time for Partial Default specifying the time given to cure, if any.

# 2.2.3 <u>Shortened Time for Partial Default in the Case of Critical Path Delay</u>

In the case of critical path delay, the District may correct any of the deficiencies described in Article 2.2 without prejudice to other remedies providing service of written notice of critical path delay to the Contractor with a specific description of the critical path delay items noting the line item or area of Work that is on the critical path and prescribe the length of shortened time to cure, if any.

# 2.2.4 <u>Written Notice of Partial Default to be Deducted by Deductive Change Order</u>

The District shall have the right to determine the reasonable value of the Article 2.2 Partial Default Work, or if there is an actual value for the Work, shall use that value and issue a Deductive Change Orders under Article 7.7.4

# ARTICLE 3 THE CONTRACTOR

# 3.1 SUPERVISION AND CONSTRUCTION PROCEDURES

#### 3.1.1 <u>Contractor</u>

The Contractor shall continually supervise and direct the Work using the Contractor's best skill and attention. The Contractor shall be solely responsible for and have control over construction means, methods, techniques, sequences, procedures; and shall coordinate all portions of the Work under the Contract, unless the Contract Documents give other specific instructions concerning these matters. The Contractor shall not perform the Work without utilizing the Contract Documents or, where required, approved Submittals, Shop Drawings, or samples for any such portion of the Work. If any of the Work is performed by contractors retained directly by the District, Contractor shall be responsible for the coordination and sequencing of the work of those other contractors so as to avoid any impact on the Project Schedule pursuant to the requirements of Article 6 and Article 8. Specific duties of the Contractor shall include those set out in Section 43 of Title 21 of the California Code of Regulations and Section 4-343 of Title 24 of the California Code of Regulations. These duties include, but are not limited to the following:

- 3.1.1.1 Responsibilities. It is the duty of the Contractor to complete the Work covered by his or her Contract in accordance with the approved Plans and Specifications. The Contractor in no way is relieved of any responsibility by the activities of the Architect, Engineer, Inspector or DSA in the performance of their duties.
- 3.1.1.2 Performance of the Work. The Contractor shall carefully study the approved Plans and Specifications and shall plan its schedule of operations well ahead of time. If at any time it is discovered that work is being done which is not in accordance with the approved Plans and Specifications, the Contractor shall correct the Work immediately.

#### 3.1.2 Contractor Responsibility to Study the Plans and Specifications

All inconsistencies or timing or sequences which appear to be in error in the Plans and Specifications shall promptly be called to the attention of the Architect or, Engineer, for interpretation or correction. Local conditions which may affect the structure shall be brought to the Architect's attention at once. In no case, shall the instruction of the Architect be construed to cause work to be done which is not in conformity with the approved Plans, Specifications, change orders, construction change documents, and as required by law. (See Title 24, Section 4-343)

# 3.1.3 <u>All Work Under the Direction of Inspector</u>

Pursuant to Title 24 requirements, the Contractor shall not carry on Work except with the knowledge of the Inspector. (See Title 24 generally)

# 3.1.4 <u>Contractor to Establish Timing and Protocol with Inspector</u>

Contractor shall establish a protocol for requesting inspection with Inspector so as to not delay the Work and provide adequate time for the Inspector to perform inspection. If such a protocol is not established ahead of time, Inspector may utilize the time criteria set by Title 24 of 48 hours in advance of submitting form DSA 156 for each new area. DSA requirements under PR 13-01 specifically gives the

Special Inspector fourteen (14) days to post to the DSA website. Contractor is responsible for delays and for failure to plan.

For some Projects, there may be a need to incrementally install certain assemblies. It is up to Contractor to identify areas and assemblies that may be constructed incrementally. Contractor must identify and establish incremental areas of construction and establish protocols with Inspector for DSA 152 approvals so they may be presented to DSA. (See PR-13 item 1.17 for further discussion)

# 3.1.5 <u>Verified Reports</u>

The Contractor shall make and submit to the office from time to time, verified reports as required in Title 24 Section 4-366. As part of the Close-Out of the Project (see Article 9.9), Contractor shall be required to execute a Form 6-C as required under Title 24 Sections 4-343.

Contractor shall fully comply with any and all reporting requirements of Education Code sections 17315, et seq., in the manner prescribed by Title 24, as applicable.

# 3.1.6 <u>Contractor Responsibility</u>

The Contractor shall be responsible to the District for acts and omissions of the Contractor's employees, Subcontractors, material and equipment suppliers, and their agents, employees, invitees, and other persons performing portions of the Work under direct or indirect contract with the Contractor or any of its Subcontractors.

# 3.1.7 Obligations not Changed by Architect's Actions

The Contractor shall not be relieved of obligations to perform the Work in accordance with the Contract Documents either by activities or duties of the Architect in the Architect's administration of the Contract or by tests, inspections, or approvals required or performed by persons other than the Contractor.

# 3.1.8 <u>Acceptance/Approval of Work</u>

The Contractor shall be responsible to determine when any completed portions of the Work already performed under this Contract or provided pursuant to Article 6 are suitable to receive subsequent Work thereon.

#### 3.2 SUPERVISION

#### 3.2.1 <u>Full Time Supervision</u>

Unless personally present on the Project site where the Work is being performed, the Contractor shall keep on the Work at all times during its progress a competent, English speaking construction Superintendent satisfactory to the District. The Superintendent shall be present on a full-time basis, shall be dedicated exclusively to the Project and shall not share superintendency duties with another project or job. The Superintendent shall not be replaced except with written consent of the District. The Superintendent shall represent the Contractor in its absence and shall be fully authorized to receive and fulfill any instruction from the Architect, the Inspector, the District or any other District Representative (including CM in the cases where the District has a CM representative). All Requests for Information shall be originated by the Superintendent and responses thereto shall be given to the Superintendent. No Work

shall begin on any day by any Subcontractor or other person on the Project site until the Superintendent has arrived, or shall any Work continue during the day after the Superintendent has departed from the Project site. The Superintendent shall have authority to bind Contractor through the Superintendent's acts. The Superintendent shall represent the Contractor, and communications given to the Superintendent shall be binding on the Contractor. Before commencing the Work, Contractor shall give written notice to District (and CM representative) and Architect of the name and a Statement of Qualifications of such superintendent. Superintendent shall not be changed except with written consent of District, unless a superintendent proves to be unsatisfactory to Contractor and ceases to be in its employ, in which case, Contractor shall notify District and Architect in writing. Contractor shall provide a replacement superintendent approved by the District prior to performing additional work.

# 3.2.2 <u>Staff</u>

Notwithstanding other requirements of the Contract Documents, the Contractor and each Subcontractor shall: (1) furnish a competent and adequate staff as necessary for the proper administration, coordination, supervision, and superintendence of its portion of the Work; (2) organize the procurement of all materials and equipment so that the materials and equipment will be available at the time they are needed for the Work; and (3) keep an adequate force of skilled and fit workers on the job to complete the Work in accordance with all requirements of the Contract Documents.

#### 3.2.3 Right to Remove

District shall have the right, but not the obligation, to require the removal from the Project of any superintendent, staff member, agent, or employee of any Contractor, Subcontractor, material or equipment supplier.

# 3.3 <u>LABOR AND MATERIALS</u>

#### 3.3.1 Contractor to Provide

Unless otherwise provided in the Contract Documents, the Contractor shall provide and pay for labor, material, equipment, tools, construction equipment and machinery, water, heat, air conditioning, utilities, transportation, and other facilities, services and permits necessary for proper execution and completion of the Work whether temporary or permanent and whether or not incorporated or to be incorporated in the Work.

#### 3.3.2 Quality

Unless otherwise specified, all materials and equipment to be permanently installed in the Project shall be new and shall be of the highest quality or as specifically stated in the Contract Documents. The Contractor shall, if requested, furnish satisfactory evidence as to kind and quality of all materials and equipment within ten (10) days of a written request by the District, including furnishing the District with bona fide copies of invoices for materials or services provided on the Project. All labor shall be performed by workers skilled in their respective trades, and shall be of the same or higher quality as with the standards of other school construction.

#### 3.3.3 Replacement

Any work, materials, or equipment, which do not conform to these requirements or the standards set forth in the Contract Documents, may be disapproved by the District, in which case, they shall be removed and replaced by the Contractor at no additional cost or extension of time to the District.

#### 3.3.4 Discipline

The Contractor shall enforce strict discipline and good order among the Contractor's and Subcontractor's employees, and other persons carrying out the Contract. The Contractor shall not permit employment of unfit persons or persons not skilled in tasks assigned to them. As used in this subsection, "unfit" includes any person who the District concludes is improperly skilled for the task assigned to that person, who fails to comply with the requirements of this article, or who creates safety hazards which jeopardize other persons and/or property.

# 3.3.5 <u>Fingerprinting (Applicable at the time Project is Occupied and on all Projects where Workers will come in Contact with Pupils, such as Modernization Projects)</u>

If applicable, Contractor shall comply with the applicable provisions of Education Code section 45125.1 in a method as determined by the District. Pursuant to Education Code section 45125.1, Contractor shall either conduct criminal background checks of all employees of Contractor assigned to the Project site, and shall certify that no employees who have been convicted of serious or violent felonies, as specified in Education Code section 45125.1, will have contact with pupils, by utilizing the Certification Regarding Background Checks and the corresponding Attachment "A" as found in the Contract Documents or shall be separated by a physical barrier from students.

If it is determined that Contractor must provide certification of employees, as part of such certification, Contractor must provide the District with a list of all employees providing services pursuant to this Agreement, and designate which sites such employees will be assigned. In performing the services set forth in this Agreement, Contractor shall not utilize any employees who are not included on the above-referenced list.

At District's sole discretion, District may make a finding, as authorized under Education Code section 45125.1, that Contractor's employees will have only "limited contact" with pupils. Contractor's failure to comply with this law shall be considered a material breach of this Agreement upon where this Agreement may be terminated, at District's sole discretion, without any further compensation to Contractor.

In the case of new construction Projects where there are no students, if the Project Schedule provides for Beneficial Occupancy or portions of the Project or if the Project should be delayed, then Contractor, at no additional costs, shall meet the requirements of either fingerprinting or providing a physical barrier as required by the District.

#### 3.3.6 Noise, Drugs, Tobacco, and Alcohol

Contractor shall take all steps necessary to insure that employees of Contractor or any of its Subcontractors' employees do not use, consume, or work under the influence of any alcohol, tobacco or illegal drugs while on the Project. Contractor shall further prevent any of its employees or its Subcontractor employees from playing any recorded music devices or radios or wearing any radio headphone devices for entertainment while working on the Project. Likewise, Contractor shall prevent its employees or

Subcontractor's employees from bringing any animal onto the Project. Contractors shall not violate any written school policies.

# 3.3.7 Delivery of Material

Contractor shall place orders for materials or equipment so that the Work may be completed in accordance with the Construction schedule for the Work as set forth in Article 8 of this Agreement. Contractor shall, upon demand from the Architect, furnish to the Architect documentary evidence including, but not limited to purchase orders, invoices, bills of materials, work orders and bills of lading, showing that orders have been placed. Contractor shall have a system to receive materials and to ensure that the proper materials are being delivered, including in the case of critical materials to the Project, checking the delivery against Shop Drawings and ensuring that the materials meet the requirements of not only the Plans and Specifications, but also the approved Shop Drawings and Submittals and in conformance with Contractor's plan for delivery of materials (including but not limited to Contractor's representations in the Schedules for the Project and Contractor's equipment and materials schedule under Article 3.7.2.2). Contractor shall be responsible for all costs of accepting non-conforming materials delivered to the Project given Contractor's responsibilities and system for acceptance of deliveries. Contractor shall notify Inspector and District Representative (including CM) as early as possible, in writing, of the delivery of materials for the Project. The deliveries shall include documentation identifying the shipment sufficiently so that the Inspector, Architect or District Representative (including CM) may review the materials that are received. Under no circumstances shall materials be delivered to the Project site that are meant for another Project.

# 3.3.8 <u>Liens and Other Security Interests of Subcontractors and Material Suppliers</u>

No material, supplies, or equipment for the Work shall be purchased subject to any chattel mortgage or under a conditional sale or other agreement by which an interest therein or in any part thereof is retained by seller or supplier. Contractor warrants good title to all material, supplies, and equipment installed or incorporated in Work and agrees upon completion of all Work to deliver premises, together with all improvements and appurtenances constructed or placed thereon by it, to District free from any claims, security interests, liens, or charges. Contractor further agrees that neither it nor any person, firm, or corporation furnishing any materials or labor for any Work covered by this Contract shall have any right to place a lien upon the premises or any improvement or appurtenance thereof, except that Contractor may install metering devices or other equipment of a utility company or political subdivision, title to which is commonly retained by the utility company or political subdivision. In event of installation of any such metering device or equipment, Contractor shall advise District as to its owner within five (5) days of such installation in writing, prior to making the installation.

Contractor agrees to indemnify, defend and hold the District harmless from any liens, stop notices, or assertion of security interests, including judgments and levies. If after written notice Contractor fails to address the lien, stop notice, or other security interest, the District may proceed to address the lien, stop notice or claim and seek reimbursement from Contractor.

#### 3.3.9 Title to Materials

The title to new materials or equipment for the Work of this Contract shall remain with Contractor until incorporated in the Work of this Contract until final acceptance of the Project; no part of said materials shall be removed from its place of storage, and Contractor shall keep an accurate inventory of all said materials and equipment in a manner satisfactory to the District or its authorized representative. Responsibility for materials remains with Contractor and Contractor shall replace materials in case of loss.

District similarly may pay for materials stored off site, but Contractor shall remain responsible for the materials that are stored off site.

#### 3.3.10 Assemblies

For all material and equipment specified or indicated in the Drawings, the Contractor shall provide all labor, materials, equipment, and services necessary, (including engineering as specifically required with Shop Drawings or Deferred Approvals) for complete assemblies and complete working systems. Incidental items not indicated on the Drawings, nor mentioned in the Specifications, that can legitimately and reasonably be inferred to belong to the Work described, or be necessary in good practice to provide a complete assembly or system, shall be furnished as though itemized in the Contract Documents in every detail. In all instances, material and equipment shall be installed in strict accordance with each manufacturer's most recent published recommendations and Specifications.

# 3.3.11 Noise Control

The Contractor shall be responsible for the installation of noise reducing devices on construction equipment. Contractor shall comply with the requirements of the city and county having jurisdiction with regard to noise ordinances governing construction sites and activities. Construction equipment noise is subject to the control of the Environmental Protection Agency's Noise Control Program (Part 204 of Title 40, Code of Federal Regulations). If school is in session at any point during the progress of the Project, and, in the District's reasonable discretion, the noise from such Work disrupts or disturbs the students or faculty or the normal operation of the school, at the District's request, the Contractor shall schedule the performance of all such Work around normal school hours or make other arrangements so that the Work does not cause such disruption or disturbance. There are specific periods of testing at operational schools and it is critical that Contractor control noise during periods of testing. In no event shall Contractor have a right to receive additional compensation or an extension to the Contract time as a result of any such rescheduling or the making of such arrangements. These controls shall be implemented during site preparation and construction. All noise related issues, including school operations, and noise during testing should be detailed in the Schedule provided pursuant to Article 8

# 3.4 WARRANTY

The Contractor warrants to the District and Architect that material and equipment furnished under the Contract will be of the highest quality and new unless otherwise required or permitted by the Contract Documents, that the Work will be free from defects not inherent in the quality required or permitted, and that the Work will conform with the requirements of the Contract Documents. Work not conforming to these requirements, including substitutions not properly approved and authorized, may be considered defective. Contractor's warranty to District includes, but is not limited to, the following representations:

- 3.4.1 In addition to any other warranties provided elsewhere, Contractor shall, and hereby does, warrant all Work after the date of Notice of Completion of Work by District and shall repair or replace any or all such Work, together with any other Work, which may be displaced in so doing that may prove defective in workmanship or materials within a one (1) year period from date of Final Completion which shall be no later than the final date of Punch List as noted at Article 9.11) without expense whatsoever to District, ordinary wear and tear, unusual abuse or neglect excepted. District will give notice of observed defects with reasonable promptness. Contractor shall notify District upon completion of repairs.
- 3.4.2 <u>In the event of failure of Contractor to comply with above mentioned conditions within one week after being notified in writing, District is hereby authorized to proceed to have defects repaired</u>

and made good at expense of Contractor who hereby agrees to pay costs and charges therefore immediately on demand.

- 3.4.3 If, in the opinion of the District, defective Work creates a dangerous condition or requires immediate correction or attention to prevent further loss to the District, the District will attempt to give the notice required by this Article. If the Contractor cannot be contacted or does not comply with the District's requirements for correction within a reasonable time as determined by the District, the District may, notwithstanding the provisions of this article, proceed to make such correction or attention which shall be charged against Contractor. Such action by the District will not relieve the Contractor of the guarantee provided in this Article or elsewhere in this Contract.
- 3.4.4 This Article does not in any way limit the guarantee on any items for which a longer warranty is specified or on any items for which a manufacturer gives a guarantee for a longer period. Contractor shall furnish District all appropriate guarantee or warranty certificates upon completion of the project.

# 3.5 TAXES

Contractor will pay all applicable Federal, State, and local taxes on all materials, labor, or services furnished by it, and all taxes arising out of its operations under the Contract Documents. District is exempt from Federal Excise Tax, and a Certificate of Exemption shall be provided upon request.

# 3.6 PERMITS, FEES AND NOTICES

# 3.6.1 Payment

The Contractor shall secure and pay for all permits and governmental fees, licenses, and inspections necessary for proper execution and completion of the Work which are necessary after execution of the Contract and are legally required by any authority having jurisdiction over the Project, except those required by the Division of the State Architect (DSA). District shall be responsible for all testing and inspection as required by the DSA on-site or within the distance limitations set forth in Article 13.5.2, unless a different mileage range is specified in the Supplemental Conditions.

3.6.1.1 DSA Fees. DSA policy is to charge CCD review fees for processing and approval of changes in the Plans and Specifications through the Construction Change Document process. Contractor is specifically directed to the current DSA IR A-30 which provides fee structure and charges that will be incurred for proceeding with respect to the CCD process, a process that must be followed for each change in the Plans and Specifications.

# 3.6.2 <u>Compliance</u>

The Contractor shall comply with and give notices required by any law, ordinance, rule, regulation, and lawful order of public authorities bearing on performance of the Work. Specifically, the Division of State Architect provides State oversight of the Project and enforcement of Title 24 rules and regulations. Contractor is directed to the DSA website. There will be local governmental oversight from City, County or both. Finally, Regional Water Quality Control Board, State Fire Marshall, local fire marshal, Department of Industrial Relations, Department of Labor Standards Enforcement, and Air Quality Management District (Local and State) are some of the agencies that provide oversight and may require specific permits, fees, or provide oversight over the Project. Contractor represents understanding and specialized knowledge of the rules governing school districts and Contractor shall maintain compliance

over the applicable rules and will file all documents required in order to ensure compliance with State, local, and other rules that apply to the Project.

# 3.6.3 <u>Responsibility</u>

The Contractor shall perform all Work in conformance with every law, statute, ordinance, building code, rule or regulation. The Contractor shall assume full responsibility for such Work and shall bear the attributable cost of correction or project delay.

Pursuant to Title 24 Section 4-343(b):

"Contractor shall carefully study the approved Plans and Specifications and shall plan a schedule of operations well ahead of time.... All inconsistencies or items which appear to be in error in the Plans and Specifications shall be promptly called to the attention of the architect or registered engineer, through the inspector, for interpretation or correction."

To help Contractor plan its operations, Contractor is directed to study the current version of the DSA 152 Inspection Card Manual identifying the exact steps the Inspector is to follow in the review and sign off process for the DSA 152. The DSA 152 Inspection Card Manual provides specific detail as to the order of operations, review items and compliance items beyond the Specifications and Plans which are reviewed for DSA compliance. The most current version of this manual is located on DSA's website.

Contractor is also specifically directed to the time periods for posting of Special Inspection Reports and Inspector Notifications under DSA PR 13-01 since the timing of Inspection is not a Governmental Entity related delay.

#### 3.7 SUBMITTALS REQUIRED AT THE COMMENCEMENT OF THE PROJECT

#### 3.7.1 Requirements Within Ten (10) Calendar Days

Within ten (10) calendar days after Notice to Proceed, Contract shall submit the following:

- 3.7.1.1 Detailed Schedule of Values (See Article 9.2)
- 3.7.1.2 Submittal Listing and Schedule for Submittals
- 3.7.1.3 Critical Path Baseline Schedule (See Article 8)

#### 3.7.2 Requirements Within Thirty-Five (35) Calendar Days

Within thirty-five (35) calendar days after Notice to Proceed, Contractor shall submit the following:

- 3.7.2.1 All Submittals for the Project except those specifically agreed upon by District and Architect, in writing, and shall be specifically incorporated into the Submittal section of the Schedule so as to not delay the Work. The agreement to allow a later Submittal does not mean that Article 3.3.7 is waived. Contractor shall order materials and ensure prices are honored and secured for the Project.
  - a. Structural Steel may be included as a later Submittal than 35 days if Structural Steel is a significant portion of the Work, at least one or some of the Project is a

structural steel structural system, or as specifically agreed upon by the Architect or District.

- b. It is specifically agreed that submissions of structural steel Submittals shall not be piecemeal (unless some portion is requested separately by the District or Architect), shall provide complete designs, shall be stamped by the structural steel Subcontractor, Contractor, and structural steel Subcontractor's structural engineer at time of submission and as further addressed in Article 3.9.
- c. In no case shall the submission of structural steel Drawings delay the critical path for the schedule. If a Milestone is provided for submission of complete structural steel Shop Drawings, then the date shall be no later than as set forth in the Milestone
- 3.7.2.2 Exceptions to Submittal Within Thirty-Five (35) Days by Written Agreement. A written request detailing the specific reasons for a submission later than 35 days due to complexity of design or non-critical path status of the Submittal shall be submitted at the time the Baseline Schedule is submitted. The Baseline Schedule shall not include a delayed Submittal until written agreement is provided. In addition to the request for providing a Submittal after the thirty-five (35) day period, a copy of the Contract with the Subcontractor who shall be performing the Submittal, a written statement from the Subcontractor verifying that work has commenced on the Submittal and providing Subcontractor's own schedule of Milestones and completion dates, and a corresponding Submittal designation in the Schedule as required under Article 8. Approval of a delayed Submittal shall not result in any increase in the Contract Price or result in an extension of time for the completion of the Project.
- 3.7.2.3 Piecemeal Submissions of Submittals. Piecemeal Submittals mean providing portions of Shop Drawings or Submittals as they are being completed. The submission of piecemeal Submittals results in the appearance of a submission when there is inadequate information for the Architect or Engineer to adequately review a submission. Piecemeal differs from submission of complete buildings or phases of buildings or complete assemblies. The Architect may agree to allow submission of single buildings or areas as long as the Submittals are complete.

#### 3.8 DOCUMENTS, SAMPLES, AND COMPUTER AT THE SITE

The Contractor shall maintain at the Site for the District one current copy of the California Building Code, Titles 19 and 24 of the California Code of Regulations, any other document required by DSA, and one record copy of the Drawings, Specifications, Addenda, Change Orders, and other Modifications, in good order and marked currently to record changes and selections made during construction. In addition, the Contractor shall maintain at the Site approved Shop Drawings, Product Data, Samples, and similar required Submittals. These documents shall be available to the Architect and shall be delivered to the Architect for delivery to the District upon completion of the Work.

Contractor shall have an operational computer with internet access so Contractor can review and post documents as required for the Project, including but not limited to the filing and posting of DSA required documents for the Project.

Contractor shall be prepared to review documents posted to the DSA Project website.

# 3.9 <u>SUBMITTALS INCLUDING SHOP DRAWINGS, PRODUCT DATA, AND SAMPLES</u>

# 3.9.1 Definitions

- 3.9.1.1 Deferred Approvals. Approval of certain aspects of the construction may be deferred until the construction Contract has been awarded. To facilitate the design process, DSA grants Deferred Approval to the design and detailing of certain elements of the Project at the request of the Architect or Engineer of Record. Design elements that may be deferred may include, but are not limited to access floors, bleachers, elevator guide rails and related elevator systems, exterior wall systems precast concrete, glass fiber reinforced concrete, etc., skylights, window wall systems, storefronts, stage rigging, and other systems as noted in the Contract Documents. (Also see Article 1.2.2.2 and 3.9.3)
- 3.9.1.2 Shop Drawings. The term "Shop Drawings" as used herein means Drawings, diagrams, equipment or product schedules, and other data, which are prepared by Contractor, Subcontractors, manufacturers, suppliers, or distributors illustrating some portion of the Work, and includes: illustrations; fabrication, erection, layout and setting Drawings; manufacturer's standard Drawings; schedules; descriptive literature, instructions, catalogs, and brochures; performance and test data including charts; wiring and control diagrams; and all other Drawings and descriptive data pertaining to materials, equipment, piping, duct and conduit systems, and methods of construction as may be required to show that the materials, equipment, or systems and their position conform to the requirements of the Contract Documents.
- 3.9.1.3 *Manufactured* applies to standard units usually mass-produced, and "Fabricated" means items specifically assembled or made out of selected materials to meet individual design requirements. Shop Drawings shall: establish the actual detail of all manufactured or Fabricated items, indicate proper relation to adjoining work, amplify design details of mechanical and electrical systems and equipment in proper relation to physical spaces in the structure, and incorporate minor changes of design or construction to suit actual conditions.
- 3.9.1.4 Submittals is a term used interchangeably and sometimes refers to Shop Drawings, Product Data, and samples since all Subcontractor submissions are tracked in a Submittal Log and may include any of the noted items. However, generally, a Submittal is a manufacturer's product information and Product Data including description, characteristics, size, physical characteristics, and requirements to prepare the jobsite for receiving of the particular manufactured item.
- 3.9.1.5 Samples. The term "samples" as used herein are physical examples furnished by Contractor to illustrate materials, equipment, or quality and includes natural materials, Fabricated items, equipment, devices, appliances, or parts thereof as called for in the Specifications, and any other samples as may be required by the Architect to determine whether the kind, quality, construction, finish, color, and other characteristics of the materials, etc., proposed by the Contractor conform to the required characteristics of the various parts of the Work. All Work shall be in accordance with the approved samples.

#### 3.9.2 Shop Drawings.

3.9.2.1 When Shop Drawings Are Required. Shop Drawings are required for prefabricated components and for installation and coordination of these prefabricated components into the Project. In addition, Shop Drawings, are prepared to address the actual size and installation of components from various Subcontractors and provides an opportunity for the Contractor to coordinate and address

conflicts between the subcontracting trades. In some cases, each Subcontractor or trade will provide Shop Drawings in a BIM format or other format as agreed by District.

- 3.9.2.2 Purpose for Shop Drawings. Shop Drawings are the Contractor's manufacturer, Subcontractor, supplier, vendor or the Contractor's detailed drawings showing particularized method for assembly, specifics to a manufacturer, manufacturer component installation requirements, specifics as to a manufactured item, alterations to a manufactured, a custom created item, or drawn version of more detailed information expanding on the Architect's design shown in the Contact Documents. The Shop Drawings address the appearance, performance, size, weight, characteristics and prescriptive descriptions associated with the Contractor or Contractor's Subcontractor's plan for installation or assembly based on the design in the Specifications and Contract Documents. The Shop Drawing often is more detailed than the information shown in the Contract Documents to give the Architect and Engineer the opportunity to review the fabricator's version of the product (along with particulars specific to that particular product), prior to fabrication. References to the Contract Documents, Construction Documents, Drawings, Plans, and Specifications assist the Architect and Engineer in their review of the Shop Drawings. Attachment of manufacturer's material Specifications, "catalog cut sheets," and other manufacturer's information may be provided to accompany Shop Drawings. Because Shop Drawings facilitate the Architect's and Engineer's approval of the system, they should be as clear and complete as possible so they may be reviewed by Architect or Engineer for the Project.
- 3.9.2.3 Shop Drawing Requirements. The Contractor shall obtain and submit with Shop Drawings all seismic and other calculations and all Product Data from equipment manufacturers. "Product Data" as used herein are illustrations, standard schedules, performance charts, instructions, brochures, diagrams, and other information furnished by the Contractor to illustrate a material, product, or system for some portion of the Work.
- 3.9.2.4 Not a Reproduction of Architectural or Engineering Drawings. The Shop Drawings are not a reproduction of the architectural or engineering Drawings. Instead, they must show more detail than the Construction Documents and details the fabrication and/or installation of the items to the manufacturer's production crew or Contractor's installation crews.
- 3.9.2.5 Shop Drawings Engineering Requirements: Some Shop Drawings require an engineer stamp to be affixed on the Drawings and calculations. In such cases, a current and valid engineering stamp shall be affixed by a California registered engineer. No out of State engineers shall stamp Shop Drawings. (See DSA IR A-18). In most cases, an engineer means California registered mechanical, structural, electrical or plumbing engineer. California Registered Civil Engineers will not be accepted for structural details unless specifically approved by DSA.
- 3.9.2.6 DSA Approvals Required Prior to Work. No work on a Shop Drawing that requires DSA approval may proceed until DSA approval is received. Contractor has provided DSA approval time and allowed adequate time for corrections in Contractor's Schedule as required pursuant to Article 8.
- 3.9.2.7 Shop Drawing Identification. All Shop Drawings must be properly identified with the name of the Project and dated, and accompanied by a letter of transmittal referring to the name of the Project and to the Specification section number for identification of each item clearly stating in narrative form, as well as "clouding" all qualifications, departures, or deviations from the Contract Documents. Shop Drawings, for each section of the Work shall be numbered consecutively and the numbering system shall be retained throughout all revisions. All Subcontractor submissions shall be made through the Contractor. Each drawing shall have a clear space for the stamps of Architect and Contractor.

## 3.9.3 <u>Deferred Approvals</u>

Deferred approvals shall be submitted and processed to ensure all DSA and other governmental approvals are secured so as to not delay the Project. There may be additional requirements for Deferred Approvals at Division 1 of the Specifications. All Deferred Approvals shall be prepared by Contractor or Contractor's agent early enough so as to not delay the Project. Contractor is aware that Title 24 California Code of Regulations Section 4-317 have specific requirements for Deferred Approval as to governing agencies and as to the Architect and Engineer for the Project. As a result, any delay associated with the time for approval by applicable agencies or by the Architect or Architect's consultants shall be Contractor's. Contractor is required to comply with inclusion of Deferred Approvals in the Schedule as required under Article 3.9.6DSA Approvals Required Prior to Work. No work on a Deferred Approval item may proceed on the components until DSA approval is received. Contractor has provided DSA approval time and allowed adequate time for any DSA revisions in Contractor's Schedule as required pursuant to Article 8.

## 3.9.4 <u>Submittals and Samples</u>

- 3.9.4.1 *Information Required with Submittals*: Manufacturer, trade name, model or type number and quantities: Information provided must be of sufficient detail to allow Architect and Engineer to compare the submitted item with the specified products and acceptable products listed, in the Specifications and addenda.
- 3.9.4.2 Description of Use and Performance Characteristics: Information should be furnished describing the normal use and expected performance of the product. The Architect and Contractor review this information to confirm that the product is appropriate for the intended use.
- 3.9.4.3 Size and Physical Characteristics: The size and physical characteristics, such as adjustment capabilities, which is reviewed by both the Contractor and Architect. The Contractor has the most available information for comparing adjoining materials and equipment. The Contractor also needs to know the size and weight of the equipment for lifting and handling considerations.
- 3.9.4.4 *Finish Characteristics:* The Architect reviews the available finishes and selects the appropriate finish, if the finish was not previously specified in the documents. The Contractor should confirm that finish requirements in the Specifications are being met by the product.
- 3.9.4.5 *Contractor Responsible for Jobsite Dimensions*: Some material is Custom-Fabricated to job conditions, requiring dimensions from the jobsite. These jobsite dimensions are provided by the Contractor as part of the Contractor's responsibilities for the Project and shall be provided prior to release of the product for manufacture. Contractor shall not rely on Architect or Engineers to provide jobsite dimensions.
- 3.9.4.6 Full Range of Samples Required (When Specific Items Not Specified). Except in cases where the exact color and type of item is specified since the District is utilizing items Standardized or pre-selected by District, the full range of color, graining, texture, or other characteristics are anticipated for review in finished products, a sufficient number of samples of the specified materials shall be furnished by the Contractor to indicate the full range of characteristics which will be present in the finished products. Products delivered or erected without Submittal and approval without providing a full range of samples shall be subject to rejection. Except for range samples, and unless otherwise called for in the various sections of the Specifications or Specification Section 1, samples shall be submitted in duplicate.

- 3.9.4.7 *Labeling of Samples*. All samples shall be marked, tagged, or otherwise properly identified with the name of the submitting party, the name of the Project, the purpose for which the samples are submitted and the date.
- 3.9.4.8 *Transmittal letter.* All samples shall be accompanied by a letter of transmittal containing similar information, together with the Specification section number.
- 3.9.4.9 *Labels and Instructions*. All samples of materials shall be supplied with the manufacturer's descriptive labels and application instructions. Each tag or sticker shall have clear space for the review stamps of Contractor and Architect.
- 3.9.4.10 Architect's Review. The Architect will review and, if appropriate, approve submissions and will return them to the Contractor with the Architect's stamp and signature applied thereto, indicating the timing for review and appropriate action in compliance with the Architect's (or District's) standard procedures. In the cases where a CM is hired by the District, CM may be the party that receives and performance logging and initial processing of the Samples. CM may, in some cases, reject samples that are not in conformance with Contract requirements.

### 3.9.5 Submittal Submission Procedure

- 3.9.5.1 Transmittal Letter and Other Requirements. All Submittals must be properly identified with the name of the Project and dated, and each lot submitted must be accompanied by a letter of transmittal referring to the name of the Project and to the Specification section number for identification of each item clearly stating in narrative form, as well as "clouding" on the submissions, all qualifications, departures, or deviations from the Contract Documents. Shop Drawings, for each section of the Work shall be numbered consecutively and the numbering system shall be retained throughout all revisions. All Subcontractor submissions shall be made through the Contractor. Each drawing shall have a clear space for the stamps of Architect and Contractor. Refer to Division 1. In the case where a CM is hired on the Project, the CM may be designated to receive the Submittals for the Project, log the Submittals, and in some cases reject Submittals that do not conform to Contract requirements. Submittal Procedures for further information.
- 3.9.5.2 *Copies Required*. Each Submittal shall include one (1) legible, reproducible (if electronic is available, electronic copies shall also be provided) and five (5) legible prints of each drawing or schedule, table, cut sheet, etc., including fabrication, erection, layout and setting drawings, and such other drawings as required under the various sections of the Specifications, until final acceptance thereof is obtained. Subcontractor shall submit copies, in an amount as requested by the Contractor, of: (1) manufacturers' descriptive data for materials, equipment, and fixtures, including catalog sheets showing dimensions, performance, characteristics, and capacities; (2) wiring diagrams and controls; (3) schedules; (4) all seismic calculations and other calculations; and (5) other pertinent information as required by the District or Architect. (See also Division 1)
- 3.9.5.3 *Corrections.* The Contractor shall make all corrections required by Architect, District or CM and shall resubmit, as required by Architect or CM, corrected copies of Shop Drawings or new samples until approved. Contractor shall direct specific attention in writing or on resubmitted Shop Drawings to revisions other than the corrections required by the Architect on previous submissions. Professional services required for more than one (1) re-review of required Submittals of Shop Drawings, Product Data, or samples are subject to charge to the Contractor pursuant to Article 4.5.

- 3.9.5.4 Approval Prior to Commencement of Work. No portion of the Work requiring a Shop Drawing or sample submission or other Submittal shall be commenced until the submission has been reviewed by Contractor and Architect (and CM, if applicable) and approved by Architect (and CM where applicable) unless specifically directed in writing by the Architect. All such portions of the Work shall be in accordance with approved Shop Drawings and samples.
- 3.9.5.5 *District's Property*. All Submittals, Shop Drawings, computer disks, BIM modeling information, clash checks, schedules, annotated Specifications, samples and other Submittals shall become the District's property upon receipt by the District or Architect.

## 3.9.6 <u>Schedule Requirements for Submittals</u>

Contractor shall obtain and shall submit all required Submittals (i.e. Shop Drawings, Deferred Approvals, Samples, etc.), in accordance with Contractor's "Schedule for Submission of Shop Drawings and Samples" as required in the scheduling portion of the General Conditions at Articles 8 and the Specifications (as long as the Specifications do not conflict with General Conditions. In the case of conflict, the conflicting provision shall be controlled by the General Conditions and the remaining Specifications sections shall be interpreted as if the general conditions language is inserted) with such promptness as to cause no delay in its own Work or in that of any other contractor or subcontractor but in no event later than thirty five (35) days after the Notice to Proceed is issued except in the specific cases noted as an exception under Article 3.7.2.1. No extensions of time will be granted to Contractor or any Subcontractor because of its failure to have Shop Drawings and samples submitted in accordance with Division 1 and the Schedule. Each Subcontractor shall submit all Shop Drawings, samples, and manufacturer's descriptive data for the review of the District, the Contractor, and the Architect through the Contractor.

3.9.6.1 *Consideration of Schedule.* Contractor has considered lead times, DSA or other agency governmental review times, Architect or Engineer review times, manufacturing seasons, and specific long lead procurement concerns for all submittals for the Project.

## 3.9.7 <u>General Submittal Requirements</u>

- 3.9.7.1 Contractor Submittal Representations and Coordination. By submitting Shop Drawings, Product Data, samples, etc., the Contractor represents that it has determined and verified all materials, field measurements, catalog numbers, related field construction criteria, and other relevant data in connection with each such submission, and that it has checked, verified, and coordinated the information contained within such Submittals with the requirements of the Work and of the Contract Documents, including the construction schedule.
- 3.9.7.2 *Contractor Coordination.* Contractor shall stamp, sign, and date each Submittal indicating its representation that the Submittal meets all of the requirements of the Contract Documents and evidence Contractor's review through execution of the following stamp to be placed on each Shop Drawings:

"[Contractor] has reviewed and approved the field dimensions and the construction criteria, and has also made written notation regarding any information in the Shop Drawings and Submittals that does not conform to the Contract Documents. This Shop Drawing or Submittal has been coordinated with all other Shop Drawings and Submittals received to date by me as Contractor and this duty of coordination has not been delegated to Subcontractors, material suppliers, the Architect, or the Engineers on this Project.

Signature of Contractor and date

3.9.7.3 *No Deviation from Contract Documents*. The submission of the Shop Drawings, Product Data, samples, etc., shall not deviate from the *requirements* of the Contract Documents including detailing and design intent which is specifically outlined in Contract Documents except as specifically authorized by the Architect or through an accepted substitution pursuant to Article 3.10.4. All deviations from the Contract Documents shall be narratively described in a transmittal accompanying the

Shop Drawings. However, Shop Drawings shall not be used as a means of requesting a substitution, the procedure for which is defined in Article 3.10.4, "Substitutions."

3.9.7.4 Contractor Responsibility for Shop Drawings Conformance to Contract Documents. Review by District and Architect shall not relieve the Contractor or any Subcontractor from its responsibility in preparing and submitting proper Shop Drawings in accordance with the Contract Documents.

3.9.7.5 *Incomplete Submittals*. Any submission, which in Architect's opinion is incomplete, contains errors, or has been checked superficially, will be returned not reviewed by the Architect for resubmission by the Contractor. Refer to Submittal Procedures of the Specifications for additional information. The Contractor shall be responsible for any related delays and shall not be the basis for any Claim.

3.9.7.6 Shop Drawings and Submittals Shall Not Be Used as a Method to Make a Substitution. Shop Drawings and Submittals shall not be used as a means of requesting a substitution or to make changes in the Contract Documents. If changes are made to the Contract Documents through the Shop Drawings, the Architect shall have the right to reject the Submittal. If the Architect does not note the deviation from the approved Plans and Specifications, the Contractor is still responsible for the change and the Architect or the District may require the Shop Drawings be revised to properly reflect the approved Contract Documents. The Architect or District may also require that the Contractor bear all costs under Article 4.5 and consequential damages associated with a CCD to revise Plans and Specifications to accommodate the deviation from approved Plans and Specifications.

3.9.7.7 Extent of Review. In reviewing Shop Drawings, the Architect will not verify dimensions and field conditions. The Architect will review and approve Shop Drawings, Product Data, samples, etc., for aesthetics and for conformance with the design concept of the Work and the information in the Contract Documents. The Architect's review shall neither be construed as a complete check which relieves the Contractor, Subcontractor, manufacturer, fabricator, or supplier from responsibility for any deficiency that may exist or from any departures or deviations from the requirements of the Contract Documents unless the Contractor has, in writing, called the Architect's attention to the deviations at the time of submission. The Architect's review shall not relieve the Contractor or Subcontractors from responsibility for errors of any sort in Shop Drawings or schedules, for proper fitting of the Work, coordination of the differing Subcontractor trades and Shop Drawings and Work which is not indicated on the Shop Drawings at the time of submission of Shop Drawings. Contractor and Subcontractors shall be solely responsible for any quantities which may be shown on the Submittals or Contract Documents.

### 3.10 SUBSTITUTIONS

3.10.1 <u>Definition</u>

A Substitution is a change in product, material, equipment, or method of construction from those required by the Construction Documents proposed by the Contractor. For this Project, a Substitution is subject to the filing of a Construction Substitution Request Form at the time of bid and meeting the requirements of this Article.

### 3.10.2 One Product Specified

Unless the Specifications state that no substitution is permitted, whenever the Contract Documents indicate any specific article, device, equipment, product, material, fixture, patented process, form, method, or type of construction or any specific name, make, trade name, or catalog number, with or without the words "or equal," such specification shall be deemed to be used for the purpose of facilitating description of the material, process, or article desired and shall be deemed to be followed by the words "or equal." Subject to the requirements of properly submitting a Substitution Request for as Addressed in Article 3.10.4, the Contractor may, unless otherwise stated, offer any material, process, article, etc., which shall be materially equal or better in every respect to that so indicated or specified ("Specified Item") and will completely accomplish the purpose of the Contract Documents.

## 3.10.3 <u>Products Specified Which Are Commercially Unavailable</u>

If the Contractor fails to make a request for substitutions for products, prior to the submission of its bid, and such products subsequently become commercially unavailable, the Contractor may request a substitution for such commercially unavailable item. The decision to grant this request is solely at the District's discretion. The written approval of the District, consistent with the procedure for Change Orders, shall be required for the use of a proposed substitute material. The District may condition its approval of the substitution upon the delivery to District of an extended warranty or other assurances of adequate performance of the substitution as well as an equitable deduction in the Contract Price should the substituted item cost less than the Specified Item. All risks of delay due the approval of a requested substitution by the DSA, or any other governmental agency having jurisdiction, shall be on the requesting party. All additional costs, DSA review costs, all procurement and construction delays, and all costs for review by the Architect or its consultants shall be the responsibility of the Contractor and will be deducted from Contractor's pay request.

## 3.10.4 <u>Substitution Request Form</u>

Requests for substitutions of products, materials, or processes in place of a Specified Item must be in writing on the District's Substitution Request Form ("Request Form") at the time of submitting bids to the District, except as provided for in Article 3.10.3.

The Request Form must be accompanied by evidence as to whether the proposed substitution:

- a. Is equal in quality/service/ability to the Specified Item;
- b. Will entail no changes in detail, construction, and scheduling of related work;
- c. Will be acceptable in consideration of the required design and artistic effect;
- d. Will provide no cost disadvantage to the District;

- e. Will require no excessive or more expensive maintenance, including adequacy and availability of replacement parts; and
- f. Will required no change of the construction schedule.

In completing the Request Form, the bidder must state, with respect to each requested substitution, whether the bidder will agree to provide the Specified Item in the event that the District denies the bidder's request for such requested substitution. In the event that the bidder has agreed in the Request Form to provide the Specified Item and the District denies the bidder's requested substitution for a Specified Item, the bidder shall provide the Specified Item without any additional cost or charge to the District.

After bids are opened, the apparent lowest bidder shall provide, within five (5) days of opening such bids, any and all Drawing, Specifications, samples, performance data, calculations, and other information, as may be required to assist the Architect, CM and the District in determining whether the proposed substitution is acceptable. The burden of establishing these facts shall be upon the bidder.

After the District's receipt of such evidence by the bidder, the District will make its final decision as to whether the bidder's request for substitution for any Specified Items will be granted. The decision as to whether a proposed request for substitution is equal to a Specified Item shall be at the sole discretion of the District. Any request for substitution that is granted by the District shall be documented and processed though a Change Order. Contractor must submit a complete Submittal of the requested substitution and a Shop Drawing showing configuration, dimensions, and other critical information associated with the substitution that meets the requirements of Article 3.9. The District may condition its approval of any substitution upon delivery to the District of an extended warranty or other assurances of adequate performance of the substitution. Any and all risks of delay due to approval by the DSA or any other governmental agency having jurisdiction shall be on the bidder.

If the Architect and District accept a proposed substitution, the Contractor agrees to pay for all DSA review costs, engineering and design services, including, without limitation, compensation to the Architect and affected engineers for their required time to process such substitution through the Division of the State Architect, if required, and to make all changes and adjustments in materials or the work of all trades directly or indirectly affected by the substituted item or items at no cost to the District.

### 3.10.5 <u>Substitution Requests After Bid</u>

The District, in its sole discretion, may accept a request for substitution by the Contractor or may request Contractor substitute a specified item. Any substitutions requested after bids are opened shall be subject to the same conditions and requirements set forth in Article 3.10.4 above. If any substitutions, that in the District or Architect's determination, results in a credit to the District, the credit amount shall be agreed upon in writing, otherwise, the request for substitution shall be deemed denied.

### 3.11 INTEGRATION OF WORK

#### 3.11.1 Scope

The Contractor shall be responsible for cutting, fitting, or patching to complete the Work and to make all parts fit together properly. Contractor shall be responsible for ensuring that all trades are coordinated and scheduled so as to ensure the timely and proper execution of the work. When modifying existing work or installing new Work adjacent to existing work, Contractor shall match, as closely as conditions of Site and materials will allow, the finishes, textures, and colors of the original work, refinishing existing work at no additional cost to District. All cost caused by defective or ill-timed work shall be borne by Contractor. Contractor shall be solely responsible for protecting existing work on adjacent properties and shall obtain all required permits for shoring and excavations near property lines.

### 3.11.2 Structural Members

New or existing structural members and elements, including reinforcing bars and seismic bracing, shall not be cut, bored, or drilled except by written authority of the Architect. Work done contrary to such authority is at the Contractor's risk and subject to replacement at its own expense without reimbursement under the Contract. Schedule delays resulting from Agency approvals for unauthorized work shall be the Contractor's responsibility.

### 3.11.3 <u>Subsequent Removal</u>

Permission to patch any areas or items of the Work shall not constitute a waiver of the District's or the Architect's right to require complete removal and replacement of the areas of items of the Work if, in the opinion of the Architect or the District, the patching does not satisfactorily restore quality and appearance of the Work or does not otherwise conform to the Contract Documents.

### 3.12 <u>CLEANING UP</u>

#### 3.12.1 Contractor's Responsibility to Clean Up

Contractor at all times shall keep premises free from debris such as waste, dust, excess water, storm water runoffs, rubbish, and excess materials and equipment. Contractor shall not leave debris under, in, or about the premises, but shall promptly remove same from the premises and dispose of it in a lawful manner. Disposal receipts or dump tickets shall be furnished to the Architect within five (5) days of request.

Contractor shall remove rubbish and debris resulting from the Work on a daily basis. Contractor shall maintain the structures and Site in a clean and orderly condition at all times until acceptance of the Project by the District. Contractor shall keep its access driveways and adjacent streets, sidewalks, gutters and drains free of rubbish, debris and excess water by cleaning and removal each day. All concrete, sidewalks, and paths of travel shall be broom cleaned daily.

## 3.12.2 <u>General Final Clean-Up</u>

Upon completion of Work, Contractor shall employ experience workers or professional cleaners for final cleaning. Contractor shall clean each surface to the condition expected in a normal, commercial, building cleaning and maintenance program including, but not limited to, the performed of the following:

a. Clean interior and exterior of buildings, including fixtures, equipment, walls, floors, ceilings, roofs, window sills and ledges, horizontal projections, and any

areas where debris has collected, so surfaces are free from foreign material or discoloration:

- b. Clean the Project site. The grounds should be cleared of any Contractor equipment, raked clean of debris and trash removed. Sweep paved areas broom clean;
- c. Repair or replace any damaged materials. Replace any chipped or broken glass;
- d. Remove any and all stains;
- e. Remove labels that aren't permanent labels;
- f. Clean and polish all glass, plumbing fixtures, equipment, finish hardware and similar finish surfaces. Remove any glazing compounds;
- g. Remove temporary utilities, fencing, barricades, planking, sanitary facilities and similar temporary facilities from Site;
- h. Remove temporary film that remains on any hardware, doors or other surfaces; and
- i. Seal the bottom and tops of all doors.

## 3.12.3 Special Clean-Up.

In addition to the general cleaning, the following special cleaning shall be done at the completion of the Work in accordance with the Specifications including, but not limited to:

- a. Remove putty stains from glazing, then wash and polish glazing;
- b. Remove marks, stains, fingerprints and other soil or dirt from painted, stained or decorated work;
- c. Remove temporary protection and clean and polish floors and waxed surfaces;
- d. Clean and polish hardware and plumbing trim; remove stains, dust, dirt, plaster and paint;
- e. Wipe surfaces of mechanical and electrical equipment;
- f. Remove spots, soil, plaster and paint from tile work, and wash tile;
- g. Clean all fixtures and equipment, remove excess lubrication, clean light fixtures and lamps, polish metal surfaces;
- h. Vacuum-clean carpeted surfaces; and
- i. Remove debris from roofs, down spout and drainage system.

### 3.12.4 Failure to Cleanup

If the Contractor fails to clean up as provided in the Contract Documents, the District may do so, and the cost thereof shall be the responsibility of the Contractor pursuant to Article 2.2 and seek a Deductive Change Order.

## 3.13 ACCESS TO WORK

The Contractor shall provide the District, the Architect, Engineers and the Inspector of Record, access to the Work in preparation and progress wherever located. Contractor shall provide safe and proper facilities for such access so that District's representatives may perform their functions.

CONTRACTOR IS AWARE THAT THIS CONTRACT MAY BE SPLIT INTO SEVERAL PHASES AS ADDRESSED IN ARTICLE 6.

# 3.13.1 <u>Special Inspection, Inspections or Tests Out of State, Out of Country or Remote from Project</u>

If Contractor has a Subcontractor or supplier that requires in plant or special inspections or inspections or tests that are out of the country, out of the state, or a distance of more than 200 miles from the Project site, the Special Inspector or Inspector shall be provided access so the special inspection or inspection may occur in the remote location. In some cases, the DSA Inspector may also require access in addition to Special Inspectors and individuals performing tests. Inspections/tests shall occur during normal work hours. (See also Article 4.3.6)

## 3.14 ROYALTIES AND PATENTS

## 3.14.1 Payment and Indemnity for Infringement

Contractor shall hold and save the District and its officers, agents, and employees, the Construction Manager, the Architect, and the Architect's consultants harmless from liability of any nature or kind, including cost and expense, for or on account of any patented or unpatented invention, process, article, or appliance manufactured or used in the performance of the Contract, including its use by the District, unless otherwise specifically provided in the Contract Documents, and unless such liability arises from the sole negligence, or active negligence, or willful misconduct of the District, the Architect, or the Architect's consultants.

### 3.14.2 Review

The review by the Architect of any method of construction, invention, appliance, process, article, device, or material of any kind shall be for its adequacy for the Work and shall not be an approval for the use by the Contractor in violation of any patent or other rights of any person or entity.

## 3.15 <u>INDEMNIFICATION</u>

### 3.15.1 Contractor

See Agreement Form. Contractor shall ensure that its contract with each of its Subcontractors contains provisions requiring the Subcontractors to defend, indemnify and hold harmless the District, Architect, Construction Manager, Inspector, the State of California to a minimum level as set forth in this Article and consistent with the indemnity and hold harmless language in the Agreement Form.

The Contractor's and Subcontractors' obligation to defend, indemnify and hold harmless the District, Architect, Inspector, the State of California and their officers, employees, agents and independent contractors hereunder shall include, without limitation, any and all claims, damages, and costs for the following: (1) any damages or injury to or death of any person, and damage or injury to, loss (including theft), or loss of use of, any property; (2) breach of any warranty, express or implied; (3) failure of the Contractor or Subcontractors to comply with any applicable governmental law, rule, regulation, or other requirement; (4) products installed in or used in connection with the Work; and (5) any claims of violation of the Americans with Disabilities Act ("ADA")

### 3.16 SUBMISSION OF DAILY REPORTS

### 3.16.1 General

By 10:00 a.m. on the following business day, the Contractor shall submit a Daily Report to the Inspector and copy the Architect for the previous day's Work. If there is a Construction Manager, the original Daily Report is to be provided to the Construction Manager and copies sent to the Architect and the Inspector. Daily Reports shall be prepared on forms approved by the District, together with applicable delivery tickets, listing all labor, materials, and equipment involved for that day. The District reserves the right to note inconsistencies or inaccuracies in the Daily Reports. In such cases, pertinent notes shall be entered by each party to explain points which cannot be resolved that day. Each party shall retain a signed copy of the report. Daily Reports by Subcontractors or others shall be submitted through the Contractor.

### 3.16.2 Labor

The Daily Report shall show names of workers, classifications, hours worked and hourly rate. The locations where work occurred shall also be identified in the Daily Report. Project superintendent expenses are not allowed.

#### 3.16.3 Materials

The Daily Report required shall describe and list quantities of materials used and unit costs.

### 3.16.4 Equipment

The Daily Report required shall show type of equipment, size, identification number, and hours of operation, including loading and transportation, if applicable, and hourly/daily cost. Move-on and move-off fees shall be noted.

### 3.16.5 Other Services and Expenditures

Other services and expenditures shall be described in the Daily Report in detail as the District requires.

### 3.16.6 Failure to Submit Daily Report

If Contractor does not submit its Daily Report by 10 am the next business day, the Inspector of Record shall prepare a Daily Report addressing each of the above items. The cost for the Inspector's services to prepare the Daily Report shall be addressed through a Deductive Change Order under Article 7.7.4.

### 3.17 AS-BUILT DRAWINGS AND ANNOTATED SPECIFICATIONS

Throughout the duration of the Project, Contractor shall maintain on a current basis an accurate and complete set of As-Built Drawings (and Annotated Specifications) clearly showing all changes, revisions to Specifications and substitutions during construction, including, without limitation, field changes and the final location of all electrical and mechanical equipment, utility lines, ducts, outlets, structural members, walls, partitions, and other significant features. In case a Specification allows Contractor to elect one of several brands, makes, or types of material or equipment, the annotations shall show which of the allowable items the Contractor has furnished. The Contractor will update the As-Built Drawings and Annotated Specifications as often as necessary to keep them current, but no less often than weekly.

Contractor shall update As-Built Drawings with complete information on an area of Work at or near the time when the Work is being performed and prior to any DSA 152 sign off and prior to any Work being covered.

The As-Built Drawings and Annotated Specifications shall be kept at the Site and available for review and inspection by the District and the Architect. Failure to maintain and update the As-Built Drawings is a basis to withhold Progress Payments pursuant to Article 9.6.

## 3.17.1 <u>Upon Beneficial Occupancy</u>

Contractor shall obtain and pay for reproducible Plans upon Beneficial Occupancy. Contractor shall deliver Plans to District Representative (Construction Manager if one is hired for the Project).

## 3.17.2 <u>As-Builts at Completion of Work</u>

Upon completion of the Work and prior to and as a condition precedent to Application for Retention Payment, the Contractor will provide one neatly prepared and complete set of As-Built Drawings and Annotated Specifications to the District. Contractor shall certify the As-Builts as a complete and accurate reflection of the actual construction conditions of the Work by affixing a stamp indicating the Drawings are As-Builts and certifying accuracy on the final set of As-Builts. Failure to deliver a complete As-Built set of Drawings may result in significant withholdings to ensure Work is properly documented. (See Article 9.9.2)

### 3.17.3 Log of Control and Survey Documentation

Contractor shall complete and maintain an accurate log or all control and survey documentation for the Project as the Work progresses. All reference and control points shall be recorded on the As-Built Drawings. The basis of elevations shall be one of the established benchmarks that must be maintained on the As-Builts.

## 3.17.4 <u>Record Coordinates for Key Items</u>

Contractor shall record, by coordinates, all utilities on-site with top of pipe elevations, major grade and alignment changes, rim, grate or top of curb and flow line elevations of all drainage structures and sewer manholes. Contractor shall update record information at or near the time when work is occurring in an area and prior to DSA 152 sign off on any category of Work and prior to covering the Work.

### 3.17.5 BIM As-Built Drawings

If BIM is utilized for the Project, then an electronic version of such As-Built Drawings and Annotated Specifications will be delivered to District (in an acceptable format to District).

### 3.18 EQUIPMENT MANUALS

Contractor shall obtain and furnish three (3) complete sets of manuals containing the manufacturers' instructions for maintenance and operation of each item of equipment and apparatus furnished under the Contract Documents and any additional data specifically requested under the various sections of the Specifications for each division of the Work. The manuals shall be arranged in logical, sequential order, labeled, indexed, and placed in three-ring binders. At the completion of its Work, the Contractor shall certify, by endorsement thereon, that each of the manuals is complete, accurate, and covers all of its Work. Prior to submittal of Contractor's Application for Retention Payment, and as a further condition to its approval by the Architect, each Subcontractor shall deliver the manuals, arranged in logical, sequential order, labeled, indexed, endorsed, and placed in three-ring binders, to the Contractor, who shall assemble these manuals for all divisions of the Work, review them for completeness, and submit them to the District through the Architect.

### 3.19 <u>DIR REGISTRATION</u>

Strict compliance with all DIR registration requirements in accordance with Labor Code sections 1725.5 and 1771.1 is a material obligation of the Contractor and all of its subcontractors (of any tier) under the Contract Documents. The foregoing includes, without limitation, compliance with DIR registration requirements at all times during performance of the Work by the Contractor and all of its subcontractors of any tier. The failure of the Contractor and all subcontractors of any tier to be properly registered with DIR at all times during performance of the Work is a material breach of the Contract and subject to termination for cause.

An affirmative and ongoing obligation of the Contractor under the Contract Documents is the verification that all subcontractors of any tier are at all times during performance of the Work are in full and strict compliance with the DIR registration requirements. The Contractor shall not permit or allow any subcontractor of any tier to perform any Work without the Contractor's verification that all subcontractors are in full and strict compliance with the DIR registration requirements. Any subcontractors of any tier not properly registered with DIR shall be substituted in accordance with Labor Code section 1771.1. Contractor or its subcontractors of any tier shall not be entitled to any additional costs or time arising from or in any way related to compliance with the DIR registration requirements.

## ARTICLE 4 ADMINISTRATION OF THE CONTRACT AND CLAIMS

### 4.1 <u>ARCHITECT</u>

### 4.1.1 Replacement of Architect

In the case of the termination of the Architect, the District may appoint an Architect or another construction professional or may perform such functions with its own licensed professional personnel. The status of the replacement Architect under the Contract Documents shall be the same as that of the former Architect.

### 4.2 ARCHITECT'S ADMINISTRATION OF THE CONTRACT

### 4.2.1 Status

Pursuant to Titles 2 of the California Code of Regulations and as required pursuant to the Field Act, Education Code 17280 et seq., the Architect will provide administration of the Contract Documents and the Work, and will be the District's representative during construction, as well as during the one (1) year period following the commencement of any warranties. The Architect will have authority to act on behalf of the District only to the extent provided in the Contract Documents.

## 4.2.2 <u>Site Visits</u>

The Architect will visit the Site at intervals necessary in the judgment of the Architect to become generally familiar with the progress and quality of the Work and to determine in general if the Work is being performed in accordance with the Contract Documents and as otherwise required by DSA.

## 4.2.3 <u>Limitations of Construction Responsibility</u>

The Architect, District and CM shall not have control over, charge of, or be responsible for construction means, methods, techniques, schedules, sequences or procedures, fabrication, procurement, shipment, delivery, receipt, installation, or for safety precautions and programs in connection with the Work, since these are solely the Contractor's responsibility under the Contract Documents. The Architect, District and CM shall not be responsible for the Contractor's, Subcontractors', material or equipment suppliers', or any other person's schedules or failure to carry out the Work in accordance with the Contract Documents. The Architect, District and CM shall not have control over or charge of acts or omissions of the Contractor, Subcontractors, their agents or employees, or any other persons or entities performing or supplying portions of the Work. The Contractor shall not be relieved of obligations to perform the Work in accordance with the Contract Documents either by activities or duties of the Architect, District or CM in the Architect, District or CM's administration of the Contract Documents, or by tests, inspections, or approvals required or performed by persons other than the Contractor.

### 4.2.4 Communications Facilitating Contract Administration

Except where a CM is on the Project, or as otherwise provided in the Contract Documents or when direct communications are warranted by special circumstances, the District and the Contractor shall communicate through the Architect. In the cases where a CM is hired for the Project, all communication shall be through the CM (unless otherwise directed) with copies to the District, Architect

and Inspector. Where direct communication is necessary between the District and the Contractor, the District's communication shall be through the District's authorized designated person. The Architect and CM shall be promptly informed, and shall receive copies of all written communications. Contractor shall not rely upon any communications from the District that is not from the District's Representative. Communications by and with the Architect's consultants shall be through the Architect. Communications by and with Subcontractors and material or equipment suppliers shall be through the Contractor. In the case where a CM is hired for the Project, the CM shall be the main point of contact for communication of information. Copies should be sent to the Architect, District Representative and Inspector.

### 4.2.5 Payment Applications

The Architect will review and make recommendations to the District regarding the amounts due the Contractor on the Certificates for Payment pursuant to Article 9.3.4 and subject to the Inspector's review, (CM review, if applicable) and Architect's observation. This review of Payment Applications is sometimes called a "Pencil Draft." Return of a Pencil Draft shall constitute the District's dispute of the Payment Application that has been submitted. Contractor shall promptly respond to Pencil Drafts or Contractor's Payment Applications may be delayed. Contractor's failure to promptly respond to a Pencil Draft shall qualify as a delay in the Prompt Payment of a Request for Payment or Request for Retention.

### 4.2.6 Rejection of Work

In addition to the rights, duties, and obligations of the Inspector under this Article, the Architect may recommend to the District that the District reject Work which does not conform to the Contract Documents. Whenever the Architect considers it necessary or advisable to achieve the intent of the Contract Documents, the Architect (and/or CM) may recommend to the District that the District require additional inspection or testing of the Work in accordance with Article 13.5, whether or not such Work is Fabricated, installed, or completed. District may have Non-conforming Work removed and replaced pursuant to Article 9.7. However, neither this authority of the Architect (or CM) nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Architect (or CM) to the Contractor, Subcontractors, material and equipment suppliers, their agents or employees, or other persons performing portions of the Work.

Contractor shall, without charge, replace or correct Work found by the District to not be in conformance to Contract requirements. Contractor shall promptly segregate and remove rejected materials from the Project site.

This section is does not address a Notice of Non-Compliance and the remedies associated with a Notice of Non-Compliance which are addressed at Article 7.1.2

### 4.2.7 Warranties upon Completion

The Architect (and where applicable CM), in conjunction with the Inspector will conduct field reviews of the Work to determine the date of Substantial Completion and of Final Completion, shall receive and forward to the District for the District's review written warranties and related documents required by the Contract and assembled by the Contractor, and will issue a final Certificate for Payment when the Architect believes the Work has been completed in compliance with the requirements of the Contract Documents (See Article 9.11 for Close-Out). The handling by the Architect (or where applicable CM) of such warranties, maintenance manuals, or similar documents shall not diminish or transfer to the Architect any responsibilities or liabilities required by the Contract Documents of the Contractor or other entities, parties, or persons performing or supplying the Work.

On some Projects, the District will take a phased occupancy of the Project. In those cases, the District may commence the running of warranties on the buildings, or phases that are accepted after Punch List is completed and the District has accepted Completion of the separate phase. A separate Notice of Completion may be filed for the separate building or phase of work and warranties shall commence for the separate phase only to the extent that warranties do not require coordination or connection to other buildings or other parts of the site and only if the warranted item is completed to its entirety in the segregated building or phased area.

If written warranties are not provided at the time the Punch List is nearing completion, Architect (with recommendations from the CM and Inspector) shall determine the dollar value of the warranties and shall make recommendation for withholdings necessary to effectuate the transfer of such warranties to the District for future use as part of the Punch List for the Project pursuant to Article 9.6.

Warranties are not commenced through utilizing of equipment for testing and operation as necessary to acclimate buildings or where necessary to test systems.

## 4.2.8 <u>Interpretation</u>

The Architect will interpret and decide matters concerning performance and requirements of the Contract Documents. Architect shall make clarifications as necessary to interpret the Contract Documents.

## 4.3 **PROJECT INSPECTOR**

### 4.3.1 General

One or more Project Inspectors employed by the District and approved by the Division of the State Architect will be assigned to the Work in accordance with the requirements of Title 24 of the California Code of Regulations. The Inspector(s) duties are as specifically defined in Title 24 Section 4-333 and 4-342 and in DSA IR A-8.

### 4.3.2 Inspector's Duties and DSA Noted Timelines for Inspection

All Work shall be under the observation of the Inspector. Contractor shall establish a protocol for requesting inspection with Inspector so as to not delay the Work and provide adequate time for the Inspector to perform inspection. If such a protocol is not established ahead of time, Inspector may utilize the time criteria set by Title 24 of 48 hours in advance of submitting form DSA 156 for each new area. The Inspector shall have free access to any or all parts of the Work at any time. The Contractor shall furnish the Inspector such information as may be necessary to keep the Inspector fully informed regarding progress and manner of Work and character of materials. Such observations shall not, in any way, relieve the Contractor from responsibility for full compliance with all terms and conditions of the Contract, or be construed to lessen to any degree the Contractor's responsibility for providing efficient and capable superintendence. The Inspector is not authorized to make changes in the Drawings or Specifications nor shall the Inspector's approval of the Work and methods relieve the Contractor of responsibility for the correction of subsequently discovered defects, or from its obligation to comply with the Contract Documents.

Inspector shall electronically post DSA required documents on the DSA electronic posting website. It is the Contractor's responsibility to determine the status of posting and determine if all the

criteria for sign off of a category of Work on the Project Inspection Card (Form DSA 152) as defined more thoroughly in the most current version of the DSA 152 manual posted on the DSA website.

Inspector may collaborate with Contractor about approval of areas that may be constructed and approved incrementally under the DSA 152 card pursuant to the guidelines of PR-13 at Article 1.17. Inspector shall work with Contractor to present incremental approval proposals to DSA.

## 4.3.3 <u>Inspector's Authority to Reject or Stop Work</u>

The Inspector shall have the authority to reject Work whenever provisions of the Contract Documents are not being complied with, and Contractor shall instruct its Subcontractors and employees accordingly. In addition, the Inspector may stop any Work that poses a probable risk of harm to persons or property. The Contractor shall instruct its employees, Subcontractors, material and equipment suppliers, etc., accordingly. The absence of any Stop Work Order or rejection of any portion of the Work shall not relieve the Contractor from any of its obligations pursuant to the Contract Documents.

## 4.3.4 <u>Inspector's Facilities</u>

Within seven (7) days after the notice to proceed, the Contractor shall provide the Inspector with the temporary facilities as required. More specific requirements for the Inspector facilities may be further described under Division 1 of the Specifications.

### 4.3.5 Testing Times

The District will provide inspection and testing at its cost during the normal eight (8) hour day Monday through Friday (except holidays). Work by the Contractor outside of the normal eight (8) hour day shall constitute an authorization from the Contractor to the District to provide inspection and testing as required outside of the normal eight (8) hour day. Contractor shall provide adequate time for inspections so as to not delay the Work. An advanced timing protocol may be established pursuant to Article 4.3.2. If the Contractor is behind Schedule, then it is incumbent on the Contractor to provide advance forecast through look ahead of the anticipated date for inspection so the Inspector may plan their activities so as to not delay the Project. Contractor shall reimburse District for any additional costs associated with inspection and testing (including re-inspection and re-testing) outside the normal eight-hour day and for any retests caused by the Contractor.

It is the Contractor's responsibility to request special inspections with sufficient time so all testing may be timely completed and posted so work may proceed and the Inspector's signature is attached to the Project Inspection Card (Form 152). Specifically, timely request for special inspection under the DSA Verified Report Forms 291 (laboratory), DSA Verified Report Form 292 (Special Inspection), and DSA Verified Report 293 (geotechnical) since DSA requirements under PR 13-01 specifically gives the Special Inspections 14 days to post to the DSA website. Failure to plan and pay (if applicable) for quicker delivery of Special Inspections may be counted as Float, but is not considered Governmental Delay Float under Article 8.1.4.

# 4.3.6 <u>Special Inspections, Inspections or Tests Out of State, Out of Country or Remote from Project</u>

If Contractor has a Subcontractor or supplier that requires in plant or special inspections, inspections or tests that are out of the country, out of the state or a distance of more than 200 miles from the Project Site, the District shall provide the Special Inspector or individual performing tests time for

inspection and testing during normal work hours. Contractor, however, is responsible for the cost of travel, housing, food, out of area premiums that may be in the Inspector/Testing Agreement with District, or other expenses necessary to ensure proper inspection, special inspection or testing is provided by a DSA Certified Inspector, Special Inspector, or individual performing tests. In some cases, all three (DSA Inspector, Special Inspector, and Tester) may be required. In addition, if the DSA Certified Inspector, Special Inspector, or individual performing test has contractual travel clauses or special rates for out of town inspection, Contractor is responsible for all costs associated with the contractual travel costs in addition to all other costs. Arrangements for inspection and/or testing shall be made far enough in advance so as to not delay the Work.

### 4.4 STOP WORK ORDER

DSA may issue a Stop Work Order, or an Order to Comply, when either (1) the Work proceeds without DSA approval; (2) the Work proceeds without a DSA Inspector of Record, or (3) where DSA determines that the Work is not being performed in accordance with applicable rules and regulations, and would compromise the structural integrity of the Project or would endanger lives. If a Stop Work Order is issued, the Work in the affected area shall cease until DSA withdraws the Stop Work Order. Pursuant to Education Code section 17307.5(b), the District shall not be held liable in any action filed against the District for any delays caused by compliance with the Stop Work Order, except to the extent that an error or omission by the District is the basis for the issuance of the Stop Work Order.

Examples of Stop Work Orders that may be issued by DSA include DSA Bulletin 07-04 and Policy 10-01, the installation of automatic fire sprinkler systems without approved Plans, covering Work that has not been approved by Inspector on DSA Project Inspection Card (Form 152).

# 4.5 <u>RESPONSIBILITY FOR ADDITIONAL CHARGES INCURRED BY THE DISTRICT</u> FOR PROFESSIONAL SERVICES

If at any time prior to the completion of the requirements under the Contract Documents, the District is required to provide or secure additional professional services (including CM, Inspection, Architect, Engineering and Special Consultant Services) for any reason by any act of the Contractor, the District may seek a Deductive Change Order for any costs incurred for any such additional services, which costs shall be deducted from the next progress payment. A Deductive Change Order shall be independent from any other District remedies and shall not be considered a waiver of any District rights or remedies. If payments then or thereafter due to the Contractor are not sufficient to cover such amounts, the Contractor shall pay the difference to the District. Additional services shall include, but shall not be limited to, the following:

- a. Services made necessary by the default of the Contractor (Article 14 or Article 2.2).
- b. Services made necessary due to the defects or deficiencies in the Work of the Contractor (Article 2.2 and Article 9.6).
- c. Spurious or frivolous RFI's issued that do not conform to the requirements of Article 7.4. Issuance of the same RFI after receiving an answer from the Architect or Engineer
- d. Review of Schedules that are provided by Contractor that do not Conform with the Requirements of Article 8.

- e. Preparation of a CCD or ICD to correct a Contractor Deficiency, or Contractor Caused Notice of Non-Compliance (See Article 7.3).
- f. Review of Incomplete Shop Drawings or Submittals, including the submission of Piecemeal Shop Drawings or Submittals unless piecemeal Submittals are specifically agreed upon by District (See Article 3.9)
- g. Services required by failure of the Contractor to perform according to any provision of the Contract Documents.
- h. Services in connection with evaluating substitutions of products, materials, equipment, Subcontractors' proposed by the Contractor, and making subsequent revisions to Drawings, Specifications, obtaining DSA approvals, DSA costs for review of CCD's, other governmental agency review costs, and providing other documentation required (except for the situation where the specified item is no longer manufactured or available). (See Article 3.10)
- i. Services for evaluating and processing Claims or Disputes submitted by the Contractor in connection with the Work outside the established Change Order process.
- j. Services required by the failure of the Contractor to prosecute the Work in a timely manner in compliance within the specified time of completion.
- k. Services in conjunction with the testing, adjusting, balancing and start-up of equipment other than the normal amount customarily associated for the type of Work involved.
- 1. Services in conjunction with more than one (1) re-review of Submittals of Shop Drawings, Product Data, samples, RFI's etc.

## 4.6 **DISPUTES AND CLAIMS**

### 4.6.1 <u>Decision of Architect</u>

"Disputes" or "Claims" as defined in Article 4.6.9.1 between District and Contractor involving money or time, including those alleging an error or omission by the Architect shall be referred initially to the Architect for action as provided in Article 4.6.2 within ten (10) days after Contractor's Article 7 request for Change is denied. If there is a CM, the CM shall receive the Dispute and may review and also assemble opinions and documents to assist the Architect. A decision by the Architect, as provided in Article 4.6.5, shall be required as a condition precedent to proceeding with remedies set forth in Article 4.6.9 as to all such matters arising prior to the date Retention Payment Application is due, regardless of whether such matters relate to execution and progress of the Work, or the extent to which the Work has reached Final Completion.

The condition precedent of an Architect decision shall be waived if: (1) the position of Architect is vacant; (2) the Architect has failed to take action required under Article 4.6.5 within the time periods required therein; or (3) the Dispute or Claim relates to a stop notice claim not arising from any extra Change Order or Immediate Change Directive for which approval has not been provided.

### 4.6.2 <u>Architect's Review</u>

The Architect (and CM) will review the Dispute and take one or more of the following preliminary actions upon receipt of a Dispute: (1) request additional supporting data from the claimant; (2) submit a schedule to the parties indicating when the Architect expects to take action; (3) reject the Dispute in whole or in part, stating reasons for rejection; (4) recommend approval of the Dispute; or (5) suggest a compromise. The Architect may also, but is not obligated to, notify the Surety, if any, of the nature and amount of the Dispute.

4.6.2.1 *Architectural Immunity*. Architect review of Disputes and Claims shall be impartial and meant to resolve Disputes and Claims. Pursuant to the case, <u>Huber, Hunt & Nichols, Inc. v. Moore</u> (1977) 67 Cal.App.3d 278, the Architect is provided a quasi-judicial immunity for interpreting and deciding Disputes and Claims between the District and Contractor.

### 4.6.3 Documentation if Resolved

If a Dispute has been resolved, the Architect (and/or CM) will prepare a Change Order or obtain appropriate documentation to document the terms for Board approval.

### 4.6.4 Actions if Not Resolved

If a Dispute has not been resolved and all documentation requested pursuant to Article 4.6.2 has been provided, the Contractor shall, within ten (10) days after the Architect's initial response, assemble all the documents involved in the Dispute including copies of all back-up documentation of costs and the basis for the Dispute and take one or more of the following actions: (1) modify the initial Dispute; (2) notify the Architect that the initial Dispute stands; or (3) supplement with additional supporting data and re-submit to the Architect under Article 4.6.2.

### 4.6.5 <u>Architect's Written Decision</u>

If a Dispute has not been resolved after consideration of the foregoing and of other evidence presented by the parties or requested by the Architect, the Architect (or Architect through CM) shall provide a written decision twenty (20) days after compliance with Article 4.6.4. Upon expiration of such time period, the Architect (or Architect through CM) will render to the parties its written decision relative to the Dispute, including any change in the Contract Sum or Contract Time or both. The Architect may also request reasonable additional time to complete Architect's written decision.

If the resolution of the Dispute by the Architect is not satisfactory to the Contractor and copies of all back-up documentation of costs and the basis for the Dispute is fully articulated in a package of material that is complete, the Contractor may then submit a Claim to the District under Article 4.6.9.

#### 4.6.6 Continuing Contract Performance

Pending final resolution of a Dispute or Claim, including, negotiation, mediation, arbitration, or litigation, the Contractor shall proceed diligently with performance of the Contract, and the District shall continue to make any undisputed payments in accordance with the Contract (less any withholdings or offsets). If the Claim is not resolved, Contractor agrees it will neither rescind the Contract nor stop the progress of the work, but Contractor's sole remedy shall be to submit such controversy to determination by a court of competent jurisdiction in the county where the Project is located, after the Project has been completed, and not before.

4.6.6.1 District's Option to Submit Individual Disputes to Arbitration during Claims and Disputes Process. At the District's sole option, in order to more efficiently resolve Claims during the Project and prior to the completion of the Claims Process, pursuant to Government Code section 9201, the District may submit individual Disputes or Claims for binding arbitration and Contractor agrees to the resolution of for each individual Dispute or Claim by an Arbitrator, including resolution of time and delays. If binding arbitration is utilized for individual Disputes or Claims, such resolution is full and final as to that particular Dispute or Claim. THIS INDIVIDUAL DISPUTE ARBITRATION PROCESS IS NOT AN ARBITRATION CLAUSE AND SHALL NOT BE CONSTRUED AS AN AGREEMENT TO ARBITRATE. THIS INDIVIDUAL DISPUTES ARBITRATION PROCESS IS FOR THE SOLE PURPOSE OF STREAMLINING AND RESOLVING DISPUTES OR CLAIMS DURING CONSTRUCTION AND SHALL BE REQUESTED ON SPECIFIC INDIVIDUAL ITEMS BY THE DISTRICT PRIOR TO RETENTION PAYMENT (EVEN IF THERE ARE DEDUCTIONS MADE FROM RETENTION PAYMENT) WHICH REPRESENTS THE FINAL COMPLETION OF THE PROJECT.

- a. If there is no Retention remaining on the Project, individual Disputes initiated prior to Project Final Completion shall continue until a final disposition of the Arbitration or resolution of the individual Claim or Dispute.
- b. <u>No Tolling</u>. The Arbitration process shall not toll the Disputes or Claims process under Article 4.6 or the requirement to submit Claims to Court under Article 4.6.9.5.

# 4.6.7 <u>Claims for Concealed Trenches or Excavations Greater Than Four Feet Below the Surface</u>

When any excavation or trenching extends greater than four feet below the surface or if any condition involving hazardous substances are encountered:

- a. <u>Immediately upon discovery</u>, The Contractor shall promptly, and before the following conditions are disturbed, notify the District, by telephone and in writing, of the condition except:
  - 1. If such condition is a hazardous waste condition, Contractor's bid includes removal or disposal of hazardous substances. Material that the Contractor believes may be a material that is hazardous waste, as defined in Section 25117 of the Health and Safety Code, is required to be removed to a Class I, Class II, or Class III disposal site in accordance with the provisions of existing law. In such case, the notice bulletin procedures of Article 7 apply.
  - 2. Subsurface or latent physical conditions at the Site differing from those indicated in the Drawings, Specifications, Soils Report, and from Contractor's own investigation under Article 2.1.
  - 3. Unknown physical conditions at the Site of any unusual nature, different materially from those ordinarily encountered and generally recognized as inherent in Work of the character provided for in the Contract.

- b. <u>The District shall investigate the conditions</u>, and if District finds that the conditions do materially so differ, do involve hazardous waste, and cause a decrease or increase in the Contractor's cost of, or the time required for, performance of any part of the Work shall issue a Change Order or Construction Change Document under the procedures described in the Contract.
- c. <u>In the event that a dispute</u> arises between the public entity or District and the Contractor whether the conditions materially differ, involve hazardous waste, or cause a decrease or increase in the Contractor's cost of, or time required for, performance of any part of the Work, the Contractor shall not be excused from any scheduled Completion Date provided for by the Contract, but shall proceed with all Work to be performed under the Contract. The Contractor shall retain any and all rights provided either by Contract or by law which pertain to the resolution of disputes and protests between the contracting parties.

### 4.6.8 <u>Dispute Concerning Extension of Time.</u>

If Contractor and District cannot agree upon an extension of time, whether compensable or not, then Contractor must have first completed the procedures set forth in Article 8.4. Upon completion of the procedures set forth under Article 8.4, Contractor must then comply with the requirements in this Article including those set forth under Article 4.6.9.

#### 4.6.9 Claims Procedures

Pursuant to the remedies under Public Contract Code section 9201 and Government Code section 930.2, Contractor, through execution of this Agreement, also agrees to comply with the Claims requirements of Article 4.6 to quickly and efficiently resolve Disputes and Claims. Further, to provide a level of accuracy to the records submitted, the District shall have the right to audit books and records pursuant to Article 13.11 based on the actual costs incurred and to reduce the uncertainty in resolving Disputes and Claims with limited information.

### 4.6.9.1 Procedure Applicable to All Claims

- a. <u>Definition of Claim</u>: A "Claim" is where a Dispute between the parties rises to the level where backup documentation is assembled and provided to the District as a separate demand by the Contractor for: (1) a time extension, including, without limitation, for relief from damages or penalties for delay assessed by the District under the Contract; (2) payment by the District of money or damages arising from Work done by, or on behalf of, the Contractor pursuant to the Contract and payment for which is not otherwise expressly provided for or to which the Contractor is not otherwise entitled to; or (3) an amount of payment disputed by the District. If the Claim is for damages associated with a DSA Stop Work Order, the Contractor shall not be entitled to a request for Compensation, but shall be entitled to utilize Governmental Delay Float (See Article 8.1.4.1.)
- b. <u>Filing Claim Is Not Basis to Discontinue Work</u>: The Contractor shall promptly comply with Work under the Contract or Work requested by the District even though a written Claim has been filed. The Contractor and the District shall make good faith efforts to resolve any and all Claims that may arise during the performance of the Work covered by this Contract.

- c. <u>Claim Notification</u>: The Contractor shall within seven (7) calendar days after the written decision of the Architect, or if the time period for Architect's decision has passed under Article 4.6.5, submit a notification in writing sent by registered mail or certified mail with return receipt requested, with the District (and the District's CM) stating clearly the basis for the Claim and including all relevant and required documents. If the notification is not submitted within seven (7) days after the written decision of the Architect or the passage of time under Article 4.6.5, the Contractor shall be deemed to have waived all right to assert the Claim, and the Claim shall be denied. Claims submitted after the Retention Payment date shall also be considered null and void by the District. All Claims shall be reviewed pursuant to Articles 4.6.1 through 4.6.5.
- d. The Formal Notification of Claim must be presented as follows:
  - (1) The term "Claim" must be at the top of the page in no smaller than 20-point writing.
  - (2) All documentation submitted pursuant to Article 4.6 to the Architect shall be submitted with the "Claim."
  - (3) A stack of documents, copy of all Project documents, or the submission of random documents shall not constitute an adequate reference to supporting documentation.
  - (4) Any additional or supporting documentation that Contractor believes is relevant should be submitted at this time.
- e. Reasonable Documents to Support Claim: The Contractor shall furnish reasonable documentation to support the Claim. The Contractor shall provide all written detailed documentation which supports the Claim, including but not limited to: arguments, justifications, cost, estimates, Schedule analysis and detailed documentation. The format of the required reasonable documentation to support the Claim shall include, without limitation:
  - 1. Cover letter.
  - 2. Summary of factual basis of Claim and amount of Claim.
  - 3. Summary of the basis of the Claim, including the specific clause and section under the Contract under which the Claim is made.
  - 4. Documents relating to the Claim, including:
    - a. Specifications sections in question.
    - b. Relevant portions of the Drawings
    - c. Applicable Clarifications (RFI's)
    - d. Other relevant information, including responses that were received.
    - e. Contractor Analysis of Claim merit.

- (a) Contractor's analysis of any Subcontractor Vendor Claims that are being passed through.
- (b) Any analysis performed by outside consultants
- (c) Any legal analysis that Contractor deems relevant
- f. Break down of all costs associated with the Claim.
- g. For Claims relating to time extensions, an analysis and supporting documentation evidencing any effect upon the critical path in conformance with the requirements of Article 8.4 chronology of events and related correspondence.
- h. Applicable Daily Reports and logs.
  - (a) If the Daily Reports or Logs are not available, lost or destroyed, there shall be a presumption that the lost documentation was unfavorable to the Contractor. See California Civil Jury Instruction 204.
- i. For Claims involving overhead, cost escalation, acceleration, disruption or increased costs, a full version of job costs reports organized by category of work or Schedule of Values with budget information tracked against actual costs. Any and all supporting back-up data, including the original bid (and associated original unaltered metadata).
  - (a) The metadata and bid information shall be provided confidentially and subject to a protective order to prevent dissemination to other contractors or to the public. However, the bid documentation should remain intact and available for review and inspection in case of this type of increased cost Claim.
  - (b) This data on the bid shall be made available to any District attorneys or experts and shall also be utilized as evidence for any legal proceedings.
  - (c) If the bid documentation is not available, lost or destroyed, there shall be a presumption that the lost bid documentation was unfavorable to the Contractor. See California Civil Jury Instruction 204.
- f. <u>Certification</u>: The Contractor (and Subcontractors, if applicable) shall submit with the Claim a certification under penalty of perjury:
  - 2. That the Contractor has reviewed the Claim and that such Claim is made in good faith;
  - 3. Supporting data are accurate and complete to the best of the Contractor's knowledge and belief;

- 4. The amount requested accurately reflects the amount of compensation for which the Contractor believes the District is liable.
- 5. That the Contractor is familiar with Government Code sections 12650 et seq. and Penal Code section 72 and that false claims can lead to substantial fines and/or imprisonment.
- g. <u>Signature of Certification</u>: If the Contractor is not an individual, the certification shall be executed by an officer or general partner of the Contractor having overall responsibility for the conduct of the Contractor's affairs.
- h. Upon receipt of a Claim and all supporting documents as required above, the District shall conduct a reasonable review of the Claim and, within a period not to exceed 45 days, shall provide the Contractor a written statement identifying what portion of the Claim is disputed and what portion is undisputed. Upon receipt of a Claim, the District and Contractor may, by mutual agreement, extend the time period provided in this paragraph.
- i. If the District needs approval from its governing Board to provide the Contractor a written statement identifying the disputed portion and the undisputed portion of the Claim, and the governing Board does not meet within the 45 days or within the mutually agreed to extension of time following receipt of a Claim sent by registered mail or certified mail, return receipt requested, the District shall have up to three days following the next duly publicly noticed meeting of the governing Board after the 45-day period, or extension, expires to provide the Contractor a written statement identifying the disputed portion and the undisputed portion.
- j. Any payment due on an undisputed portion of the Claim shall be processed and made within 60 days after the District issues its written statement. If the District fails to issue a written statement, paragraph p below shall apply.
- k. If the Contractor disputes the District's written response, or if the District fails to respond to a Claim issued pursuant to this Article 4.6.9 within the time prescribed, the Contractor may demand in writing an informal conference to meet and confer for settlement of the issues in dispute. Upon receipt of a demand in writing sent by registered mail or certified mail, return receipt requested, the District shall schedule a meet and confer conference within 30 days for settlement of the Claim.
- 1. Within 10 business days following the conclusion of the meet and confer conference, if the Claim or any portion of the Claim remains in dispute, the District shall provide the Contractor a written statement identifying the portion of the Claim that remains in dispute and the portion that is undisputed. Any payment due on an undisputed portion of the Claim shall be processed and made within 60 days after the District issues its written statement. Any disputed portion of the Claim, as identified by the Contractor in writing, shall be submitted to nonbinding mediation, with the District and the Contractor sharing the associated costs equally. The District and Contractor shall mutually agree to a mediator within 10 business days after the disputed portion of the Claim has been identified in writing. If the parties cannot agree upon a mediator, each party shall select a mediator and those

mediators shall select a qualified neutral third party to mediate with regard to the disputed portion of the Claim. Each party shall bear the fees and costs charged by its respective mediator in connection with the selection of the neutral mediator. If mediation is unsuccessful, the parts of the Claim remaining in dispute shall be subject to applicable procedures in Article 4.6.9.5.

- m. For purposes of this Article 4.6.9, mediation includes any nonbinding process, including, but not limited to, neutral evaluation or a dispute review board, in which an independent third party or board assists the parties in dispute resolution through negotiation or by issuance of an evaluation. Any mediation utilized shall conform to the timeframes in this section.
- n. Unless otherwise agreed to by the District and the Contractor in writing, the mediation conducted pursuant to this Article 4.6.9 shall excuse any further obligation under Section 20104.4 to mediate after litigation has been commenced.
- o. This Claims process does not preclude the District from requiring arbitration of disputes under private arbitration or the Public Works Contract Arbitration Program, if mediation under this Article 4.6.9 does not resolve the parties' Claim. This Claims process does not preclude the District from submitting individual Disputes or Claims to binding arbitration pursuant to Article 4.6.9.4 below.
- p. Failure by the District to respond to a Claim from the Contractor within the time periods described in this subdivision or to otherwise meet the time requirements of this Article 4.6.9 shall result in the Claim being deemed rejected in its entirety. A Claim that is denied by reason of the District's failure to have responded to a Claim, or its failure to otherwise meet the time requirements of this Article 4.6.9, shall not constitute an adverse finding with regard to the merits of the Claim or the responsibility or qualifications of the Contractor.
- q. If a subcontractor or a lower tier subcontractor lacks legal standing to assert a Claim against a District because privity of contract does not exist, the Contractor may present to the District a Claim on behalf of a subcontractor or lower tier subcontractor. A subcontractor may request in writing, either on his or her own behalf or on behalf of a lower tier subcontractor, that the Contractor present a Claim for work which was performed by the subcontractor or by a lower tier subcontractor on behalf of the subcontractor. The subcontractor requesting that the Claim be presented to the District shall furnish reasonable documentation to support the Claim. Within 45 days of receipt of this written request, the Contractor shall notify the subcontractor in writing as to whether the Contractor presented the Claim to the District and, if the Contractor did not present the Claim, provide the subcontractor with a statement of the reasons for not having done so.
- r. Upon receipt of a Claim, the parties may mutually agree to waive, in writing, mediation and proceed directly to the commencement of a civil action or binding arbitration, as applicable.
- s. The Contractor's Claim shall be denied if it fails to follow the requirements of this Article.

- 4.6.9.2 District (through CM or District's Agent or Attorney) May Request Additional Information. Within thirty (30) days of receipt of the Claim and the information under this Article, the District may request in writing any additional documentation supporting the Claim or documentation relating to defenses to the Claim which the District may assert. If additional documents are required, the time in which the Claim is evaluated may be extended by a reasonable time so the Claim and additional documents may be reviewed.
- 4.6.9.3 Claims Procedures in Addition to Government Code Claim. Nothing in the Claims procedures set forth in this Article 4 of the General Conditions shall act to waive or relieve the Contractor from meeting the requirements set forth in Government Code section 900 et seq.
- 4.6.9.4 *Binding Arbitration of Individual Claim Issues*. To expedite resolution of Claims pursuant to Public Contract Code section 9201, at the District's sole option, the District may submit individual Claims to Arbitration prior to Retention Payment consistent with the requirements of Article 4.6.6.1.
- 4.6.9.5 Resolution of Claims in Court of Competent Jurisdiction. If Claims are not resolved under the procedure set forth and pursuant to Article 4.6.9, such Claim or controversy shall be submitted to a court in the County of the location of the Project after the Project has been completed, and not before.
- 4.6.9.6 Warranties, Guarantees and Obligations. The duties and obligations imposed by these General Conditions and the rights and remedies available hereunder to the parties hereto, and, in particular but without limitation, the warranties, guarantees and obligations imposed upon Contractor by the General Conditions and amendments thereto; and all of the rights and remedies available to District and Architect thereunder, are in addition to, and are not to be construed in any way as a limitation of, any rights and remedies available to any or all of them which are otherwise imposed or available by laws or regulations by special warranty or guarantee or by other provisions of the Contract Documents, and the provisions of this Article will be as effective as if repeated specifically in the Contract Documents in connection with each particular duty, obligation, right and remedy to which they apply.

## ARTICLE 5 SUBCONTRACTORS

### 5.1 <u>DEFINITIONS</u>

## 5.1.1 Subcontractual Relations Bound to Same Contract Terms at General Contractor

By appropriate agreement, written where legally required for validity, the Contractor shall require each Subcontractor, to the extent of the Work to be performed by the Subcontractor, to be bound to the Contractor by terms of the Contract Documents, and to assume toward the Contractor all the same obligations and responsibilities, assumed by Contractor pursuant to the Contract Documents. Each subcontract agreement shall preserve and protect the rights of the District and the Architect under the Contract Documents with respect to the Work to be performed by the Subcontractor so that subcontracting thereof will not prejudice such rights. Where appropriate, the Contractor shall require each Subcontractor to enter into similar agreements with Sub-subcontractors. The Contractor shall make available to each proposed Subcontractor, prior to the execution of the subcontract agreement, copies of the Contract Documents to which the Subcontractor will be bound. Upon written request of the Subcontractor, the Contractor shall identify to the Subcontractor the terms and conditions of the proposed subcontract agreement, which may be at variance with the Contract Documents. Subcontractors shall similarly make copies of applicable portions of such documents available to their respective proposed Sub-subcontractors.

### 5.1.2 Subcontractor Licenses and DIR Registration

All Subcontractors shall be properly licensed by the California State Licensing Board. All Subcontractors (of any tier) performing any portion of the Work must comply with the Labor Code sections 1725.5 and 1771.1 and must be properly and currently registered with the California Department of Industrial Relations and qualified to perform public works pursuant to Labor Code section 1725.5 throughout the duration of the Project. No portion of the Work is permitted to be performed by a Subcontractor of any tier unless the subcontractor is properly registered with DIR. Any Subcontractors of any tier not properly registered with DIR shall be substituted in accordance with Labor Code section 1771.1.

#### 5.1.3 Substitution of Subcontractor

Substitution of Subcontractors shall be permitted only as authorized under Public Contract Code §§ 4107 et seq. Any substitutions of Subcontractors shall not result in any increase in the Contract Price or result in the granting of any extension of time for the completion of the Project.

## 5.1.4 Contingent Assignment of Subcontracts and Other Contracts

Each subcontract, purchase order, vendor contract or agreement for any portion of the Work is hereby assigned by the Contractor to the District provided that:

- a. Such assignment is effective only after Termination of this Contract with the Contractor by the District as provided under Article 14 and only for those subcontracts and other contracts and agreements that the District accepts by notifying the Subcontractor or Materialman (as may be applicable) in writing; and
- b. Such assignment is subject to the prior rights of the Surety(ies) obligated under the Payment Bond and Performance Bond.

c.	The Contractor shall include adequate provisions for this contingent assignment of subcontracts and other contracts and agreements in each such document.

## ARTICLE 6 CONSTRUCTION BY DISTRICT OR BY SEPARATE CONTRACTORS

## 6.1 <u>DISTRICT'S RIGHT TO PERFORM CONSTRUCTION AND TO AWARD SEPARATE</u> CONTRACTS

## 6.1.1 Separate Contracts.

- 6.1.1.1 District reserves the right to let other contracts in connection with this Work. Contractor shall afford other contractors reasonable opportunity for (1) introduction and storage of their materials; (2) access to the Work; and (3) execution of their work. Contractor shall properly connect and coordinate its work with that of other Contractors.
- 6.1.1.2 If any part of Contractor's Work depends on proper execution or results of any other contractor, the Contractor shall inspect and within seven (7) days or less, report to Architect, in writing, any defects in such work that render it unsuitable for proper execution of Contractor's Work. Contractor will be held accountable for damages to District for that Work which it failed to inspect or should have inspected. Contractor's failure to inspect and report shall constitute its acceptance of other contractors' Work as fit and proper for reception of its Work, except as to defects which may develop in other contractors' work after execution of Contractor's work.
- 6.1.1.3 To ensure proper execution of its subsequent Work, Contractor shall measure and inspect Work already in place and shall at once report to the Architect in writing any discrepancy between executed Work as built and the Contract Documents.
- 6.1.1.4 Contractor shall ascertain to its own satisfaction the scope of the Project and nature of any other contracts that have been or may be awarded by District in prosecution of the Project and the potential impact of such Work on the Baseline Schedule or Schedule updates.
- 6.1.1.5 Nothing herein contained shall be interpreted as granting to Contractor the exclusive occupancy at the site of Project. Contractor shall not cause any unnecessary hindrance or delay to any other contractor working on the Project Site. If execution of any contract by the District is likely to cause interference with Contractor's performance of this Contract, once Contractor provides District timely written notice and identifies the Schedule Conflict, District shall decide which contractor shall cease work temporarily and which contractor shall continue, or whether Work can be coordinated so that contractors may proceed simultaneously.
- 6.1.1.6 District shall not be responsible for any damages suffered or extra costs incurred by Contractor resulting directly or indirectly from award or performance or attempted performance of any other contract or contracts at the Project necessary for the performance of the Project (examples include Electrical Utility Contractor, separate offsite contractor, a separate grading contractor, furniture installation etc.)

CONTRACTOR IS AWARE THAT THIS CONTRACT MAY BE SPLIT INTO SEVERAL PHASES BASED ON DOCUMENTATION PROVIDED WITH THIS BID OR DISCUSSED AT THE JOB WALK. CONTRACTOR HAS MADE ALLOWANCE FOR ANY DELAYS OR DAMAGES WHICH MAY ARISE FROM COORDINATION WITH CONTRACTORS REQUIRED FOR OTHER PHASES. IF ANY DELAYS SHOULD ARISE FROM ANOTHER CONTRACTOR

WORKING ON A DIFFERENT PHASE, CONTRACTOR'S SOLE REMEDY FOR DAMAGES, INCLUDING DELAY DAMAGES, SHALL BE AGAINST THE CONTRACTOR WHO CAUSED SUCH DAMAGE AND NOT THE DISTRICT. CONTRACTOR SHALL PROVIDE ACCESS TO OTHER CONTRACTORS FOR OTHER PHASES AS NECESSARY TO PREVENT DELAYS AND DAMAGES TO OTHER CONTRACTORS WORKING ON OTHER PHASES OF CONSTRUCTION.

## 6.1.2 <u>District's Right to Carry Out the Work</u>

(See Article 2.2)

## 6.1.3 <u>Designation as Contractor</u>

When separate contracts are awarded to contractors on the Project Site, the term "Contractor" in the Contract Documents in each case shall mean the Contractor who executes each separate District/Contractor Agreement.

### 6.1.4 District Notice to the Contractor of Other Contractors

The Contractor shall have overall responsibility to reasonably coordinate and schedule Contractor's activities with the activities of the District's forces and of each separate contractor with the Work of the Contractor, who shall cooperate with them. The Contractor shall participate with other separate contractors and the District in reviewing their construction schedules when:

- a. Notice is provided in the Contract Documents of other scope of Work,
- b. In the case where there is known Work to be performed by other Contractors
- c. For outside contractors hired by utilities
- d. Where the Contract Document provides "Work by Others" or "By Others"
- e. Where specifically noted during the Pre-Bid Conference
- f. Where specifically noted in the Mandatory Job Walk
- g. By CO or ICD,
- h. With respect to the installation of:
  - 1. Furniture,
  - 2. Electronics and networking equipment,
  - 3. Cabling,
  - 4. Low voltage,
  - 5. Off-site work,
  - 6. Grading (when by a separate contractor),
  - 7. Environmental remediation when excluded by the Contract Documents (i.e. asbestos, lead or other hazardous waste removal)
  - 8. Deep cleaning crews,
  - 9. Commissioning and testing,
  - 10. Keying and re-keying,

### 11. Programming

- 6.1.4.1 Exception where no Coordination is Required on the Part of the Contractor for Turn Key Operations. If the Contractor has specifically outlined a "Turn Key" or "Complete Delivery" of a final completed operational school in writing as part of the Baseline Schedule.
- 6.1.4.2 The Contractor shall make any revisions to the Baseline Schedule (or Schedule Update) and Contract Sum deemed necessary after a joint review and mutual agreement. The Baseline Schedule (or Schedule Update) shall then constitute the Schedules to be used by the Contractor, separate contractors, and the District until subsequently revised. Additionally, Contractor shall coordinate with Architect, District, and Inspector to ensure timely and proper progress of Work.

## 6.2 CONSTRUCTIVE OWNERSHIP OF PROJECT SITE AND MATERIAL

Upon commencement of Work, the Contractor becomes the constructive owner of the entire site, improvements, material and equipment on Project site. Contractor must ensure proper safety and storage of all materials and assumes responsibility as if Contractor was the owner of the Project site. All risk of loss or damage shall be borne by Contractor during the Work until the date of Completion. As constructive owner of the Project site, Contractor must carry adequate insurance in case of calamity and is not entitled to rely on the insurance requirements as set forth in this Agreement as being adequate coverage in case of calamity.

## 6.3 <u>DISTRICT'S RIGHT TO CLEAN UP</u>

If a dispute arises among the Contractor, separate contractors, and the District as to the responsibility under their respective contracts for maintaining the premises and surrounding area free from waste materials and rubbish as described in Article 3.12, the District may clean up and allocate the cost among those it deems responsible.

# ARTICLE 7 CHANGES IN THE WORK

### 7.1 CHANGES

### 7.1.1 No Changes Without Authorization

There shall be no change whatsoever in the Drawings, Specifications, or in the Work without an executed Change Order, Change Order Request, Immediate Change Directive, or order by the Architect for a minor change in the Work as herein provided. District shall not be liable for the cost of any extra work or any substitutions, changes, additions, omissions, or deviations from the Drawings and Specifications unless the District's Governing Board or designated representative with delegated authority (subject to Board ratification) has authorized the same and the cost thereof approved in writing by Change Order or executed Construction Change Document. No extension of time for performance of the Work shall be allowed hereunder unless claim for such extension is made at the time changes in the Work are ordered, and such time duly adjusted in writing in the Change Order. The provisions of the Contract Documents shall apply to all such changes, additions, and omissions with the same effect as if originally embodied in the Drawings and Specifications. Notwithstanding anything to the contrary in this Article 7, all Change Orders shall be prepared and issued by the Architect and shall become effective when executed by the District's Governing Board, the Architect, and the Contractor.

Should any Change Order result in an increase in the Contract Price, the cost of such Change Order shall be agreed to, in writing, in advance by Contractor and District and be subject to the monetary limitations set forth in Public Contract Code section 20118.4 (Please check with the District since there are different interpretations of the limitations of Public Contract Code section 20118.4 depending on the County the Project is located). In the event that Contractor proceeds with any change in Work without first notifying District and obtaining the Architect's and District's consent to a Change Order, Contractor waives any Claim of additional compensation for such additional work and Contractor takes the risk that a Notice of Non-Compliance may issue, a critical path Project delay may occur, and the Contractor will also be responsible for the cost of preparation and DSA CCD review fees for a corrective DSA approved Construction Change Document.

CONTRACTOR UNDERSTANDS, ACKNOWLEDGES, AND AGREES THAT THE REASON FOR THIS NOTICE REQUIREMENT IS SO THAT DISTRICT MAY HAVE AN OPPORTUNITY TO ANALYZE THE WORK AND DECIDE WHETHER THE DISTRICT SHALL PROCEED WITH THE CHANGE ORDER OR ALTER THE PROJECT SO THAT SUCH CHANGE IN WORK BECOMES UNNECESSARY AND TO AVOID THE POSSIBLE DELAYS ASSOCIATED WITH THE ISSUANCE OF A NOTICE OF NON-COMPLIANCE.

#### 7.1.2 Notices of Non-Compliance

Contractor deviation or changes from approved Plans and Specifications may result in the issuance of a Notice of Non-Compliance (See DSA Form 154). Contractor is specifically notified that deviations from the Plans and Specifications, whether major or minor, may result in the requirement to obtain a DSA Construction Change Document to correct the Notice of Non-Compliance. (See Article 7.3.1 for Definition of CCD). In some cases, the lack of a DSA approved CCD AND verification from the Inspector that a Notice of Non-Compliance has been corrected may result in a critical path delay to the next stage of Work on the Project. Specifically, a deviation from approved Plans and Specifications may prevent

approval of the category of Work listed in the DSA 152 Project Inspection Card. Any delays that are caused by the Contractor's deviation from approved Plans and Specifications shall be the Contractor's responsibility.

### 7.1.3 Architect Authority

The Architect will have authority to order minor changes in the Work that do not involve DSA Approval not involving any adjustment in the Contract Sum, or an extension of the Contract Time.

## 7.2 CHANGE ORDERS ("CO")

A CO is a written instrument prepared by the Architect and signed by the District (as authorized by the District's Governing Board), the Contractor, and the Architect stating their agreement upon all of the following:

- a. A description of a change in the Work;
- b. The amount of the adjustment in the Contract Sum, if any; and
- c. The extent of the adjustment in the Contract Time, if any.

A CO may be comprised of ICD's, Response to RFP's and COR's

## 7.3 <u>CONSTRUCTION CHANGE DOCUMENT (CCD Category A, and CCD Category B) and</u> IMMEDIATE CHANGE DIRECTIVE (ICD)

## 7.3.1 <u>Definitions</u>

- 7.3.1.1 Construction Change Document (CCD). A Construction Change Document is a DSA term that is utilized to address changes to the DSA approved Plans and Specifications. There are two types of Construction Change Documents. (1) DSA approved CCD Category A for Work affecting structural, access compliance or fire/ life safety of the Project which will require a DSA approval; and, (2) CCD Category B for work NOT affecting structural safety, access compliance or fire/ life safety that will not require a DSA approval (except to confirm that no approval is required). Both CCD Category A and Category B shall be set forth in DSA Form 140 and submitted to DSA as required.
- 7.3.1.2 *Immediate Change Directive (ICD)*. An Immediate Change Directive is a written order to the Contractor prepared by the Architect and signed by the District (and CM if there is a CM on the Project) and the Architect, directing a change in the Work and stating a proposed basis for adjustment, if any, in the Contract Sum or Contract Time, or both. The District may by ICD, without invalidating the Contract, direct immediate changes in the Work within the general scope of the Contract consisting of additions, deletions, or other revisions within. If applicable, the Contract Sum and Contract Time will be adjusted accordingly.

In the case of an Immediate Change Directive being issued, Contractor must commence Work immediately or delays from failure to perform the ICD shall be the responsibility of Contractor and the failure to move forward with Work immediately shall also be grounds for Termination under Article 14.

An ICD does not automatically trigger an Article 7.6 Dispute or Claim. Contractor must timely follow the procedures outlined at Article 7.6 and 4.6 where applicable.

Refer to Division 1 and Supplementary General Conditions for a copy of the proposed Immediate Change Directive form.

### 7.3.2 <u>Use to Direct Change</u>

An ICD shall be used to move work forward immediately and to avoid delay. In some cases, an ICD shall be issued in the absence of agreement on the terms of a CO, COR, or RFP. A copy of an ICD form is provided in the Supplementary General Conditions and Division 1. The anticipated not to exceed price for the Work will be inserted into the ICD. In the case of an ICD issued to correct Contractor Deficiencies or to correct a Contractor caused Notice of Non-Compliance, the ICD may be issued with \$0 and no additional time. Contract may prepare a COR associated with the ICD pursuant to Article 7. However, Contractor shall proceed with all Work required under an Approved ICD immediately upon issuance. Failure to proceed with the Work under an ICD shall be grounds for Termination for Cause under Article 14 or take over the Work under Article 2.2.

If adequate time exists, an ICD may be subject of an RFP for pricing and determination if any time that may be required. However, if an RFP is not completed, Contractor shall immediately commence Work when an ICD is issued. If the RFP is incomplete, it may still be completed to be submitted for pricing purposes as long as the RFP is submitted within the timeline provided by the RFP, or within 10 days following issuance of the ICD.

## 7.3.3 <u>ICD Issued Over a Notice of Non-Compliance or to Cover Work Subject to a DSA 152</u> Sign Off

In some cases, an ICD shall be for the purpose of proceeding with Work to keep the Project on Schedule and as an acknowledgement by the District that Contractor is proceeding with Work contrary to a Notice of Non-Compliance, prior to issuance of a DSA approved CCD Category A, or to direct the covering of Work which has not yet received a DSA 152 Inspection Approval to move forward.

- 7.3.3.1 Contractor Compliance with all Aspects of an ICD. Contractor is to undertake the ICD and comply with all aspects of the Work outlined in the ICD. Inspector is to inspect the Work pursuant to the ICD. Failure to follow the ICD may result in deduction of the ICD Work under Article 2.2 or Termination of the Contractor pursuant to Article 14.
- 7.3.3.2 Exception in the Case of DSA Issued Stop Work Order. Contractor must proceed with an ICD even if a CCD has not been approved by DSA except in the case of a DSA issued Stop Work Order. If a DSA Stop Work Order is issued, Contractor must stop work and wait further direction from the District.
- 7.3.3.3 ICD Due to Contractor Deficiency or Contractor Caused Notice of Non-Compliance. If an ICD is issued to correct a Contractor Deficiency or a Contractor caused notice of Non-Compliance, Contractor specifically acknowledges responsibility for all consequential damages associated with the Contractor Deficiency or Contractor caused Notice of Non-Compliance and all consequential damages and costs incurred to correct the deficiency under Article 4.5

## 7.4 REQUEST FOR INFORMATION ("RFI")

### 7.4.1 Definition

A RFI is a written request prepared by the Contractor requesting the Architect to provide additional information necessary to clarify or amplify an item which the Contractor believes is not clearly shown or called for in the Drawings or Specifications, or to address problems which have arisen under field conditions.

- 7.4.1.1 A RFI shall not be used as a vehicle to generate time extensions.
- 7.4.1.2 Resubmission of the same or similar RFI is not acceptable. RFI's that are similar should be addressed in Project meetings where the requestor (Contractor, Subcontractor or vendor) is able to address the particular issue with the Architect or Engineer and a resolution addressed in the minutes.
- 7.4.1.3 A RFI response applicable to a specific area cannot be extended to other situations unless specifically addressed in writing within the RFI or in a separate RFI.
- 7.4.1.4 RFI's should provide a proposed solution and should adequately describe the problem that has arisen.

## 7.4.2 <u>Scope</u>

The RFI shall reference all the applicable Contract Documents including Specification section, detail, page numbers, Drawing numbers, and sheet numbers, etc. The Contractor shall make suggestions and interpretations of the issue raised by the RFI. An RFI cannot modify the Contract Cost, Contract Time, or the Contract Documents.

### 7.4.3 Response Time

The Architect must respond to a RFI within a reasonable time after receiving such request. If the Architect's response results in a change in the Work, then such change shall be effected by a written CO, COR RFP or ICD, if appropriate. If the Architect cannot respond to the RFI within a reasonable time, the Architect shall notify the Contractor, with a copy to the Inspector and the District, of the amount of time that will be required to respond.

### 7.4.4 Costs Incurred

The Contractor shall be responsible for any costs incurred for professional services as more fully set forth in Article 4.5, which shall be subject to a Deductive Change Order, if an RFI requests an interpretation or decision of a matter where the information sought is equally available to the party making such request. District, at its sole discretion, shall issue a Deductive Change Order to Contractor for all such professional services arising from this Article.

### 7.5 REQUEST FOR PROPOSAL ("RFP")

### 7.5.1 <u>Definition</u>

A RFP is a written request prepared by the Architect (and/or CM) requesting the Contractor to submit to the District and the Architect an estimate of the effect of a proposed change on the Contract Price and (if applicable) the Contract Time. If Architect issues a Bulletin, the Changed items in the Bulletin shall be addressed as an RFP and all responses shall be prepared to a Bulletin as addressed in this Article 7.5. A form RFP is included in the Division 1 documents.

## 7.5.2 <u>Scope</u>

A RFP shall contain adequate information, including any necessary Drawings and Specifications, to enable Contractor to provide the cost breakdowns required by Article 7.7. The Contractor shall not be entitled to any Additional Compensation for preparing a response to an RFP, whether ultimately accepted or not.

## 7.5.3 <u>Response Time</u>

Contractor shall respond to an RFP within ten (10) days or the time period otherwise set forth in the RFP.

# 7.6 CHANGE ORDER REQUEST ("COR")

#### 7.6.1 Definition

A COR is a written request prepared by the Contractor supported by backup documentation requesting that the District and the Architect issue a CO based upon a proposed change, cost, time, or cost and time that may be incurred on the Project or arising from an RFP, ICD, or CCD.

## 7.6.2 Changes in Price

A COR shall include breakdowns per Article 7.7 to validate any change in Contract Price due to proposed change or Claim.

## 7.6.3 Changes in Time

A COR shall also include any additional time required to complete the Project only if the delay is a critical path delay. Any additional time requested shall not be the number of days to make the proposed change, but must be based upon the impact to the Project Schedule as defined in Article 8. A schedule fragnet showing the time delay must be submitted with the COR. Any changes in time will be granted only if there is an impact to the critical path. If Contractor fails to request a time extension in a COR, then the Contractor is thereafter precluded from requesting or claiming a delay.

# 7.7 <u>COST OF CHANGE ORDERS</u>

## 7.7.1 Scope

Within ten (10) days after a request is made for a change that impacts the Contract Sum as defined in Article 9.1, the critical path, or the Contract Time as defined in Article 8.1.1, the Contractor shall provide the District and the Architect, with a written estimate of the effect of the proposed CO upon the Contract Sum and the actual cost of construction, which shall include a complete itemized cost breakdown of all labor and material showing actual quantities, hours, unit prices, and wage rates required for the change, and the effect upon the Contract Time of such CO. Changes may be made by District by an

appropriate written CO, or, at the District's option, such changes shall be implemented immediately upon the Contractor's receipt of an appropriate written Construction Change Document.

District may, as provided by law and without affecting the validity of this Agreement, order changes, modification, deletions and extra work by issuance of written CO or CCD from time to time during the progress of the Project, Contract Sum being adjusted accordingly. All such Work shall be executed under conditions of the original Agreement except that any extension of time caused thereby shall be adjusted at time of ordering such change. District has discretion to order changes on a "time and material" basis with adjustments to time made after Contractor has justified through documentation the impact on the critical path of the Project.

7.7.1.1 *Time and Material Charges*. If the District Orders Work on a "time and material" basis, timesheets shall be signed daily by the Inspector or District Representative at or near the time the Work is actually undertaken and shall show the hours worked, and the Work actually completed. No time sheets shall be signed the next day. A copy shall be provided to the Person signing the document at the time the document is signed, but not before 10 am the following day.

## 7.7.2 Determination of Cost

The amount of the increase or decrease in the Contract Price from a CO or COR, if any, shall be determined in one or more of the following ways as applicable to a specific situation:

- a. <u>Mutual acceptance</u> of a lump sum properly itemized and supported by sufficient substantiating data to permit evaluation. If an agreement cannot be reached within fifteen (15) days after submission and negotiation of Contractor's proposal, Contractor may submit pursuant to Article 7.7.3. Submission of sums which have no basis in fact are at the sole risk of Contractor and may be a violation of the False Claims Act set forth under Government Code section 12650 et seq.);
  - 1. If the District objects to 7.7.2(a) as a method for submission due to inaccuracies in the submitted amount, overstatement of manpower or time required to perform the CO, or unreliability of the data provided, the District may either have the Architect or a professional estimator determine the cost for the CO, and the applicable time extension, or the Contractor shall utilize Article 7.7.2(d) or 7.7.3.
  - 2. Once the District provides a written objection to use of Article 7.7.2(a) due to unreliability of the estimated price, the Contractor shall no longer utilize mutual acceptance of a lump sum as a method for submission of CO's and shall provide a breakdown of estimated or actual costs pursuant to Article 7.7.2(d) or 7.7.3
- b. By unit prices contained in Contractor's original bid and incorporated in the Project documents or fixed by subsequent agreement between District and Contractor;
- c. Cost to be determined in a manner agreed upon by the parties and a mutually acceptable fixed or percentage fee. However, in the case of disagreement, Contractor must utilize the procedure under Article 7.7.3; or

- d. By cost of material and labor and percentage of overhead and profit. If the value is determined by this method the following requirements shall apply:
  - 1. Basis for Establishing Costs
    - (1) <u>Labor will be the cost for wages</u> prevailing locally for each craft or type of workers at the time the extra Work is done, plus employer payments of payroll taxes and workers compensation insurance (exclude insurance costs as part of the overhead and profit mark-up), health and welfare, pension, vacation, apprenticeship funds, and other direct costs resulting from Federal. State, or local laws, as well as assessments or benefits required by lawful collective bargaining agreements. In no case shall the total labor costs exceed the applicable prevailing wage rate for that particular classification. The use of a labor classification which would increase the extra Work cost will not be permitted unless the Contractor establishes the necessity for such additional costs. Labor costs for equipment operators and helpers shall be reported only when such costs are not included in the invoice for equipment rental.
    - (2) Materials shall be at invoice or lowest current price at which such materials are locally available and delivered to the Site in the quantities involved, plus sales tax, freight, and delivery. The District reserves the right to approve materials and sources of supply or to supply materials to the Contractor if necessary for the progress of the Work. No markup shall be applied to any material provided by the District.
    - (3) <u>Tool and Equipment Rental</u>. No payment will be made for the use of tools which have a replacement value of \$250 or less.

Regardless of ownership, the rates to be used in determining equipment rental costs shall not exceed listed rates prevailing locally at equipment rental agencies or distributors at the time the Work is performed. Rates applied shall be appropriate based on actual equipment need and usage. Monthly, weekly or other extended use rates that results in the lowest cost shall be applied if equipment is used on site for extended periods.

The rental rates paid shall include the cost of fuel, oil, lubrication, supplies, small tools, necessary attachments, repairs and maintenance of any kind, depreciation, storage, insurance, and all incidentals.

Necessary loading and transportation costs for equipment used on the extra Work shall be included. If equipment is used intermittently and, when not in use, could be returned to its rental source at less expense to the District than holding it at the Work Site, it shall be returned unless the Contractor elects to keep it at the Work Site at no expense to the District.

All equipment shall be acceptable to the Inspector, in good working condition, and suitable for the purpose for which it is to be used. Manufacturer's ratings and modifications shall be used to classify equipment, and equipment shall be powered by a unit of at least the minimum rating recommended by the manufacturer.

If tool and equipment charges are part of a Dispute, Claim, or Appeal, the District reserves the right to utilize actual costs for tools and equipment or a depreciation rate for equipment based on audit finding under Article 13.11 and deduct any rental charges that exceed actual or depreciated costs.

- e. <u>Other Items</u>. The District may authorize other items which may be required on the extra work. Such items include labor, services, material, and equipment which are different in their nature from those required by the Work, and which are of a type not ordinarily available from the Contractor or any of the Subcontractors. Invoices covering all such items in detail shall be submitted with the request for payment.
- f. <u>Invoices</u>. Vendors' invoices for material, equipment rental, and other expenditures shall be submitted with the COR. If the request for payment is not substantiated by invoices or other documentation, the District may establish the cost of the item involved at the lowest price which was current at the time of the Daily Report.
- g. Overhead. Overhead, including direct and indirect costs, shall be submitted with the COR and include: field overhead, home office overhead, off-site supervision, CO preparation/negotiation/research, time delays, Project interference and disruption, additional guaranty and warranty durations, on-site supervision, additional temporary protection, additional temporary utilities, additional material handling costs, liability and property damage insurance, and additional safety equipment costs.

## 7.7.3 Format for COR or CO's

The following format shall be used as applicable by the District and the Contractor to communicate proposed additions to the Contract. All costs submitted shall be actual costs and labor shall be unburdened labor. Refer to Division 1 for a copy of the Construction Change Order form.

(a)	Material (attach itemized quantity and unit cost plus sales tax)	<u>EXTRA</u>	<u>CREDIT</u>
(b)	Labor Not to Exceed Applicable Prevailing Wage Rates (attach itemized hours and rates)		
(c)	Equipment (attach invoices)		
(d)	Subtotal		

		<u>EXTRA</u>	<u>CREDIT</u>
(e)	If Subcontractor performed work, add Subcontractor's overhead and profit to		
	portions performed by Subcontractor, not to		
	exceed 10% of item (d).		
(f)	Subtotal		
(g)	Contractor's Overhead and Profit: Not to		
	exceed 10% of Item (d) if Contractor performed the work. No more than 5% of		
	Îtem (d) if Subcontractor performed the		
	work. If work was performed by Contractor and Subcontractors, portions performed by		
	Contractor shall not exceed 10% of Item (d), and portions performed by Subcontractor		
	shall not exceed 10% of Item (d).		
(h)	Subtotal		
(i)	Bond not to exceed one percent (1%) of Item		
	(h)		
(k)	TOTAL		
(1)	Time/ Days		

The undersigned Contractor approves the foregoing Change Order or Immediate Change Directive as to the changes, if any, and the Contract price specified for each item and as to the extension of time allowed, if any, for completion of the entire Work on account of said Change Order or Immediate Change Directive, and agrees to furnish all labor, materials and service and perform all Work necessary to complete any additional Work specified therein, for the consideration stated herein. It is understood that said Change Order or Immediate Change Directive shall be effective when approved by the Governing Board of the District.

It is expressly understood that the value of such extra Work or changes, as determined by any of the aforementioned methods, expressly includes any and all of the Contractor's costs and expenses, both direct and indirect, resulting from additional time required on the Project or resulting from delay to the Project. Any costs, expenses, damages or time extensions not included are deemed waived.

The Contractor expressly acknowledges and agrees that any change in the Work performed shall not be deemed to constitute a delay or other basis for claiming additional compensation based on theories including, but not limited to, acceleration, suspension or disruption to the Project.

7.7.3.1 Adjustment for Time and Compensable Delay. A CO shall also include any additional time required to complete the Project. Any additional time requested shall not be the number of days to make the proposed change, but must be based upon the impact to the Project Schedule as defined

in Article 8 of the General Contract. A schedule fragnet showing the time delay must be submitted with the CO. Any changes in time will be granted only if there is an impact to the critical path. If Contractor fails to request a time extension in a CO, then the Contractor is thereafter precluded from requesting or claiming a delay.

## 7.7.4 <u>Deductive Change Orders</u>

All Deductive Change Order(s) must be prepared utilizing the form under Article 7.7.3 (a) – (d) only, setting forth the actual costs incurred. Except in the case of an Article 2.2 or 9.6 Deductive Change Order where no mark-up shall be allowed, Contractor will be allowed a maximum of 5% total profit and overhead.

For unilateral Deductive Change Orders, or where credits are due from Contractor for Allowances, Deductive Items, Inspection, Damage, DSA CCD review costs, Architect or Inspector costs for after hours or corrective services, Work removed from the Agreement under Article 2.2 or Article 9.6, there shall be no mark-up.

District may, any time after a Deductive Change Order is presented to Contractor by District for items under Article 2.2 or Article 9.6 or if there is disagreement as to the Deductive Change Order, issue a unilateral Deductive Change Order on the Project and deduct the Deductive Change Order from a Progress Payment, Final Payment, or Retention.

## 7.7.5 Discounts, Rebates, and Refunds

For purposes of determining the cost, if any, of any change, addition, or omission to the Work hereunder, all trade discounts, rebates, refunds, and all returns from the sale of surplus materials and equipment shall accrue and be credited to the Contractor, and the Contractor shall make provisions so that such discounts, rebates, refunds, and returns may be secured, and the amount thereof shall be allowed as a reduction of the Contractor's cost in determining the actual cost of construction for purposes of any change, addition, or omissions in the Work as provided herein. All CO's are subject to Audit under Article 13.11 for discounts, rebates and refunds.

## 7.7.6 <u>Accounting Records</u>

With respect to portions of the Work performed by CO's and CCD's on a time-and-materials, unit-cost, or similar basis, the Contractor shall keep and maintain cost-accounting records in a format consistent with accepted accounting standards and satisfactory to the District, which shall be available to the District on the same terms as any other books and records the Contractor is required to maintain under the Contract Documents.

Any time and material charges shall require Inspector's signature on time and material cards showing the hours worked and the Work actually completed. (See Article 7.7.1.1)

## 7.7.7 <u>Notice Required</u>

If the Contractor desires to initiate a Dispute for an increase in the Contract Price, or any extension in the Contract Time for completion, Contractor shall notify the applicable party responsible for addressing the Dispute or Claim pursuant to Article 4.6. No Claim or Dispute shall be considered unless made in accordance with this subparagraph. Contractor shall proceed to execute the Work even though the

adjustment may not have been agreed upon. Any change in the Contract Price or extension of the Contract Time resulting from such Claim shall be authorized by a CO.

# 7.7.8 <u>Applicability to Subcontractors</u>

Any requirements under this Article 7 shall be equally applicable to CO's, COR's or ICD's issued to Subcontractors by the Contractor to the same extent required by the Contractor.

## 7.7.9 Alteration to Change Order Language

Contractor shall not alter or reserve time in COR's, CO's or ICD's. Contractor shall execute finalized CO's and proceed under Article 7.7.7 and Article 4.6 with proper notice. If Contractor intends to reserve time without an approved CPM schedule prepared pursuant to Article 8 or without submitting a fragnet showing delay to critical path, then Contractor may be prosecuted pursuant to the False Claim Act.

# ARTICLE 8 TIME AND SCHEDULE

## 8.1 **DEFINITIONS**

#### 8.1.1 Contract Time

Contractor shall perform and reach Substantial Completion (See Article 1.1.46) within the time specified in the Agreement Form. Moreover, Contractor shall perform its Work in strict accordance with the Project Milestones in the Contract Documents and shall proceed on a properly developed and approved Baseline Schedule, which represents the Contractor's view of the practical way in which the Work will be accomplished. Note that Contract Time includes and incorporates all Float and other Baseline inclusions as noted in Article 8.3.2.1 and as otherwise specifically noted in Article 8.

## 8.1.2 Notice to Proceed

District may give a Notice to Proceed within ninety (90) days of the award of the bid by District. Once Contractor has received the notice to proceed, Contractor shall complete the Work in the period of time referenced in the Contract Documents.

In the event that District desires to postpone the giving of the Notice to Proceed beyond this three-month period, it is expressly understood that with reasonable notice to the Contractor, the giving of the date to proceed may be postponed by District. It is further expressly understood by Contractor, that Contractor shall not be entitled to any claim of additional compensation as a result of the postponement of the giving of the notice to proceed

If the Contractor believes that a postponement will cause a hardship to Contractor, Contractor may terminate the Contract with written notice to District within 10 days after receipt by Contractor of District's notice of postponement. It is further understood by Contractor that in the event that Contractor terminates the Contract as a result of postponement by the District, the District shall only be obligated to pay Contractor for the Work that Contractor had performed at the time of notification of postponement and the grounds for notification and hardship shall be subject to Audit pursuant to Article 13.11. Should Contractor terminate the Contract as a result of a notice of postponement, District may award the Contract to the next lowest responsible bidder.

## 8.1.3 <u>Computation of Time</u>

The term "day" as used in the Contract Documents shall mean calendar day unless otherwise specifically defined.

## 8.1.4 <u>Float</u>

Float is time the total number of days an activity may be extended or delayed without delaying the Completion Date shown in the schedule. Float will fall into three categories: (1) Rain Days; (2) Governmental Delays; and, (3) Project Float. Project Float and Rain Days are owned by the Project and may be utilized as necessary for critical path delays once the days become available for consumption (i.e. the Rain Day arrives and is not utilized since rain did not occur or Work was performed on the interior of a building). However, Governmental Delay float shall not be utilized for purposes other than to address critical path delays that arise due to approvals, Inspector approvals or verifications on governmental forms.

8.1.4.1 Governmental Delay Float. It is anticipated that there will be governmental generated delays. Specific to DSA approvals, it is anticipated that no less than twelve (12) days per calendar year shall be set aside as Governmental Float to be utilized on critical path delays. A pro-rated number of days shall be calculated based on length of Contract Time. (For example, a two (2) year Contract Time shall require twenty-four (24) days of Governmental Float. If the Contract Time is 182 days, then the Contract Time shall require six (6) days of Governmental Float) This Governmental Delay float must be incorporated into the schedule and should be incorporated in each critical activity as Contractor deems fit. Specifically, major categories of Work under the DSA 152 (Project Inspection Card) should be allocated Governmental Delay Float at the Contractor's discretion. Governmental Delay Float on the Project may exceed 12 days per one (1) year period, but Contractor is required to include not be less than 12 days of Governmental Delay Float during each one (1) year period.

Contractor's failure to establish a protocol for requesting inspections is not grounds to utilize Governmental Delay Float. As noted in Article 3.1.4, 48 hours advance notice of commencing Work on a new area is required after submitting form DSA 156 and under PR 13-01 Special Inspection reports are not required to be posted until at least 14 days after the Work was inspected. Failure to plan, and pay (if applicable) for quicker delivery of Special Inspections is not Governmental Delay Float under Article 8.1.4.1. If Governmental Delay Float is not utilized, this float is carried through to other DSA 152 categories of inspection and consumed over the course of the Project

Governmental Delay Float may be utilized for a DSA Stop Work Order regardless of fault as defined under Education Code section 17307.5(b).

8.1.4.2 *Inclement Weather (Rain Days)*. The Contractor will only be allowed a time extension for unusually severe weather if it results in precipitation or other conditions which in the amount, frequency, or duration is in excess of the norm at the location and time of year in question as established by NOAA weather data. No less than 22 calendar days for each calendar year for Southern California will be allotted for in the Contractor's schedule for each winter weather period or carried at the end of the schedule as Rain Float. Float for weather days in other geographical regions shall be adjusted based on NOAA weather data for the geographical location. Contractor has anticipated all the days it takes to dry out and re-prepare areas that may be affected by weather delays which extend beyond the actual weather days. The weather days shall be shown on the schedule and if not used will become float for the Project's use. The Contractor will not be allowed a day-for-day weather delay for periods noted as float in the Schedule. The Contractor is expected to work seven (7) days per week (if necessary, irrespective of inclement weather), to maintain access, and to protect the Work under construction from the effects of inclement weather. Additional days beyond the NOAA shall be considered under the same criteria that weather days are granted below.

A Rain Day shall be granted by Architect or CM if the weather prevents the Contractor from beginning Work at the usual daily starting time, or prevents the Contractor from proceeding with seventy-five (75%) of the normal labor and equipment force towards completion of the day's current controlling item on the accepted schedule for a period of at least five hours, and the crew is dismissed as a result thereof, the Architect will designate such time as unavoidable delay and grant one (1) critical path activity calendar-day extension if there is no available float for the calendar year.

8.1.4.3 *Project Float.* The Contractor may determine some activities require a lesser duration than allocated and may set aside float in the Project Schedule. There shall be no early completion. Instead, to the extent float is either addressed at the end of the Project or throughout each category of critical path work, Project float may be used as necessary during the course of the Project and allocated on a first,

come first serve basis. However, the use of float does not extend to Governmental Delay Float, which shall only be used for Governmental Delays.

# 8.2 HOURS OF WORK

#### 8.2.1 Sufficient Forces

Contractors and Subcontractors shall continuously furnish sufficient forces to ensure the prosecution of the Work in accordance with the Construction Schedule.

## 8.2.2 Performance During Working Hours

Work shall be performed during regular working hours as permitted by the appropriate governmental agency except that in the event of an emergency, or when required to complete the Work in accordance with job progress, Work may be performed outside of regular working hours with the advance written consent of the District and approval of any required governmental agencies.

#### 8.2.3 Costs for After Hours Inspections

If the Work done after hours is required by the Contract Documents, a Recovery Schedule, or as a result of the Contractor's failure to plan, and inspection must be conducted outside the Inspector's regular working hours, the costs of any after hour inspections, shall be borne by the Contractor.

If the District allows the Contractor to do Work outside regular working hours for the Contractor's convenience, the costs of any inspections required outside regular working hours shall be invoiced to the Contractor by the District and a Deductive Change Order shall be issued from the next Progress Payment.

If the Contractor elects to perform Work outside the Inspector's regular working hours, costs of any inspections required outside regular working hours shall be invoiced to the Contractor by the District and a Deductive Change Order from the next Progress Payment as a Deductive Change Order.

## 8.3 PROGRESS AND COMPLETION

## 8.3.1 Time of the Essence

Time limits stated in the Contract Documents are of the essence to the Contract. By executing the Agreement, the Contractor confirms that the Contract Time is a reasonable period for performing the Work.

## 8.3.2 <u>Baseline Schedule Requirements</u>

8.3.2.1 *Timing*: Within ten (10) calendar days after Notice to Proceed, Contractor shall submit a practical schedule showing the order in which the Contractor proposes to perform the Work, and the dates on which the Contractor contemplates starting and completing the salient categories of the Work. This first schedule which outlines the Contractor's view of the practical way in which the Work will be accomplished is the Baseline Schedule. If the Contractor Fails to submit the Baseline Schedule within the ten (10) days noted, then District may withhold processing and approval of progress payments pursuant to Article 9.4 and 9.6.

- 8.3.2.2 District Review and Approval: District, Architect and CM will review both a paper and electronic copy of Baseline Schedule and may provide comments as noted in this Article and either approve or disapprove the Baseline Schedule. All Schedules shall be prepared using an electronic scheduling program acceptable to District. All Schedules shall be delivered in an electronic format usable by the District. All logic ties and electronic information shall be included in the electronic copy of the Baseline Schedule that is delivered to the District.
- 8.3.2.3 Schedule Must Be Within the Given Contract Time. The Baseline Schedule shall not exceed time limits set forth in the Contract Documents and shall comply with all of the scheduling requirements as set forth in the Specifications and Contract Documents.
- 8.3.2.4 Submittals Must Be Incorporated (See Articles 3.7 and 3.9): Contractor shall include Submittals as line items in the Baseline Schedule as required under Article 3.7.2 and 3.9.6. Submittals shall not delay the Work, Milestones, or the Completion Date. Failure to include Submittals in the Baseline Schedule shall be deemed a material breach by the Contractor.
- 8.3.2.5 Float Must Be Incorporated. The Baseline Schedule must indicate the beginning and completion of all phases of construction and shall use the "critical path method" (commonly called CPM) for the value reporting, planning and scheduling, of all Work required under the Contract Documents. The Baseline Schedule must incorporate all Milestones in the Project and apply Governmental Float at each Milestone in the Contractor's discretion. The Baseline Schedule shall incorporate any Schedule provided by the District as part of the bid and shall note durations that will not be adequate or should be shortened based on Contractor's review. These changes shall be identified and incorporated into Contractor's Baseline Schedule as long as requested changes are made within 10 days after the District chooses to move forward with the Project. Scheduling is necessary for the District's adequate monitoring of the progress of the Work and shall be prepared in accordance with the time frame described in this Article 8. The Architect may disapprove of any Schedule or require modification to it if, in the opinion of the Architect or District, adherence to the any Schedule prepared by the Contractor will not cause the Work to be completed in accordance with the Agreement.
- 8.3.2.6 *No Early Completion.* Contractor shall not submit any Schedule showing early completion without indicating float time through the date set for Project completion by District. Contractor's Baseline Schedule shall account for all days past early completion as float which belongs to the Project. Usage of float shall not entitle Contractor to any delay Claim or damages due to delay.
- 8.3.2.7 *Use of Schedule Provided in Bid Documents.* In some cases, the bid will include a preliminary schedule indicating Milestones and construction sequences for the Project along with general timing for the Project. The preliminary schedule is not intended to serve as the Baseline Schedule utilized for construction. It is up to the Contractor to study and develop a Baseline Schedule to address the actual durations and sequences of Work that is anticipated while maintaining the Milestones provided by the District. Contract shall obtain information from Contractor's Subcontractors and vendors on the planning, progress, delivery of equipment, coordination, and timing of availability of Subcontractors so a practical plan of Work is fully developed and represented in the Baseline Schedule.
- 8.3.2.8 Incorrect Logic, Durations, Sequences, or Critical Path. The District may reject or indicate durations, sequences, critical path or logic are not acceptable and request changes. The electronic copy of the Baseline Schedule shall have adequate information so logic ties, duration, sequences and critical path may be reviewed electronically. Contractor is to diligently rebuild and resubmit the Baseline Schedule to represent the Contractor's plan to complete the Work and maintain Milestones at the next progress meeting, or before the next progress meeting. If Contractor is not able to build a Baseline

Schedule that is acceptable to the District or Architect, the District reserves the right to utilize the unapproved originally submitted Baseline Schedule (See Article 8.3.2.12) and the comments submitted to hold Contractor accountable for timely delivery of Work and maintenance of Milestones. Furthermore, Contractor's representations in the Baseline Schedule, if unacceptable, may also be used as a basis for termination of the Contract under Article 14 if Contractor fails to adequately maintain the Schedule and falls significantly behind without undertaking the efforts to either submit and follow a Recovery Schedule or fail to submit a Recovery Schedule and make no effort toward recovery on the Project.

- 8.3.2.9 *Contractor Responsibility Even if Schedule Issues Are Not Discovered.* Failure on the Part of the District to discover errors or omissions in any Schedules submitted shall not be construed to be an approval of the error or omission and any flawed Schedule is not grounds for a time extension.
  - 8.3.2.9 <u>Inclusions in Baseline Schedule.</u> In addition to scheduling requirements set forth at Article 8.3.2, Contractor is specifically directed to include (broken out separately) in Contractor's Baseline Schedule and all Schedule updates, the following items required pursuant to these General Conditions, including but not limited to:
  - 1. Rain Day Float (excluding inclement weather) as required under Article 8.1.4.2. For example, if the NOAA provides 22 days of Rain Days, all 22 days must be incorporated and noted in the Baseline Schedule. Further, any days required to clean-up or dry out shall be included for operations that are likely to require a clean-up or dry out period. Days that are not utilized shall be considered float owned by the Project.
  - 2. Governmental Delay Float under Article 8.1.4.1. This Governmental Delay Float shall only be utilized for Governmental Delays and shall not be considered available float owned by the Project. This float shall only be distributed to the Project upon the completion of the Project and shall be used to offset Liquidated Damages and shall not generate compensable delays.
  - 3. Submittal and Shop Drawing schedule under Article 3.9.
  - 4. Deferred Approvals under Article 3.9.
  - 5. Time for separate contractors, including furniture installation and start up activities, under Article 6.1.
  - 6. Coordination and timing of any Drawings, approvals, notifications, permitting, connection, and testing for all utilities for the Project. (See Article 2.1.4).
  - 7. Testing, special events, or school activities
- 8.3.2.10 Failure to include Mandatory Schedule Items. District may withhold payment pursuant to Articles 9.3, 9.4 and 9.6. In lieu of withholding payment for failure to include Mandatory Schedule Items, after the District or Architect has notified the Contractor of failure to meet the Baseline Schedule or Updated Schedule requirements and provided a written notification of this failure and provided a written notice of Schedule preparation errors, and the Contractor fails to correct the noted deficiencies or

the Contractor does not provide an updated Baseline Schedule correcting the deficiencies, then Contractor shall not be granted an extension of time for failure to obtain necessary items and approvals under Article 8.3.2 and for the time required for failure to comply with laws, building codes, and other regulations (including Title 24 of the California Code of Regulations). Contractor shall maintain all required Article 8.3.2 Schedule items in the Baseline Schedule and indicate any days that have been used as allowed in Article 8. If Contractor fails to include all Article 8.3.2 items in its Baseline Schedule or Schedule Updates and the District either utilizes an Unapproved Schedule under Article 8.3.2.12 or does not object to the inclusion of required scheduling items, then all mandatory Schedule inclusions, including float, shall be utilized in the District's discretion. If the Contract Time is exceeded, then Contractor shall be subject to the assessment of Liquidated Damages pursuant to Article 8.4.

- 8.3.2.11 Failure to Meet Requirements. Failure of the Contractor to provide proper Schedules as required by this Article and Article 9 is a material breach of the Contract and grounds for Termination pursuant to Article 14. The District, at its sole discretion, may choose, instead, to withhold, in whole or in part, any Progress Payments or Retention amounts otherwise payable to the Contractor.
- 8.3.2.12 *Use of an Unapproved Baseline Schedule.* If the Baseline Schedule submitted by the Contractor is unacceptable to the District (i.e. failing to meet the requirements of Article 8.3.2) and Contractor does not incorporate or address the written comments to the Baseline Schedule and a Baseline Schedule is not approved, but due to extreme necessity, the District moves forward without an approved Baseline Schedule, Contractor shall diligently revise and meet Schedule update requirements of Article 8 and incorporate all Article 8.3.2 comments in all updates). However, for purposes of Termination pursuant to Article 14, the unapproved Baseline Schedule initially submitted shall be treated as the Baseline Schedule with durations shortened or revised to accommodate all float, all mandatory Schedule requirements under Article 8.3.2, any requirements in the Contract Documents, and all revisions by the District or Architect.

# 8.3.3 <u>Update Schedules</u>

8.3.3.1 *Updates Shall Be Based on Approved Baseline Schedule*. Except in the case where there has not been agreement as to a Baseline Schedule, the approved Baseline Schedule shall be used to build future Schedule updates. Schedule updates shall be a CPM based Schedule consistent with the Baseline Schedule requirements of 8.3.2

In the case that no Baseline has been approved, Schedule updates shall be provided monthly and each update shall incorporate all comments and revisions noted as not complying with the requirements of Article 8.3.2. Contractor shall be held to the Article 8.3.2.12 unapproved Baseline Schedule, inclusive of all Milestones, float, comments and revisions by the District and Architect, all required Baseline Schedule Inclusions under Article 8.3.2, and any requirements in the Contract Documents.

- 8.3.3.2 Schedule Updates. Contractor shall update the approved Schedule each month to address actual start dates and durations, the percent complete on activities, actual completion dates, estimated remaining duration for the Work in progress, estimated start dates for Work scheduled to start at future times and changes in duration of Work items
- 8.3.3.3 Listing of Items Causing Delays. Schedule updates shall provide a listing of activities which are causing delay in the progress of Work and a narrative shall be provided showing a description of problem areas, anticipated delays, and impacts on the Construction Schedule. Simply stating "District Delay" or "Architect Delay" shall be an inadequate listing. Delays shall only be listed if they meet the requirements of Article 8.4.

- 8.3.3.4 Recovery Schedule. In addition to providing a schedule update every thirty (30) days, the Contractor, if requested by the Architect or District, shall take the steps necessary to improve Contractor's progress and demonstrate to the District and Architect that the Contractor has seriously considered how the lost time, the Completion Date, or the Milestones that are required to be met within the terms of the Contract. Contractor shall immediately provide a Recovery Schedule showing how Milestones and the Completion Date will be met. In no case, shall a Recovery Schedule be provided later than ten (10) days following the request for a Recovery Schedule from the Architect or District.
  - a. <u>Failure to Provide a Recovery Schedule</u>. Shall subject Contractor to the assessment of Liquidated Damages for failure to meet the Contract Time. Refusal or failure to provide a Recovery Schedule shall be considered a substantial failure of performance and a material breach of Contract and may result in Termination of the Contract pursuant to Article 14.
  - b. <u>Recovery Schedule Acceleration without Additional Cost.</u> The District may require Contractor prepare a Recovery Schedule showing how the Project shall be accelerated, without any additional cost to the District. The District may order, without additional cost, the following:
    - 1. Increase the number of shifts;
    - 2. Utilize overtime to recover the approved Schedule; and/or
    - 3. Increase the days when Work occurs, including weekends, at the Project and at any manufacturer's plant.
  - c. <u>Recovery Schedule Acceleration without Additional Cost.</u> If Contractor disputes that the Recovery Schedule acceleration shall be issued without additional costs, the Contractor shall submit concurrent with Recovery Schedule acceleration notice pursuant to Articles 8.4.3 and 8.4.4.

## 8.4 EXTENSIONS OF TIME - LIQUIDATED DAMAGES

# 8.4.1 <u>Liquidated Damages</u>

CONTRACTOR AND DISTRICT HEREBY AGREE THAT THE EXACT AMOUNT OF DAMAGES FOR FAILURE TO COMPLETE THE WORK WITHIN THE TIME SPECIFIED IS EXTREMELY DIFFICULT OR IMPOSSIBLE TO DETERMINE. IF THE WORK IS NOT SUBSTANTIALLY COMPLETED IN THE TIME SET FORTH IN THE AGREEMENT, IT IS UNDERSTOOD THAT THE DISTRICT WILL SUFFER DAMAGES. IT BEING IMPRACTICAL AND UNFEASIBLE TO DETERMINE THE AMOUNT OF ACTUAL DAMAGE, IT IS AGREED THE CONTRACTOR SHALL PAY TO THE DISTRICT THE AMOUNT LIQUIDATED DAMAGES SET FORTH IN THE AGREEMENT, FOR EACH CALENDAR DAY OF DELAY IN REACHING SUBSTANTIAL COMPLETION (SEE ARTICLE 1.1.46). CONTRACTOR AND ITS SURETY SHALL BE LIABLE FOR THE AMOUNT THEREOF PURSUANT TO GOVERNMENT CODE SECTION 53069.85.

8.4.2 <u>Delay</u>

Except and only to the extent provided under Article 7 and Article 8, by signing the Agreement, Contractor agrees to bear the risk of delays to Completion of the Work and that Contractor's bid for the Project was made with full knowledge of this risk.

In agreeing to bear the risk of delays to complete the Work, Contractor understands that, except and only to the extent provided otherwise in Article 7 and 8, the occurrence of events that delay the Work shall not excuse Contractor from its obligation to achieve Completion of the Project within the Contract Time, and shall not entitle the Contractor to an adjustment to the Contract time.

#### 8.4.3 Excusable Delay

Contractor shall not be charged for Liquidated Damages because of any delays in completion of Work which are not the fault or negligence of Contractor or its Subcontractors, arising from Rain Float or Project Float, including acts of God, as defined in Public Contract Code section 7105, acts of enemy, epidemics and quarantine restrictions. Contractor shall within five (5) calendar days of beginning of any such delay notify District in writing of causes of delay; thereupon District shall ascertain the facts and extent of delay and grant extension of time for completing Work when, in its judgment, the findings of fact justify such an extension. Extensions of time shall apply only to that portion of Work affected by delay, and shall not apply to other portions of Work not so affected. An extension of time may only be granted after proper compliance with Article 8.3 requiring preparation and submission of a properly prepared CPM schedule.

- 8.4.3.1 Excusable Delay Is Not Compensable. No extended overhead, general conditions costs, impact costs, out-of-sequence costs or any other type of compensation, by any name or characterization, shall be paid to the Contractor for any delay to any activity not designated as a critical path item on the latest approved Project schedule.
- 8.4.3.2 *Notification.* The Contractor shall notify the Architect in writing of any anticipated delay and its cause, in order that the Architect may take immediate steps to prevent, if possible, the occurrence or continuance of delay, and may determine whether the delay is to be considered avoidable or unavoidable, how long it continues, and to what extent the prosecution and completion of the Work might be delayed thereby.
- 8.4.3.3 Extension Request. In the event the Contractor requests an extension of Contract time for unavoidable delay, such request shall be submitted in accordance with the provisions in the Contract Documents governing changes in Work (See Article 7). When requesting time, i.e., extensions, for proposed Change Orders, they must be submitted with the proposed Change Order with full justification and documentation. If the Contractor fails to submit justification with the proposed Change Order it waives its right to a time extension at a later date. Such justification must be based on the official Contract schedule as updated at the time of occurrence of the delay or execution of Work related to any changes to the scope of Work. Blanket or general claims for extra days without specific detailed information as required herein or a blanket or general reservation of rights do not fulfill the requirements of this Article and shall be denied. The justification must include, but is not limited to, the following information:
  - a. The duration of the activity relating to the changes in the Work and the resources (manpower, equipment, material, etc.) required to perform these activities within the stated duration.
  - b. Logical ties to the official Baseline Schedule or Approved Updated Schedule for the proposed changes and/or delay showing the activity/activities in the schedule

whose start or completion dates are affected by the change and/or delay. (A fragnet of any delay of over ten (10) days must be provided.)

The Contractor and District understand and expressly agree that insofar as Public Contract Code section 7102 may apply to changes in the Work or delays under this Contract, the actual delays and damages, if any, and time extensions are intended to, and shall provide, the exclusive and full method of compensation for changes in the Work and construction delays.

## 8.4.4 Notice by Contractor Required

The Contractor shall within five (5) calendar days of beginning of any such delay notify the District in writing of causes of delay with justification and supporting documentation. In the case of a Recovery Schedule pursuant to Article 8.3.3.4, Contractor shall submit written notice concurrent with the Recovery Schedule. District will then ascertain the facts and extent of the delay and grant an extension of time for completing the Work when, in its judgment, the findings of fact justify such an extension. Extensions of time shall apply only to that portion of the Work affected by the delay and shall not apply to other portions of the Work not so affected.

Claims relating to time extensions shall be made in accordance with applicable provisions of Article 7.

- 8.4.4.1 *Adjustment for Compensable Delays*. The Schedule may be adjusted for a delay if, and only if, Contractor undertakes the following:
  - a. Contractor submits a timely COR or CO pursuant to the requirements of Article 7.
  - b. Contractor submits a fragnet showing the critical path delay caused by the COR, CO, Changed Condition, CCD, or ICD
  - c. Contractor has addressed all required float days in the Fragnet.
  - d. Contractor submits a complete breakdown of all costs incurred utilizing the format of Article 7.3.3

# 8.4.5 <u>No Additional Compensation for Coordinating Governmental Submittals and the</u> Resulting Work

CONTRACTOR HAS PLANNED ITS WORK AHEAD OF TIME AND IS AWARE THAT GOVERNMENTAL AGENCIES, SUCH AS THE GAS COMPANIES, ELECTRICAL UTILITY COMPANIES, WATER DISTRICTS AND OTHER AGENCIES MAY HAVE TO APPROVE CONTRACTOR PREPARED DRAWINGS OR APPROVE A PROPOSED INSTALLATION. CONTRACTOR HAS INCLUDED DELAYS AND DAMAGES WHICH MAY BE CAUSED BY SUCH AGENCIES IN CONTRACTOR'S BID AND HAS INCLUDED ADEQUATE TIME IN THE CONTRACTOR'S BASELINE SCHEDULE. FAILURE TO ADEQUATELY PLAN AND SCHEDULE IS NOT A BASIS TO USE GOVERNMENTAL DELAY FLOAT.

## 8.4.6 <u>District Right to Accelerate the Work</u>

The District may direct the Contractor to meet schedule requirements when the Work has been delayed. The District shall compensate the Contractor for the additional costs incurred by acceleration

to the extent that such costs are directly attributable to the acceleration and are incurred through no fault or negligence of the Contractor.

8.4.6.1 *Management of Acceleration*. Contractor acceleration shall not include Work that is part of the scope of Work detailed in the Plans and Specifications. Instead, the acceleration costs shall be premium or overtime and quantifiable additional work added to the Project meant to accelerate the Project. Contractor is directed to keep consistent crews on the Project so time can be tracked. If crews are circulated off the Project or crews brought in only for overtime, the District may be charged for Contract Work and not accelerated time. In such case, the District may object to the costs submitted.

8.4.6.2 *Costs for Acceleration*. Cost for Acceleration shall be supported by backup documentation, and time sheets signed by the Inspector for each day work has been performed, at or near the time when the Work was performed. A listing on the time sheet shall document all labor, materials and services utilized that day and provide areas of work, and amount of work performed. Contractor shall comply with submission requirements of Article 7.7.

# ARTICLE 9 PAYMENTS AND COMPLETION

## 9.1 CONTRACT SUM

The Contract Sum or Contract Price is stated in the Agreement and, including authorized adjustments, is the total amount payable by the District to the Contractor for performance of the Work under the Contract Documents.

## 9.2 COST BREAKDOWN

## 9.2.1 <u>Required Information</u>

Contractor shall furnish the following:

- a. Within ten (10) days after Notice to Proceed, a detailed breakdown of the Contract Price (hereinafter "Schedule of Values") for each Project, Site, building, Milestone or other meaningful method to measure the level of Project Completion as determined by the District shall be submitted as a Submittal for the Project.;
- b. Within ten (10) days after the date of the Notice to Proceed, a schedule of estimated monthly payment requests due the Contractor showing the values and construction time of the various portions of the Work to be performed by it and by its Subcontractors or material and equipment suppliers containing such supporting evidence as to its correctness as the District may require;
- c. Within ten (10) days after the date of the Notice to Proceed, address, telephone number, telecopier number, California State Contractors License number, classification and monetary value of all subcontracts for parties furnishing labor, material, or equipment for completion of the Project.

## 9.2.2 Information and Preparation of Schedule of Values

- 9.2.2.1 *Break Down of Schedule of Values*. Schedule of Values shall be broken down by Project, site, building, Milestone, or other meaningful method to measure the level of Project Completion as determined by the District.
- 9.2.2.2 Based on Contractor Bid Costs. The Schedule of Values shall be based on the costs from Contractor's bid to the District. However, the submission of the Schedule of Values shall not be front loaded so the Contractor is paid a greater value than the value of the Work actually performed and shall not shift funds from parts of the Project that are later to Work that is performed earlier.
- 9.2.2.3 <u>Largest Dollar Value for Each Line Item</u>. Identify Subcontractors and materials suppliers proposed to provide portions of Work equal to or greater than ten thousand dollars (\$10,000) or one-half of one percent (0.5%) of their Contract Price, whichever is less.
- 9.2.2.4 *Allowances*. Any Allowances provided for in the Contract shall be a line item in the Schedule of Values.

9.2.2.5 *Labor and Materials Shall Be Separate*. Labor and Materials shall be broken into two separate line items unless specifically agreed in writing by the District.

# 9.2.3 <u>District Approval Required</u>

The District shall review all submissions received pursuant to Article 9.2 in a timely manner. All submissions must be approved by the District before becoming the basis of any payment.

## 9.3 PROGRESS PAYMENTS

#### 9.3.1 Payments to Contractor

Unless there is a resolution indicating that the Work for the Project is substantially complex, within thirty-five (35) days after approval of the Request for Payment, Contractor shall be paid a sum equal to ninety-five percent (95%) of the value of the Work performed (as certified by Architect and Inspector and verified by Contractor) up to the last day of the previous month, less the aggregate of previous payments. In the case of a Project designated substantially complex, the sum paid to the Contractor shall be equal to ninety percent (90%) of the value of the Work performed (as certified by the Architect and Inspector and verified by Contractor). The value of the Work completed shall be the Contractor's best estimate. Work completed as estimated shall be an approximation or estimate only and no mistake, inaccuracy, error or falsification in said any approved estimate shall operate to release the Contractor, or any Surety upon any bond, from damages arising from such Work, or from the District's enforcement of each and every provision of this Contract including but not limited to the Performance Bond and Payment Bond. The District shall have the right to subsequently to correct any mistake, inaccuracy, error or falsification made or otherwise set forth in any approved Request for Payment and such correction may occur in any future Payment Application or in the Retention Payment to the Contractor. No Surety upon any bond shall be relieved, released or exonerated of its obligations under this Contract or any applicable bond when the District is unable to correct an overpayment to the Contractor due to any abandonment by the Contractor or termination by the District.

The Contractor shall not be entitled to have any payment requests processed, or be entitled to have any payment made for Work performed, so long as any lawful or proper direction given by the District concerning the Work, or any portion thereof, remains incomplete.

Notwithstanding anything to the contrary stated above, the Contractor may include in its Request for Payment the value of any structural steel, glue laminated beams, trusses, bleachers and other such custom-made materials prepared specifically for the Project and unique to the Project so long as all of the following requirements are satisfied:

- a. The aggregate cost of materials stored off-site shall not exceed Twenty-Five Thousand Dollars (\$25,000) at any time or as otherwise agreed to be District in writing;
- b. Title to such materials shall be vested in the District as evidenced by documentation satisfactory in form and substance to the District, including, without limitation, recorded financing statements, UCC filings and UCC searches;
- c. With each Contractor Request for Payment, the Contractor shall submit to the District a written list identifying each location where materials are stored off-site (which must be a bonded warehouse) and the value of the materials at each

location. The Contractor shall procure insurance satisfactory to the District (in its reasonable discretion) for materials stored off-site in an amount not less than the total value thereof:

- d. The consent of any Surety shall be obtained to the extent required prior to payment for any materials stored off-site;
- e. Representatives of the District shall have the right to make inspections of the storage areas at any time; and
- f. Such materials shall be: (1) protected from diversion, destruction, theft and damage to the reasonable satisfaction of the District; (2) specifically marked for use on the Project; and (3) segregated from other materials at the storage facility.

## 9.3.2 Purchase of Materials and Equipment and Cost Fluctuations

The Contractor is required to order, obtain, and store materials and equipment sufficiently in advance of its Work at no additional cost or advance payment from District to assure that there will be no delays. Contractor understands that materials fluctuate in value and shall have adequately addressed market fluctuations through agreements with Contractor vendors or by other means. Contractor further understands and incorporates into Contractor's bid cost any wage rate increases during the Project for the Contractor's labor force as well as all other Subcontractor and vendor labor forces. District shall not be responsible for market fluctuations in costs or labor rate increases during the Project. Contractor further has incorporated any and all cost increases in areas of Work where there may be schedule variations so that cost increases are not passed through to the District.

#### 9.3.3 No Waiver

No payment by District hereunder shall be interpreted so as to imply that District has inspected, approved, or accepted any part of the Work. Contractor specifically understands that Title 24 Section 4-343 which states:

"It is the duty of the contractor to complete the work covered by his or her contract in accordance with the approved Plans and Specifications therefore. The contractor in no way is relieved of any responsibility by the activities of the Architect, Engineer, Inspector or DSA in the performance of such duties... In no case, however, shall the instruction of the Architect or registered Engineer be construed to cause work to be done with is not in conformity with the approved Plans, Specifications, and change orders..."

Notwithstanding any payment, the District may enforce each and every provision of this Contract which includes, but is not limited to, the Performance Bond and Payment Bond. The District may correct any error subsequent to any payment. In no event shall the Contractor or the Surety be released or exonerated from performance under this Contract when the District overpays the Contractor based upon any mistake, inaccuracy, error or falsification in any estimate that is included in any Request for Payment.

#### 9.3.4 Issuance of Certificate of Payment

The Architect shall, within seven (7) days after receipt of the Contractor's Application for Payment, either approve such payment or notify the Contractor in writing of the Architect's reasons for

withholding approval in whole or in part as provided in Article 9.6. The review of the Contractor's Application for Payment by the Architect is based on the Architect's observations at the Project and the data comprising the Application for Payment that the Work has progressed to the point indicated and that, to the best of the Architect's knowledge, information, and belief, the quality of the Work is in accordance with the Contract Documents. In some cases, the Architect may act upon or rely on the evaluation of the Work by the Inspector. This review of Payment Applications is sometimes called a "Pencil Draft." District's return of a Pencil Draft shall constitute the District's dispute of the Payment Application that has been submitted. Contractor shall promptly respond to Pencil Drafts or Contractor's Payment Applications may be delayed. Contractor's failure to promptly respond to a Pencil Draft shall qualify as a delay in the prompt payment of a Request for Payment or Request for Retention. The foregoing representations are subject to: (1) an evaluation of the Work for conformance with the Contract Documents, (2) results of subsequent tests and inspections, (3) minor deviations from the Contract Documents correctable prior to completion, and (4) specific qualifications expressed by the Architect. The issuance of a Certificate for Payment will further constitute the Contractor's verified representation that the Contractor is entitled to payment in the amount certified.

## 9.3.5 Payment of Undisputed Contract Payments

In accordance with Public Contract Code section 7100, payments by the District to the Contractor for any and all undisputed amounts (including all Progress Payments, Final Payments or Retention Payment) is contingent upon submission of a proper and accurate Payment Application and the Contractor furnishing the District with a release of all Claims against the District related to such undisputed amounts. Disputed Contract Claims in stated amounts may be specifically excluded by the Contractor from the operation of the release. If, however, the Contractor specifically excludes any Claims, the Contractor shall provide details such as a specific number of disputed days or costs of any such exclusion in accordance with Articles 4.6 and 7.7.

## 9.4 APPLICATIONS FOR PROGRESS PAYMENTS

#### 9.4.1 Procedure

- 9.4.1.1 Application for Progress. On or before the fifth (5th) day of each calendar month during the progress of the Work, Contractor shall submit to the Architect an itemized Application for Progress Payment for operations completed. Such application shall be notarized, if required, and supported by the following or such portion thereof as Architect requires:
  - 1. The amount paid to the date of the Payment Application to the Contractor, to all its Subcontractors, and all others furnishing labor, material, or equipment for its Contract;
  - 2. The amount being requested under the Payment Application by the Contractor on its own behalf and separately stating the amount requested on behalf of each of the Subcontractors and all others furnishing labor, material, and equipment under the Contract;
  - 3. The balance that will be due to each of such entities after said payment is made;
  - 4. A certification that the As-Built Drawings and Annotated Specifications are current:

- 5. Itemized breakdown of Work done for the purpose of requesting partial payment;
- 6. An updated or approved Baseline Schedule or other Schedule updates in conformance with Article 8;
- 7. Failure to submit an updated Schedule for the month or any previous month;
- 8. The additions to and subtractions from the Contract Price and Contract Time;
- 9. A summary of the Retention held;
- 10. Material invoices, evidence of equipment purchases, rentals, and other support and details of cost as the District may require from time to time;
- 11. The percentage of completion of the Contractor's Work by line item;
- 12. An updated Schedule of Values from the preceding Application for Payment;
- 13. Prerequisites for Progress Payments; and
- 14. Any other information or documents reasonably requested by the District, Architect, Inspector or CM (if applicable).
- 9.4.1.2 *First Payment Request.* The following items, if applicable, must be completed before the first payment request will be accepted for processing:
  - 1. Installation of the Project sign;
  - 2. Receipt by Architect of Submittals;
  - 3. Installation of field office;
  - 4. Installation of temporary facilities and fencing;
  - 5. Submission of documents listed in the Article 9.2 relating to Contract Price breakdown;
  - 6. Preliminary schedule analysis, due within 10 days after Notice to Proceed;
  - 7. Contractor's Baseline Schedule (to be CPM based in conformance with Article 8);
  - 8. Schedule of unit prices, if applicable;
  - 9. Submittal Schedule;
  - 10. Copies of necessary permits;

- 11. Copies of authorizations and licenses from governing authorities;
- 12. Initial progress report;
- 13. Surveyor qualifications;
- 14. Written acceptance of District's survey of rough grading, if applicable;
- 15. List of all Subcontractors, with names, license numbers, telephone numbers, and scope of work;
- 16. All bonds and insurance endorsements; and
- 17. Resumes of General Contractor's Project Manager, and if applicable, job site secretary, record documents recorder, and job site Superintendent.
- 9.4.1.3 *Second Payment Request.* The second payment request will not be processed until all Submittals and Shop Drawings have been accepted for review by the Architect.
- 9.4.1.4 *All Payment Requests.* No payment requests will be processed unless Contractor has submitted copies of the certified payroll records for the Work which correlates to the payment request and a proper CPM schedule pursuant to Article 8 is submitted.
  - 9.4.1.5 Final Payment Application (90% or 95%). See Article 9.11.1
  - 9.4.1.6 Final Payment Application (100%). See Article 9.11.3

## 9.5 STOP NOTICE CLAIMS AND WARRANTY OF TITLE

The Contractor warrants title to all Work. The Contractor further warrants that all Work is free and clear of liens, claims, security interests, stop notices, or encumbrances in favor of the Contractor, Subcontractors, material and equipment suppliers, or other persons or entities making a claim by reason of having provided labor, materials, and equipment relating to the Work. Failure to keep work free of liens, stop notices, claims, security interests or encumbrances is grounds to make a claim against Contractor's Payment and Performance Bond to immediately remedy and defend.

If a lien or stop notice of any nature should at any time be filed against the Work or any District property, by any entity which has supplied material or services at the request of the Contractor, Contractor and Contractor's Surety shall promptly, on demand by District and at Contractor's and Surety's own expense, take any and all action necessary to cause any such lien or stop notice to be released or discharged immediately therefrom.

If the Contractor fails to furnish to the District within ten (10) calendar days after written demand by the District, satisfactory evidence that a lien or stop notice has been so released, discharged, or secured, then District may discharge such indebtedness and deduct the amount required therefor, together with any and all losses, costs, damages, and attorney's fees and expense incurred or suffered by District from any sum payable to Contractor under the Contract. In addition, any liens, stop notices, claims, security interests or encumbrances shall trigger the indemnification requirements under Article 3.15 and the Agreement Form, and shall act as a trigger under Civil Code section 2778 and 2779 requiring reimbursement for any and all costs following the District's written demand has been made. Any withholdings by the District for

stop notices in accordance with Civil Code section 9358 shall not be a basis by the Contractor to make a Claim for interest penalties under Public Contract Code sections 7107 or 20104.50.

# 9.6 DECISIONS TO WITHHOLD PAYMENT

#### 9.6.1 Reasons to Withhold Payment

The District may withhold payment in whole, or in part, to the extent reasonably necessary to protect the District if, in the District's opinion, the representations to the District required by Article 9.4 cannot be made. The District may withhold payment, in whole, or in part, to such extent as may be necessary to protect the District from loss because of, but not limited to:

- a. Defective Work not remedied;
- b. Stop notices served upon the District;
- c. Liquidated Damages assessed against the Contractor;
- d. The cost of Completion of the Contract if there exists reasonable doubt that the Work can be Completed for the unpaid balance of any Contract Price or by the completion date;
- e. Damage to the District or other contractor;
- f. Unsatisfactory prosecution of the Work by the Contractor;
- g. Failure to store and properly secure materials;
- h. Failure of the Contractor to submit on a timely basis, proper and sufficient documentation required by the Contract Documents, including, without limitation, acceptable monthly progress schedules, Shop Drawings, Submittal schedules, Schedule of Values, Product Data and samples, proposed product lists, executed Change Order, Construction Change Documents, and verified reports;
- i. Failure of the Contractor to maintain As-Built Drawings;
- j. Erroneous estimates by the Contractor of the value of the Work performed, or other false statements in a Payment Application;
- k. Unauthorized deviations from the Contract Documents (including but not limited to Unresolved Notices of Deviations (DSA Form 154));
- 1. Failure of the Contractor to prosecute the Work in a timely manner in compliance with established progress schedules and completion dates.
- m. Failure to properly pay prevailing wages as defined in Labor Code section 1720, et seq.;
- n. Failure to properly maintain or clean up the Site;

- o. Payments to indemnify, defend, or hold harmless the District;
- p. Any payments due to the District including but not limited to payments for failed tests, or utilities changes or permits;
- q. Failure to submit an acceptable Baseline Schedule or any Schedule or Schedule update in accordance with Article 8;
- r. Failure to pay Subcontractor or suppliers as required by Article 9.8.1
- s. Failure to secure warranties, including the cost to pay for warranties;
- t. Failure to provide releases from material suppliers or Subcontractors when requested to do so;
- u. Items deducted pursuant to Article 2.2;
- v. Incomplete Punch List items under Article 9.9.1.1 which have gone through the Article 2.2 process; or
- w. Allowances that have not been used.

## 9.6.2 Reallocation of Withheld Amounts

District may, in its discretion, apply any withheld amount to payment of outstanding claims or obligations as defined in Article 9.6.1 and 9.5. In so doing, District shall make such payments on behalf of Contractor. If any payment is so made by District, then such amount shall be considered as a payment made under Contract by District to Contractor and District shall not be liable to Contractor for such payments made in good faith. Such payments may be made without prior judicial determination of claim or obligation. District will render Contractor an accounting of such funds disbursed on behalf of Contractor.

If Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents or fails to perform any provision thereof, District may, after ten (10) calendar days written notice to the Contractor and without prejudice to any other remedy make good such deficiencies. The District shall adjust the total Contract price by reducing the amount thereof by the cost of making good such deficiencies. If District deems it inexpedient to correct Work which is damaged, defective, or not done in accordance with Contract provisions, an equitable reduction in the Contract Price (of at least 150% of the estimated reasonable value of the nonconforming Work) shall be made therefor.

## 9.6.3 Payment After Cure

When the grounds for declining approval are removed, payment shall be made for amounts withheld because of them. No interest shall be paid on any retainage or amounts withheld due to the failure of the Contractor to perform in accordance with the terms and conditions of the Contract Documents.

# 9.7 <u>NONCONFORMING WORK</u>

Contractor shall promptly remove from premises all Work identified by District as failing to conform to the Contract whether incorporated or not. Contractor shall promptly replace and re-execute its

own Work to comply with the Contract without additional expense to District and shall bear the expense of making good all Work of other contractors destroyed or damaged by such removal or replacement.

If Contractor does not remove such Work which has been identified by District as failing to conform to the Contract Documents within a reasonable time, fixed by written notice, District may remove it and may store the material at Contractor's expense. If Contractor does not pay expenses of such removal within ten (10) calendar days' time thereafter, District may, upon ten (10) calendar days' written notice, sell such materials at auction or at private sale and shall account for net proceeds thereof, after deducting all costs and expenses that should have been borne by Contractor.

## 9.8 SUBCONTRACTOR PAYMENTS

## 9.8.1 <u>Payments to Subcontractors</u>

No later than ten (10) days after receipt, or pursuant to Business and Professions Code section 7108.5, the Contractor shall pay to each Subcontractor, out of the amount paid to the Contractor on account of such Subcontractor's portion of the Work, the amount to which said Subcontractor is entitled. The Contractor shall, by appropriate agreement with each Subcontractor, require each Subcontractor to make payments to Sub-subcontractors in a similar manner.

## 9.8.2 <u>No Obligation of District for Subcontractor Payment</u>

The District shall have no obligation to pay, or to see to the payment of, money to a Subcontractor except as may otherwise be required by law.

## 9.8.3 Payment Not Constituting Approval or Acceptance

An approved Request for Payment, a progress payment, a Certificate of Substantial Completion, or partial or entire use or occupancy of the Project by the District shall not constitute acceptance of Work that is not in accordance with the Contract Documents.

## 9.8.4 Joint Checks

District shall have the right, if necessary for the protection of the District, to issue joint checks made payable to the Contractor and Subcontractors and material or equipment suppliers. The joint check payees shall be responsible for the allocation and disbursement of funds included as part of any such joint payment. In no event shall any joint check payment be construed to create any contract between the District and a Subcontractor of any tier, any obligation from the District to such Subcontractor, or rights in such Subcontractor against the District. The District may choose to issue joint checks at District's sole discretion and only after all the requirements of that particular school district and county are specifically met. Some school districts cannot issue joint checks, so the ability to issue joint checks depends on the school district and the specific circumstances.

## 9.9 COMPLETION OF THE WORK

## 9.9.1 Close-Out Procedures

9.9.1.1 *Incomplete Punch Items*. When the Contractor considers the Work Substantially Complete (See Article 1.1.46 for definition of Substantially Complete), the Contractor shall prepare and submit to the District a comprehensive list of minor items to be completed or corrected

(hereinafter "Incomplete Punch Items" or "Punch List"). The Contractor and/or its Subcontractors shall proceed promptly to complete and correct the Incomplete Punch Items listed. Failure to include an item on such list does not alter the responsibility of the Contractor to complete all Work in accordance with the Contract Documents. Contractor is aware that Title 24 Section 4-343(a) provides:

"RESPONSIBILITIES. IT IS THE DUTY OF THE CONTRACTOR TO COMPLETE THE WORK COVERED BY HIS OR HER CONTRACT IN ACCORDANCE WITH THE APPROVED PLANS AND SPECIFICATIONS THEREFOR. THE CONTRACTOR IN NO WAY IS RELIEVED OF ANY RESPONSIBILITY BY THE ACTIVITIES OF THE ARCHITECT, ENGINEER, INSPECTOR OR DSA IN THE PERFORMANCE OF SUCH DUTIES.

9.9.1.2 Punch List Is Prepared Only After the Project Is Substantially Complete. If any of the conditions noted in Article 1.1.46 as defining Substantial Completion are not met, the Inspector, Architect or District may reject Contractor's Incomplete Punch Items as premature. If the Architect and Inspector commence review of Incomplete Punch Items, all rights are reserved until the Project actually meets the definition of Substantially Complete. Liquidated Damages, warranties, and other contractual rights are not affected by Incomplete Punch Items unless otherwise addressed in these General Conditions.

Once the Inspector and the Architect determine the Project is Substantially Complete, a Certificate of Substantial Completion shall be issued. The Inspector and Architect shall prepare a Punch List of items which is an inspection report of the Work, if any, required in order to complete the Contract Documents and ensure compliance with the DSA Approved Plans so the Project may be Completed by the Contractor and a final DSA Close-Out is approved. When all Work for the Project is Complete, including Punch Lists and all Work complies with the approved Contract Documents and Change Orders, the Project has reached Final Completion.

9.9.1.3 Time for Completion of Punch List. Contractor shall only be given a period of no more than thirty (30) days to complete the Punch List for the Project. During the Punch List period, the Contractor's Superintendent and Project Manager shall remain engaged in the Project and shall not be removed or replaced. If the Punch List is not completed at the end of the Punch List time, then Contractor shall issue a valued Punch List within 5 days after the date the Punch List time ends. If Contractor does not issue such a list, the District or Architect may issue a valued Punch List to the Contractor and withhold up to 150% of the value of the Punch List Work pursuant to Article 2.2 of this Agreement.

Failure to issue a timely written request for additional time to complete Punch List shall result in the deletion of the remaining Punch List Work pursuant to Article 2.2 and the issuance of a Deductive Change Order.

- Extension of Time to Complete Punch List. If Contractor cannot finish the Punch List Work during the time period allotted under Article 9.9.1.3, the Contractor may make a written request for a Non-Compensable Punch List time extension accompanied by an estimate of the number of additional days it will take to complete the Punch List Work for a written consent from the District to allow continued Punch List Work. Punch List time extensions are a maximum of thirty (30) days for each request and must be accompanied by an itemized valued Punch List.
- b. If there is no valued Punch List accompanying any request or if Contractor intends to undertake Punch List without the continued support and supervision of its

Superintendent and Project Manager (as required under Article 3.2), the District, Construction Manager or Architect may issue a valued Punch List, reject the Punch List Time Extension and deduct 150% of the valued Punch List pursuant to Article 2.2 and proceed to Close-Out the Project. Contractor shall cease work on the Project and proceed to complete Contractor's Retention Payment Application and complete the Work for the Project required pursuant to Article 9.11.3.

9.9.1.4 District Rejection of Written Request for Punch List Time Extensions. Following sixty (60) Days of Punch List under Article 9.9.1.3, the District has the option of rejecting Punch List Time Extension requests. The District may proceed under Article 2.2 and deduct the value of remaining Punch List Work pursuant to Article 2.2. If the District rejects the Punch List Time Extension request then Contractor shall cease Work on the Project and proceed to Final Inspection pursuant to Article 9.11.2.

9.9.1.5 Punch List Liquidated Damages to Compensate for Added District Project Costs. If the total time utilized for Punch List exceeds sixty (60) days [the thirty (30) day period under Article 9.9.1.3 plus an additional thirty (30) day period that has been requested in writing], and the District grants an additional written Punch List Time Extension that exceeds sixty (60) days of Punch List, then Contactor shall be charged Liquidated Damages of at least \$750 per day for continued Punch List Work to partially compensate the Inspector, Architect, and Construction Manager's extended time on the Project. This Punch List Liquidated Damage number is based on anticipated cost for an Inspector on site and additional costs for the Architect and Construction Manager to reinspect Punch List items and perform the administration of the Close-out.

Contractor received thirty (30) days without any charges for Punch List Liquidated Damages and is placed on notice pursuant to this Article 9.9.1.5 that \$750 is due for each day of Punch List that exceeds sixty (60) days at \$750, a cost much lower than typical (and actual) costs for Inspection, Architect and Construction Manager time required during Punch List. Starting at ninety (90) days of Punch List (an excessive number of days to complete Punch List), the District shall be entitled to adjust Punch List Liquidated Damages to an estimate of the actual costs incurred to oversee, monitor and inspect the Punch List. If costs exceed \$750 per day, the anticipated extended contract charges for Inspection, Architect, Construction Manager, and any other costs that will be incurred due to the extended Punch List shall be itemized and a daily rate of Punch List Liquidated Damages shall be presented in writing to the Contractor within five (5) days following the receipt of a written request for Punch List Time Extension by the Contractor that extends the Punch List time beyond ninety (90) days. This written notice of actual Punch List Liquidated Damages may be provided to the Contractor at any time following the first written request for Punch List Time extension requested under Article 9.9.1.3. The adjusted actual Punch List Liquidated Damage amount shall be applicable as Punch List Liquidated Damages commencing on the ninetieth (90th) day of Punch List.

### 9.9.2 Close-Out Requirements for Final Completion of the Project

- a. <u>Utility Connections</u>. Buildings shall be connected to water, gas, sewer, and electric services, complete and ready for use. Service connections shall be made and existing services reconnected
- b. <u>As-Builts Up to Date and Complete</u>. The intent of this procedure is to obtain an exact "As-Built" record of the Work upon completion of the project. The following information shall be carefully and correctly drawn on the prints and all items shall be accurately located and dimensioned from finished surfaces of building walls on all As-Built Drawings

- 1. The exact location and elevations of all covered utilities, including valves, cleanouts, etc. must be shown on As-Built Drawings
- 2. Contractor is liable and responsible for inaccuracies in As-Built Drawings, even though they become evident at some future date.
- 3. Upon completion of the Work and as a condition precedent to approval of Retention Payment, Contractor shall obtain the Inspector's approval of the "As-Built" information. When completed, Contractor shall deliver corrected sepias and/or a Diskette with an electronic file in a format acceptable to the District.
- 4. District may withhold the cost to hire a draftsman and potholing and testing service to complete Record As-Built Drawings at substantial cost if the Contractor does not deliver a complete set of Record As-Built Drawings. This shall result in withholding of between \$10,000 to \$20,000 per building that does not have a corresponding Record As Built Drawing.
- c. <u>Any Work not installed</u> as originally indicated on Drawings
- d. <u>All DSA Close-Out requirements</u> (See DSA Certification Guide) Contractor is also specifically directed to Item 3.2 in the DSA Certification Guide and the applicable certificates for the DSA-311 form.
- e. <u>Submission of Form 6-C.</u> Contractor shall be required to execute a Form 6-C as required under Title 24 Sections 4-343. The Contractor understands that the filing with DSA of a Form 6-C is a requirement to obtain final DSA Approval of the construction by Contractor and utilized to verify under penalty of perjury that the Work performed by Contractor complies with the DSA approved Contract Documents. The failure to file a DSA Form 6C has two consequences. First, the Construction of the Project will not comply with the design immunity provisions of Government Code section 830.6 and exposes the District and the individual Board members to personal liability for injuries that occur on the Project.

Secondly, under DSA IR A-20, since the Project cannot be Certified by DSA, no future or further Projects will be authorized so Contractor will have essentially condemned the campus from any future modernization or addition of new classrooms through their failure to file the DSA Form 6C.

- 1. Execution of the DSA Form 6-C is Mandatory. Refusal to execute the Form 6-C, which is a Final DSA Verified Report that all Work performed complies with the DSA approved Contract Documents is a violation of Education Code section 17312 and shall be referred to the Attorney General for Prosecution.
- 2. Referral to the District Attorney for Extortion. If the Contractor's refusal to execute the DSA Form 6C is to leverage a Dispute, Claim or Litigation, then the matter shall also be referred to the District Attorney for prosecution for extortion.

- 3. Contractor shall be Responsible for All Costs to Certify the Project. The District may certify the Project complies with Approved Plans and Specifications by utilizing the procedures under the Project Certification Guide (located at the DSA website). All costs for professionals, inspection, and testing required for an alternate Project Certification shall be the Contractor's responsibility and the District reserves its right to institute legal action against the Contractor and Contractor's Surety for all costs to certify the Project and all costs to correct Non-Compliant Work that is discovered during the Alternate Certification Process.
- f. ADA Work that must be corrected to receive DSA certification. See Article 12.2.
- g. <u>Maintenance Manuals</u>. At least thirty (30) days prior to final inspection, three (3) copies of complete operations and maintenance manuals, repair parts list, service instructions for all electrical and mechanical equipment, and equipment warranties shall be submitted. All installation, operating, and maintenance information and Drawings shall be bound in 8½" x 11" binders. Provide a table of contents in front and all items shall be indexed with tabs. Each manual shall also contain a list of Subcontractors, with their addresses and the names of persons to contact in cases of emergency. Identifying labels shall provide names of manufactures, their addresses, ratings, and capacities of equipment and machinery.
  - 1. Maintenance manuals shall also be delivered in electronic media for the Project. Any demonstration videos shall also be provided on electronic media.
- h. <u>Inspection Requirements</u>. Before calling for final inspection, Contractor shall determine that the following Work has been performed:
  - 1. The Work has been completed;
  - 2. All fire/life safety items are completed and in working order;
  - 3. Mechanical and electrical Work complete, fixtures in place, connected and tested;
  - 4. Electrical circuits scheduled in panels and disconnect switches labeled;
  - 5. Painting and special finishes complete;
  - 6. Doors complete with hardware, cleaned of protective film relieved of sticking or binding and in working order;
  - 7. Tops and bottoms of doors sealed;
  - 8. Floors waxed and polished as specified;
  - 9. Broken glass replaced and glass cleaned;

- 10. Grounds cleared of Contractor's equipment, raked clean of debris, and trash removed from Site:
- 11. Work cleaned, free of stains, scratches, and other foreign matter, replacement of damaged and broken material;
- 12. Finished and decorative work shall have marks, dirt and superfluous labels removed;
- 13. Final cleanup, as in Article 3.12;
- 14. All Work pursuant to Article 9.11.2; and
- 15. Furnish a letter to District stating that the District's Representative or other designated person or persons have been instructed in working characteristics of mechanical and electrical equipment.

## 9.9.3 Costs of Multiple Inspections

More than two (2) requests of the District to make inspections required under Article 9.9.1 shall be considered an additional service of Architect, Inspector, Engineer or other consultants shall be the Contractor's responsibility pursuant to Article 4.5 and all subsequent costs will be prepared as a Deductive Change Order.

## 9.10 PARTIAL OCCUPANCY OR USE

## 9.10.1 <u>District's Rights</u>

The District may occupy or use any completed or partially completed portion of the Work at any stage. The District and the Contractor shall agree in writing to the responsibilities assigned to each of them for payments, security, maintenance, heat, utilities, damage to the Work, insurance, the period for correction of the Work, and the commencement of warranties required by the Contract Documents. If District and Contractor cannot agree as to responsibilities such disagreement shall be resolved pursuant to Article 4.6. When the Contractor considers a portion complete, the Contractor shall prepare and submit a Punch List to the District as provided under Article 9.9.1.

## 9.10.2 Inspection Prior to Occupancy or Use

Immediately prior to such partial occupancy or use, the District, the Contractor, and the Architect shall jointly inspect the area to be occupied or portion of the Work to be used in order to determine and record the condition of the Work.

#### 9.10.3 No Waiver

Unless otherwise agreed upon, partial occupancy or use of a portion or portions of the Work shall not constitute acceptance of Work not complying with the requirements of the Contract Documents.

# 9.11 COMPLETION AND FINAL PAYMENT

# 9.11.1 <u>Final Payment (90% Billing if Substantially Complex Finding and 95% Billing If No Finding Is Made)</u>

The following items must be completed before the Final Payment Application will be accepted for processing at Substantial Completion of the Project:

- a. Inspector sign-off of each item in the DSA 152 Project Inspection Card;
- b. The Project has reached the Punch List items under Article 9.9.1.2 and the Project has been determined to be Substantially Complete under Article 1.1.46;
- c. Removal of temporary facilities and services;
- d. Testing, adjusting and balance records are complete;
- e. Removal of surplus materials, rubbish, and similar elements;
- f. Changeover of door locks;
- g. Deductive items pursuant to Article 9.6 and Article 2.2; and
- h. Completion and submission of all final Change Orders for the Project.

# 9.11.2 <u>Final Inspection (Punch List Completion)</u>

Contractor shall comply with Punch List procedures under Article 9.9.1.1, and maintain the presence of Project Superintendent and Project Manager (not replacement project superintendent or project manager) until the Punch List is complete to ensure proper and timely completion of the Punch List. Under no circumstances shall Contractor demobilize its forces prior to completion of the Punch List.

Upon completion of the Work under Article 9.9.1, the Contractor shall notify the District and Architect, who shall again inspect such Work. If the Architect and the District find the Work contained in the Punch List acceptable under the Contract Documents, the Work shall have reached Final Completion. Architect shall notify Contractor, who shall then submit to the Architect its Application for Retention Payment. This Application for Retention Payment shall contain any deductions under Article 9.6, including but not limited to incomplete Punch List items under Article 9.9.1.

Upon receipt and approval of Application for Retention Payment, the Architect shall issue a Form 6 stating that to the best of its knowledge, information, and belief, and on the basis of its observations, inspections, and all other data accumulated or received by the Architect in connection with the Work, such Work has been completed in accordance with the Contract Documents. The District shall thereupon inspect such Work and either accept the Work as complete or notify the Architect and the Contractor in writing of reasons why the Work is not complete. Upon acceptance of the Work of the Contractor as fully complete (which, absent unusual circumstances, will occur when the Punch List items have been satisfactorily completed), the District shall record a Notice of Completion with the County Recorder, and the Contractor shall, upon receipt of payment from the District, pay the amounts due Subcontractors.

If the Architect and the District find that the Work contained in the Punch List is unacceptable, then Contractor shall issue a valued Punch List within 5 days after the date the Punch List time ends. If Contractor does not issue such a list, the District or Architect may issue a valued Punch List to the Contractor and withhold up to 150% of the value of the Punch List Work pursuant to Article 2.2 of this Agreement.

## 9.11.3 Retainage (100% Billing for the Entire Project)

The retainage, less any amounts disputed by the District or which the District has the right to withhold pursuant to the Contract Documents (including but not limited to incomplete Punch List items under Article 9.9.1), shall be paid after approval by the District of the Application for Retention Payment, after the satisfaction of the conditions set forth in Article 9, the Final Inspection under Article 9.11.2 is completed, and after thirty-five (35) days after the acceptance of the Work and recording of the Notice of Completion by District. No interest shall be paid on any retainage, or on any amounts withheld due to a failure of the Contractor to perform, in accordance with the terms and conditions of the Contract Documents, except as provided to the contrary in any escrow agreement between the District and the Contractor.

- a. <u>Procedures for Application for Retention Payment.</u> The following conditions must be fulfilled prior to release of Retention Payment:
  - 1. A full and final waiver or release of all stop notices in connection with the Work shall be submitted by Contractor, including a release of stop notice in recordable form, together with (to the extent permitted by law) a copy of the full and final release of all Stop Notice rights.
  - 2. The Contractor shall have made all corrections, including all Punch List Items, to the Work which are required to remedy any defects therein, to obtain compliance with the Contract Documents or any requirements of applicable codes and ordinances, or to fulfill any of the orders or directions of District required under the Contract Documents.
  - 3. Each Subcontractor shall have delivered to the Contractor all written guarantees, warranties, applications, releases from the Surety and warranty bonds (if applicable) required by the Contract Documents for its portion of the Work.
  - 4. Contractor must have completed all requirements set forth in Article 9.9
  - 5. Contractor must have issued a Form 6C for the Project.
  - 6. The Contractor shall have delivered to the District all manuals and materials required by the Contract Documents.
  - 7. The Contractor shall have completed final clean up as required by Article 3.12

8. Contractor shall have all deductive items under Article 9.6 and Article 2.2 submitted as part of the Retention Payment.

# 9.11.4 Recording of a Notice of Completion After Punch List Period and Final Inspection.

When the Work, or designated portion thereof, is complete or the District has completed the Article 9.6and/or the Article 2.2 process, whichever occurs first, the District will file either a Notice of Completion or a Notice of Completion noting valued Punch List items. Valued Punch List items will be deducted from the Retention Payment.

During the time when Work is being performed on the Punch List, the Project does not meet the definition of "Complete" under Public Contract Code section 7107(c)(1) even if there is "beneficial occupancy" of the Project since that has been no "cessation of labor" on the Project. Completion of Punch List under this Article is not "testing, startup, or commissioning by the public entity or its agent." In other words, the continuing Punch List Work is Contractor labor on the Project until each and every item of Punch List Work is complete or the time periods under Article 9.9.1 have expired.

## 9.11.5 Warranties

Warranties required by the Contract Documents shall commence on the date of Completion of the entire Work. Warranty periods DO NOT commence at Substantial Completion or when a particular Subcontractor work is complete. No additional charges, extras, Change Orders, or Claims may be sought for warranties commencing from the Notice of Completion.

District shall have the right to utilize equipment, test, and operate as necessary for acclimation, or testing without voiding or starting warranties. Taking beneficial occupancy shall not start warranties except in the case where the District agrees, in writing, that warranties shall commence running or where the District is taking phased occupancy of specific buildings or areas and completes separate Punch Lists as further addressed in Article 4.2.7.

# 9.11.6 <u>Time for Submission of Application for Final Payment and Retention Payment (Unilateral Processing of Final and Retention Payment Application).</u>

If Contractor submits a Final Payment Application which fails to include deductive items under Article 9.6, the District or Architect shall note this defective request for Final Payment Application. The Contractor shall be notified that specific deductive items shall be included in the Final Payment Application. If Contractor either continues to submit the Final Payment Application without deductive items under Article 9.6, or a period of 14 calendar days passes after Contractor is provided written notice of deductive items for inclusion in Final Payment Application, then District may either alter the Final Payment Application and recalculate the math on the Final Payment Application to address the Article 9.6 deductive items or process a unilateral Final Payment Application.

#### 9.11.7 Unilateral Release of Retention

After the recordation of the Notice of Completion, or within sixty (60) days following the completion of the Punch List or the expiration of the time for completion of Punch List under Article 9.9.1, if Contractor does not make an Application for Release of Retention, the District may unilaterally release retention less any deducts under Article 9.6 and/or Article 2.2, withholds due to stop notices, or withholdings due to other defective Work on the Project. District may also choose to unilaterally release Retention after deduction of 150% of any disputed items, which may also include items under Article 9.6

and 2.2. If a deduction pursuant to Article 9.6 is made from Retention, a letter deducting specific valued items shall be considered a notice of Default under the terms of the Escrow Agreement.

# 9.12 SUBSTITUTION OF SECURITIES

The District will permit the substitution of securities in accordance with the provisions of Public Contract Code section 22300 as set forth in the form contained in the Bid Documents.

## ARTICLE 10 PROTECTION OF PERSONS AND PROPERTY

## 10.1 SAFETY PRECAUTIONS AND PROGRAMS

## 10.1.1 Contractor Responsibility

The Contractor shall be responsible for all damages to persons or property that occur as a result of its fault or negligence in connection with the prosecution of this Contract and shall take all necessary measures and be responsible for the proper care and protection of all materials delivered and Work performed until completion and final acceptance by the District. All Work shall be solely at the Contractor's risk, with the exception of damage to the Work caused by "acts of God" as defined in Public Contract Code section 7105(b)(2).

Contractor shall take, and require Subcontractor to take, all necessary precautions for safety of workers on the Work and shall comply with all applicable federal, state, local and other safety laws, standards, orders, rules, regulations, and building codes to prevent accidents or injury to persons on, about, or adjacent to premises where Work is being performed and to provide a safe and healthful place of employment. In addition to meeting all requirements of OSHA, Cal-OSHA, state, and local codes, Contractor shall furnish, erect and properly maintain at all times, as directed by District or Architect or required by conditions and progress of Work, all necessary safety devices, safeguards, construction canopies, signs, audible devices for protection of the blind, safety rails, belts and nets, barriers, lights, and watchmen for protection of workers and the public, and shall post danger signs warning against hazards created by such features in the course of construction. Contractor shall designate a responsible member of its organization on the Work, whose duty shall be to post information regarding protection and obligations of workers and other notices required under occupational safety and health laws, to comply with reporting and other occupational safety requirements, and to protect the life, safety and health of workers. The name and position of person so designated shall be reported to District by Contractor. Contractor shall correct any violations of safety laws, rules, orders, standards, or regulations. Upon the issuance of a citation or notice of violation by the Division of Occupational Safety and Health, such violation shall be corrected promptly.

## 10.1.2 Subcontractor Responsibility

Contractor shall require that Subcontractors participate in, and enforce, the safety and loss prevention programs established by the Contractor for the Project, which will cover all Work performed by the Contractor and its Subcontractors. Each Subcontractor shall designate a responsible member of its organization whose duties shall include loss and accident prevention, and who shall have the responsibility and full authority to enforce the program. This person shall attend meetings with the representatives of the various Subcontractors employed to ensure that all employees understand and comply with the programs.

## 10.1.3 <u>Cooperation</u>

All Subcontractors and material or equipment suppliers shall cooperate fully with Contractor, the District, and all insurance carriers and loss prevention engineers.

# 10.1.4 <u>Accident Reports</u>

Subcontractors shall immediately, within two (2) days, report in writing to the Contractor all accidents whatsoever arising out of, or in connection with, the performance of the Work, whether on or off the Site, which caused death, personal injury, or property damage, giving full details and statements of witnesses. In addition, if death or serious injuries or serious damages are caused, the accident shall be reported within four (4) days by telephone or messenger. Contractor shall thereafter immediately, within two (2) days, report the facts in writing to the District and the Architect giving full details of the accident.

#### 10.1.5 First-Aid Supplies at Site

The Contractor will provide and maintain at the Site first-aid supplies which complies with the current Occupational Safety and Health Regulations.

## 10.1.6 <u>Material Safety Data Sheets and Compliance with Proposition 65</u>

Contractor is required to have material safety data sheets available in a readily accessible place at the job site for any material requiring a material safety data sheet per the Federal "hazard communication" standard, or employees' "right-to-know law." The Contractor is also required to properly label any substance brought into the job site, and require that any person working with the material, or within the general area of the material, is informed of the hazards of the substance and follows proper handling and protection procedures.

Contractor is required to comply with the provisions of California Health and Safety Code section 25249, et seq., which requires the posting and giving of notice to persons who may be exposed to any chemical known to the State of California to cause cancer. The Contractor agrees to familiarize itself with the provisions of this Section, and to comply fully with its requirements.

#### 10.1.7 Non-Utilization of Asbestos Material

NO ASBESTOS OR ASBESTOS-CONTAINING PRODUCTS SHALL BE USED IN THIS CONSTRUCTION OR IN ANY TOOLS, DEVICES, CLOTHING, OR EQUIPMENT USED TO EFFECT THIS CONSTRUCTION.

Asbestos and/or asbestos-containing products shall be defined as all items containing, but not limited to, chrysotile, amosite, anthophyllite, tremolite, and antinolite.

Any or all material containing greater than one-tenth of one percent (>.1%) asbestos shall be defined as asbestos-containing material.

All Work or materials found to contain asbestos or Work or material installed with asbestos-containing equipment will be immediately rejected and this Work will be removed at no additional cost to the District.

Decontamination and removal of Work found to contain asbestos or Work installed with asbestos-containing equipment shall be done only under supervision of a qualified consultant, knowledgeable in the field of asbestos abatement and accredited by the Environmental Protection Agency.

The asbestos removal contractor shall be an EPA accredited contractor qualified in the removal of asbestos and shall be chosen and approved by the asbestos consultant, who shall have sole discretion and final determination in this matter.

The asbestos consultant shall be chosen and approved by the District, who shall have sole discretion and final determination in this matter.

The Work will not be accepted until asbestos contamination is reduced to levels deemed acceptable by the asbestos consultant.

Interface of Work under this Contract with Work containing asbestos shall be executed by the Contractor at his risk and at his discretion, with full knowledge of the currently accepted standards, hazards, risks, and liabilities associated with asbestos work and asbestos-containing products. By execution of this Contract, the Contractor acknowledges the above and agrees to hold harmless District and its assigns for all asbestos liability which may be associated with this work and agrees to instruct his employees with respect to the above-mentioned standards, hazards, risks, and liabilities.

## 10.2 SAFETY OF PERSONS AND PROPERTY

#### 10.2.1 The Contractor

The Contractor shall take reasonable precautions for the safety of, and shall provide reasonable protection to prevent damage, injury, or loss to:

- a. Employees on the Work and other persons who may be affected thereby;
- b. The Work, material, and equipment to be incorporated therein, whether in storage on or off the Site, under the care, custody, or control of the Contractor or the Contractor's Subcontractors or Sub-subcontractors; and
- c. Other property at the Site or adjacent thereto such as trees, shrubs, lawns, walks, pavement, roadways, structures, and utilities not designated for removal, relocation, or replacement in the course of construction.

Contractor is constructive owner of Project site as more fully discussed in Article 6.2.

#### 10.2.2 Contractor Notices

The Contractor shall give notices and comply with applicable laws, ordinances, rules, regulations, and lawful orders of public authorities bearing on the safety of persons or property or their protection from damage, injury, or loss.

#### 10.2.3 Safety Barriers and Safeguards

The Contractor shall erect and maintain, as required by existing conditions and performance of the Contract, reasonable safeguards for safety and protection, including posting danger signs and other warnings against hazards, promulgating safety regulations, and notifying owners and users of adjacent sites and utilities.

## 10.2.4 <u>Use or Storage of Hazardous Material</u>

When use or storage of explosives, other hazardous materials or equipment, or unusual methods are necessary for execution of the Work, the Contractor shall exercise utmost care and carry on such activities under supervision of properly qualified personnel. The Contractor shall notify the District

any time that explosives or hazardous materials are expected to be stored on Site. Location of storage shall be coordinated with the District and local fire authorities.

## 10.2.5 Protection of Work

The Contractor and Subcontractors shall continuously protect the Work, the District's property, and the property of others, from damage, injury, or loss arising in connection with operations under the Contract Documents. The Contractor and Subcontractors, at their own expense, shall make good any such damage, injury, or loss, except such as may be solely due to, or caused by, agents or employees of the District.

The Contractor, at Contractor's expense, will remove all mud, water, or other elements as may be required for the proper protection and prosecution of its Work.

Contractor shall take adequate precautions to protect existing roads, sidewalks, curbs, pavements, utilities, adjoining property and structures (including, without limitation, protection from settlement or loss of lateral support), and to avoid damage thereto, and repair any damage thereto caused by construction operations. All permits, licenses, or inspection fees required for such repair Work shall be obtained and paid for by Contractor.

#### 10.2.6 Requirements for Existing Sites

Contractor shall (unless waived by the District in writing):

- a. When performing construction on existing sites, become informed and take into specific account the maturity of the students on the Site; and perform Work which may interfere with school routine before or after school hours, enclose working area with a substantial barricade, and arrange Work to cause a minimum amount of inconvenience and danger to students and faculty in their regular school activities. The Contractor shall comply with Specifications and directives of the District regarding the timing of certain construction activities in order to avoid unnecessary interference with school functioning.
- b. Avoid performing any Work that will disturb students during testing.
- c. Provide substantial barricades around any shrubs or trees indicated to be preserved.
- d. Deliver materials to building area over route designated by Architect.
- e. Take preventive measures to eliminate objectionable dust, noise, or other disturbances.
- f. Confine apparatus, the storage of materials, and the operations of workers to limits indicated by law, ordinances, permits or directions of Architect; and not interfere with the Work or unreasonably encumber premises or overload any structure with materials; and enforce all instructions of District and Architect regarding signs, advertising, fires, and smoking and require that all workers comply with all regulations while on the Project site.

- g. Take care to prevent disturbing or covering any survey markers, monuments, or other devices marking property boundaries or corners. If such markers are disturbed by accident, they shall be replaced by an approved land surveyor or civil engineer and all maps and records required therefrom shall be filed with county and local authorities, at no cost to the District. All filing and plan check fees shall be paid by Contractor.
- h. Provide District on request with Contractor's written safety program and safety plan for each site.

## 10.2.7 Shoring and Structural Loading

The Contractor shall not impose structural loading upon any part of the Work under construction or upon existing construction on or adjacent to the Site in excess of safe limits, or loading such as to result in damage to the structural, architectural, mechanical, electrical, or other components of the Work. The design of all temporary construction equipment and appliances used in construction of the Work and not a permanent part thereof, including, without limitation, hoisting equipment, cribbing, shoring, and temporary bracing of structural steel, is the sole responsibility of the Contractor. All such items shall conform with the requirements of governing codes and all laws, ordinances, rules, regulations, and orders of all authorities having jurisdiction. The Contractor shall take special precautions, such as shoring of masonry walls and temporary tie bracing of structural steel Work, to prevent possible wind damage during construction of the Work. The installation of such bracing or shoring shall not damage the Work in place or the Work installed by others. Any damage which does occur shall be promptly repaired by the Contractor at no cost to the District.

#### 10.2.8 Conformance within Established Limits

The Contractor and Subcontractors shall confine their construction equipment, the storage of materials, and the operations of workers to the limits indicated by laws, ordinances, permits, and the limits established by the District or the Contractor, and shall not unreasonably encumber the premises with construction equipment or materials.

#### 10.2.9 Subcontractor Enforcement of Rules

Subcontractors shall enforce the District's and the Contractor's instructions, laws, and regulations regarding signs, advertisements, fires, smoking, the presence of liquor, and the presence of firearms by any person at the Site.

#### 10.2.10 Site Access

The Contractor and the Subcontractors shall use only those ingress and egress routes designated by the District, observe the boundaries of the Site designated by the District, park only in those areas designated by the District, which areas may be on or off the Site, and comply with any parking control program established by the District, such as furnishing license plate information and placing identifying stickers on vehicles.

#### 10.2.11 Security Services.

The Contractor shall be responsible for providing security services for the Site as needed for the protection of the Site and as determined in the District's sole discretion.

## 10.3 EMERGENCIES

## 10.3.1 <u>Emergency Action</u>

In an emergency affecting the safety of persons or property, the Contractor shall take any action necessary, at the Contractor's discretion, to prevent threatened damage, injury, or loss. Additional compensation or extension of time claimed by the Contractor on account of an emergency shall be determined as provided in Article 7.

#### 10.3.2 Accident Reports

The Contractor shall promptly report in writing to the District all accidents arising out of or in connection with the Work, which caused death, personal injury, or property damage, giving full details and statements of any witnesses in conformance with Article 10.1.4. In addition, if death, serious personal injuries, or serious property damages are caused, the accident shall be reported in accordance with Article 10.1.4, immediately by telephone or messenger to the District.

## 10.4 <u>HAZARDOUS MATERIALS</u>

#### 10.4.1 Discovery of Hazardous Materials

In the event the Contractor encounters or suspects the presence on the job site of material reasonably believed to be asbestos, polychlorinated biphenyl (PCB), or any other material defined as being hazardous by § 25249.5 of the California Health and Safety Code, which has not been rendered harmless, the Contractor shall immediately stop Work in the area affected and report the condition to the District and the Architect in writing, whether or not such material was generated by the Contractor or the District. The Work in the affected area shall not thereafter be resumed, except by written agreement of the District and the Contractor, if in fact the material is asbestos, polychlorinated biphenyl (PCB), or other hazardous material, and has not been rendered harmless. The Work in the affected area shall be resumed only in the absence of asbestos, polychlorinated biphenyl (PCB), or other hazardous material, or when it has been rendered harmless by written agreement of the District and the Contractor.

#### 10.4.2 Hazardous Material Work Limitations

In the event that the presence of hazardous materials is suspected or discovered on the Site (except in cases where asbestos and other hazardous material Work in the Contractor's responsibility), the District shall retain an independent testing laboratory to determine the nature of the material encountered and whether corrective measures or remedial action is required. The Contractor shall not be required pursuant to Article 7 to perform without consent any Work in the affected area of the Site relating to asbestos, polychlorinated biphenyl (PCB), or other hazardous material, until any known or suspected hazardous material has been removed, or rendered harmless, or determined to be harmless by District, as certified by an independent testing laboratory and approved by the appropriate government agency.

## 10.4.3 <u>Indemnification by Contractor for Hazardous Material Caused by Contractor</u>

In the event the hazardous materials on the Project Site is caused by the Contractor, the Contractor shall pay for all costs of testing and remediation, if any, and shall compensate the District for any additional costs incurred as a result of Contractor's generation of hazardous material on the Project Site. In addition, the Contractor shall defend, indemnify and hold harmless District and its agents, officers,

and employees from and against any and all claims, damages, losses, costs and expenses incurred in connection with, arising out of, or relating to, the presence of hazardous material on the Project Site.

## 10.4.4 Terms of Hazardous Material Provision

The terms of this Hazardous Material provision shall survive the completion of the Work and/or any termination of this Contract.

## ARTICLE 11 INSURANCE AND BONDS

#### 11.1 CONTRACTOR'S LIABILITY INSURANCE

#### 11.1.1 Insurance Requirements

Before the commencement of the Work, the Contractor shall purchase from and maintain in a company or companies lawfully authorized to do business in California with a financial rating of at least an A-VIII status as rated in the most recent edition of Best's Insurance Reports or as amended by the Supplementary General Conditions, such insurance as will protect the District from claims set forth below, which may arise out of or result from the Contractor's Work under the Contract and for which the Contractor may be legally liable, whether such Work are by the Contractor, by a Subcontractor, by anyone directly or indirectly employed by any of them, or by anyone for whose acts any of them may be liable. Any required insurance shall not contain any exclusion that applies to the type of work performed by the Contractor under the Contract Documents.

- a. Claims for damages because of bodily injury, sickness, disease, or death of any person District would require indemnification and coverage for employee claim;
- b. Claims for damages insured by usual personal injury liability coverage, which are sustained by a person as a result of an offense directly or indirectly related to employment of such person by the Contractor or by another person;
- c. Claims for damages because of injury or destruction of tangible property, including loss of use resulting therefrom, arising from operations under the Contract Documents:
- d. Claims for damages because of bodily injury, death of a person, or property damage arising out of the ownership, maintenance, or use of a motor vehicle, all mobile equipment, and vehicles moving under their own power and engaged in the Work:
- e. Claims involving contractual liability applicable to the Contractor's obligations under the Contract Documents, including liability assumed by and the indemnity and defense obligations of the Contractor and the Subcontractors; and
- f. Claims involving Completed Operations, Independent Contractors' coverage, and Broad Form property damage, without any exclusions for collapse, explosion, demolition, underground coverage, and excavating. (XCU)
- g. Claims involving sudden or accidental discharge of contaminants or pollutants.

## 11.1.2 Specific Insurance Requirements

Contractor shall take out and maintain and shall require all Subcontractors, if any, whether primary or secondary, to take out and maintain:

Comprehensive General Liability Insurance with a combined single limit per occurrence of not less than \$2,000,000.00 or Commercial General Liability Insurance which provides limits of not less than:

(a)	Per limit)	occurrenc	ee (combine	ed single	\$2,000,000.00
(b)	Project only)	Specific	Aggregate (for	this Project	\$2,000,000.00
(c)	Products (aggregate	and )	Completed	Operations	\$2,000,000.00
(d)	Personal Limit	and	Advertisin	ng Injury	\$1,000,000.00
				•••••	

## **Insurance Covering Special Hazards**

The following Special hazards shall be covered by riders or riders to above mentioned public liability insurance or property damage insurance policy or policies of insurance, in amounts as follows:

(a)	Automotive amounts	and	truck	where	operated	in	\$1,000,000.00
(b)	Material amounts	Hoist	W	here	used	in	\$1,000,000.00
(c)	Explosion, Co (XCU coverage)	ollapse an	d Underg	ground			\$1,000,000.00
(d)	Hazardous Materials						\$1,000,000.00

In addition, provide Excess Liability Insurance coverage in the amount of Four Million Dollars (\$4,000,000.00).

## 11.1.3 <u>Subcontractor Insurance Requirements</u>

The Contractor shall require its Subcontractors to take out and maintain public liability insurance and property damage insurance required under Article 11.1 in like amounts. A "claims made" or

modified "occurrence" policy shall not satisfy the requirements of Article 11.1 without prior written approval of the District.

## 11.1.4 Additional Insured Endorsement Requirements

The Contractor shall name, on any policy of insurance required under Article 11.1, the District, CM, Architect, Inspector, the State of California, their officers, employees, agents, volunteers and independent contractors as additional insureds. Subcontractors shall name the Contractor, the District, Architect, Inspector, the State of California, their officers, employees, agents, volunteers and independent contractors as additional insureds. The Additional Insured Endorsement included on all such insurance policies shall be an ISO CG 20 10 (04/13), or an ISO CG 20 38 (04/13), or their equivalent as determined by the District in its sole discretion, and must state that coverage is afforded the additional insured with respect to claims arising out of operations performed by or on behalf of the insured. If the additional insureds have other insurance which is applicable to the loss, such other insurance shall be on an excess or contingent basis. The insurance provided by the Contractor pursuant to 11.1 must be designated in the policy as primary to any insurance obtained by the District. The amount of the insurer's liability shall not be reduced by the existence of such other insurance.

## 11.2 WORKERS' COMPENSATION INSURANCE

During the term of this Contract, the Contractor shall provide workers' compensation and employer's liability insurance for all of the Contractor's employees engaged in Work under this Contract on or at the Site of the Project and, in case any of the Contractor's Work is subcontracted, the Contractor shall require the Subcontractor to provide workers' compensation insurance for all the Subcontractor's employees engaged in Work under the subcontract. Any class of employee or employees not covered by a Subcontractor's insurance shall be covered by the Contractor's insurance. In case any class of employees engaged in Work under this Contract on or at the Site of the Project is not protected under the Workers' Compensation laws, the Contractor shall provide or cause a Subcontractor to provide insurance coverage for the protection of those employees not otherwise protected. The Contractor shall file with the District certificates of insurance as required under Article 11.7 and in compliance with Labor Code § 3700.

Workers' compensation limits as required by the Labor Code, but not less than \$1,000,000 and employers' liability limits of \$1,000,000 per accident for bodily injury or disease.

#### 11.3 SEXUAL ABUSE/MOLESTATION

The CONTRACTOR shall carry and maintain during the term of this agreement a policy with a limit of not less than \$1,000,000 sexual abuse injury limit of insurance. The contractor shall hold the district harmless from any and all claims for injury, damage, and loss.

## 11.4 <u>BUILDER'S RISK/ "ALL RISK" INSURANCE</u>

## 11.4.1 <u>Course-of-Construction Insurance Requirements</u>

The Contractor, during the progress of the Work and until final acceptance of the Work by District upon completion of the entire Contract, shall maintain Builder's Risk, Course of Construction or similar first party property coverage issued on a replacement cost value basis consistent with the total replacement cost of all insurable Work and the Project included within the Contract Documents. Coverage is to insure against all risks of accidental direct physical loss, and must include, by the basic grant of

coverage or by endorsement, the perils of vandalism, malicious mischief (both without any limitation regarding vacancy or occupancy), fire, sprinkler leakage, civil authority, sonic boom, earthquake, flood, collapse, wind, lightning, smoke and riot. The coverage must include debris removal, demolition, increased costs due to enforcement of building ordinance and law in the repair and replacement of damage and undamaged portions of the property, and reasonable costs for the Architect's and engineering services and expenses required as a result of any insured loss upon the Work and Project which is the subject of the Contract Documents, including completed Work and Work in progress, to the full insurable value thereof. Such insurance shall include the District and the Architect as additional named insureds, and any other person with an insurable interest as designated by the District.

The Contractor shall submit to the District for its approval all items deemed to be uninsurable. The risk of the damage to the Work due to the perils covered by the "Builder's Risk/All Risk" Insurance, as well as any other hazard which might result in damage to the Work, is that of the Contractor and the Surety, and no Claims for such loss or damage shall be recognized by the District nor will such loss or damage excuse the complete and satisfactory performance of the Contract by the Contractor.

## 11.5 FIRE INSURANCE

Before the commencement of the Work, the Contractor shall procure, maintain, and cause to be maintained at the Contractor's expense, fire insurance on all Work subject to loss or damage by fire. The amount of fire insurance shall be sufficient to protect the Project against loss or damage in full until the Work is accepted by the District. This requirement may be waived upon confirmation by the District that such coverage is provided under the Builder's Risk Insurance being provided.

## 11.6 <u>AUTOMOBILE LIABILITY</u>

11.6.1 The District, Architect and Construction Manager, Inspectors, their directors, officers, employees, agents and volunteers shall be covered as additional insureds with respect to the ownership, operation, maintenance, use, loading or unloading of any auto owned, leased, hired or borrowed by the Contractor or for which the Contractor is responsible. Such insurance coverage shall be primary and non-contributory insurance as respects the District, Architect, Construction Manager, Project Inspector, their directors, officers, employees, agents and volunteers, or if excess, shall stand in an unbroken chain of coverage excess of the Contractor's scheduled underlying coverage. Any insurance or self-insurance maintained by the District, Architect, Construction Manager, Project Inspector, their directors, officers, employees, agents and volunteers shall be excess of the Contractor's insurance and shall not be called upon to contribute with it. The insurer shall agree to waive all rights of subrogation against the District, Architect, Construction Manager, Project Inspector, their directors, officers, employees, agents and volunteers for losses paid under the terms of the insurance policy that arise from Work performed by the Contractor.

11.6.2 Insurance Services Office Business Auto Coverage Form Number CA 0001, Code 1 (any auto) is required. Comprehensive Automobile Liability insurance to include all autos, owned, non-owned, and hired, with limits of \$1,000,000 per accident for bodily injury and property damage.

## 11.7 OTHER INSURANCE

The Contractor shall provide all other insurance required to be maintained under applicable laws, ordinances, rules, and regulations.

#### 11.8 PROOF OF INSURANCE

The Contractor shall not commence Work nor shall it allow any Subcontractor to commence Work under this Contract until all required insurance and certificates have been obtained and delivered in duplicate to the District for approval subject to the following requirements:

a. Certificates and insurance policies shall include the following clause:

"This policy and any coverage shall not be suspended, voided, non-renewed, canceled, or reduced in required limits of liability or amounts of insurance or coverage until notice has been mailed via certified mail to the District. Date of cancellation or reduction may not be less than thirty (30) days after the date of mailing notice."

- b. Certificates of insurance shall state in particular those insured, the extent of insurance, location and operation to which the insurance applies, the expiration date, and cancellation and reduction notices.
- c. Certificates of insurance shall clearly state that the District and the Architect are named as additional insureds under the policy described and that such insurance policy shall be primary to any insurance or self-insurance maintained by District.
- d. The Contractor and its Subcontractors shall produce a certified copy of any insurance policy required under this Section upon written request of the District.

#### 11.9 COMPLIANCE

In the event of the failure of Contractor to furnish and maintain any insurance required by this Article 11, the Contractor shall be in default under the Contract. Compliance by Contractor with the requirement to carry insurance and furnish certificates or policies evidencing the same shall not relieve the Contractor from liability assumed under any provision of the Contract Documents, including, without limitation, the obligation to defend and indemnify the District and the Architect.

## 11.10 WAIVER OF SUBROGATION

Contractor waives (to the extent permitted by law) any right to recover against the District for damages to the Work, any part thereof, or any and all claims arising by reason of any of the foregoing, but only to the extent that such damages and/or claims are covered by property insurance and only to the extent of such coverage (which shall exclude deductible amounts) by insurance actually carried by the District.

The provisions of this Article are intended to restrict each party to recovery against insurance carriers only to the extent of such coverage and waive fully and for the benefit of each, any rights and/or claims which might give rise to a right of subrogation in any insurance carrier. The District and the Contractor shall each obtain in all policies of insurance carried by either of them, a waiver by the insurance companies thereunder of all rights of recovery by way of subrogation for any damages or claims covered by the insurance.

## 11.11 PERFORMANCE AND PAYMENT BONDS

#### 11.11.1 Bond Requirements

Unless otherwise specified in the Supplemental Conditions, prior to commencing any portion of the Work, the Contractor shall furnish separate Payment and Performance Bonds for its portion of the Work which shall cover 100% faithful performance of and payment of all obligations arising under the Contract Documents and/or guaranteeing the payment in full of all claims for labor performed and materials supplied for the Work. All bonds shall be provided by a corporate Surety authorized and admitted to transact business in California as sureties.

To the extent, if any, that the Contract Price is increased in accordance with the Contract Documents, the Contractor shall, upon request of the District, cause the amount of the bonds to be increased accordingly and shall promptly deliver satisfactory evidence of such increase to the District. To the extent available, the bonds shall further provide that no change or alteration of the Contract Documents (including, without limitation, an increase in the Contract Price, as referred to above), extensions of time, or modifications of the time, terms, or conditions of payment to the Contractor will release the Surety. If the Contractor fails to furnish the required bonds, the District may terminate the Contract for cause.

#### 11.11.2 Surety Qualification

Only bonds executed by admitted Surety insurers as defined in Code of Civil Procedure § 995.120 shall be accepted. Surety must be a California-admitted Surety and listed by the U.S. Treasury with a bonding capacity in excess of the Project cost.

## 11.11.3 <u>Alternate Surety Qualifications</u>

If a California-admitted Surety insurer issuing bonds does not meet these requirements, the insurer will be considered qualified if it is in conformance with § 995.660 of the California Code of Civil Procedure and proof of such is provided to the District.

## ARTICLE 12 UNCOVERING AND CORRECTION OF WORK

#### 12.1 COMPLIANCE WITH TITLE 24 INSTALLATION REQUIREMENTS

Contractor is aware of the requirements governing Contractor's Work under title 24 Section 4-343 which provides, in pertinent part:

#### 4-343. Duties of the Contractor.

- (a) **Responsibilities**. It is the duty of the contractor to complete the Work covered by his or her contract in accordance with the approved Plans and Specifications therefore. The contractor in no way is relieved of any responsibility by the activities of the architect, engineer, Inspector or DSA in the performance of such duties.
- (b) **Performance of the Work.** The contractor shall carefully study the approved Plans and Specifications and shall plan a schedule of operations well ahead of time. If at any time it is discovered that Work is being done which is not in accordance with the approved Plans and Specifications, the contractor shall correct the Work immediately. All inconsistencies or items which appear to be in error in the Plans and Specifications shall be promptly called to the attention of the architect or registered engineer, through the Inspector, for interpretation or correction. In no case, however, shall the instruction of the architect or registered engineer be construed to cause Work to be done which is not in conformity with the approved Plans, Specifications, and Change Orders. The contractor must notify the Project Inspector, in advance, of the commencement of construction of each and every aspect of the Work.

#### 12.1.1 Issuance of Notices of Non-Compliance

The Inspector may issue a Notice of Non-Compliance on the Project indicating deviation from Plans and Specifications. It is Contractor's responsibility to correct all deviations from the approved Plans and Specifications unless the District has issued an Immediate Change Directive. In such case, the Contractor shall proceed with the Work with the understandings of the District as set forth in the ICD and as specifically noted in Article 7.3.

## 12.2 SPECIAL NOTICE OF AMERICAN'S WITH DISABILITIES ACT

Some of the requirements in the Plans and Specifications are meant to comply with the Americans with Disabilities Act ("ADA"). The requirements of the ADA are technical in nature and may appear to be minor in nature (i.e. whether a walkway or ramp has a 2% cross-slope). Contractor is warned that even the slightest deviation from the specific requirements from the ADA is considered a Civil Rights violation and subjects the District to fines of three times actual damages sustained by a handicap individual or up to \$4,000 per violation and attorney's fees required to enforce the ADA violation. As a result of the significant liability and exposure associated with ADA aspects of the Contract, Contractor shall take special care to meet all ADA requirements detailed in the Plans and Specifications. Failure to comply with ADA rules that results in a Notice of Non-Compliance shall be repaired to meet ADA requirements promptly. In addition, any ADA violations that are not identified by Inspector or Architect that are later identified shall be repaired and charged back to the Contractor through a Deductive Change Order.

#### 12.2.1 Indemnification of ADA Claims

Contractor shall indemnify, hold harmless and defend the District from ADA claims arising from the failure to comply with the Plans and Specifications. Further, any withholdings for ADA violations under Article 9.6 shall include potential redesign costs and an accelerated repair costs due to the potential for ADA claims arising from DSA posting of ADA violations on the Project.

## 12.3 UNCOVERING OF WORK

#### 12.3.1 Uncovering Work for Required Inspections

Work shall not be covered without the Inspector's review and the Architect's knowledge that the Work conforms with the requirements of the approved Plans and Specifications (except in the case of an ICD under Article 7.3). Inspector must be timely notified of inspections and of new areas so Work can be inspected at least 48 hours before opening a new area (For example, see DSA Form 156 for Commencement/Completion of Work Notification which requires "at least 48 hours" advance notification of a new area). An Inspector must comply with DSA protocols for signing each category or phase of Work under DSA Form 152 (in compliance with the Form 152 Manual) or a Notice of Deviation (DSA Form 154) will be issued requiring the Work that was not inspected be uncovered for inspection. Thus, if a portion of the Work is covered without inspection or Architect approval, is subject to a Notice of Non-Compliance for being undertaken without inspection, or otherwise not in compliance with the Contract Documents, after issuance of a Written Notice of Non-Compliance (Form 154) or a written notice to uncover Work, Contractor shall promptly uncover all Work (which includes furnishing all necessary facilities, labor, and material) for the Inspector's or the Architect's observation and such Work shall be replaced at the Contractor's expense without change in the Contract Sum or Time.

## 12.3.2 <u>Costs for Inspections Not Required</u>

If a portion of the Work has been covered is believed to be Non-Conforming to the Plans and Specifications, even if the Form 152 for the category of Work has been signed by the Inspector, the Inspector or the Architect may request to see such Work, and it shall be promptly uncovered by the Contractor. If such Work is in accordance with the Contract Documents, costs of uncover and replacement shall, by appropriate Change Order and shall, be charged to the District. If such Work is not in accordance with Contract Documents, the Contractor shall be responsible for all costs to uncover the Work, delays incurred to uncover the Work, and Contractor shall pay all costs to correct the Non-Conforming construction condition unless the condition was caused by the District or a separate contractor, in which event the District shall be responsible for payment of such costs to the Contractor.

#### 12.4 CORRECTION OF WORK

#### 12.4.1 Correction of Rejected Work

The Contractor shall promptly correct the Work rejected by the Inspector or the District upon recommendation of the Architect as failing to conform to the requirements of the Contract Documents, whether observed before or after Completion and whether or not Fabricated, installed, or completed. The Contractor shall bear costs of correcting the rejected Work, including cost for delays that may be incurred by Contractor or Subcontractors, the cost for additional testing, inspections, and compensation for the Inspector's or the Architect's services and expenses made necessary thereby (including costs for preparing a CCD, DSA CCD review fees, and additional inspection and special inspection costs).

## 12.4.2 <u>One-Year Warranty Corrections</u>

If, within one (1) year after the date of Completion of the Work or a designated portion thereof, or after the date for commencement of warranties established under Article 9.9.1, or by the terms of an applicable special warranty required by the Contract Documents, any of the Work is found to be not in accordance with the requirements of the Contract Documents, the Contractor shall correct it promptly after receipt of written notice from the District to do so unless the District has previously given the Contractor a written acceptance of such condition. This period of one (1) year shall be extended with respect to portions of the Work first performed after Completion by the period of time between Completion and the actual performance of the Work. This obligation under this Article 12.4.2 shall survive acceptance of the Work under the Contract and termination of the Contract. The District shall give such notice promptly after discovery of the condition.

## 12.4.3 District's Rights if Contractor Fails to Correct

If the Contractor fails to correct nonconforming Work within a reasonable time, the District may correct the Work and seek a Deductive Change Order, pursuant to Article 9.6 or Article 2.2.

# ARTICLE 13 MISCELLANEOUS PROVISIONS

#### 13.1 GOVERNING LAW

The Contract shall be governed by the law of the place where the Project is located.

#### 13.2 SUCCESSORS AND ASSIGNS

The District and the Contractor respectively bind themselves, their partners, successors, assigns, and legal representatives to the other party hereto and to partners, successors, assigns, and legal representatives of such other party in respect to covenants, agreements, and obligations contained in the Contract Documents. Neither party to the Contract shall assign the Contract as a whole without written consent of the other. If either party attempts to make such an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the Contract.

#### 13.3 WRITTEN NOTICE

In the absence of specific notice requirements in the Contract Documents, written notice shall be deemed to have been duly served if delivered in person to the individual, member of the firm or entity, or to an officer of the corporation for which it was intended, or if delivered at or sent by registered or certified mail to the last business address known to the party giving notice.

## 13.4 RIGHTS AND REMEDIES

#### 13.4.1 <u>Duties and Obligations Cumulative</u>

Duties and obligations imposed by the Contract Documents and rights and remedies available thereunder shall be in addition to and not a limitation of duties, obligations, rights, and remedies otherwise imposed or available by law.

#### 13.4.2 No Waiver

No action or failure to act by the Inspector, the District, or the Architect shall constitute a waiver of a right or duty afforded them under the Contract Documents, nor shall such action or failure to act constitute approval of or acquiescence in a breach thereunder, except as may be specifically agreed in writing.

#### 13.5 TESTS AND INSPECTIONS

## 13.5.1 <u>Compliance</u>

Tests, inspections, and approvals of portions of the Work required by the Contract Documents will comply with Division 1, Title 24, and with all other laws, ordinances, rules, regulations, or orders of public authorities having jurisdiction.

## 13.5.2 <u>Independent Testing Laboratory</u>

The District will select and pay an independent testing laboratory to conduct all tests and inspections. Selection of the materials required to be tested shall be made by the laboratory or the District's representative and not by the Contractor. See Articles 3.13.1 and 4.3.6 regarding costs or expenses of inspection or testing outside of the Project Site.

#### 13.5.3 Advance Notice to Inspector

The Contractor shall notify the Inspector a sufficient time in advance of its readiness for required observation or inspection so that the Inspector may arrange for same. The Contractor shall notify the Inspector a sufficient time in advance of the manufacture of material to be supplied under the Contract Documents which must, by terms of the Contract Documents, be tested in order that the Inspector may arrange for the testing of the material at the source of supply.

## 13.5.4 <u>Testing Off-Site</u>

Any material shipped by the Contractor from the source of supply, prior to having satisfactorily passed such testing and inspection or prior to the receipt of notice from said Inspector that such testing and inspection will not be required, shall not be incorporated in the Work.

## 13.5.5 <u>Additional Testing or Inspection</u>

If the Inspector, the Architect, the District, or public authority having jurisdiction determines that portions of the Work require additional testing, inspection, or approval not included under Article 13.5.1, the Inspector will, upon written authorization from the District, make arrangements for such additional testing, inspection, or approval. The District shall bear such costs except as provided in Articles 13.5.6 and 13.5.7.

## 13.5.6 Costs for Retesting

If such procedures for testing, inspection, or approval under Articles 13.5.1 and 13.5.2 reveal failure of the portions of the Work to comply with requirements established by the Contract Documents, the Contractor shall bear all costs arising from such failure, including those of re-testing, reinspection, or re-approval, including, but not limited to, compensation for the Architect's services and expenses. Any such costs shall be paid by the District, invoiced to the Contractor, and deducted from the next Progress Payment.

## 13.5.7 Costs for Premature Test

In the event the Contractor requests any test or inspection for the Project and is not completely ready for the inspection, the Contractor shall be invoiced by the District for all costs and expenses resulting from that testing or inspection, including, but not limited to, the Inspector's and Architect's fees and expenses, and the amount of the invoice shall be deducted from the next Progress Payment.

#### 13.6 TRENCH EXCAVATION

#### 13.6.1 Trenches Greater Than Five Feet

Pursuant to Labor Code section 6705, if the Contract Price exceeds \$25,000 and involves the excavation of any trench or trenches five (5) feet or more in depth, the Contractor shall, in advance of excavation, submit to the District or a registered civil or structural engineer employed by the District or Architect, a detailed plan showing the design of shoring for protection from the hazard of caving ground during the excavation of such trench or trenches.

#### 13.6.2 Excavation Safety

If such plan varies from the Shoring System Standards established by the Construction Safety Orders, the plan shall be prepared by a registered civil or structural engineer, but in no case shall such plan be less effective than that required by the Construction Safety Orders. No excavation of such trench or trenches shall be commenced until said plan has been accepted by the District or by the person to whom authority to accept has been delegated by the District.

#### 13.6.3 No Tort Liability of District

Pursuant to Labor Code § 6705, nothing in this Article shall impose tort liability upon the District or any of its employees.

#### 13.6.4 No Excavation without Permits

The Contractor shall not commence any excavation Work until it has secured all necessary permits including the required CAL OSHA excavation/shoring permit. Any permits shall be prominently displayed on the Site prior to the commencement of any excavation.

## 13.7 WAGE RATES, TRAVEL, AND SUBSISTENCE

#### 13.7.1 Wage Rates

Pursuant to the provisions of Article 2 (commencing at § 1720), Chapter 1, Part 7, Division 2, of the Labor Code, the District has obtained the general prevailing rate of per diem wages and the general prevailing rate for holiday and overtime work in the locality in which this public works project is to be performed for each craft, classification, or type of worker needed for this Project from the Director of the Department of Industrial Relations ("Director"). These rates are on file at the administrative office of the District and are also available from the Director of the Department of Industrial Relations. Copies will be made available to any interested party on request. The Contractor shall post a copy of such wage rates at appropriate, conspicuous, weatherproof points at the Site.

Any worker employed to perform Work on the Project, but such Work is not covered by any classification listed in the published general prevailing wage rate determinations or per diem wages determined by the Director of the Department of Industrial Relations, shall be paid not less than the minimum rate of wages specified therein for the classification which most nearly corresponds to the employment of such person in such classification.

#### 13.7.2 Holiday and Overtime Pay

Holiday and overtime work, when permitted by law, shall be paid for at the rate set forth in the prevailing wage rate determinations issued by the Director of the Department of Industrial Relations or at least one and one-half (1½) times the specified basic rate of per diem wages, plus employer payments, unless otherwise specified in the Contract Documents or authorized by law.

## 13.7.3 Wage Rates Not Affected by Subcontracts

The Contractor shall pay and shall cause to be paid each worker engaged in the execution of the Work on the Project not less than the general prevailing rate of per diem wages determined by the Director, regardless of any contractual relationship which may be alleged to exist between the Contractor or any Subcontractor and such workers.

## 13.7.4 Per Diem Wages

The Contractor shall pay and shall cause to be paid to each worker needed to execute the Work on the Project per diem wages including, but not limited to, employer payments for health and welfare, pensions, vacation, travel time and subsistence pay as provided for in Labor Code §1773.1.

#### 13.7.5 Forfeiture and Payments

Pursuant to Labor Code §1775, the Contractor shall forfeit to the District, not more than Two Hundred Dollars (\$200.00) for each calendar day, or portion thereof, for each worker paid less than the prevailing wages rates as determined by the Director of the Department of Industrial Relations, for the work or craft in which the worker is employed for any Work done under the Agreement by the Contractor or by any Subcontractor under it. The amount of the penalty shall be determined by the Labor Commissioner and shall be based on consideration of: (1) whether the Contractor or Subcontractor's failure to pay the correct rate of per diem wages was a good faith mistake and, if so, the error was promptly and voluntarily correct upon being brought to the attention of the Contractor or Subcontractor; and (2) whether the Contractor or Subcontractor has a prior record of failing to meet its prevailing wage obligations.

## 13.7.6 <u>Monitoring and Enforcement by Labor Commissioner</u>

Monitoring and enforcement of the prevailing wage laws and related requirements will be performed by the Labor Commissioner/ Department of Labor Standards Enforcement (DLSE). The Contractor and all subcontractors shall be required to furnish, at least monthly, certified payroll records directly to the Labor Commissioner in accordance with Labor Code section 1771.4. All payroll records shall be furnished in a format required by the Labor Commissioner. The Contractor and all subcontractors must sign up for, and utilize, the Labor Commissioner's electronic certified payroll records submission system. The District will have direct and immediate access to all CPRs for the Project that are submitted through the Labor Commissioner's system. The District can use this information for any appropriate purpose, including monitoring compliance, identifying suspected violations, and responding to Public Records Act requests.

The Labor Commissioner/ DLSE may conduct various compliance monitoring and enforcement activities including, but not limited to, confirming the accuracy of payroll records, conducting worker interviews, conducting audits, requiring submission of itemized statements prepared in accordance with Labor Code section 226, and conducting random in-person inspections of the Project site ("On-Site Visits"). On-Site Visits may include inspections of records, inspections of the Work site and observation of work activities, interviews of workers and others involved with the Project, and any other activities deemed necessary by the Labor Commissioner/DLSE to ensure compliance with prevailing wage requirements. The Labor Commissioner/DLSE shall have free access to any construction site or other place of labor and may obtain any information or statistics pertaining to the lawful duties of the Labor Commissioner/DLSE.

Any lawful activities conducted or any requests made by the Labor Commissioner/DLSE shall not be the basis for any delays, claims, costs, damages or liability of any kind against the District by the Contractor. Contractor and all subcontractors shall cooperate and comply with any lawful requests by the Labor Commissioner/DLSE. The failure of the Labor Commissioner, DLSE, or any other entity related to the Department of Industrial Relations to comply with any requirement imposed by the California Code of Regulations, Title 8, Chapter 8 shall not of itself constitute a defense to the failure to pay prevailing wages or to comply with any other obligation imposed by Division 2, Part 7, Chapter 1 of the Labor Code.

Prior to commencing any Work on the Project, the Contractor shall post the required notice/poster required under the California Code of Regulations and Labor Code section 1771.4 in both English and Spanish at a conspicuous, weatherproof area at the Project site. The required notice/poster is available on the Labor Commissioner's website.

## 13.8 RECORDS OF WAGES PAID

#### 13.8.1 Payroll Records

a. Pursuant to §1776 of the Labor Code, the Contractor and each Subcontractor shall keep an accurate payroll record showing the name, address, social security number, work classification and straight time and overtime hours worked each day and week, and the actual per diem wages paid to each journeyman, apprentice, worker or other employee employed by him or her in connection with the Project.

All payroll records as specified in Labor Code §1776 of the Contractor and all Subcontractors shall be certified and furnished directly to the Labor Commissioner in accordance with Labor Code §1771.4(a)(3) on a monthly basis (or more frequently if required by the District or the Labor Commissioner) and in a format prescribed by the Labor Commissioner. Payroll records as specified in Labor Code §1776 shall be certified and submitted to the District with each application for payment. All payroll records shall be available for inspection at all reasonable hours at the principal office of the Contractor on the following basis:

- 1. A certified copy of an employee's payroll record shall be made available for inspection or furnished to the employee or his or her authorized representative on request.
- 2. A certified copy of all payroll records shall be made available for inspection or furnished upon request to a representative of District, the Division of Labor Standards Enforcement or the Division of Apprenticeship Standards of the Department of Industrial Relations.
- 3. A certified copy of all payroll records shall be made available upon request by the public for inspection or for copies thereof. However, a request by the public shall be made through the District, the Division of Apprenticeship Standards or the Division of Labor Standards Enforcement. If the requested payroll records have not been provided pursuant to Paragraph (2) above, the requesting party shall, prior to being provided the records, reimburse the costs, according to law for the preparation by the Contractor, Subcontractor(s), and the entity through

which the request was made. The public shall not be given access to such records at the principal office of the Contractor.

- b. The certified payroll records shall be on forms provided by the Division of Labor Standards Enforcement or shall contain the same information as the forms provided by the Division of Labor Standards Enforcement.
- c. The Contractor or Subcontractor(s) shall file a certified copy of all payroll records with the entity that requested such records within 10 calendar days after receipt of a written request.
- d. Any copy of records made available for inspection as copies and furnished upon request to the public or any public agency by the District, the Division of Apprenticeship Standards or the Division of Labor Standards Enforcement shall be marked or obliterated to prevent disclosure of an individual's name, address and social security number. The name and address of the Contractor awarded the Contract or the Subcontractor(s) performing the Contract shall not be marked or obliterated. Any copy of records made available for inspection by, or furnished to, a joint labor-management committee established pursuant to the federal Labor Management Cooperation Act of 1978 (Section 175a of Title 29 of the United States Code) shall be marked or obliterated only to prevent disclosure of an individual's name and social security number. Notwithstanding any other provision of law, agencies that are included in the Joint Enforcement Strike Force on the Underground Economy established pursuant to Section 329 of the Unemployment Insurance Code and other law enforcement agencies investigating violations of law shall, upon request, be provided non-redacted copies of certified payroll records.
- e. The Contractor shall inform the District of the location of all payroll records, including the street address, city and county, and shall, within five working days, provide a notice of a change of location and address.
- f. The Contractor or Subcontractor(s) shall have 10 calendar days in which to comply subsequent to receipt of a written notice requesting payroll records. In the event that the Contractor or Subcontractor(s) fails to comply within the 10-day period, the Contractor or Subcontractor(s) shall, as a penalty to the District, forfeit One Hundred Dollars (\$100.00) for each calendar day, or portion thereof, for each worker, until strict compliance is effectuated. Upon the request of the Division of Apprenticeship Standards or the Division of Labor Standards Enforcement, these penalties shall be withheld from progress payments then due.

Responsibility for compliance with this Article shall rest upon the Contractor.

## 13.8.2 Withholding of Contract Payments & Penalties

The District may withhold or delay contract payments to the Contractor and/or any Subcontractor if:

- a. The required prevailing rate of per diem wages determined by the Director of the Department of Industrial Relations is not paid to all workers employed on the Project; or
- b. The Contractor or Subcontractor(s) fail to submit all required certified payroll records with each application for payment, but not less than once per month; or
- c. The Contractor or Subcontractor(s) submit incomplete or inadequate payroll records; or
- d. The Contractor or Subcontractor(s) fail to comply with the Labor Code requirements concerning apprentices; or
- e. The Contractor or Subcontractor(s) fail to comply with any applicable state laws governing workers on public works projects.

## 13.9 APPRENTICES

## 13.9.1 Apprentice Wages and Definitions

All apprentices employed by the Contractor to perform services under the Contract shall be paid the standard wage paid to apprentices under the regulations of the craft or trade for which he or she is employed, and as determined by the Director of the Department of Industrial Relations, and shall be employed only at the craft or trade to which he or she is registered. Only apprentices, as defined in §3077 of the Labor Code, who are in training under apprenticeship standards that have been approved by the Chief of the Division of Apprenticeship Standards and who are parties to written apprenticeship agreements under Chapter 4 (commencing with §3070) of Division 3, are eligible to be employed under this Contract. The employment and training of each apprentice shall be in accordance with the apprenticeship standards and apprentice agreements under which he or she is training, or in accordance with the rules and regulations of the California Apprenticeship Council.

#### 13.9.2 Employment of Apprentices

Contractor agrees to comply with the requirements of Labor Code §1777.5. The Contractor awarded the Project, or any Subcontractor under him or her, when performing any of the Work under the Contract or subcontract, employs workers in any apprenticeable craft or trade, the Contractor and Subcontractor shall employ apprentices in the ratio set forth in Labor Code §1777.5. The Contractor or any Subcontractor must apply to any apprenticeship program in the craft or trade that can provide apprentices to the Project site for a certificate approving the contractor or subcontractor under the apprenticeship standards for the employment and training of apprentices in the area or industry affected. However, the decision of the apprenticeship program to approve or deny a certificate shall be subject to review by the Administrator of Apprenticeship. The apprenticeship program or programs, upon approving the Contractor or Subcontractor, shall arrange for the dispatch of apprentices to the Contractor or Subcontractor upon the Contractor's or Subcontractor's request. "Apprenticeable craft or trade" as used in this Article means a craft or trade determined as an apprenticeable occupation in accordance with the rules and regulations prescribed by the California Apprenticeship Council. The ratio of work performed by apprentices to journeyman employed in a particular craft or trade on the Project shall be in accordance with Labor Code §1777.5.

## 13.9.3 <u>Submission of Contract Information</u>

Prior to commencing Work on the Project, the Contractor and Subcontractors shall submit contract award information to the applicable apprenticeship program(s) that can supply apprentices to the Project and make the request for the dispatch of apprentices in accordance with the Labor Code. The information submitted shall include an estimate of journeyman hours to be performed under the Contact, the number of apprentices proposed to be employed, and the approximate dates the apprentices would be employed. A copy of this information shall also be submitted to the District if requested. Within 60 days after concluding Work on the Project, the Contractor and Subcontractors shall submit to the District, if requested, and to the apprenticeship program a verified statement of the journeyman and apprentice hours performed on the Project.

## 13.9.4 Apprentice Fund

The Contractor or any Subcontractor under him or her, who, in performing any of the Work under the Contract, employs journeymen or apprentices in any apprenticeable craft or trade shall contribute to the California Apprenticeship Council the same amount that the Director determines is the prevailing amount of apprenticeship training contributions in the area of the Project. The Contractor and Subcontractors may take as a credit for payments to the California Apprenticeship Council any amounts paid by the Contractor or Subcontractor to an approved apprenticeship program that can supply apprentices to the Project. The Contractor and Subcontractors may add the amount of the contributions in computing his or her bid for the Contract.

## 13.9.5 Prime Contractor Compliance

The responsibility of compliance with Article 13 and §1777.5 of the Labor Code for all apprenticeable occupations is with the Prime Contractor. Any Contractor or Subcontractor that knowingly violates the provisions of this Article or Labor Code §1777.5 shall be subject to the penalties set forth in Labor Code §1777.7.

## 13.10 ASSIGNMENT OF ANTITRUST CLAIMS

#### 13.10.1 Application

Pursuant to Government Code § 4551, in entering into a public works contract or a subcontract to supply goods, services, or materials pursuant to a public works contract, the Contractor or Subcontractor offers and agrees to assign to the District all rights, title, and interest in and to all causes of action it may have under Section 4 of the Clayton Act, (15 U.S.C. § 15) or under the Cartwright Act (Chapter 2 [commencing with § 16700] of Part 2 of Division 7 of the Business and Professions Code), arising from the purchase of goods, services, or materials pursuant to the public works contract or the subcontract. This assignment shall be made and become effective at the time the awarding body tenders Retention Payment to the Contractor, without further acknowledgment by the parties. If the District receives, either through judgment or settlement, a monetary recovery for a cause of action assigned under Chapter 11 (commencing with § 4550) of Division 5 of Title 1 of the Government Code, the assignor shall be entitled to receive reimbursement for actual legal costs incurred and may, upon demand, recover from the District any portion of the recovery, including treble damages, attributable to overcharges that were paid by the assignor but were not paid by the District as part of the bid price, less the expenses incurred in obtaining that portion of the recovery.

#### 13.10.2 Assignment of Claim

Upon demand in writing by the assignor, the District shall, within one (1) year from such demand, reassign the cause of action assigned pursuant to this Article if the assignor has been or may have been injured by the violation of law for which the cause of action arose and the District has not been injured thereby or the District declines to file a court action for the cause of action.

## 13.11 STATE AND DISTRICT CONDUCTED AUDITS

Pursuant to and in accordance with the provisions of Government Code § 10532, or any amendments thereto, all books, records, and files of the District, the Contractor, or any Subcontractor connected with the performance of this Contract involving the expenditure of state funds in excess of Ten Thousand Dollars (\$10,000.00), including, but not limited to, the administration thereof, shall be subject to the examination and audit of the Office of the Auditor General of the State of California for a period of five (5) years after Retention Payment is made or a Notice of Completion is Recorded, whichever occurs first. Contractor shall preserve and cause to be preserved such books, records, hard drives, electronic media, and files for the audit period.

Pursuant to the remedies under Public Contract Code section 9201 and Government Code section 930.2, Contractor, through execution of this Agreement, also agrees the District shall have the right to review and audit, upon reasonable notice, the books and records of the Contractor concerning any monies associated with the Project. The purpose of this "Audit" is to quickly and efficiently resolve Disputes based on the actual costs incurred and to reduce the uncertainty in resolving Disputes with limited information. The District shall perform any audits at its own cost and any such audit shall be performed by an independent auditor, having no direct or indirect relationship with the functions or activities being audited or with the business conducted by the Contractor or District. In the event the independent auditor determines that Change Orders, response to Request for Proposals, Claims, Appeal of Claims, or other requests for payment are in error, or have has any other concerns or questions, the Auditor shall report the results of the Audit findings to the District and provide a copy to the Contractor after giving the District Board the opportunity for at least 10 days review. If the Contractor disputes the findings of the independent auditor, such dispute shall be handled in the manner set forth under Article 4.6.2 entitled Disputes.

If Contractor having agreed to the terms of this Contract fails to produce books or records requested by Auditor, such failure to produce books or records that were required to be preserved for audit, it shall be presumed that the information contained in the withheld books or records were unfavorable to the Contractor and the Auditor shall note this refusal in the results of the Audit findings for further evaluation by the District and the District's Board. The refusal to release records that are concerning monies associated with the Project may be used as a grounds to debar the Contractor under Article 15 for failure to preserve records under Article 13.11 and the failure to produce required audit records may also be used as a grounds for a negative finding against the Contractor depending on the significance of the records that are withheld by Contractor. Failure to produce job cost data tied to job cost categories and budgets shall be presumed an intentional failure to produce key audit records. Similarly, failure to produce Daily Reports (prepared at or near the time of the Work actually took place (See Article 3.16) shall be presumed an intentional failure to produce key audited records.

If Contractor is seeking costs for inefficiency, home office overhead, or unanticipated increased costs due to delays or acceleration, Contractor shall also produce copies of the original bid tabulation utilized in submitting Contractor's bid for the Project. This document shall be considered confidential and shall not be subject to disclosure through a Public Records Act and shall not be distributed to anyone other than the District and the District's counsel. This bid tabulation shall only be used in litigation, arbitration, evaluation of Claims or Disputes, Audit, and trial. If the records for the bid tabulation are kept on a computer, the Contractor shall also produce all metadata (in native format) that accompanies

the bid tabulation for inspection to prove the authenticity of the underlying bid tabulation. Failure to produce the bid tabulation for review of inefficiency, home office overhead, or unanticipated increased costs due to delays or accelerations shall be considered material evidence that the bid tabulation was not favorable to the Contractor. This evidence shall be entered as a jury instruction for trial that the bid tabulation was not produced and the bid tabulation information was unfavorable to the Contractor. The evidence may also be used in debarment proceedings, and noted as an exception to an Audit findings.

Upon notification of Contractor concerning the results of the audit and a reasonable time has passed for Contractor to respond to the Audit findings and if either there is no Dispute of the Audit findings under Article 4.6 or if the result after utilizing the Disputes Clause confirms the Audit findings, the District may seek reimbursement for overstated Claims, Change Orders, or Appeal of Claims and may also undertake debarment proceedings under Article 15 of these General Conditions.

## 13.12 STORM WATER POLLUTION PREVENTION

#### 13.12.1 Application

This Section addresses the preparation, implementation and monitoring of a Storm Water Pollution Prevention Plan (SWPPP) for the purpose of preventing the discharge of pollutants from the construction site. This includes the elimination of pollution discharges such as improper dumping, spills or leakage from storage tanks or transfer areas. The District will not issue a Notice to Proceed until Contractor has prepared by a qualified individual and obtained approval of the Permit Registration Documents ("PRDs") that include a Notice of Intent, Construction Risk Calculation, Site Map, SWPPP, Annual Fee and any additional required documents from all applicable Local Governing Agencies including the Regional Water Quality Control Board. The Contractor shall also secure a certification that the Project has met all of the conditions of the General Construction Activity Storm Water Permit (GCASP) and comply with all applicable local, state and federal regulations governing storm water pollution prevention.

## 13.12.2 References and Materials

- California Stormwater Quality Association New Development and Redevelopment Best Management Practice Handbook
- 2009 California Stormwater Quality Association Construction BMP Handbook.
- State Water Resources Control Board (2009). Order 2009-0009-DWQ, NPDES General Permit No. CAS000002: Waste Discharge Requirements for Discharges of Storm Water Runoff Associated with Construction and Land Disturbing Activities. Available on-line at:
- http://www.waterboards.ca.gov/water\_issues/programs/stormwater/construction.shtml.- Use materials of a class, grade and type needed to meet the performance described in the BMP Handbook.

## 13.12.3 <u>Preparation and Approval</u>

The Contractor shall prepare by a qualified individual the PRDs that include a Notice of Intent, Construction Risk Calculation, Site Map, SWPPP, Annual Fee and any additional required documents. The Contractor's Qualified SWPPP Developer ("QSD") shall prepare the Storm Water Pollution Prevention Plan (SWPPP) as required to comply with storm water pollution regulations for project sites with storm water discharges associated with construction activity such as clearing or demolition,

grading, excavation and other land disturbances. The SWPPP shall apply to all areas that are directly related to construction activity, including but not limited to staging areas, storage yards, material borrow areas, and access roads.

- 13.12.3.1 The Contractor shall prepare and submit to the Local Governing Agencies and the District the SWPPP for review and approval if the project sites, new or existing, with land disturbance of 1 or more acres (or less than 1 acres if part of a common plan of development); the construction activity that results in land surface disturbances of less than one acre is part of a larger common plan of development or sale of one or more acres of disturbed land surface; or the construction activity associated with Linear Underground/Overhead Projects ("LUPs") including, but not limited to, those activities necessary for the installation of underground and overhead linear facilities (e.g., conduits, substructures, pipelines, towers, poles, cables, wires, connectors, switching, regulating and transforming equipment and associated ancillary facilities) and include, but are not limited to, underground utility mark-out, potholing, concrete and asphalt cutting and removal, trenching, excavation, boring and drilling, access road and pole/tower pad and cable/wire pull station, substation construction, substructure installation, construction of tower footings and/or foundations, pole and tower installations, pipeline installations, welding, concrete and/or pavement repair or replacement, and stockpile/borrow locations.
- 13.12.3.2 The Contractor shall also pay annual renewal fee(s) until the contract is completed and make all such checks payable to the State Water Resources Control Board. The Notice of Intent must be submitted at least two weeks prior to the commencement of construction activities.
- 13.12.3.3 The Contractor shall prepare the SWPPP by following the format in Sections 2, 3, 4 and Appendices A through F of the California Stormwater BMP Handbook Construction, January 2009 edition, published by the California Stormwater Quality Association. The publication is available from:

California Stormwater Quality Association P.O. Box 2105 Menlo Park, CA 94026-2105 Phone: (650) 366-1042 E-mail: info@casqa.org

or

https://www.casqa.org/store/products/tabid/154/p-167-construction-handbookportal-initial-subscription.aspx

- 13.12.3.4 Where land disturbance is less than 1 acre, any BMPs indicated in the BMP Handbook needed to prevent or minimize storm water pollution shall be implemented at no extra cost to the District.
- 13.12.3.5 Within two weeks after Award of Contract by the District, the Contractor shall submit to the District's Civil Engineer one copy of the PRDs including the SWPPP for review. After the District's approval, the Contractor shall provide approved copies of the SWPPP as follows: one copy each to the Project Inspector, Construction Manager, Architect, Commissioned Architect and District's Civil Engineer.

#### 13.12.4 Implementation

The Contractor shall implement the Storm Water Pollution Prevention Plan by doing the following:

- a. Obtain a Waste Discharger Identification (WDID) number from the SWRCB before beginning construction. This number will be issued once your PRDs are administratively accepted and fee is received.
- b. Keep the SWPPP, REAPs, monitoring data on the construction site.
- c. Employ a Qualified SWPPP Practitioner (QSP) to implement the SWPPP during construction and develop Rain Event Action Plans ("REAPs").
- d. Install, inspect, maintain and monitor BMPs required by the General Permit.
- e. Install perimeter controls prior to starting other construction work at the site.
- f. Contain on-site storm water at the jobsite. Do not drain on-site water directly into the storm drain.
- g. Implement the SWPPP.
- h. Provide SWPPP and BMP implementation training for those responsible for implementing the SWPPP.
- i. Designate trained personnel for the proper implementation of the SWPPP.
- Conduct monitoring, as required, and assess compliance with the Numeric Action Levels (NALs) or Numeric Effluent Limitations (NELs) appropriate to your project.
- k. Report monitoring data:
  - 1. Maintain a paper or electronic copy of all required records for three years from the date generated or date submitted, whichever is last. These records must be available at the construction site until construction is completed.
  - 2. Have a QSD revise the SWPPP as needed to reflect the phases of construction and to suit changing site conditions and instances when properly installed systems are ineffective.
  - 3. Assist the District with entering any necessary data or information into the Stormwater Multi-Application and Reporting System ("SMARTS") system.
- 1. At the end of Construction Contract:
  - 1. Submit Notice of Termination (NOT) into the SMARTS when construction is complete and conditions of termination listed in the NOT have been satisfied. A copy of the NOT can be found at:

http://www.waterboards.ca.gov/water\_issues/programs/stormwater/const ruction.shtml.

- 2. Leave in place storm water pollution prevention controls needed for post-construction storm water management and remove those that are not needed as determined by the District. Thereafter, left-in-place controls will be maintained by the District.
- 3. Provide Site Monitoring Reports, SWPPP revisions, Compliance Certifications and related documents to the District. Post-construction storm water operation and management plan as mentioned in the compliance certifications are considered to be in place at the end of the Construction Contract.

#### 13.12.5 Monitoring

The Contractor shall conduct examination of storm water pollution prevention controls as required by the State Water Resources Control Board (2009). Order 2009-0009-DWQ, NPDES General Permit No. CAS000002: Waste Discharge Requirements for Discharges of Storm Water Runoff Associated with Construction and Land Disturbing Activities. This includes properly qualified personnel performing all required monitoring, testing, inspections and monitoring. The Contractor shall also conduct examination of storm water pollution prevention controls, as well as before and after each storm event in compliance with the State Water Resources Control Board Order No. 2009-0009-DWQ, National Pollutant Discharge Elimination System General Permit No. CAS000002, Waste Discharge Requirements for Discharges of Storm Water Runoff Associated with Construction and Land Disturbance Activities (General Permit) (SWRCB, 2009).and at least once each 24-hour period during extended storm events to identify BMP effectiveness and implement repairs or BMP changes as soon as feasible. All maintenance related to a storm event should be completed within 48 hours of the storm event. The Contactor shall also prepare and maintain, at the jobsite, a log of each inspection using Site Monitoring Report forms.

#### 13.12.6 Liabilities and Penalties

- a. Review of the SWPPP and inspection logs by the District shall not relieve the Contractor from liabilities arising from non-compliance with storm water pollution regulations.
- b. Payment of penalties for non-compliance by the Contractor shall be the sole responsibility of the Contractor and will not be reimbursed by the District.
- c. Compliance with the Clean Water Act pertaining to construction activity is the sole responsibility of the Contractor. For any fine(s) levied against the District due to non-compliance by the Contractor, the District will deduct from the final payment due the Contractor the total amount of the fine(s) levied on the District, plus legal and associated costs.
- d. The Contractor shall submit to the District a completed NOI for change of information (Construction Site Information and Material Handling/Management Practices).

## ARTICLE 14 TERMINATION OR SUSPENSION OF THE CONTRACT

## 14.1 TERMINATION BY THE CONTRACTOR FOR CAUSE

#### 14.1.1 Grounds for Termination

The Contractor may terminate the Contract if the Work is stopped for a period of thirty (30) consecutive days through no act or fault of the Contractor, a Subcontractor, a Sub-subcontractor, their agents or employees, or any other persons performing portions of the Work for whom the Contractor is contractually responsible, for only the following reasons:

- a. Issuance of an order of a court or other public authority having jurisdiction; or
- b. An act of the United State or California government, such as a declaration of national emergency.

#### 14.1.2 Notice of Termination

If one of the above reasons exists, the Contractor may, upon written notice of seven (7) additional days to the District, terminate the Contract and recover from the District payment for Work executed and for reasonable costs verified by the Architect with respect to materials, equipment, tools, construction equipment, and machinery, including reasonable overhead, profit, and damages.

#### 14.2 TERMINATION BY THE DISTRICT FOR CAUSE

#### 14.2.1 Grounds for Termination

The District may terminate the Contractor and/or this Contract for the following reasons:

- a. Persistently or repeatedly refuses or fails to supply enough properly skilled workers or proper materials;
- b. Persistently or repeatedly is absent, without excuse, from the job site;
- c. Fails to make payment to Subcontractors, suppliers, materialmen, etc.;
- d. Persistently disregards laws, ordinances, rules, regulations, or orders of a public authority having jurisdiction;
- e. Fails to provide a schedule or fails or refuses to update schedules required under the Contract;
- f. Falls behind on the Project and refuses or fails to undertake a Recovery Schedule;
- g. If the Contractor has been debarred from performing Work
- h. Becomes bankrupt or insolvent, including the filing of a general assignment for the benefit of creditors; or

i. Otherwise is in substantial breach of a provision of the Contract Documents.

#### 14.2.2 Notification of Termination

When any of the above reasons exist, the District may, without prejudice to any other rights or remedies of the District and after giving the Contractor and the Contractor's Surety written notice of seven (7) days, terminate the Contractor and/or this Contract and may, subject to any prior rights of the Surety:

- a. Take possession of the Project and of all material, equipment, tools, and construction equipment and machinery thereon owned by the Contractor;
- b. Accept assignment of Subcontracts. Contractor acknowledges and agrees that if the District (in its sole and absolute discretion) decides to takeover completion of the Project, the Contractor agrees to immediately assign all subcontracts to the District which the District has chosen to accept;
- c. Complete the Work by any reasonable method the District may deem expedient, including contracting with a replacement contractor or contractors; and,
- d. Agree to accept a takeover and completion arrangement with Surety that is acceptable to the District Board.

## 14.2.3 <u>Takeover and Completion of Work after Termination for Cause</u>

A Termination for Cause is an urgent matter which requires immediate radiation since Project Work is open and incomplete, the site is subject to vandalism and theft, the Project site is considered a public nuisance, and there is a possibility of injury and deterioration of the Project Work and materials. Thus, the District shall be entitled to enter a takeover contract to either remediate the unfinished condition or complete the Work for this Project.

#### 14.2.4 Payments Withheld

If the District terminates the Contract for one of the reasons stated in Article 14.2.1, the Contractor shall not be entitled to receive further payment until the Work is complete. All costs associated with the termination and completion of the Project shall be the responsibility of the Contractor and/or its Surety.

#### 14.2.5 Payments upon Completion

If the unpaid balance of the Contract Sum exceeds costs of completing the Work, including compensation for professional services and expenses made necessary thereby, such excess shall be paid to the Contractor. If such costs exceed the unpaid balance, the Contractor and its Surety shall pay the difference to the District. The amount to be paid to the Contractor, or District, as the case may be, shall be certified by the Architect upon application. This payment obligation shall survive completion of the Contract.

#### 14.3 TERMINATION OF CONTRACT BY DISTRICT (CONTRACTOR NOT AT FAULT)

## 14.3.1 <u>Termination for Convenience</u>

District may terminate the Contract upon fifteen (15) calendar days of written notice to the Contractor and use any reasonable method the District deems expedient to complete the Project, including contracting with replacement contractor or contractors, if it is found that reasons beyond the control of either the District or Contractor make it impossible or against the District's interest to complete the Project. In such a case, the Contractor shall have no Claims against the District except for: (1) the actual cost for approved labor, materials, and services performed in accordance with the Contract Documents which have not otherwise been previously paid for and which are supported and documented through timesheets, invoices, receipts, or otherwise; and (2) profit and overhead of ten percent (10%) of the approved costs in item (1); and (3) termination cost of five percent (5%) of the approved costs in item (1). Contractor acknowledges and agrees that if the District (in its sole and absolute discretion) decides to takeover completion of the Project, the Contractor agrees to immediately assign all subcontracts to the District which the District has chosen to accept.

## 14.3.2 <u>Non-Appropriation of Funds/ Insufficient Funds</u>

In the event that sufficient funds are not appropriated to complete the Project or the District determines that sufficient funds are not available to complete the Project, District may terminate or suspend the completion of the Project at any time by giving written notice to the Contractor. In the event that the District exercises this option, the District shall pay for any and all work and materials completed or delivered onto the site for which value is received, and the value of any and all work then in progress and orders actually placed which cannot be canceled up to the date of notice of termination. The value of work and materials not otherwise already paid for by the District up to the time of termination under this Paragraph shall include a factor of fifteen percent (15%) for the Contractor's overhead and profit and there shall be no other costs or expenses paid to Contractor. All work, materials and orders paid for pursuant to this provision shall become the property of the District. District may, without cause, order Contractor in writing to suspend, delay or interrupt the Project in whole or in part for such period of time as District may determine. Adjustment shall be made for increases in the cost of performance of the Agreement caused by suspense, delay or interruption.

## 14.4 <u>REMEDIES OTHER THAN TERMINATION</u>

If a default occurs, the District may, without prejudice to any other right or remedy, including, without limitation, its right to terminate the Contract pursuant to Article 14.2, do any of the following:

- a. Permit the Contractor to continue under this Contract, but make good such deficiencies or complete the Contract by whatever method the District may deem expedient, and the cost and expense thereof shall be deducted from the Contract Price or paid by the Contractor to the District on demand;
- b. If the workmanship performed by the Contractor is faulty or defective materials are provided, erected or installed, then the District may order the Contractor to remove the faulty workmanship or defective materials and to replace the same with work or materials that conform to the Contract Documents, in which event the Contractor, at its sole costs and expense, shall proceed in accordance with the District's order and complete the same within the time period given by the District in its notice to the Contractor; or
- c. Initiate procedures to declare the Contractor a non-responsible bidder for a period of two (2) to five (5) years thereafter.

All amounts expended by the District in connection with the exercise of its rights hereunder shall accrue interest from the date expended until paid to the District at the maximum legal rate. The District may retain or withhold any such amounts from the Contract Price. If the Contractor is ordered to replace any faulty workmanship or defective materials pursuant to Paragraph (b) above, the Contractor shall replace the same with new work or materials approved by the Architect and the District, and, at its own cost, shall repair or replace, in a manner and to the extent the Architect and the District shall direct, all Work or material that is damaged, injured or destroyed by the removal of said faulty workmanship or defective material, or by the replacement of the same with acceptable work or materials. In no event shall anything in this Article be deemed to constitute a waiver by the District of any other rights or remedies that it may have at law or in equity, it being acknowledged and agreed by the Contractor that the remedies set forth in this Article are in addition to, and not in lieu of, any other rights or remedies that the District may have at law or in equity.

## ARTICLE 15 DEBARMENT

# 15.1 <u>DEBARMENT MEANS THERE HAS BEEN A FINDING THAT THE CONTRACTOR IS</u> NOT RESPONSIBLE.

During the course of the Project, or if it is determined through Change Orders, Claims, or Audit that a Contractor is not responsible, the District may, in addition to other remedies provided in the Contract, debar the Contractor from bidding or proposing on, or being awarded, and/or performing work on District contracts for a specified period of time, which generally will not exceed five (5) years, but may exceed five (5) years or be permanent if the circumstances warrant such debarment. In addition to the debarment proceeding, a finding that a Contractor is to be debarred shall result in the termination of any or all existing Contracts the Contractor may have with the District.

#### 15.2 BOARD FINDING

The District may debar a Contractor if the Board, or the Board's delegatee, in its discretion, finds the Contractor has done any of the following:

- 15.2.1 <u>Intentionally or with reckless disregard, violated any term of the Contract with the</u>
  District
- 15.2.2 <u>Committed an acts or omission which reflects on the Contractor's quality, fitness or capacity to perform Work for the District;</u>
- 15.2.3 <u>Committed an act or offense which indicates a lack of business integrity or business</u> honesty; or,
  - 15.2.4 Made or submitted a false claim against the District or any other public entity.

#### 15.3 HEARING AND PRESENTATION OF EVIDENCE

If there is evidence that the Contractor may be subject to debarment, the District shall notify the Contractor in writing of the evidence which is the basis for the proposed debarment and shall advice the Contractor of the scheduled date for a debarment hearing before the District Board or its delegated designee.

The District Board, or designee, shall conduct a hearing where evidence on the proposed debarment is presented. The Contractor or the Contractor's representative shall be given an opportunity to submit evidence at the hearing. The Contractor shall be provided an adequate amount of time to prepare and object to evidence presented. A tentative proposed decision shall be issued as a tentative decision and the District shall be entitled to modify, deny or adopt the proposed decision. The proposed decision shall contain a recommendation regarding whether the Contractor should be debarred, and, if so, the appropriate length of time of the debarment. The Contractor and the District shall be provided an opportunity to object to the tentative proposed decision for a period of 15 days. If additional evidence is presented, the District shall evaluate this evidence and either issue an amended ruling, issue the same ruling, or call a further hearing.

If a Contractor has been debarred for a period of longer than five (5) years, that Contractor may after the debarment has been in effect for at least five (5) years, submit a written request for review of the debarment determination to reduce the period of debarment or terminate the debarment. The District may,

in its discretion, reduce the period of debarment or terminate the debarment if it finds that the Contractor has adequately demonstrated one or more of the following: (1) elimination of the grounds for which the debarment was imposed; (2) a bona fide change in ownership or management; (3) material evidence discovered after debarment was imposed; or (4) any other reason that is in the best interests of the District.

The District will consider a request for review of a debarment determination only where: (1) the Contractor has been debarred for a period longer than five (5) years; (2) the debarment has been in effect for at least five (5) years; and (3) the request is in writing, states one or more of the grounds for reduction of the debarment period or termination of the debarment, and includes supporting documentation. Upon receiving an appropriate request, the District will provide notice of the hearing on the request. At the hearing, the District shall review evidence on the proposed reduction of debarment period. This hearing shall be conducted and the request for review decided by the District pursuant to the same procedures as for a debarment hearing.

The District's proposed decision shall contain a recommendation on the request to reduce the period of debarment or terminate the debarment.

The terms shall also apply to Subcontractors of Contractor.

The following supplements modify the General Conditions. Where a portion of the General Conditions is modified and or deleted by these Supplementary Conditions, the unaltered portions of the General Conditions shall remain in effect.

## LABOR CODE SECTION 1771.4 [INCLUDE ONLY IF THE PROJECT IS SUBJECT TO A PLA, PSA OR SIMILAR AGREEMENT]

If the District has entered into a collective bargaining agreement that binds all contractors and subcontractors performing work on this project that includes a mechanism for resolving disputes about the payment of wages, the requirements set forth in Labor Code section 1771.49(a) do not apply on this Project.

#### ARTICLE 3 - THE CONTRACTOR

Article 3.10.4 Add the following: The Contractor shall require all Subcontractors to prepare and submit to the Contractor, within <u>fifteen (15)</u> days of execution of the Subcontract, comprehensive lists, in quadruplicate, of the manufacturers and products proposed for the Project, including information on materials, equipment, and fixtures required by the Contract Documents, as may be required for the Contractor's or Architect's approval.

#### **ARTICLE 8 – TIME**

Article 8.4.1 Liquidated Damages – Contractor will be liable to the District for Liquidated Damages pursuant to Article 8.4 for each calendar day of delay in the amount set forth in the Agreement Form.

#### ARTICLE 11 – INSURANCE AND BONDS

Article 11.11Performance and Payment Bonds – The number of executed copies of the Performance Bond and the Payment Bond required is three (3).

#### **ARTICLE 13**

This Project is subject to prevailing wages monitoring and enforcement by the Division of Industrial Relations (DIR) within the Division of Labor Standards Enforcement pursuant to Title 8, California Code of Regulations, and Section 16450 et seq.

The Contractor and all Subcontractors shall be required to furnish, at least monthly, electronic certified payroll records directly to the Labor Commissioner/ DIR in accordance with Title 8, California Code of Regulations, Section 16450 et seq. All payroll records shall be furnished in a format prescribed by Title 8, California Code of Regulations, and Section 16401. The Contractor and all Subcontractors must enroll in DIR's eCPR system to submit electronic certified payroll records. The District will have direct and immediate access to all CPRs for the Project that are submitted through the eCPR system. The District can use this information for any appropriate purpose, including monitoring compliance, identifying suspected violations, and responding to Public Records Act requests.

The DIR may conduct various compliance monitoring and enforcement activities including, but not limited to, confirming the accuracy of payroll records, conducting worker interviews, conducting audits, requiring submission of itemized statements prepared in accordance with Labor Code section 226, and conducting random in-person inspections of the Project site ("On-Site Visits"). On-Site Visits may include inspections

of records, inspections of the work site and observation of work activities, interviews of workers and others involved with the Project, and any other activities deemed necessary by the DIR to ensure compliance with prevailing wage requirements. The DIR shall have free access to any construction site or other place of labor and may obtain any information or statistics pertaining to the lawful duties of the Labor Commissioner.

Any lawful activities conducted or any requests made by the DIR shall not be the basis for any delays, claims, costs, damages or liability of any kind against the District by the Contractor. Contractor and all Subcontractors shall cooperate and comply with any lawful requests by the DIR. The failure of the DIR, the Division of Labor Standards Enforcement, or any other part of the Department of Industrial Relations to comply with any requirement imposed by the California Code of Regulations, Title 8, Chapter 8 shall not of itself constitute a defense to the failure to pay prevailing wages or to comply with any other obligation imposed by Division 2, Part 7, Chapter 1 of the Labor Code.

Prior to commencing any work on the Project, the Contractor shall post the notice/poster required under Title 8, California Code of Regulations, Section 16451(d) in both English and Spanish at a conspicuous, weatherproof area at the Project site. The required notice/poster is available on the DIR website, at the Division of Labor Standards Enforcement District Offices.

Division 1 Forms

## IMMEDIATE CONSTRUCTION CHANGE DIRECTIVE NO.

PROJECT:
TO:
You are hereby directed to provide the extra work necessary to comply with this ICD.
DESCRIPTION OF CHANGE:
COST (This cost shall not be exceeded):
TIME FOR COMPLETION:
NOTE:
Pursuant to Article 7.3.1.2 An Immediate Change Directive is a written order to the Contractor prepared by the Architect and signed by the District (and CM if there is a CM on the Project) and the Architect, directing a change in the Work and stating a proposed basis for adjustment, if any, in the Contract Sum or Contract Time, or both. The District may by ICD, without invalidating the Contract, direct immediate changes in the Work within the general scope of the Contract consisting of additions, deletions, or other revisions within If applicable, the Contract Sum and Contract Time will be adjusted accordingly. CONTRACTOR SHALL PROCEED WITH WORK SET FORTH IN THIS ICD IMMEDIATELY UPON RECEIPT OR THE DISTRICT MAY EITHER HOLD THE CONTRACTOR IN EITHER PARTIAL DEFAULT PURSUANT TO ARTICLE 2.2 OR TOTAL DEFAULT PURSUANT TO ARTICLE 14.
Architect
District

## CERTIFICATE OF SUBSTANTIAL COMPLETION

PROJECT:
TO:
As the Architect for the Project described above, the Project has reached Substantial Completion. Substantial Completion is not reached unless and until each of the following three (3) conditions have been met: (1) all contractually required items have been installed with the exception of only minor and Incomplete Punch Items (See Article 9.9 of the General Conditions); (2) All Fire/Life Safety Systems have been installed, and are working and signed off on the DSA Form 152 Inspection Card, all building systems including mechanical, electrical and plumbing are all functioning; and (3) the Project is fit for occupancy and its intended use
I certify that the Project has reached Substantial Completion as defined above on the following date:
·
Architect
Date